

P.O. Box 982 El Paso, Texas 79960-0982

May 31, 2023

The Honorable Kimberly D. Bose, Secretary Federal Energy Regulatory Commission 888 First Street, NE Washington, DC 20426

RE: El Paso Electric Company, Docket No. ER23-

El Paso Electric Company Interconnection Queue Proposed Tariff Revisions

Dear Secretary Bose:

Pursuant to Section 205 of the Federal Power Act ("FPA")¹ and Part 35 of the Federal Energy Regulatory Commission's ("Commission" or "FERC") Regulations,² El Paso Electric Company ("EPE") hereby submits proposed modifications to Attachment M of its Open Access Transmission Tariff ("OATT"). The modifications are necessary for EPE to improve and expedite its interconnection queue process. The revised tariff sheets are submitted in eTariff format, with proposed effective date on or before August 1, 2023.

I. COMMUNICATIONS

Communications regarding this filing should be sent to the following individuals:

Cynthia Henry
Vice President – General Counsel
El Paso Electric Company
P.O. Box 982
El Paso, TX 79960
Tel: (915) 351-4201
E-mail: cynthia.henry@epelectric.com

Milena Yordanova
Senior Attorney – Federal Regulation
El Paso Electric Company
P.O. Box 982
El Paso, TX 79960
Tel: (202) 302-7746
E-mail: milena.yordanova@epelectric.com

II. DESCRIPTION OF THE FILING PARTY

EPE is a vertically integrated electric utility whose primary business is serving native load in far west Texas and southern New Mexico. EPE provides service to about 455,000 customers in

_

¹ 16 U.S.C. § 824d.

² 18 C.F.R. Part 35 (2021).

an area of approximately 10,000 square miles. EPE owns distribution facilities through which it provides service to its customers at retail rates and owns transmission facilities over which it offers service under its OATT.

III. DESCRIPTION OF THE FILING

A. Need for Reform

EPE currently processes interconnection requests in a cluster manner, with spring and fall cluster windows, consistent with EPE's standard LGIP. When a higher-queued generator in a cluster modifies its project or withdraws from the queue, all equal-queued projects in the same cluster and all lower-queued projects may need to be restudied. Within EPE's area, not all generators requesting interconnection service have identified specific customers in need of supply; therefore, EPE expects that some but not all the generating projects making interconnection requests are likely to reach commercial operation.³ In EPE's experience, generation developers do not initiate construction before securing a contract for sale of the output of their facilities. As a result, some projects complete the study process and execute an LGIA but then request multiple extensions of the commercial operation date and/or place in suspension their LGIAs for up to three years, only to withdraw from the queue or resubmit the project for a new queue position when the suspension period expires. These projects interrupt the efficient processing of interconnection requests and delay ready-to-interconnect generation from advancing through the interconnection process.

EPE has determined that these issues require process improvement. They cannot be effectively solved by increased personnel. Although increased personnel might aid in the task of repeatedly performing re-studies, increased personnel would not prevent the interconnection delays being experienced by ready-to-interconnect projects as they sit in the queue behind projects that are higher-queued but not ready to move forward. With this filing, EPE transitions from a first-come, first-served interconnection cluster process to a first-ready, first-served cluster approach, recognized by the Commission as necessary and beneficial to both interconnection customers and transmission providers.⁴ EPE's proposed revisions are consistent with those

³ Other utilities have described similar problems with their first-come, first-served interconnection process. *See Public Service Company of Colorado, Revisions to Large Generator Interconnection Procedures*, Section Need for Reform, (Sept. 9, 2019) ("PSCo 2019 Interconnection Reform") (Explaining that PSCo serves approximately 6,900 MW of native load in its balancing authority area and it has over 22,000 MW of generation interconnection requests pending in its LGIP interconnection queue.).

⁴ Improvements to Generator Interconnection Procedures and Agreements, Notice of Proposed Rulemaking, 179 FERC ¶ 61,194, PP 37-39, P. 4, n 8 (2022) ("LGIA NOPR") (Stating "A first-ready, first-served cluster study process includes the following elements: increased access to information prior to entering the queue; a mechanism to study interconnection requests in groups; and increased financial commitments and readiness requirements to enter and proceed through the queue. To contrast, the existing first-come, first-served serial study process assigns interconnection requests an individual queue position based solely on the date of entry into the queue and does not

approved for similarly situated utilities⁵ and with the Commission's proposed vision to reform the LGIA by requiring, among other things, transmission providers to implement a first-ready, firstserved cluster study process and impose more stringent financial commitments and readiness requirements on those seeking interconnection, including increased study deposits, more stringent site control requirements, a commercial readiness framework, and effective withdrawal penalties.⁶

B. Summary of EPE's Proposal

A detailed list of all proposed revisions is included in Appendix A. Generally, EPE is proposing two study processes that will replace the existing interconnection system impact study: (1) the Informational Interconnection Study; and (2) the Definitive Interconnection Study Process. ⁷ The Informational Interconnection Study is intended for customers that are not yet ready to enter the interconnection queue and would like the opportunity to evaluate their project's interconnection feasibility prior to entering the interconnection queue. Therefore, these interconnection customers will not be assigned a queue position, and they will not be subject to the tariff's new readiness demonstrations and withdrawal penalties. In contrast, the Definitive Interconnection Study Process is intended for projects that are ready to move toward interconnection. Such projects are assigned a queue position and are then studied in a clustered Definitive Interconnection System Impact Study (DISIS) followed by a facilities study.

i. Informational Interconnection Study Process

The two study processes will provide more flexibility to interconnection customers who may be at varying stages of their project development. Prospective interconnection customers interested in the Informational Interconnection Study process will be able to request a reasonable number of Informational Interconnection Studies during the period from May 1st to November 1st

include access to information prior to entering the queue.) See also PP. 3, 39 (Preliminarily finding that "the Commission's pro forma LGIP and LGIA are unjust, unreasonable, unduly discriminatory, and preferential and that reforms are needed to allow interconnection customers to interconnect in a reliable, efficient, timely manner, thereby ensuring that rates, terms, and conditions for Commission-jurisdictional services remain just and reasonable and not unduly discriminatory or preferential.").

⁵ PSCo 2019 Interconnection Reform, Order on Tariff Filing, 169 FERC ¶ 61,182 (2019) (Approving PSCo's 2019 Interconnection Reform). EPE's proposal is based almost entirely on the elements of PSCo's 2019 interconnection queue reform that have been approved by the Commission. At this time, EPE is not proposing to include elements of PSCo's subsequent interconnection reform filed on December 14, 2022, and currently pending in Docket ER23-629-000.

⁶ LGIA NOPR P. 39. See also P. 6 (Explaining that "that transmission providers have undertaken efforts to address interconnection queue management issues. This NOPR is not intended to divert or slow the potential progress represented by those efforts. We will review any filings that result from those efforts based on the record before us in those proceedings and not based on whether they comply with the proposed reforms in this NOPR.").

⁷ Proposed LGIP, Definitions.

each year.⁸ An Informational Interconnection Study will consist of power flow and voltage analyses and a customer will be required to submit a separate Informational Interconnection Requests for a single site.⁹ The Informational Interconnection Study will allow the interconnection customer to study several interconnection scenarios and the effect of other clustered generation on a specific interconnection request.¹⁰ The Interconnection Studies are not queued, are for informational purposes only, and will be performed at the interconnection customer's expense.¹¹ Prior to conducting the Informational Interconnection Study, the interconnection customer and EPE will enter into an Informational Interconnection Study Agreement and provide a deposit for the study.¹² EPE will use reasonable efforts to complete the Informational Interconnection Study within seventy (70) calendar days from the date of the Informational Interconnection Study Agreement.¹³

ii. Definitive Interconnection System Impact Study Process

When a project is ready to interconnect, an interconnection customer must submit a valid interconnection request to be considered in a cluster under the DISIS queued process. Consistent with its existing practice, EPE proposes a DISIS Request Window that will be open semi-annually during a sixty (60) calendar day period. The spring window period will open on February 1st and close on March 31st, while the fall window period will open on August 1st and close on September 30th. Once the DISIS Request Window closes, EPE will work with the interconnection customers to test models, verify data, and generally prepare for the start of DISIS work. At the end of this process, all valid interconnection requests with an executed DISIS study agreement shall be included in that DISIS cluster.

⁸ Proposed LGIP, Section 6.1.

⁹ *Id*.

¹⁰ *Id. See also* Section 6.2.

¹¹ *Id.*, Section 6.2.

¹² Proposed LGIP, Section 6.1.

¹³ Proposed LGIP, Section 6.3.

¹⁴ Proposed LGIP, Section 4.2.

¹⁵ *Id*.

¹⁶ *Id*.

¹⁷ Id., Section 7.1. Further, Transmission Provider shall determine each interconnection customer's share of the cost of the DISIS cluster study by allocating the applicable study costs on a per capita basis based on number of interconnection requests included in the applicable cluster. The interconnection facilities study portion of the Definitive Interconnection Study Process is handled in the same manner. The interconnection facilities study is a

EPE will perform a DISIS to evaluate the impact of the proposed interconnections on the reliability of the transmission system.¹⁸ The Definitive Interconnection Study Process consists of up to three DISIS phases, as well as a subsequent interconnection facilities study phase. EPE's study work occurs in the following sequence: (1) an initial power-flow and voltage study expected to identify the majority of required network upgrades; (2) a stability and short circuit study, which is based on the traditional system impact study but adds stability and short circuit analysis to the power-flow analysis; and, if necessary, (3) a restudy if there are project withdrawal(s). 19 In order to be studied, interconnection customers will be required to provide, among other things: (1) a study deposit; (2) a completed application, including applicable technical information needed for modeling; (3) a demonstration of site control for in percentages that increase as an interconnection request moves through the study process; (4) a point of interconnection; (5) the point of delivery if the request is for network resource interconnection service; (6) the generating facility size in MW; and (7) readiness milestones.²⁰ The DISIS study report will provide a list of facilities that are required as a result of the interconnection request and a non-binding good faith estimate of cost responsibility and a non-binding good faith estimated time to construct.²¹ The definitive study process then continues to the interconnection facilities study phase, and leads to the submission of a Large Generator Interconnection Agreement.

To deter speculative projects from negatively affecting the queue, EPE is proposing a withdrawal penalty to be assessed if an interconnection customer withdraws in the definitive study process from the queue or the generating facility does not otherwise reach commercial operation, unless the withdrawal does not negatively affect the timing or cost of equal or lower queued projects. The penalties increase progressively as projects advance through the project milestones of development towards achieving commercial operation. ²² Specifically, if the customer provided a demonstration of readiness, then the customer's penalty assessment will be equal to the higher of the study deposit or one (1) times of its actual allocated cost of the Definitive Interconnection Study Process. ²³ If the customer did not provide a demonstration of readiness, the penalty will

cluster study, and study costs are allocated to the participating interconnection customers in the same way as in the DISIS cluster study.

¹⁸ Proposed LGIP, Section 7.3.

¹⁹ *Id*.

²⁰ Proposed LGIP, Sections 7.4, 7.7. Section 7.4 refers to an "At-A-Glance Reference Sheet," made part of the filing. It provides an overview and timeline of the Definitive Interconnection Study Process, including the phases and associated milestones.

²¹ Proposed LGIP, Section 7.3.

²² Proposed LGIP, Section 3.7.1.

²³ Proposed LGIP, Section 3.7.1.1.

depend on the phase in which the customer withdraws from the interconnection queue. For example, the penalties are capped at \$1 million for withdrawal at the initial stages of the project but gradually increase to up to \$2.5 million if withdrawal occurs during the facilities study phase of the interconnection process. Once the customer has an executed LGIA, the penalty is calculated as the higher of the study deposit or nine (9) times its actual allocated cost of the Definitive Interconnection Study Process.²⁴ Any withdrawal penalty revenues will be used to fund generation interconnection studies.²⁵

Given that only projects ready to move forward are expected to enter the queue, EPE does not expect that the withdrawal penalty provisions of the LGIP will be triggered often. The limits of the withdrawal penalties are consistent with limits previously approved by the Commission. Specifically, EPE is proposing penalty limits consistent with these approved for PSCo in 2019.²⁶

Finally, as it was true for the PSCo's LGIP reform approved by the Commission in 2019, EPE's proposal allows a resource planning entity to request the initiation of a Resource Solicitation Cluster.²⁷ The Resource Solicitation Cluster will respect queue positions and will have the same requirements as the DISIS cluster, with both being subject to the same Definitive Interconnection Study Process described above. Study results for a Resource Solicitation Cluster are provided to the Resource Planning Entity for use in its Resource Solicitation Process.²⁸

Accordingly, EPE is submitting a revised Attachment M (Large Generator Interconnection Procedures and Agreement) in clean eTariff format and in marked format. For a full list of the tariff changes, please refer to Appendix A.

IV. PUBLIC PROCESS

On March 27, 2023, EPE posted a notice informing interested parties that EPE is planning to propose revisions to its interconnection process and posted a draft of the proposed tariff

²⁴ *Id*.

²⁵ Proposed LGIP, Section 3.7.1.2.

²⁶ See PSCo 2019 Interconnection Reform. See also Public Service Company of Colorado, Revisions to Attachment N Large Generator Interconnection Procedures Response to Deficiency Notice and Amended Filing, Response to Deficiency Letter, at 2-3 (April 3, 2023) (Explaining the reasons for requesting an increase of the 2019 Commission-approved withdrawal penalty levels.).

²⁷ Proposed LGIP, Section 4.2.1.

²⁸ *Id*.

revisions.²⁹ On April 6, 2023, EPE held a public meeting, providing an overview of the project and answering questions from stakeholders. In the weeks that followed the stakeholder meeting, EPE received limited written comments and responded to a few questions submitted by stakeholders.30

V. **EFFECTIVE DATE**

EPE is requesting that Attachment M becomes effective on or before August 1, 2023 when the proposed fall application window begins. This will allow EPE and interconnection customers to prepare for the timely implementation of the proposal and will facilitate the transition to the new requirements.

VI. **CONCLUSION**

Pursuant to Section 205 of the FPA and Part 35 of the Commission's Regulations, EPE submits the instant filing to revise Attachment M of its OATT with an effective date as described above. Any correspondence or other communications concerning this filing may be directed to the undersigned at the mailing address shown in the letterhead.

Respectfully submitted,

/s/**Milena Yordanova** /

Milena Yordanova Senior Attorney – Federal Regulation El Paso Electric Company milena.yordanova@epelectric.com

²⁹ See EPE's OASIS website at https://www.oasis.oati.com/epe/index.html. A redlined version of the proposed tariff was posted in folder "Generation Interconnection/Generation Interconnection Queue."

³⁰ Id. See folder "Generation Interconnection/Stakeholder Meeting."

TABLE OF CONTENTS	TABLE OF CONTENTS	
Table of Contents	The Table of Content is being revised to reflect changes in Sections 3.7, 4, 5, 6, 7, 10, 13, Appendix 2, 4, and 5.	
Section 1 - Definitions		
Cluster	Adding a definition for Cluster: "Cluster shall mean a group of Interconnection Requests (one or more) that are studied together for the purpose of conducting Interconnection Studies."	
Cluster Study	Adding a definition for Cluster Study: "Cluster Study shall mean an Interconnection Study evaluating one or more Interconnection Requests."	
Clustering	Updating the definition of Clustering by replacing "Interconnection System Impact Study" with "Interconnection Studies."	
Control Area	Updating the last two sentences sentence to clarify that "A Control Area must be certified by an Applicable NERC Regional Reliability Entity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC."	
Customer Engagement Window	Adding a definition for Customer Engagement Window: "Customer Engagement Window shall have the meaning set forth in Section 4.2.1 of the LGIP."	
Definitive Interconnection Study Process	Adding a definition for Definitive Interconnection Study Process: "Definitive Interconnection Study Process ("Definitive Interconnection Study") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study."	
Definitive Interconnection System Impact Study	Adding the definition for Definitive Interconnection System Impact Study: Definitive Interconnection System Impact Study ("DISIS") shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System."	
Definitive Interconnection System Impact Study Agreement	Adding a definition for Definitive Interconnection System Impact Study Agreement: Definitive Interconnection System Impact Study Agreement ("DISIS Agreement") shall mean the form of agreement	

	1: 1: 4 1: 0 04 LODG 1 1: 4 D 7:11
	contained in Appendix 2 of the LGIP for conducting the Definitive Interconnection System Impact Study."
Definitive Interconnection	Adding a definition for Definitive Interconnection System Impact
System Impact Study Cluster	Study Cluster: "Definitive Interconnection System Impact Study
System Impact Stady Claster	Cluster ("DISIS Cluster") shall mean an engineering study that
	evaluates the impact of the proposed interconnection(s) on the safety
	and reliability of the Transmission System and, if applicable, an
	Affected System."
DISIS Request Window	Adding a definition for DISIS Request Window: "DISIS Request
•	Window shall have the meaning set forth in Section 4.2.1 of the
	LGIP."
Deposit in Lieu of Site Control	Removing the definition for Deposit in Lieu of Site Control.
Financial Security	Adding a definition for Financial Security: "Financial Security shall
	have the meaning set forth in Section 7.7.1 of the LGIP."
Informational Interconnection	Adding a definition for Informational Interconnection Study:
Study	"Informational Interconnection Study shall mean a System Impact
,	Study analysis based on generation and POI assumptions specified by
	Interconnection Customer in the Informational Interconnection Study
	Agreement. The informational study shall be based on an approved
	WECC base case."
Informational Interconnection	Adding a definition for Informational Interconnection Study
Study Agreement	Agreement: "Informational Interconnection Study Agreement shall
	mean the form of agreement contained in Appendix 5 of the LGIP for
	conducting the Informational Interconnection Study."
Interconnection Facilities	Removing from the definition: "Interconnection Facilities may be
Internation Facilities Chade	shared by more than one Generating Facility in a Cluster."
Interconnection Facilities Study	Updating "Interconnection System Impact Study" to "Definitive
Interconnection St. 1	Interconnection System Impact Study."
Interconnection Study	Updating the definition to add Informational Interconnection
	Study" and to add "Definitive" to "Interconnection System
7	Impact Study."
Interconnection Study	Updating the definition to add Informational Interconnection
Agreement	Study" and to add "Definitive" to "Interconnection System
	Impact Study."
Interconnection System Impact Study	Removing the definition Interconnection System Impact Study.
Interconnection System Impact	Removing the definition Interconnection System Impact Study
Study Agreement	Agreement.
-	

Large Generating Facility	Adding to the definition: "The terms, Large Generating Facility and
	Generating Facility, are used interchangeably to mean any
	Generating Facility that is subject to the LGIP."
Material Modification	Updating the definition to clarify that modifications are those
	modifications that have a material impact on the cost or timing
	of any Interconnection Request with an equal or later Queue
	Position rather than priority date.
OASIS	Adding a definition of OASIS: "OASIS shall mean the Transmission
	Provider's Open Access Same-Time Information System."
Optional Interconnection Study	Removing the definition Optional Interconnection Study.
Optional Interconnection Study	Removing the definition Optional Interconnection Study Agreement.
Agreement Phase	Adding a definition for Phase "Phase ("Phase 1 Phase 2 Phase 2 or
rnase	Adding a definition for Phase: "Phase ("Phase 1, Phase 2, Phase 3 or Phase 4") shall mean a distinct part of the Definitive Study Process
	as described in Section 7."
Queue	Adding a definition for Queue: "Queue shall mean a queue for valid
Queue	Interconnection Requests for the Definitive Interconnection Study
	Process."
Queue Position	Adding to the definition for Queue Position: "in the Definitive
Quodo I obidion	Interconnection Study Process. The Queue Position" and "Customer
	satisfied all of the requirements of this Attachment M to enter the
	Definitive Study Process" and removing "that" "of receipt of"
	"valid" and "Request by the Transmission Provider."
Resource Plan	Adding a definition for Resource Plan: "Resource Plan shall mean
	any process authorized or required by Applicable Laws and
	Regulations for, <i>inter alia</i> , the selection of Generating Facilities."
Resource Planning Entity	Adding a definition for Resource Planning Entity: "Resource
	Planning Entity shall mean any entity required to develop a Resource
	Plan or conduct a Resource Solicitation Process."
Resource Solicitation Cluster	Adding a definition for Resource Solicitation Cluster: "Resource
	Solicitation Cluster shall mean a Cluster Study associated with a
	Resource Plan or related process."
Resource Solicitation Process	Adding a definition for Resource Solicitation Process: "Resource
	Solicitation Process shall mean any process authorized or required by
	Applicable Laws and Regulations for the acquisition of Network
	Resources."
Scoping Meeting	Adding to the definition: "the proposed interconnection request," and "affect" and removing "impact".
Site Control	Adding to the definition: "the exclusive land right to develop,
	construct, operate, ad maintain the Generating Facility over the term
	of expected operation of the Generating Facility. Site Control shall

	include the right to develop, construct, operate and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by" "establishing" "of sufficient sized to construct and operate" "and associated Interconnection Facilities" "interest in a" "of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities" "any other documentation that clearly demonstrates the right of the" "to exclusively occupy a site of sufficient size to construct and operate the Generating Facility" and "Site Control f or any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all co-located projects that meet the aforementioned provisions of this Site Control definition." Removing "reasonably demonstrating" "for the purpose of constructing" "an exclusivity or other business relationship between Interconnection Customer" and "and the entity having the right to sell, lease or grant Interconnection Customer the right to possess or occupy a site for such purpose."		
Withdrawal Penalty	Adding the definition Withdrawal Penalty: "Withdrawal Penalty shall have the meaning set forth in Section 3.7.1 of the LGIP."		
Section 2. Scope and Applicati	· ·		
2.1 - Application of Standard Large Generator Interconnection Procedures	Section 2.1 - Application of Standard Large Generator Interconnection Procedures is being revised to add "Small Generating Facilities requesting NRIS shall be processed under the LGIP."		
2.3 Base Case Data	Capitalized Interconnection Study.		
Section 3. Interconnection Rec	Section 3. Interconnection Requests		
3.1 - General	Adding "Definitive" and "Process". Removing "required" and changing "Studies to Study". Changed "Point(s)" to "Point. Adding "Definitive" and "For purposes of clustering Interconnection Service requests, Transmission Provider may make reasonable changes to the requested Point(s) of Interconnection to facilitate efficient interconnection of Interconnection Customers at common points of interconnection. Transmission Provider shall notify Interconnection Customers in writing of any intended changes to the requested		

Filed Date: 05/31/2023

Appendix A

	Point(s) of Interconnection and the Point(s) of Interconnection shall only change upon mutual agreement."
	Removing "Where an Interconnection Customer has submitted multiple alternative Interconnection Requests (seeking the study of either multiple alternative projects, or the study of a single project at multiple alternative Points of Interconnection), and the Scoping Meeting fails to result in a single definitive project at a single definitive Point of Interconnection, the Transmission Provider will seek to accommodate such multiple alternative requests, as follows: • The System Impact Study will be performed in serial fashion to produce a separate version of the study for each of the Interconnection Customer's alternative requests. • The cost of the first version of the study will be allocated equally among the participants of the System Impact Study Cluster. The full cost of producing each subsequent alternative version of the study will be paid by the Interconnection Customer pursuing the alternative request(s). • No more than three alternative Interconnection Requests will be permitted in any single study cluster. • After receiving the results of the System Impact Study (in its multiple serial versions), the Interconnection Customer must designate a single Interconnection Request in order to be eligible to proceed to the Facility Study phase. Alternative Interconnection Requests will not be accommodated in the Facility Study cluster."
3.2 – Identification of Types of Interconnection Services	Adding "below" "may designate only one type of" "each separate Interconnection Request in the Queue. The Type of" "must be finalized on submission of the executed Definitive System Impact" and "and may only be changed after the start of the Definitive Study Process between Phase 2 and Phase 3 of the Definitive Interconnection Study Process and only if a Cluster must be restudied in Phase 3 and otherwise may not be changed" while deleting "provided, however, any" "requesting Network Resource" "may also request that it be concurrently studied" "Energy Resource" "up to the point when an Interconnection Facility" "is executed" and "Interconnection Customer may then elect to proceed with Network Resource Interconnection Service or to proceed under a lower level of interconnection service to the extent that only certain upgrades will be completed."

3.2.2.1 – The Product	Adding "If the Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resource within Transmission Provider's Control Area, the Interconnection Customer must provide the point of delivery or the geographic location on EPE's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area."
3.2.2.2 – The Study	Adding "output" and "is" while deleting "are."
3.4.1 – Initiating an Interconnection Request	Adding "An Interconnection Customer wishing to join the Definitive Interconnection Study Process shall submit, and complete its Interconnection Request to Transmission Provider within, and no later than the close of the DISIS Request Window."
3.4.1 – Initiating an Interconnection Request	Adding "An Interconnection Customer wishing to join the Definitive Interconnection Study Process shall submit a completed Interconnection Request to Transmission Provider within, and no later than the close of, the DISIS Request Window." Deleting "In the event the Interconnection Request is withdrawn or deemed withdrawn before the Start Date, \$25,000 of the Initial Deposit will be non-refundable. If the Interconnection Customer has not demonstrated Site Control as of the Start Date, \$25,000 of the Deposit in Lieu of Site Control shall be non-refundable." And "In the event the Interconnection Request is withdrawn or deemed withdrawn after the execution of the Facilities Study Agreement, in accordance with Section 8.1, \$50,000 of the Initial Deposit will be non-refundable. If the Interconnection Customer has not demonstrated Site Control at the time of this withdrawal, \$50,000 of the Deposit in Lieu of Site Control shall be non-refundable." And "These non-refundable amounts shall not be cumulative and shall be in addition to the actual costs incurred by Transmission Provider up to the date of the withdrawal."
3.4.1 (ii)	Capitalize A, add "(including all technical information);" and delete "and."
3.4.1 (iii)	Capitalize A, add "as defined in Section 1 and addressed further in Section 7.7.7 of the LGIP" and "Specifications for acceptable site size for the purposes of demonstrating Site Control are posted on Transmission Provider's OASIS website. Interconnection Customer may propose alternative specifications for site size to those posted on OASIS for Transmission Provider approval. In the event Transmission Provider and Interconnection Customer cannot reach

	agreement related to adequacy of site size, Transmission Provider will accept a Professional Engineer (licensed in the state of Texas or New Mexico, as applicable depending on the state in which the Generating Facility is to be located) stamped site plan drawing that depicts the proposed generation arrangement and specifies the maximum facility output for that arrangement" and delete "or a Deposit in Lieu of Site Control in the amount of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater. Such Deposits shall be applied toward any Interconnection Studies pursuant to the Interconnection Request."
3.4.1 (iv)	Adding (iv) A Point of Interconnection.
3.4.1 (v)	Adding (v) "If the request is for NRIS and if Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resourced within Transmission Provider's Control Area, the point of delivery or the geographic location on Transmission Provider's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area."
3.4.1 (vi)	Adding (vi) "A Generating Facility size (MW) (and requested Interconnection Service amount if the requested Interconnection Service is less than the Generating Facility Capacity)."
3.4.1 (vii)	Adding (vii) "One of the following Readiness Milestone ("M1") options totaling the entire capacity of the Generating Facility (or requested Interconnection Service amount if the requested Interconnection Service amount is less than the Generating Facility Capacity or security equal to one times the study deposit described above (see Initial Deposit) in the form of an irrevocable letter of credit or cash <i>in lieu of</i> the Readiness Milestone. The security is refunded to the Interconnection Customer according to Section 7.7.5. a. Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale (1) of the constructed Generating Facility, or (2) of the Generating Facility's energy, or (3) of the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
	b. Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or

	c. Provisional Large Generator Interconnection Agreement filed with FERC that contains a commitment to move forward with constructing the Generating Facility and is not suspended"
3.4.1 (viii)	Adding (viii) "Security equal to one times the study Initial Deposit in the form of an irrevocable letter of credit or cash. The security is refunded to the Interconnection Customer according to Section 7.7."
3.4.2 - Acknowledgement of Interconnection Request	Adding "the close" and "DISIS Request Window" and deleting "receipt" and "request."
3.4.3 - Deficiencies in Interconnection Request	Adding "the close" "DISIS" "Window" "At any time, during the studies, if Transmission Provider identifies issues with technical data provided by Interconnection Customer, Interconnection Customer and Transmission Provider shall work expeditiously and in good faith to remedy any data issues." and "Transmission Provider shall determine if the information contained in the Interconnection Request is adequately sufficient to start the Definitive System Impact Study by the close of the Customer Engagement Window." Deleting "receipt" and "initial Interconnection."
3.4.4 – Scoping Meeting	Adding "the close" "the DISIS" "Window" "host an open Scoping Meeting for all Interconnection Requests received in that DISIS Request Window. If requested by" "Transmission Provider" "also hold individual customer specific Scoping Meetings, which must" "requested" "fifteen (15) business days after" "close of the DISIS Request Window" and "pursuant to Section 7.2". Deleting "receipt" "a valid Interconnection" "establish a date agreeable to" "for the Scoping Meeting, and such date" "thirty (30) Calendar Days from receipt of the valid Interconnection Request, unless otherwise mutually agreed upon by" "Parties" and "and one or more available alternative Point(s) of Interconnection."
3.5.1 – OASIS Posting	Deleting "and" and "Interconnection Study reports and Optional."
3.5.2.1(A) - Interconnection	Adding "Phase 1" and "and the number of Interconnection Requests
System Impact Studies	that had Phase 2 and Phase 3 Studies completed within Transmission
processing time	Provider's coordinated region during the reporting quarter," while deleting "Interconnection System Impact."
3.5.2.1(B) - Interconnection System Impact Studies processing time	Adding "Phase 1" "System" and "and the number of Interconnection Requests that had Phase 2 and Phase 3 Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than one hundred fifty (150) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact

	Study Agreements," while deleting "Interconnection System Impact."
3.5.2.1(C) - Interconnection System Impact Studies processing time	Adding in "Definitive" "one hundred fifty (150)" and "one hundred fifty (150)" while deleting "ninety (90)" and "ninety (90)."
3.5.2.1(D) - Interconnection System Impact Studies processing time	Adding "Definitive."
3.5.4(i)	Adding in "150" and "90" while deleting "45" and "180."
3.6 – Coordination with Affected Systems	Adding "It is the responsibility of the Affected System Owner to provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to (i) complete any interconnection studies and (ii) construct any necessary interconnection facilities and network upgrades needed to reliably interconnect at the requested service level."
3.7 – Withdrawal	Deleting "an" and "or study payments" while adding "In the case of a withdrawal," "(ii) impose the Withdrawal Penalty described in Section 3.7.1; (iii) refund any security after settling the final invoice (see Section 7.7.5);" "of the refundable" "study" and "share of the."
3.7.1 – Withdrawal Penalty	Adding new subsection 3.7.1, adding "Interconnection Customers shall be subject to a Withdrawal Penalty if they withdraw their request from the Queue or the Generating Facility does not otherwise reach Commercial Operation unless the withdrawal does not negatively affect the timing or cost of equal or lower queued projects."
3.7.1.1 – Calculation of the Withdrawal Penalty	Adding subsection 3.7.1.1, adding "If the Interconnection Customer provided a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be equal to the higher of the study deposit or one (1) times of its actual allocated cost of the Definitive Interconnection Study Process. If the Interconnection Customer did not provide a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be dependent on the Phase in which the Interconnection Customer withdraws and shall be calculated as follows: 1. If the Interconnection Customer withdraws in Phase 1 (after M1, but before M2), the Withdrawal Penalty shall be the higher of the study deposit or two (2) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one (1) million dollars. 2. If the Interconnection Customer withdraws in Phase 2 (after M2, but before M3), the Withdrawal Penalty shall be the higher of the study deposit

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

	or three (3) times its actual allocated cost of the Definitive
	Interconnection Study Process. This amount shall be capped at one
	and one half (1.5) million dollars. 3. If the Interconnection Customer
	withdraws in Phase 3 (after M3, but before M4), the study cost
	obligation shall be the higher of the study deposit or five (5) times its
	actual allocated cost of the Definitive Interconnection Study Process.
	This amount shall be capped at two (2) million dollars. 4. If the
	Interconnection Customer withdraws in Phase 4 (after M4, but before
	M5), the Withdrawal Penalty shall be the higher of the study deposit
	or seven (7) times its actual allocated cost of the Definitive
	Interconnection Study Process. This amount shall be capped at two and a half (2.5) million dollars.
	The Withdrawal Penalty for any customer that has executed an LGIA
	is the higher of the study deposit or nine (9) times its actual allocated
	cost of the Definitive Interconnection Study Process."
3.7.1.2 Distribution of the	Adding subsection 3.7.1.2, adding "Any Withdrawal Penalty
Withdrawal Penalty	revenues shall be used to fund generation interconnection studies.
	Withdrawal Penalty revenues shall first be applied, in the form of a
	bill credit, to not-yet-invoiced study costs for other Interconnection
	Customers in the same cluster, and to the extent that such studies are
	fully credited, shall be applied to study costs of future clusters in
	Queue order. Withdrawn Interconnection Customers shall not receive
	a bill credit associated with Withdrawal Penalties. Distribution of
	Withdrawal Penalty revenues to a specific study shall not exceed the
	total actual study costs. Allocation of Withdrawal Penalty revenues
	within a cluster to a specific customer shall be comparable to the
	allocation of study costs described in Section 4.2. Specifically, the
	Withdrawal Penalty revenue distribution to each customer in a
	specific cluster, shall be (1) fifty percent (50%) on a per capita basis
	based on number of Interconnection Requests in the applicable
	Cluster; and (2) fifty percent (50%) to Interconnection Customers on
	a pro-rata basis based on requested megawatts included in the
	applicable Cluster. Distribution of Withdrawal Penalty revenue
	associated with Readiness Milestone 5 shall not be distributed to the
	remaining customers in that cluster until all customers in that cluster
	have reached Commercial Operation and thereafter shall be distributed as described above. Transmission Provider shall not
	change the distribution of Withdrawal Penalty revenue without
	authorization by the Commission. Transmission Provider shall post
	the Withdrawal Penalty balance on its OASIS site."
	one of the area of the control of th

Section 4. Interconnection Request Evaluation Process

Section 4.	Changed the name from "Queue Position" to "Interconnection Request Evaluation Process."
Section 4.1	Changed the name from "General" to "Queue Position". Created subsections 4.1 and 4.2 incorporating and modifying previous 4.1 text.
Section 4.1.1 Assignment of Queue Position	Adding "as follows: the Queue Position within the Queue shall be assigned" "all items" "pursuant" "the provisions of Section 3.4. There is no queue for informational," and deleting "the valid Interconnection Request; provided that, if the sole reason an Interconnection Request is not valid is the lack of" "information on the application form, and Interconnection Customer provides such information in accordance with Section 3.4.3, then Transmission Provider shall assign Interconnection Customer a Queue Position based on the date the application form was originally filed. Moving a Point of Interconnection shall result in a lowering of Queue Position if it is deemed a Material Modification under Section 4.4.3." "determine the order of performing the" and "and determination of cost responsibility for the facilities necessary to accommodate the Interconnection Request."
4.1.2 – Higher Queue Position	Adding "Queue Position assigned to an" and "assigned a", capitalized "Queue" and deleting "queued" "queued" and "Transmission Provider may allocate the cost of the common upgrades for clustered Interconnection Requests without regard to Queue Position."
4.2 - Clustering	Adding "shall be considered equally queued but Clusters initiated earlier in time shall be considered to have a higher Queue Position than clusters initiated later. The Queue Position of" "Request shall have no bearing on the allocation of the cost of the upgrades identified in the applicable" "Study (such costs will be allocated among Interconnection Requests" and "4.2.2. Moving a Point of Interconnection shall result in a loss of Queue Position if it is deemed a Material Modification under Section 4.4.37.4" while deleting "Interconnection Requests are to be studied in clusters for the purpose of the Interconnection System Impact Study and the Interconnection Facilities Study. Clustering shall be implemented on the basis of" "and geographic location of the proposed Interconnection Point on the Transmission Provider's Transmission System and/or the interconnecting voltage

	level. All Interconnection Requests received during the second and third quarters of a given year (i.e., beginning April 1 and closing September 30) will be grouped into one "Queue" "Window," and all Interconnection Requests received during the fourth quarter of a year and the first quarter of the following year (i.e., beginning October 1 and closing March 31 of the following year) will be placed into a separate "Queue Cluster Window." Interconnection Requests shall be grouped in their respective Queue Cluster Window and by geographical areas and/or the interconnecting voltage level, and shall be studied together for Network Resource Interconnection Service without regard to the nature of the requested Interconnection Service, whether Energy Resource Interconnection Service or Network Resource Interconnection Service. The deadline for completing all Interconnection System Impact Studies for which" "System Impact Study Agreement has been executed during a Queue" "Window shall be" and "4.4.37.4, for all Interconnection Requests assigned to the same Queue Cluster Window. Transmission Provider may study an Interconnection Request separately to the extent warranted by Good Utility Practice based upon the electrical remoteness of the proposed Large Generating Facility. Any changes to the established Queue Cluster Window interval and opening or closing dates shall be announced with a posting on Transmission Provider's OASIS beginning at least one hundred and eighty (180) Calendar Days in advance of the change and continuing
	thereafter through the end date of the first Queue Cluster Window that is to be modified."
4.2 - Clustering and the Definitive Study Process	Added "The diagram attached in the At-A-Glance Reference Sheet provides an overview of the Definitive Interconnection Study Process. Transmission Provider shall accept Interconnection Requests during a (60) Calendar Day period referred to as the "DISIS Request Window." DISIS Request Window shall open annually on February 1st and close March 31th or the following Business Day if March 30th falls on a weekend or NERC recognized holiday. A second DISIS Request Wind shall open annually on August 1st and close on September 30th or the following Business Day if September 30th falls on a Saturday or Sunda If one or more valid requests are received, for sixty (60) Calendar Days following the close of the DISIS Request Window (the "Customer Engagement Window"), Transmission Provider shall work with applicable Interconnection Customers to test models, verify data, hold stakeholder meetings (including Scoping Meetings,

4.2.1 – Cluster Window

Transition Period

Appendix A

as appropriate), work with requestors to address any modeling irregularities identified by Transmission Provider in its evaluation of the Interconnection Request, and generally prepare for the start of the Definitive Interconnection System Impact Study" and "Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable study costs to Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study is a cluster study, and s	
the Interconnection Request, and generally prepare for the start of the Definitive Interconnection System Impact Study" and "Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable cluster. The Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study c	as appropriate), work with requestors to address any modeling
Definitive Interconnection System Impact Study" and "Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable cluster. The Interconnection Requests included in the applicable cluster. The Interconnection Requests included in the paplicable cluster. The Interconnection Requests included in the participating Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are al	irregularities identified by Transmission Provider in its evaluation of
"Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first	the Interconnection Request, and generally prepare for the start of the
all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window de	Definitive Interconnection System Impact Study" and
all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window de	"Notwithstanding the preceding sentence and upon written consent of
Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Requests: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Wi	9 1 9
to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows."	<u> </u>
Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Study Process is handled in the same manner. The Interconnection Customers in the same way as in the DISIS Cluster study' while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	` '
list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Realities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows."	•
amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows."	<u> </u>
state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	· ·
(vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	, , , , , , , , , , , , , , , , , , , ,
type such as wind, natural gas, coal, or solar. At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \
At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	9 9
Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	•
the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	·
in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	• • •
Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
Definitive Interconnection System Impact Study described in more detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
detail in Section 7. Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	· · · · · · · · · · · · · · · · · · ·
Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	·
applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	-
manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	•
participating Interconnection Customers in the same way as in the DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
DISIS Cluster study" while deleting "The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	•
Window will commence upon the first Window deadline following Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
Commission approval of the Queue Cluster Windows." Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	=
Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering	
	 Commission approval of the Queue Cluster Windows."
and the Definitive Study Process.	Subheading 4.2.1 is deleted and the text now part of 4.2 – Clustering
	 and the Definitive Study Process.

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

4.2.1 – Initiation of a Resource Solicitation Cluster

Added subsection 4.2.1 and "Upon request of a Resource Planning Entity, Transmission Provider may initiate the study of a Resource Solicitation Cluster. The Resource Solicitation Cluster shall respect Queue Position and shall be studied as its own Cluster. Within ten (10) Business Days of receipt of a request to perform a Resource Solicitation Cluster that includes valid Interconnection Requests as described in Section 3.4, Transmission Provider and Resource Planning Entity shall meet to determine a mutually agreeable scope of study and timeframe to initiate the Resource Solicitation Cluster. The timeline shall indicate the close of the Customer Engagement Window for that Resource Solicitation Cluster. Thereafter the Definitive Interconnection System Impact Study shall proceed as described in Section 7.

In order to initiate Transmission Provider's study of Interconnection Requests made in connection with a Resource Solicitation Process Resource Planning Entity must: (a) act as the authorized representative for all Interconnection Requests submitted to the Resource Solicitation Cluster; (b) submit all Interconnection Requests arising from the Resource Solicitation Process at the same time to ensure an equal Queue Position for all Generating Facilities included in the Resource Solicitation Cluster; (c) cooperate with Transmission Provider in conducting the studies; and (d) request a reasonable number of different combinations of such Interconnection Requests to meet Resource Planning Entity's identified need and assumptions in the Resource Solicitation Process. Such studies in connection with a Resource Solicitation Process shall be implemented based upon Queue Position (relative to higher or lower queued clusters) and shall consider Resource Planning Entity's needs and assumptions identified in the Resource Solicitation Process. Transmission Provider shall not be obligated to implement a Resource Solicitation Cluster more than once a year. The Resource Planning Entity may submit for inclusion in the Resource Solicitation Process an Interconnection Request for a Generating Facility that already has a higher Queue Position pursuant to Section 4.1.1. A Generating Facility that initially is associated with a Queue Position through the Resource Solicitation Process may also reserve a lower Queue Position separate from the Resource Solicitation Process pursuant to Section 4.1.1. In either case, Interconnection Customer must meet all requirements associated with maintaining each Queue Position for the Generating Facility. In the event a Generating Facility has multiple Queue Positions, it shall not be double counted in the study models.

	A Generating Facility in the Resource Solicitation Process is subject to study according to the Queue Position of the Resource Solicitation Cluster. A Generating Facility that is not a part of the Resource Solicitation Process is also subject to study according to its Queue Position. All studies must be performed in accordance with the provisions of the LGIP, and may not be delayed as a result of the Resource Solicitation Process. After Transmission Provider completes the Definitive Interconnection System Impact Studies for the requested combinations, the results will be provided (Phase 1 Reports, Phase 2 Reports, Phase 3 Reports, etc.; as applicable under Section 7.4) to the Resource Planning Entity for use in the Resource Solicitation Process. The results will be posted on Transmission Provider's OASIS consistent with the posting of other study results. After receipt of the Phase 1 Report, Resource Planning Entity must select one of the studied combinations to advance to Phase 2 of the study process. After receipt of the Phase 2 Report, Resource Planning Entity must select one of the studied combinations prior to the commencement of any Interconnection Facilities Study associated with the Resource Solicitation Process. Prior to the completion of the Interconnection Facilities Study of all of the components of the selected combination, Resource Planning Entity may replace components, subject to any necessary Re-Study pursuant to Sections 7.6 or 8.5. While conducting the Definitive Interconnection Study Process, Transmission Provider may suspend further action on the Interconnection Requests in the Resource Solicitation Process that are not included in the selected combination. Once a Generating Facility is rejected in the Resource Solicitation Process, the Generating Facility is selected by Resource Planning Entity at the conclusion of the Resource Solicitation Process, the Generating Facility may no longer maintain more than one Queue Position."
4.4 – Modifications	Adding in "Subject to the foregoing sentence, and provided, however, they do not result in a material modification," and "and
	potentially impacted Interconnection Customers in the same Cluster," while deleting "and."
4.4.1	Adding in "No later than forty (40) Calendar Days after the close of the DISIS Request Window and" "Definitive" and "the."
4.4.2	Changing "Facility" to "Facilities."
4.4.3	Deleted "7.2."
4.4.4	Added in "the modification of."

	Added in "The initial requested Commercial Operation Date used for this calculation is determined from the date proposed in the initial Interconnection Request. Such cumulative extensions are inclusive of extensions requested after execution by Interconnection Customer of the LGIA." while deleting "Large." connection Requests Submitted Prior to Effective Date of Standard
Large Generator Interconnecti	
5.1	Changed the name from "Queue Position for Pending Requests" to "Transition to the First-Ready, First-Served Process."
5.1.1	Added in "subject to the requirements set forth in Sections 5.1.1.1 and 5.1.1.2. An Interconnection Customer that fails to meet these requirements shall have its Interconnection Request deemed withdrawn pursuant to Section 3.7, without prejudice to the Interconnection Customer's ability to request an Informational Interconnection Study for which there is no Queue Position. Any unused deposit amounts of withdrawn Interconnection Requests shall be returned pursuant to Section 3.7. If an Interconnection Customer elects to continue with an Informational Interconnection Study or alternatively with a Transitional Interconnection Facilities Study or a Transitional Cluster Study as described below, Transmission Provider shall retain the current study deposits, and Interconnection Customer shall be responsible for the entire cost of all studies
	pursuant to Sections 6 and 4."
5.1.1.1 – Transitional Projects	Added name "Transitional Projects" to the subsection, and added "An Interconnection Customer that has a) a final System Impact Study Report that identifies facilities required to feasibly interconnect and b) an Interconnection Facilities Study Agreement that was executed prior to the effective date of this Section 5 may opt to continue with the Interconnection Facilities Study process if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Interconnection Facilities Study Agreement in the form of Appendix 4.1 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All of the following are required: a) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades identified in the System Impact Study Report. The deposit shall be equal to one hundred percent (100%) of the costs identified for Transmission Provider's Interconnection Facilities and Network Upgrades in the System Impact Study Report and will be reconciled to actual costs after the associated facilities are in service. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied

Filed Date: 05/31/2023

Appendix A

	towards future construction costs described in the LGIA. If the
	Interconnection Customer withdraws or otherwise does not reach
	Commercial Operation, the deposit is fully refundable once the final
	invoice for study costs and Withdrawal Penalty is settled. The deposit
	shall be in the form of an irrevocable letter of credit upon which the
	Transmission Provider may draw or cash.
	b) Exclusive Site Control for the entire Generating Facility and any
	Interconnection Customer's Interconnection Facilities.
	c) Interconnection Customer shall provide one following:
	i. A contract, binding upon the parties to the contract, for sale of the
	Generating Facility's energy, or the entire constructed Generating
	Facility; where the term of sale is not less than five (5) years, or
	ii. Reasonable evidence that the Generating Facility is included in an
	approved Resource Plan or Resource Solicitation Process, or
	iii. An executed Provisional Large Generator Interconnection
	Agreement filed with FERC. Such an agreement shall not be
	suspended and shall include a commitment to construct the
	Generating Facility.
	All LGIA negotiations shall be completed and the LGIA executed (or
	filed unexecuted) within sixty (60) Calendar Days of the publication
	of the final Interconnection Facilities Study Report or the
	Interconnection Request shall be deemed withdrawn pursuant to
	Section 3.7 unless extended by mutual agreement of Transmission
	Provider and Interconnection Customer. A change in the Commercial
	Operation Date shall not delay the construction of facilities if such
	delay negatively affects lower or equal queued projects. If the
	Interconnection Customer withdraws or otherwise does not reach
	Commercial Operation, a Withdrawal Penalty equal to nine (9) times
	the Interconnection Customer's total study cost is imposed." While
	deleting "If an Interconnection Study Agreement has not been
	executed as of the effective date of this LGIP, then such
	Interconnection Study, and any subsequent Interconnection Studies,
	shall be processed in accordance with this LGIP."
5.1.1.2 - Combined System	Added name "Combined System Impact and Interconnection
Impact and Interconnection	Facilities Transitional Cluster Study" to subsection and added "An
Facilities Transitional Cluster	Interconnection Customer with an assigned Queue Position prior to
Study	the effective date of this Section 5 may opt to enter the combined
	system impact and interconnection facilities transitional cluster study
	("Transitional Cluster Study") if the Interconnection Customer: (1)
	meets each of the following requirements that demonstrate readiness;
	and (2) executes a Transitional Cluster Study Agreement in the form
	of Appendix 4.2 to the LGIP within thirty (30) Calendar Days of the

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

Effective Date of this LGIP. All Interconnection Requests that enter the Transitional Cluster Study shall be considered to have an equal Queue Position, and identified upgrade costs shall be allocated according to Section 4.2.2 of the LGIP. The Transitional Cluster Study costs shall be allocated according to the method described in Section 4.2. Interconnection Customer may make a one-time extension to its requested Commercial Operation date upon entry into the Transitional Cluster Study and such an extension shall not be past May 1, 2027.

All of the following are required:

- a) Choice of requesting either ERIS or NRIS
- b) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades expected to be identified in the Transitional Cluster Study. The deposit shall be equal to five million dollars (\$5,000,000) and be in the form of an irrevocable letter of credit upon which the Transmission Provider may draw or cash. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied towards future construction costs described in the LGIA. Any amounts in excess of the actual construction costs shall be returned to the customer. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, the deposit is fully refundable once the final invoice for study costs and Withdrawal Penalty is settled.
- c) Exclusive Site Control for the entire Generating Facility.
- d) Interconnection Customer shall provide one of the following:
- i. A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or ii. Reasonable evidence that the Generating Facility is included in an
- approved Resource Plan or Resource Solicitation Process, or iii. An executed Provisional Large Generator Interconnection Agreement filed with FERC that is not in suspension with 1) a commitment to construct the facility, 2) a Commercial Operation Date no later than May 1, 2027 and 3) a security deposit in addition to the five million dollars identified in 5.1.1.2.b

where the total security deposit represents a reasonable estimation of the potential costs that could be ultimately allocated to the project in the transitional cluster study" and "After the Transitional Cluster Study report is published, the remaining process shall proceed according to Section 11 of this LGIP. All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the tender of the draft LGIA or the

	Interconnection Request is deemed withdrawn unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of Transmission Provider's Interconnection Facilities or Network Upgrades if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit (see Section 3.4) is imposed" while deleting "If an agreement termed an Interconnection Feasibility Study Agreement has been executed as of the effective date of this LGIP, and the study has been initiated, then such study shall be completed under the terms of the Agreement."
5.1.1.3	Added in "An Interconnection Customer under an LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 (and not in suspension) that has not built its Large Generating Facility shall have 60 Calendar Days from the effective date of this Section 5.1.1.3 to submit to Transmission Provider a written commitment to (a) start construction of the Generating Facility no later than two (2) years of the effective date of this Section 5.1.1.3 and reach Commercial Operation no later than three (3) years of the effective date of this Section 5.1.1.3 and (b) pay a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit if it withdraws, terminates its LGIA, or otherwise does not reach Commercial Operation. If it does not timely make this submission, the LGIA is deemed terminated, and no Withdrawal Penalty shall be assessed. A deemed termination shall be without prejudice to the Interconnection Customer's ability to submit a new Interconnection Request when it is willing and able to satisfy the tariff criteria under the Definitive Interconnection Process. For any LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 and currently in suspension, the Interconnection Customer shall have 60 Calendar Days from lifting its suspension to make the submission to the Transmission Provider addressed above."
5.1.1.4	Removed subsection 5.1.1.4, deleted "If an Interconnection Study Agreement has been executed prior to the effective date of this LGIP, such Interconnection Study shall be completed in accordance with the terms of such agreement. With respect to any remaining studies for which an Interconnection Customer has not signed an Interconnection Study Agreement prior to the effective date of the LGIP, Transmission Provider must go forward with the completion

<u> </u>	
	of the necessary Interconnection Studies (for which it does not have a
	signed Interconnection Studies Agreement) in accordance with this LGIP."
5.1.1.5	Removed subsection 5.1.1.5, deleted "If an LGIA has been submitted
	to FERC for approval before the effective date of the LGIP, then the
	LGIA would be grandfathered."
Section 6. Informational In	terconnection Study
6.1 - Informational	Adding "At any time, a customer may request, and Transmission
Interconnection Study	Provider (either itself or through a consultant) shall perform a
Agreement	reasonable number of Informational Interconnection Studies from
8	May 1 to November 1 each year. An Informational Interconnection
	Study shall consist of analyses of the type performed in Phase I of the
	DISIS (power flow and voltage analyses). Interconnection Customer
	shall submit a separate Informational Interconnection Request for
	each site and may submit multiple Informational Interconnection
	Requests for a single site. Interconnection Customer must submit a
	deposit with each Informational Interconnection Request even when
	more than one request is submitted for a single site. An Informational
	Interconnection Request to evaluate one site at two different voltage
	levels shall be treated as two Informational Interconnection Requests.
	The request shall use the form in Appendix 5.1 of the LGIP and shall
	describe the assumptions that Interconnection Customer wishes
	Transmission Provider to study within the scope described in Section
	6.2 of the LGIP below. Within five (5) Business Days after receipt of
	a request for an Informational Interconnection Study, Transmission
	Provider shall provide to Interconnection Customer an Informational
	Interconnection Study Agreement in the form of Appendix 5.2.
	The Informational Interconnection Study Agreement shall: (i) include
	the scope of work for the Informational Interconnection Study (ii)
	specify the technical data that Interconnection Customer must
	provide, (iii) specify the Informational Interconnection Study case
	and assumptions, and (iv) identify the Transmission Provider's
	estimate of the cost of the Informational Interconnection Study. To
	the extent known by Transmission Provider, such estimate shall
	include any costs expected to be incurred by any Affected System
	whose participation is necessary to complete the Informational
	Interconnection Study. Notwithstanding the above, Transmission
	Provider shall not be required as a result of an Informational
	Interconnection Study request to conduct any additional
	Interconnection Studies with respect to any other Interconnection
	Request.
	request.

	Interconnection Customer shall execute the Informational Interconnection Study Agreement within ten (10) Business Days of receipt of an agreed upon scope of work and deliver the Informational Interconnection Study Agreement, the technical data, and a study deposit to Transmission Provider in the amount of \$100,000.00."
6.2 – Scope of Informational Interconnection Study	Adding in "The intent of the Informational Interconnection Study is to aid Interconnection Customer in its business decisions related to interconnection of generation facilities prior to entering the Definitive Interconnection Process. The Informational Interconnection Study shall consist of analysis based on the assumptions and scope of work specified by Interconnection Customer in the Informational Interconnection Study Agreement. The Informational Interconnection Study shall identify the potential Transmission Provider's Interconnection Facilities and the Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the results and assumptions of the Informational Interconnection Study. The Informational Interconnection Study shall be performed solely for informational purposes. Transmission Provider shall use Reasonable Efforts to coordinate the study with any Affected Systems that may be affected by the types of Interconnection Services that are being studied. Transmission Provider shall utilize existing studies to the extent
6.3 –Informational	practicable in conducting the Informational Interconnection Study." Adding in "The executed Informational Interconnection Study
Interconnection Study	Agreement, the deposit, and technical and other data called for
Procedures.	therein must be provided to Transmission Provider within ten (10) Business Days of Interconnection Customer receipt of the Informational Interconnection Request. Transmission Provider shall use Reasonable Efforts to complete the Informational Interconnection Study within 70 Calendar Days from the date of the Informational Interconnection Study Agreement. If Transmission Provider is unable to complete the Informational Interconnection Study within such time period, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required. Any difference between the study payment and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation and work papers and databases or data developed in the preparation of the

Open Access Transmission Tariff ATTACHMENT M

Large Generator Interconnection Procedures and Agreement

	Informational Interconnection Study, subject to confidentiality
	arrangements consistent with Section 13.1."
Section 7. Definitive Interconne	ection System Impact Study
7.1 - Definitive Interconnection	Adding "Definitive" to subsection title and adding in "Unless
System Impact Study	otherwise agreed, pursuant to the Scoping Meeting provided for in
Agreement	Section 3.4.4, within thirty (30) Calendar Days acknowledgement of
	a valid Interconnection Request indicating that a Definitive
	Interconnection System Impact Study is to be performed,
	Transmission Provider shall provide to Interconnection Customer a
	DISIS Agreement in the form of Appendix 2 to this LGIP. The DISIS
	Agreement shall provide that Interconnection Customer shall
	compensate Transmission Provider for the actual cost of the DISIS.
	No later than Fifteen (15) Business Days after the close of the DISIS
	Request Window, Transmission Provider shall provide to
	Interconnection Customer a non-binding updated good faith estimate
	of the cost and timeframe for completing the Definitive
	Interconnection System Impact Study" while deleting
	"Simultaneously with the acknowledgement of a valid
	Interconnection Request, Transmission Provider shall provide to
	Interconnection Customer an Interconnection System Impact Study
	Agreement in the form of Appendix 2 of this LGIP. The
	Interconnection System Impact Study Agreement shall specify that
	Interconnection Customer is responsible for the actual cost of the
	Interconnection System Impact Study. Within five (5) Business Days
	following the Scoping Meeting, Interconnection Customer shall
	specify for inclusion in the attachment to the Interconnection System
	Impact Study Agreement the Point(s) of Interconnection. Within five
	(5) Business Days following Transmission Provider's receipt of such
	designation, Transmission Provider shall tender to Interconnection
	Customer the Interconnection System Impact Study Agreement
	signed by Transmission Provider, which includes a good faith
	estimate of the cost and estimated timeframe for completing the
	Interconnection System Impact Study."
7.2 - Execution of Definitive	Added "Definitive" to subsection title and added in "DISIS"
Interconnection System	"DISIS" and while deleting "Interconnection System Impact Study"
Impact Study Agreement	"Interconnection System Impact Study" and "On or before the return
	of the executed Interconnection System Impact Study Agreement to
	Transmission Provider, Interconnection Customer shall provide the
	technical data called for in Appendix 1, Attachment A.
	If Interconnection Customer does not provide all such technical data
	called for in Appendix 1, Attachment A when it delivers the
	Interconnection System Impact Study Agreement, Transmission

Filed Date: 05/31/2023

Appendix A

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

Provider shall notify Interconnection Customer of the deficiency within five (5) Business Days of the receipt of the executed Interconnection System Impact Study Agreement and Interconnection Customer shall cure the deficiency within ten (10) Business Days of receipt of the notice, provided, however, such deficiency does not include failure to deliver the executed Interconnection System Impact Study Agreement. If the Interconnection System Impact Study uncovers any unexpected result(s) not contemplated during the Scoping Meeting, a substitute Point of Interconnection identified by either Interconnection Customer or Transmission Provider, and acceptable to the other, such acceptance not to be unreasonably withheld, will be substituted for the designated Point of Interconnection specified above without loss of Queue Position, and restudies shall be completed pursuant to Section 7.6 as applicable. For the purpose of this Section 7.2, if Transmission Provider and Interconnection Customer cannot agree on the substituted Point of Interconnection, then Interconnection Customer may direct that one of the alternatives as specified pursuant to Section 3.4.4, shall be the substitute."

7.3 - Scope of Definitive Interconnection System Impact Study

Adding "Definitive" to subsection title and adding in "The Definitive Interconnection System Impact Study shall evaluate the impact of the proposed interconnection on the reliability of the Transmission System. The Definitive Interconnection System Impact Study will consider the Base Case as well as all generating facilities (and with respect to (iii) below, any identified Network Upgrades associated with such higher queued requests) that, on the date the DISIS Request Window closes: (i) are existing and directly interconnected to the Transmission System; (ii) are existing and interconnected to Affected Systems and may have an impact on the Interconnection Request; (iii) have a pending higher queued Interconnection Request to interconnect to the Transmission System; and (iv) have no Queue Position but have executed an LGIA or requested that an unexecuted LGIA be filed with FERC. Generating facilities with pending higher or equal queued NRIS requests, or requests associated with Firm Transmission Service shall generally be modeled at full output while existing generation may be re-dispatched to accommodate new requests in the model. Higher queued ERIS requests or in-service ERIS generators without associated Firm Transmission Service may be dispatched at zero in some study models. As discussed in more detail in Section 7.4 below, the Definitive

Interconnection System Impact Study is a phased study where the first phase (Phase 1) consists of a power flow and voltage analysis

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

that is followed by a phase (Phase 2) that consists of a short circuit analysis and a stability analysis, and may also include another appropriate study (for example, an electromagnetic transient analysis, if appropriate). Any DISIS re-studies (Phase 3) shall consist of a power flow/voltage analysis, a short circuit analysis, and/or a stability analysis as needed. The Definitive Interconnection System Impact Study report will state the assumptions upon which it is based; state the results of the analyses; and provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to correct any problems identified in those analyses and implement the interconnection. For purposes of determining necessary Interconnection Facilities and Network Upgrades, the Definitive Interconnection System Impact Study shall consider the level of Interconnection Service requested by the Interconnection Customer, unless otherwise required to study the full Generating Facility Capacity due to safety or reliability concerns. The Definitive Interconnection System Impact Study will provide a list of facilities that are required as a result of the Interconnection Request and a non-binding good faith estimate of cost responsibility and a non-binding good faith estimated time to construct" while deleting "The Interconnection System Impact Study shall evaluate the impact of the proposed interconnection on the reliability of the Transmission System. The Interconnection System Impact Study will consider the Base Case as well as all generating facilities (and with respect to (iii) below, any identified Network Upgrades associated with such higher queued interconnection) that, on the date the Interconnection System Impact Study is commenced: (i) are directly interconnected to the Transmission System; (ii) are interconnected to Affected Systems and may have an impact on the Interconnection Request; (iii) have a pending higher queued Interconnection Request to interconnect to the Transmission System; and (iv) have no Queue Position but have executed an LGIA or requested that an unexecuted LGIA be filed with FERC.

The Interconnection System Impact Study will consist of a short circuit analysis, a stability analysis, and a power flow analysis. The Interconnection System Impact Study will state the assumptions upon which it is based; state the results of the analyses; and provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to correct any problems identified in those analyses and implement the interconnection. For

Filed Date: 05/31/2023

Appendix A

	purposes of determining necessary Interconnection Facilities and Network Upgrades, the System Impact Study shall consider the level of Interconnection Service requested by the Interconnection Customer, unless otherwise required to study the full Generating Facility Capacity due to safety or reliability concerns. The Interconnection System Impact Study will provide a list of facilities that are required as a result of the Interconnection Request and a non-binding good faith estimate of cost responsibility and a non-binding
	good faith estimated time to construct."
7.4 - Definitive Interconnection System Impact Study Procedures	Adding "Definitive" to subsection title and adding in "Definitive" "DISIS" and "Interconnection Requests for DISIS may be submitted only within the DISIS Request Window and Transmission Provider shall initiate the Definitive Interconnection Study Process pursuant to Section 4.2. The At-A-Glance Reference Sheet attached provides an overview and timeline of the Definitive Interconnection Study Process, including the phases and milestones associated with the Definitive Interconnection System Impact Study. a. The DISIS Cluster shall consist of all eligible Interconnection Requests that have executed a DISIS Agreement and have provided all required information before the close of the DISIS Request Window. Transmission Provider shall use Reasonable Efforts to complete the first phase (Phase 1) consisting of a power flow and voltage analysis within ninety (90) Calendar Days. The Phase 1 Report shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades that are expected to be required as a result of the Interconnection Request(s) and a non-binding good-faith indicative level estimate of cost responsibility and a non-binding good-faith estimated time to construct. Transmission Provider shall meet with each cluster participant ("Phase 1 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 1 results on OASIS. b. Within thirty (30) Calendar Days of the Phase 1 Report, all Interconnection Customers are required to provide Readiness Milestone 2 ("M2") and continued evidence of Site Control as described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section
	3.7.

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

c. Interconnection Customers whose M2 and Site Control are accepted by Transmission Provider shall continue in to the second phase ("Phase 2") of the Definitive Interconnection System Impact Study. Phase 2 consists of an updated power flow/voltage analysis (if necessary), stability analysis and short circuit analysis, and any other appropriate study (for example, an electromagnetic transient analysis, if appropriate) for the Interconnection Customers remaining in the DISIS Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 2 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that DISIS Cluster at the requested Interconnection Service level and shall provide nonbinding estimates for required upgrades. The Phase 2 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 2 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 2 results on OASIS.

d. Within thirty (30) Calendar Days of the Phase 2 Report, each Interconnection Customer is required to provide Readiness Milestone 3 ("M3") and additional evidence of Site Control described in Section 7.7.7. Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

i. If all Interconnection Customers in the Cluster provide M3 and no Interconnection Customers withdraw from the Queue at this stage, the Definitive Interconnection Study Process advances to the Interconnection Facilities Study (Section 8). Transmission Provider shall electronically notify Interconnection Customers in the Cluster that Phase 3 is not required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3. ii. If one or more Interconnection Customer withdraws from the Cluster, Transmission Provider shall determine if a full system impact re-study is necessary. If Transmission Provider determines a re-study is not necessary and Phase 3 is not required, Transmission Provider shall provide an updated Phase 2 Report within thirty (30)

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

Calendar Days of such determination and the Definitive Interconnection Study Process advances to the Interconnection Facilities Study (Section 8). When the updated Phase 2 report is issued, Transmission Provider shall electronically notify Interconnection Customers in the Cluster that Phase 3 is not required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3. iii. If one or more Interconnection Customers withdraws from the Cluster and Transmission Provider determines a full system impact re-study is necessary, Transmission Provider will continue with System Impact re-studies ("Phase 3") as described in Section 7.4.e below, until Transmission Provider determines that no further restudies are required. If a customer withdraws after Section 7.4.d.i or Section 7.4.d.ii or during the Interconnection Facilities Study and Transmission Provider determines system impact level studies are necessary, the Cluster shall be restudied under the terms of Phase 3. Transmission Provider shall electronically notify Interconnection Customers in the Cluster and post on OASIS that a re-study is required. Interconnection Customers that have elected NRIS may make a onetime change between Phase 2 and Phase 3 (before the restudy starts) to ERIS if they notify Transmission Provider of such change in election within five (5) Business Days of the Transmission Provider's notification the first re-study is required. e. Interconnection Customers whose M3 and additional evidence of Site Control is accepted by Transmission Provider shall continue with the third phase ("Phase 3") of the Definitive Interconnection System Impact Study. Phase 3 may consist of updated power flow/voltage analysis, stability analysis, and/or short circuit and other analyses if necessary for the Interconnection Customers remaining in the Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 3 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 3 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 3 Report Meeting") within ten

(10) Business Days of publishing the DISIS Phase 3 results on

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

OASIS. If additional re-studies are required before moving to Phase 4 below, within thirty (30) Calendar Days of the Phase 3 Report, all Interconnection Customers are required to provide an updated Readiness Milestone 3 ("M3"). Readiness Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7. by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7. Transmission Provider shall electronically notify Interconnection Customers in the Cluster when no further re-studies are required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3. f. Within twenty (20) Calendar Days of the notice that no System Impact re-studies are needed, each Interconnection Customer is required to provide Readiness Milestone 4 ("M4"), Site Control requirements described in Section 7.7.7, and an executed Interconnection Facilities Agreement in the form of Appendix 3 (completed and including all required data identified therein). Readiness Milestones for the Definitive Interconnection System Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7. g. Twenty (20) Calendar Days after the notice that no further Re-Studies are needed, Transmission Provider shall proceed with the Interconnection Facilities Study phase ("Phase 4") of the Definitive Study Process, described in detail in Section 8 below. An additional study deposit is not required for Phase 4" "DISIS" "DISIS Cluster" "DISIS" and "DISIS" while changing "required" to "indicated" "Customer" to "Customer(s)" and while deleting "study" and "Transmission Provider shall use Reasonable Efforts to complete the Interconnection System Impact Study within ninety (90) Calendar Days after the receipt of the Interconnection System Impact Study Agreement or notification to proceed, study payment, and technical data. If Transmission Provider uses Clustering, Transmission Provider shall use Reasonable Efforts to deliver a completed Interconnection System Impact Study within ninety (90) Calendar Days after the close of the Queue Cluster Window" "Interconnection System Impact Study" "Interconnection System Impact Study"

	"Interconnection System Impact Study" and "Interconnection System
	Impact Study."
7.5 - Meeting with Transmission Provider	Adding in "DISIS" "and posting the report on OASIS" and "study" while changing "an to "a" and while deleting "Interconnection System Impact Study."
7.6 – Re-Study	Adding in "Definitive" "other than the re-study described above in 7.4.e" "or equal priority" "ensure such" "take" "one hundred fify" "150" and "be borne by Interconnection Customer(s) being restudied" while deleting "or re-designation of the Point of Interconnection pursuant to Section 7.2" "complete the" "sixty" "60" and "first utilize the funds remaining of the non-refundable portion of the withdrawing applicant's Initial Deposit as per Section 3.3.1, then the funds remaining of the Initial Deposit(s) of all Interconnection Customer(s) being re-studied, and, if these funds are insufficient, the Interconnection Customer is responsible for the remaining actual cost incurred for the re-study."
7.7 – Readiness Milestones and Site Control	Added 7.7 "Readiness Milestones are required throughout the Definitive Interconnection Study Process to demonstrate readiness. A customer that does not sufficiently demonstrate readiness by providing a Readiness Milestones is subject to withdrawal as described in Section 3.7 which may include additional penalties. There are three Readiness Milestone options that demonstrate readiness through the study process (i.e., for Readiness Milestones 1 (M1) through Readiness Milestones 4 (M4)."
7.7.1 – Readiness Milestone 1 ("M1")	Added 7.7.1 "M1 is satisfied by any one of the three options below (also described in 3.4.1) at Interconnection Customer's option. M1 may also be satisfied by providing additional security described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness. a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years. b) Reasonable evidence the project has been selected in a Resource Plan or Resource Solicitation Process; or c) Provisional Large Generator Interconnection Agreement accepted for filing at FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility."

Filed Date: 05/31/2023

Appendix A

7.7.2 – Readiness Milestone 2	Added 7.7.2 "M2 is satisfied by any one of the three options below at
("M2")	Interconnection Customer's option. M2 may also be satisfied by
	providing additional security as described in Section 7.7.5 in lieu of
	providing one of the three options to demonstrate readiness.
	a) Executed term sheet (or comparable evidence) related to a
	contract, binding upon the parties to the contract, for sale of (i) the
	constructed Generating Facility, (ii) the Generating Facility's energy,
	or (iii) the Generating Facility's ancillary services if the Generating
	Facility is an electric storage resource; where the term of sale is not
	less than five (5) years.
	` / •
	b) Reasonable evidence that the project has been selected in a
	Resource Plan or Resource Solicitation Process; or
	c) Provisional Large Generator Interconnection Agreement accepted
	for filing at FERC. Such an agreement shall not be suspended and
	shall include a commitment to construct the Generating Facility."
7.7.3 Readiness Milestone 3	Added 7.7.3 "M3 is satisfied by any one of the three options below at
("M3")	Interconnection Customer's option. M3 may also be satisfied by
	providing additional security described in Section 7.7.5 in lieu of
	providing one of the three options to demonstrate readiness.
	a) Executed contract, binding upon the parties to the contract, for sale
	of (i) the constructed Generating Facility, (ii) the Generating
	Facility's energy, or (iii) the Generating Facility's ancillary services
	if the Generating Facility is an electric storage resource; where the
	term of sale is not less than five (5) years.
	b) Reasonable evidence that the project has been selected in an
	approved Resource Plan or Resource Solicitation Process; or
	c) An unsuspended Provisional Large Generator Interconnection
	Agreement accepted for filing by FERC with reasonable evidence
	that the Generating Facility and Interconnection Facilities have
	commenced design and engineering."
7.7.4 – Readiness Milestone 4	Added 7.7.4 "M4 is satisfied by any one of the three options below at
("M4")	Interconnection Customer's option. M4 may also be satisfied by
(·)	providing additional security as described in Section 7.7.5 below in
	lieu of providing one of the three options to demonstrate readiness.
	a) Executed contract, binding upon the parties to the contract, for sale
	of (i) the constructed Generating Facility, (ii) the Generating
	Facility's energy, or (iii) the Generating Facility's ancillary services
	and capacity if the Generating Facility is an electric storage resource;
	where the term of sale is not less than five (5) years;
	b) Reasonable evidence that the project has been selected in an
	approved Resource Plan or Resource Solicitation Process; or
	approved Resource Fian of Resource Solicitation Process; of

Filed Date: 05/31/2023

Appendix A

	c) An unsuspended Provisional Large Generator Interconnection
	Agreement accepted for filing by FERC with reasonable evidence
	that the Generating Facility and Interconnection Facilities have
	commenced construction."
7.7.5 Security Requirements	Added 7.7.5 "A table showing the security required in each milestone
	is provided in the At-A-Glance attachment. The security amount is
	dependent on if the customer provided a Readiness Milestone and the
	study phase the customer is entering. All security described below
	shall be in the form of an irrevocable letter of credit upon which
	Transmission Provider may draw or cash. The security is refunded to
	the Interconnection Customer upon withdrawal, LGIA termination,
	or Commercial Operation after any final invoice is settled. If cash is
	provided as security, it shall be refunded plus interest, where the
	interest is calculated in accordance with the methodology set forth in
	FERC's regulations at 18 CFR § 35.19a(a)(2)(iii) from the date the
	security is received to the date that it is refunded. Security may be
	drawn upon if costs under this LGIP including the LGIA remain unpaid as per this LGIP and the attached LGIA. As part of a valid
	interconnection request all Interconnection Customers must provide
	security equal to the study deposit amount as described in Section
	3.4. The security provided in Section 3.4.1 will be applied towards
	the amount of security required for M5. An Interconnection
	Customer may opt to provide security in lieu of providing Readiness
	Milestones 1 through 4, as described above in Sections 7.7.1, 7.7.2,
	7.7.3, and 7.7.4. The security provided is applied towards the security
	amount required for each successive milestone if the Interconnection
	Customer does not withdraw from the queue. For example, the
	security provided for M2 is applied to the amount of security
	required for M3.
	In lieu of providing a demonstration of readiness for Milestones 1
	through 4, the amount of security required is a multiple of the study
	deposit described in Section 3.4 and is in addition to the security
	required for all Interconnection Customers under Sections 3.4. The
	additional amount of security required for each milestone for
	Interconnection Customers that do not provide a demonstration of
	readiness is: M1 = 1 times the study deposit amount
	M1 - 1 times the study deposit amount M2 = 2 times the study deposit amount
	M3 = 4 times the study deposit amount
	M4 = 6 times the study deposit amount
	For clarity, the total (i.e., inclusive of the security required under
	Section 3.4) amount of security required for each milestone for
L	, , , , , , , , , , , , , , , , , , , ,

Filed Date: 05/31/2023

Appendix A

Open Access Transmission Tariff ATTACHMENT M Large Generator Interconnection Procedures and Agreement

Interconnection Customers that do not	provide a demonstration of

readiness is:

M1 = 2 times the study deposit amount

M2 = 3 times the study deposit amount

M3 = 5 times the study deposit amount

M4 = 7 times the study deposit amount

All Interconnection Customers are required to provide security in order to satisfy Readiness Milestone 5 (M5) when the LGIA is executed as described in Section 11.3. The amount of security required for M5 is equal to nine (9) times the Interconnection Customer's share of the Definitive Interconnection Study Process study costs. If this amount is not known, the study deposit amount shall be used as an estimate of study cost until such amounts are known. If initially estimated, M5 shall be updated when the final invoice for actual study costs is issued. As this M5 amount is the total security required to satisfy Readiness Milestone 5, any security provided pursuant to Sections 3.4 and 7.7 shall be applied towards the Readiness Milestone 5 amount when the LGIA is executed. The Interconnection Customer shall only be responsible to provide the incremental amount of security to the Transmission Provider and any excess security provided shall be refunded to the Interconnection Customer. Transmission Provider shall refund all security provided under this section to the Interconnection Customer upon achieving Commercial Operation."

7.7.6 [Reserved for Future Use]

7.7.7 – Site Control

Added 7.7.6

Added 7.7.7 "In addition to the above Readiness Milestones, Site Control is

required to determine increased readiness through the Definitive Interconnection Study Process. Additional information on Site Control is posted on Transmission Provider's OASIS.

- a) Before entering Phase 1 (concurrent with M1) demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- b) Before entering Phase 2 (concurrent with M2): continued demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- c) Before entering Phase 3 (concurrent with M3): demonstration of 60% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- d) Before entering Phase 4 (concurrent with M4): demonstration of 75% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.

	e) Before executing an LGIA (concurrent with M5): demonstration of
	90% Site Control and 50% Site Control of Interconnection
	Customer's Interconnection Facilities is required."
Section 8. Interconnection Faci	
8.1 - Interconnection Facilities	Added in "notice" "(s) that Phase 3 is complete or not required"
Study Agreement	"Within five (5) Business Days following the open DISIS results
7.29.20	(Phase 2 or Phase 3) meeting, Transmission Provider shall provide to
	Interconnection Customer a non-binding good faith estimate of the
	cost and timeframe for completing the Interconnection Facilities
	Study." "twenty" and "Readiness Milestone 4 and the Site Control
	requirements described in Section 7.7.7. Interconnection Customers
	that do not provide the Readiness Milestone (or additional security in
	lieu of the Readiness Milestone) and provide Site Control described
	in Section 7.7.7 by the required date shall be deemed withdrawn from
	the Queue pursuant to Section 3.7."
	Deleting "delivery of the Interconnection System Impact Study"
	"Within three (3) Business Days following the Interconnection
	System Impact Study results meeting, Transmission Provider shall
	provide to Interconnection Customer a non-binding good faith
	estimate of the cost and timeframe for completing the
	Interconnection Facilities Study" and "thirty."
8.1.1	Deleted 8.1.1 "Transmission Provider shall invoice Interconnection
	Customer on a monthly basis for the work to be conducted on the
	Interconnection Facilities Study each month. Interconnection
	Customer shall pay invoiced amounts within thirty (30) Calendar
	Days of receipt of invoice. Transmission Provider shall continue to
	hold the amounts on deposit until settlement of the final invoice."
8.2 – Scope of Interconnection	Added "provide a non-binding" "of" "Phase 1 and Phase 2 (and/or
Facilities Study	Phase 3) Reports (as appropriate)" and "Interconnection." Removed
	"Interconnection System Impact Study" and changed "Facility" to "Facilities."
8.3 - Interconnection Facilities	Added "acceptance of the Interconnection Facilities Agreement and
Study Procedures	Readiness Milestone 4" "identified" "Interconnection Facilities
	Study" "consider" "completing" "Interconnection Facilities Study"
	and "Study Report." While deleting "receipt of an executed
	Interconnection Facilities Study Agreement." "or at any time
	Transmission Provider determines that it will not meet the required
	time frame for completing the Interconnection Facilities Study"
	"required" and "included."
8.5 – Re-Study	Added "or equal priority" "Re-studies that require re-running the
	system impact study analysis may take longer than sixty days" and
	"be" while deleting "first utilize the funds remaining of the

	withdrawing applicant's Initial Deposit as per Section 3.3.1, then the
	funds remaining of the Initial Deposit(s) of all Interconnection
	Customer(s) being re-studied, and, if these funds are insufficient, the
	remaining cost is." Capitalized "Queue."
Section 9. Engineering & Procu	
Section 9 - Engineering &	Added "Readiness" and Capitalized "Milestones."
Procurement ('E&P') Agreement	Added Readilless and Capitalized Wilestolles.
Section 10. Optional Intercon	naction Study
Section 10. Optional Interconn	Changed the name from "Optional Interconnection Study" to
Section 10.	"Reserved for future use"
10.1 Ontional Interconnection	
10.1 - Optional Interconnection	Deleted 10.1 "On or after the date when Interconnection Customer
Study Agreement	receives Interconnection System Impact Study results,
	Interconnection Customer may request, and Transmission Provider
	shall perform a reasonable number of Optional Studies. The request
	shall describe the assumptions that Interconnection Customer wishes
	Transmission Provider to study within the scope described in Section
	10.2. Within five (5) Business Days after receipt of a request for an
	Optional Interconnection Study, Transmission Provider shall provide
	to Interconnection Customer an Optional Interconnection Study
	Agreement in the form of Appendix 5.
	The Optional Interconnection Study Agreement shall: (i) specify the technical data that Interconnection Customer must provide for each phase of the Optional Interconnection Study, (ii) specify
	Interconnection Customer's assumptions as to which Interconnection Requests with earlier queue priority dates will be excluded from the Optional Interconnection Study case and assumptions as to the type of interconnection service for Interconnection Requests remaining in the Optional Interconnection Study case, and (iii) Transmission Provider's estimate of the cost of the Optional Interconnection Study.
	To the extent known by Transmission Provider, such estimate shall include any costs expected to be incurred by any Affected System
	whose participation is necessary to complete the Optional
	Interconnection Study. Notwithstanding the above, Transmission
	Provider shall not be required as a result of an Optional
	Interconnection Study request to conduct any additional
	Interconnection Studies with respect to any other Interconnection
	Request.
	Interconnection Customer shall execute the Optional Interconnection
	Study Agreement within ten (10) Business Days of receipt and
	deliver the Optional Interconnection Study Agreement, the technical
	data and a \$25,000 deposit to Transmission Provider."

10.2 - Scope of Optional	Deleted 10.2 "The Optional Interconnection Study will consist of a	
Interconnection Study	sensitivity analysis based on the assumptions specified by	
	Interconnection Customer in the Optional Interconnection Study	
	Agreement. The Optional Interconnection Study will also give a	
	good faith estimate of Transmission Provider's Interconnection	
	Facilities and the Network Upgrades, and the estimated cost thereof,	
	that may be required to provide Interconnection Service based upon	
	the results of the Optional Interconnection Study. The Optional	
	Interconnection Study shall be performed solely for informational	
	purposes. Transmission Provider shall use Reasonable Efforts to	
	coordinate the study with any Affected Systems that may be affected	
	by the types of Interconnection Services that are being studied.	
	Transmission Provider shall utilize existing studies to the extent	
	practicable in conducting the Optional Interconnection Study."	
10.3 - Optional Interconnection	Deleted 10.3 "The executed Optional Interconnection Study	
Study Procedures	Agreement, the prepayment, and technical and other data called for	
	therein must be provided to Transmission Provider within ten (10)	
	Business Days of Interconnection Customer receipt of the Optional	
	Interconnection Study Agreement. Transmission Provider shall use	
	Reasonable Efforts to complete the Optional Interconnection Study	
	within a mutually agreed upon time period specified within the	
	Optional Interconnection Study Agreement. If Transmission	
	Provider is unable to complete the Optional Interconnection Study	
	within such time period, it shall notify Interconnection Customer and	
	provide an estimated completion date and an explanation of the	
	reasons why additional time is required. Any difference between the	
	study deposit and the actual cost of the study shall be paid to	
	Transmission Provider or refunded to Interconnection Customer, as	
	appropriate. Upon request, Transmission Provider shall provide	
	Interconnection Customer supporting documentation and workpapers	
	and databases or data developed in the preparation of the Optional	
	Interconnection Study, subject to confidentiality arrangements	
	consistent with Section 13.1."	
Section 11. Standard Large Generator Interconnection Agreement (LGIA)		
11.1 - Tender	Added "or after the Interconnection Customer notifies Transmission	
	Provider that it will provide no comments" and "so that Transmission	
	Provider may prepare a final LGIA. Transmission Provider shall	
	tender a final LGIA, together with final appendices, within thirty (30)	
	Calendar Days after the expiration of the thirty (30) Calendar Day	
	window in which the Interconnection Customer is to return	
	completed draft appendices" while deleting "execute and."	
11.2 – Negotiation	Added "Interconnection Customer" while deleting "it."	

11.3 – Execution and Filing	Added "as defined in Section 7.7.7 and" "Readiness Milestone 5" "equal to nine (9) times that Interconnection Customer's share of the Definitive Interconnection Study Process study" "If the actual study costs are not known" "study costs shall be estimated as the study deposit described in Section 3.4, and the M5 amount shall be updated when the study costs are known. If the Interconnection Customer does not reach Commercial Operation, upon payment of any final invoice, including any Withdrawal Penalty, Readiness Milestone 5 shall be refunded to the Interconnection Customer, including any accumulated interest, if applicable. If the Interconnection Customer reaches Commercial Operation, Readiness Milestone 5 is refunded to the Interconnection Customer including any accumulated interest, if applicable. Within fifteen (15) Business Days after receipt of the final LGIA," "(applicable to Generating Facilities that require fuel transportation; not available for wind or solar resources)" "(applicable to Generating Facilities that require cooling water for the production process; not available for wind or solar resources)" and "At the same time that Interconnection Customer makes its submission of the items identified above (due within fifteen (15) Business Days after receipt of the final LGIA)" while deleting "or" "ing of \$250,000, non-refundable additional" and "which shall be
Cartina 12 Canadana di an af Tara	applied toward future construction."
Section 12. Construction of 1ra Upgrades	nsmission Provider's Interconnection Facilities and Network
12.2.4 - Amended Definitive	Added "Definitive" to subsection title. Added "Definitive" and
Interconnection System Impact Study	"may" while deleting "will" and changing "An" to "A."
Section 13. Miscellaneous	
13.3 – Obligations for Study Costs and Withdrawal Penalty	Added "and Withdrawal Penalty" to subsection title. Added "and the Withdrawal Penalty, as applicable" "as well as the Withdrawal Penalty, if applicable" and "If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided under this LGIP to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security" as well as deleting "submitted as per Section 3.3.1, less any part of that Deposit deemed non-refundable as per Section 3.3.1" and changing "Study" to "Study(ies)."
13.5.5 - Non-Binding Dispute Resolution Procedures	Capitalized subsection title. Changed "days" to "Calendar Days."

ATTACHMENT M

Large Generator Interconnection Procedures and Agreement

Standard Large Generator Interconnection Procedures (LGIP)

(Applicable to Generating Facilities that exceed 20 MW)

TABLE OF CONTENTS

Section 1.	Definitions	
Section 2.	Scope and Application	
2.1	Application of Standard Large Generator Interconnection Procedures	
2.2		
2.3	•	
2.4	No Applicability to Transmission Service	
Section 3.	Interconnection Requests	
3.1	General	
3.2	Identification of Types of Interconnection Services	
	3.2.1 Energy Resource Interconnection Service	
	3.2.1.1 The Product	
	3.2.1.2 The Study	
	3.2.2 Network Resource Interconnection Service	
	3.2.2.1 The Product	
	3.2.2.2 The Study	
3.3	Utilization of Surplus Interconnection Service	
	3.3.1 Surplus Interconnection Service Requests	
3.4	Valid Interconnection Request	
	3.4.1 Initiating an Interconnection Request	
	3.4.2 Acknowledgment of Interconnection Request	
	3.4.3 Deficiencies in Interconnection Request	
	3.4.4 Scoping Meeting	
3.5	OASIS Posting	
	3.5.1 OASIS Posting	
	3.5.2 Requirement to Post Interconnection Study Metrics	
	3.5.2.1 Interconnection System Impact Studies Processing Time	
	3.5.2.2 Interconnection Facilities Studies Processing Time	
	3.5.2.3 Interconnection Service Requests Withdrawn from	
	Interconnection Queue	
3.6	Coordination with Affected Systems	
3.7	Withdrawal	
	3.7.1 Withdrawal Penalty	
	3.7.1.1 Calculation of the Withdrawal Penalty	
	3.7.1.2 Distribution of the Withdrawal Penalty	
3.8	Identification of Contingent Facilities	
Section 4.	Queue Position Interconnection Request Evaluation Process	
4.1	General Queue Position	
	4.1.1 Assignment of Queue Position	

	4.1.2 Higher Queue Position
4.2	Clustering and the Definitive Study Process
	4.2.1 <u>Initiation of a Resource Solicitation</u> Cluster Window Transition Period
	4.2.2 Network Upgrades Cost Allocation
4.3	Transferability of Queue Position
4.4	Modifications
Section 5.	Procedures for Interconnection Requests Submitted Prior to Effective Date of Standard Large Generator Interconnection Procedures
5.1	Queue Position for Pending Requests Transition to the First-Ready, First-
Served Prod	<u>cess</u>
5.2	New Transmission Provider
Section 6.	[Reserved for Future Use] Informational Interconnection Study
6.1	Informational Interconnection Study Agreement
6.2	Scope of Informational Interconnection Study
6.3	Informational Interconnection Study Procedures
Section 7.	<u>Definitive</u> Interconnection System Impact Study
7.1	Definitive Interconnection System Impact Study Agreement
7.2	Execution of Definitive Interconnection System Impact Study Agreement
7.3	Scope of Definitive Interconnection System Impact Study
7.4	Definitive Interconnection System Impact Study Procedures
7.5	Meeting with Transmission Provider
7.6	Re-Study
7.7	Readiness Milestones and Site Control
·-	7.7.1 Readiness Milestone 1 ("M1")
	7.7.2 Readiness Milestone 2 ("M2")
	7.7.3 Readiness Milestone 3 ("M3")
	7.7.4 Readiness Milestone 4 ("M3")
	7.7.5 Security Requirements
	7.7.6 [Reserved for Future Use]
	7.7.7 Site Control
Section 8.	Interconnection Facilities Study
8.1	Interconnection Facilities Study Agreement
8.2	Scope of Interconnection Facilities Study
8.3	Interconnection Facilities Study Procedures
8.4	Meeting with Transmission Provider
8.5	Re-Study
Section 9.	Engineering & Procurement ('E&P') Agreement

Section 10.	Optional Interconnection Study[Reserved for Future Use]
10.1	Optional Interconnection Study Agreement
10.2	Scope of Optional Interconnection Study
10.3	Optional Interconnection Study Procedures
Section 11.	Standard Large Generator Interconnection Agreement (LGIA)
11.1	Tender
	Negotiation
	Execution and Filing
11.4	Commencement of Interconnection Activities
Section 12.	Construction of Transmission Provider's Interconnection Facilities and Network Upgrades
12.1	Schedule
12.2	Construction Sequencing 12.2.1 General
	 12.2.2 Advance Construction of Network Upgrades that are an Obligation of an Entity other than Interconnection Customer 12.2.3 Advancing Construction of Network Upgrades that are Part of an Expansion Plan of the Transmission Provider
	Expansion Plan of the Transmission Provider 12.2.4 Amended Interconnection System Impact Study
Section 13.	Miscellaneous
13.1	Confidentiality
	13.1.1 Scope
	13.1.2 Release of Confidential Information
	13.1.3 Rights
	13.1.4 No Warranties
	13.1.5 Standard of Care
	13.1.6 Order of Disclosure
	13.1.7 Remedies
	13.1.8 Disclosure to FERC, its Staff, or a State
13.2	Delegation of Responsibility
13.3	Obligation for Study Costs and Withdrawal Penalty
13.4	Third Parties Conducting Studies
13.5	Disputes
	13.5.1 Submission
	13.5.2 External Arbitration Procedures
	13.5.3 Arbitration Decisions
	13.5.4 Costs
	13.5.5 Non-binding dispute resolution procedures
13.6	Local Furnishing Bonds

- 13.6.1 Transmission Providers That Own Facilities Financed by Local Furnishing Bonds
- 13.6.2 Alternative Procedures for Requesting Interconnection Service
- Appendix 1 Interconnection Request for a Large Generating Facility
- Appendix 2 <u>Definitive</u> Interconnection System Impact Study Agreement
- Appendix 3 Interconnection Facilities Study Agreement
- Appendix 4 Transitional Study Agreements Reserved for future use
 - 4.1 Transitional Facilities Interconnection Study Agreement
 - 4.2 Transitional Cluster Study Agreement
- Appendix 5 Optional Interconnection Informational Study Forms-
 - 5.1 Informational Interconnection Study Request
 - 5.2 Informational Interconnection Study Agreement

Appendix 6 - Standard Large Generator Interconnection Agreement

Appendix 7 - Interconnection Procedures for a Wind Generating Plant

Section 1. Definitions

Adverse System Impact shall mean the negative effects due to technical or operational limits on conductors or equipment being exceeded that may compromise the safety and reliability of the electric system.

Affected System shall mean an electric system other than the Transmission Provider's Transmission System that may be affected by the proposed interconnection.

Affected System Operator shall mean the entity that operates an Affected System.

Affiliate shall mean, with respect to a corporation, partnership or other entity, each such other corporation, partnership or other entity that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such corporation, partnership or other entity.

Ancillary Services shall mean those services that are necessary to support the transmission of capacity and energy from resources to loads while maintaining reliable operation of the Transmission Provider's Transmission System in accordance with Good Utility Practice.

Applicable Laws and Regulations shall mean all duly promulgated applicable federal, state and local laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders, permits and other duly authorized actions of any Governmental Authority.

Applicable Reliability Council shall mean the reliability council applicable to the Transmission System to which the Generating Facility is directly interconnected.

Applicable Reliability Standards shall mean the requirements and guidelines of NERC, the Applicable Reliability Council, and the Control Area of the Transmission System to which the Generating Facility is directly interconnected.

Base Case shall mean the base case power flow, short circuit, and stability data bases used for the Interconnection Studies by the Transmission Provider or Interconnection Customer.

Breach shall mean the failure of a Party to perform or observe any material term or condition of the Standard Large Generator Interconnection Agreement.

Breaching Party shall mean a Party that is in Breach of the Standard Large Generator Interconnection Agreement.

Business Day shall mean Monday through Friday, excluding Federal Holidays.

Calendar Day shall mean any day including Saturday, Sunday or a Federal Holiday.

Cluster shall mean a group of Interconnection Requests (one or more) that are studied together for the purpose of conducting Interconnection Studies.

Cluster Study shall mean an Interconnection Study evaluating one or more Interconnection Requests.

Clustering shall mean the process whereby a group of Interconnection Requests is studied together, instead of serially, for the purpose of conducting the Interconnection System Impact Studyies.

Commercial Operation shall mean the status of a Generating Facility that has commenced generating electricity for sale, excluding electricity generated during Trial Operation.

Commercial Operation Date of a unit shall mean the date on which the Generating Facility commences Commercial Operation as agreed to by the Parties pursuant to Appendix E to the Standard Large Generator Interconnection Agreement.

Confidential Information shall mean any confidential, proprietary or trade secret information of a plan, specification, pattern, procedure, design, device, list, concept, policy or compilation relating to the present or planned business of a Party, which is designated as confidential by the Party supplying the information, whether conveyed orally, electronically, in writing, through inspection, or otherwise.

Contingent Facilities shall mean those unbuilt Interconnection Facilities and Network Upgrades upon which the Interconnection Request's costs, timing, and study findings are dependent, and if delayed or not built, could cause a need for Re-Studies of the Interconnection Request or a reassessment of the Interconnection Facilities and/or Network Upgrades and/or costs and timing.

Control Area shall mean an electrical system or systems bounded by interconnection metering and telemetry, capable of controlling generation to maintain its interchange schedule with other Control Areas and contributing to frequency regulation of the interconnection. A Control Area must be certified by an Applicable NERC Regional Reliability Council Entity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC.

<u>Customer Engagement Window</u> shall have the meaning set forth in Section 4.2 of the LGIP.

Default shall mean the failure of a Breaching Party to cure its Breach in accordance with Article 17 of the Standard Large Generator Interconnection Agreement.

Definitive Interconnection Study Process ("Definitive Interconnection Study") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study.

<u>Definitive Interconnection System Impact Study ("DISIS")</u> shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System.

<u>Definitive Interconnection System Impact Study Agreement ("DISIS</u>

<u>Agreement") shall mean the form of agreement contained in the Appendices of the LGIP for conducting the Definitive Interconnection System Impact Study.</u>

<u>Shall mean an engineering study that evaluates the impact of the proposed interconnection(s) on the safety and reliability of the Transmission System and, if applicable, an Affected System.</u>

DISIS Request Window shall have the meaning set forth in Section 4.2 of the LGIP.

Deposit in Lieu of Site Control shall mean a deposit of \$160,000 for an Interconnection Request of up to 75 MW, or \$250,000 for an Interconnection Request of 75 MW or greater, that Interconnection Customer submits to Transmission Provider in place of demonstrating Site Control.

Dispute Resolution shall mean the procedure for resolution of a dispute between the Parties in which they will first attempt to resolve the dispute on an informal basis.

Distribution System shall mean the Transmission Provider's facilities and equipment used to transmit electricity to ultimate usage points such as homes and industries directly from nearby generators or from interchanges with higher voltage transmission networks which transport bulk power over longer distances. The voltage levels at which distribution systems operate differ among areas.

Distribution Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Distribution System at or beyond the Point of Interconnection to facilitate interconnection of the Generating Facility and render the transmission service necessary to effect Interconnection Customer's wholesale sale of electricity in interstate commerce. Distribution Upgrades do not include Interconnection Facilities.

Effective Date shall mean the date on which the Standard Large Generator Interconnection Agreement becomes effective upon execution by the Parties subject to acceptance by FERC, or if filed unexecuted, upon the date specified by FERC.

Emergency Condition shall mean a condition or situation: (1) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (2) that, in the case of a Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to Transmission Provider's Transmission System, Transmission Provider's Interconnection Facilities or the electric systems of others to which the Transmission Provider's Transmission System is directly connected; or (3) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Generating Facility or Interconnection Customer's Interconnection Facilities. System restoration and black start shall be considered Emergency Conditions; provided that Interconnection Customer is not obligated by the Standard Large Generator Interconnection Agreement to possess black start capability.

Energy Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to connect its Generating Facility to the Transmission Provider's Transmission System to be eligible to deliver the Generating Facility's electric output using the existing firm or nonfirm capacity of the Transmission Provider's Transmission System on an as available basis. Energy Resource Interconnection Service in and of itself does not convey transmission service.

Engineering & Procurement (E&P) Agreement shall mean an agreement that authorizes the Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection in order to advance the implementation of the Interconnection Request.

Environmental Law shall mean Applicable Laws or Regulations relating to pollution or protection of the environment or natural resources.

Federal Power Act shall mean the Federal Power Act, as amended, 16 U.S.C. §§ 791a et seq.

FERC shall mean the Federal Energy Regulatory Commission (Commission) or its successor.

Force Majeure shall mean any act of God, labor disturbance, act of the public enemy, war, insurrection, riot, fire, storm or flood, explosion, breakage or accident to machinery or equipment, any order, regulation or restriction imposed by governmental, military or lawfully established civilian authorities, or any other cause beyond a Party's control. A Force Majeure event does not include acts of negligence or intentional wrongdoing by the Party claiming Force Majeure.

Generating Facility shall mean Interconnection Customer's device for the production and/or storage for later injection of electricity identified in the Interconnection Request, but shall not include the Interconnection Customer's Interconnection Facilities.

Generating Facility Capacity shall mean the net capacity of the Generating Facility and the aggregate net capacity of the Generating Facility where it includes multiple energy production devices.

Good Utility Practice shall mean any of the practices, methods and acts engaged in or approved by a significant portion of the electric industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act to the exclusion of all others, but rather to be acceptable practices, methods, or acts generally accepted in the region.

Governmental Authority shall mean any federal, state, local or other governmental regulatory or administrative agency, court, commission, department, board, or other governmental subdivision, legislature, rulemaking board, tribunal, or other governmental authority having jurisdiction over the Parties, their respective facilities, or the respective services they provide, and exercising or entitled to exercise any administrative, executive, police, or taxing authority or power; provided, however, that such term does not include Interconnection Customer, Transmission Provider, or any Affiliate thereof.

Hazardous Substances shall mean any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "radioactive substances," "contaminants," "pollutants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, or any other chemical,

material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

Informational Interconnection Study shall mean an analysis based on assumptions specified by Interconnection Customer in the Informational Interconnection Study Agreement.

<u>Informational Interconnection Study Agreement</u> shall mean the form of agreement contained in Appendix 5 of the LGIP for conducting the Informational Interconnection Study.

Initial Deposit shall mean a deposit of \$160,000 for an Interconnection Request of up to 75 MW, or \$250,000 for an Interconnection Request of 75 MW or greater, that Interconnection Customer submits to Transmission Provider upon initiating an Interconnection Request.

Initial Synchronization Date shall mean the date upon which the Generating Facility is initially synchronized and upon which Trial Operation begins.

In-Service Date shall mean the date upon which the Interconnection Customer reasonably expects it will be ready to begin use of the Transmission Provider's Interconnection Facilities to obtain back feed power.

Interconnection Customer shall mean any entity, including the Transmission Provider, Transmission Owner or any of the Affiliates or subsidiaries of either, that proposes to interconnect its Generating Facility with the Transmission Provider's Transmission System.

Interconnection Customer's Interconnection Facilities shall mean all facilities and equipment, as identified in Appendix A of the Standard Large Generator Interconnection Agreement, that are located between the Generating Facility and the Point of Change of Ownership, including any modification, addition, or upgrades to such facilities and equipment necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Customer's Interconnection Facilities are sole use facilities.

Interconnection Facilities shall mean the Transmission Provider's Interconnection Facilities and the Interconnection Customer's Interconnection Facilities. Collectively, Interconnection Facilities include all facilities and equipment between the Generating Facility and the Point of Interconnection, including any modification, additions or upgrades that are necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Facilities are sole use facilities and shall not include Distribution

Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Interconnection Facilities Study shall mean a study conducted by the Transmission Provider or a third party consultant for the Interconnection Customer to determine a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades as identified in the Definitive Interconnection System Impact Study), the cost of those facilities, and the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System. The scope of the study is defined in Section 8 of the Standard Large Generator Interconnection Procedures.

Interconnection Facilities Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Facilities Study.

Interconnection Request shall mean an Interconnection Customer's request, in the form of Appendix 1 to the Standard Large Generator Interconnection Procedures, in accordance with the Tariff, to interconnect a new Generating Facility, or to increase the capacity of, or make a Material Modification to the operating characteristics of, an existing Generating Facility that is interconnected with the Transmission Provider's Transmission System.

Interconnection Service shall mean the service provided by the Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to the Transmission Provider's Transmission System and enabling it to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Standard Large Generator Interconnection Agreement and, if applicable, the Transmission Provider's Tariff.

Interconnection Study shall mean any of the following studies: the Informational Interconnection Study, the Definitive Interconnection System Impact Study, and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection Study Agreement shall mean any of the following studies: the <u>Informational Interconnection Study Agreement, the Definitive</u> Interconnection System Impact Study and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection System Impact Study shall mean a technical and engineering study that evaluates the impact of the proposed interconnection on the safety and reliability of Transmission Provider's Transmission System and, if applicable, an Affected System. The study shall identify a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades) to interconnect the

Generating Facility, a good faith estimate of the cost of those facilities, and a good faith estimate of the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System.

Interconnection System Impact Study Agreement shall mean the form of agreement contained in Appendix 2 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection System Impact Study.

IRS shall mean the Internal Revenue Service.

Joint Operating Committee shall be a group made up of representatives from Interconnection Customers and the Transmission Provider to coordinate operating and technical considerations of Interconnection Service.

Large Generating Facility shall mean a Generating Facility having a Generating Facility Capacity of more than 20 MW.

Loss shall mean any and all losses relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's performance, or non performance of its obligations under the Standard Large Generator Interconnection Agreement on behalf of the Indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the Indemnifying Party.

Material Modification shall mean those modifications that have a material impact on the cost or timing of any Interconnection Request with an equal or later Qqueue priority datePosition.

Metering Equipment shall mean all metering equipment installed or to be installed at the Generating Facility pursuant to the Standard Large Generator Interconnection Agreement at the metering points, including but not limited to instrument transformers, MWh-meters, data acquisition equipment, transducers, remote terminal unit, communications equipment, phone lines, and fiber optics.

NERC shall mean the North American Electric Reliability Council or its successor organization.

Network Resource shall mean any designated generating resource owned, purchased, or leased by a Network Customer under the Network Integration Transmission Service Tariff. Network Resources do not include any resource, or any portion thereof, that is committed for sale to third parties or otherwise cannot be called upon to meet the Network Customer's Network Load on a non-interruptible basis.

Network Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to integrate its Large Generating Facility with the Transmission Provider's Transmission System (1) in a manner comparable to that in which the Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an RTO or ISO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service in and of itself does not convey transmission service.

Network Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Transmission System required at or beyond the point at which the Interconnection Facilities connect to the Transmission Provider's Transmission System to accommodate the interconnection of the Large Generating Facility to the Transmission Provider's Transmission System.

Notice of Dispute shall mean a written notice of a dispute or claim that arises out of or in connection with the Standard Large Generator Interconnection Agreement or its performance.

OASIS shall mean the Transmission Provider's Open Access Same-Time Information System.

Optional Interconnection Study shall mean a sensitivity analysis based on assumptions specified by the Interconnection Customer in the Optional Interconnection Study Agreement.

Optional Interconnection Study Agreement shall mean the form of agreement contained in Appendix 5 of the Standard Large Generator Interconnection Procedures for conducting the Optional Interconnection Study.

Party or Parties shall mean Transmission Provider, Transmission Owner, Interconnection Customer or any combination of the above.

Permissible Technological Advancement shall mean a modification to a specific technology type submitted in an Interconnection Customer's Interconnection Request that: (1) is not a Material Modification; (2) is not a change in generation technology or fuel type, but may include advancements to other technology such as, e.g., advancements to turbines, inverters, plant supervisory controls, or other technological advancements that may affect a generating facility's ability to produce ancillary services; (3) results in electrical performance that is equal to or better than the electrical performance expected prior to the technology change; (4) does not degrade the electrical characteristics of the generating equipment (e.g., the ratings, impedances, efficiencies, capabilities, and performance of the equipment under steady state and dynamic

conditions); (5) does not increase the Interconnection Customer's requested Interconnection Service and (6) does not cause any reliability concerns (i.e., materially impact the transmission system with regard to short circuit capability limits, steady-state thermal or voltage limits or dynamic system stability and response).

Phase ("Phase 1, Phase 2, Phase 3 or Phase 4") shall mean a distinct part of the Definitive Study Process as described in Section 7.

Point of Change of Ownership shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Customer's Interconnection Facilities connect to the Transmission Provider's Interconnection Facilities.

Point of Interconnection shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Facilities connect to the Transmission Provider's Transmission System.

Provisional Interconnection Service shall mean Interconnection Service provided by Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to Transmission Provider's Transmission System and enabling that Transmission System to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Provisional Large Generator Interconnection Agreement and, if applicable, the Tariff.

Provisional Large Generator Interconnection Agreement shall mean the interconnection agreement for Provisional Interconnection Service established between Transmission Provider and/or the Transmission Owner and the Interconnection Customer. This agreement shall take the form of the Large Generator Interconnection Agreement, modified for provisional purposes.

Queue shall mean a queue for valid Interconnection Requests for the Definitive Interconnection Study Process.

Queue Position shall mean the order of a valid Interconnection Request, relative to all other pending valid Interconnection Requests, in the Definitive Interconnection

Study Process. The Queue Positionthat is established based upon the date and time of receipt of the valid Interconnection Customer satisfied all of the requirements of this Attachment M to enter the Definitive Study ProcessRequest by the Transmission-Provider.

Reasonable Efforts shall mean, with respect to an action required to be attempted

or taken by a Party under the Standard Large Generator Interconnection Agreement, efforts that are timely and consistent with Good Utility Practice and are otherwise substantially equivalent to those a Party would use to protect its own interests.

Resource Plan shall mean any process authorized or required by Applicable Laws and Regulations for, *inter alia*, the selection of Generating Facilities.

Resource Planning Entity shall mean any entity required to develop a Resource Plan or conduct a Resource Solicitation Process.

Resource Solicitation Cluster shall mean a Cluster Study associated with a Resource Plan or related process.

Resource Solicitation Process shall mean any process authorized or required by Applicable Laws and Regulations for the acquisition of Network Resources.

Scoping Meeting shall mean the meeting between representatives of the Interconnection Customer and Transmission Provider conducted for the purpose of discussing the proposed interconnection request, alternative interconnection options, to exchange information including any transmission data and earlier study evaluations that would be reasonably expected to impact affect such interconnection options, to analyze such information, and to determine the potential feasible Points of Interconnection.

Site Control shall mean the exclusive land right to develop, construct, operate, ad maintain the Generating Facility over the term of expected operation of the Generating Facility. Site Control shall include the right to develop, construct, operate and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by documentation establishingreasonably demonstrating: (1) ownership of, a leasehold interest in, or a right to develop a site of sufficient sized to construct and operate for the purpose of constructing the Generating Facility and associated Interconnection Facilities; (2) an option to purchase or acquire a leasehold interest in a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities for such purpose; or (3) any other documentation that clearly demonstrates the right of theanexclusivity or other business relationship between Interconnection Customer to exclusively occupy a site of sufficient size to construct and operate the Generating Facilityand the entity having the right to sell, lease or grant Interconnection Customer the right to possess or occupy a site for such purpose. Site Control f or any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all co-located projects that meet the aforementioned provisions of this Site Control definition.

Small Generating Facility shall mean a Generating Facility that has a Generating Facility Capacity of no more than 20 MW.

Stand Alone Network Upgrades shall mean Network Upgrades that are not part of an Affected System that an Interconnection Customer may construct without affecting day-to-day operations of the Transmission System during their construction. Both the Transmission Provider and the Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify them in Appendix A to the Standard Large Generator Interconnection Agreement. If the Transmission Provider and Interconnection Customer disagree about whether a particular Network Upgrade is a Stand Alone Network Upgrade, the Transmission Provider must provide the Interconnection Customer a written technical explanation outlining why the Transmission Provider does not consider the Network Upgrade to be a Stand Alone Network Upgrade within 15 days of its determination.

Standard Engineering and Procurement Agreement shall mean the form of engineering and procurement agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Agreement (LGIA) shall mean the form of interconnection agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Procedures (LGIP) shall mean the interconnection procedures applicable to an Interconnection Request pertaining to a Large Generating Facility that are included in the Transmission Provider's Tariff.

Start Date shall mean that actual date of the start of the System Impact Study as evidenced by the initiation of the development of the required Base Case(s), or the initiation of the technical study work, i.e., start of the powerflow/stability computer runs, whichever is applicable.

Surplus Interconnection Service shall mean any unneeded portion of Interconnection Service established in a Large Generator Interconnection Agreement, such that if Surplus Interconnection Service is utilized, the total amount of Interconnection Service at the Point of Interconnection would remain the same.

System Protection Facilities shall mean the equipment, including necessary protection signal communications equipment, required to protect (1) the Transmission Provider's Transmission System from faults or other electrical disturbances occurring at the Generating Facility and (2) the Generating Facility from faults or other electrical system disturbances occurring on the Transmission Provider's Transmission System or on

other delivery systems or other generating systems to which the Transmission Provider's Transmission System is directly connected.

Tariff shall mean the Transmission Provider's Tariff through which open access transmission service and Interconnection Service are offered, as filed with FERC, and as amended or supplemented from time to time, or any successor tariff.

Transmission Owner shall mean an entity that owns, leases or otherwise possesses an interest in the portion of the Transmission System at the Point of Interconnection and may be a Party to the Standard Large Generator Interconnection Agreement to the extent necessary.

Transmission Provider shall mean the public utility (or its designated agent) that owns, controls, or operates transmission or distribution facilities used for the transmission of electricity in interstate commerce and provides transmission service under the Tariff. The term Transmission Provider should be read to include the Transmission Owner when the Transmission Owner is separate from the Transmission Provider.

Transmission Provider's Interconnection Facilities shall mean all facilities and equipment owned, controlled, or operated by the Transmission Provider from the Point of Change of Ownership to the Point of Interconnection as identified in Appendix A to the Standard Large Generator Interconnection Agreement, including any modifications, additions or upgrades to such facilities and equipment. Transmission Provider's Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Transmission System shall mean the facilities owned, controlled or operated by the Transmission Provider or Transmission Owner that are used to provide transmission service under the Tariff.

Trial Operation shall mean the period during which Interconnection Customer is engaged in on-site test operations and commissioning of the Generating Facility prior to Commercial Operation.

Withdrawal Penalty shall have the meaning set forth in Section 3.7.1 of the LGIP.

Section 2. Scope and Application

2.1 Application of Standard Large Generator Interconnection Procedures
Sections 2 through 13 apply to processing an Interconnection Request
pertaining to a Large Generating Facility. Small Generating Facilities
requesting NRIS shall be processed under the LGIP.

2.2 Comparability

Transmission Provider shall receive, process and analyze all Interconnection Requests in a timely manner as set forth in this LGIP. Transmission Provider will use the same Reasonable Efforts in processing and analyzing Interconnection Requests from all Interconnection Customers, whether the Generating Facilities are owned by Transmission Provider, its subsidiaries or Affiliates or others.

2.3 Base Case Data

Transmission Provider shall maintain, consistent with Applicable Laws and Regulations and Applicable Reliability Standards, base power flow, short circuit and stability databases, including all underlying assumptions, and contingency list on either its OASIS site or a password-protected website, subject to confidentiality provisions in LGIP Section 13.1. In addition, Transmission Provider shall maintain network models and underlying assumptions on either its OASIS site or a password-protected website. Such network models and underlying assumptions should reasonably represent those used during the most recent Linterconnection Sstudy and be representative of current system conditions. If Transmission Provider posts this information on a password-protected website, a link to the information must be provided on Transmission Provider's OASIS site. Transmission Provider is permitted to require that Interconnection Customers, OASIS site users and password-protected website users sign a confidentiality agreement before the release of commercially sensitive information or Critical Energy Infrastructure Information in the Base Case data. Such databases and lists, hereinafter referred to as Base Cases, shall include all (1) generation projects and (2) transmission projects, including merchant transmission projects that are proposed for the Transmission System for which a transmission expansion plan has been submitted and approved by the applicable authority.

2.4 No Applicability to Transmission Service

Nothing in this LGIP shall constitute a request for transmission service or confer upon an Interconnection Customer any right to receive transmission service.

Section 3. Interconnection Requests

3.1 General

An Interconnection Customer shall submit to Transmission Provider an Interconnection Request in the form of Appendix 1 to this LGIP and the Initial Deposit of \$160,000 for an Interconnection Request of up to 75 MW

or \$250,000 for an Interconnection Request of 75 MW or greater. Transmission Provider shall apply the Initial Deposit toward the cost of the Definitiverequired Interconnection Studyies Process. Interconnection Customer shall submit a separate Interconnection Request for each site and may submit multiple Interconnection Requests for a single site. Interconnection Customer must submit an Initial Deposit with each Interconnection Request even when more than one request is submitted for a single site. An Interconnection Request to evaluate one site at two different voltage levels shall be treated as two Interconnection Requests. Similarly, an Interconnection Request to evaluate one interconnection at two or more different sites (two or more Points of Interconnection) shall be treated as separate Interconnection Requests.

At Interconnection Customer's option, Transmission Provider and Interconnection Customer will identify alternative Point(s) of Interconnection and configurations at the Scoping Meeting to evaluate in this process and attempt to eliminate alternatives in a reasonable fashion given resources and information available. Interconnection Customer will select the definitive Point(s) of Interconnection to be studied no later than the execution of the Definitive Interconnection System Impact Study Agreement. For purposes of clustering Interconnection Service requests, Transmission Provider may make reasonable changes to the requested Point(s) of Interconnection Customers at common points of interconnection.

Transmission Provider shall notify Interconnection Customers in writing of any intended changes to the requested Point(s) of Interconnection and the Point(s) of Interconnection shall only change upon mutual agreement.

Where an Interconnection Customer has submitted multiple alternative Interconnection Requests (seeking the study of either multiple alternative projects, or the study of a single project at multiple alternative Points of Interconnection), and the Scoping Meeting fails to result in a single definitive project at a single definitive Point of Interconnection, the Transmission Provider will seek to accommodate such multiple alternative requests, as follows:

- The System Impact Study will be performed in serial fashion to produce a separate version of the study for each of the Interconnection Customer's alternative requests.
- The cost of the first version of the study will be allocated equally among the participants of the System Impact Study Cluster. The full cost of producing each subsequent alternative version of the study will be paid by the Interconnection Customer pursuing the

alternative request(s).

- No more than three alternative Interconnection Requests will be permitted in any single study cluster.
- After receiving the results of the System Impact Study (in its multiple serial versions), the Interconnection Customer must designate a single Interconnection Request in order to be eligible to proceed to the Facility Study phase. Alternative Interconnection Requests will not be accommodated in the Facility Study cluster.

Transmission Provider shall use the process below to consider requests for Interconnection Service below the Generating Facility Capacity. These requests for Interconnection Service shall be studied at the level of Interconnection Service requested for purposes of Interconnection Facilities, Network Upgrades, and associated costs but may be subject to other studies at the full Generating Facility Capacity to ensure safety and reliability of the system, with the study costs borne by the Interconnection Customer. If after the additional studies are complete, Transmission Provider determines that additional Network Upgrades are necessary, then Transmission Provider must: (1) specify which additional Network Upgrade costs are based on which studies; and (2) provide a detailed explanation of why the additional Network Upgrades are necessary. Any Interconnection Facility and/or Network Upgrade costs required for safety and reliability also will be borne by the Interconnection Customer. Interconnection Customers may be subject to additional control technologies as well as testing and validation of those technologies consistent with Article 6 of the LGIA. The necessary control technologies and protection systems shall be established in Appendix C of the executed, or requested to be filed unexecuted, LGIA.

3.2 Identification of Types of Interconnection Services

At the time the Interconnection Request is submitted, Interconnection Customer must request either Energy Resource Interconnection Service or Network Resource Interconnection Service, as described below.; provided, however.any Interconnection Customer may designate only one type of-equesting Network Resource Interconnection Service <a href="may also request-that-it-be-concurrently-studied-for-each separate Interconnection Request-in-the-Queue. The type of Energy Resource Interconnection Service must-be-final.org/must-be-final.org/however, any Interconnection Request in the Queue. The type of Energy Resource Interconnection Service must-be-final.org/must-be-final.org/however, any Interconnection Service may also request-that it be-concurrently-studied-for each separate Interconnection Service must-be-final.org/of-the Definitive Study Process between Phase 2 and Phase 3 of the Definitive Interc

only if a Cluster must be re-studied in Phase 3 and otherwise may not be changed. Interconnection Customer may then elect to proceed with Network Resource Interconnection Service or to proceed under a lower level of interconnection service to the extent that only certain upgrades will be completed.

3.2.1 Energy Resource Interconnection Service

3.2.1.1 The Product.

Energy Resource Interconnection Service allows
Interconnection Customer to connect the Large
Generating Facility to the Transmission System and be
eligible to deliver the Large Generating Facility's
output using the existing firm or non-firm capacity of
the Transmission System on an "as available" basis.
Energy Resource Interconnection Service does not in
and of itself convey any right to deliver electricity to
any specific customer or Point of Delivery.

3.2.1.2 The Study.

The study consists of short circuit/fault duty, steady state (thermal and voltage) and stability analyses. The short circuit/fault duty analysis would identify direct Interconnection Facilities required and the Network Upgrades necessary to address short circuit issues associated with the Interconnection Facilities. The stability and steady state studies would identify necessary upgrades to allow full output of the proposed Large Generating Facility and would also identify the maximum allowed output, at the time the study is performed, of the interconnecting Large Generating Facility without requiring additional Network Upgrades.

3.2.2 Network Resource Interconnection Service

3.2.2.1 The Product.

Transmission Provider must conduct the necessary studies and construct the Network Upgrades needed to integrate the Large Generating Facility: (1) in a manner comparable to that in which Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an ISO or RTO with

market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service Aallows Interconnection Customer's Large Generating Facility to be designated as a Network Resource, up to the Large Generating Facility's full output, on the same basis as existing Network Resources interconnected to Transmission Provider's Transmission System, and to be studied as a Network Resource on the assumption that such a designation will occur. If the Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resource within Transmission Provider's Control Area, the Interconnection Customer must provide the point of delivery or the geographic location on EPE's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area.

3.2.2.2 The Study.

The Interconnection Study for Network Resource Interconnection Service shall assure that Interconnection Customer's Large Generating Facility meets the requirements for Network Resource Interconnection Service and as a general matter, that such Large Generating Facility's interconnection is also studied with Transmission Provider's Transmission System at peak load, under a variety of severely stressed conditions, to determine whether, with the Large Generating Facility at full output, the aggregate of generation in the local area can be delivered to the aggregate of load on Transmission Provider's Transmission System, consistent with Transmission Provider's reliability criteria and procedures. This approach assumes that some portion of existing Network Resources' output areis displaced by the output of Interconnection Customer's Large Generating Facility. Network Resource Interconnection Service in and of itself does not convey any right to deliver electricity to any specific customer or Point of Delivery. The Transmission Provider may also study the Transmission System

under non-peak load conditions. However, upon request by the Interconnection Customer, the Transmission Provider must explain in writing to the Interconnection Customer why the study of non-peak load conditions is required for reliability purposes.

3.3 Utilization of Surplus Interconnection Service.

Transmission Provider will use the process below to allow an Interconnection Customer to utilize or transfer Surplus Interconnection Service at an existing Point of Interconnection. The original Interconnection Customer or one of its affiliates shall have priority to utilize Surplus Interconnection Service. If the existing Interconnection Customer or one of its affiliates does not exercise its priority, then that service may be made available to other potential Interconnection Customers.

3.3.1 Surplus Interconnection Service Requests.

Surplus Interconnection Service requests may be made by the existing Interconnection Customer whose Generating Facility is already interconnected or one of its affiliates. Surplus Interconnection Service requests also may be made by another Interconnection Customer. Transmission Provider shall use the process in Section 3.3.2 in evaluating Interconnection Requests for Surplus Interconnection Service. Studies for Surplus Interconnection Service shall consist of reactive power, short circuit/fault duty, stability analyses, and any other appropriate studies. Steady-state (thermal/voltage) analyses may be performed as necessary to ensure that all required reliability conditions are studied. If the Surplus Interconnection Service was not studied under off-peak conditions, offpeak steady state analyses shall be performed to the required level necessary to demonstrate reliable operation of the Surplus Interconnection Service. If the original System Impact Study is not available for the Surplus Interconnection Service, both off-peak and peak analysis may need to be performed for the existing Generating Facility associated with the request for Surplus Interconnection Service. The reactive power, short circuit/fault duty, stability, and steady-state analyses for Surplus Interconnection Service will identify any additional Interconnection Facilities and/or Network Upgrades necessary.

3.3.2 Process for Securing Surplus Interconnection Service.

An existing (original) Interconnection Customer whose Large Generating Facility is already interconnected, may make available its Surplus Interconnection Service at the existing Point of Interconnection for the Large Generating Facility using the process outlined in this Section 3.3.2.

The original Interconnection Customer may retain any surplus for itself, or may transfer it to an entity of its own choosing, whether an affiliate or a non-affiliate. The amount of Surplus Interconnection Service made available cannot exceed the total amount of Interconnection Service at the Point of Interconnection established in the original Interconnection Customer's Large Generator Interconnection Agreement.

- A. Either the original Interconnection Customer, its affiliate or a thirdparty Interconnection Customer (transferee) shall notify the Transmission Provider in writing of a request to utilize or transfer any Surplus Interconnection Service made available by the original Interconnection Customer at the existing Point of Interconnection. The written Surplus Interconnection Service request should include, at a minimum, the following information; (1) the amount of Surplus Interconnection Service (in MW) proposed to be made available; (2) the start date and end date that the Surplus Interconnection Service is to be made available; (3) the type of service (Energy Resource Interconnection Service or Network Resource Interconnection Service) to be made available as Surplus Interconnection Service, provided however, that if the original Interconnection Customer's Large Generator Interconnection Agreement provides for Energy Resource Interconnection Service, any Surplus Interconnection Service must also be Energy Resource Interconnection Service; (4) a one-line diagram illustrating how the Generating Facility that is to use the Surplus Interconnection Service will connect to the original Interconnection Customer's Interconnection Facilities; (5) details on the Generating Facility that is to use the Surplus Interconnection Service (type of resource, characteristics, modeling information); (6) the signature of the original Interconnection Customer if the original Interconnection Customer is not the entity submitting the Surplus Interconnection Service request; and (7) any other relevant conditions of the original Interconnection Customer on its proposed utilization or transfer. To aid the original Interconnection Customer in initiating this process, Transmission Provider may post on OASIS a blank request form for Surplus Interconnection Service, identifying the required fields of data set forth above.
- B. Transmission Provider shall process such surplus interconnection service requests separately from other requests pending in its non-surplus interconnection queue. All requests for Surplus Interconnection Service will be subject to technical studies, as necessary, to ensure the reliable use of Surplus Interconnection Service. The Transmission Provider shall evaluate if the original

System Impact Studies are still applicable and sufficient to accommodate the new request, and may require new or additional studies to evaluate system impacts. Examples of circumstances in which the Transmission Provider would require new studies include, but are not limited to, circumstances in which the original System Impact Study(ies) are more than 3 years old and/or circumstances in which new generation facilities have been added to the Transmission Provider's system since the original Interconnection Customer's Large Generating Facility was studied. Examples of studies that may be necessary and appropriate to examine the potential for system impacts include, but are not limited to:

- i. Powerflow studies (steady-state, thermal/voltage),
- ii. Stability studies (voltage/angular),
- iii. Short circuit/fault duty studies,
- iv. Reactive power studies
- v. Transient, harmonic and/or sub-transient studies, particularly if the generation technology of the generation facility using the Surplus Interconnection Service is different from the generation technology in the original Interconnection Customer's Large Generation Interconnection Agreement.
- C. Should Transmission Provider determine that studies are necessary to evaluate a request for Surplus Interconnection Service, Transmission Provider will collect from the requestor a study deposit of \$25,000, subject to true-up to reflect the actual cost of the studies.
- D. If the technical studies show that the addition of Network Upgrades will be required for the use of Surplus Interconnection Service due to different technology or short circuit requirements, or for other reasons, Surplus Interconnection Service shall be available up to the amount that can be accommodated without Network Upgrades. The Generating Facility receiving Surplus Interconnection Service is to interconnect to the original Interconnection Customer's Interconnection Facilities. Changes or additions to the original Interconnection Customer's Interconnection Facilities may be necessary and permissible. The need for changes to other facilities identified in the original Interconnection Customer's Interconnection Agreement, including Network Upgrades, would make Surplus Interconnection Service unavailable.
- E. An agreement for Surplus Interconnection Service shall be developed between the Transmission Provider, the original Interconnection Customer and the Surplus Interconnection Service

Customer. The Transmission Provider will file the agreement for Surplus Interconnection Service with the Commission. The agreement will contain the parameters of the Surplus Interconnection Service, its level of service, term of service, and any conditions arising from the results of the Transmission Provider's evaluation of system impacts. Upon Commission approval of the Surplus Interconnection Agreement and concurrent with the effective date of the Surplus Interconnection Service, the original Interconnection Customer's Interconnection Agreement shall be deemed to be adjusted to lower the level of Interconnection Service by an amount equal to the Surplus Interconnection Service, without the need for any filings or approvals.

- F. The use of Surplus Interconnection Service does not convey any promise or grant of transmission service.
- G. Surplus Interconnection Service cannot be offered until all facilities required for the original Interconnection Customer's interconnection service (including all Contingent Facilities) are constructed and In Service.
- H. Surplus Interconnection Service cannot be offered if the original Interconnection Customer's Generating Facility is scheduled to retire and permanently cease Commercial Operation before the Surplus Interconnection Service Customer's Generating Facility begins Commercial Operation.

3.4 Valid Interconnection Request

3.4.1 Initiating an Interconnection Request

An Interconnection Customer wishing to join the Definitive
Interconnection Study Process shall submit a completed
Interconnection Request to Transmission Provider within, and no
later than the close of, the DISIS Request Window. To initiate an
Interconnection Request, Interconnection Customer must submit all
of the following:

- (i) (i) Initial Deposit of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater;
- (ii) (ii) aA completed application in the form of Appendix 1 (including all technical information); and
- (iii) (iii) A demonstration of Site Control as defined in Section 1

and addressed further in Section 7.7.7 of the LGIPor a Deposit in Lieu of Site Control in the amount of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater. Such Deposits shall be applied toward any Interconnection Studies pursuant to the Interconnection Request. Specifications for acceptable site size for the purposes of demonstrating Site Control are posted on Transmission Provider's OASIS website. Interconnection Customer may propose alternative specifications for site size to those posted on OASIS for Transmission Provider approval. In the event Transmission Provider and Interconnection Customer cannot reach agreement related to adequacy of site size, Transmission Provider will accept a Professional Engineer (licensed in the state of Texas or New Mexico, as applicable depending on the state in which the Generating Facility is to be located) stamped site plan drawing that depicts the proposed generation arrangement and specifies the maximum facility output for that arrangement;

(iv) A Point of Interconnection;

- (v) If the request is for NRIS and if Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resourced within Transmission Provider's Control Area, the point of delivery or the geographic location on Transmission Provider's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area;
- (vi) A Generating Facility size (MW) (and requested

 Interconnection Service amount if the requested

 Interconnection Service is less than the Generating Facility

 Capacity);
- (vii) One of the following Readiness Milestone ("M1") options
 totaling the entire capacity of the Generating Facility (or
 requested Interconnection Service amount if the requested
 Interconnection Service amount is less than the Generating
 Facility Capacity or security equal to one times the study
 deposit described above (see Initial Deposit) in the form of an
 irrevocable letter of credit or cash in lieu of the Readiness

Milestone. The security is refunded to the Interconnection Customer according to Section 7.7.5.

- a. Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale (1) of the constructed Generating Facility, or (2) of the Generating Facility's energy, or (3) of the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
- b. Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c. Provisional Large Generator Interconnection

 Agreement filed with FERC that contains a

 commitment to move forward with constructing the

 Generating Facility and is not suspended; and
- (viii) Security equal to one times the study Initial Deposit in the form of an irrevocable letter of credit or cash. The security is refunded to the Interconnection Customer according to Section 7.7.

In the event the Interconnection Request is withdrawn or deemed withdrawn before the Start Date, \$25,000 of the Initial Deposit will-be non-refundable. If the Interconnection Customer has not demonstrated Site Control as of the Start Date, \$25,000 of the Deposit in Lieu of Site Control shall be non-refundable.

In the event the Interconnection Request is withdrawn or deemed withdrawn after the execution of the Facilities Study Agreement, in accordance with Section 8.1, \$50,000 of the Initial Deposit will be non-refundable. If the Interconnection Customer has not demonstrated Site Control at the time of this withdrawal, \$50,000 of the Deposit in Lieu of Site Control shall be non-refundable.

These non-refundable amounts shall not be cumulative and shall be in addition to the actual costs incurred by Transmission Provider up to the date of the withdrawal.

The expected In-Service Date of the new Large Generating Facility or increase in capacity of the existing Generating Facility shall be no

more than the process window for the regional expansion planning period (or in the absence of a regional planning process, the process window for Transmission Provider's expansion planning period) not to exceed seven years from the date the Interconnection Request is received by Transmission Provider, unless Interconnection Customer demonstrates that engineering, permitting and construction of the new Large Generating Facility or increase in capacity of the existing Generating Facility will take longer than the regional expansion planning period. The In-Service Date may succeed the date the Interconnection Request is received by Transmission Provider by a period up to ten years, or longer where Interconnection Customer and Transmission Provider agree, such agreement not to be unreasonably withheld.

3.4.2 Acknowledgment of Interconnection Request

Transmission Provider shall acknowledge receipt of the Interconnection Request within five (5) Business Days of the DISIS Request Windowrequest and attach a copy of the received Interconnection Request to the acknowledgement.

3.4.3 Deficiencies in Interconnection Request

An Interconnection Request will not be considered to be a valid request until all items in Section 3.4.1 have been received by Transmission Provider. If an Interconnection Request fails to meet the requirements set forth in Section 3.4.1, Transmission Provider shall notify Interconnection Customer within five (5) Business Days of the closereceipt of the DISISinitial Interconnection Request Window of the reasons for such failure and that the Interconnection Request does not constitute a valid request. Interconnection Customer shall provide Transmission Provider the additional requested information needed to constitute a valid request within ten (10) Business Days after receipt of such notice. At any time, if Transmission Provider identifies issues with technical data provided by Interconnection Customer, Interconnection Customer and Transmission Provider shall work expeditiously and in good faith to remedy any data issues. If the data issues are remedied by the Interconnection Customer without adversely affecting the Transmission Provider's study timeline, the remedy will be accommodated. Failure by Interconnection Customer to comply with this Section 3.4.3 shall be treated in accordance with Section 3.7.

Transmission Provider shall determine if the information contained

in the Interconnection Request is adequately sufficient to start the Definitive System Impact Study by the close of the Customer Engagement Window.

3.4.4 Scoping Meeting

Within ten (10) Business Days after the closereceipt of the DISISavalid Interconnection Request Window, Transmission Provider shall host a Scoping Meeting for all Interconnection Requests received in that DISIS Request Window. If requested by establish a date agreeable to Interconnection Customer, Transmission Provider for the Scoping Meeting, and such date shall also hold individual customer specific Scoping Meetings, which must be requested no later than fifteen (15) business days after thirty (30) Calendar Days from receipt of the valid Interconnection Request, unless otherwise mutually agreed upon by the close of the DISIS Request Window Parties.

The purpose of the Scoping Meeting shall be to discuss alternative interconnection options, to exchange information including any transmission data that would reasonably be expected to impact such interconnection options, to analyze such information and to determine the potential feasible Points of Interconnection. Transmission Provider and Interconnection Customer will bring to the meeting such technical data, including, but not limited to: (i) general facility loadings, (ii) general instability issues, (iii) general short circuit issues, (iv) general voltage issues, and (v) general reliability issues as may be reasonably required to accomplish the purpose of the meeting. Transmission Provider and Interconnection Customer will also bring to the meeting personnel and other resources as may be reasonably required to accomplish the purpose of the meeting in the time allocated for the meeting. On the basis of the meeting, Interconnection Customer shall designate its Point of Interconnection, and one or more available alternative Point(s) of Interconnection. The duration of the meeting shall be sufficient to accomplish its purpose.

3.5 OASIS Posting

3.5.1 OASIS Posting

Transmission Provider will maintain on its OASIS a list of all Interconnection Requests. The list will identify, for each Interconnection Request: (i) the maximum summer and winter megawatt electrical output; (ii) the location by county and state; (iii) the station or transmission line or

lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the status of the Interconnection Request, including Queue Position; (vi) the type of Interconnection Service being requested; and (vii) the availability of any studies related to the Interconnection Request; (viii) the date of the Interconnection Request; (ix) the type of Generating Facility to be constructed (combined cycle, base load or combustion turbine and fuel type); and (x) for Interconnection Requests that have not resulted in a completed interconnection, an explanation as to why it was not completed. Except in the case of an Affiliate, the list will not disclose the identity of Interconnection Customer until Interconnection Customer executes an LGIA or requests that Transmission Provider file an unexecuted LGIA with FERC. Before holding a Scoping Meeting with its Affiliate, Transmission Provider shall post on OASIS an advance notice of its intent to do so. Transmission Provider shall post to its OASIS site any deviations from the study timelines set forth herein. Interconnection Study reports and Optional Interconnection Study reports shall be posted to Transmission Provider's OASIS site subsequent to the meeting between Interconnection Customer and Transmission Provider to discuss the applicable study results. Transmission Provider shall also post any known deviations in the Large Generating Facility's In-Service Date.

3.5.2 Requirement to Post Interconnection Study Metrics

Transmission Provider will maintain on its OASIS or its website summary statistics related to processing Interconnection Studies pursuant to Interconnection Requests, updated quarterly. If Transmission Provider posts this information on its website, a link to the information must be provided on Transmission Provider's OASIS site. For each calendar quarter, Transmission Providers must calculate and post the information detailed in sections 3.5.2.1 through 3.5.2.3. Where an Interconnection Customer is a member of a Queue Cluster and such customer tenders to Transmission Provider an executed study agreement before the close of the Queue Cluster Window, Transmission Provider will deem its date of receipt of the study agreement to be the day immediately after the close of the Queue Cluster Window for purposes of calculating and posting processing time under Sections 3.5.2.1 and 3.5.2.2.

3.5.2.1 Interconnection System Impact Studies processing time.

(A) Number of Interconnection Requests that had <u>Phase</u>
<u>Interconnection System Impact</u> Studies completed within Transmission Provider's coordinated region during the

reporting quarter, and the number of Interconnection
Requests that had Phase 2 and Phase 3 Studies completed
within Transmission Provider's coordinated region during the
reporting quarter,

- (B) Number of Interconnection Requests that had Phase 1-Interconnection System Impact Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than ninety (90) Calendar Days after the close of the Queue Cluster Window, or more than ninety (90) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact Study Agreements, whichever is later, and the number of Interconnection Requests that had Phase 2 and Phase 3 Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than one hundred fifty (150) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact Study Agreements,
- (C) At the end of the reporting quarter, the number of active valid Interconnection Requests with ongoing incomplete System Impact Studies where such Interconnection Requests had executed Definitive Interconnection System Impact Study Agreements received by Transmission Provider more than one hundred fifty (150) ninety (90) Calendar Days after the close of the Queue Cluster Window, or more than one hundred fifty (150) ninety (90) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection Impact Study Agreements, whichever is later, before the reporting quarter end,
- (D) Mean time (in days), <u>Definitive</u> Interconnection System Impact Studies completed within Transmission Provider's coordinated region during the reporting quarter, from the date when Transmission Provider received the executed Interconnection System Impact Study Agreement(s) to the date when Transmission Provider provided the completed Interconnection System Impact Study to the Interconnection Customer(s),

(E) Percentage of Interconnection System Impact Studies exceeding the timelines set forth above to complete this reporting quarter, calculated as the sum of 3.5.2.1(B) plus 3.5.2.1(C) divided by the sum of 3.5.2.1(A) plus 3.5.2.1(C)).

3.5.2.2 Interconnection Facilities Studies processing time.

- (A) Number of Interconnection Requests that had Interconnection Facilities Studies that are completed within Transmission Provider's coordinated region during the reporting quarter and tendered to the Interconnection Customer in draft form,
- (B) Number of Interconnection Requests that had Interconnection Facilities Studies that are completed within Transmission Provider's coordinated region during the reporting quarter that were completed and tendered to the Interconnection Customer in draft form more than ninety (90) Calendar Days after receipt by Transmission Provider of the Interconnection Customer's executed Interconnection Facilities Study Agreement,
- (C) At the end of the reporting quarter, the number of active valid Interconnection Service requests with ongoing incomplete Interconnection Facilities Studies where such Interconnection Requests had executed Interconnection Facilities Studies Agreement received by Transmission Provider more than ninety (90) Calendar Days before the reporting quarter end,
- (D) Mean time (in days), for Interconnection Facilities Studies completed within Transmission Provider's coordinated region during the reporting quarter, calculated from the date when Transmission Provider received the executed Interconnection Facilities Study Agreement to the date when Transmission Provider provided the completed draft Interconnection Facilities Study to the Interconnection Customer,
- (E) Percentage of delayed Interconnection Facilities Studies this reporting quarter, calculated as the sum of 3.5.2.2(B) plus 3.5.2.2(C) divided by the sum of 3.5.2.2(A) plus 3.5.2.2(C)).

3.5.2.3 Interconnection Service requests withdrawn from Interconnection queue.

- (A) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter,
- (B) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of any interconnection studies or execution of any interconnection study agreements,
- (C) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of an Interconnection System Impact Study,
- (D) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of an Interconnection Facilities Study,
- (E) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue after execution of a generator interconnection agreement or Interconnection Customer requests the filing of an unexecuted, new interconnection agreement,
- (F) Mean time (in days), for all withdrawn Interconnection Requests, from the date when the request was determined to be valid to when Transmission Provider received the request to withdraw from the queue.

3.5.3

Transmission Provider is required to post on OASIS or its website the measures in paragraph 3.5.2.1(A) through paragraph 3.5.2.3(F) for each calendar quarter within 30 days of the end of the calendar quarter. Transmission Provider will keep the quarterly measures posted on OASIS or its website for three calendar years with the first required report to be in the first quarter of 2020. If Transmission Provider retains this information on its website, a link to the information must be provided on Transmission Provider's OASIS site.

3.5.4

In the event that any of the values calculated in paragraphs 3.5.2.1(E), or 3.5.2.2(E) exceeds 25 percent for two consecutive calendar quarters, Transmission Provider will have to comply with the measures below for the next four consecutive calendar quarters and must continue reporting this information until Transmission Provider reports four consecutive calendar quarters without the values calculated in 3.5.2.1(E), or 3.5.2.2(E) exceeding 25 percent for two consecutive calendar quarters:

- (i) Transmission Provider must submit a report to the Commission describing the reason for each study or group of clustered studies pursuant to an Interconnection Request that exceeded its deadline (i.e., 45, 90, 150 or 180-90 days) for completion (excluding any allowance for Reasonable Efforts). Transmission Provider must describe the reasons for each study delay and any steps taken to remedy these specific issues and, if applicable, prevent such delays in the future. The report must be filed at the Commission within 45 days of the end of the calendar quarter.
- (ii) Transmission Provider shall aggregate the total number of employee-hours and third party consultant hours expended towards interconnection studies within its coordinated region that quarter and post on OASIS or its website. If Transmission Provider posts this information on its website, a link to the information must be provided on Transmission Provider's OASIS site. This information is to be posted within 30 days of the end of the calendar quarter.

3.6 Coordination with Affected Systems

Transmission Provider will coordinate the conduct of any studies required to determine the impact of the Interconnection Request on Affected Systems with Affected System Operators and, if possible, include those results (if available) in its applicable Interconnection Study within the time frame specified in this LGIP. Transmission Provider will include such Affected System Operators in all meetings held with Interconnection Customer as required by this LGIP. Interconnection Customer will cooperate with Transmission Provider in all matters related to the conduct of studies and the determination of modifications to Affected Systems. A Transmission Provider which may be an Affected System shall cooperate with Transmission Provider with whom interconnection has been requested in all matters related to the conduct of studies and the determination of

modifications to Affected Systems. It is the responsibility of the Affected System owner to provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to (i) complete any interconnection studies and (ii) construct any necessary interconnection facilities and network upgrades needed to reliably interconnect at the requested service level.

3.7 Withdrawal

Interconnection Customer may withdraw its Interconnection Request at any time by written notice of such withdrawal to Transmission Provider. In addition, if Interconnection Customer fails to adhere to all requirements of this LGIP, except as provided in Section 13.5 (Disputes), Transmission Provider shall deem the Interconnection Request to be withdrawn and shall provide written notice to Interconnection Customer of the deemed withdrawal and an explanation of the reasons for such deemed withdrawal. Upon receipt of such written notice, Interconnection Customer shall have fifteen (15) Business Days in which to either respond with information or actions that cures the deficiency or to notify Transmission Provider of its intent to pursue Dispute Resolution.

Withdrawal shall result in the loss of Interconnection Customer's Queue Position. If an Interconnection Customer disputes the withdrawal and loss of its Queue Position, then during Dispute Resolution, Interconnection Customer's Interconnection Request is eliminated from the Qqueue until such time that the outcome of Dispute Resolution would restore its Queue Position. An Interconnection Customer that withdraws or is deemed to have withdrawn its Interconnection Request shall pay to Transmission Provider all costs that Transmission Provider prudently incurs with respect to that Interconnection Request prior to Transmission Provider's receipt of notice described above. Interconnection Customer must pay all monies due to Transmission Provider before it is allowed to obtain any Interconnection Study data or results.

In the case of a withdrawal, Transmission Provider shall (i) update the OASIS Queue Position posting; (ii) impose the Withdrawal Penalty described in Section 3.7.1; (iii) refund any security after settling the final invoice (see Section 7.7.5); and (ivi) refund to Interconnection Customer any of the refundable portion of Interconnection Customer's study depositor study payments that exceeds the share of the costs that Transmission Provider has incurred (less any non-refundable amounts described above in this LGIPSection 3.3.1). The application of interest on amounts subject to refund shall be made in a manner consistent with Commission precedent

and Interest on a refund shall be calculated in accordance with section 35.19a(a)(2) of FERC's regulations. In the event of such withdrawal, Transmission Provider, subject to the confidentiality provisions of Section 13.1, shall provide, at Interconnection Customer's request, all information that Transmission Provider developed for any completed study conducted up to the date of withdrawal of the Interconnection Request.

3.7.1 Withdrawal Penalty

Interconnection Customers shall be subject to a Withdrawal Penalty if they withdraw their request from the Queue or the Generating Facility does not otherwise reach Commercial Operation unless the withdrawal does not negatively affect the timing or cost of equal or lower queued projects.

3.7.1.1 Calculation of the Withdrawal Penalty.

If the Interconnection Customer provided a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be equal to the higher of the study deposit or one (1) times of its actual allocated cost of the Definitive Interconnection Study Process. If the Interconnection Customer did not provide a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be dependent on the Phase in which the Interconnection Customer withdraws and shall be calculated as follows: 1. If the Interconnection Customer withdraws in Phase 1 (after M1, but before M2), the Withdrawal Penalty shall be the higher of the study deposit or two (2) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one (1) million dollars. 2. If the Interconnection Customer withdraws in Phase 2 (after M2, but before M3), the Withdrawal Penalty shall be the higher of the study deposit or three (3) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one and one half (1.5) million dollars. 3. If the Interconnection Customer withdraws in Phase 3 (after M3, but before M4), the study cost obligation shall be the higher of the study deposit or five (5) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at two (2) million dollars. 4. If the Interconnection Customer withdraws in Phase 4 (after M4, but before M5), the Withdrawal Penalty shall be the higher of the study deposit or seven (7) times its actual allocated cost of the Definitive

Filed Date: 05/31/2023

Interconnection Study Process. This amount shall be capped at two and a half (2.5) million dollars.

The Withdrawal Penalty for any customer that has executed an LGIA is the higher of the study deposit or nine (9) times its actual allocated cost of the Definitive Interconnection Study Process.

3.7.1.2 Distribution of the Withdrawal Penalty.

Any Withdrawal Penalty revenues shall be used to fund generation interconnection studies. Withdrawal Penalty revenues shall first be applied, in the form of a bill credit, to not-yet-invoiced study costs for other Interconnection Customers in the same cluster, and to the extent that such studies are fully credited, shall be applied to study costs of future clusters in Queue order. Withdrawn Interconnection Customers shall not receive a bill credit associated with Withdrawal Penalties. Distribution of Withdrawal Penalty revenues to a specific study shall not exceed the total actual study costs. Allocation of Withdrawal Penalty revenues within a cluster to a specific customer shall be comparable to the allocation of study costs described in Section 4.2. Specifically, the Withdrawal Penalty revenue distribution to each customer in a specific cluster, shall be (1) fifty percent (50%) on a per capita basis based on number of Interconnection Requests in the applicable Cluster; and (2) fifty percent (50%) to Interconnection Customers on a pro-rata basis based on requested megawatts included in the applicable Cluster. Distribution of Withdrawal Penalty revenue associated with Readiness Milestone 5 shall not be distributed to the remaining customers in that cluster until all customers in that cluster have reached Commercial Operation and thereafter shall be distributed as described above. Transmission Provider shall not change the distribution of Withdrawal Penalty revenue without authorization by the Commission. Transmission Provider shall post the Withdrawal Penalty balance on its OASIS site.

3.8 Identification of Contingent Facilities.

The Transmission Provider shall identify Contingent Facilities in the System Impact Study Report (including any restudy reports) using the steps outlined below. The method permits the parties to determine why a specific Contingent Facility was identified and how it relates to the Interconnection Request.

3.8.1 Baseline assumptions.

Transmission Provider uses a technical screening process to identify Contingent Facilities that starts with the baseline assumption that the following are in service: (i) Generating Facilities that are directly interconnected to the Transmission System; (ii) Generating Facilities that are interconnected to Affected Systems where the Affected System(s) have communicated to Transmission Provider, or Transmission Provider otherwise has determined, that such facilities may have an impact on the Interconnection Request; (iii) Generating Facilities that have a pending higher-queued Interconnection Request to interconnect to the Transmission System and their associated Interconnection Facilities and Network Upgrade requirements, to the extent those higher-queued Interconnection Requests have been the subject of a System Impact Study and/or a Facilities Study; (iv) Generating Facilities that executed an interconnection agreement, or requested that an unexecuted interconnection agreement be filed with FERC, and their associated Interconnection Facilities and Network Upgrades; (v) higher-queued requests for transmission service and their associated facilities or upgrade requirements to the extent they have an impact on the Interconnection Request and to the extent that those higher-queued requests for transmission service have been the subject of a System Impact Study and/or a Facilities Study; (vi) Transmission Provider's transmission expansion plan components; and (vii) the transmission expansion plan components of third-party transmission providers, to the extent that Transmission Provider has determined that they have an impact on the Interconnection Request.

With respect to the treatment of higher-queued requests for interconnection service and/or transmission service, in situations in which (a) the higher-queued requests have not yet proceeded through the System Impact Study process to reach the stage where facilities necessary to accommodate the requests have been identified, or (b) facilities associated with higher-queued requests for service change as a result of queue withdrawal or otherwise, Transmission Provider shall adjust its baseline assumptions with the most current information available and produce a re-study System Impact Study Report on Interconnection Customer's Interconnection Request. The re-study report will use the method described in this Section to permit the parties to determine why a specific Contingent Facility was identified and how it relates to the Interconnection Request.

3.8.2. Technical Screening Process.

The technical screening process for identifying Contingent Facilities is comprised of the following steps:

Step 1: Identify Potential Contingent Facilities.

Transmission Provider will review all applicable Interconnection and Transmission Service study results for higher-queued Interconnection or Transmission Service Requests to identify any unbuilt Interconnection Facilities and/or Network Upgrades as potential Contingent Facilities to be evaluated pursuant to Steps 2-5 below.

Step 2: Remove a Potential Contingent Facility and Perform Applicable Contingency Analyses. The Transmission Provider will take a potential Contingent Facility (and its associated unbuilt Generating Facility) out of service in its study model and: (a) perform steady state, short circuit, voltage stability, and/or transient stability analyses to determine if the Transmission System demonstrates acceptable pre- and post-contingency system performance, in accordance with current Transmission Provider, WECC, NERC, or Reliability Coordinator criteria or standards; and (b) document the resulting Transmission System performance deficiencies following the analysis in Step 2(a). In implementing this step in situations in which higher-queued clustered requests involve a potential Contingent Facility, Transmission Provider will remove all unbuilt Interconnection Facilities and/or Network Upgrades in the cluster, as well as their associated Generating Facilities in the cluster.

Step 3: Add the proposed Generating Facility into Model and Rerun Contingency Analyses. Transmission Provider will add the proposed Generating Facility into the model after taking the potential Contingent Facility and its associated Generating Facility out of service as provided in Step 2 above, and: (a) perform the same analysis for the added proposed Generating Facility as the analysis outlined in Step 2(a) for the removed potential Contingent Facility; and (b) document the resulting Transmission System performance deficiencies following the analysis in Step 3(a).

Step 4: Apply Threshold and Categorize. If the Transmission System performance deficiencies observed in Step 3(b) are: (a) exacerbated by one percent (1%) or greater than the Transmission System performance deficiencies initially observed in Step 2(b), then the potential Contingent Facility that is individually evaluated in

Step 2 will be deemed a Contingent Facility; or (b) exacerbated by less than one percent (1%) than the Transmission System performance deficiencies initially observed in Step 2(b), then the potential Contingent Facility that is individually evaluated in Step 2 will not be deemed a Contingent Facility. The only variation from this analysis will apply to short circuit criteria. For the performance of the short circuit analysis of the Transmission Provider's Transmission System in its Balancing Authority Area, all generation directly connected to the Transmission Provider's Transmission System is assumed to be connected and synchronized to the grid. The fault current at any substation that contains one or more circuit breakers exceeding 95% of the interruption rating of any circuit breaker in the substation where the fault is taken will be considered a Contingent Facility.

- Step 5: Repeat for Each Identified Potential Contingent Facility. Transmission Provider will repeat Steps 2-4 for each potential Contingent Facility identified in Step 1.
- **Step 6: Per Se Contingent Facilities.** Notwithstanding Steps 1-5, an Interconnection Facility or Network Upgrade of a higher-queued request for service shall automatically be deemed a Contingent Facility if such Interconnection Facility or Network Upgrade would be necessary for the proper functioning of the proposed Generating Facility's System Protection Facilities (as defined in Appendix 6 to Attachment M of Transmission Provider's OATT).
- 3.8.3. The Interconnection System Impact Study report will list Contingent Facilities in an appendix, which will include: (a) a description of each Contingent Facility; and (b) the Interconnection Request, transmission service request or planned project for which the Contingent Facility was initially required. This list of Contingent Facilities is subject to updates if a System Impact Study is re-studied pursuant to the Tariff under the LGIP or the provisions governing transmission service requests. In addition, where the Transmission Provider has identified an Affected System pursuant to Section 3.6 and facilities have been identified to mitigate adverse impacts on an Affected System, such facilities shall be included on the list of Contingent Facilities to the extent they have an impact on the Interconnection Request.
- **3.8.4.** If requested by the Interconnection Customer, and if readily available

and not commercially sensitive, Transmission Provider will also provide an estimate of the costs of and the in-service date for each Contingent Facility, which may be subject to later updates if a Contingent Facility's estimated costs and in-service dates change.

Section 4. <u>Interconnection Request Evaluation Process Queue Position</u>

4.1 Queue Position General

4.1.1 Assignment of Queue Postion

Transmission Provider shall assign a Queue Position <u>as follows: the Queue Position within the Queue shall be assigned</u> based upon the date and time of receipt of <u>all items</u>the valid Interconnection Request; provided that, if the sole reason an Interconnection Request is not valid is the lack of required <u>pursuantinformation</u> on the application form, and Interconnection Customer provides such information in accordance with Section 3.4.3, then Transmission Provider shall assign Interconnection Customer a Queue Position based on the date the application form was originally filed. Moving a Point of Interconnection shall result in a lowering of Queue Position if it is deemed a Material Modification under Section 4.4.3.

The Queue Position of each Interconnection Request will be used to the provisions of Section 3.4. There is no queue for informational determine the order of performing the Interconnection Studies and determination of cost responsibility for the facilities necessary to accommodate the Interconnection Request.

4.1.2 Higher Queue Position

A higher <u>Queue Position assigned to an queued</u> Interconnection Request is one that has been placed "earlier" in the <u>Qqueue in relation to another</u> Interconnection Request that is <u>assigned a lower queued.</u>

Transmission Provider may allocate the cost of the common upgrades for clustered Interconnection Requests without regard to Queue Position.

4.2 Clustering

Interconnection Requests are to be studied in clusters for the purpose of the Interconnection System Impact Study and the Interconnection Facilities Study.

Clustering shall be implemented on the basis of Queue Position. All requests studied in a and geographic location of the proposed Interconnection Point on the Transmission Provider's Transmission Systemand/or the interconnecting voltage level. All Interconnection Requests received during the second and third quarters of a given year (i.e., beginning April 1 and closing September 30) will be grouped into one "Queue Cluster shall be considered equally queued but Clusters initiated earlier in time shall be considered to have a higher Queue Position than clusters initiated later. The Queue Position of Window," and all-Interconnection Requests received during the fourth quarter of a year and the first quarter of the following year (i.e., beginning October 1 and closing-March 31 of the following year) will be placed into a separate "Queue-Cluster Window." Interconnection Requests shall be grouped in their respective Queue Cluster Window and by geographical areas and/or theinterconnecting voltage level, and shall be studied together for Network Resource Interconnection Service without regard to the nature of the requested Interconnection Service, whether Energy Resource Interconnection Service or Network Resource Interconnection Service. The deadline for completing all Interconnection System Impact Studies for which an Interconnection Request shall have no bearing on the allocation of the cost of the upgrades identified in the applicable System Impact Study Agreement has been executed during a Queue Cluster Study (such costs will be allocated among Interconnection Requests Window shall be in accordance with Section 4.2.2. Moving a Point of Interconnection shall result in a loss of Queue Position if it is deemed a Material Modification under Section 4.4.37.4, for all Interconnection Requests assigned to the same Queue Cluster Window. Transmission Provider may study an-Interconnection Request separately to the extent warranted by Good Utility-Practice based upon the electrical remoteness of the proposed Large Generating Facility.

Any changes to the established Queue Cluster Window interval and opening or closing dates shall be announced with a posting on Transmission Provider's OASIS beginning at least one hundred and eighty (180) Calendar Days in advance of the change and continuing thereafter through the end date of the first Queue Cluster Window that is to be modified.

4.2 Clustering and the Definitive Study Process

The diagram attached in the At-A-Glance Reference Sheet provides an overview of the Definitive Interconnection Study Process.

4.2.1 Cluster Window Transition Period

Transmission Provider shall accept Interconnection Requests during an approximate sixty (60) Calendar Day period referred to as the "DISIS Request Window." A DISIS Request Window shall open annually on or about February 1st and close on March 31th or the following Business Day if March 31th falls on a weekend or NERC recognized holiday. A second DISIS Request Window shall open annually on August 1st and close on September 30th or the following Business Day if September 30th falls on a Saturday or Sunday.

If one or more valid requests are received, for sixty (60) Calendar Days following the close of the DISIS Request Window (the "Customer Engagement Window"), Transmission Provider shall work with applicable Interconnection Customers to test models, verify data, hold stakeholder meetings (including Scoping Meetings, as appropriate), work with requestors to address any modeling irregularities identified by Transmission Provider in its evaluation of the Interconnection Request, and generally prepare for the start of the Definitive Interconnection System Impact Study. The first Queue Cluster Window will commence upon the first Window deadline following Commission approval of the Oueue Cluster-Windows. Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each <u>Interconnection Request: (i) the requested amount of Interconnection</u> Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar.

At the end of the Customer Engagement Window, all Interconnection
Requests deemed sufficient that have an executed DISIS Agreement shall
be included in that DISIS Cluster. Any Interconnection Requests not
deemed sufficient or undergoing Dispute Resolution at the close of the
Customer Engagement Window shall not be included in that DISIS Cluster.
Immediately following the Customer Engagement Window, Transmission
Provider shall initiate the Definitive Interconnection System Impact Study
described in more detail in Section 7.

Filed Date: 05/31/2023

Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study.

4.2.1 Initiation of a Resource Solicitation Cluster

Upon request of a Resource Planning Entity, Transmission Provider may initiate the study of a Resource Solicitation Cluster. The Resource Solicitation Cluster shall respect Queue Position and shall be studied as its own Cluster. Within ten (10) Business Days of receipt of a request to perform a Resource Solicitation Cluster that includes valid Interconnection Requests as described in Section 3.4, Transmission Provider and Resource Planning Entity shall meet to determine a mutually agreeable scope of study and timeframe to initiate the Resource Solicitation Cluster. The timeline shall indicate the close of the Customer Engagement Window for that Resource Solicitation Cluster. Thereafter the Definitive Interconnection System Impact Study shall proceed as described in Section 7.

In order to initiate Transmission Provider's study of Interconnection Requests made in connection with a Resource Solicitation Process Resource Planning Entity must: (a) act as the authorized representative for all Interconnection Requests submitted to the Resource Solicitation Cluster; (b) submit all Interconnection Requests arising from the Resource Solicitation Process at the same time to ensure an equal Queue Position for all Generating Facilities included in the Resource Solicitation Cluster; (c) cooperate with Transmission Provider in conducting the studies; and (d) request a reasonable number of different combinations of such Interconnection Requests to meet Resource Planning Entity's identified need and assumptions in the Resource Solicitation Process. Such studies in connection with a Resource Solicitation Process shall be implemented based upon Oueue Position (relative to higher or lower queued clusters) and shall consider Resource Planning Entity's needs and assumptions identified in the Resource Solicitation Process. Transmission Provider shall not be obligated to implement a Resource Solicitation Cluster more than once a year.

The Resource Planning Entity may submit for inclusion in the Resource Solicitation Process an Interconnection Request for a Generating Facility that already has a higher Queue Position pursuant to Section 4.1.1. A Generating Facility that initially is associated with a Queue Position through the Resource Solicitation Process may also reserve a lower Queue Position separate from the Resource Solicitation Process pursuant to Section 4.1.1. In either case, Interconnection Customer must meet all requirements associated with maintaining each Queue Position for the Generating Facility. In the event a Generating Facility has multiple Queue Positions, it shall not be double counted in the study models.

A Generating Facility in the Resource Solicitation Process is subject to study according to the Queue Position of the Resource Solicitation Cluster. A Generating Facility that is not a part of the Resource Solicitation Process is also subject to study according to its Queue Position. All studies must be performed in accordance with the provisions of the LGIP, and may not be delayed as a result of the Resource Solicitation Process.

After Transmission Provider completes the Definitive
Interconnection System Impact Studies for the requested
combinations, the results will be provided (Phase 1 Reports, Phase 2
Reports, Phase 3 Reports, etc.; as applicable under Section 7.4) to
the Resource Planning Entity for use in the Resource Solicitation
Process. The results will be posted on Transmission Provider's
OASIS consistent with the posting of other study results.

After receipt of the Phase 1 Report, Resource Planning Entity must select one of the studied combinations to advance to Phase 2 of the study process. After receipt of the Phase 2 Report, Resource Planning Entity must select one of the studied combinations prior to the commencement of any Interconnection Facilities Study associated with the Resource Solicitation Process. Prior to the completion of the Interconnection Facilities Study of all of the components of the selected combination, Resource Planning Entity may replace components, subject to any necessary Re-Study pursuant to Sections 7.6 or 8.5. While conducting the Definitive Interconnection Study Process, Transmission Provider may suspend further action on the Interconnection Requests in the Resource Solicitation Process that are not included in the selected combination. Once a Generating Facility is rejected in the Resource Solicitation Process, the Generating Facility shall lose the Queue

Position it held as part of the Resource Solicitation Process. If a Generating Facility is selected by Resource Planning Entity at the conclusion of the Resource Solicitation Process, the Generating Facility may no longer maintain more than one Queue Position.

4.2.2 Network Upgrades Cost Allocation

For Network Upgrades identified in a Queue Cluster study, the Transmission Provider shall calculate each Interconnection Customer's share of Network Upgrade costs in the following manner:

- (a) The costs for station equipment, including all switching stations, will be allocated on a *pro rata* basis based on the number of Generating Facilities interconnecting at an individual station.
- (b) The costs for all transmission lines, transformers and voltagesupport related Network Upgrades will be allocated on a *pro rata* basis based on the proportional capacity of each individual Generating Facility in the Queue Cluster requiring such Network Upgrades.

4.3 Transferability of Queue Position

An Interconnection Customer may transfer its Queue Position to another entity only if such entity acquires the specific Generating Facility identified in the Interconnection Request and the Point of Interconnection does not change.

4.4 Modifications

Interconnection Customer shall submit to Transmission Provider, in writing, modifications to any information provided in the Interconnection Request. Interconnection Customer shall retain its Queue Position if the modifications are in accordance with Sections 4.4.1, 4.4.2 or 4.4.5, or are determined not to be Material Modifications pursuant to Section 4.4.3.

Notwithstanding the above, during the course of the Interconnection Studies, either Interconnection Customer or Transmission Provider may identify changes to the planned interconnection that may improve the costs and benefits (including reliability) of the interconnection, and the ability of the proposed change to accommodate the Interconnection Request. Subject to the foregoing sentence, and provided, however, they do not result in a

material modification, tTo the extent the identified changes are acceptable to Transmission Provider, and Interconnection Customer and potentially impacted Interconnection Customers in the same Cluster, such acceptance not to be unreasonably withheld, Transmission Provider shall modify the Point of Interconnection and/or configuration in accordance with such changes and proceed with any re-studies necessary to do so in accordance with Section 7.6 and Section 8.5 as applicable and Interconnection Customer shall retain its Queue Position.

- A.4.1 No later than thirty (30) Calendar Days after the close of the DISIS

 Request Window and pPrior to the return of the executed Definitive
 Interconnection System Impact Study Agreement to Transmission
 Provider, the modifications permitted under this Section shall
 include specifically: (a) a decrease of up to 60 percent of electrical
 output (MW) of the proposed project, through either (1) a decrease
 in plant size or (2) a decrease in Interconnection Service level
 (consistent with the process described in Section 3.1) accomplished
 by applying Transmission Provider-approved injection-limiting
 equipment; (b) modifying the technical parameters associated with
 the Large Generating Facility technology or the Large Generating
 Facility step-up transformer impedance characteristics; and (c)
 modifying the interconnection configuration. For plant increases, the
 incremental increase in plant output will go to the end of the Qqueue
 for the purposes of cost allocation and study analysis.
- **4.4.2** Prior to the return of the executed Interconnection Facilityies Study Agreement to Transmission Provider, the modifications permitted under this Section shall include specifically: (a) additional 15 percent decrease of electrical output of the proposed project through either (1) a decrease in plant size (MW) or (2) a decrease in Interconnection Service level (consistent with the process described in Section 3.1) accomplished by applying Transmission Providerapproved injection-limiting equipment; (b) Large Generating Facility technical parameters associated with modifications to Large Generating Facility technology and transformer impedances; provided, however, the incremental costs associated with those modifications are the responsibility of the requesting Interconnection Customer; and (c) a Permissible Technological Advancement for the Large Generating Facility after the submission of the Interconnection Request. Section 4.4.6 specifies a separate technological change procedure including the requisite information and process that will be followed to assess whether the Interconnection Customer's proposed technological advancement

- under Section 4.4.2(c) is a Material Modification. Section 1 contains a definition of Permissible Technological Advancement.
- 4.4.3 Prior to making any modification other than those specifically permitted by Sections 4.4.1, 4.4.2, and 4.4.6, Interconnection Customer may first request that Transmission Provider evaluate whether such modification is a Material Modification. In response to Interconnection Customer's request, Transmission Provider shall evaluate the proposed modifications prior to making them and inform Interconnection Customer in writing of whether the modifications would constitute a Material Modification. Any change to the Point of Interconnection, except those deemed acceptable under Sections 4.4.1, 7.2 or so allowed elsewhere, shall constitute a Material Modification. Interconnection Customer may then withdraw the proposed modification or proceed with a new Interconnection Request for such modification.
- **4.4.4** Upon receipt of Interconnection Customer's request for modification permitted under this Section 4.4, Transmission Provider shall commence and perform any necessary additional studies as soon as practicable, but in no event shall Transmission Provider commence such studies later than thirty (30) Calendar Days after receiving notice of the modification of Interconnection Customer's request. Any additional studies resulting from such modification shall be done at Interconnection Customer's cost.
- 4.4.5 Extensions of less than three (3) cumulative years in the Commercial Operation Date of the Large Generating Facility to which the Interconnection Request relates are not material and should be handled through construction sequencing; provided, however, that extensions may necessitate a determination of whether additional studies are required pursuant to Applicable Laws and Regulations and Applicable Reliability Standards. The initial requested Commercial Operation Date used for this calculation is determined from the date proposed in the initial Interconnection Request. Such cumulative extensions are inclusive of extensions requested after execution by Interconnection Customer of the LGIA.
- 4.4.6 Technological Change Procedure.
 - **4.4.6.1 Interconnection Customer Technological Advancement Request.**

- (a) At any time after the submission of an Interconnection Request, but before the execution of an Interconnection Facility Study Agreement by Interconnection Customer, an Interconnection Customer may submit a written request pursuant to this Section to include additional or substituted technological components for its Large Generating Facility that differ from the description of the Large Generating Facility in its Interconnection Request.
- (b) To timely perfect its Technological Advancement Request, Interconnection Customer shall submit the following to Transmission Provider: (i) completed Technological Advancement Request form submitted on the request template provided by Transmission Provider on its OASIS site; (ii) a \$25,000 deposit; (iii) an updated version of the Interconnection Request for a Large Generating Facility, found at Appendix 1 of this LGIP, that reflects the data associated with the change in technology that Interconnection Customer seeks to incorporate; (iv) to the extent applicable, updated modeling data.

Updated modeling data is applicable if the technology change results in a change in the time of use of the facility, i.e., from peak to off peak, or capacity factor of the facility; a change in the steady-state thermal and/or voltage limits of the facility; a change in harmonics; a possible increase in the short circuit capability of the facility; or a possible change in the transient or dynamic response of the facility.

Where a technology change does not fall into one of the categories above, but is a technology change that may reduce the Network Upgrades previously identified in the interconnection study process associated with the Interconnection Customer's Large Generating Facility, the Transmission Provider may require a Technological Advancement Study using updated modeling data for purposes of implementing the identification of Network Upgrades necessary to accommodate the Interconnection Request under this LGIP and for the implementation of the crediting provisions set forth in Section 11.4 of the LGIA. Among the changes that Transmission Provider may evaluate under this Section 4.4.6 include, but are not limited to, advancements that improve, rather than degrade the electrical

characteristics of the generating equipment (e.g., the ratings, impedances, efficiencies, capabilities, and performance of the equipment under steady state and dynamic conditions) without changing the real power (MW) output or reactive power (MVAR) output of the Large Generating Facility.

- (c) Interconnection Customer's Technological Advancement Request must demonstrate how the proposed technological advancement (i) results in equal to or better electrical performance, (ii) does not increase the Interconnection Customer's requested interconnection service, and (iii) does not cause any reliability concerns (i.e., material impacts to the transmission system, including impacts to short circuit capability limits, steady-state thermal and voltage limits, or dynamic system stability and response).
- (d) An Interconnection Customer may have no more than one Technological Advancement Request pending at any one time. If the Technological Advancement Request is submitted during the time allocated under the LGIP for Interconnection Customer to execute and return a System Impact or Facilities Study Agreement to Transmission Provider, the deadline for execution and return of the System Impact or Facilities Study Agreement will be suspended while Transmission Provider analyzes the Technological Advancement Request in accordance with Section 4.4.6.
- (e) If Transmission Provider is performing an Interconnection System Impact Study, or other study for the Interconnection Request at the time that Interconnection Customer submits a Technological Advancement Request, Transmission Provider shall suspend work on any such pending studies until it has completed its analysis of the Technological Advancement Request and any Technological Advancement Study.
- (f) Interconnection Customer shall fill in, sign and submit, together with its Technological Advancement Request, a Technological Advancement Study Agreement. A form of Technological Advancement Study Agreement is posted on Transmission Provider's OASIS.

4.4.6.2 Initial Analysis of Technological Advancement Request.

(a) After the Interconnection Customer's Technological

Advancement Request is received pursuant to Section 4.4.6.1, the Transmission Provider will perform an initial analysis to determine whether the proposed technological advancement is a Permissible Technological Advancement without the need of additional study.

- (b) If the Transmission Provider determines on the basis of its initial analysis that Interconnection Customer has demonstrated that the proposed technological advancement is a Permissible Technological Advancement without the need for additional study, the Transmission Provider will incorporate the technological advancement into Interconnection Customer's Interconnection Request.
- (c) If the Transmission Provider determines on the basis of its initial analysis that Interconnection Customer has not demonstrated that the proposed technological advancement is a Permissible Technological Advancement, then the Technological Advancement Request will be treated as a request for modification of the Interconnection Request under Section 4.4.3.
- (d) If the Transmission Provider determines on the basis of its initial analysis that further study is required to conclude whether the Technological Advancement Request is a Permissible Technological Advancement, Transmission Provider will require that a Technological Advancement Study be performed at the sole expense of the Interconnection Customer consistent with Sections 4.4.6.3, 4.4.6.4, 4.4.6.5.
- (e) Any difference between the deposit provided under Section 4.4.6.1(b) and the actual cost of providing the initial analysis of the request shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate, along with an invoice describing any charges.

4.4.6.3 Technological Advancement Study Notification:

If after its initial analysis of a Technological Advancement Request, Transmission Provider determines that a Technological Advancement Study is necessary to determine whether the requested technological advancement constitutes a Permissible Technological Advancement, Transmission Provider shall notify Interconnection Customer in writing that such a study is necessary, and shall perform such study pursuant to Section 4.4.6.4.

4.4.6.4 Technological Advancement Study Procedures:

- (a) The Technological Advancement Study shall seek to determine (i) whether the proposed technological advancement is a Permissible Technological Advancement, by focusing on whether the proposed technological advancement will result in equal or better electrical performance than the Large Generating Facility described in the Interconnection Request, and whether the proposed technological advancement will cause any reliability concerns (i.e., material impacts to the transmission system, including impacts to short circuit capability limits, steady-state thermal and voltage limits, or dynamic system stability and response); and (ii) if the proposed technological advancement is determined not to be a Permissible Technological Advancement, whether the proposed technological advancement is a Material Modification. The Technological Advancement Study may include steady-state, reactive power, short circuit/fault duty, stability analyses, and any other appropriate studies that Transmission Provider deems necessary to determine whether the technological advancement results in electrical performance that is equal to or better than the electrical performance expected prior to the technology change, and whether such technological advancement causes any reliability concerns.
- (b) Interconnection Customer shall cooperate with Transmission Provider to provide any additional information that Transmission Provider may require to complete the Technological Advancement Study. If the Transmission Provider determines that it requires additional technical information to complete the Technological Advancement Study, Transmission Provider shall notify the Interconnection Customer of the additional technical information required, and Interconnection Customer shall work in good faith with Transmission Provider to promptly provide such information.
- (c) Upon completion of the Technological Advancement Study, Transmission Provider shall provide Interconnection Customer notice of its study conclusions. Upon request,

Transmission Provider shall also provide Interconnection Customer supporting documentation, subject to confidentiality arrangements consistent with Section 13.1.

- (d) If the Technological Advancement Study determines that the proposed technological advancement is either (i) a Permissible Technological Advancement, or (ii) is not a Permissible Technological Advancement but does not constitute a Material Modification, then the Interconnection Request shall be amended to reflect the technological advancement.
- (e) If the Technological Advancement Study determines that the proposed technological advancement is not a Permissible Technological Advancement and also constitutes a Material Modification, Transmission Provider shall provide an explanation for this conclusion. Interconnection Customer may then withdraw the proposed modification or proceed with a new Interconnection Request for such modification.
- (f) Any difference between the deposit provided under Section 4.4.6.1(b) and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate, along with an invoice describing any charges.

4.4.6.5 Time for Completing Initial Analysis and Technological Advancement Study.

Within thirty (30) Calendar Days of receipt of the Interconnection Customer's Technological Advancement Request submitted pursuant to Section 4.4.6.1, Transmission Provider shall complete all analysis and study obligations under this Section 4.4.6 and determine whether the Technological Advancement Request is a Permissible Technological Advancement or Material Modification, subject to Interconnection Customer's performance of its obligations under Section 4.4.6.1(f) and Section 4.4.6.4(b).

4.4.6.6 Treatment of Other Interconnection Studies During and After Technological Advancement Study.

Upon completion of the Transmission Provider's initial analysis of a Technological Advancement Request and any Technological Advancement Study, Transmission Provider and Interconnection Customer shall amend any existing Interconnection System Impact Study Agreement, or other Interconnection Study Agreements as necessary to incorporate elements of the requested technological advancement or the results of the Technological Advancement Study. Transmission Provider may require additional time or information to complete or re-run studies that were suspended during the pendency of the Technological Advancement Request. A single study may be suspended no more than once as a result of a Technological Advancement Request. If a subsequent Technological Advancement Request is received by Transmission Provider in such circumstances, Transmission Provider will process the subsequent Technological Advancement Request as soon as it completes the study that had been previously suspended.

Section 5. Procedures for Interconnection Requests Submitted Prior to Effective Date of Standard Large Generator Interconnection Procedures

5.1 Queue Position for Pending Requests Transition to the First-Ready, First-Served Process

5.1.1 Any Interconnection Customer assigned a Queue Position prior to the effective date of this LGIP shall retain that Queue Position_subject to the requirements set forth in Sections 5.1.1.1 and 5.1.1.2. An Interconnection Customer that fails to meet these requirements shall have its Interconnection Request deemed withdrawn pursuant to Section 3.7, without prejudice to the Interconnection Customer's ability to request an Informational Interconnection Study for which there is no Queue Position. Any unused deposit amounts of withdrawn Interconnection Requests shall be returned pursuant to Section 3.7. If an Interconnection Customer elects to continue with an Informational Interconnection Study or alternatively with a Transitional Interconnection Facilities Study or a Transitional Cluster Study as described below, Transmission Provider shall retain the current study deposits, and Interconnection Customer shall be responsible for the entire cost of all studies pursuant to Sections 6 and 4.

5.1.1.1 Transitional Projects.

An Interconnection Customer that has a) a final System Impact Study Report that identifies facilities required to feasibly interconnect and b) an Interconnection Facilities Study Agreement that was executed prior to the effective date of this Section 5 may opt to continue with the Interconnection Facilities Study process if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Interconnection Facilities Study Agreement in the form of Appendix 4.1 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All of the following are required:

- a) A deposit on the Transmission Provider's Interconnection
 Facilities and Network Upgrades identified in the System Impact
 Study Report. The deposit shall be equal to one hundred percent
 (100%) of the costs identified for Transmission Provider's
 Interconnection Facilities and Network Upgrades in the System
 Impact Study Report and will be reconciled to actual costs after the
 associated facilities are in service. If the Interconnection Customer
 does not withdraw, the deposit shall be reconciled with and applied
 towards future construction costs described in the LGIA. If the
 Interconnection Customer withdraws or otherwise does not reach
 Commercial Operation, the deposit is fully refundable once the final
 invoice for study costs and Withdrawal Penalty is settled. The
 deposit shall be in the form of an irrevocable letter of credit upon
 which the Transmission Provider may draw or cash.
- b) Exclusive Site Control for the entire Generating Facility and any Interconnection Customer's Interconnection Facilities.
- c) Interconnection Customer shall provide one following:
 - i. A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or
 - ii. Reasonable evidence that the Generating Facility is included in an approved Resource Plan or Resource Solicitation Process, or
 - iii. An executed Provisional Large Generator
 Interconnection Agreement filed with FERC. Such an

agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the publication of the final Interconnection Facilities Study Report or the Interconnection Request shall be deemed withdrawn pursuant to Section 3.7 unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of facilities if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's total study cost is imposed.

5.1.1.1 If an Interconnection Study Agreement has not been executed as of the effective date of this LGIP, then such Interconnection Study, and any subsequent Interconnection Studies, shall be processed in accordance with this LGIP.

5.1.1.2 <u>Combined System Impact and Interconnection</u> Facilities Transitional Cluster Study.

An Interconnection Customer with an assigned Queue Position prior to the effective date of this Section 5 may opt to enter the combined system impact and interconnection facilities transitional cluster study ("Transitional Cluster Study") if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Cluster Study Agreement in the form of Appendix 4.2 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All Interconnection Requests that enter the Transitional Cluster Study shall be considered to have an equal Queue Position, and identified upgrade costs shall be allocated according to Section 4.2.2 of the LGIP. The Transitional Cluster Study costs shall be allocated

according to the method described in Section 4.2. Interconnection Customer may make a one-time extension to its requested Commercial Operation date upon entry into the Transitional Cluster Study and such an extension shall not be past May 1, 2027. All of the following are required: a) Choice of requesting either ERIS or NRIS b) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades expected to be identified in the Transitional Cluster Study. The deposit shall be equal to five million dollars (\$5,000,000) and be in the form of an irrevocable letter of credit upon which the Transmission Provider may draw or cash. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied towards future construction costs described in the LGIA. Any amounts in excess of the actual construction costs shall be returned to the customer. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, the deposit is fully refundable once the final invoice for study costs and Withdrawal Penalty is settled. c) Exclusive Site Control for the entire Generating Facility. d) Interconnection Customer shall provide one of the following: A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or Reasonable evidence that the Generating Facility is included in an approved

Resource Plan or Resource Solicitation

Process, or

Generator Interconnection Agreement
filed with FERC that is not in suspension
with 1) a commitment to construct the
facility, 2) a Commercial Operation Date
no later than May 1, 2027 and 3) a
security deposit in addition to the five
million dollars identified in 5.1.1.2.b
where the total security deposit
represents a reasonable estimation of the
potential costs that could be ultimately
allocated to the project in the transitional
cluster study.

If an agreement termed an Interconnection Feasibility—Study Agreement has been executed as of the effective—date of this LGIP, and the study has been initiated, then such study shall be completed under the terms of the Agreement.

After the Transitional Cluster Study report is published, the remaining process shall proceed according to Section 11 of this LGIP. All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the tender of the draft LGIA or the Interconnection Request is deemed withdrawn unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of Transmission Provider's Interconnection Facilities or Network Upgrades if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit (see Section 3.4) is imposed.

Filed Date: 05/31/2023

and in effect or made effective unexecuted prior to the effective date of this Section 5 (and not in suspension) that has not built its Large Generating Facility shall have 60 Calendar Days from the effective date of this Section 5.1.1.3 to submit to Transmission Provider a written commitment to (a) start construction of the Generating Facility no later than two (2) years of the effective date of this Section 5.1.1.3 and reach Commercial Operation no later than three (3) years of the effective date of this Section 5.1.1.3 and (b) pay a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit if it withdraws, terminates its LGIA, or otherwise does not reach Commercial Operation. If it does not timely make this submission, the LGIA is deemed terminated, and no Withdrawal Penalty shall be assessed. A deemed termination shall be without prejudice to the Interconnection Customer's ability to submit a new Interconnection Request when it is willing and able to satisfy the tariff criteria under the Definitive Interconnection Process.

For any LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 and currently in suspension, the Interconnection Customer shall have 60 Calendar Days from lifting its suspension to make the submission to the Transmission Provider addressed above. If an agreement termed an Interconnection Feasibility Study Agreement has been executed as of the effective date of this LGIP, but the study has not been initiated, then any study shall be performed under the terms of this LGIP.

5.1.1.4 If an Interconnection Study Agreement has been executed prior to the effective date of this LGIP, such Interconnection Study shall be completed in accordance with the terms of such agreement. With respect to any remaining studies for which an Interconnection Customer has not signed an Interconnection Study Agreement prior to the effective date of the LGIP, Transmission Provider must go forward with the completion of the necessary

Interconnection Studies (for which it does not have a signed Interconnection Studies Agreement) in accordance with this LGIP.

5.1.1.5 If an LGIA has been submitted to FERC for approval before the effective date of the LGIP, then the LGIA would be grandfathered.

5.2 New Transmission Provider

If Transmission Provider transfers control of its Transmission System to a successor Transmission Provider during the period when an Interconnection Request is pending, the original Transmission Provider shall transfer to the successor Transmission Provider any amount of the deposit or payment with interest thereon that exceeds the cost that it incurred to evaluate the request for interconnection (less any non-refundable amounts). Any difference between such net amount and the deposit or payment required by this LGIP shall be paid by or refunded to the Interconnection Customer, as appropriate. The original Transmission Provider shall coordinate with the successor Transmission Provider to complete any Interconnection Study, as appropriate, that the original Transmission Provider has begun but has not completed. If Transmission Provider has tendered a draft LGIA to Interconnection Customer but Interconnection Customer has not either executed the LGIA or requested the filing of an unexecuted LGIA with FERC, unless otherwise provided, Interconnection Customer must complete negotiations with the successor Transmission Provider.

Section 6. [Reserved for Future Use] Informational Interconnection Study

6.1 Informational Interconnection Study Agreement.

At any time, a customer may request, and Transmission Provider (either itself or through a consultant) shall perform a reasonable number of Informational Interconnection Studies from May 1 to November 1 each year. An Informational Interconnection Study shall consist of analyses of the type performed in Phase I of the DISIS (power flow and voltage analyses). Interconnection Customer shall submit a separate Informational Interconnection Request for each site and may submit multiple Informational Interconnection Requests for a single site. Interconnection Customer must submit a deposit with each Informational Interconnection Request even when more than one request is submitted for a single site. An Informational Interconnection Request to evaluate one site at two different voltage levels shall be treated as two Informational Interconnection Requests. The request shall use the form in Appendix 5.1 of the LGIP and shall describe the assumptions that Interconnection Customer wishes

Filed Date: 05/31/2023

Transmission Provider to study within the scope described in Section 6.2 of the LGIP below. Within five (5) Business Days after receipt of a request for an Informational Interconnection Study, Transmission Provider shall provide to Interconnection Customer an Informational Interconnection Study Agreement in the form of Appendix 5.2.

The Informational Interconnection Study Agreement shall: (i) include the scope of work for the Informational Interconnection Study (ii) specify the technical data that Interconnection Customer must provide, (iii) specify the Informational Interconnection Study case and assumptions, and (iv) identify the Transmission Provider's estimate of the cost of the Informational Interconnection Study. To the extent known by Transmission Provider, such estimate shall include any costs expected to be incurred by any Affected System whose participation is necessary to complete the Informational Interconnection Study. Notwithstanding the above, Transmission Provider shall not be required as a result of an Informational Interconnection Study request to conduct any additional Interconnection Studies with respect to any other Interconnection Request.

Interconnection Customer shall execute the Informational Interconnection Study Agreement within ten (10) Business Days of receipt of an agreed upon scope of work and deliver the Informational Interconnection Study Agreement, the technical data, and a study deposit to Transmission Provider in the amount of \$100,000.00.

6.2 Scope of Informational Interconnection Study.

The intent of the Informational Interconnection Study is to aid Interconnection Customer in its business decisions related to interconnection of generation facilities prior to entering the Definitive Interconnection Process. The Informational Interconnection Study shall consist of analysis based on the assumptions and scope of work specified by Interconnection Customer in the Informational Interconnection Study Agreement. The Informational Interconnection Study shall identify the potential Transmission Provider's Interconnection Facilities and the Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the results and assumptions of the Informational Interconnection Study. The Informational Interconnection Study shall be performed solely for informational purposes. Transmission Provider shall use Reasonable Efforts to coordinate the study with any Affected Systems that may be affected by the types of Interconnection Services that are being studied. Transmission Provider shall utilize existing studies to the extent practicable in conducting the Informational Interconnection Study.

6.3 Informational Interconnection Study Procedures.

The executed Informational Interconnection Study Agreement, the deposit, and technical and other data called for therein must be provided to Transmission Provider within ten (10) Business Days of Interconnection Customer receipt of the Informational Interconnection Request. Transmission Provider shall use Reasonable Efforts to complete the Informational Interconnection Study within 70 Calendar Days from the date of the Informational Interconnection Study Agreement. If Transmission Provider is unable to complete the Informational Interconnection Study within such time period, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required. Any difference between the study payment and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation and work papers and databases or data developed in the preparation of the Informational Interconnection Study, subject to confidentiality arrangements consistent with Section 13.1.

Section 7. <u>Definitive</u> Interconnection System Impact Study

7.1 <u>Definitive</u> Interconnection System Impact Study Agreement

Unless otherwise agreed, pursuant to the Scoping Meeting provided for in Section 3.4.4, within thirty (30) Calendar Days acknowledgement of a valid Interconnection Request indicating that a Definitive Interconnection System Impact Study is to be performed, Transmission Provider shall provide to Interconnection Customer a DISIS Agreement in the form of Appendix 2 to this LGIP. The DISIS Agreement shall provide that Interconnection Customer shall compensate Transmission Provider for the actual cost of the DISIS. No later than Fifteen (15) Business Days after the close of the DISIS Request Window, Transmission Provider shall provide to Interconnection Customer a non-binding updated good faith estimate of the cost and timeframe for completing the Definitive Interconnection System Impact Study. Simultaneously with the acknowledgement of a valid-Interconnection Request, Transmission Provider shall provide to Interconnection Customer an Interconnection System Impact Study Agreement in the form of Appendix 2 of this LGIP. The Interconnection System Impact Study Agreement shall specify that Interconnection Customer is responsible for the actual cost of the Interconnection System-Impact Study. Within five (5) Business Days following the Scoping-Meeting, Interconnection Customer shall specify for inclusion in the attachment to the Interconnection System Impact Study Agreement the

Point(s) of Interconnection. Within five (5) Business Days following Transmission Provider's receipt of such designation, Transmission Provider shall tender to Interconnection Customer the Interconnection System Impact Study Agreement signed by Transmission Provider, which includes a good faith estimate of the cost and estimated timeframe for completing the Interconnection System Impact Study.

7.2 Execution of <u>Definitive</u> Interconnection System Impact Study
Agreement Interconnection Customer shall execute the

<u>DISIS Interconnection System Impact Study</u> Agreement and deliver the
executed <u>DISIS Interconnection System Impact Study</u> Agreement to
Transmission Provider no later than thirty (30) Calendar Days after its
receipt.

On or before the return of the executed Interconnection System Impact Study Agreement to Transmission Provider, Interconnection Customer shall provide the technical data called for in Appendix 1, Attachment A.

If Interconnection Customer does not provide all such technical data called for in Appendix 1, Attachment A when it delivers the Interconnection System Impact Study Agreement, Transmission Provider shall notify Interconnection Customer of the deficiency within five (5) Business Days of the receipt of the executed Interconnection System Impact Study Agreement and Interconnection Customer shall cure the deficiency within ten (10) Business Days of receipt of the notice, provided, however, such deficiency does not include failure to deliver the executed Interconnection System Impact Study Agreement.

If the Interconnection System Impact Study uncovers any unexpected result(s) not contemplated during the Scoping Meeting, a substitute Point of Interconnection identified by either Interconnection Customer or Transmission Provider, and acceptable to the other, such acceptance not to be unreasonably withheld, will be substituted for the designated Point of Interconnection specified above without loss of Queue Position, and restudies shall be completed pursuant to Section 7.6 as applicable. For the purpose of this Section 7.2, if Transmission Provider and Interconnection Customer cannot agree on the substituted Point of Interconnection, then Interconnection Customer may direct that one of the alternatives as specified pursuant to Section 3.4.4, shall be the substitute.

7.3 Scope of <u>Definitive</u> Interconnection System Impact Study

The Definitive Interconnection System Impact Study shall evaluate the impact of the proposed interconnection on the reliability of the

Transmission System. The Definitive Interconnection System Impact Study will consider the Base Case as well as all generating facilities (and with respect to (iii) below, any identified Network Upgrades associated with such higher queued requests) that, on the date the DISIS Request Window closes: (i) are existing and directly interconnected to the Transmission System; (ii) are existing and interconnected to Affected Systems and may have an impact on the Interconnection Request; (iii) have a pending higher queued Interconnection Request to interconnect to the Transmission System; and (iv) have no Queue Position but have executed an LGIA or requested that an unexecuted LGIA be filed with FERC. Generating facilities with pending higher or equal queued NRIS requests, or requests associated with Firm Transmission Service shall generally be modeled at full output while existing generation may be re-dispatched to accommodate new requests in the model. Higher queued ERIS requests or in-service ERIS generators without associated Firm Transmission Service may be dispatched at zero in some study models.

As discussed in more detail in Section 7.4 below, the Definitive Interconnection System Impact Study is a phased study where the first phase (Phase 1) consists of a power flow and voltage analysis that is followed by a phase (Phase 2) that consists of a short circuit analysis and a stability analysis, and may also include another appropriate study (for example, an electromagnetic transient analysis, if appropriate). Any DISIS re-studies (Phase 3) shall consist of a power flow/voltage analysis, a short circuit analysis, and/or a stability analysis as needed. The Definitive Interconnection System Impact Study report will state the assumptions upon which it is based; state the results of the analyses; and provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to correct any problems identified in those analyses and implement the interconnection. For purposes of determining necessary Interconnection Facilities and Network Upgrades, the Definitive Interconnection System Impact Study shall consider the level of Interconnection Service requested by the Interconnection Customer, unless otherwise required to study the full Generating Facility Capacity due to safety or reliability concerns. The Definitive Interconnection System Impact Study will provide a list of facilities that are required as a result of the Interconnection Request and a non-binding good faith estimate of cost responsibility and a non-binding good faith estimated time to construct. The Interconnection System Impact Study shall evaluate the impact of the proposed interconnection on the reliability of the Transmission System. The Interconnection System Impact Study will consider the Base Case aswell as all generating facilities (and with respect to (iii) below, any

identified Network Upgrades associated with such higher queued interconnection) that, on the date the Interconnection System Impact Study is commenced: (i) are directly interconnected to the Transmission System; (ii) are interconnected to Affected Systems and may have an impact on the Interconnection Request; (iii) have a pending higher queued Interconnection Request to interconnect to the Transmission System; and (iv) have no Queue Position but have executed an LGIA or requested that an unexecuted LGIA be filed with FERC.

The Interconnection System Impact Study will consist of a short circuit analysis, a stability analysis, and a power flow analysis. The Interconnection System Impact Study will state the assumptions uponwhich it is based; state the results of the analyses; and provide the requirements or potential impediments to providing the requestedinterconnection service, including a preliminary indication of the cost and length of time that would be necessary to correct any problems identified in those analyses and implement the interconnection. For purposes of determining necessary Interconnection Facilities and Network Upgrades, the System Impact Study shall consider the level of Interconnection Service requested by the Interconnection Customer, unless otherwise required tostudy the full Generating Facility Capacity due to safety or reliability concerns. The Interconnection System Impact Study will provide a list of facilities that are required as a result of the Interconnection Request and a non-binding good faith estimate of cost responsibility and a non-bindinggood faith estimated time to construct.

7.4 **Definitive** Interconnection System Impact Study Procedures

Transmission Provider shall coordinate the <u>Definitive</u> Interconnection System Impact Study with any Affected System that is affected by the Interconnection Request pursuant to Section 3.6 above. Transmission Provider shall utilize existing studies to the extent practicable when it performs the <u>studyDISIS</u>. <u>Interconnection Requests for DISIS may be submitted only within the DISIS Request Window and Transmission Provider shall initiate the Definitive Interconnection Study Process pursuant to Section 4.2.</u>

The At-A-Glance Reference Sheet attached provides an overview and timeline of the Definitive Interconnection Study Process, including the phases and milestones associated with the Definitive Interconnection System Impact Study.

a. The DISIS Cluster shall consist of all eligible Interconnection Requests that have executed a DISIS Agreement and have provided all required

information before the close of the DISIS Request Window. Transmission Provider shall use Reasonable Efforts to complete the first phase (Phase 1) consisting of a power flow and voltage analysis within ninety (90) Calendar Days. The Phase 1 Report shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades that are expected to be required as a result of the Interconnection Request(s) and a non-binding good-faith indicative level estimate of cost responsibility and a non-binding good-faith estimated time to construct. Transmission Provider shall meet with each cluster participant ("Phase 1 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 1 results on OASIS.

- b. Within thirty (30) Calendar Days of the Phase 1 Report, all Interconnection

 Customers are required to provide Readiness Milestone 2 ("M2") and
 continued evidence of Site Control as described in Section 7.7.

 Interconnection Customers that do not provide the Readiness Milestone (or
 provide security in lieu of the Readiness Milestone) or do not provide Site
 Control described in Section 7.7.7 by the required date shall be deemed
 withdrawn from the Queue pursuant to Section 3.7.
- Interconnection Customers whose M2 and Site Control are accepted by Transmission Provider shall continue in to the second phase ("Phase 2") of the Definitive Interconnection System Impact Study. Phase 2 consists of an updated power flow/voltage analysis (if necessary), stability analysis and short circuit analysis, and any other appropriate study (for example, an electromagnetic transient analysis, if appropriate) for the Interconnection Customers remaining in the DISIS Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 2 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that DISIS Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 2 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 2" Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 2 results on OASIS.
 - d. Within thirty (30) Calendar Days of the Phase 2 Report, each

 Interconnection Customer is required to provide Readiness Milestone 3

 ("M3") and additional evidence of Site Control described in Section 7.7.7.

Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

- i. If all Interconnection Customers in the Cluster provide M3 and no
 Interconnection Customers withdraw from the Queue at this stage,
 the Definitive Interconnection Study Process advances to the
 Interconnection Facilities Study (Section 8). Transmission Provider
 shall electronically notify Interconnection Customers in the Cluster
 that Phase 3 is not required and simultaneously provide the
 Interconnection Facilities Agreement in the form of Appendix 3.
- ii. If one or more Interconnection Customer withdraws from the
 Cluster, Transmission Provider shall determine if a full system
 impact re-study is necessary. If Transmission Provider determines a
 re-study is not necessary and Phase 3 is not required, Transmission
 Provider shall provide an updated Phase 2 Report within thirty (30)
 Calendar Days of such determination and the Definitive
 Interconnection Study Process advances to the Interconnection
 Facilities Study (Section 8). When the updated Phase 2 report is
 issued, Transmission Provider shall electronically notify
 Interconnection Customers in the Cluster that Phase 3 is not required
 and simultaneously provide the Interconnection Facilities Agreement
 in the form of Appendix 3.
- If one or more Interconnection Customers withdraws from the Cluster and Transmission Provider determines a full system impact re-study is necessary, Transmission Provider will continue with System Impact re-studies ("Phase 3") as described in Section 7.4.e below, until Transmission Provider determines that no further restudies are required. If a customer withdraws after Section 7.4.d.i or Section 7.4.d.ii or during the Interconnection Facilities Study and Transmission Provider determines system impact level studies are necessary, the Cluster shall be restudied under the terms of Phase 3. Transmission Provider shall electronically notify Interconnection Customers in the Cluster and post on OASIS that a re-study is required. Interconnection Customers that have elected NRIS may make a onetime change between Phase 2 and Phase 3 (before the restudy starts) to ERIS if they notify Transmission Provider of such change in election within five (5) Business Days of the Transmission Provider's notification the first re-study is required.

Filed Date: 05/31/2023

- Interconnection Customers whose M3 and additional evidence of Site Control is accepted by Transmission Provider shall continue with the third phase ("Phase 3") of the Definitive Interconnection System Impact Study. Phase 3 may consist of updated power flow/voltage analysis, stability analysis, and/or short circuit and other analyses if necessary for the Interconnection Customers remaining in the Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 3 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 3 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 3 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 3 results on OASIS. If additional re-studies are required before moving to Phase 4 below, within thirty (30) Calendar Days of the Phase 3 Report, all Interconnection Customers are required to provide an updated Readiness Milestone 3 ("M3"). Readiness Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7. by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7. Transmission Provider shall electronically notify Interconnection Customers in the Cluster when no further re-studies are required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.
- f. Within twenty (20) Calendar Days of the notice that no System Impact restudies are needed, each Interconnection Customer is required to provide Readiness Milestone 4 ("M4"), Site Control requirements described in Section 7.7.7, and an executed Interconnection Facilities Agreement in the form of Appendix 3 (completed and including all required data identified therein). Readiness Milestones for the Definitive Interconnection System Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

g. Twenty (20) Calendar Days after the notice that no further Re-Studies are needed, Transmission Provider shall proceed with the Interconnection Facilities Study phase ("Phase 4") of the Definitive Study Process, described in detail in Section 8 below. An additional study deposit is not required for Phase 4.

Transmission Provider shall use Reasonable Efforts to complete the Interconnection System Impact Study within ninety (90) Calendar Days after the receipt of the Interconnection System Impact Study Agreement or notification to proceed, study payment, and technical data. If Transmission-Provider uses Clustering, Transmission Provider shall use Reasonable Efforts to deliver a completed Interconnection System Impact Study within ninety (90) Calendar Days after the close of the Queue Cluster Window.

At the request of Interconnection Customer or at any time Transmission Provider determines that it will not meet the required indicated time frame for completing the Interconnection System Impact StudyDISIS, Transmission Provider shall notify Interconnection Customer(s) as to the schedule status of the Interconnection System Impact StudyDISIS Cluster. If Transmission Provider is unable to complete the Interconnection System Impact StudyDISIS within the time period, it shall notify Interconnection Customer(s) and provide an estimated completion date with an explanation of the reasons why additional time is required. Upon request, Transmission Provider shall provide Interconnection Customer all supporting documentation, workpapers and relevant pre-Interconnection Request and post-Interconnection Request power flow, short circuit and stability databases for the Interconnection System Impact StudyDISIS, subject to confidentiality arrangements consistent with Section 13.1.

7.5 Meeting with Transmission Provider

Within ten (10) Business Days of providing an <u>DISIS</u> Interconnection—System Impact Study report to Interconnection Customer and posting the report on OASIS, Transmission Provider and Interconnection Customer shall meet to discuss the <u>study</u> results of the Interconnection System Impact Study, unless otherwise mutually agreed upon by the Parties.

7.6 Re-Study

If Re-Study <u>Definitive</u> of the Interconnection System Impact Study <u>other</u> than the re-study described above in 7.4.e is required due to a higher <u>or equal priority</u> queued project dropping out of the qQueue, or a modification of a higher queued project subject to 4.4, or re-designation of the Point of Interconnection pursuant to Section 7.2 Transmission Provider shall notify Interconnection Customer(s) in writing. The Transmission Provider shall

use Reasonable Efforts to ensure such complete the Re-Study in take no longer than one hundred fifty sixty (60150) Calendar Days from the date of notice. Any cost of Re-Study shall be borne by Interconnection Customer(s) being re-studied first utilize the funds remaining of the non-refundable portion of the withdrawing applicant's Initial Deposit as per Section 3.3.1, then the funds remaining of the Initial Deposit(s) of all Interconnection Customer(s) being re-studied, and, if these funds are insufficient, the Interconnection Customer is responsible for the remaining actual cost incurred for the re-study.

7.7 Readiness Milestones and Site Control

Readiness Milestones are required throughout the Definitive
Interconnection Study Process to demonstrate readiness. A customer that
does not sufficiently demonstrate readiness by providing a Readiness
Milestones is subject to withdrawal as described in Section 3.7 which may
include additional penalties.

There are three Readiness Milestone options that demonstrate readiness through the study process (i.e., for Readiness Milestones 1 (M1) through Readiness Milestones 4 (M4).

7.7.1 Readiness Milestone 1 ("M1").

M1 is satisfied by any one of the three options below (also described in 3.4.1) at Interconnection Customer's option. M1 may also be satisfied by providing additional security described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness.

- a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.
- b) Reasonable evidence the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c) Provisional Large Generator Interconnection Agreement
 accepted for filing at FERC. Such an agreement shall not be
 suspended and shall include a commitment to construct the
 Generating Facility.

7.7.2 Readiness Milestone 2 ("M2").

M2 is satisfied by any one of the three options below at Interconnection Customer's option. M2 may also be satisfied by providing additional security as described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

- a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of
 (i) the constructed Generating Facility, (ii) the Generating
 Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.
- b) Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c) Provisional Large Generator Interconnection Agreement
 accepted for filing at FERC. Such an agreement shall not be
 suspended and shall include a commitment to construct the
 Generating Facility.

7.7.3 Readiness Milestone 3 ("M3").

M3 is satisfied by any one of the three options below at Interconnection Customer's option. M3 may also be satisfied by providing additional security described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

- a) Executed contract, binding upon the parties to the contract,
 for sale of (i) the constructed Generating Facility, (ii) the
 Generating Facility's energy, or (iii) the Generating Facility's
 ancillary services if the Generating Facility is an electric
 storage resource; where the term of sale is not less than five
 (5) years.
- b) Reasonable evidence that the project has been selected in an approved Resource Plan or Resource Solicitation Process; or
- c) An unsuspended Provisional Large Generator Interconnection

 Agreement accepted for filing by FERC with reasonable
 evidence that the Generating Facility and Interconnection
 Facilities have commenced design and engineering.

7.7.4 Readiness Milestone 4 ("M4").

M4 is satisfied by any one of the three options below at Interconnection Customer's option. M4 may also be satisfied by providing additional security as described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness.

- a) Executed contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services and capacity if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
- b) Reasonable evidence that the project has been selected in an approved Resource Plan or Resource Solicitation Process; or
- c) An unsuspended Provisional Large Generator Interconnection

 Agreement accepted for filing by FERC with reasonable
 evidence that the Generating Facility and Interconnection
 Facilities have commenced construction.

7.7.5 Security Requirements.

A table showing the security required in each milestone is provided in the At-A-Glance Reference Sheet attachment. The security amount is dependent on if the customer provided a Readiness Milestone and the study phase the customer is entering. All security described below shall be in the form of an irrevocable letter of credit upon which Transmission Provider may draw or cash. The security is refunded to the Interconnection Customer upon withdrawal, LGIA termination, or Commercial Operation after any final invoice is settled. If cash is provided as security, it shall be refunded plus interest, where the interest is calculated in accordance with the methodology set forth in FERC's regulations at 18 CFR § 35.19a(a)(2)(iii) from the date the security is received to the date that it is refunded. Security may be drawn upon if costs under this LGIP including the LGIA remain unpaid as per this LGIP and the attached LGIA. As part of a valid interconnection request all Interconnection Customers must provide security equal to the study deposit amount as described in Section 3.4. The security provided in Section 3.4.1 will be applied towards the amount of security required for M5. An Interconnection Customer may opt to provide security in lieu of providing Readiness Milestones 1 through 4, as described above in Sections 7.7.1, 7.7.2, 7.7.3, and 7.7.4. The security provided is

applied towards the security amount required for each successive milestone if the Interconnection Customer does not withdraw from the queue. For example, the security provided for M2 is applied to the amount of security required for M3.

In lieu of providing a demonstration of readiness for Milestones 1 through 4, the amount of security required is a multiple of the study deposit described in Section 3.4 and is in addition to the security required for all Interconnection Customers under Sections 3.4. The additional amount of security required for each milestone for Interconnection Customers that do not provide a demonstration of readiness is:

M1 = 1 times the study deposit amount

M2 = 2 times the study deposit amount

M3 = 4 times the study deposit amount

M4 = 6 times the study deposit amount

For clarity, the total (i.e., inclusive of the security required under Section 3.4) amount of security required for each milestone for Interconnection Customers that do not provide a demonstration of readiness is:

M1 = 2 times the study deposit amount

M2 = 3 times the study deposit amount

M3 = 5 times the study deposit amount

M4 = 7 times the study deposit amount

All Interconnection Customers are required to provide security in order to satisfy Readiness Milestone 5 (M5) when the LGIA is executed as described in Section 11.3. The amount of security required for M5 is equal to nine (9) times the Interconnection Customer's share of the Definitive Interconnection Study Process study costs. If this amount is not known, the study deposit amount shall be used as an estimate of study cost until such amounts are known. If initially estimated, M5 shall be updated when the final invoice for actual study costs is issued. As this M5 amount is the

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

total security required to satisfy Readiness Milestone 5, any security provided pursuant to Sections 3.4 and 7.7 shall be applied towards the Readiness Milestone 5 amount when the LGIA is executed. The Interconnection Customer shall only be responsible to provide the incremental amount of security to the Transmission Provider and any excess security provided shall be refunded to the Interconnection Customer. Transmission Provider shall refund all security provided under this section to the Interconnection Customer upon achieving Commercial Operation.

7.7.6 [Reserved for Future Use]

7.7.7 Site Control.

In addition to the above Readiness Milestones, Site Control is required to determine increased readiness through the Definitive Interconnection Study Process. Additional information on Site Control is posted on Transmission Provider's OASIS.

- a) Before entering Phase 1 (concurrent with M1) demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- b) Before entering Phase 2 (concurrent with M2): continued demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- c) Before entering Phase 3 (concurrent with M3): demonstration of 60% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- d) Before entering Phase 4 (concurrent with M4): demonstration of 75% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- e) Before executing an LGIA (concurrent with M5):
 demonstration of 90% Site Control and 50% Site Control of
 Interconnection Customer's Interconnection Facilities is
 required.

Section 8. Interconnection Facilities Study

8.1 Interconnection Facilities Study Agreement

Simultaneously with the noticedelivery of the Interconnection System- Impact Study to Interconnection Customer(s) that Phase 3 is complete or not required, Transmission Provider shall provide to Interconnection Customer an Interconnection Facilities Study Agreement in the form of Appendix 3 to this LGIP. Within five (5) Business Days following the DISIS results (Phase 2 or Phase 3) meeting, Transmission Provider shall provide to Interconnection Customer a non-binding good faith estimate of the cost and timeframe for completing the Interconnection Facilities Study. The Interconnection Facilities Study Agreement shall provide that Interconnection Customer shall compensate Transmission Provider for the actual cost of the Interconnection Facilities Study. Within three (3) Business Days following the Interconnection System Impact Study results meeting, Transmission Provider shall provide to Interconnection Customer a non-binding good faith estimate of the cost and timeframe for completing the Interconnection Facilities Study. Interconnection Customer shall execute the Interconnection Facilities Study Agreement and deliver the executed Interconnection Facilities Study Agreement to Transmission Provider within thirty-twenty (3020) Calendar Days after its receipt, together with the required technical data, Readiness Milestone 4 and the Site Control requirements described in Section 7.7.7. Interconnection Customers that do not provide the Readiness Milestone (or additional security in lieu of the Readiness Milestone) and provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

8.1.1 Transmission Provider shall invoice Interconnection Customer on a monthly basis for the work to be conducted on the Interconnection Facilities Study each month. Interconnection Customer shall pay invoiced amounts within thirty (30) Calendar Days of receipt of invoice. Transmission Provider shall continue to hold the amounts on deposit until settlement of the final invoice.

8.2 Scope of Interconnection Facilities Study

The Interconnection Facilities Study shall specify and provide a non-binding estimate of the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the Phase 1 and Phase 2 (and/or Phase 3) Reports (as appropriate) Interconnection—System Impact Study in accordance with Good Utility Practice to physically and electrically connect the Interconnection Facilityies to the Transmission System. The Interconnection Facilities Study shall also identify the electrical switching configuration of the connection equipment, including, without limitation: the transformer, switchgear, meters, and other station equipment; the nature and estimated cost of any Transmission

Provider's Interconnection Facilities and Network Upgrades necessary to accomplish the interconnection; and an estimate of the time required to complete the construction and installation of such facilities. The Interconnection Facilities Study will also identify any potential control equipment for requests for Interconnection Service that are lower than the Generating Facility Capacity.

8.3 Interconnection Facilities Study Procedures

Transmission Provider shall coordinate the Interconnection Facilities Study with any Affected System pursuant to Section 3.6 above.

Transmission Provider shall utilize existing studies to the extent practicable in performing the Interconnection Facilities Study. Transmission Provider shall use Reasonable Efforts to complete the study and issue a draft Interconnection Facilities Study report to Interconnection Customer within ninety (90) Calendar Days after acceptance of the Interconnection Facilities Agreement and Readiness Milestone 4receipt of an executed Interconnection Facilities Study Agreement.

At the request of Interconnection Customer-or at any time Transmission-Provider determines that it will not meet the required time frame for completing the Interconnection Facilities Study, Transmission Provider shall notify Interconnection Customer as to the schedule status of the Interconnection Facilities Study. If Transmission Provider is unable to complete the Interconnection Facilities Study and issue a draft Interconnection Facilities Study report within the time identified required, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required.

Interconnection Customer may, within thirty (30) Calendar Days after receipt of the draft Interconnection Facilities Study report, provide written comments to Transmission Provider, which Transmission Provider shall considerinelude in completing the final Interconnection Facilities Study report. Transmission Provider shall issue the final Interconnection Facilities Study report within fifteen (15) Business Days of receiving Interconnection Customer's comments or promptly upon receiving Interconnection Customer's statement that it will not provide comments. Transmission Provider may reasonably extend such fifteen-day period upon notice to Interconnection Customer if Interconnection Customer's comments require Transmission Provider to perform additional analyses or make other significant modifications prior to the issuance of the final Interconnection Facilities Study rReport. Upon request, Transmission Provider shall provide Interconnection Customer supporting

documentation, workpapers, and databases or data developed in the preparation of the Interconnection Facilities Study, subject to confidentiality arrangements consistent with Section 13.1.

8.4 Meeting with Transmission Provider

Within ten (10) Business Days of providing a draft Interconnection Facilities Study report to Interconnection Customer, Transmission Provider and Interconnection Customer shall meet to discuss the results of the Interconnection Facilities Study, unless otherwise mutually agreed upon by the Parties.

8.5 Re-Study

If Re-Study of the Interconnection Facilities Study is required due to a higher or equal priority queued project dropping out of the qQueue or a modification of a higher queued project pursuant to Section 4.4, Transmission Provider shall so notify Interconnection Customer(s) in writing. The Transmission Provider shall use Reasonable Efforts to complete the Re-Study in no longer than sixty (60) Calendar Days from the date of notice. Re-studies that require re-running the system impact study analysis may take longer than sixty days. Any cost of Re-Study shall befirst utilize the funds remaining of the withdrawing applicant's Initial Deposit as per Section 3.3.1, then the funds remaining of the Initial Deposit(s) of all Interconnection Customer(s) being re-studied, and, if these funds are insufficient, the remaining cost is borne by the Interconnection Customer(s) being re-studied.

Section 9. Engineering & Procurement ('E&P') Agreement

Prior to executing an LGIA, an Interconnection Customer may, in order to advance the implementation of its interconnection, request and Transmission Provider shall offer the Interconnection Customer, an E&P Agreement that authorizes Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection. However, Transmission Provider shall not be obligated to offer an E&P Agreement if Interconnection Customer is in Dispute Resolution as a result of an allegation that Interconnection Customer has failed to meet any Readiness Mmilestones or comply with any prerequisites specified in other parts of the LGIP. The E&P Agreement is an optional procedure and it will not alter the Interconnection Customer's Queue Position or In-Service Date. The E&P Agreement shall provide for Interconnection Customer to pay the cost of all activities authorized by Interconnection Customer and to make advance payments or provide other satisfactory security for such costs.

Interconnection Customer shall pay the cost of such authorized activities and any cancellation costs for equipment that is already ordered for its interconnection, which cannot be mitigated as hereafter described, whether or not such items or equipment later become unnecessary. If Interconnection Customer withdraws its application for interconnection or either Party terminates the E&P Agreement, to the extent the equipment ordered can be canceled under reasonable terms, Interconnection Customer shall be obligated to pay the associated cancellation costs. To the extent that the equipment cannot be reasonably canceled, Transmission Provider may elect: (i) to take title to the equipment, in which event Transmission Provider shall refund Interconnection Customer any amounts paid by Interconnection Customer for such equipment and shall pay the cost of delivery of such equipment, or (ii) to transfer title to and deliver such equipment to Interconnection Customer, in which event Interconnection Customer shall pay any unpaid balance and cost of delivery of such equipment.

Section 10. Optional Interconnection Study [Reserved for Future Use]

10.1 Optional Interconnection Study Agreement

On or after the date when Interconnection Customer receives
Interconnection System Impact Study results, Interconnection Customer
may request, and Transmission Provider shall perform a reasonable number
of Optional Studies. The request shall describe the assumptions that
Interconnection Customer wishes Transmission Provider to study within the
scope described in Section 10.2. Within five (5) Business Days after
receipt of a request for an Optional Interconnection Study, Transmission
Provider shall provide to Interconnection Customer an Optional
Interconnection Study Agreement in the form of Appendix 5.

The Optional Interconnection Study Agreement shall: (i) specify the technical data that Interconnection Customer must provide for each phase of the Optional Interconnection Study, (ii) specify Interconnection Customer's assumptions as to which Interconnection Requests with earlier queue priority dates will be excluded from the Optional Interconnection Study case and assumptions as to the type of interconnection service for Interconnection Requests remaining in the Optional Interconnection Study case, and (iii) Transmission Provider's estimate of the cost of the Optional Interconnection Study. To the extent known by Transmission Provider, such estimate shall include any costs expected to be incurred by any Affected System whose participation is necessary to complete the Optional Interconnection Study. Notwithstanding the above, Transmission Provider shall not be required as a result of an Optional Interconnection Study

Filed Date: 05/31/2023

request to conduct any additional Interconnection Studies with respect to any other Interconnection Request.

Interconnection Customer shall execute the Optional Interconnection Study Agreement within ten (10) Business Days of receipt and deliver the Optional Interconnection Study Agreement, the technical data and a \$25,000 deposit to Transmission Provider.

10.2 Scope of Optional Interconnection Study

The Optional Interconnection Study will consist of a sensitivity analysis-based on the assumptions specified by Interconnection Customer in the Optional Interconnection Study Agreement. The Optional Interconnection Study will also give a good faith estimate of Transmission Provider's Interconnection Facilities and the Network Upgrades, and the estimated cost thereof, that may be required to provide Interconnection Service based upon the results of the Optional Interconnection Study. The Optional Interconnection Study shall be performed solely for informational purposes. Transmission Provider shall use Reasonable Efforts to coordinate the study with any Affected Systems that may be affected by the types of Interconnection Services that are being studied. Transmission Provider shall utilize existing studies to the extent practicable in conducting the Optional Interconnection Study.

10.3 Optional Interconnection Study Procedures

The executed Optional Interconnection Study Agreement, the prepayment, and technical and other data called for therein must be provided to Transmission Provider within ten (10) Business Days of Interconnection Customer receipt of the Optional Interconnection Study Agreement. Transmission Provider shall use Reasonable Efforts to complete the Optional Interconnection Study within a mutually agreed upon time periodspecified within the Optional Interconnection Study Agreement. If Transmission Provider is unable to complete the Optional Interconnection Study within such time period, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasonswhy additional time is required. Any difference between the study depositand the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation and workpapers and databases or data developed in the preparation of the Optional Interconnection Study, subject to confidentiality arrangements consistent with Section 13.1.

Section 11. Standard Large Generator Interconnection Agreement (LGIA)

11.1 Tender

Interconnection Customer shall tender comments on the draft Interconnection Facilities Study Report within thirty (30) Calendar Days of receipt of the report. Within thirty (30) Calendar Days after the comments are submitted or after the Interconnection Customer notifies Transmission Provider that it will provide no comments, Transmission Provider shall tender a draft LGIA, together with draft appendices. The draft LGIA shall be in the form of Transmission Provider's FERC-approved standard form LGIA, which is in Appendix 6. Interconnection Customer shall execute and return the completed draft appendices within thirty (30) Calendar Days so that Transmission Provider may prepare a final LGIA. Transmission Provider shall tender a final LGIA, together with final appendices, within thirty (30) Calendar Days after the expiration of the thirty (30) Calendar Day window in which the Interconnection Customer is to return completed draft appendices.

11.2 Negotiation

Notwithstanding Section 11.1, at the request of Interconnection Customer Transmission Provider shall begin negotiations with Interconnection Customer concerning the appendices to the LGIA at any time after Interconnection Customer executes the Interconnection Facilities Study Agreement. Transmission Provider and Interconnection Customer shall negotiate concerning any disputed provisions of the appendices to the draft LGIA for not more than sixty (60) Calendar Days after tender of the final Interconnection Facilities Study Report. If Interconnection Customer determines that negotiations are at an impasse, #Interconnection Customer may request termination of the negotiations at any time after tender of the draft LGIA pursuant to Section 11.1 and request submission of the unexecuted LGIA with FERC or initiate Dispute Resolution procedures pursuant to Section 13.5. If Interconnection Customer requests termination of the negotiations, but within sixty (60) Calendar Days thereafter fails to request either the filing of the unexecuted LGIA or initiate Dispute Resolution, it shall be deemed to have withdrawn its Interconnection Request. Unless otherwise agreed by the Parties, if Interconnection Customer has not executed the LGIA, requested filing of an unexecuted LGIA, or initiated Dispute Resolution procedures pursuant to Section 13.5 within sixty (60) Calendar Days of tender of draft LGIA, it shall be deemed to have withdrawn its Interconnection Request. Transmission Provider shall provide to Interconnection Customer a final LGIA within fifteen (15) Business Days after the completion of the negotiation process.

11.3 Execution and Filing

Within fifteen (15) Business Days after receipt of the final LGIA, Interconnection Customer shall provide Transmission Provider (A) reasonable evidence that continued Site Control as defined in Section 7.7.7 andor (B) post Readiness Milestone 5 ing of \$250,000, non-refundable additional security equal to nine (9) times that Interconnection Customer's share of the Definitive Interconnection Study Process study, which shall be applied toward future construction costs. If the actual study costs are not known aAt the same time, study costs shall be estimated as the study deposit described in Section 3.4, and the M5 amount shall be updated when the study costs are known. If the Interconnection Customer does not reach Commercial Operation, upon payment of any final invoice, including any Withdrawal Penalty, Readiness Milestone 5 shall be refunded to the Interconnection Customer, including any accumulated interest, if applicable. If the Interconnection Customer reaches Commercial Operation, Readiness Milestone 5 is refunded to the Interconnection Customer including any accumulated interest, if applicable. Within fifteen (15) Business Days after receipt of the final LGIA, Interconnection Customer also shall provide reasonable evidence that one or more of the following milestones in the development of the Large Generating Facility, at Interconnection Customer election, has been achieved: (i) the execution of a contract for the supply or transportation of fuel to the Large Generating Facility (applicable to Generating Facilities that require fuel transportation; not available for wind or solar resources); (ii) the execution of a contract for the supply of cooling water to the Large Generating Facility (applicable to Generating Facilities that require cooling water for the production process; not available for wind or solar resources); (iii) execution of a contract for the engineering for, procurement of major equipment for, or construction of, the Large Generating Facility; (iv) execution of a contract for the sale of electric energy or capacity from the Large Generating Facility; or (v) application for an air, water, or land use permit.

At the same time that Interconnection Customer makes its submission of the items identified above (due within fifteen (15) Business Days after receipt of the final LGIA), Interconnection Customer shall either: (i) execute two originals of the tendered LGIA and return them to Transmission Provider; or (ii) request in writing that Transmission Provider file with FERC an LGIA in unexecuted form. As soon as practicable, but not later than ten (10) Business Days after receiving either the two executed originals of the tendered LGIA (if it does not conform with a FERC-approved standard form of interconnection agreement) or the request to file an unexecuted LGIA, Transmission Provider shall file the LGIA with FERC, together with its explanation of any matters as to which Interconnection Customer and Transmission Provider disagree and support

for the costs that Transmission Provider proposes to charge to Interconnection Customer under the LGIA. An unexecuted LGIA should contain terms and conditions deemed appropriate by Transmission Provider for the Interconnection Request. If the Parties agree to proceed with design, procurement, and construction of facilities and upgrades under the agreed-upon terms of the unexecuted LGIA, they may proceed pending FERC action.

11.4 Commencement of Interconnection Activities

If Interconnection Customer executes the final LGIA, Transmission Provider and Interconnection Customer shall perform their respective obligations in accordance with the terms of the LGIA, subject to modification by FERC. Upon submission of an unexecuted LGIA, Interconnection Customer and Transmission Provider shall promptly comply with the unexecuted LGIA, subject to modification by FERC.

Section 12. Construction of Transmission Provider's Interconnection Facilities and Network Upgrades

12.1 Schedule

Transmission Provider and Interconnection Customer shall negotiate in good faith concerning a schedule for the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades.

12.2 Construction Sequencing

12.2.1 General

In general, the In-Service Date of an Interconnection Customers seeking interconnection to the Transmission System will determine the sequence of construction of Network Upgrades.

12.2.2 Advance Construction of Network Upgrades that are an Obligation of an Entity other than Interconnection Customer

An Interconnection Customer with an LGIA, in order to maintain its In-Service Date, may request that Transmission Provider advance to the extent necessary the completion of Network Upgrades that: (i) were assumed in the Interconnection Studies for such Interconnection Customer, (ii) are necessary to support such In-Service Date, and (iii) would otherwise not be completed, pursuant to a contractual obligation of an entity other than Interconnection Customer

that is seeking interconnection to the Transmission System, in time to support such In-Service Date. Upon such request, Transmission Provider will use Reasonable Efforts to advance the construction of such Network Upgrades to accommodate such request; provided that Interconnection Customer commits to pay Transmission Provider: (i) any associated expediting costs and (ii) the cost of such Network Upgrades.

Transmission Provider will refund to Interconnection Customer both the expediting costs and the cost of Network Upgrades, in accordance with Article 11.4 of the LGIA. Consequently, the entity with a contractual obligation to construct such Network Upgrades shall be obligated to pay only that portion of the costs of the Network Upgrades that Transmission Provider has not refunded to Interconnection Customer. Payment by that entity shall be due on the date that it would have been due had there been no request for advance construction. Transmission Provider shall forward to Interconnection Customer the amount paid by the entity with a contractual obligation to construct the Network Upgrades as payment in full for the outstanding balance owed to Interconnection Customer. Transmission Provider then shall refund to that entity the amount that it paid for the Network Upgrades, in accordance with Article 11.4 of the LGIA.

12.2.3 Advancing Construction of Network Upgrades that are Part of an Expansion Plan of the Transmission Provider

An Interconnection Customer with an LGIA, in order to maintain its In-Service Date, may request that Transmission Provider advance to the extent necessary the completion of Network Upgrades that: (i) are necessary to support such In-Service Date and (ii) would otherwise not be completed, pursuant to an expansion plan of Transmission Provider, in time to support such In-Service Date. Upon such request, Transmission Provider will use Reasonable Efforts to advance the construction of such Network Upgrades to accommodate such request; provided that Interconnection Customer commits to pay Transmission Provider any associated expediting costs. Interconnection Customer shall be entitled to transmission credits, if any, for any expediting costs paid.

12.2.4 — Amended <u>Definitive</u> Interconnection System Impact Study An <u>Definitive</u> Interconnection System Impact Study <u>maywill</u>

be amended to determine the facilities necessary to support the requested In-Service Date. This amended study will include those transmission and Large Generating Facilities that are expected to be in service on or before the requested In-Service Date.

Section 13. Miscellaneous

13.1 Confidentiality

Confidential Information shall include, without limitation, all information relating to a Party's technology, research and development, business affairs, and pricing, and any information supplied by either of the Parties to the other prior to the execution of an LGIA.

Information is Confidential Information only if it is clearly designated or marked in writing as confidential on the face of the document, or, if the information is conveyed orally or by inspection, if the Party providing the information orally informs the Party receiving the information that the information is confidential.

If requested by either Party, the other Party shall provide in writing, the basis for asserting that the information referred to in this Section warrants confidential treatment, and the requesting Party may disclose such writing to the appropriate Governmental Authority. Each Party shall be responsible for the costs associated with affording confidential treatment to its information.

The release of Confidential Information shall be subject to Applicable Laws and Regulations and Applicable Reliability Standards.

13.1.1 Scope

Confidential Information shall not include information that the receiving Party can demonstrate: (1) is generally available to the public other than as a result of a disclosure by the receiving Party; (2) was in the lawful possession of the receiving Party on a non-confidential basis before receiving it from the disclosing Party; (3) was supplied to the receiving Party without restriction by a third party, who, to the knowledge of the receiving Party after due inquiry, was under no obligation to the disclosing Party to keep such information confidential; (4) was independently developed by the receiving Party without reference to Confidential Information of the disclosing Party; (5) is, or becomes, publicly known,

through no wrongful act or omission of the receiving Party or Breach of the LGIA; or (6) is required, in accordance with Section 13.1.6, Order of Disclosure, to be disclosed by any Governmental Authority or is otherwise required to be disclosed by law or subpoena, or is necessary in any legal proceeding establishing rights and obligations under the LGIA. Information designated as Confidential Information will no longer be deemed confidential if the Party that designated the information as confidential notifies the other Party that it no longer is confidential.

13.1.2 Release of Confidential Information

Neither Party shall release or disclose Confidential Information to any other person, except to its Affiliates (limited by the Standards of Conduct requirements), employees, consultants, or to parties who may be or considering providing financing to or equity participation with Interconnection Customer, or to potential purchasers or assignees of Interconnection Customer, on a need-to-know basis in connection with these procedures, unless such person has first been advised of the confidentiality provisions of this Section 13.1 and has agreed to comply with such provisions. Notwithstanding the foregoing, a Party providing Confidential Information to any person shall remain primarily responsible for any release of Confidential Information in contravention of this Section 13.1.

13.1.3 Rights

Each Party retains all rights, title, and interest in the Confidential Information that each Party discloses to the other Party. The disclosure by each Party to the other Party of Confidential Information shall not be deemed a waiver by either Party or any other person or entity of the right to protect the Confidential Information from public disclosure.

13.1.4 No Warranties

By providing Confidential Information, neither Party makes any warranties or representations as to its accuracy or completeness. In addition, by supplying Confidential Information, neither Party obligates itself to provide any particular information or Confidential Information to the other Party nor to enter into any further agreements or proceed with any other relationship or joint venture.

13.1.5 Standard of Care

Each Party shall use at least the same standard of care to protect Confidential Information it receives as it uses to protect its own Confidential Information from unauthorized disclosure, publication or dissemination. Each Party may use Confidential Information solely to fulfill its obligations to the other Party under these procedures or its regulatory requirements.

13.1.6 Order of Disclosure

If a court or a Government Authority or entity with the right, power, and apparent authority to do so requests or requires either Party, by subpoena, oral deposition, interrogatories, requests for production of documents, administrative order, or otherwise, to disclose Confidential Information, that Party shall provide the other Party with prompt notice of such request(s) or requirement(s) so that the other Party may seek an appropriate protective order or waive compliance with the terms of the LGIA. Notwithstanding the absence of a protective order or waiver, the Party may disclose such Confidential Information which, in the opinion of its counsel, the Party is legally compelled to disclose. Each Party will use Reasonable Efforts to obtain reliable assurance that confidential treatment will be accorded any Confidential Information so furnished.

13.1.7 Remedies

The Parties agree that monetary damages would be inadequate to compensate a Party for the other Party's Breach of its obligations under this Section 13.1. Each Party accordingly agrees that the other Party shall be entitled to equitable relief, by way of injunction or otherwise, if the first Party Breaches or threatens to Breach its obligations under this Section 13.1, which equitable relief shall be granted without bond or proof of damages, and the receiving Party shall not plead in defense that there would be an adequate remedy at law. Such remedy shall not be deemed an exclusive remedy for the Breach of this Section 13.1, but shall be in addition to all other remedies available at law or in equity. The Parties further acknowledge and agree that the covenants contained herein are necessary for the protection of legitimate business interests and are reasonable in scope. No

Party, however, shall be liable for indirect, incidental, or consequential or punitive damages of any nature or kind resulting from or arising in connection with this Section 13.1.

13.1.8 Disclosure to FERC, its Staff, or a State

Notwithstanding anything in this Section 13.1 to the contrary, and pursuant to 18 CFR section 1b.20, if FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to the LGIP, the Party shall provide the requested information to FERC or its staff, within the time provided for in the request for information. In providing the information to FERC or its staff, the Party must, consistent with 18 CFR section 388.112, request that the information be treated as confidential and non public by FERC and its staff and that the information be withheld from public disclosure. Parties are prohibited from notifying the other Party prior to the release of the Confidential Information to FERC or its staff. The Party shall notify the other Party to the LGIA when its is notified by FERC or its staff that a request to release Confidential Information has been received by FERC, at which time either of the Parties may respond before such information would be made public, pursuant to 18 CFR section 388.112. Requests from a state regulatory body conducting a confidential investigation shall be treated in a similar manner, consistent with applicable state rules and regulations.

Party claims is competitively sensitive, commercial or financial information ("Confidential Information") shall not be disclosed by the other Party to any person not employed or retained by the other Party, except to the extent disclosure is (i) required by law; (ii) reasonably deemed by the disclosing Party to be required to be disclosed in connection with a dispute between or among the Parties, or the defense of litigation or dispute; (iii) otherwise permitted by consent of the other Party, such consent not to be unreasonably withheld; or (iv) necessary to fulfill its obligations under this LGIP or as a transmission service provider or a Control Area operator including disclosing the Confidential Information to an RTO or ISO or to a subregional, regional or national reliability organization or planning group. The Party asserting

confidentiality shall notify the other Party in writing of the information it claims is confidential. Prior to any disclosures of the other Party's Confidential Information under this subparagraph, or if any third party or Governmental Authority makes any request or demand for any of the information described in this subparagraph, the disclosing Party agrees to promptly notify the other Party in writing and agrees to assert confidentiality and cooperate with the other Party in seeking to protect the Confidential Information from public disclosure by confidentiality agreement, protective order or other reasonable measures.

- This provision shall not apply to any information that was or is hereafter in the public domain (except as a result of a Breach of this provision).
- 13.1.11 Transmission Provider shall, at Interconnection Customer's election, destroy, in a confidential manner, or return the Confidential Information provided at the time of Confidential Information is no longer needed.

13.2 Delegation of Responsibility

Transmission Provider may use the services of subcontractors as it deems appropriate to perform its obligations under this LGIP. Transmission Provider shall remain primarily liable to Interconnection Customer for the performance of such subcontractors and compliance with its obligations of this LGIP. The subcontractor shall keep all information provided confidential and shall use such information solely for the performance of such obligation for which it was provided and no other purpose.

13.3 Obligation for Study Costs and Withdrawal Penalty

Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Interconnection Studies and the Withdrawal Penalty, as applicable. Any difference between the Initial Deposit submitted as per Section 3.3.1, less any part of that Deposit deemed non-refundable as per Section 3.3.1, and the actual cost of the Interconnection Study(ies) shall be paid by or refunded, except as otherwise provided herein, to Interconnection Customer. Any invoices for Interconnection Studies shall include a detailed and itemized accounting of the cost of each Interconnection Study as well as the Withdrawal Penalty, if applicable. Interconnection Customer shall pay any such undisputed costs within thirty (30)

Calendar Days of receipt of an invoice therefor. Transmission Provider shall not be obligated to perform or continue to perform any studies unless Interconnection Customer has paid all undisputed amounts in compliance herewith. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided under this LGIP to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

13.4 Third Parties Conducting Studies

If (i) at the time of the signing of an Interconnection Study Agreement there is disagreement as to the estimated time to complete an Interconnection Study, (ii) Interconnection Customer receives notice pursuant to Sections 7.4 or 8.3 that Transmission Provider will not complete an Interconnection Study within the applicable timeframe for such Interconnection Study, or (iii) Interconnection Customer receives neither the Interconnection Study nor a notice under Sections 7.4 or 8.3 within the applicable timeframe for such Interconnection Study, then Interconnection Customer may require Transmission Provider to utilize a third party consultant reasonably acceptable to Interconnection Customer and Transmission Provider to perform such Interconnection Study under the direction of Transmission Provider. At other times, Transmission Provider may also utilize a third party consultant to perform such Interconnection Study, either in response to a general request of Interconnection Customer, or on its own volition.

In all cases, use of a third party consultant shall be in accord with Article 26 of the LGIA (Subcontractors) and limited to situations where Transmission Provider determines that doing so will help maintain or accelerate the study process for Interconnection Customer's pending Interconnection Request and not interfere with Transmission Provider's progress on Interconnection Studies for other pending Interconnection Requests. In cases where Interconnection Customer requests use of a third party consultant to perform such Interconnection Study, Interconnection Customer and Transmission Provider shall negotiate all of the pertinent terms and conditions, including reimbursement arrangements and the estimated study completion date and study review deadline. Transmission Provider shall convey all workpapers, data bases, study results and

all other supporting documentation prepared to date with respect to the Interconnection Request as soon as practicable upon Interconnection Customer's request subject to the confidentiality provision in Section 13.1. In any case, such third party contract may be entered into with either Interconnection Customer or Transmission Provider at Transmission Provider's discretion. In the case of (iii) Interconnection Customer maintains its right to submit a claim to Dispute Resolution to recover the costs of such third party study. Such third party consultant shall be required to comply with this LGIP, Article 26 of the LGIA (Subcontractors), and the relevant Tariff procedures and protocols as would apply if Transmission Provider were to conduct the Interconnection Study and shall use the information provided to it solely for purposes of performing such services and for no other purposes. Transmission Provider shall cooperate with such third party consultant and Interconnection Customer to complete and issue the Interconnection Study in the shortest reasonable time.

13.5 Disputes

13.5.1 Submission

In the event either Party has a dispute, or asserts a claim, that arises out of or in connection with the LGIA, the LGIP, or their performance, such Party (the "disputing Party") shall provide the other Party with written notice of the dispute or claim ("Notice of Dispute"). Such dispute or claim shall be referred to a designated senior representative of each Party for resolution on an informal basis as promptly as practicable after receipt of the Notice of Dispute by the other Party. In the event the designated representatives are unable to resolve the claim or dispute through unassisted or assisted negotiations within thirty (30) Calendar Days of the other Party's receipt of the Notice of Dispute, such claim or dispute may, upon mutual agreement of the Parties, be submitted to arbitration and resolved in accordance with the arbitration procedures set forth below. In the event the Parties do not agree to submit such claim or dispute to arbitration, each Party may exercise whatever rights and remedies it may have in equity or at law consistent with the terms of this LGIA.

13.5.2 External Arbitration Procedures

Any arbitration initiated under these procedures shall be conducted before a single neutral arbitrator appointed by the Parties. If the Parties fail to agree upon a single arbitrator within ten (10) Calendar Days of the submission of the dispute to arbitration, each Party shall choose one arbitrator who shall sit on a three-member arbitration panel. The two arbitrators so chosen shall within twenty (20) Calendar Days select a third arbitrator to chair the arbitration panel. In either case, the arbitrators shall be knowledgeable in electric utility matters, including electric transmission and bulk power issues, and shall not have any current or past substantial business or financial relationships with any party to the arbitration (except prior arbitration). The arbitrator(s) shall provide each of the Parties an opportunity to be heard and, except as otherwise provided herein, shall conduct the arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("Arbitration Rules") and any applicable FERC regulations or RTO rules; provided, however, in the event of a conflict between the Arbitration Rules and the terms of this Section 13, the terms of this Section 13 shall prevail.

13.5.3 Arbitration Decisions

Unless otherwise agreed by the Parties, the arbitrator(s) shall render a decision within ninety (90) Calendar Days of appointment and shall notify the Parties in writing of such decision and the reasons therefor. The arbitrator(s) shall be authorized only to interpret and apply the provisions of the LGIA and LGIP and shall have no power to modify or change any provision of the LGIA and LGIP in any manner. The decision of the arbitrator(s) shall be final and binding upon the Parties, and judgment on the award may be entered in any court having jurisdiction. The decision of the arbitrator(s) may be appealed solely on the grounds that the conduct of the arbitrator(s), or the decision itself, violated the standards set forth in the Federal Arbitration Act or the Administrative Dispute Resolution Act. The final decision of the arbitrator must also be filed with FERC if it affects jurisdictional rates, terms and conditions of service, Interconnection Facilities, or Network Upgrades.

13.5.4 Costs

Each Party shall be responsible for its own costs incurred during the arbitration process and for the following costs, if applicable: (1) the cost of the arbitrator chosen by the Party to sit on the three member panel and one half of the cost of the third arbitrator chosen; or (2) one half the cost of the single arbitrator jointly chosen by the Parties.

13.5.5 Non-bBinding dDispute rResolution pProcedures.

If a Party has submitted a Notice of Dispute pursuant to section 13.5.1, and the Parties are unable to resolve the claim or dispute through unassisted or assisted negotiations within the thirty (30) Calendar Days provided in that section, and the Parties cannot reach mutual agreement to pursue the section 13.5 arbitration process, a Party may request that Transmission Provider engage in Non-binding Dispute Resolution pursuant to this section by providing written notice to Transmission Provider ("Request for Non-binding Dispute Resolution"). Conversely, either Party may file a Request for Non-binding Dispute Resolution pursuant to this section without first seeking mutual agreement to pursue the section 13.5 arbitration process. The process in section 13.5.5 shall serve as an alternative to, and not a replacement of, the section 13.5 arbitration process. Pursuant to this process, a Transmission Provider must within 30 Calendar Delays of receipt of the Request for Non-binding Dispute Resolution appoint a neutral decision-maker that is an independent subcontractor that shall not have any current or past substantial business or financial relationships with either Party. Unless otherwise agreed by the Parties, the decision-maker shall render a decision within sixty (60) Calendar Days of appointment and shall notify the Parties in writing of such decision and reasons therefore. This decisionmaker shall be authorized only to interpret and apply the provisions of the LGIP and LGIA and shall have no power to modify or change any provision of the LGIP and LGIA in any manner. The result reached in this process is not binding, but, unless otherwise agreed, the Parties may cite the record and decision in the non-binding dispute resolution process in future dispute resolution processes, including in a section 13.5 arbitration, or in a Federal Power Act section 206 complaint. Each Party shall be responsible for its own costs incurred during the process and the cost of the

decision-maker shall be divided equally among each Party to the dispute.

13.6 Local Furnishing Bonds

13.6.1 Transmission Providers That Own Facilities Financed by Local Furnishing Bonds

This provision is applicable only to a Transmission Provider that has financed facilities for the local furnishing of electric energy with tax-exempt bonds, as described in Section 142(f) of the Internal Revenue Code ("local furnishing bonds"). Notwithstanding any other provision of this LGIA and LGIP, Transmission Provider shall not be required to provide Interconnection Service to Interconnection Customer pursuant to this LGIA and LGIP if the provision of such Transmission Service would jeopardize the tax-exempt status of any local furnishing bond(s) used to finance Transmission Provider's facilities that would be used in providing such Interconnection Service.

13.6.2 Alternative Procedures for Requesting Interconnection Service

If Transmission Provider determines that the provision of Interconnection Service requested by Interconnection Customer would jeopardize the tax exempt status of any local furnishing bond(s) used to finance its facilities that would be used in providing such Interconnection Service, it shall advise the Interconnection Customer within thirty (30) Calendar Days of receipt of the Interconnection Request.

Interconnection Customer thereafter may renew its request for interconnection using the process specified in Section 5.2(ii) of the Transmission Provider's Tariff.

At-A-Glance Reference Sheet

1

N	<u> Tilestone</u>	Total Security Required	Total Security Required	Demonstration of	Site Control
		(Multiple of Section 3.4)	(Multiple of Section 3.4)	Site Control for All	of ICIF's
		Study Deposit) If	Study Deposit)	Fuel Types	
		Demonstration of	If Demonstration of		
		Readiness	Readiness IS NOT		
		IS Provided	<u>Provided</u>		
\mathbf{M}^{2}		<u>1x</u>	<u>2x</u>	<u>50%</u>	<u>0%</u>
\mathbf{M}_{2}^{\prime}	2	<u>1x</u>	<u>3x</u>	<u>50%</u>	<u>0%</u>
M.	3	<u>1x</u>	<u>5x</u>	<u>60%</u>	<u>0%</u>
M	1	<u>1x</u>	<u>7x</u>	<u>75%</u>	0%
M:	5	9 <u>x</u>	<u>9x</u>	<u>90%</u>	<u>50%</u>

Phase 1: Power Flow/Voltage: Within 90 calendar days

- Transmission Provider to perform Power Flow and Voltage Analyses.
- Transmission Provider to complete Phase 1 report within 90 calendar days after the close of the Queue Cluster Window.
- Transmission Provider to hold results meeting within 10 business days of posting of DISIS Phase 1 results.
- Interconnection Customer to demonstrate M2 Readiness within 30 calendar days following the Phase 1 report.

Phase 2: Stability/Short Circuit: Within 150 calendar days

- Transmission Provider to complete Phase 2 analyses within 150 calendar days after receipt of all Interconnection Customers' executed Interconnection System Impact Study Agreements.
- Transmission Provider to hold Phase 2 results meeting within 10 business days of posting of Phase 2 report.
- Interconnection Customer to demonstrate M3 (if Re-Study is necessary) within 30 calendar days following the Phase 2
 report, or demonstrate M4 Readiness (if no Re-Study is necessary) within 20 calendar days following Transmission
 Provider's notice that no Re-Study is needed.

**Phase 3: Iterative Cluster System Impact Re-Study: Within 150 calendar days

May not be necessary

- If a Re-Study is needed, Transmission Provider perform Phase 3 Re-Study within 150 calendar days after acceptance of all Interconnection Customers' executed Interconnection System Impact Study Agreements.
- Transmission Provider to hold results meeting within 10 business days of posting results.
- Interconnection Customer to demonstrate M4 Readiness within 20 calendar days following Transmission Provider's notice that no further System Impact Re-Study is needed.

Phase 4: Facilities Study: Within 90 calendar days

- Transmission Provider to complete Facilities Study, complete and submit draft Facilities Study Report to Interconnection
 Customer within 90 calendar days after acceptance of all Interconnection Facilities Study Agreements.
- Transmission Provider to hold results meeting within 10 business days of posting of Phase 4 report.
- Interconnection Customer has opportunity to provide written comments of Facilities Study Report to Transmission Provider within 30 calendar days of receipt of draft Facilities Study Report.
- Transmission Provider to issue Final Facilities Study Report to Customer within 15 business days of receiving
 Interconnection Customer's comments or promptly upon receiving Interconnection Customer's statement that it will not provide comments.

Phase 5: LGIA

- Transmission Provider to provide Interconnection Customer with draft LGIA within 30 calendar days of receipt of Interconnection Customer's Facilities Study comments.
- Interconnection Customer to return completed draft appendices within 30 calendar days of receipt of draft LGIA.
- Deadline for LGIA negotiations to be completed is within 60 calendar days after tender by Transmission Provider of the final Interconnection Facilities Study Report.
- Interconnection Customer to satisfy Readiness Milestone 5 within 15 business days of receiving final LGIA.
- Deadline for Interconnection Customer to execute the LGIA (or instruct that it be filed unexecuted) is 15 business days after receipt of the final LGIA.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Appendices

APPENDIX 1 to LGIP INTERCONNECTION REQUEST FOR A LARGE GENERATING FACILITY

- The undersigned Interconnection Customer submits this request to interconnect its
 Large Generating Facility with Transmission Provider's Transmission System pursuant to a Tariff.
- 2. This Interconnection Request is for (check one):
 - A proposed new Large Generating Facility.
 - An increase in the generating capacity or a Material Modification of an existing Generating Facility.
 - ____ A Generating Facility proposed for inclusion in a resource solicitation process.
- 3. The type of interconnection service requested (check one):
 - **Energy Resource Interconnection Service**
 - ____ Network Resource Interconnection Service
- 4. Interconnection Customer provides the following information:
 - a. Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility;
 - Maximum summer at degrees C and winter at degrees C megawatt
 electrical output of the proposed new Large Generating Facility or the amount of megawatt increase in the generating capacity of an existing Generating Facility;
 - c. General description of the equipment configuration;
 - d. Commercial Operation Date (Day, Month, and Year);
 - e. Name, address, telephone number, and e-mail address of Interconnection Customer's contact person;
 - f. Approximate location of the proposed Point of Interconnection (optional);
 - g. Interconnection Customer Data (set forth in Attachment A)
 - h. Primary frequency response operating range for electric storage resources.
 - i. Requested capacity (in MW) of Interconnection Service (if lower than the Generating Facility Capacity).
- 5. Interconnection Customer provides applicable study deposit amount as specified in the LGIP.

\$160,000 for requests of less than 75 MW \$250,000 for requests of 75 MW and greater

6. Interconnection Customer provides Readiness Milestone 1 (M1) as specified in the LGIP.

M1 is satisfied by any one of the three options below (also described in the LGIP) at Interconnection Customer's option. M1 may also be satisfied by providing additional

	security described in Section 7.7.5 in lieu of providing one of the three options to				
	demonstrate readiness.				
 i. Executed term sheet (or comparable evidence) related to a contract upon the parties to the contract, for sale of (i) the constructed Generating (ii) the Generating Facility's energy, or (iii) the Generating Facility's services if the Generating Facility is an electric storage resource; where of sale is not less than five (5) years; 					
	ii. Reasonable evidence the project has been selected Resource Plan or Resource Solicitation Process; or	d in a			
	iii. Provisional Large Generator Interconnection Agreement of the suspender shall include a commitment to construct the Generating Facility.				
<u>7.</u>	Interconnection Customer provides security equal to one times the study of described in Section 3 of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the form of an irrevocable letter of credit or called the security of the LGIP in the security of the				
8.	If requesting NRIS: Interconnection Customer provides the expected point of deliver within the Transmission Provider's Control Area.	very to			
9.	Interconnection Customer provides Evidence of Site Control as specified in the and Transmission Provider's business practices posted on OASIS, as applicable.	e LGIP			
<u>10.</u>	. This Interconnection Request shall be submitted to the representative indicated below:				
	[To be completed by Transmission Provider]				
<u>11.</u>	Representative of Interconnection Customer to contact:				
	To be completed by Interconnection Customer				
<u>12.</u>	This Interconnection Request is submitted by:				
<u>Name</u>	of Interconnection Customer:				
By (sig	gnature):				
<u>Name</u>	e (type or print):				
Title:_					
Date:					

Attachment A to Appendix 1 Interconnection Request

LARGE GENERATING FACILITY DATA

Power Factor	UNIT RATINGS	<u>Voltage</u>		
	Connection (e.g. Wye) Frequency, Hertz	Field Volts		
	`F	I IEIU VOIIS		
Primary frequency response	nse operating range	for electric storage resources.		
Minimum State of Charge: Maximum State of Charge:				
COMBINED TURE	BINE-GENERATOR-E	XCITER INERTIA DATA		
Inertia Constant, H = Moment-of-Inertia, WR ² =	kW sec/kVA lb. ft. ²			
REACTA	NCE DATA (PER UNI	T-RATED KVA)		
DIRECT AXIS QUADRATURE AXIS				
Synchronous – saturated	<u>X_{dv}</u>	<u>X_{av}</u>		
Synchronous – unsaturated	<u>X_{di}</u>	<u>X_{qi}</u>		
<u>Transient – saturated</u>	<u>X'_{dv}</u>	<u>X'_qv</u>		
<u>Transient – unsaturated</u>	<u>X'_{di}</u>	<u>X'_{qi}</u>		
<u>Subtransient – saturated</u>	<u>X"_{dv}</u>	<u>X"_{qv}</u>		
Subtransient – unsaturated	<u>X"_{di}</u>	<u>X"_{qi}</u>		
Negative Sequence – saturated	<u>X2_v</u>			
Negative Sequence – unsaturated	X2 _i			
Zero Sequence – saturated	<u>X0_v</u>			
Zero Sequence – unsaturated	<u>X0</u> ,			
Leakage Reactance	<u>XI_m</u>			

	FIELD TIME	CONSTANT	DATA (SEC)	
Open Circuit				
	Short Circuit Transient	<u>T'_{do} </u>		
	hort Circuit Transient	<u> </u>		
Short Circuit S		<u>" a1</u>	 Т"	
Open Circuit S		<u>T'_{d2}</u>	<u> </u>	
	Il Short Circuit Transient	<u>T"_{do}</u>		
<u>Ento to reduce</u>				
	<u>ARMATURE TI</u>	ME CONSTA	ANT DATA (SEC)	
Three Phase S Line to Line S Line to Neutra				
NOTE: If requ	ested information is not ap	plicable, indica	ate by marking "N/A."	
			CONFIGURATION ACILITY DATA	
	ARMATURE WINDIN	G RESISTAN	ICE DATA (PER UNIT)	
Positive Negative Zero	R ₁ R ₂ R ₀			
Field Current : Field Current : Three Phase : Field Winding	me Thermal Capacity I ₂ ² t = at Rated kVA, Armature Vo at Rated kVA and Armature Armature Winding Capacita Resistance = ohr ding Resistance (Per Phas	oltage and PF e Voltage, 0 P ance = ns °C	PF = amps microfarad C_	
		CURVES		
		apability, Ca	apacity Temperature Correction operating range for multiple curve	
	GENERATOR STEP-U	IP TRANSFO	RMER DATA RATINGS	
Capacity	Self-cooled/Maximum	<u>Nameplate</u>		
	/ k\/	/Δ		

Voltage Ratio (Generator Side/System side/Tertiary)				
		kV		
Winding Connections (Low V/High V/Tertiary V (Delta or Wye))				
Fixed Taps Available				
Present Tap Setting				
If more than one transformer stage is used to deliver the output from the proposed Generating Facility to the Transmission System, please provide the information above for each transformer or transformer type.				
	<u>IMPEDANCI</u>	<u>E</u>		
Positive Z ₁ (on self-cooled kVA rating)	%		X/R	
Zero Z ₀ (on self-cooled kVA rating)	%		<u>X/R</u>	
EXCITA	TION SYSTE	EM DATA		
Identify appropriate IEEE model block diagram of excitation system and power system stabilizer (PSS) for computer representation in power system stability simulations and the corresponding excitation system and PSS constants for use in the model.				
GOVER	RNOR SYSTE	M DATA		
Identify appropriate IEEE model block diagram of governor system for computer representation in power system stability simulations and the corresponding governor system constants for use in the model.				
WIND AND OTHER NON-SYNCHRONOUS GENERATORS				
Number of generators to be interco	nnected pur	suant to this	Interconnection	Request:
Elevation: Single	Phase	Three Phas	se	
Inverter manufacturer, model name, number, and version:				
List of adjustable setpoints for the protective equipment or software:				

Note: A completed General Electric Company Power Systems Load Flow (PSLF) data sheet or other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate to the proposed device, then they shall be provided with the Interconnection Request and discussed at Scoping Meeting.

Project Information: Site Control and Adequacy

Total acres required to construct the Generating Facility:
Total acres under site control for the Generating Facility at the time of application:
Is Site Control required for Interconnection Facilities, i.e. transmission gen-tie or substation, to interconnect the Generating Facility? Y N
If yes, how many miles of gen-tie right-of-way are required?
What is the total number of acres required to build the gen-tie?
How many miles of gen-tie right-of-way are under Site Control at the time of this application?
List any local, state, or federal government permits required to construct the Generating Facility
and any applicable Interconnection Facilities, i.e. transmission gen-tie:

INDUCTION	I GENERATORS
-----------	--------------

(*) Field Volts:	
(*) Field Amperes:	
(*) Motoring Power (kW):	
(*) Neutral Grounding Resistor (If App	olicable:
(*) I ₂ ² t or K (Heating Time Constant):	
(*) Rotor Resistance:	
(*) Stator Resistance:	
(*) Stator Reactance:	
(*) Rotor Reactance:	_
(*) Magnetizing Reactance:	
(*) Short Circuit Reactance:	
(*) Exciting Current:	
(*) Temperature Rise:	
(*) Frame Size:	
(*) Design Letter:	
(*) Reactive Power Required In Vars	(No Load):
(*) Reactive Power Required In Vars	(Full Load):
(*) Total Rotating Inertia, H:	Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Interconnection Request to determine if the information designated by (*) is required.

APPENDIX 2 to LGIP DEFINITIVE INTERCONNECTION SYSTEM IMPACT STUDY AGREEMENT

			<u>s made an</u>	d entered into this		, 20by
	<u>etweer</u>			, a		existing under the laws
<u>of</u>	the	State	of		("Interconnection	
			a			laws of the State of
						<u>mer and Transmission</u>
<u>Provi</u>	<u>der eac</u>	<u>ch may be</u>	referred to	as a "Party," or co	<u>llectively as the "Partie</u>	<u>es."</u>
				RECITAL	<u>-S</u>	
WHE	REAS.	Interconn	ection Cus	stomer is proposin	g to develop a Large	Generating Facility or
						consistent with the
					on Customer dated	
mer	301111000		ot oabiiite	od by intercention	orr odotomor datod	<u>j dire</u>
WHE	REAS.	Interconn	ection Cu	stomer desires to	interconnect the Lard	ge Generating Facility
			System; a			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
WHE	REAS.	Interconr	nection Cu	ustomer has requi	ested Transmission F	Provider to perform a
_						of interconnecting the
					em, and of any Affected	
WHE	REAS.	Interconi	nection C	ustomer commits	to provide certain F	Readiness Milestones
_					s as described in the L	
NOW	, THEF	REFORE.	in conside	ration of and subje	ct to the mutual cover	nants contained herein
		greed as				
1.0	Whe	n used in	this Agree	ement, with initial o	apitalization, the term	s specified shall have
	the n	neanings i	ndicated in	Transmission Pro	vider's FERC-approve	d LGIP.
2.0	Interd	connection	<u>ı Custome</u>	r elects and Transr	nission Provider shall	cause to be performed
	a Do	efinitive I	nterconnec	ction System Imp	act Study consisten	t with this LGIP in
	acco	rdance wit	h the Tarif	<u>f.</u>		
				_		
3.0	The	scope of t	he Definitiv	ve Interconnection	System Impact Study	shall be subject to the
	assu	mptions se	et forth in A	Attachment A to this	s Agreement.	
4.0	The	<u>Definitive</u>	<u>Interconne</u>	ection System Impa	act Study shall be bas	ed upon the technical
	inforr	mation pro	ovided by	Interconnection	Customer in the Inte	rconnection Request,
	subje	ect to any	modification	ons in accordance	with Section 4.4 of th	e LGIP. Transmission
	Provi	ider rese	rves the	right to reques	st additional technic	al information from
						consistent with Good
	Utility	/ Practice	during the	course of the Def	nitive Interconnection	System Impact Study.
						accordance with the
	LGIP	, the time	to comple	te the Definitive In	terconnection System	Impact Study may be

5.0 The Definitive Interconnection System Impact Study reports (though Phase 2, and

extended.

including Phase 3 if applicable) shall provide the following information as appropriate:

- identification of any circuit breaker short circuit capability limits exceeded as a result of the interconnection;
- identification of any thermal overload or voltage limit violations resulting from the interconnection;
- identification of any instability or inadequately damped response to system disturbances resulting from the interconnection; and
- description and non-binding, good faith estimated cost of facilities required to interconnect the Large Generating Facility to the Transmission System and to address the identified short circuit, instability, and power flow issues.
- Interconnection Customer shall provide the deposit as specified in Section 3.4 of the LGIP for the performance of the Definitive Interconnection System Impact Study and the Interconnection Facilities Study. Transmission Provider's good faith estimate for the time of completion of the Definitive Interconnection System Impact Study (Phase 2) is [insert date].

Upon receipt of the Interconnection Facilities Study results (Phase 4 Results), or withdrawal of the Interconnection Request, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Definitive Interconnection System Impact Study and the actual costs of the Interconnection Facilities Study, and the Withdrawal Penalty, as applicable, allocated according to the LGIP.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, except as otherwise provided herein. As provided in the LGIP, Interconnection Customer has thirty (30) Calendar Days of receipt of an invoice from Transmission Provider to pay any undisputed costs. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

Miscellaneous. The Definitive Interconnection System Impact Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, that are consistent with regional practices, Applicable Laws and Regulations and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS THEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]						
Ву:	Ву:					
Title:	Title:					
Date:	Date:					
[Insert name of Interconn	ection Customer]					
By:						
Title:						
Date:						

Attachment A to Appendix 2 **Definitive Interconnection System Impact Study Agreement**

ASSUMPTIONS USED IN CONDUCTING THE DEFINITIVE INTERCONNECTION SYSTEM IMPACT STUDY

The Definitive Interconnection System Impact Study shall be based upon the information set forth in the Interconnection Request(s) and results of applicable prior studies, subject to any modifications in accordance with Section 4.4 of the LGIP, and the following assumptions:

<u>Designation of Point of Interconnection and configuration to be studied.</u>

[Above assumptions to be completed by Interconnection Customer and other assumptions to be provided by Interconnection Customer and Transmission Provider]

APPENDIX 3 to LGIP INTERCONNECTION FACILITIES STUDY AGREEMENT

THIS	AGREE	MENT	is made	and entered i	nto this_	day of	, 2	20 by
and	between			, a		organized and	existing under	the laws
of	the	State	of		,	("Interconnection	Customer,")	and
				а		existing under the	e laws of the	State of
			, ("Trar	smission Prov	vider "). Ir	nterconnection Custo	omer and Tran	smission
Prov	ider each	n may b	e referred	to as a "Party	," or coll	ectively as the "Parti	es."	

RECITALS

WHEREAS, Interconnection Customer is proposing to develop a Large Generating Facility or generating capacity addition to an existing Generating Facility consistent with the Interconnection Request submitted by Interconnection Customer dated _____; and

WHEREAS, Interconnection Customer desires to interconnect the Large Generating Facility with the Transmission System under (check one only):

Network Resource Interconnection Service

Energy Resource Interconnection Service:

WHEREAS, Transmission Provider has completed a Definitive Interconnection System Impact Study (the "System Impact Study") and provided the results of said study to Interconnection Customer; and

WHEREAS, Interconnection Customer has requested Transmission Provider to perform an Interconnection Facilities Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the Definitive Interconnection System Impact Study in accordance with Good Utility Practice to physically and electrically connect the Large Generating Facility to the Transmission System.

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- <u>2.</u>0 Interconnection Customer elects and Transmission Provider shall cause an Interconnection Facilities Study consistent with this LGIP to be performed in accordance with the Tariff.
- The scope of the Interconnection Facilities Study shall be subject to the assumptions set 3.0 forth in Attachment A to this Agreement.
- The Interconnection Facilities Study report (i) shall provide a description of, estimated cost of, schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the Definitive Interconnection System Impact Study.

- 5.0 Interconnection Customer shall meet the requirements specified under the LGIP prior to the performance of the Interconnection Facilities Study. The time for completion of the Interconnection Facilities Study is specified in Attachment A.
- 6.0 Interconnection Customer shall have provided the deposit as specified in Section 3.4.1 of the LGIP for the performance of the Definitive Interconnection System Impact Study and the Interconnection Facilities Study.

Upon receipt of the Interconnection Facilities Study results (Phase 4 Results), Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Definitive Interconnection System Impact Study and the actual costs of the Interconnection Facilities Study, which includes costs allocated according to the LGIP, and the Withdrawal Penalty calculated pursuant to Section 3.7.1, if applicable.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, except as otherwise provided herein. As provided in the LGIP, Interconnection Customer has thirty (30) Calendar Days of receipt of an invoice from Transmission Provider to pay any undisputed costs. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]						
By:	By:					
Title:	Title:					
Date:	Date:					
[Insert name of Interconnection Cus	tomer]					
By:						
Title:						
Date:						

Attachment A to Appendix 3 **Interconnection Facilities Study Agreement**

DATA FORM TO BE PROVIDED BY INTERCONNECTION CUSTOMER WITH THE INTERCONNECTION FACILITIES STUDY AGREEMENT

Provide location plan and simplified one-line diagram of the plant and station facilities. For staged projects, please indicate future generation, transmission circuits, etc.

One set of metering is required for each generation connection to the new ring bus or existing Transmission Provider station. Number of generation connections:

On the one line diagram indicate the generation capacity attached at each metering location. (Maximum load on CT/PT)

On the one line diagram indicate the location of auxiliary power. (Minimum load on CT/PT)

<u>Amps</u>
Will an alternate source of auxiliary power be available during CT/PT maintenance? Yes No
Will a transfer bus on the generation side of the metering require that each meter set be designed for the total plant generation? Yes No (Please indicate on one line diagram).
What type of control system or PLC will be located at Interconnection Customer's Large Generating Facility?
What protocol does the control system or PLC use?
Please provide a 7.5-minute quadrangle of the site. Sketch the plant, station, transmission line, and property line.
Physical dimensions of the proposed interconnection station:
Bus length from generation to interconnection station:
Line length from interconnection station to Transmission Provider's transmission line.
Tower number observed in the field. (Painted on tower leg)*
Number of third party easements required for transmission lines*:

* To be completed in coordination with Transmission Provider.

Is the Large Generating Facility in Transmission Provider's service area?

Yes No Local provider:

Please provide proposed schedule dates:

Begin Construction Date:

Generator step-up transformer receives back feed power Date:

Generation Testing Date:

Commercial Operation Date:

Document Accession #: 20230531-5351 Filed

Filed Date: 05/31/2023

APPENDIX 4.1 to LGIP

<u>Transitional Interconnection Facilities Study Agreement</u>

THIS	AGREE	<u>MENT</u>	is made	and entered	into this_	day of	, 2	20by
and b	<u>etween</u>			, a		organized and	existing under	the laws
of	the	State	of		,	("Interconnection	Customer,")	and
				а		existing under th	e laws of the	State of
			, ("Trar	smission Pro	vider"). Ir	nterconnection Custo	omer and Trans	smission
Provi	der each	n mav b	e referred	d to as a "Par	tv." or coll	ectively as the "Part	ies."	

RECITALS

WHEREAS, Interconnection Customer is proposing to develop a Large Generating Facility or generating capacity addition to an existing Generating Facility consistent with the Interconnection Request submitted by Interconnection Customer dated ; and

<u>WHEREAS</u>, Interconnection Customer desires to interconnect the Large Generating Facility with the Transmission System; and

WHEREAS, Interconnection Customer has requested Transmission Provider to continue processing its Interconnection Facilities Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the System Impact Study in accordance with Good Utility Practice to physically and electrically connect the Large Generating Facility to the Transmission System; and

WHEREAS, Interconnection Customer has executed and Transmission Provider has accepted an Interconnection Facilities Study Agreement prior to the effective date of Transmission Provider's Definitive Interconnection Study Process; and

WHEREAS, Interconnection Customer has provided certain requirements described in the LGIP including a deposit on Transmission Provider's Interconnection Facilities and Network Upgrades.

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- 1.0 When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- 2.0 Interconnection Customer elects and Transmission Provider shall cause to be performed an Interconnection Facilities Study consistent with this LGIP in accordance with the Tariff.
- 3.0 The scope of the Interconnection Facilities Study shall be subject to the assumptions set forth in Attachment A to this Agreement which shall be the same assumptions as the previous Interconnection Facilities Study Agreement.
- 4.0 The Interconnection Facilities Study report (i) shall provide a description, estimated cost of (consistent with Attachment A), schedule for required facilities to interconnect the

Filed Date: 05/31/2023

- <u>Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the most recently published System Impact Study.</u>
- 5.0 Interconnection Customer has met certain requirements described in the LGIP. The time for completion of the Interconnection Facilities Study is specified in Attachment A.
- 6.0 Interconnection Customer shall have previously provided the deposit for the performance of the Interconnection Facilities Study.

Upon receipt of the Interconnection Facilities Study results, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Interconnection Facilities Study, as allocated pursuant to the LGIP.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]							
By:	By:						
Title:	Title:						
Date:	Date:						
[Insert name of Interconnection Customer]							
By:							
Title:							
Date:							

Attachment A to Appendix 4.1 Transitional Interconnection Facilities Study Agreement

ASSUMPTIONS USED IN CONDUCTING THE TRANSITIONAL INTERCONNECTION FACILITIES STUDY

[Assumptions to be completed by Interconnection Customer and Transmission Provider]

Filed Date: 05/31/2023

APPENDIX 4.2 to LGIP Transitional Cluster Study Agreement

THIS	AGREE	MENT	is made	and entered	d into this	day of	, 2	0by
and	between			, a		organized and	existing under	the laws
of	the	State	of		,	("Interconnection	Customer,")	and
				а		existing under the	e laws of the	State of
			, ("Trar	smission Pr	ovider").	Interconnection Custo	mer and Trans	mission
Prov	ider each	n may b	e referred	d to as a "Pa	rty," or co	llectively as the "Partie	es."	

RECITALS

WHEREAS, Interconnection Customer is proposing to develop a Large Generating Facility or generating capacity addition to an existing Generating Facility consistent with the Interconnection Request submitted by Interconnection Customer dated ; and

<u>WHEREAS</u>, Interconnection Customer desires to interconnect the Large Generating Facility with the Transmission System under (check one only):

Network Resource Interconnection Service

Energy Resource Interconnection Service; and

WHEREAS, Interconnection Customer has requested Transmission Provider to perform a "Transitional Cluster Study," which is a combined system impact and interconnection facility Cluster Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to physically and electrically connect the Large Generating Facility to the Transmission System; and

WHEREAS, Interconnection Customer has provided certain requirements described in the LGIP including a deposit of five million dollars (\$5,000,000) on expected Transmission Provider's Interconnection Facilities and Network Upgrades; and

WHEREAS, Interconnection Customer has a valid Queue Position as of the effective date of Transmission Provider's Definitive Interconnection Study Process; and

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- 1.0 When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- 2.0 Interconnection Customer elects and Transmission Provider shall cause to be performed a combined system impact and interconnection facility Cluster Study.
- 3.0 The Transitional Cluster Study shall be based upon the technical information provided by Interconnection Customer in the Interconnection Request. Transmission Provider reserves the right to request additional technical information from Interconnection Customer as may reasonably become necessary consistent with Good Utility Practice during the course of the Transitional Cluster Study and Interconnection Customer shall

provide such data as quickly as reasonable.

- 4.0 The Transitional Cluster Study report shall provide the following information:
 - identification of any circuit breaker short circuit capability limits exceeded as a result of the interconnection;
 - identification of any thermal overload or voltage limit violations resulting from the interconnection;
 - identification of any instability or inadequately damped response to system disturbances resulting from the interconnection and
 - shall provide a description, estimated cost of, schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the most recently published System Impact Study.
- 5.0 Interconnection Customer has met certain requirements described in the LGIP.
- 6.0 Interconnection Customer shall have previously provided a deposit for the performance of Interconnection Studies.

Upon receipt of the Transitional Cluster Study results, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Transitional Cluster Study.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]							
By:	By:						
Title:	Title:						
Date:	Date:						
[Insert name of Interconnection Customer]							
By:							
Title:							
Date:							

Attachment A to Appendix 4.2 Transitional Cluster Study Agreement

ASSUMPTIONS USED IN CONDUCTING THE TRANSITIONAL CLUSTER STUDY (A COMBINED SYSTEM IMPACT AND INTERCONNECTION FACILITIES STUDY)

[Assumptions to be completed by Interconnection Customer and Transmission Provider]

APPENDIX 5.1 to LGIP INFORMATIONAL INTERCONNECTION STUDY REQUEST

- The undersigned Interconnection Customer submits this request to evaluate the interconnection of its Generating Facility with Transmission Provider's Transmission System pursuant to the Tariff.
- The type of interconnection service to be evaluated (check one):
 - Energy Resource Interconnection Service
 - Network Resource Interconnection Service
- Interconnection Customer provides the following information:
 - Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility:
 - degrees C and winter at ___ Maximum summer at degrees C megawatt electrical output of the proposed new Large Generating Facility or the amount of megawatt increase in the generating capacity of an existing Generating Facility:
 - General description of the equipment configuration;
 - Commercial Operation Date to be studied (Day, Month, and Year);
 - Name, address, telephone number, and e-mail address of Interconnection Customer's contact person:
 - Approximate location of the proposed Point of Interconnection;
 - Interconnection Customer Data (set forth in Attachment A)
 - Primary frequency response operating range for electric storage resources.
 - Requested capacity (in MW) of Interconnection Service (if lower than the Generating Facility Capacity); and
 - A Scope of Work including any additional information that may be reasonably required.
- Study deposit amount as specified in the LGIP.
- 5. For study purposes, the point of delivery to deliver within the Control Area.
- This Informational Interconnection Study Request shall be submitted to the representative indicated below:

[To be completed by Transmission Provider]

Representative of Interconnection Customer to contact:

[To be completed by Interconnection Customer]

8.	This Interconnection Request is submitted by:
	Name of Interconnection Customer:
	By (signature):
	Name (type or print):
	Title:
	Date:

Attachment A to Appendix 5.1 Informational Interconnection Study Request

LARGE GENERATING FACILITY DATA					
	UNIT RATII	NGS			
kVA	°F	Voltage			
Power Factor					
Speed (RPM)	Connection (e.g. V				
Short Circuit Ratio	Frequency, Hertz				
Stator Amperes at Rated kVA		Field Volts			
Max Turbine MW	<u>°F</u>				
Primary frequency resp	oonse operating ra	nge for electric storage resources.			
Minimum State of Cha	arde.				
Maximum State of Cha					
Maximum State of Sine					
COMBINED TUP	RBINE-GENERATO	R-EXCITER INERTIA DATA			
Inertia Constant, H =	14/4/ 000/14/4				
Moment-of-Inertia, WR ² =	kW sec/kVA lb. ft. ²				
Woment-or-mertia, vviv =	10.11.				
REACT	ANCE DATA (PER	UNIT-RATED KVA)			
<u>DII</u>	RECT AXIS QUADI	RATURE AXIS			
Synchronous – saturated	X_{dv}	<u>X_{av.}</u>			
Synchronous – unsaturated	X_{di}	<u>X_{ai}</u>			
<u>Transient – saturated</u>	X'_{dv}	<u>X'_{qv}</u>			
<u>Transient – unsaturated</u>	X' _{di}	<u>X'_{gi}</u>			
Subtransient – saturated	X" _{dv}	<u>X"_{qv}</u>			
<u>Subtransient – unsaturated</u>	<u>X"_{di}</u>	<u>X"_{qi} </u>			
Negative Sequence – saturated	<u>X2_v</u>				
Negative Sequence – unsaturate	ed X2 _i				
Zero Sequence – saturated	<u>X0_v</u>				
Zero Sequence – unsaturated	<u>X0</u> ,				
Leakage Reactance	<u>XI_m</u>				

FIELD TIME	E CONSTAN	T DATA (S	<u>EC)</u>
Open Circuit	T' _{do}	<u>T'</u>	<u>qo</u>
Three-Phase Short Circuit Transient	<u>T'_{d3}</u>	<u>T'</u>	<u></u>
Line to Line Short Circuit Transient	<u>T'_{d1}</u>		
Short Circuit Subtransient	<u>T"_d</u>	<u></u>	<u>g</u>
Open Circuit Subtransient	<u>T'_{d2}</u>		
Line to Neutral Short Circuit Transient	T" _{do}	<u>T</u> '	<u>qo</u>
ARMATURE T	IME CONST	ANT DATA	(SEC)
NOTE: If requested information is not ap	oplicable, indi	cate by ma	rking "N/A."
MW CAPABILITY LARGE GEN	AND PLAN		
ARMATURE WINDIN	NG RESISTA	NCE DATA	(PER UNIT)
$\begin{array}{cc} \underline{\text{Positive}} & \underline{R_1} \\ \underline{\text{Negative}} & \underline{R_2} \\ \underline{\text{Zero}} & \underline{R_0} \\ \end{array}$			
Rotor Short Time Thermal Capacity $I_2^2 t$: Field Current at Rated kVA, Armature Volenield Current at Rated kVA and Armature Three Phase Armature Winding Capacity Field Winding Resistance = Oh Armature Winding Resistance (Per Phase	oltage and PF re Voltage, 0 ance = ms	PF =	amps
	<u>CURVES</u>		
Provide Saturation, Vee, Reactive Control Designate normal and emergency Hydronics			
GENERATOR STEP-U	UP TRANSFO	ORMER DA	TA RATINGS
Capacity Self-cooled/Maximum	Nameplate		

Identify appropriate IEEE model block diagram of governor system for computer representation in power system stability simulations and the corresponding governor system constants for use in the model.

WIND GENERATORS

Number of generators to b	e interconnected	pursuant to	this Interconne	ection Request:		
Elevation:	Single Phase	Three	Phase	_		
Inverter manufacturer, model name, number, and version:						
List of adjustable setpoints for the protective equipment or software:						

Note: A completed General Electric Company Power Systems Load Flow (PSLF) data sheet or other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate to the proposed device, then they shall be provided with the study request and discussed with the Transmission Provider in advance of the study.

INDUCTION GENERATORS	INDU	CTIC) NC	GEN	ERA1	TORS
----------------------	------	------	------	-----	------	-------------

/*) Field Volte:	
(*) Field Volts:	
(*) Field Amperes:	
(*) Motoring Power (kW):	
(*) Neutral Grounding Resistor (If Ap	plicable:
(*) I ₂ ² t or K (Heating Time Constant)	
(*) Rotor Resistance:	
(*) Stator Resistance:	
(*) Stator Reactance:	
(*) Rotor Reactance:	
(*) Magnetizing Reactance:	
(*) Short Circuit Reactance:	
(*) Exciting Current:	
(*) Temperature Rise:	
(*) Frame Size:	
(*) Design Letter:	
(*) Reactive Power Required In Vars	(No Load):
(*) Reactive Power Required In Vars	(Full Load):
(*) Total Rotating Inertia, H:	Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Informational Interconnection Study Request to determine if the information designated by (*) is required.

THIS AGREEMENT is made and entered into this day of

APPENDIX 5.2 to LGIP INFORMATIONAL INTERCONNECTION STUDY AGREEMENT

20

THIS AGE	REEMENT is made and entered into this	day of	, 20	_ by and
<u>between</u>	, a	organ	ized and existi	ng under
the laws	of the State of	, ("Interconne	ection Custome	er,") and
	a	existing under	the laws of the	State of
	_, ("Transmission Provider""). Interconne	ction Customer and	Transmission	Provider
each may	be referred to as a "Party," or collectively a	as the "Parties."		
	RECITAL	<u>s</u>		
	S, Interconnection Customer is evaluating		Generating F	acility or
generating	g capacity addition to an existing Generatin	g Facility and		
	S, Interconnection Customer is proposing sion System; and	g to evaluate an in	terconnection	with the
WHEREA	S, Interconnection Customer has submitte	d to Transmission Pr	ovider an Infor	mational
	ection Study Interconnection Request; and		ovider dir iiiler	mational
		•		
NOW. TH	EREFORE , in consideration of and subject	t to the mutual cove	nants containe	ed herein
	s agree as follows:			
	hen used in this Agreement, with initial ca e meanings indicated in Transmission Prov			nall have
2.0 Int	erconnection Customer elects and Transm	ission Provider shall	cause an Infor	mational
	rerconnection Study consistent with this LC uriff.	GIP to be performed	in accordance	with the
3.0 Th	ne scope of the Informational Intercor	unaction Study sha	ll he subject	to the

The Informational Interconnection Study shall be performed solely for informational

The Informational Interconnection Study report shall provide a sensitivity analysis based

on the assumptions specified by Interconnection Customer in Attachment A to this Agreement. The Informational Interconnection Study shall identify Transmission Provider's Interconnection Facilities and the Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the assumptions

assumptions set forth in Attachment A to this Agreement.

specified by Interconnection Customer in Attachment A.

4.0

5.0

purposes.

Interconnection Customer shall provide a deposit as identified in the LGIP for the 6.0 performance of the Informational Interconnection Study. Transmission Provider's good faith estimate for the time of completion of the Informational Interconnection Study is [insert date].

Upon receipt of the Informational Interconnection Study, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Informational Interconnection Study.

Any difference between the initial deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.

7.0 Miscellaneous. The Informational Interconnection Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

	isert name of Transmission Provider	or Transmission Owner, if applicable
By:	<u>r. </u>	By:
<u>Title</u>	le:	Title:
_ <u>Date</u>	ate:	Date:
[lns	sert name of Interconnection Custo	mer]
By:	<u>r. </u>	_
<u>Title</u>	le:	_
Date	ate:	
		
Apr	ppendix 1. INTERCONNECTION	REQUEST FOR A LARGE GENERATING
	ACILITY APPI INTERCONNE	ENDIX 1 to LGIP ECTION REQUEST FOR A NERATING FACILITY
FAC	APPI INTERCONNE LARGE GE . The undersigned Interconnection	ENDIX 1 to LGIP ECTION REQUEST FOR A
FAC	APPI INTERCONNE LARGE GE The undersigned Interconnection Large Generating Facility with pursuant to a Tariff. This Interconnection Request is A proposed new Large	ENDIX 1 to LGIP ECTION REQUEST FOR A NERATING FACILITY In Customer submits this request to interconnect its Transmission Provider's Transmission System For (check one): Generating Facility. Erating capacity or a Material Modification of an

4.	— Check here only if Interconnection Customer requesting Network Resource
	Interconnection Service also seeks to have its Generating Facility studied for
	Energy Resource Interconnection Service.
	Energy Resource Interconnection Service.
5.	Interconnection Customer provides the following information:
	a. Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility;
	b. Maximum summer at degrees C and winter at degrees C megawatt electrical output of the proposed new Large Generating Facility or the amount of megawatt increase in the generating eapacity of an existing Generating Facility;
	c. General description of the equipment configuration;
	d. Commercial Operation Date (Day, Month, and Year);
	e. Name, address, telephone number, and e-mail address of Interconnection Customer's contact person;
	f. Approximate location of the proposed Point of Interconnection (optional); and
	g. Interconnection Customer Data (set forth in Attachment A).
	h. Primary frequency response operating range for electric storage resources.
	i. Requested capacity (in MW) of Interconnection Service (if lower than the Generating Facility Capacity).
6.	Applicable Initial Deposit amount as specified in the LGIP.
7.	Evidence of Site Control as specified in the LGIP (check one) Is attached to this Interconnection Request Is NOT attached to this Interconnection Request, but a Deposit in Lieu of Site Control is provided
8.	This Interconnection Request shall be submitted to the representative indicated below:

	[To be completed by Transmission Provider]
9	Representative of Interconnection Customer to contact:
	-{To be completed by Interconnection Customer}
10.	This Interconnection Request is submitted by:
	Name of Interconnection Customer:
	By (signature):
	Name (type or print):
	Title:
	Date:

Attachment A. LARGE GENERATING FACILITY DATA

Attachment A to Appendix 1 Interconnection Request

LARGE GENERATING FACILITY DATA UNIT RATINGS

kVA °F	Voltage	
Power Factor		
Speed (RPM)		
Connection (e.g. Wye)		
Short Circuit Ratio	Freq	luency, Hertz
Stator Amperes at Rated kVA_	Field	d Volts
Max Turbine MW		
Primary frequency response op	erating range for elect	tric storage resources:
Minimum State of Charg	ge:	
Maximum State of Char	·ge:	
COMPINED TUDDI	NE CENEDATOD E	
COMBINED TURBIN	NE-GENEKA I UK-I	EXCITER INERTIA DA
Inertia Constant, H =	kW se	ec/kVA-
		1b ft 2
Moment of Inertia, WR2 =	CE DATA (PER UN	IT-RATED KVA)
Moment-of-Inertia, WR2 =	CE DATA (PER UN	
Moment-of-Inertia, WR2 =	CE DATA (PER UN DIRECT AXIS	IT-RATED KVA) —QUADRATURE AXI
Moment-of-Inertia, WR2 = REACTANO Synchronous - saturated	CE DATA (PER UN DIRECT AXIS X _{dv}	IT-RATED KVA) —QUADRATURE AXI —X _{qv}
REACTANO Synchronous - saturated Synchronous - unsaturated	CE DATA (PER UN — DIRECT AXIS — X _{dv} — — — — — — — — — — — — — — — — — — —	IT-RATED KVA) —QUADRATURE AXI —X _{qv} ————————————————————————————————————
Synchronous - saturated Synchronous - unsaturated Transient - saturated X'd	CE DATA (PER UN) DIRECT AXIS X _{dv} X _{di}	IT-RATED KVA) —QUADRATURE AXI —X _{qv} ———— —X _{qi} ————
Synchronous – saturated Synchronous – unsaturated Transient – saturated Transient – unsaturated	CE DATA (PER UN DIRECT AXIS X _{dv} X _{di} X' _{qv} X' _{di}	IT-RATED KVA) —QUADRATURE AXI —X _{qv} ————————————————————————————————————
Synchronous – saturated Synchronous – unsaturated Transient – saturated Transient – unsaturated Subtransient – saturated	CE DATA (PER UN) DIRECT AXIS X _{dv} X _{di} X' _{qv} X' _{di} X' _{di}	IT-RATED KVA) —QUADRATURE AXI —X _{qv} ——— —X _{qi} ——— —X' _{qi} ——— —X' _{qi} ————
Synchronous - saturated Synchronous - unsaturated Transient - saturated X'd Transient - unsaturated Subtransient - unsaturated Subtransient - unsaturated	CE DATA (PER UN) DIRECT AXIS X _{dv}	IT-RATED KVA) — QUADRATURE AXI — X _{qv} — — — — — — — — — — — — — — — — — — —
Synchronous – saturated Synchronous – unsaturated Transient – saturated Transient – unsaturated Subtransient – saturated	CE DATA (PER UN) DIRECT AXIS X _{dv} X _{di} X' _{qv} X' _{di} X' _{di} X'' _{di} X'' _{di} X'' _{di}	T-RATED KVA) — QUADRATURE AXI — X _{qv} — — — — — — — — — — — — — — — — — — —
Synchronous - saturated Synchronous - unsaturated Transient - saturated Transient - unsaturated Subtransient - unsaturated Subtransient - unsaturated Subtransient - unsaturated Negative Sequence - saturated	CE DATA (PER UN) DIRECT AXIS X _{dv}	IT-RATED KVA) — QUADRATURE AXI — X _{qv} — — — — — — — — — — — — — — — — — — —
Synchronous – saturated Synchronous – unsaturated Transient – saturated Subtransient – unsaturated Subtransient – unsaturated Subtransient – unsaturated Negative Sequence – saturated Negative Sequence – unsaturated	CE DATA (PER UN) DIRECT AXIS X _{dv}	T-RATED KVA

FIELD TIME CONSTANT DATA (SEC)

Open Circuit T'do	-T'do
Three-Phase Short Circuit Transient	-T'd3
Line to Line Short Circuit Transient	<u></u>
Line to Neutral Short Circuit Transient	<u>-T'd1</u>
Short Circuit Subtransient	
Open Circuit Subtransient	

ARMATURE TIME CONSTANT DATA (SEC)

Three Phase Short Circuit	$-T_{e}^{3}$
Line to Line Short Circuit	
Line to Neutral Short Circuit	—T _e ¹

NOTE: If requested information is not applicable, indicate by marking "N/A."

MW CAPABILITY AND PLANT CONFIGURATION LARGE GENERATING FACILITY DATA

ARMATURE WINDING RESISTANCE DATA (PER UNIT)

	4
Positive	<u>D</u>
1 OSITIVE	Ι
Negative	R
riogative	14
Zero	R
	1 VH

그 그 그 그 그 그 그 그 그 그 그 그 그 그 그 그 그 그 그	
Rotor Short Time Thermal Capacity I ₂ t =	
	amps
Field Current at Rated kVA and Armature Voltage, 0 PF =	amps
Three Phase Armature Winding Capacitance =microfarac]
Field Winding Resistance = ohms °C	
Armature Winding Resistance (Per Phase) =ohms	<u> °C </u>

CURVES

Provide Saturation, Vee, Reactive Capability, Capacity Temperature Correction curves. Designate normal and emergency Hydrogen Pressure operating range for multiple curves.

GENERATOR STEP-UP TRANSFORMER DATA RATINGS				
Capacity Self-cooled/-				
Maximum Nameplate				
/kVA				
Voltage Ratio(Generator Side/System side/Tertiary)/kV				
Winding Connections (Low V/High V/Tertiary V (Delta or Wye))				
Fixed Taps Available				
Present Tap Setting				
IMPEDANCE-				
Positive Z ₁ (on self-cooled kVA rating) % X/R				
Zero Z ₀ (on self-cooled kVA rating) % X/R				

EXCITATION SYSTEM DATA

Identify appropriate IEEE model block diagram of excitation system and power systemstabilizer (PSS) for computer representation in power system stability simulations and the corresponding excitation system and PSS constants for use in the model.

GOVERNOR SYSTEM DATA

Identify appropriate IEEE model block diagram of governor system for computerrepresentation in power system stability simulations and the corresponding governorsystem constants for use in the model.

WIND CENERATORS

Number of generators to be interest.	onnected pursuant to this l	Interconnection Request:
Elevation:	Single Phase	Three Phase
Inverter manufacturer, model nam	ne, number, and version:	
List of adjustable setpoints for the	protective equipment or s	oftware:

Note: A completed General Electric Company Power Systems Load Flow (PSLF) datasheet or other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate tothe proposed device, then they shall be provided and discussed at Scoping Meeting.

INDUCTION GENERATORS

(*) Field Volts:
(*) Field Amperes:
(*) Motoring Power (kW):
(*) Neutral Grounding Resistor (If Applicable):
(*) 122t or K (Heating Time Constant):
(*) Rotor Resistance:
(*) Stator Resistance:
(*) Stator Reactance:
(*) Rotor Reactance:
(*) Magnetizing Reactance:
(*) Short Circuit Reactance:
(*) Exciting Current:
(*) Temperature Rise:
(*) Frame Size:
(*) Design Letter:
(*) Reactive Power Required In Vars (No Load):
(*) Reactive Power Required In Vars (Full Load):
(*) Total Rotating Inertia, H:Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Interconnection Request to determine if the information designated by (*) is required.

Appendix 2. INTERCONNECTION SYSTEM IMPACT STUDY AGREEMENT

APPENDIX 2 to LGIP -INTERCONNECTION SYSTEM IMPACT STUDY AGREEMENT-

THI	S AGREEMENT is	s made and entered	into this	 day of,
20 by ar	nd between	, a		organized and
existing unc	ler the laws of the St	tate of	, ("Int	erconnection-
Customer ") and	2	exis	ting under the laws of
the State of		, ("Transmissic	o n Provider "). Ir	nterconnection-
Customer as the "Part	nd Transmission Pro	vider each may be	referred to as a "	Party," or collectively
		RECITALS	}-	
Generating consistent v	EREAS, Interconnect ;	ng capacity addition	rto an existing G	enerating Facility
	EREAS, Interconnec Facility with the Tra			ect the Large
perform an	Interconnection Syst	tem Impact Study t	o assess the impa	mission Provider to act of interconnecting ny Affected Systems;
	V, THEREFORE, i erein the Parties agre		and subject to the	e mutual covenants
1.0		Agreement, with in anings indicated in		on, the terms specified ovider's FERC
2.0	be performed an I	Customer elects and interconnection Systems LGIP in accordance	tem Impact Stud	
	3.0	shall be subject	e Interconnection to the assumption this Agreement	
4.0	The Interconnection	on System Impact (Study will be bas	ed upon the technical-

information provided by Interconnection Customer in the Interconnection Request, subject to any modifications in accordance with Section 4.4 of the LGIP. Transmission Provider reserves the right to request additional technical information from Interconnection Customer as may reasonably become necessary consistent with Good Utility Practice during the course of the Interconnection Customer System Impact Study. If Interconnection Customer modifies its designated Point of Interconnection, Interconnection Request, or the technical information provided therein is modified, the time to complete the Interconnection System Impact Study may be extended.

The Interconnection System Impact Study report shall provide the

- following information: identification of any circuit breaker short circuit capability limitsexceeded as a result of the interconnection; identification of any thermal overload or voltage limit violationsresulting from the interconnection; identification of any instability or inadequately damped response to system disturbances resulting from the interconnection and description and non-binding, good faith estimated cost of facilities required to interconnect the Large Generating Facility to the Transmission System and to address the identified short circuit, instability, and power flow issues. Transmission Provider's good faith estimate for the time of completion of the Interconnection System Impact Study is [insert date]. Upon receipt of the Interconnection System Impact Study, Transmission Provider shall charge and Interconnection Customer shall pay the actualcosts of the Interconnection System Impact Study. Any difference between the Initial Deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, less any amounts deemed non-refundable as per Section 3.3.1 of this LGIP.
 - 7.0 Miscellaneous. The Interconnection System Impact Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, that are consistent with regional

practices, Applicable Laws and Regulations and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be

consistent with the provisions of the LGIP and the LGIA.			
IN WITNESS THEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.			
[Insert name of Transmission Provide	er or Transmission Owner, if applicable]		
Ву:	By:		
Title:	Title:		
Date:	Date:		
Insert name of Interconnection Cust By:	-		
Title:	_		
Date:			

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Attachment A. ASSUMPTIONS USED

Attachment A To Appendix 2
-Interconnection System Impact
-Study Agreement

ASSUMPTIONS USED IN CONDUCTING THE INTERCONNECTION SYSTEM IMPACT STUDY

The Interconnection System Impact Study will be based upon the following assumptions:
 Designation of Point of Interconnection and configuration to be studied. Designation of alternative Point(s) of Interconnection and configuration.
[Above assumptions to be completed by Interconnection Customer and other assumptions to be provided by Interconnection Customer and Transmission Provider]

Appendix 3. INTERCONNECTION FACILITIES STUDY AGREEMENT

APPENDIX 3 to LGIP INTERCONNECTION FACILITIES STUDY AGREEMENT

THIS AGE	REEMENT is made and	entered into this	day of	, 20
by and between_		, a		organized and
existing under the	e laws of the State of		, ("Interconnec	etion-
Customer,") and		a	existing ur	nder the laws of
the State of		, ("Transmis	ssion Provider ")	
	Customer and Transmiss tively as the "Parties."	ion Provider eac l	h may be referre	d to as a
	RI	ECITALS-		
Facility or genera	erconnection Customer inting capacity addition to on Request submitted by	o an existing Gen	erating Facility	consistent with
	erconnection Customer of Fransmission System;	desires to interco	nnect the Large	Generating-
	nsmission Provider has m Impact Study") and p Customer; and			
WHEREAS. Inte	erconnection Customer l	nas elected		rconnection
	n with its election betwe		-	
	sted Transmission Provid		•	
	and estimate the cost of	•		
	c needed to implement t l			
Impact Study in a	eccordance with Good U	tility Practice to	physically and e	electrically
connect the Large	e Generating Facility to	the Transmission	System.	
	ORE, in consideration	of and subject to	the mutual cove	enants contained
herein the Parties	agreed as follows:			
sha l	en used in this Agreeme Il have the meanings ind roved LGIP.		1	
	erconnection Customer e erconnection Facilities S			

	be performed in accordance with the Tariff.
3.0	The scope of the Interconnection Facilities Study shall be subject to the assumptions set forth in Attachment A and the data provided in Attachment B to this Agreement.
4.0	The Interconnection Facilities Study report (i) shall provide a description, estimated cost of (consistent with Attachment A), schedule for required facilities to interconnect the Large Generating Facility to the Transmission-System and (ii) shall address the short circuit, instability, and power flow issues identified in the Interconnection System Impact Study.
5.0	The time for completion of the Interconnection Facilities Study is specified in Attachment A.
	Transmission Provider shall invoice Interconnection Customer on a monthly basis for the work to be conducted on the Interconnection-Facilities Study each month. Interconnection Customer shall pay invoiced amounts within thirty (30) Calendar Days of receipt of invoice. Transmission Provider shall continue to hold the amounts on deposit until-settlement of the final invoice.
6.0	Miscellaneous. The Interconnection Facility Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.
	VITNESS WHEREOF, the Parties have caused this Agreement to be duly their duly authorized officers or agents on the day and year first above
	ne of Transmission Provider or Transmission Owner, if applicable
5y: Fitle:	By:
)ate:	Date:

Incort nama	of Intorconnaction	Customorl
msert manne	or mitter commection	Customer

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Attachment A. INTERCONNECTION CUSTOMER SCHEDULE ELECTION FOR CONDUCTING THE INTERCONNECTION FACILITIES STUDY

Attachment A To Appendix 3
Interconnection Facilities
Study Agreement

INTERCONNECTION CUSTOMER SCHEDULE ELECTION FOR CONDUCTING THE INTERCONNECTION FACILITIES STUDY

Transmission Provider shall use Reasonable Efforts to complete the study and issue a draft Interconnection Facilities Study report to Interconnection Customer within ninety (90) Calendar Days after a receipt of an executed copy of this Interconnection Facilities Study Agreement.

Attachment B. DATA FORM TO BE PROVIDED BY INTERCONNECTION-CUSTOMER WITH THE INTERCONNECTION FACILITIES STUDY AGREEMENT

Attachment B to Appendix 3
-Interconnection Facilities
Study Agreement

DATA FORM TO BE PROVIDED BY INTERCONNECTION CUSTOMER-WITH THE INTERCONNECTION FACILITIES STUDY AGREEMENT

Provide location plan and simplified one-line diagram of the plant and station facilities. For staged projects, please indicate future generation, transmission circuits, etc.

One set of metering is required for each generation connection to the new ring bus or existing Transmission Provider station. Number of generation connections: _____

On the one line diagram indicate the generation capacity attached at each metering location. (Maximum load on CT/PT)

On the one line diagram indicate the location of auxiliary power. (Minimum load on CT/PT) Amps

•
Will an alternate source of auxiliary power be available during CT/PT maintenance?
Yes No-
Will a transfer bus on the generation side of the metering require that each meter set be designed for the total plant generation? Yes
No (Please indicate on one line diagram).
What type of control system or PLC will be located at Interconnection Customer's Large Generating Facility?
What protocol does the control system or PLC use?

Please provide a 7.5-minute quadrangle of the site. Sketch the plant, station, transmission

line, and property line.

Physical dimensions of the proposed interconnection station:				
Bus length from generation to interconnection station:				
Line length from interconnection station to Transmission Provider's transmission line.				
Tower number observed in the field. (Painted on tower leg)*				
Number of third party easements required for transmission lines*:				
* To be completed in coordination with Transmission Provider.				
Is the Large Generating Facility in the Transmission Provider's service area? ———————————————————————————————————				
Please provide proposed schedule dates:				
Begin Construction	Date:			
Generator step-up transformer receives back feed power	Date:			
Generation Testing	Date:			
Commercial Operation	Date:			

Appendix 4. [Reserved for Future Use] - Appendix 5. OPTIONAL INTERCONNECTION STUDY AGREEMENT

APPENDIX 5 to LGIP OPTIONAL INTERCONNECTION STUDY AGREEMENT

	S AGREEMENT is made and entered into this day of,
	nd between, a, and existing under the laws of the State of,
_	
	ection Customer,") and a a ader the laws of the State of , ("Transmission Provider"). Interconnection
_	nd Transmission Provider each may be referred to as a "Party," or collectively
as the "Part	
	RECITALS-
	EREAS, Interconnection Customer is proposing to develop a Large
	Facility or generating capacity addition to an existing Generating Facility
	with the Interconnection Request submitted by Interconnection Customer
dated	;
/	
	EREAS, Interconnection Customer is proposing to establish an
interconnec	etion with the Transmission System; and
WH	EREAS, Interconnection Customer has submitted to Transmission Provider
	nection Request; and
an intercon	nection request, and
WH	EREAS, on or after the date when Interconnection Customer receives the
	etion System Impact Study results, Interconnection Customer has further
	hat Transmission Provider prepare an Optional Interconnection Study;
•	
NO 1	W, THEREFORE, in consideration of and subject to the mutual covenants
contained h	erein the Parties agree as follows:
1.0	When used in this Agreement, with initial capitalization, the terms specified
1.0	shall have the meanings indicated in Transmission Provider's FERC-
	approved LGIP.
2.0	Interconnection Customer elects and Transmission Provider shall cause an
	Optional Interconnection Study consistent with Section 10.0 of this LGIP to
	be performed in accordance with the Tariff.

- 3.0 The scope of the Optional Interconnection Study shall be subject to the assumptions set forth in Attachment A to this Agreement.
- 4.0 The Optional Interconnection Study shall be performed solely for informational purposes.
- 5.0 The Optional Interconnection Study report shall provide a sensitivity analysis based on the assumptions specified by Interconnection Customer in Attachment A to this Agreement. The Optional Interconnection Study will identify Transmission Provider's Interconnection Facilities and the Network Upgrades, and the estimated cost thereof, that may be required to provide interconnection service based upon the assumptions specified by Interconnection Customer in Attachment A.
- 6.0 Interconnection Customer shall provide a deposit of \$25,000 for the performance of the Optional Interconnection Study. Transmission Provider's good faith estimate for the time of completion of the Optional Interconnection Study is (insert date).
- Upon receipt of the Optional Interconnection Study, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Optional Study.
- Any difference between the initial payment and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.
- 7.0 Miscellaneous. The Optional Interconnection Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

Insert name of	Transmission Provider o	r Transmission	Owner, if applicable
By:	B	y:	

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Title:	Title:
Date:	_ Date:
[Insert name of Interconnection Custo	mer]
Ву:	
Title:	-
Date:	-

Appendix 6. STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT (LGIA)

Appendix 6 to the Standard Large Generator Interconnection Procedures

STANDARD LARGE GENERATOR **INTERCONNECTION AGREEMENT (LGIA)**

(Applicable to Generating Facilities that exceed 20 MW)

Note: This form of agreement is also used for a Small Generating Facility seeking **Network Resource Interconnection Service. Small Generating Facilities are** studied under the Large Generator Interconnection Procedures.

Appendix 6. STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT (LGIA)

TABLE OF CONTENTS

Recitals

Article 1.	Definitions
Article 2.	Effective Date, Term, and Termination
2.1	Effective Date
2.2	Term of Agreement
2.3	Termination Procedures
	2.3.1 Written Notice
	2.3.2 Default
2.4	Termination Costs
2.5	
2.6	Survival
Article 3.	Regulatory Filings
3.1	Filing
Article 4.	Scope of Service
4.1	Interconnection Product Options
	4.1.1 Energy Resource Interconnection Service
	4.1.1.1 The Product
	4.1.1.2 Transmission Delivery Service Implications
	4.1.2 Network Resource Interconnection Service
	4.1.2.1 The Product
	4.1.2.2 Transmission Delivery Service Implications
4.2	Provision of Service
4.3	Performance Standards
4.4	No Transmission Delivery Service
4.5	Interconnection Customer Provided Services
Article 5.	Interconnection Facilities Engineering, Procurement, and Construction
5.1	Options
	5.1.1 Standard Option
	5.1.2 Alternate Option
	5.1.3 Option to Build
	5.1.4 Negotiated Option
5.2	General Conditions Applicable to Option to Build

- 5.4 Power System Stabilizers
- 5.5 Equipment Procurement
- 5.6 Construction Commencement
- 5.7 Work Progress
- 5.8 Information Exchange
- 5.9 Other Interconnection Options
 - 5.9.1 Limited Operation
 - 5.9.2 Provisional Interconnection Service
- 5.10 Interconnection Customer's Interconnection Facilities ('ICIF')
 - 5.10.1 Interconnection Customer's Interconnection Facility Specifications
 - 5.10.2 Transmission Provider's Review
 - 5.10.3 ICIF Construction
- 5.11 Transmission Provider's Interconnection Facilities Construction
- 5.12 Access Rights
- 5.13 Lands of Other Property Owners
- 5.14 Permits
- 5.15 Early Construction of Base Case Facilities
- 5.16 Suspension
 - 5.16.1 Effect of Missed Interconnection Customer Milestones
 - 5.16.2 Effect of Suspension; Parties' Obligations
- 5.17 Taxes
 - 5.17.1 Interconnection Customer Payments Not Taxable
 - 5.17.2 Representations and Covenants
 - 5.17.3 Indemnification for the Cost Consequences of Current Tax Liability Imposed Upon the Transmission Provider
 - 5.17.4 Tax Gross-Up Amount
 - 5.17.5 Private Letter Ruling or Change or Clarification of Law
 - 5.17.6 Subsequent Taxable Events
 - 5.17.7 Contests
 - 5.17.8 Refund
 - 5.17.9 Taxes Other Than Income Taxes
 - 5.17.10Transmission Owners Who Are Not Transmission Providers
- 5.18 Tax Status
- 5.19 Modification
 - 5.19.1 General.
 - 5.19.2 Standards
 - 5.19.3 Modification Costs
- Article 6. Testing and Inspection
 - 6.1 Pre-Commercial Operation Date Testing and Modifications
 - 6.2 Post-Commercial Operation Date Testing and Modifications
 - 6.3 Right to Observe Testing

6.4	Right to	Inspect

Article 7.	Metering

- General 7.1
- 7.2 **Check Meters**
- 7.3 Standards
- 7.4 Testing of Metering Equipment
- 7.5 Metering Data

Article 8. Communications

- 8.1 **Interconnection Customer Obligations**
- Remote Terminal Unit 8.2
- 8.3 No Annexation
- 8.4 Provision of Data from a Variable Energy Resource

Article 9. **Operations**

- 9.1 General
- Control Area Notification 9.2
- 9.3 **Transmission Provider Obligations**
- 9.4 **Interconnection Customer Obligations**
- 9.5 Start-Up and Synchronization
- Reactive Power and Primary Frequency Response 9.6
 - 9.6.1 Power Factor Design Criteria
 - 9.6.2 Voltage Schedules
 - 9.6.2.1 Voltage Regulators
 - 9.6.3 Payment for Reactive Power
 - 9.6.4 Primary Frequency Response
 - 9.6.4.1 Governor or Equivalent Controls
 - 9.6.4.2 Timely and Sustained Response
 - 9.6.4.3 Exemptions
 - 9.6.4.4 Electric Storage Resources
- Outages and Interruptions 9.7
 - 9.7.1 Outages.
 - 9.7.1.1 Outage Authority and Coordination
 - 9.7.1.2 Outage Schedules
 - 9.7.1.3 Outage Restoration
 - 9.7.2 Interruption of Service
 - 9.7.3 Under-Frequency and Over Frequency Conditions
 - 9.7.4 System Protection and Other Control Requirements 9.7.4.1 System Protection Facilities
 - 9.7.5 Requirements for Protection
 - 9.7.6 Power Quality
- Switching and Tagging Rules 9.8

9.9	Use of Interconnection Facilities by Third Parties
	9.9.1 Purpose of Interconnection Facilities
	9.9.2 Third Party Users
9.10	Disturbance Analysis Data Exchange
Article 10.	Maintenance
10.1	Transmission Provider Obligations
10.2	Interconnection Customer Obligations
10.3	Coordination
10.4	Secondary Systems
10.5	Operating and Maintenance Expenses
Article 11.	Performance Obligation
11.1	Interconnection Customer Interconnection Facilities
11.2	Transmission Provider's Interconnection Facilities
11.3	Network Upgrades and Distribution Upgrades
11.4	Transmission Credits
	11.4.1 Repayment of Amounts Advanced for Network Upgrades
	11.4.2 Special Provisions for Affected Systems
11.5	Provision of Security
11.6	Interconnection Customer Compensation
11.6.1	Interconnection Customer Compensation for Actions During Emergency
	Condition
Article 12.	Invoice
12.1	General
12.2	Final Invoice
12.3	Payment
12.4	Disputes
Article 13.	Emergencies
13.1	Definition
13.2	Obligations
13.3	Notice
13.4	Immediate Action
13.5	Transmission Provider Authority
	13.5.1 General
	13.5.2 Reduction and Disconnection
13.6	Interconnection Customer Authority
13.7	Limited Liability

Regulatory Requirements and Governing Law Regulatory Requirements Article 14.

14.1

14.2 Governing Law

- Article 15. Notices
 - 15.1 General
 - 15.2 Billings and Payments
 - 15.3 Alternative Forms of Notice
 - 15.4 Operations and Maintenance Notice
- Article 16. Force Majeure
 - 16.1 Force Majeure
- Article 17. Default
 - 17.1 Default
 - 17.1.1 General
 - 17.1.2 Right to Terminate
- Article 18. Indemnity, Consequential Damages and Insurance
 - Indemnity 18.1
 - 18.1.1 Indemnified Person
 - 18.1.2 Indemnifying Party
 - 18.1.3 Indemnity Procedures
 - Consequential Damages 18.2
 - 18.3 Insurance
- Article 19. Assignment
 - 19.1 Assignment
- Article 20. Severability
 - Severability 20.1
- Article 21. Comparability
 - Comparability 21.1
- Confidentiality Article 22.
 - Confidentiality 22.1
 - 22.1.1 Term
 - 22.1.2 Scope
 - 22.1.3 Release of Confidential Information
 - 22.1.4 Rights
 - 22.1.5 No Warranties
 - 22.1.6 Standard of Care
 - 22.1.7 Order of Disclosure
 - 22.1.8 Termination of Agreement

	22.1.9 Remedies22.1.10 Disclosure to FERC, its Staff, or a State
Article 23.	Environmental Releases
Article 24.	Information Requirements
24.1	Information Acquisition
24.2	Information Submission by Transmission Provider
24.3	Updated Information Submission by Interconnection Customer
24.4	Information Supplementation
Article 25.	Information Access and Audit Rights
25.1	Information Access
25.2	Reporting of Non-Force Majeure Events
25.3	E
25.4	8
	25.4.1 Audit Rights Period for Construction-Related Accounts and Records
	25.4.2 Audit Rights Period for All Other Accounts and Records
25.5	Audit Results
Article 26.	Subcontractors
26.1	General
26.2	Responsibility of Principal
26.3	No Limitation by Insurance
Article 27.	Disputes
27.1	_
27.2	External Arbitration Procedures
27.3	Arbitration Decisions
27.4	Costs
Article 28.	Representations, Warranties, and Covenants
28.1	General
	28.1.1 Good Standing
	28.1.2 Authority
	28.1.3 No Conflict
	28.1.4 Consent and Approval
Article 29.	Joint Operating Committee
29.1	Joint Operating Committee
Article 30.	Miscellaneous
30.1	Binding Effect

- 30.2 Conflicts
- 30.3 Rules of Interpretation
- 30.4 Entire Agreement
- 30.5 No Third Party Beneficiaries
- 30.6 Waiver
- 30.7 Headings
- 30.8 Multiple Counterparts
- 30.9 Amendment
- 30.10 Modification by the Parties
- 30.11 Reservation of Rights
- 30.12 No Partnership
- Appendix A Interconnection Facilities, Network Upgrades, and Distribution Upgrades
- Appendix B Milestones
- Appendix C Interconnection Details
- Appendix D Security Arrangements Details
- Appendix E Commercial Operation Date
- Appendix F Addresses for Delivery of Notices and Billings
- Appendix G Interconnection Requirements for a Wind Generating Plant

Standard Large Generator Interconnection Agreement

STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT

THIS STANDADD I ADOE CENEDATOD INTEDCONNECTION

THIS STANDARD LARGE GENERATOR INTERCONNECTION
AGREEMENT ("Agreement") is made and entered into this day of
20, by and between, a
organized and existing under the laws of the State/Commonwealth of
("Interconnection Customer" with a Large Generating Facility), and
, a organized
and existing under the laws of the State/Commonwealth of
("Transmission Provider and/or Transmission Owner"). Interconnection Customer and
Transmission Provider each may be referred to as a "Party" or collectively as the
"Parties."

Recitals

WHEREAS, Transmission Provider operates the Transmission System; and

WHEREAS, Interconnection Customer intends to own, lease and/or control and operate the Generating Facility identified as a Large Generating Facility in Appendix C to this Agreement; and,

WHEREAS, Interconnection Customer and Transmission Provider have agreed to enter into this Agreement for the purpose of interconnecting the Large Generating Facility with the Transmission System;

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein, it is agreed:

When used in this Standard Large Generator Interconnection Agreement, terms with initial capitalization that are not defined in Article 1 shall have the meanings specified in the Article in which they are used or the Open Access Transmission Tariff (Tariff).

Article 1. Definitions

Adverse System Impact shall mean the negative effects due to technical or operational limits on conductors or equipment being exceeded that may compromise the safety and reliability of the electric system.

Affected System shall mean an electric system other than the Transmission Provider's Transmission System that may be affected by the proposed interconnection. **Affected System Operator** shall mean the entity that operates an Affected System.

Affiliate shall mean, with respect to a corporation, partnership or other entity, each such other corporation, partnership or other entity that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such corporation, partnership or other entity.

Ancillary Services shall mean those services that are necessary to support the transmission of capacity and energy from resources to loads while maintaining reliable operation of the Transmission Provider's Transmission System in accordance with Good Utility Practice.

Applicable Laws and Regulations shall mean all duly promulgated applicable federal, state and local laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders, permits and other duly authorized actions of any Governmental Authority.

Applicable Reliability Council shall mean the reliability council applicable to the Transmission System to which the Generating Facility is directly interconnected.

Applicable Reliability Standards shall mean the requirements and guidelines of NERC, the Applicable Reliability Council, and the Control Area of the Transmission System to which the Generating Facility is directly interconnected.

Base Case shall mean the base case power flow, short circuit, and stability data bases used for the Interconnection Studies by the Transmission Provider or Interconnection Customer.

Breach shall mean the failure of a Party to perform or observe any material term or condition of the Standard Large Generator Interconnection Agreement.

Breaching Party shall mean a Party that is in Breach of the Standard Large Generator Interconnection Agreement.

Business Day shall mean Monday through Friday, excluding Federal Holidays.

Calendar Day shall mean any day including Saturday, Sunday or a Federal Holiday.

Cluster shall mean a group of Interconnection Requests that are studied together for the purpose of conducting the Interconnection Studies. A Cluster is sometimes

referred to as a Queue Cluster.

Clustering shall mean the process whereby a group of Interconnection Requests is studied together, instead of serially, for the purpose of conducting the Interconnection System Impact Studyies.

Commercial Operation shall mean the status of a Generating Facility that has commenced generating electricity for sale, excluding electricity generated during Trial Operation.

Commercial Operation Date of a unit shall mean the date on which the Generating Facility commences Commercial Operation as agreed to by the Parties pursuant to Appendix E to the Standard Large Generator Interconnection Agreement.

Confidential Information shall mean any confidential, proprietary or trade secret information of a plan, specification, pattern, procedure, design, device, list, concept, policy or compilation relating to the present or planned business of a Party, which is designated as confidential by the Party supplying the information, whether conveyed orally, electronically, in writing, through inspection, or otherwise.

Control Area shall mean an electrical system or systems bounded by interconnection metering and telemetry, capable of controlling generation to maintain its interchange schedule with other Control Areas and contributing to frequency regulation of the interconnection. A Control Area must be certified by the Applicable NERC Regional Reliability CouncilEntity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC.

Default shall mean the failure of a Breaching Party to cure its Breach in accordance with Article 17 of the Standard Large Generator Interconnection Agreement.

Dispute Resolution shall mean the procedure for resolution of a dispute between the Parties in which they will first attempt to resolve the dispute on an informal basis.

<u>Study</u>") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study Process.

<u>Definitive Interconnection System Impact Study ("DISIS")</u> shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System.

Definitive Interconnection System Impact Study Agreement ("DISIS Agreement") shall mean the form of agreement contained in the Appendix of the LGIP for conducting the Definitive Interconnection System Impact Study.

DISIS Request Window shall have the meaning set forth in Section 4.2 of the LGIP.

Distribution System shall mean the Transmission Provider's facilities and equipment used to transmit electricity to ultimate usage points such as homes and industries directly from nearby generators or from interchanges with higher voltage transmission networks which transport bulk power over longer distances. The voltage levels at which distribution systems operate differ among areas.

Distribution Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Distribution System at or beyond the Point of Interconnection to facilitate interconnection of the Generating Facility and render the transmission service necessary to effect Interconnection Customer's wholesale sale of electricity in interstate commerce. Distribution Upgrades do not include Interconnection Facilities.

Effective Date shall mean the date on which the Standard Large Generator Interconnection Agreement becomes effective upon execution by the Parties subject to acceptance by FERC, or if filed unexecuted, upon the date specified by FERC.

Emergency Condition shall mean a condition or situation: (1) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (2) that, in the case of a Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to Transmission Provider's Transmission System, Transmission Provider's Interconnection Facilities or the electric systems of others to which the Transmission Provider's Transmission System is directly connected; or (3) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Generating Facility or Interconnection Customer's Interconnection Facilities. System restoration and black start shall be considered Emergency Conditions; provided, that Interconnection Customer is not obligated by the Standard Large Generator Interconnection Agreement to possess black start capability.

Energy Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to connect its Generating Facility to the Transmission Provider's Transmission System to be eligible to deliver the Generating Facility's electric output using the existing firm or nonfirm capacity of the Transmission

Provider's Transmission System on an as available basis. Energy Resource Interconnection Service in and of itself does not convey transmission service.

Engineering & Procurement (E&P) Agreement shall mean an agreement that authorizes the Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection in order to advance the implementation of the Interconnection Request.

Environmental Law shall mean Applicable Laws or Regulations relating to pollution or protection of the environment or natural resources.

Federal Power Act shall mean the Federal Power Act, as amended, 16 U.S.C. §§ 791a et seq.

FERC shall mean the Federal Energy Regulatory Commission (Commission) or its successor.

Force Majeure shall mean any act of God, labor disturbance, act of the public enemy, war, insurrection, riot, fire, storm or flood, explosion, breakage or accident to machinery or equipment, any order, regulation or restriction imposed by governmental, military or lawfully established civilian authorities, or any other cause beyond a Party's control. A Force Majeure event does not include acts of negligence or intentional wrongdoing by the Party claiming Force Majeure.

Generating Facility shall mean Interconnection Customer's device for the production and/or storage for later injection of electricity identified in the Interconnection Request, but shall not include the Interconnection Customer's Interconnection Facilities.

Generating Facility Capacity shall mean the net capacity of the Generating Facility and the aggregate net capacity of the Generating Facility where it includes multiple energy production devices.

Good Utility Practice shall mean any of the practices, methods and acts engaged in or approved by a significant portion of the electric industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act to the exclusion of all others, but rather to be acceptable practices, methods, or acts generally accepted in the region.

Governmental Authority shall mean any federal, state, local or other

governmental regulatory or administrative agency, court, commission, department, board, or other governmental subdivision, legislature, rulemaking board, tribunal, or other governmental authority having jurisdiction over the Parties, their respective facilities, or the respective services they provide, and exercising or entitled to exercise any administrative, executive, police, or taxing authority or power; provided, however, that such term does not include Interconnection Customer, Transmission Provider, or any Affiliate thereof.

Hazardous Substances shall mean any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "radioactive substances," "contaminants," "pollutants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, or any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

Initial Synchronization Date shall mean the date upon which the Generating Facility is initially synchronized and upon which Trial Operation begins.

In-Service Date shall mean the date upon which the Interconnection Customer reasonably expects it will be ready to begin use of the Transmission Provider's Interconnection Facilities to obtain back feed power.

Interconnection Customer shall mean any entity, including the Transmission Provider, Transmission Owner or any of the Affiliates or subsidiaries of either, that proposes to interconnect its Generating Facility with the Transmission Provider's Transmission System. Where the Interconnection Customer and the Transmission Provider are the same entity, application of the following provisions set forth herein is not required: provisions governing the posting of security (in Articles 5 and 11), provisions governing taxes and reimbursements for tax liability (in Article 5), provisions governing indemnity, consequential damages and insurance (in Article 18), and provisions governing billing and payment (in Articles 12 and 15) (invoices are not required, but may be generated and tendered for administrative use to aid in the identification and accounting of costs).

Interconnection Customer's Interconnection Facilities shall mean all facilities and equipment, as identified in Appendix A of the Standard Large Generator Interconnection Agreement, that are located between the Generating Facility and the Point of Change of Ownership, including any modification, addition, or upgrades to such facilities and equipment necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Customer's Interconnection Facilities are sole use facilities.

Interconnection Facilities shall mean the Transmission Provider's Interconnection Facilities and the Interconnection Customer's Interconnection Facilities. Collectively, Interconnection Facilities include all facilities and equipment between the Generating Facility and the Point of Interconnection, including any modification, additions or upgrades that are necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Interconnection Facilities Study shall mean a study conducted by the Transmission Provider or a third party consultant for the Interconnection Customer to determine a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades as identified in the Definitive Interconnection System Impact Study), the cost of those facilities, and the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System. The scope of the study is defined in Section 8 of the Standard Large Generator Interconnection Procedures.

Interconnection Facilities Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Facilities Study.

Interconnection Feasibility Study shall mean a preliminary evaluation of the system impact and cost of interconnecting the Generating Facility to the Transmission-Provider's Transmission System, the scope of which is described in Section 6 of the Standard Large Generator Interconnection Procedures.

Interconnection Feasibility Study Agreement shall mean the form of agreement contained in Appendix 2 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Feasibility Study.

Interconnection Request shall mean an Interconnection Customer's request, in the form of Appendix 1 to the Standard Large Generator Interconnection Procedures, in accordance with the Tariff, to interconnect a new Generating Facility, or to increase the capacity of, or make a Material Modification to the operating characteristics of, an existing Generating Facility that is interconnected with the Transmission Provider's Transmission System.

Interconnection Service shall mean the service provided by the Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to the Transmission Provider's Transmission System and enabling it to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Standard Large Generator Interconnection Agreement and, if

applicable, the Transmission Provider's Tariff.

Interconnection Study shall mean any of the following studies: the <u>Informational</u> Interconnection <u>Feasibility</u> Study, the <u>Definitive</u> Interconnection System Impact Study, and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection System Impact Study Agreement shall mean any engineering study that evaluates the impact of the following agreements: the Informational proposed interconnection on the safety and reliability of Transmission Provider's Transmission System and, if applicable, an Affected System. The study shall identify and detail the system impacts that would result if the Generating Facility were interconnected without project modifications or system modifications, focusing on the Adverse System Impacts identified in the Interconnection Feasibility Study Agreement, the Definitive Interconnection System Impact Study Agreement, or the Interconnection Facilities Study Agreement to study potential impacts, including but not limited to those identified in the Scoping Meeting as described in the Standard Large Generator Interconnection Procedures.

Interconnection System Impact Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection System Impact Study.

IRS shall mean the Internal Revenue Service.

Joint Operating Committee shall be a group made up of representatives from Interconnection Customers and the Transmission Provider to coordinate operating and technical considerations of Interconnection Service.

Large Generating Facility shall mean a Generating Facility having a Generating Facility Capacity of more than 20 MW.

Loss shall mean any and all losses relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's performance, or non performance of its obligations under the Standard Large Generator Interconnection Agreement on behalf of the indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the indemnifying Party.

Material Modification shall mean those modifications that have a material impact on the cost or timing of any Interconnection Request with a later <u>or equal</u> queue priority date.

Metering Equipment shall mean all metering equipment installed or to be installed at the Generating Facility pursuant to the Standard Large Generator Interconnection Agreement at the metering points, including but not limited to instrument transformers, MWh-meters, data acquisition equipment, transducers, remote terminal unit, communications equipment, phone lines, and fiber optics.

NERC shall mean the North American Electric Reliability Council or its successor organization.

Network Resource shall mean any designated generating resource owned, purchased, or leased by a Network Customer under the Network Integration Transmission Service Tariff. Network Resources do not include any resource, or any portion thereof, that is committed for sale to third parties or otherwise cannot be called upon to meet the Network Customer's Network Load on a non-interruptible basis.

Network Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to integrate its Large Generating Facility with the Transmission Provider's Transmission System (1) in a manner comparable to that in which the Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an RTO or ISO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service in and of itself does not convey transmission service.

Network Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Transmission System required at or beyond the point at which the Interconnection Facilities connect to the Transmission Provider's Transmission System to accommodate the interconnection of the Large Generating Facility to the Transmission Provider's Transmission System.

Notice of Dispute shall mean a written notice of a dispute or claim that arises out of or in connection with the Standard Large Generator Interconnection Agreement or its performance.

OASIS shall mean the Transmission Provider's Open Access Same-Time
Informational Systemptional Interconnection Study shall mean a sensitivity analysis-based on assumptions specified by the Interconnection Customer in the Optional Interconnection Study Agreement.

Optional Interconnection Study Agreement shall mean the form of agreement contained in Appendix 5 of the Standard Large Generator Interconnection Procedures for conducting the Optional Interconnection Study.

Party or Parties shall mean Transmission Provider, Transmission Owner,

Interconnection Customer or any combination of the above.

Phase ("Phase 1, Phase 2, Phase 3 or "Phase 4") shall mean a distinct part of the Definitive Study Process as described in Sections 7 and 8 of the LGIP.

Point of Change of Ownership shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Customer's Interconnection Facilities connect to the Transmission Provider's Interconnection Facilities.

Point of Interconnection shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Facilities connect to the Transmission Provider's Transmission System.

Provisional Interconnection Service shall mean Interconnection Service provided by Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to Transmission Provider's Transmission System and enabling that Transmission System to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Provisional Large Generator Interconnection Agreement and, if applicable, the Tariff.

Provisional Large Generator Interconnection Agreement shall mean the interconnection agreement for Provisional Interconnection Service established between Transmission Provider and/or the Transmission Owner and the Interconnection Customer. This agreement shall take the form of the Large Generator Interconnection Agreement, modified for provisional purposes. Provisional Large Generator Interconnection Agreements are not eligible for suspension.

Queue shall mean a queue for valid Interconnection Requests for the Definitive Interconnection Study Process.

Queue Position shall mean the order of a valid Interconnection Request, relative to all other pending valid Interconnection Requests, in the Definitive Interconnection

Study Process. The Queue Position that is established based upon the date and time Interconnection Customer satisfies all of receipt of the requirements of the LGIP to enter the Definitive Interconnection Study Processvalid Interconnection Request by the Transmission Provider. Valid Interconnection Requests in a single Cluster are considered equal-queued.

Readiness Milestone(s) shall have the meaning set forth in the LGIP.

Reasonable Efforts shall mean, with respect to an action required to be attempted or taken by a Party under the Standard Large Generator Interconnection Agreement, efforts that are timely and consistent with Good Utility Practice and are otherwise substantially equivalent to those a Party would use to protect its own interests.

Resource Plan shall mean any process authorized or required by Applicable Laws and Regulations for, *inter alia*, the selection of Generating Facilities interconnected to the Transmission System of Transmission Provider.

Resource Planning Entity shall mean any entity subject to or conducting a Resource Solicitation Process.

Resource Solicitation Cluster shall mean a Cluster Study associated with a Resource Planning Process.

Resource Solicitation Process shall mean any process authorized or required by Applicable Laws and Regulations for the acquisition of Network Resources by an entity interconnected to the Transmission System of Transmission Provider.

Scoping Meeting shall mean the meeting between representatives of the Interconnection Customer and Transmission Provider conducted for the purpose of discussing the proposed Interconnection Request, alternative interconnection options, to exchange information including any transmission data and earlier study evaluations that would be reasonably expected to affectimpact such interconnection options, to analyze such information, and to determine the potential feasible Points of Interconnection.

Site Control shall mean the exclusive land right to develop, construct, operate, and maintain the Generating Facility over the term of expected operation of the Generating Facility. Site Control shall include the right to develop, construct, operate, and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by documentation reasonably demonstrating: (1) ownership of, a leasehold interest in, or a right to develop a site for the purpose of sufficient size to constructing and operate the Generating Facility and associated Interconnection Customer's Interconnection Facilities; (2) an option to purchase or acquire a leasehold interest in a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities for such purpose; or (3) any other documentation that clearly demonstrates the right of the exclusivity or other business relationship between Interconnection Customer to exclusively and the entity having the right to sell, lease or grant Interconnection Customer the right to possess or occupy a site of sufficient size to construct and operate the Generating Facility. Site Control for any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all colocated projects that meet the aforementioned provisions of this Site Control definitionfor such purpose.

Small Generating Facility shall mean a Generating Facility that has a Generating Facility Capacity of no more than 20 MW.

Stand Alone Network Upgrades shall mean Network Upgrades that are not part of an Affected System that an Interconnection Customer may construct without affecting day-to-day operations of the Transmission System during their construction. Both the Transmission Provider and the Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify them in Appendix A to the Standard Large Generator Interconnection Agreement. If the Transmission Provider and Interconnection Customer disagree about whether a particular Network Upgrade is a Stand Alone Network Upgrade, the Transmission Provider must provide the Interconnection Customer a written technical explanation outlining why the Transmission Provider does not consider the Network Upgrade to be a Stand Alone Network Upgrade within 15 days of its determination.

Standard Large Generator Interconnection Agreement (LGIA) shall mean the form of interconnection agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Procedures (LGIP) shall mean the interconnection procedures applicable to an Interconnection Request pertaining to a Large Generating Facility that are included in the Transmission Provider's Tariff.

Surplus Interconnection Service shall mean any unneeded portion of Interconnection Service established in a Large Generator Interconnection Agreement, such that if Surplus Interconnection Service is utilized the total amount of Interconnection Service at the Point of Interconnection would remain the same.

System Protection Facilities shall mean the equipment, including necessary protection signal communications equipment, required to protect (1) the Transmission Provider's Transmission System from faults or other electrical disturbances occurring at the Generating Facility and (2) the Generating Facility from faults or other electrical system disturbances occurring on the Transmission Provider's Transmission System or on other delivery systems or other generating systems to which the Transmission Provider's Transmission System is directly connected.

Tariff shall mean the Transmission Provider's Tariff through which open access transmission service and Interconnection Service are offered, as filed with FERC, and as amended or supplemented from time to time, or any successor tariff.

Transmission Owner shall mean an entity that owns, leases or otherwise

possesses an interest in the portion of the Transmission System at the Point of Interconnection and may be a Party to the Standard Large Generator Interconnection Agreement to the extent necessary.

Transmission Provider shall mean the public utility (or its designated agent) that owns, controls, or operates transmission or distribution facilities used for the transmission of electricity in interstate commerce and provides transmission service under the Tariff. The term Transmission Provider should be read to include the Transmission Owner when the Transmission Owner is separate from the Transmission Provider.

Transmission Provider's Interconnection Facilities shall mean all facilities and equipment owned, controlled or operated by the Transmission Provider from the Point of Change of Ownership to the Point of Interconnection as identified in Appendix A to the Standard Large Generator Interconnection Agreement, including any modifications, additions or upgrades to such facilities and equipment. Transmission Provider's Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Transmission System shall mean the facilities owned, controlled or operated by the Transmission Provider or Transmission Owner that are used to provide transmission service under the Tariff.

Trial Operation shall mean the period during which Interconnection Customer is engaged in on-site test operations and commissioning of the Generating Facility prior to Commercial Operation.

Variable Energy Resource shall mean a device for the production of electricity that is characterized by an energy source that: (1) is renewable; (2) cannot be stored by the facility owner or operator; and (3) has variability that is beyond the control of the facility owner or operator.

Withdrawal Penalty shall have the meaning set forth in Section 3.7 of the LGIP.

Article 2. Effective Date, Term, and Termination

- **2.1 Effective Date**. This LGIA shall become effective upon execution by the Parties subject to acceptance by FERC (if applicable), or if filed unexecuted, upon the date specified by FERC. Transmission Provider shall promptly file this LGIA with FERC upon execution in accordance with Article 3.1, if required.
- **2.2 Term of Agreement**. Subject to the provisions of Article 2.3, this LGIA shall remain in effect for a period of ten (10) years from the Effective Date or such other longer period as Interconnection Customer may request (Term to be

specified in individual agreements) and shall be automatically renewed for each successive one-year period thereafter.

2.3 Termination Procedures.

- 2.3.1 Written Notice. This LGIA may be terminated by Interconnection Customer after giving Transmission Provider ninety (90) Calendar Days advance written notice, or by Transmission Provider notifying FERC after the Generating Facility permanently ceases Commercial Operation. This LGIA shall be terminated by Transmission Provider if the Generating Facility or a portion of the Generating Facility fails to achieve Commercial Operation by the Commercial Operation Date established in accordance with Section 4.4.5 of the LGIP, including any extension provided thereunder.
- **2.3.2 Default**. Either Party may terminate this LGIA in accordance with Article 17.
- **2.3.3.** Notwithstanding Articles 2.3.1 and 2.3.2, no termination shall become effective until the Parties have complied with all Applicable Laws and Regulations applicable to such termination, including the filing with FERC of a notice of termination of this LGIA, which notice has been accepted for filing by FERC.
- Article 2.3 above, each Party shall pay all costs incurred (including any cancellation costs relating to orders or contracts for Interconnection Facilities and equipment) or charges assessed by the other Party, as of the date of the other Party's receipt of such notice of termination, that are the responsibility of the Terminating Party under this LGIA. In the event of termination by a Party, the Parties shall use commercially Reasonable Efforts to mitigate the costs, damages and charges arising as a consequence of termination. Upon termination of this LGIA, unless otherwise ordered or approved by FERC:
 - 2.4.1 With respect to any portion of Transmission Provider's Interconnection Facilities that have not yet been constructed or installed, Transmission Provider shall to the extent possible and with Interconnection Customer's authorization cancel any pending orders of, or return, any materials or equipment for, or contracts for construction of, such facilities; provided that in the event Interconnection Customer elects not to authorize such cancellation, Interconnection Customer shall assume all payment obligations with respect to such materials, equipment, and contracts, and Transmission Provider shall deliver such material and equipment, and, if

necessary, assign such contracts, to Interconnection Customer as soon as practicable, at Interconnection Customer's expense. To the extent that Interconnection Customer has already paid Transmission Provider for any or all such costs of materials or equipment not taken by Interconnection Customer, Transmission Provider shall promptly refund such amounts to Interconnection Customer, less any costs, including penalties incurred by Transmission Provider to cancel any pending orders of or return such materials, equipment, or contracts.

If an Interconnection Customer terminates this LGIA, it shall be responsible for all costs incurred in association with that Interconnection Customer's interconnection, including any cancellation costs relating to orders or contracts for Interconnection Facilities and equipment, and other expenses including any Network Upgrades for which Transmission Provider has incurred expenses and has not been reimbursed by Interconnection Customer.

- **2.4.2** Transmission Provider may, at its option, retain any portion of such materials, equipment, or facilities that Interconnection Customer chooses not to accept delivery of, in which case Transmission Provider shall be responsible for all costs associated with procuring such materials, equipment, or facilities.
- 2.4.3 With respect to any portion of the Interconnection Facilities, and any other facilities already installed or constructed pursuant to the terms of this LGIA, Interconnection Customer shall be responsible for all costs associated with the removal, relocation or other disposition or retirement of such materials, equipment, or facilities.
- 2.4.4 Transmission Provider shall refund the security provided under Section
 10.3 of the LGIP including any accumulated interest, if applicable.

 Notwithstanding the foregoing, prior to remitting such security, plus accumulated interest if applicable, Transmission Provider shall offset against such security, and accumulated interest, any unpaid costs or penalties arising out of this Agreement or the LGIP. Monies due the Interconnection Customer shall be remitted within 90 Calendar Days of termination.
- 2.4.5 In the event a Withdrawal Penalty applies to a termination under this section, the Withdrawal Penalty shall have the meaning of and be calculated pursuant to Section 3.7.1 of the LGIP.
- **2.5 Disconnection**. Upon termination of this LGIA, the Parties will take all

appropriate steps to disconnect the Large Generating Facility from the Transmission System. All costs required to effectuate such disconnection shall be borne by the terminating Party, unless such termination resulted from the non-terminating Party's Default of this LGIA or such non-terminating Party otherwise is responsible for these costs under this LGIA.

2.6 Survival. This LGIA shall continue in effect after termination to the extent necessary to provide for final billings and payments and for costs incurred hereunder, including billings and payments pursuant to this LGIA; to permit the determination and enforcement of liability and indemnification obligations arising from acts or events that occurred while this LGIA was in effect; and to permit each Party to have access to the lands of the other Party pursuant to this LGIA or other applicable agreements, to disconnect, remove or salvage its own facilities and equipment.

Article 3. Regulatory Filings

3.1 Filing. Transmission Provider shall file this LGIA (and any amendment hereto) with the appropriate Governmental Authority, if required. Interconnection Customer may request that any information so provided be subject to the confidentiality provisions of Article 22. If Interconnection Customer has executed this LGIA, or any amendment thereto, Interconnection Customer shall reasonably cooperate with Transmission Provider with respect to such filing and to provide any information reasonably requested by Transmission Provider needed to comply with applicable regulatory requirements.

Article 4. Scope of Service

4.1 Interconnection Product Options. Interconnection Customer has selected the following (checked) type of Interconnection Service:

4.1.1 Energy Resource Interconnection Service

4.1.1.1 The Product. Energy Resource Interconnection Service allows Interconnection Customer to connect the Large Generating Facility to the Transmission System and be eligible to deliver the Large Generating Facility's output using the existing firm or non-firm capacity of the Transmission System on an "as available" basis. To the extent Interconnection Customer wants to receive Energy Resource Interconnection Service, Transmission Provider shall construct facilities identified in Attachment A.

4.1.1.2

Transmission Delivery Service Implications. Under Energy Resource Interconnection Service, Interconnection Customer will be eligible to inject power from the Large Generating Facility into and deliver power across the interconnecting Transmission Provider's Transmission System on an "as available" basis up to the amount of MWs identified in the applicable stability and steady state studies to the extent the upgrades initially required to qualify for Energy Resource Interconnection Service have been constructed. Where eligible to do so (e.g., PJM, ISONE, NYISO), Interconnection Customer may place a bid to sell into the market up to the maximum identified Large Generating Facility output, subject to any conditions specified in the interconnection service approval, and the Large Generating Facility will be dispatched to the extent Interconnection Customer's bid clears. In all other instances, no transmission delivery service from the Large Generating Facility is assured, but Interconnection Customer may obtain Point-to-Point Transmission Service, Network Integration Transmission Service, or be used for secondary network transmission service, pursuant to Transmission Provider's Tariff, up to the maximum output identified in the stability and steady state studies. In those instances, in order for Interconnection Customer to obtain the right to deliver or inject energy beyond the Large Generating Facility Point of Interconnection or to improve its ability to do so, transmission delivery service must be obtained pursuant to the provisions of Transmission Provider's Tariff. The Interconnection Customer's ability to inject its Large Generating Facility output beyond the Point of Interconnection, therefore, will depend on the existing capacity of Transmission Provider's Transmission System at such time as a transmission service request is made that would accommodate such delivery. The provision of firm Point-to-Point Transmission Service or Network Integration Transmission Service may require the construction of additional Network Upgrades.

4.1.2 Network Resource Interconnection Service.

4.1.2.1 The Product. Transmission Provider must conduct the necessary studies and construct the Network Upgrades needed to integrate the Large Generating Facility (1) in a manner

comparable to that in which Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an ISO or RTO with market based congestion management, in the same manner as all Network Resources. To the extent Interconnection Customer wants to receive Network Resource Interconnection Service, Transmission Provider shall construct the facilities identified in Attachment A to this LGIA.

4.1.2.2 Transmission Delivery Service Implications. Network

Resource Interconnection Service allows Interconnection Customer's Large Generating Facility to be designated by any Network Customer under the Tariff on Transmission Provider's Transmission System as a Network Resource, up to the Large Generating Facility's full output, on the same basis as existing Network Resources interconnected to Transmission Provider's Transmission System, and to be studied as a Network Resource on the assumption that such a designation will occur. Although Network Resource Interconnection Service does not convey a reservation of transmission service, any Network Customer under the Tariff can utilize its network service under the Tariff to obtain delivery of energy from the interconnected Interconnection Customer's Large Generating Facility in the same manner as it accesses Network Resources. A Large Generating Facility receiving Network Resource Interconnection Service may also be used to provide Ancillary Services after technical studies and/or periodic analyses are performed with respect to the Large Generating Facility's ability to provide any applicable Ancillary Services, provided that such studies and analyses have been or would be required in connection with the provision of such Ancillary Services by any existing Network Resource. However, if an Interconnection Customer's Large Generating Facility has not been designated as a Network Resource by any load, it cannot be required to provide Ancillary Services except to the extent such requirements extend to all generating facilities that are similarly situated. The provision of Network Integration Transmission Service or firm Point-to-Point Transmission Service may require additional studies and the construction of additional upgrades. Because such studies and upgrades would be associated with a request for delivery service under the Tariff, cost responsibility for the studies and upgrades

would be in accordance with FERC's policy for pricing transmission delivery services. Network Resource Interconnection Service does not necessarily provide Interconnection Customer with the capability to physically deliver the output of its Large Generating Facility to any particular load on Transmission Provider's Transmission System without incurring congestion costs. In the event of transmission constraints on Transmission Provider's Transmission System, Interconnection Customer's Large Generating Facility shall be subject to the applicable congestion management procedures in Transmission Provider's Transmission System in the same manner as Network Resources.

There is no requirement either at the time of study or interconnection, or at any point in the future, that Interconnection Customer's Large Generating Facility be designated as a Network Resource by a Network Service Customer under the Tariff or that Interconnection Customer identify a specific buyer (or sink). To the extent a Network Customer does designate the Large Generating Facility as a Network Resource, it must do so pursuant to Transmission Provider's Tariff.

Once an Interconnection Customer satisfies the requirements for obtaining Network Resource Interconnection Service, any future transmission service request for delivery from the Large Generating Facility within Transmission Provider's Transmission System of any amount of capacity and/or energy, up to the amount initially studied, will not require that any additional studies be performed or that any further upgrades associated with such Large Generating Facility be undertaken, regardless of whether or not such Large Generating Facility is ever designated by a Network Customer as a Network Resource and regardless of changes in ownership of the Large Generating Facility. However, the reduction or elimination of congestion or redispatch costs may require additional studies and the construction of additional upgrades.

To the extent Interconnection Customer enters into an arrangement for long term transmission service for deliveries from the Large Generating Facility outside Transmission

Provider's Transmission System, such request may require additional studies and upgrades in order for Transmission Provider to grant such request.

- **4.2 Provision of Service**. Transmission Provider shall provide Interconnection Service for the Large Generating Facility at the Point of Interconnection.
- 4.3 Performance Standards. Each Party shall perform all of its obligations under this LGIA in accordance with Applicable Laws and Regulations, Applicable Reliability Standards, and Good Utility Practice, and to the extent a Party is required or prevented or limited in taking any action by such regulations and standards, such Party shall not be deemed to be in Breach of this LGIA for its compliance therewith. If such Party is a Transmission Provider or Transmission Owner, then that Party shall amend the LGIA and submit the amendment to FERC for approval.
- **4.4 No Transmission Delivery Service**. The execution of this LGIA does not constitute a request for, nor the provision of, any transmission delivery service under Transmission Provider's Tariff, and does not convey any right to deliver electricity to any specific customer or Point of Delivery.
- **4.5 Interconnection Customer Provided Services.** The services provided by Interconnection Customer under this LGIA are set forth in Article 9.6 and Article 13.5.1. Interconnection Customer shall be paid for such services in accordance with Article 11.6.

Article 5. Interconnection Facilities Engineering, Procurement, and Construction

5.1 **Options**. Unless otherwise mutually agreed to between the Parties, Interconnection Customer shall select the In-Service Date, Initial Synchronization Date, and Commercial Operation Date; and either Standard Option or Alternate Option set forth below for completion of Transmission Provider's Interconnection Facilities and Network Upgrades as set forth in Appendix A, and such dates and selected option shall be set forth in Appendix B, Milestones. At the same time, Interconnection Customer shall indicate whether it elects to exercise the Option to Build set forth in Article 5.1.3 below. If the dates designated by Interconnection Customer are not acceptable to Transmission Provider, Transmission Provider shall so notify Interconnection Customer within thirty (30) Calendar Days. Upon receipt of the notification that Interconnection Customer's designated dates are not acceptable to Transmission Provider, the Interconnection Customer shall notify Transmission Provider within thirty (30) Calendar Days whether it elects to exercise the Option to Build if it has not already elected to exercise the Option to Build.

- 5.1.1 Standard Option. Transmission Provider shall design, procure, and construct Transmission Provider's Interconnection Facilities and Network Upgrades, using Reasonable Efforts to complete Transmission Provider's Interconnection Facilities and Network Upgrades by the dates set forth in Appendix B, Milestones. Transmission Provider shall not be required to undertake any action which is inconsistent with its standard safety practices, its material and equipment specifications, its design criteria and construction procedures, its labor agreements, and Applicable Laws and Regulations. In the event Transmission Provider reasonably expects that it will not be able to complete Transmission Provider's Interconnection Facilities and Network Upgrades by the specified dates, Transmission Provider shall promptly provide written notice to Interconnection Customer and shall undertake Reasonable Efforts to meet the earliest dates thereafter.
- **5.1.2 Alternate Option**. If the dates designated by Interconnection Customer are acceptable to Transmission Provider, Transmission Provider shall so notify Interconnection Customer within thirty (30) Calendar Days, and shall assume responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities by the designated dates.

If Transmission Provider subsequently fails to complete Transmission Provider's Interconnection Facilities by the In-Service Date, to the extent necessary to provide back feed power; or fails to complete Network Upgrades by the Initial Synchronization Date to the extent necessary to allow for Trial Operation at full power output, unless other arrangements are made by the Parties for such Trial Operation; or fails to complete the Network Upgrades by the Commercial Operation Date, as such dates are reflected in Appendix B, Milestones; Transmission Provider shall pay Interconnection Customer liquidated damages in accordance with Article 5.3, Liquidated Damages, provided, however, the dates designated by Interconnection Customer shall be extended day for day for each day that the applicable RTO or ISO refuses to grant clearances to install equipment.

5.1.3 Option to Build. Interconnection Customer shall have the option to assume responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades on the dates specified in Article 5.1.2. Transmission Provider and Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify such Stand Alone Network Upgrades in Appendix A. Except for Stand Alone Network Upgrades, Interconnection Customer shall have no right to construct Network Upgrades under this option.

- 5.1.4 Negotiated Option. If the dates designated by Interconnection Customer are not acceptable to Transmission Provider, the Parties shall in good faith attempt to negotiate terms and conditions (including revision of the specified dates and liquidated damages, the provision of incentives, or the procurement and construction of all facilities other than Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades if the Interconnection Customer elects to exercise the Option to Build under Article 5.1.3). If the Parties are unable to reach agreement on such terms and conditions, then, pursuant to article 5.1.1 (Standard Option), Transmission Provider shall assume responsibility for the design, procurement and construction of all facilities other than Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades if the Interconnection Customer elects to exercise the Option to Build.
- **5.2 General Conditions Applicable to Option to Build**. If Interconnection Customer assumes responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades,
 - (1) Interconnection Customer shall engineer, procure equipment, and construct Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades (or portions thereof) using Good Utility Practice and using standards and specifications provided in advance by Transmission Provider;
 - (2) Interconnection Customer's engineering, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades shall comply with all requirements of law to which Transmission Provider would be subject in the engineering, procurement or construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
 - (3) Transmission Provider shall review and approve the engineering design, equipment acceptance tests, and the construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
 - (4) prior to commencement of construction, Interconnection Customer shall provide to Transmission Provider a schedule for construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades, and shall promptly respond to requests for information from Transmission Provider;

- (5) at any time during construction, Transmission Provider shall have the right to gain unrestricted access to Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades and to conduct inspections of the same;
- (6) at any time during construction, should any phase of the engineering, equipment procurement, or construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades not meet the standards and specifications provided by Transmission Provider, Interconnection Customer shall be obligated to remedy deficiencies in that portion of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
- (7) Interconnection Customer shall indemnify Transmission Provider for claims arising from Interconnection Customer's construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades under the terms and procedures applicable to Article 18.1 Indemnity;
- (8) Interconnection Customer shall transfer control of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades to Transmission Provider;
- (9) Unless Parties otherwise agree, Interconnection Customer shall transfer ownership of Transmission Provider's Interconnection Facilities and Stand-Alone Network Upgrades to Transmission Provider;
- (10) Transmission Provider shall approve and accept for operation and maintenance Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades to the extent engineered, procured, and constructed in accordance with this Article 5.2; and
- (11) Interconnection Customer shall deliver to Transmission Provider "asbuilt" drawings, information, and any other documents that are reasonably required by Transmission Provider to assure that the Interconnection Facilities and Stand Alone Network Upgrades are built to the standards and specifications required by Transmission Provider.
- (12) If Interconnection Customer exercises the Option to Build pursuant to Article 5.1.3, Interconnection Customer shall pay Transmission Provider an agreed upon amount of [\$ PLACEHOLDER] for Transmission Provider to execute the responsibilities enumerated to Transmission Provider under Article 5.2. Transmission Provider shall invoice Interconnection Customer

for this total amount to be divided on a monthly basis pursuant to Article 12.

5.3 Liquidated Damages. The actual damages to Interconnection Customer, in the event Transmission Provider's Interconnection Facilities or Network Upgrades are not completed by the dates designated by Interconnection Customer and accepted by Transmission Provider pursuant to subparagraphs 5.1.2 or 5.1.4, above, may include Interconnection Customer's fixed operation and maintenance costs and lost opportunity costs. Such actual damages are uncertain and impossible to determine at this time. Because of such uncertainty, any liquidated damages paid by Transmission Provider to Interconnection Customer in the event that Transmission Provider does not complete any portion of Transmission Provider's Interconnection Facilities or Network Upgrades by the applicable dates, shall be an amount equal to ½ of 1 percent per day of the actual cost of Transmission Provider's Interconnection Facilities and Network Upgrades, in the aggregate, for which Transmission Provider has assumed responsibility to design, procure and construct.

However, in no event shall the total liquidated damages exceed 20 percent of the actual cost of Transmission Provider's Interconnection Facilities and Network Upgrades for which Transmission Provider has assumed responsibility to design, procure, and construct. The foregoing payments will be made by Transmission Provider to Interconnection Customer as just compensation for the damages caused to Interconnection Customer, which actual damages are uncertain and impossible to determine at this time, and as reasonable liquidated damages, but not as a penalty or a method to secure performance of this LGIA. Liquidated damages, when the Parties agree to them, are the exclusive remedy for the Transmission Provider's failure to meet its schedule.

No liquidated damages shall be paid to Interconnection Customer if: (1) Interconnection Customer is not ready to commence use of Transmission Provider's Interconnection Facilities or Network Upgrades to take the delivery of power for the Large Generating Facility's Trial Operation or to export power from the Large Generating Facility on the specified dates, unless Interconnection Customer would have been able to commence use of Transmission Provider's Interconnection Facilities or Network Upgrades to take the delivery of power for Large Generating Facility's Trial Operation or to export power from the Large Generating Facility, but for Transmission Provider's delay; (2) Transmission Provider's failure to meet the specified dates is the result of the action or inaction of Interconnection Customer or any other Interconnection Customer who has entered into an LGIA with Transmission Provider or any cause beyond Transmission Provider's reasonable control or reasonable ability to cure; (3) the Interconnection Customer has assumed responsibility for the design, procurement

- and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades; or (4) the Parties have otherwise agreed.
- 5.4 Power System Stabilizers. The Interconnection Customer shall procure, install, maintain and operate Power System Stabilizers in accordance with the guidelines and procedures established by the Applicable Reliability Council. Transmission Provider reserves the right to reasonably establish minimum acceptable settings for any installed Power System Stabilizers, subject to the design and operating limitations of the Large Generating Facility. If the Large Generating Facility's Power System Stabilizers are removed from service or not capable of automatic operation, Interconnection Customer shall immediately notify Transmission Provider's system operator, or its designated representative. The requirements of this paragraph shall not apply to wind generators.
- 5.5 Equipment Procurement. If responsibility for construction of Transmission Provider's Interconnection Facilities or Network Upgrades is to be borne by Transmission Provider, then Transmission Provider shall commence design of Transmission Provider's Interconnection Facilities or Network Upgrades and procure necessary equipment as soon as practicable after all of the following conditions are satisfied, unless the Parties otherwise agree in writing:
 - **5.5.1** Transmission Provider has completed the <u>Interconnection</u> Facilities Study pursuant to the <u>Interconnection</u> Facilities Study Agreement;
 - **5.5.2** Transmission Provider has received written authorization to proceed with design and procurement from Interconnection Customer by the date specified in Appendix B, Milestones; and
 - **5.5.3** Interconnection Customer has provided security to Transmission Provider in accordance with Article 11.5 by the dates specified in Appendix B, Milestones.
- **Construction Commencement**. Transmission Provider shall commence construction of Transmission Provider's Interconnection Facilities and Network Upgrades for which it is responsible as soon as practicable after the following additional conditions are satisfied:
 - **5.6.1** Approval of the appropriate Governmental Authority has been obtained for any facilities requiring regulatory approval;
 - **5.6.2** Necessary real property rights and rights-of-way have been obtained, to the extent required for the construction of a discrete aspect of Transmission Provider's Interconnection Facilities and Network Upgrades;

- **5.6.3** Transmission Provider has received written authorization to proceed with construction from Interconnection Customer by the date specified in Appendix B, Milestones; and
- **5.6.4** Interconnection Customer has provided security to Transmission Provider in accordance with Article 11.5 by the dates specified in Appendix B, Milestones.
- 5.7 Work Progress. The Parties will keep each other advised periodically as to the progress of their respective design, procurement and construction efforts. Either Party may, at any time, request a progress report from the other Party. If, at any time, Interconnection Customer determines that the completion of Transmission Provider's Interconnection Facilities will not be required until after the specified In-Service Date, Interconnection Customer will provide written notice to Transmission Provider of such later date upon which the completion of Transmission Provider's Interconnection Facilities will be required.
- 5.8 Information Exchange. As soon as reasonably practicable after the Effective Date, the Parties shall exchange information regarding the design and compatibility of the Parties' Interconnection Facilities and compatibility of the Interconnection Facilities with Transmission Provider's Transmission System, and shall work diligently and in good faith to make any necessary design changes.

5.9 Other Interconnection Options

- Facilities or Network Upgrades are not reasonably expected to be completed prior to the Commercial Operation Date of the Large Generating Facility, Transmission Provider shall, upon the request and at the expense of Interconnection Customer, perform operating studies on a timely basis to determine the extent to which the Large Generating Facility and Interconnection Customer's Interconnection Facilities may operate prior to the completion of Transmission Provider's Interconnection Facilities or Network Upgrades consistent with Applicable Laws and Regulations, Applicable Reliability Standards, Good Utility Practice, and this LGIA. Transmission Provider shall permit Interconnection Customer to operate the Large Generating Facility and Interconnection Customer's Interconnection Facilities in accordance with the results of such studies.
- **5.9.2 Provisional Interconnection Service**. Upon the request of Interconnection Customer, and prior to completion of requisite Interconnection Facilities, Network Upgrades, Distribution

Upgrades, or System Protection Facilities Transmission Provider may execute a Provisional Large Generator Interconnection Agreement or Interconnection Customer may request the filing of an unexecuted Provisional Large Generator Interconnection Agreement with the Interconnection Customer for limited Interconnection Service at the discretion of Transmission Provider based upon an evaluation that will consider the results of available studies. Transmission Provider shall determine, through available studies or additional studies as necessary, whether stability, short circuit, thermal, and/or voltage issues would arise if Interconnection Customer interconnects without modifications to the Generating Facility or Transmission System. Transmission Provider shall determine whether any Interconnection Facilities, Network Upgrades, Distribution Upgrades, or System Protection Facilities that are necessary to meet the requirements of NERC, or any applicable Regional Entity for the interconnection of a new, modified and/or expanded Generating Facility are in place prior to the commencement of Interconnection Service from the Generating Facility. Where available studies indicate that such, Interconnection Facilities, Network Upgrades, Distribution Upgrades, and/or System Protection Facilities that are required for the interconnection of a new, modified and/or expanded Generating Facility are not currently in place, Transmission Provider will perform a study, at the Interconnection Customer's expense, to confirm the facilities that are required for Provisional Interconnection Service. The maximum permissible output of the Generating Facility in the Provisional Large Generator Interconnection Agreement shall be studied and updated on an annual frequency at the Interconnection Customer's expense. Interconnection Customer assumes all risk and liabilities with respect to changes between the Provisional Large Generator Interconnection Agreement and the Large Generator Interconnection Agreement, including changes in output limits and Interconnection Facilities, Network Upgrades, Distribution Upgrades, and/or System Protection Facilities cost responsibilities.

5.10 Interconnection Customer's Interconnection Facilities ('ICIF').

Interconnection Customer shall, at its expense, design, procure, construct, own and install the ICIF, as set forth in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades.

5.10.1 Interconnection Customer's Interconnection Facility Specifications. Interconnection Customer shall submit initial specifications for the ICIF, including System Protection Facilities, to Transmission Provider at least

one hundred eighty (180) Calendar Days prior to the Initial Synchronization Date; and final specifications for review and comment at least ninety (90) Calendar Days prior to the Initial Synchronization Date. Transmission Provider shall review such specifications to ensure that the ICIF are compatible with the technical specifications, operational control, and safety requirements of Transmission Provider and comment on such specifications within thirty (30) Calendar Days of Interconnection Customer's submission. All specifications provided hereunder shall be deemed confidential.

- 5.10.2 Transmission Provider's Review. Transmission Provider's review of Interconnection Customer's final specifications shall not be construed as confirming, endorsing, or providing a warranty as to the design, fitness, safety, durability or reliability of the Large Generating Facility, or the ICIF. Interconnection Customer shall make such changes to the ICIF as may reasonably be required by Transmission Provider, in accordance with Good Utility Practice, to ensure that the ICIF are compatible with the technical specifications, operational control, and safety requirements of Transmission Provider.
- **5.10.3 ICIF Construction**. The ICIF shall be designed and constructed in accordance with Good Utility Practice. Within one hundred twenty (120) Calendar Days after the Commercial Operation Date, unless the Parties agree on another mutually acceptable deadline, Interconnection Customer shall deliver to Transmission Provider "as-built" drawings, information and documents for the ICIF, such as: a one-line diagram, a site plan showing the Large Generating Facility and the ICIF, plan and elevation drawings showing the layout of the ICIF, a relay functional diagram, relaying AC and DC schematic wiring diagrams and relay settings for all facilities associated with Interconnection Customer's step-up transformers, the facilities connecting the Large Generating Facility to the step-up transformers and the ICIF, and the impedances (determined by factory tests) for the associated step-up transformers and the Large Generating Facility. The Interconnection Customer shall provide Transmission Provider specifications for the excitation system, automatic voltage regulator, Large Generating Facility control and protection settings, transformer tap settings, and communications, if applicable.

5.11 Transmission Provider's Interconnection Facilities Construction.

Transmission Provider's Interconnection Facilities shall be designed and constructed in accordance with Good Utility Practice. Upon request, within one hundred twenty (120) Calendar Days after the Commercial Operation Date, unless the Parties agree on another mutually acceptable deadline, Transmission Provider

shall deliver to Interconnection Customer the following "as-built" drawings, information and documents for Transmission Provider's Interconnection Facilities (include appropriate drawings and relay diagrams).

Transmission Provider will obtain control of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades upon completion of such facilities.

- Access Rights. Upon reasonable notice and supervision by a Party, and subject to 5.12 any required or necessary regulatory approvals, a Party ("Granting Party") shall furnish at no cost to the other Party ("Access Party") any rights of use, licenses, rights of way and easements with respect to lands owned or controlled by the Granting Party, its agents (if allowed under the applicable agency agreement), or any Affiliate, that are necessary to enable the Access Party to obtain ingress and egress to construct, operate, maintain, repair, test (or witness testing), inspect, replace or remove facilities and equipment to: (i) interconnect the Large Generating Facility with the Transmission System; (ii) operate and maintain the Large Generating Facility, the Interconnection Facilities and the Transmission System; and (iii) disconnect or remove the Access Party's facilities and equipment upon termination of this LGIA. In exercising such licenses, rights of way and easements, the Access Party shall not unreasonably disrupt or interfere with normal operation of the Granting Party's business and shall adhere to the safety rules and procedures established in advance, as may be changed from time to time, by the Granting Party and provided to the Access Party.
- 5.13 Lands of Other Property Owners. If any part of Transmission Provider or Transmission Owner's Interconnection Facilities and/or Network Upgrades is to be installed on property owned by persons other than Interconnection Customer or Transmission Provider or Transmission Owner, Transmission Provider or Transmission Owner shall at Interconnection Customer's expense use efforts, similar in nature and extent to those that it typically undertakes on its own behalf or on behalf of its Affiliates, including use of its eminent domain authority, and to the extent consistent with state law, to procure from such persons any rights of use, licenses, rights of way and easements that are necessary to construct, operate, maintain, test, inspect, replace or remove Transmission Provider or Transmission Owner's Interconnection Facilities and/or Network Upgrades upon such property.
- 5.14 Permits. Transmission Provider or Transmission Owner and Interconnection Customer shall cooperate with each other in good faith in obtaining all permits, licenses and authorizations that are necessary to accomplish the interconnection in compliance with Applicable Laws and Regulations. With respect to this paragraph, Transmission Provider or Transmission Owner shall provide permitting assistance to Interconnection Customer comparable to that provided to

Transmission Provider's own, or an Affiliate's generation.

- request Transmission Provider to construct, and Transmission Provider shall construct, using Reasonable Efforts to accommodate Interconnection Customer's In-Service Date, all or any portion of any Network Upgrades required for Interconnection Customer to be interconnected to the Transmission System which are included in the Base Case of the Facilities Study for Interconnection Customer, and which also are required to be constructed for another Interconnection Customer, but where such construction is not scheduled to be completed in time to achieve Interconnection Customer's In-Service Date.
- 5.16 **Suspension**. Interconnection Customer reserves the right, upon written notice to Transmission Provider, to suspend at any time all work by Transmission Provider associated with the construction and installation of Transmission Provider's Interconnection Facilities and/or Network Upgrades required under this LGIA with the condition that Transmission System shall be left in a safe and reliable condition in accordance with Good Utility Practice and Transmission Provider's safety and reliability criteria. In such event, Interconnection Customer shall be responsible for all reasonable and necessary costs which Transmission Provider (i) has incurred pursuant to this LGIA prior to the suspension and (ii) incurs in suspending such work, including any costs incurred to perform such work as may be necessary to ensure the safety of persons and property and the integrity of the Transmission System during such suspension and, if applicable, any costs incurred in connection with the cancellation or suspension of material, equipment and labor contracts which Transmission Provider cannot reasonably avoid; provided, however, that prior to canceling or suspending any such material, equipment or labor contract. Transmission Provider shall obtain Interconnection Customer's authorization to do so.

Transmission Provider shall invoice Interconnection Customer for such costs pursuant to Article 12 and shall use due diligence to minimize its costs. In the event Interconnection Customer suspends work by Transmission Provider required under this LGIA pursuant to this Article 5.16, and has not requested Transmission Provider to recommence the work required under this LGIA on or before the expiration of three (3) years following commencement of such suspension, this LGIA shall be deemed terminated. The three-year period shall begin on the date the suspension is requested, or the date of the written notice to Transmission Provider, if no effective date is specified.

5.16.1 Effect of Missed Interconnection Customer Milestones.

If Interconnection Customer fails to provide notice of suspension pursuant to Article 5.15, and Interconnection Customer fails to fulfill

or complete any Interconnection Customer milestone provided in Appendix B, this constitutes a Breach under this LGIA. Depending upon the consequences of the Breach and effectiveness of the cure pursuant to Article 17, Transmission Provider's Appendix B milestones may be revised following consultation with Interconnection Customer, consistent with Reasonable Efforts, and in consideration of all relevant circumstances. Parties shall employ Reasonable Efforts to maintain their remaining respective Appendix B milestones.

5.16.2 Effect of Suspension; Parties' Obligations.

In the event that Interconnection Customer suspends work pursuant to this Article 5.16, the applicable construction duration, timelines and schedules set forth in Appendix B shall be suspended during the period of suspension. Should Interconnection Customer thereafter request that work be recommenced, Appendix A and Appendix B may be revised to account for construction sequencing and modified milestones. If the Commercial Operation Date is extended beyond three (3) cumulative years described in Section 4.4.5 of the LGIP and Article 2.3.1 of this LGIA, such an extension may be considered a Material Modification and result in the termination of the LGIA under Article 2.3.1. Interconnection Customer is required to maintain Site Control while this LGIA is in effect, including during suspension.

5.17 Taxes.

- 5.17.1 Interconnection Customer Payments Not Taxable. The Parties intend that all payments or property transfers made by Interconnection Customer to Transmission Provider for the installation of Transmission Provider's Interconnection Facilities and the Network Upgrades shall be non-taxable, either as contributions to capital, or as an advance, in accordance with the Internal Revenue Code and any applicable state income tax laws and shall not be taxable as contributions in aid of construction or otherwise under the Internal Revenue Code and any applicable state income tax laws.
- 5.17.2 Representations and Covenants. In accordance with IRS Notice 2001-82 and IRS Notice 88-129, Interconnection Customer represents and covenants that (i) ownership of the electricity generated at the Large Generating Facility will pass to another party prior to the transmission of the electricity on the Transmission System, (ii) for income tax purposes, the amount of any payments and the cost of

any property transferred to Transmission Provider for Transmission Provider's Interconnection Facilities will be capitalized by Interconnection Customer as an intangible asset and recovered using the straight-line method over a useful life of twenty (20) years, and (iii) any portion of Transmission Provider's Interconnection Facilities that is a "dual-use intertie," within the meaning of IRS Notice 88-129, is reasonably expected to carry only a de minimis amount of electricity in the direction of the Large Generating Facility. For this purpose, "de minimis amount" means no more than 5 percent of the total power flows in both directions, calculated in accordance with the "5 percent test" set forth in IRS Notice 88-129. This is not intended to be an exclusive list of the relevant conditions that must be met to conform to IRS requirements for non-taxable treatment.

At Transmission Provider's request, Interconnection Customer shall provide Transmission Provider with a report from an independent engineer confirming its representation in clause (iii), above.

Transmission Provider represents and covenants that the cost of Transmission Provider's Interconnection Facilities paid for by Interconnection Customer will have no net effect on the base upon which rates are determined.

5.17.3 Indemnification for the Cost Consequences of Current Tax Liability
Imposed Upon the Transmission Provider. Notwithstanding
Article 5.17.1, Interconnection Customer shall protect, indemnify
and hold harmless Transmission Provider from the cost
consequences of any current tax liability imposed against
Transmission Provider as the result of payments or property transfers
made by Interconnection Customer to Transmission Provider under
this LGIA for Interconnection Facilities, as well as any interest and
penalties, other than interest and penalties attributable to any delay
caused by Transmission Provider.

Transmission Provider shall not include a gross-up for the cost consequences of any current tax liability in the amounts it charges Interconnection Customer under this LGIA unless (i) Transmission Provider has determined, in good faith, that the payments or property transfers made by Interconnection Customer to Transmission Provider should be reported as income subject to taxation or (ii) any Governmental Authority directs Transmission Provider to report payments or property as income subject to taxation; provided, however, that Transmission Provider may require Interconnection

Customer to provide security for Interconnection Facilities, in a form reasonably acceptable to Transmission Provider (such as a parental guarantee or a letter of credit), in an amount equal to the cost consequences of any current tax liability under this Article 5.17. Interconnection Customer shall reimburse Transmission Provider for such costs on a fully grossed-up basis, in accordance with Article 5.17.4, within thirty (30) Calendar Days of receiving written notification from Transmission Provider of the amount due, including detail about how the amount was calculated.

The indemnification obligation shall terminate at the earlier of (1) the expiration of the ten year testing period and the applicable statute of limitation, as it may be extended by Transmission Provider upon request of the IRS, to keep these years open for audit or adjustment, or (2) the occurrence of a subsequent taxable event and the payment of any related indemnification obligations as contemplated by this Article 5.17.

5.17.4 Tax Gross-Up Amount. Interconnection Customer's liability for the cost consequences of any current tax liability under this Article 5.17 shall be calculated on a fully grossed-up basis. Except as may otherwise be agreed to by the parties, this means that Interconnection Customer will pay Transmission Provider, in addition to the amount paid for the Interconnection Facilities and Network Upgrades, an amount equal to (1) the current taxes imposed on Transmission Provider ("Current Taxes") on the excess of (a) the gross income realized by Transmission Provider as a result of payments or property transfers made by Interconnection Customer to Transmission Provider under this LGIA (without regard to any payments under this Article 5.17) (the "Gross Income Amount") over (b) the present value of future tax deductions for depreciation that will be available as a result of such payments or property transfers (the "Present Value Depreciation Amount"), plus (2) an additional amount sufficient to permit Transmission Provider to receive and retain, after the payment of all Current Taxes, an amount equal to the net amount described in clause (1).

For this purpose, (i) Current Taxes shall be computed based on Transmission Provider's composite federal and state tax rates at the time the payments or property transfers are received and Transmission Provider will be treated as being subject to tax at the highest marginal rates in effect at that time (the "Current Tax Rate"), and (ii) the Present Value Depreciation Amount shall be computed

by discounting Transmission Provider's anticipated tax depreciation deductions as a result of such payments or property transfers by Transmission Provider's current weighted average cost of capital. Thus, the formula for calculating Interconnection Customer's liability to Transmission Owner pursuant to this Article 5.17.4 can be expressed as follows: (Current Tax Rate x (Gross Income Amount - Present Value of Tax Depreciation))/(1Current Tax Rate). Interconnection Customer's estimated tax liability in the event taxes are imposed shall be stated in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades.

5.17.5 Private Letter Ruling or Change or Clarification of Law. At

Interconnection Customer's request and expense, Transmission Provider shall file with the IRS a request for a private letter ruling as to whether any property transferred or sums paid, or to be paid, by Interconnection Customer to Transmission Provider under this LGIA are subject to federal income taxation. Interconnection Customer will prepare the initial draft of the request for a private letter ruling, and will certify under penalties of perjury that all facts represented in such request are true and accurate to the best of Interconnection Customer's knowledge. Transmission Provider and Interconnection Customer shall cooperate in good faith with respect to the submission of such request.

Transmission Provider shall keep Interconnection Customer fully informed of the status of such request for a private letter ruling and shall execute either a privacy act waiver or a limited power of attorney, in a form acceptable to the IRS, that authorizes Interconnection Customer to participate in all discussions with the IRS regarding such request for a private letter ruling. Transmission Provider shall allow Interconnection Customer to attend all meetings with IRS officials about the request and shall permit Interconnection Customer to prepare the initial drafts of any follow-up letters in connection with the request.

5.17.6 Subsequent Taxable Events. If, within 10 years from the date on which the relevant Transmission Provider's Interconnection Facilities are placed in service, (i) Interconnection Customer Breaches the covenants contained in Article 5.17.2, (ii) a "disqualification event" occurs within the meaning of IRS Notice 88-129, or (iii) this LGIA terminates and Transmission Provider retains ownership of the Interconnection Facilities and Network Upgrades, Interconnection Customer shall pay a tax gross-up for the cost consequences of any

current tax liability imposed on Transmission Provider, calculated using the methodology described in Article 5.17.4 and in accordance with IRS Notice 90-60.

5.17.7 Contests. In the event any Governmental Authority determines that Transmission Provider's receipt of payments or property constitutes

income that is subject to taxation, Transmission Provider shall notify Interconnection Customer, in writing, within thirty (30) Calendar Days of receiving notification of such determination by a Governmental Authority. Upon the timely written request by Interconnection Customer and at Interconnection Customer's sole expense, Transmission Provider may appeal, protest, seek abatement of, or otherwise oppose such determination. Upon Interconnection Customer's written request and sole expense, Transmission Provider may file a claim for refund with respect to any taxes paid under this Article 5.17, whether or not it has received such a determination. Transmission Provider reserves the right to make all decisions with regard to the prosecution of such appeal, protest, abatement or other contest, including the selection of counsel and compromise or settlement of the claim, but Transmission Provider shall keep Interconnection Customer informed, shall consider in good faith suggestions from Interconnection Customer about the conduct of the contest, and shall reasonably permit Interconnection Customer or an Interconnection Customer representative to attend contest proceedings.

Interconnection Customer shall pay to Transmission Provider on a periodic basis, as invoiced by Transmission Provider, Transmission Provider's documented reasonable costs of prosecuting such appeal, protest, abatement or other contest. At any time during the contest, Transmission Provider may agree to a settlement either with Interconnection Customer's consent or after obtaining written advice from nationally-recognized tax counsel, selected by Transmission Provider, but reasonably acceptable to Interconnection Customer, that the proposed settlement represents a reasonable settlement given the hazards of litigation.

Interconnection Customer's obligation shall be based on the amount of the settlement agreed to by Interconnection Customer, or if a higher amount, so much of the settlement that is supported by the written advice from nationally-recognized tax counsel selected under the terms of the preceding sentence. The settlement amount shall be calculated on a fully grossed-up bases to cover any related cost

consequences of the current tax liability. Any settlement without Interconnection Customer's consent or such written advice will relieve Interconnection Customer from any obligation to indemnify Transmission Provider for the tax at issue in the contest.

- **5.17.8 Refund.** In the event that (a) a private letter ruling is issued to Transmission Provider which holds that any amount paid or the value of any property transferred by Interconnection Customer to Transmission Provider under the terms of this LGIA is not subject to federal income taxation, (b) any legislative change or administrative announcement, notice, ruling or other determination makes it reasonably clear to Transmission Provider in good faith that any amount paid or the value of any property transferred by Interconnection Customer to Transmission Provider under the terms of this LGIA is not taxable to Transmission Provider, (c) any abatement, appeal, protest, or other contest results in a determination that any payments or transfers made by Interconnection Customer to Transmission Provider are not subject to federal income tax, or (d) if Transmission Provider receives a refund from any taxing authority for any overpayment of tax attributable to any payment or property transfer made by Interconnection Customer to Transmission Provider pursuant to this LGIA, Transmission Provider shall promptly refund to Interconnection Customer the following:
 - (i) any payment made by Interconnection Customer under this Article 5.17 for taxes that is attributable to the amount determined to be non-taxable, together with interest thereon,
 - (ii) interest on any amounts paid by Interconnection Customer to Transmission Provider for such taxes which Transmission Provider did not submit to the taxing authority, calculated in accordance with the methodology set forth in FERC's regulations at 18 CFR §35.19a(a)(2)(iii) from the date payment was made by Interconnection Customer to the date Transmission Provider refunds such payment to Interconnection Customer, and
 - (iii) with respect to any such taxes paid by Transmission Provider, any refund or credit Transmission Provider receives or to which it may be entitled from any Governmental Authority, interest (or that portion thereof attributable to the payment described in clause (i), above) owed to Transmission Provider for such overpayment of taxes (including any

reduction in interest otherwise payable by Transmission Provider to any Governmental Authority resulting from an offset or credit); provided, however, that Transmission Provider will remit such amount promptly to Interconnection Customer only after and to the extent that Transmission Provider has received a tax refund, credit or offset from any Governmental Authority for any applicable overpayment of income tax related to Transmission Provider's Interconnection Facilities.

The intent of this provision is to leave the Parties, to the extent practicable, in the event that no taxes are due with respect to any payment for Interconnection Facilities and Network Upgrades hereunder, in the same position they would have been in had no such tax payments been made.

5.17.9 Taxes Other Than Income Taxes. Upon the timely request by

Interconnection Customer, and at Interconnection Customer's sole expense, Transmission Provider may appeal, protest, seek abatement of, or otherwise contest any tax (other than federal or state income tax) asserted or assessed against Transmission Provider for which Interconnection Customer may be required to reimburse Transmission Provider under the terms of this LGIA. Interconnection Customer shall pay to Transmission Provider on a periodic basis, as invoiced by Transmission Provider, Transmission Provider's documented reasonable costs of prosecuting such appeal, protest, abatement, or other contest. Interconnection Customer and Transmission Provider shall cooperate in good faith with respect to any such contest. Unless the payment of such taxes is a prerequisite to an appeal or abatement or cannot be deferred, no amount shall be payable by Interconnection Customer to Transmission Provider for such taxes until they are assessed by a final, non-appealable order by any court or agency of competent jurisdiction. In the event that a tax payment is withheld and ultimately due and payable after appeal, Interconnection Customer will be responsible for all taxes, interest and penalties, other than penalties attributable to any delay caused by Transmission Provider.

Transmission Owners Who Are Not Transmission Providers. If Transmission Provider is not the same entity as the Transmission Owner, then (i) all references in this Article 5.17 to Transmission Provider shall be deemed also to refer to and to include the Transmission Owner, as appropriate, and (ii) this LGIA shall not

become effective until such Transmission Owner shall have agreed in writing to assume all of the duties and obligations of Transmission Provider under this Article 5.17 of this LGIA.

5.18 Tax Status. Each Party shall cooperate with the other to maintain the other Party's tax status. Nothing in this LGIA is intended to adversely affect any Transmission Provider's tax exempt status with respect to the issuance of bonds including, but not limited to, Local Furnishing Bonds.

5.19 Modification.

5.19.1 General. Either Party may undertake modifications to its facilities. If a Party plans to undertake a modification that reasonably may be expected to affect the other Party's facilities, that Party shall provide to the other Party sufficient information regarding such modification so that the other Party may evaluate the potential impact of such modification prior to commencement of the work. Such information shall be deemed to be confidential hereunder and shall include information concerning the timing of such modifications and whether such modifications are expected to interrupt the flow of electricity from the Large Generating Facility. The Party desiring to perform such work shall provide the relevant drawings, plans, and specifications to the other Party at least ninety (90) Calendar Days in advance of the commencement of the work or such shorter period upon which the Parties may agree, which agreement shall not unreasonably be withheld, conditioned or delayed.

In the case of Large Generating Facility modifications that do not require Interconnection Customer to submit an Interconnection Request, Transmission Provider shall provide, within thirty (30) Calendar Days (or such other time as the Parties may agree), an estimate of any additional modifications to the Transmission System, Transmission Provider's Interconnection Facilities or Network Upgrades necessitated by such Interconnection Customer modification and a good faith estimate of the costs thereof.

- **5.19.2 Standards**. Any additions, modifications, or replacements made to a Party's facilities shall be designed, constructed and operated in accordance with this LGIA and Good Utility Practice.
- **5.19.3 Modification Costs**. Interconnection Customer shall not be directly assigned for the costs of any additions, modifications, or replacements that Transmission Provider makes to Transmission

Provider's Interconnection Facilities or the Transmission System to facilitate the interconnection of a third party to Transmission Provider's Interconnection Facilities or the Transmission System, or to provide transmission service to a third party under Transmission Provider's Tariff. Interconnection Customer shall be responsible for the costs of any additions, modifications, or replacements to Interconnection Customer's Interconnection Facilities that may be necessary to maintain or upgrade such Interconnection Customer's Interconnection Facilities consistent with Applicable Laws and Regulations, Applicable Reliability Standards or Good Utility Practice.

Article 6. Testing and Inspection

- 6.1 Pre-Commercial Operation Date Testing and Modifications. Prior to the Commercial Operation Date, Transmission Provider shall test Transmission Provider's Interconnection Facilities and Network Upgrades and Interconnection Customer shall test the Large Generating Facility and Interconnection Customer's Interconnection Facilities to ensure their safe and reliable operation. Similar testing may be required after initial operation. Each Party shall make any modifications to its facilities that are found to be necessary as a result of such testing. Interconnection Customer shall bear the cost of all such testing and modifications. Interconnection Customer shall generate test energy at the Large Generating Facility only if it has arranged for the delivery of such test energy.
- 6.2 Post-Commercial Operation Date Testing and Modifications. Each Party shall at its own expense perform routine inspection and testing of its facilities and equipment in accordance with Good Utility Practice as may be necessary to ensure the continued interconnection of the Large Generating Facility with the Transmission System in a safe and reliable manner. Each Party shall have the right, upon advance written notice, to require reasonable additional testing of the other Party's facilities, at the requesting Party's expense, as may be in accordance with Good Utility Practice.
- **Right to Observe Testing**. Each Party shall notify the other Party in advance of its performance of tests of its Interconnection Facilities. The other Party has the right, at its own expense, to observe such testing.
- 6.4 Right to Inspect. Each Party shall have the right, but shall have no obligation to:
 (i) observe the other Party's tests and/or inspection of any of its System Protection Facilities and other protective equipment, including Power System Stabilizers; (ii) review the settings of the other Party's System Protection Facilities and other protective equipment; and (iii) review the other Party's maintenance records

relative to the Interconnection Facilities, the System Protection Facilities and other protective equipment. A Party may exercise these rights from time to time as it deems necessary upon reasonable notice to the other Party. The exercise or non-exercise by a Party of any such rights shall not be construed as an endorsement or confirmation of any element or condition of the Interconnection Facilities or the System Protection Facilities or other protective equipment or the operation thereof, or as a warranty as to the fitness, safety, desirability, or reliability of same. Any information that a Party obtains through the exercise of any of its rights under this Article 6.4 shall be deemed to be Confidential Information and treated pursuant to Article 22 of this LGIA.

Article 7. Metering

- 7.1 General. Each Party shall comply with the Applicable Reliability Council requirements. Unless otherwise agreed by the Parties, Transmission Provider shall install Metering Equipment at the Point of Interconnection prior to any operation of the Large Generating Facility and shall own, operate, test and maintain such Metering Equipment. Power flows to and from the Large Generating Facility shall be measured at or, at Transmission Provider's option, compensated to, the Point of Interconnection. Transmission Provider shall provide metering quantities, in analog and/or digital form, to Interconnection Customer upon request. Interconnection Customer shall bear all reasonable documented costs associated with the purchase, installation, operation, testing and maintenance of the Metering Equipment.
- 7.2 Check Meters. Interconnection Customer, at its option and expense, may install and operate, on its premises and on its side of the Point of Interconnection, one or more check meters to check Transmission Provider's meters. Such check meters shall be for check purposes only and shall not be used for the measurement of power flows for purposes of this LGIA, except as provided in Article 7.4 below. The check meters shall be subject at all reasonable times to inspection and examination by Transmission Provider or its designee. The installation, operation and maintenance thereof shall be performed entirely by Interconnection Customer in accordance with Good Utility Practice.
- **7.3 Standards**. Transmission Provider shall install, calibrate, and test revenue quality Metering Equipment in accordance with applicable ANSI standards.
- 7.4 Testing of Metering Equipment. Transmission Provider shall inspect and test all Transmission Provider-owned Metering Equipment upon installation and at least once every two (2) years thereafter. If requested to do so by Interconnection Customer, Transmission Provider shall, at Interconnection Customer's expense, inspect or test Metering Equipment more frequently than every two (2) years.

Transmission Provider shall give reasonable notice of the time when any inspection or test shall take place, and Interconnection Customer may have representatives present at the test or inspection. If at any time Metering Equipment is found to be inaccurate or defective, it shall be adjusted, repaired or replaced at Interconnection Customer's expense, in order to provide accurate metering, unless the inaccuracy or defect is due to Transmission Provider's failure to maintain, then Transmission Provider shall pay. If Metering Equipment fails to register, or if the measurement made by Metering Equipment during a test varies by more than two percent from the measurement made by the standard meter used in the test, Transmission Provider shall adjust the measurements by correcting all measurements for the period during which Metering Equipment was in error by using Interconnection Customer's check meters, if installed. If no such check meters are installed or if the period cannot be reasonably ascertained, the adjustment shall be for the period immediately preceding the test of the Metering Equipment equal to one-half the time from the date of the last previous test of the Metering Equipment.

7.5 Metering Data. At Interconnection Customer's expense, the metered data shall be telemetered to one or more locations designated by Transmission Provider and one or more locations designated by Interconnection Customer. Such telemetered data shall be used, under normal operating conditions, as the official measurement of the amount of energy delivered from the Large Generating Facility to the Point of Interconnection.

Article 8. Communications

Interconnection Customer Obligations. Interconnection Customer shall 8.1 maintain satisfactory operating communications with Transmission Provider's Transmission System dispatcher or representative designated by Transmission Provider. Interconnection Customer shall provide standard voice line, dedicated voice line and facsimile communications at its Large Generating Facility control room or central dispatch facility through use of either the public telephone system, or a voice communications system that does not rely on the public telephone system. Interconnection Customer shall also provide the dedicated data circuit(s) necessary to provide Interconnection Customer data to Transmission Provider as set forth in Appendix D, Security Arrangements Details. The data circuit(s) shall extend from the Large Generating Facility to the location(s) specified by Transmission Provider. Any required maintenance of such communications equipment shall be performed by Interconnection Customer. Operational communications shall be activated and maintained under, but not be limited to, the following events: system paralleling or separation, scheduled and unscheduled shutdowns, equipment clearances, and hourly and daily load data.

8.2 Remote Terminal Unit. Prior to the Initial Synchronization Date of the Large Generating Facility, a Remote Terminal Unit, or equivalent data collection and transfer equipment acceptable to the Parties, shall be installed by Interconnection Customer, or by Transmission Provider at Interconnection Customer's expense, to gather accumulated and instantaneous data to be telemetered to the location(s) designated by Transmission Provider through use of a dedicated point-to-point data circuit(s) as indicated in Article 8.1. The communication protocol for the data circuit(s) shall be specified by Transmission Provider. Instantaneous bidirectional analog real power and reactive power flow information must be telemetered directly to the location(s) specified by Transmission Provider.

Each Party will promptly advise the other Party if it detects or otherwise learns of any metering, telemetry or communications equipment errors or malfunctions that require the attention and/or correction by the other Party. The Party owning such equipment shall correct such error or malfunction as soon as reasonably feasible.

- **8.3 No Annexation**. Any and all equipment placed on the premises of a Party shall be and remain the property of the Party providing such equipment regardless of the mode and manner of annexation or attachment to real property, unless otherwise mutually agreed by the Parties.
- Provision of Data from a Variable Energy Resource. The Interconnection 8.4 Customer whose Generating Facility is a Variable Energy Resource shall provide meteorological and forced outage data to the Transmission Provider to the extent necessary for the Transmission Provider's development and deployment of power production forecasts for that class of Variable Energy Resources. The Interconnection Customer with a Variable Energy Resource having wind as the energy source, at a minimum, will be required to provide the Transmission Provider with site-specific meteorological data including: temperature, wind speed, wind direction, and atmospheric pressure. The Interconnection Customer with a Variable Energy Resource having solar as the energy source, at a minimum, will be required to provide the Transmission Provider with site-specific meteorological data including: temperature, atmospheric pressure, and irradiance. The Transmission Provider and Interconnection Customer whose Generating Facility is a Variable Energy Resource shall mutually agree to any additional meteorological data that are required for the development and deployment of a power production forecast. The Interconnection Customer whose Generating Facility is a Variable Energy Resource also shall submit data to the Transmission Provider regarding all forced outages to the extent necessary for the Transmission Provider's development and deployment of power production forecasts for that class of Variable Energy Resources. The exact specifications of the meteorological and forced outage data to be provided by the Interconnection Customer to the Transmission Provider, including the frequency and timing of

data submittals, shall be made taking into account the size and configuration of the Variable Energy Resource, its characteristics, location, and its importance in maintaining generation resource adequacy and transmission system reliability in its area. All requirements for meteorological and forced outage data must be commensurate with the power production forecasting employed by the Transmission Provider. Such requirements for meteorological and forced outage data are set forth in Appendix C, Interconnection Details, of this LGIA, as they may change from time to time.

Article 9. Operations

- **9.1 General**. Each Party shall comply with the Applicable Reliability Council requirements. Each Party shall provide to the other Party all information that may reasonably be required by the other Party to comply with Applicable Laws and Regulations and Applicable Reliability Standards.
- Date, Interconnection Customer shall notify Transmission Provider in writing of the Control Area in which the Large Generating Facility will be located. If Interconnection Customer elects to locate the Large Generating Facility in a Control Area other than the Control Area in which the Large Generating Facility is physically located, and if permitted to do so by the relevant transmission tariffs, all necessary arrangements, including but not limited to those set forth in Article 7 and Article 8 of this LGIA, and remote Control Area generator interchange agreements, if applicable, and the appropriate measures under such agreements, shall be executed and implemented prior to the placement of the Large Generating Facility in the other Control Area.
- 9.3 Transmission Provider Obligations. Transmission Provider shall cause the Transmission System and Transmission Provider's Interconnection Facilities to be operated, maintained and controlled in a safe and reliable manner and in accordance with this LGIA. Transmission Provider may provide operating instructions to Interconnection Customer consistent with this LGIA and Transmission Provider's operating protocols and procedures as they may change from time to time. Transmission Provider will consider changes to its operating protocols and procedures proposed by Interconnection Customer.
- 9.4 Interconnection Customer Obligations. Interconnection Customer shall at its own expense operate, maintain and control the Large Generating Facility and Interconnection Customer's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA. Interconnection Customer shall operate the Large Generating Facility and Interconnection Customer's Interconnection Facilities in accordance with all applicable requirements of the

Control Area of which it is part, as such requirements are set forth in Appendix C, Interconnection Details, of this LGIA. Appendix C, Interconnection Details, will be modified to reflect changes to the requirements as they may change from time to time. Either Party may request that the other Party provide copies of the requirements set forth in Appendix C, Interconnection Details, of this LGIA.

- **9.5 Start-Up and Synchronization**. Consistent with the Parties' mutually acceptable procedures, Interconnection Customer is responsible for the proper synchronization of the Large Generating Facility to Transmission Provider's Transmission System.
- 9.6 Reactive Power and Primary Frequency Response.
 - 9.6.1 Power Factor Design Criteria.
 - 9.6.1.1 Synchronous Generation. Interconnection Customer shall design the Large Generating Facility to maintain a composite power delivery at continuous rated power output at the Point of Interconnection at a power factor within the range of 0.95 leading to 0.95 lagging, unless the Transmission Provider has established different requirements that apply to all generators in the Control Area on a comparable basis.
 - 9.6.1.2 Non-Synchronous Generation. Interconnection Customer shall design the Large Generating Facility to maintain a composite power delivery at continuous rated power output at the high-side of the generator substation at a power factor within the range of 0.95 leading to 0.95 lagging, unless the Transmission Provider has established a different power factor range that applies to all non-synchronous generators in the Control Area on a comparable basis. This power factor range standard shall be dynamic and can be met using, for example, power electronics designed to supply this level of reactive capability (taking into account any limitations due to voltage level, real power output, etc.) or fixed and switched capacitors, or a combination of the two. This requirement shall only apply to newly interconnecting non-synchronous generators that have not yet executed a Facilities Study Agreement as of the effective date of the Final Rule establishing this requirement (Order No. 827).
 - **9.6.2 Voltage Schedules**. Once Interconnection Customer has synchronized the Large Generating Facility with the Transmission System, Transmission

Provider shall require Interconnection Customer to operate the Large Generating Facility to produce or absorb reactive power within the design limitations of the Large Generating Facility set forth in Article 9.6.1 (Power Factor Design Criteria). Transmission Provider's voltage schedules shall treat all sources of reactive power in the Control Area in an equitable and not unduly discriminatory manner. Transmission Provider shall exercise Reasonable Efforts to provide Interconnection Customer with such schedules at least one (1) day in advance, and may make changes to such schedules as necessary to maintain the reliability of the Transmission System. Interconnection Customer shall operate the Large Generating Facility to maintain the specified output voltage or power factor at the Point of Interconnection within the design limitations of the Large Generating Facility set forth in Article 9.6.1 (Power Factor Design Criteria). If Interconnection Customer is unable to maintain the specified voltage or power factor, it shall promptly notify the System Operator.

- 9.6.2.1 **Voltage Regulators**. Whenever the Large Generating Facility is operated in parallel with the Transmission System and voltage regulators are capable of operation, Interconnection Customer shall operate the Large Generating Facility with its voltage regulators in automatic operation. If the Large Generating Facility's voltage regulators are not capable of such automatic operation, Interconnection Customer shall immediately notify Transmission Provider's system operator, or its designated representative, and ensure that such Large Generating Facility's reactive power production or absorption (measured in MVARs) are within the design capability of the Large Generating Facility's generating unit(s) and steady state stability limits. Interconnection Customer shall not cause its Large Generating Facility to disconnect automatically or instantaneously from the Transmission System or trip any generating unit comprising the Large Generating Facility for an under or over frequency condition unless the abnormal frequency condition persists for a time period beyond the limits set forth in ANSI/IEEE Standard C37.106, or such other standard as applied to other generators in the Control Area on a comparable basis.
- **9.6.3** Payment for Reactive Power. Transmission Provider is required to pay Interconnection Customer for reactive power that Interconnection Customer provides or absorbs from the Large Generating Facility when Transmission Provider requests Interconnection Customer to operate its Large Generating Facility outside the range specified in Article 9.6.1, provided that if

Transmission Provider pays its own or affiliated generators for reactive power service within the specified range, it must also pay Interconnection Customer. Payments shall be pursuant to Article 11.6 or such other agreement to which the Parties have otherwise agreed.

9.6.4 Primary Frequency Response. Interconnection Customer shall ensure the primary frequency response capability of its Large Generating Facility by installing, maintaining, and operating a functioning governor or equivalent controls. The term "functioning governor or equivalent controls" as used herein shall mean the required hardware and/or software that provides frequency responsive real power control with the ability to sense changes in system frequency and autonomously adjust the Large Generating Facility's real power output in accordance with the droop and deadband parameters and in the direction needed to correct frequency deviations. Interconnection Customer is required to install a governor or equivalent controls with the capability of operating: (1) with a maximum 5 percent droop and ± 0.036 Hz deadband; or (2) in accordance with the relevant droop, deadband, and timely and sustained response settings from an approved NERC Reliability Standard providing for equivalent or more stringent parameters. The droop characteristic shall be: (1) based on the nameplate capacity of the Large Generating Facility, and shall be linear in the range of frequencies between 59 to 61 Hz that are outside of the deadband parameter; or (2) based on an approved NERC Reliability Standard providing for an equivalent or more stringent parameter. The deadband parameter shall be: the range of frequencies above and below nominal (60 Hz) in which the governor or equivalent controls is not expected to adjust the Large Generating Facility's real power output in response to frequency deviations. The deadband shall be implemented: (1) without a step to the droop curve, that is, once the frequency deviation exceeds the deadband parameter, the expected change in the Large Generating Facility's real power output in response to frequency deviations shall start from zero and then increase (for underfrequency deviations) or decrease (for over-frequency deviations) linearly in proportion to the magnitude of the frequency deviation; or (2) in accordance with an approved NERC Reliability Standard providing for an equivalent or more stringent parameter. Interconnection Customer shall notify Transmission Provider that the primary frequency response capability of the Large Generating Facility has been tested and confirmed during commissioning. Once Interconnection Customer has synchronized the Large Generating Facility with the Transmission System, Interconnection Customer shall operate the Large Generating Facility consistent with the provisions specified in Sections 9.6.4.1 and 9.6.4.2 of this Agreement. The primary frequency response requirements contained herein shall apply to both synchronous and non-synchronous Large

Generating Facilities.

9.6.4.1 Governor or Equivalent Controls. Whenever the Large Generating Facility is operated in parallel with the Transmission System, Interconnection Customer shall operate the Large Generating Facility with its governor or equivalent controls in service and responsive to frequency. Interconnection Customer shall: (1) in coordination with Transmission Provider and/or the relevant balancing authority, set the deadband parameter to: (1) a maximum of ± 0.036 Hz and set the droop parameter to a maximum of 5 percent; or (2) implement the relevant droop and deadband settings from an approved NERC Reliability Standard that provides for equivalent or more stringent parameters. Interconnection Customer shall be required to provide the status and settings of the governor or equivalent controls to Transmission Provider and/or the relevant balancing authority upon request. If Interconnection Customer needs to operate the Large Generating Facility with its governor or equivalent controls not in service, Interconnection Customer shall immediately notify Transmission Provider and the relevant balancing authority, and provide both with the following information: (1) the operating status of the governor or equivalent controls (i.e., whether it is currently out of service or when it will be taken out of service); (2) the reasons for removing the governor or equivalent controls from service; and (3) a reasonable estimate of when the governor or equivalent controls will be returned to service. Interconnection Customer shall make Reasonable Efforts to return its governor or equivalent controls into service as soon as practicable. Interconnection Customer shall make Reasonable Efforts to keep outages of the Large Generating Facility's governor or equivalent controls to a minimum whenever the Large Generating Facility is operated in parallel with the Transmission System.

9.6.4.2 Timely and Sustained Response. Interconnection Customer shall ensure that the Large Generating Facility's real power response to sustained frequency deviations outside of the deadband setting is automatically provided and shall begin immediately after frequency deviates outside of the deadband, and to the extent the Large Generating Facility has operating capability in the direction needed to correct the frequency

deviation. Interconnection Customer shall not block or otherwise inhibit the ability of the governor or equivalent controls to respond and shall ensure that the response is not inhibited, except under certain operational constraints including, but not limited to, ambient temperature limitations, physical energy limitations, outages of mechanical equipment, or regulatory requirements. The Large Generating Facility shall sustain the real power response at least until system frequency returns to a value within the deadband setting of the governor or equivalent controls. A Commission-approved Reliability Standard with equivalent or more stringent requirements shall supersede the above requirements.

- 9.6.4.3 Exemptions. Large Generating Facilities that are regulated by the United States Nuclear Regulatory Commission shall be exempt from Sections 9.6.4, 9.6.4.1, and 9.6.4.2 of this Agreement. Large Generating Facilities that are behind the meter generation that is sized-to-load (i.e., the thermal load and the generation are near-balanced in real-time operation and the generation is primarily controlled to maintain the unique thermal, chemical, or mechanical output necessary for the operating requirements of its host facility) shall be required to install primary frequency response capability in accordance with the droop and deadband capability requirements specified in Section 9.6.4, but shall be otherwise exempt from the operating requirements in Sections 9.6.4, 9.6.4.1, 9.6.4.2, and 9.6.4.4 of this Agreement.
- 9.6.4.4 Electric Storage Resources. Interconnection Customer interconnecting an electric storage resource shall establish an operating range in Appendix C of its LGIA that specifies a minimum state of charge and a maximum state of charge between which the electric storage resource will be required to provide primary frequency response consistent with the conditions set forth in Sections 9.6.4, 9.6.4.1, 9.6.4.2, and 9.6.4.3 of this Agreement. Appendix C shall specify whether the operating range is static or dynamic, and shall consider (1) the expected magnitude of frequency deviations in the interconnection; (2) the expected duration that system frequency will remain outside of the deadband parameter in the interconnection; (3) the expected incidence of frequency deviations outside of the deadband parameter in the

interconnection; (4) the physical capabilities of the electric storage resource; (5) operational limitations of the electric storage resource due to manufacturer specifications; and (6) any other relevant factors agreed to by Transmission Provider and Interconnection Customer, and in consultation with the relevant transmission owner or balancing authority as appropriate. If the operating range is dynamic, then Appendix C must establish how frequently the operating range will be reevaluated and the factors that may be considered during its reevaluation.

Interconnection Customer's electric storage resource is required to provide timely and sustained primary frequency response consistent with Section 9.6.4.2 of this Agreement when it is online and dispatched to inject electricity to the Transmission System and/or receive electricity from the Transmission System. This excludes circumstances when the electric storage resource is not dispatched to inject electricity to the Transmission System and/or dispatched to receive electricity from the Transmission System. If Interconnection Customer's electric storage resource is charging at the time of a frequency deviation outside of its deadband parameter, it is to increase (for over-frequency deviations) or decrease (for under-frequency deviations) the rate at which it is charging in accordance with its droop parameter. Interconnection Customer's electric storage resource is not required to change from charging to discharging, or vice versa, unless the response necessitated by the droop and deadband settings requires it to do so and it is technically capable of making such a transition.

9.7 Outages and Interruptions.

9.7.1 Outages.

9.7.1.1 Outage Authority and Coordination. Each Party may in accordance with Good Utility Practice in coordination with the other Party remove from service any of its respective Interconnection Facilities or Network Upgrades that may impact the other Party's facilities as necessary to perform maintenance or testing or to install or replace equipment. Absent an Emergency Condition, the Party scheduling a removal of such facility(ies) from service will use Reasonable

Efforts to schedule such removal on a date and time mutually acceptable to the Parties. In all circumstances, any Party planning to remove such facility(ies) from service shall use Reasonable Efforts to minimize the effect on the other Party of such removal.

9.7.1.2 Outage Schedules. Transmission Provider shall post scheduled outages of its transmission facilities on the OASIS. Interconnection Customer shall submit its planned maintenance schedules for the Large Generating Facility to Transmission Provider for a minimum of a rolling twentyfour month period. Interconnection Customer shall update its planned maintenance schedules as necessary. Transmission Provider may request Interconnection Customer to reschedule its maintenance as necessary to maintain the reliability of the Transmission System; provided, however, adequacy of generation supply shall not be a criterion in determining Transmission System reliability. Transmission Provider shall compensate Interconnection Customer for any additional direct costs that Interconnection Customer incurs as a result of having to reschedule maintenance, including any additional overtime, breaking of maintenance contracts or other costs above and beyond the cost Interconnection Customer would have incurred absent Transmission Provider's request to reschedule maintenance. Interconnection Customer will not be eligible to receive compensation, if during the twelve (12) months prior to the date of the scheduled maintenance, Interconnection Customer had modified its schedule of maintenance activities.

9.7.1.3 Outage Restoration. If an outage on a Party's Interconnection Facilities or Network Upgrades adversely affects the other Party's operations or facilities, the Party that owns or controls the facility that is out of service shall use Reasonable Efforts to promptly restore such facility(ies) to a normal operating condition consistent with the nature of the outage. The Party that owns or controls the facility that is out of service shall provide the other Party, to the extent such information is known, information on the nature of the Emergency Condition, an estimated time of restoration, and any corrective actions required. Initial verbal notice shall be followed up as soon as practicable with written notice explaining the nature of the outage.

- **9.7.2 Interruption of Service**. If required by Good Utility Practice to do so, Transmission Provider may require Interconnection Customer to interrupt or reduce deliveries of electricity if such delivery of electricity could adversely affect Transmission Provider's ability to perform such activities as are necessary to safely and reliably operate and maintain the Transmission System. The following provisions shall apply to any interruption or reduction permitted under this Article 9.7.2:
 - **9.7.2.1** The interruption or reduction shall continue only for so long as reasonably necessary under Good Utility Practice;
 - 9.7.2.2 Any such interruption or reduction shall be made on an equitable, non-discriminatory basis with respect to all generating facilities directly connected to the Transmission System;
 - 9.7.2.3 When the interruption or reduction must be made under circumstances which do not allow for advance notice, Transmission Provider shall notify Interconnection Customer by telephone as soon as practicable of the reasons for the curtailment, interruption, or reduction, and, if known, its expected duration. Telephone notification shall be followed by written notification as soon as practicable;
 - 9.7.2.4 Except during the existence of an Emergency Condition, when the interruption or reduction can be scheduled without advance notice, Transmission Provider shall notify Interconnection Customer in advance regarding the timing of such scheduling and further notify Interconnection Customer of the expected duration. Transmission Provider shall coordinate with Interconnection Customer using Good Utility Practice to schedule the interruption or reduction during periods of least impact to Interconnection Customer and Transmission Provider;
 - 9.7.2.5 The Parties shall cooperate and coordinate with each other to the extent necessary in order to restore the Large Generating Facility, Interconnection Facilities, and the Transmission System to their normal operating state, consistent with system conditions and Good Utility Practice.
- 9.7.3 Under-Frequency and Over Frequency Conditions. The Transmission

System is designed to automatically activate a load-shed program as required by the Applicable Reliability Council in the event of an under-frequency system disturbance. Interconnection Customer shall implement under-frequency and over-frequency relay set points for the Large Generating Facility as required by the Applicable Reliability Council to ensure "ride through" capability of the Transmission System. Large Generating Facility response to frequency deviations of pre-determined magnitudes, both under-frequency and over-frequency deviations, shall be studied and coordinated with Transmission Provider in accordance with Good Utility Practice. The term "ride through" as used herein shall mean the ability of a Generating Facility to stay connected to and synchronized with the Transmission System during system disturbances within a range of under-frequency and over-frequency conditions, in accordance with Good Utility Practice.

9.7.4 System Protection and Other Control Requirements.

- 9.7.4.1 System Protection Facilities. Interconnection Customer shall, at its expense, install, operate and maintain System Protection Facilities as a part of the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Transmission Provider shall install at Interconnection Customer's expense any System Protection Facilities that may be required on Transmission Provider's Interconnection Facilities or the Transmission System as a result of the interconnection of the Large Generating Facility and Interconnection Customer's Interconnection Facilities.
- 9.7.4.2 Each Party's protection facilities shall be designed and coordinated with other systems in accordance with Good Utility Practice.
- **9.7.4.3** Each Party shall be responsible for protection of its facilities consistent with Good Utility Practice.
- 9.7.4.4 Each Party's protective relay design shall incorporate the necessary test switches to perform the tests required in Article 6. The required test switches will be placed such that they allow operation of lockout relays while preventing breaker failure schemes from operating and causing unnecessary breaker operations and/or the tripping of Interconnection Customer's units.

- **9.7.4.5** Each Party will test, operate and maintain System Protection Facilities in accordance with Good Utility Practice.
- 9.7.4.6 Prior to the In-Service Date, and again prior to the Commercial Operation Date, each Party or its agent shall perform a complete calibration test and functional trip test of the System Protection Facilities. At intervals suggested by Good Utility Practice and following any apparent malfunction of the System Protection Facilities, each Party shall perform both calibration and functional trip tests of its System Protection Facilities. These tests do not require the tripping of any in-service generation unit. These tests do, however, require that all protective relays and lockout contacts be activated.
- **9.7.5** Requirements for Protection. In compliance with Good Utility Practice, Interconnection Customer shall provide, install, own, and maintain relays, circuit breakers and all other devices necessary to remove any fault contribution of the Large Generating Facility to any short circuit occurring on the Transmission System not otherwise isolated by Transmission Provider's equipment, such that the removal of the fault contribution shall be coordinated with the protective requirements of the Transmission System. Such protective equipment shall include, without limitation, a disconnecting device or switch with load-interrupting capability located between the Large Generating Facility and the Transmission System at a site selected upon mutual agreement (not to be unreasonably withheld, conditioned or delayed) of the Parties. Interconnection Customer shall be responsible for protection of the Large Generating Facility and Interconnection Customer's other equipment from such conditions as negative sequence currents, over- or under-frequency, sudden load rejection, over- or under-voltage, and generator loss-of-field. Interconnection Customer shall be solely responsible to disconnect the Large Generating Facility and Interconnection Customer's other equipment if conditions on the Transmission System could adversely affect the Large Generating Facility.
- 9.7.6 Power Quality. Neither Party's facilities shall cause excessive voltage flicker nor introduce excessive distortion to the sinusoidal voltage or current waves as defined by ANSI Standard C84.1-1989, in accordance with IEEE Standard 519, or any applicable superseding electric industry standard. In the event of a conflict between ANSI Standard C84.1-1989, or any applicable superseding electric industry standard, ANSI Standard C84.1-1989, or the applicable superseding electric industry standard, shall

control.

- 9.8 Switching and Tagging Rules. Each Party shall provide the other Party a copy of its switching and tagging rules that are applicable to the other Party's activities. Such switching and tagging rules shall be developed on a non-discriminatory basis. The Parties shall comply with applicable switching and tagging rules, as amended from time to time, in obtaining clearances for work or for switching operations on equipment.
- 9.9 Use of Interconnection Facilities by Third Parties.
 - **9.9.1 Purpose of Interconnection Facilities**. Except as may be required by Applicable Laws and Regulations, or as otherwise agreed to among the Parties, the Interconnection Facilities shall be constructed for the sole purpose of interconnecting the Large Generating Facility to the Transmission System and shall be used for no other purpose.
 - 9.9.2 **Third Party Users.** If required by Applicable Laws and Regulations or if the Parties mutually agree, such agreement not to be unreasonably withheld, to allow one or more third parties to use Transmission Provider's Interconnection Facilities, or any part thereof, Interconnection Customer will be entitled to compensation for the capital expenses it incurred in connection with the Interconnection Facilities based upon the pro rata use of the Interconnection Facilities by Transmission Provider, all third party users, and Interconnection Customer, in accordance with Applicable Laws and Regulations or upon some other mutually-agreed upon methodology. In addition, cost responsibility for ongoing costs, including operation and maintenance costs associated with the Interconnection Facilities, will be allocated between Interconnection Customer and any third party users based upon the pro rata use of the Interconnection Facilities by Transmission Provider, all third party users, and Interconnection Customer, in accordance with Applicable Laws and Regulations or upon some other mutually agreed upon methodology. If the issue of such compensation or allocation cannot be resolved through such negotiations, it shall be submitted to FERC for resolution.
- 9.10 Disturbance Analysis Data Exchange. The Parties will cooperate with one another in the analysis of disturbances to either the Large Generating Facility or Transmission Provider's Transmission System by gathering and providing access to any information relating to any disturbance, including information from oscillography, protective relay targets, breaker operations and sequence of events records, and any disturbance information required by Good Utility Practice.

Article 10. Maintenance

- **10.1 Transmission Provider Obligations**. Transmission Provider shall maintain the Transmission System and Transmission Provider's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA.
- **10.2** Interconnection Customer Obligations. Interconnection Customer shall maintain the Large Generating Facility and Interconnection Customer's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA.
- **10.3** Coordination. The Parties shall confer regularly to coordinate the planning, scheduling and performance of preventive and corrective maintenance on the Large Generating Facility and the Interconnection Facilities.
- 10.4 Secondary Systems. Each Party shall cooperate with the other in the inspection, maintenance, and testing of control or power circuits that operate below 600 volts, AC or DC, including, but not limited to, any hardware, control or protective devices, cables, conductors, electric raceways, secondary equipment panels, transducers, batteries, chargers, and voltage and current transformers that directly affect the operation of a Party's facilities and equipment which may reasonably be expected to impact the other Party. Each Party shall provide advance notice to the other Party before undertaking any work on such circuits, especially on electrical circuits involving circuit breaker trip and close contacts, current transformers, or potential transformers.
- 10.5 Operating and Maintenance Expenses. Subject to the provisions herein addressing the use of facilities by others, and except for operations and maintenance expenses associated with modifications made for providing interconnection or transmission service to a third party and such third party pays for such expenses, Interconnection Customer shall be responsible for all reasonable expenses including overheads, associated with: (1) owning, operating, maintaining, repairing, and replacing Interconnection Customer's Interconnection Facilities; and (2) operation, maintenance, repair and replacement of Transmission Provider's Interconnection Facilities.

Article 11. Performance Obligation

11.1 Interconnection Customer Interconnection Facilities. Interconnection Customer shall design, procure, construct, install, own and/or control Interconnection Customer Interconnection Facilities described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades, at its sole expense.

- **11.2 Transmission Provider's Interconnection Facilities**. Transmission Provider or Transmission Owner shall design, procure, construct, install, own and/or control the Transmission Provider's Interconnection Facilities described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades, at the sole expense of the Interconnection Customer.
- 11.3 Network Upgrades and Distribution Upgrades. Transmission Provider or Transmission Owner shall design, procure, construct, install, and own the Network Upgrades and Distribution Upgrades described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades. The Interconnection Customer shall be responsible for all costs related to Distribution Upgrades. Unless Transmission Provider or Transmission Owner elects to fund the capital for the Network Upgrades, they shall be solely funded by Interconnection Customer.

11.4 Transmission Credits.

11.4.1 Repayment of Amounts Advanced for Network Upgrades.

Interconnection Customer shall be entitled to a cash repayment, equal to the total amount paid to Transmission Provider and Affected System Operator, if any, for the Network Upgrades, including any tax gross-up or other tax-related payments associated with Network Upgrades, and not refunded to Interconnection Customer pursuant to Article 5.17.8 or otherwise, to be paid to Interconnection Customer on a dollar-for-dollar basis for the nonusage sensitive portion of transmission charges, as payments are made under Transmission Provider's Tariff and Affected System's Tariff for transmission services with respect to the Large Generating Facility. Any repayment shall include interest calculated in accordance with the methodology set forth in FERC's regulations at 18 C.F.R. § 35.19a(a)(2)(iii) from the date of any payment for Network Upgrades through the date on which the Interconnection Customer receives a repayment of such payment pursuant to this subparagraph. Interconnection Customer may assign such repayment rights to any person.

Notwithstanding the foregoing, Interconnection Customer, Transmission Provider, and Affected System Operator may adopt any alternative payment schedule that is mutually agreeable so long as Transmission Provider and Affected System Operator take one of the following actions no later than five years from the Commercial Operation Date: (1) return to Interconnection Customer any amounts advanced for Network Upgrades not previously repaid, or (2) declare in writing that Transmission Provider or Affected System Operator will continue to provide payments to Interconnection Customer on a dollar-for-dollar basis for the non-usage sensitive portion of transmission charges, or develop an alternative schedule that is mutually agreeable and provides for the return of all amounts advanced for Network Upgrades not previously repaid; however, full reimbursement shall not extend beyond twenty (20) years from the Commercial Operation Date.

If the Large Generating Facility fails to achieve eCommercial eOperation, but it or another Generating Facility is later constructed and makes use of the Network Upgrades, Transmission Provider and Affected System Operator shall at that time reimburse Interconnection Customer for the amounts advanced for the Network Upgrades. Before any such reimbursement can occur, the Interconnection Customer, or the entity that ultimately constructs the Generating Facility, if different, is responsible for identifying the entity to which reimbursement must be made.

- 11.4.2 Special Provisions for Affected Systems. Unless Transmission Provider provides, under the LGIA, for the repayment of amounts advanced to Affected System Operator for Network Upgrades, Interconnection Customer and Affected System Operator shall enter into an agreement that provides for such repayment. The agreement shall specify the terms governing payments to be made by Interconnection Customer to the Affected System Operator as well as the repayment by the Affected System Operator.
- 11.4.3 Notwithstanding any other provision of this LGIA, nothing herein shall be construed as relinquishing or foreclosing any rights, including but not limited to firm transmission rights, capacity rights, transmission congestion rights, or transmission credits, that Interconnection Customer, shall be entitled to, now or in the future under any other agreement or tariff as a result of, or otherwise associated with, the transmission capacity, if any, created by the Network Upgrades, including the right to obtain cash reimbursements or transmission credits for transmission service that is not associated with the Large Generating Facility.
- 11.5 Provision of Security. At least thirty (30) Calendar Days prior to the commencement of the procurement, installation, or construction of a discrete

portion of a Transmission Provider's Interconnection Facilities, Network Upgrades, or Distribution Upgrades, Interconnection Customer shall provide Transmission Provider, at Interconnection Customer's option, a guarantee, a surety bond, letter of credit or other form of security that is reasonably acceptable to Transmission Provider and is consistent with the Uniform Commercial Code of the jurisdiction identified in Article 14.2.1. Such security for payment shall be in an amount sufficient to cover the costs for constructing, procuring and installing the applicable portion of Transmission Provider's Interconnection Facilities, Network Upgrades, or Distribution Upgrades and shall be reduced on a dollar-for-dollar basis for payments made to Transmission Provider for these purposes.

In addition:

- 11.5.1 The guarantee must be made by an entity that meets the creditworthiness requirements of Transmission Provider, and contain terms and conditions that guarantee payment of any amount that may be due from Interconnection Customer, up to an agreed-to maximum amount.
- **11.5.2** The letter of credit must be issued by a financial institution reasonably acceptable to Transmission Provider and must specify a reasonable expiration date.
- 11.5.3 The surety bond must be issued by an insurer reasonably acceptable to Transmission Provider and must specify a reasonable expiration date.
- directs Interconnection Customer to provide a service pursuant to Articles 9.6.3 (Payment for Reactive Power), or 13.5.1 of this LGIA, Transmission Provider shall compensate Interconnection Customer in accordance with Interconnection Customer's applicable rate schedule then in effect unless the provision of such service(s) is subject to an RTO or ISO FERC-approved rate schedule. Interconnection Customer shall serve Transmission Provider or RTO or ISO with any filing of a proposed rate schedule at the time of such filing with FERC. To the extent that no rate schedule is in effect at the time the Interconnection Customer is required to provide or absorb any Reactive Power under this LGIA, Transmission Provider agrees to compensate Interconnection Customer in such amount as would have been due Interconnection Customer had the rate schedule been in effect at the time service commenced; provided, however, that such rate schedule must be filed at FERC or other appropriate Governmental Authority within sixty (60) Calendar Days of the commencement of service.

11.6.1 Interconnection Customer Compensation for Actions During

Emergency Condition. Transmission Provider or RTO or ISO shall compensate Interconnection Customer for its provision of real and reactive power and other Emergency Condition services that Interconnection Customer provides to support the Transmission System during an Emergency Condition in accordance with Article 11.6.

Article 12. Invoice

- 12.1 General. Each Party shall submit to the other Party, on a monthly basis, invoices of amounts due for the preceding month. Each invoice shall state the month to which the invoice applies and fully describe the services and equipment provided. The Parties may discharge mutual debts and payment obligations due and owing to each other on the same date through netting, in which case all amounts a Party owes to the other Party under this LGIA, including interest payments or credits, shall be netted so that only the net amount remaining due shall be paid by the owing Party.
- 12.2 Final Invoice. Within six months after completion of the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades, Transmission Provider shall provide an invoice of the final cost of the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades and shall set forth such costs in sufficient detail to enable Interconnection Customer to compare the actual costs with the estimates and to ascertain deviations, if any, from the cost estimates. Transmission Provider shall refund to Interconnection Customer any amount by which the actual payment by Interconnection Customer for estimated costs exceeds the actual costs of construction within thirty (30) Calendar Days of the issuance of such final construction invoice.
- 12.3 Payment. Invoices shall be rendered to the paying Party at the address specified in Appendix F. The Party receiving the invoice shall pay the invoice within thirty (30) Calendar Days of receipt. All payments shall be made in immediately available funds payable to the other Party, or by wire transfer to a bank named and account designated by the invoicing Party. Payment of invoices by either Party will not constitute a waiver of any rights or claims either Party may have under this LGIA. If Interconnection Customer has not paid the final invoice following a withdrawal within thirty (30) Calendar Days, Transmission Provider shall draw upon the security provided under this LGIA to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

12.4 Disputes. In the event of a billing dispute between Transmission Provider and Interconnection Customer, Transmission Provider shall continue to provide Interconnection Service under this LGIA as long as Interconnection Customer: (i) continues to make all payments not in dispute; and (ii) pays to Transmission Provider or into an independent escrow account the portion of the invoice in dispute, pending resolution of such dispute. If Interconnection Customer fails to meet these two requirements for continuation of service, then Transmission Provider may provide notice to Interconnection Customer of a Default pursuant to Article 17. Within thirty (30) Calendar Days after the resolution of the dispute, the Party that owes money to the other Party shall pay the amount due with interest calculated in accord with the methodology set forth in FERC's regulations at 18 CFR § 35.19a(a)(2)(iii).

Article 13. Emergencies

- 13.1 Definition. "Emergency Condition" shall mean a condition or situation: (i) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (ii) that, in the case of Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to the Transmission System, Transmission Provider's Interconnection Facilities or the Transmission Systems of others to which the Transmission System is directly connected; or (iii) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Large Generating Facility or Interconnection Customer's Interconnection Facilities' System restoration and black start shall be considered Emergency Conditions; provided, that Interconnection Customer is not obligated by this LGIA to possess black start capability.
- **Obligations**. Each Party shall comply with the Emergency Condition procedures of the applicable ISO/RTO, NERC, the Applicable Reliability Council, Applicable Laws and Regulations, and any emergency procedures agreed to by the Joint Operating Committee.
- when it becomes aware of an Emergency Condition that affects Transmission Provider's Interconnection Facilities or the Transmission System that may reasonably be expected to affect Interconnection Customer's operation of the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Interconnection Customer shall notify Transmission Provider promptly when it becomes aware of an Emergency Condition that affects the Large Generating Facility or Interconnection Customer's Interconnection Facilities that

may reasonably be expected to affect the Transmission System or Transmission Provider's Interconnection Facilities. To the extent information is known, the notification shall describe the Emergency Condition, the extent of the damage or deficiency, the expected effect on the operation of Interconnection Customer's or Transmission Provider's facilities and operations, its anticipated duration and the corrective action taken and/or to be taken. The initial notice shall be followed as soon as practicable with written notice.

13.4 Immediate Action. Unless, in Interconnection Customer's reasonable judgment, immediate action is required, Interconnection Customer shall obtain the consent of Transmission Provider, such consent to not be unreasonably withheld, prior to performing any manual switching operations at the Large Generating Facility or Interconnection Customer's Interconnection Facilities in response to an Emergency Condition either declared by Transmission Provider or otherwise regarding the Transmission System.

13.5 Transmission Provider Authority.

13.5.1 General. Transmission Provider may take whatever actions or inactions with regard to the Transmission System or Transmission Provider's Interconnection Facilities it deems necessary during an Emergency Condition in order to (i) preserve public health and safety, (ii) preserve the reliability of the Transmission System or Transmission Provider's Interconnection Facilities, (iii) limit or prevent damage, and (iv) expedite restoration of service.

Transmission Provider shall use Reasonable Efforts to minimize the effect of such actions or inactions on the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Transmission Provider may, on the basis of technical considerations, require the Large Generating Facility to mitigate an Emergency Condition by taking actions necessary and limited in scope to remedy the Emergency Condition, including, but not limited to, directing Interconnection Customer to shut-down, startup, increase or decrease the real or reactive power output of the Large Generating Facility; implementing a reduction or disconnection pursuant to Article 13.5.2; directing Interconnection Customer to assist with blackstart (if available) or restoration efforts; or altering the outage schedules of the Large Generating Facility and Interconnection Customer's Interconnection Facilities. Interconnection Customer shall comply with all of Transmission Provider's operating instructions concerning Large Generating Facility real power and reactive power output within the manufacturer's design limitations

of the Large Generating Facility's equipment that is in service and physically available for operation at the time, in compliance with Applicable Laws and Regulations.

- 13.5.2 **Reduction and Disconnection**. Transmission Provider may reduce Interconnection Service or disconnect the Large Generating Facility or Interconnection Customer's Interconnection Facilities, when such, reduction or disconnection is necessary under Good Utility Practice due to Emergency Conditions. These rights are separate and distinct from any right of curtailment of Transmission Provider pursuant to Transmission Provider's Tariff. When Transmission Provider can schedule the reduction or disconnection in advance. Transmission Provider shall notify Interconnection Customer of the reasons, timing and expected duration of the reduction or disconnection. Transmission Provider shall coordinate with Interconnection Customer using Good Utility Practice to schedule the reduction or disconnection during periods of least impact to Interconnection Customer and Transmission Provider. Any reduction or disconnection shall continue only for so long as reasonably necessary under Good Utility Practice. The Parties shall cooperate with each other to restore the Large Generating Facility, the Interconnection Facilities, and the Transmission System to their normal operating state as soon as practicable consistent with Good Utility Practice.
- 13.6 Interconnection Customer Authority. Consistent with Good Utility Practice and the LGIA and the LGIP, Interconnection Customer may take actions or inactions with regard to the Large Generating Facility or Interconnection Customer's Interconnection Facilities during an Emergency Condition in order to (i) preserve public health and safety, (ii) preserve the reliability of the Large Generating Facility or Interconnection Customer's Interconnection Facilities, (iii) limit or prevent damage, and (iv) expedite restoration of service. Interconnection Customer shall use Reasonable Efforts to minimize the effect of such actions or inactions on the Transmission System and Transmission Provider's Interconnection Facilities. Transmission Provider shall use Reasonable Efforts to assist Interconnection Customer in such actions.
- **13.7 Limited Liability**. Except as otherwise provided in Article 11.6.1 of this LGIA, neither Party shall be liable to the other for any action it takes in responding to an Emergency Condition so long as such action is made in good faith and is consistent with Good Utility Practice.

Article 14. Regulatory Requirements and Governing Law

14.1 Regulatory Requirements. Each Party's obligations under this LGIA shall be subject to its receipt of any required approval or certificate from one or more Governmental Authorities in the form and substance satisfactory to the applying Party, or the Party making any required filings with, or providing notice to, such Governmental Authorities, and the expiration of any time period associated therewith. Each Party shall in good faith seek and use its Reasonable Efforts to obtain such other approvals. Nothing in this LGIA shall require Interconnection Customer to take any action that could result in its inability to obtain, or its loss of, status or exemption under the Federal Power Act, the Public Utility Holding Company Act of 1935, as amended, or the Public Utility Regulatory Policies Act of 1978.

14.2 Governing Law.

- **14.2.1** The validity, interpretation and performance of this LGIA and each of its provisions shall be governed by the laws of the state where the Point of Interconnection is located, without regard to its conflicts of law principles.
- **14.2.2** This LGIA is subject to all Applicable Laws and Regulations.
- 14.3.3 Each Party expressly reserves the right to seek changes in, appeal, or otherwise contest any laws, orders, rules, or regulations of a Governmental Authority.

Article 15. Notices.

- 15.1 General. Unless otherwise provided in this LGIA, any notice, demand or request required or permitted to be given by either Party to the other and any instrument required or permitted to be tendered or delivered by either Party in writing to the other shall be effective when delivered and may be so given, tendered or delivered, by recognized national courier, or by depositing the same with the United States Postal Service with postage prepaid, for delivery by certified or registered mail, addressed to the Party, or personally delivered to the Party, at the address set out in Appendix F, Addresses for Delivery of Notices and Billings.
 - Either Party may change the notice information in this LGIA by giving five (5) Business Days written notice prior to the effective date of the change.
- **15.2 Billings and Payments**. Billings and payments shall be sent to the addresses set out in Appendix F.

- **15.3 Alternative Forms of Notice**. Any notice or request required or permitted to be given by a Party to the other and not required by this Agreement to be given in writing may be so given by telephone, facsimile or email to the telephone numbers and email addresses set out in Appendix F.
- **15.4** Operations and Maintenance Notice. Each Party shall notify the other Party in writing of the identity of the person(s) that it designates as the point(s) of contact with respect to the implementation of Articles 9 and 10.

Article 16. Force Majeure

16.1 Force Majeure.

- **16.1.1** Economic hardship is not considered a Force Majeure event.
- **16.1.2** Neither Party shall be considered to be in Default with respect to any obligation hereunder, (including obligations under Article 4), other than the obligation to pay money when due, if prevented from fulfilling such obligation by Force Majeure. A Party unable to fulfill any obligation hereunder (other than an obligation to pay money when due) by reason of Force Majeure shall give notice and the full particulars of such Force Majeure to the other Party in writing or by telephone as soon as reasonably possible after the occurrence of the cause relied upon. Telephone notices given pursuant to this article shall be confirmed in writing as soon as reasonably possible and shall specifically state full particulars of the Force Majeure, the time and date when the Force Majeure occurred and when the Force Majeure is reasonably expected to cease. The Party affected shall exercise due diligence to remove such disability with reasonable dispatch, but shall not be required to accede or agree to any provision not satisfactory to it in order to settle and terminate a strike or other labor disturbance.

Article 17. Default

17.1 Default

17.1 General. No Default shall exist where such failure to discharge an obligation (other than the payment of money) is the result of Force Majeure as defined in this LGIA or the result of an act of omission of the other Party. Upon a Breach, the non-breaching Party shall give written notice of such Breach to the breaching Party. Except as

provided in Article 17.1.2, the breaching Party shall have thirty (30) Calendar Days from receipt of the Default notice within which to cure such Breach; provided however, if such Breach is not capable of cure within thirty (30) Calendar Days, the breaching Party shall commence such cure within thirty (30) Calendar Days after notice and continuously and diligently complete such cure within ninety (90) Calendar Days from receipt of the Default notice; and, if cured within such time, the Breach specified in such notice shall cease to exist.

Right to Terminate. If a Breach is not cured as provided in this article, or if a Breach is not capable of being cured within the period provided for herein, the non-breaching Party shall have the right to declare a Default and terminate this LGIA by written notice at any time until cure occurs, and be relieved of any further obligation hereunder and, whether or not that Party terminates this LGIA, to recover from the breaching Party all amounts due hereunder, plus all other damages and remedies to which it is entitled at law or in equity. The provisions of this article will survive termination of this LGIA.

Article 18. Indemnity, Consequential Damages and Insurance

- 18.1 Indemnity. The Parties shall at all times indemnify, defend, and hold the other Party harmless from, any and all damages, losses, claims, including claims and actions relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's action or inactions of its obligations under this LGIA on behalf of the Indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the indemnified Party.
 - 18.1.1 Indemnified Person. If an Indemnified Person is entitled to indemnification under this Article 18 as a result of a claim by a third party, and the Indemnifying Party fails, after notice and reasonable opportunity to proceed under Article 18.1, to assume the defense of such claim, such Indemnified Person may at the expense of the Indemnifying Party contest, settle or consent to the entry of any judgment with respect to, or pay in full, such claim.
 - **18.1.2 Indemnifying Party**. If an Indemnifying Party is obligated to indemnify and hold any Indemnified Person harmless under this Article 18, the amount owing to the Indemnified Person shall be the amount of such

Indemnified Person's actual Loss, net of any insurance or other recovery.

18.1.3 Indemnity Procedures. Promptly after receipt by an Indemnified Person of any claim or notice of the commencement of any action or administrative or legal proceeding or investigation as to which the indemnity provided for in Article 18.1 may apply, the Indemnified Person shall notify the Indemnifying Party of such fact. Any failure of or delay in such notification shall not affect a Party's indemnification obligation unless such failure or delay is materially prejudicial to the Indemnifying Party.

The Indemnifying Party shall have the right to assume the defense thereof with counsel designated by such Indemnifying Party and reasonably satisfactory to the Indemnified Person. If the defendants in any such action include one or more Indemnified Persons and the Indemnifying Party and if the Indemnified Person reasonably concludes that there may be legal defenses available to it and/or other Indemnified Persons which are different from or additional to those available to the Indemnifying Party, the Indemnified Person shall have the right to select separate counsel to assert such legal defenses and to otherwise participate in the defense of such action on its own behalf. In such instances, the Indemnifying Party shall only be required to pay the fees and expenses of one additional attorney to represent an Indemnified Person or Indemnified Persons having such differing or additional legal defenses.

The Indemnified Person shall be entitled, at its expense, to participate in any such action, suit or proceeding, the defense of which has been assumed by the Indemnifying Party.

Notwithstanding the foregoing, the Indemnifying Party (i) shall not be entitled to assume and control the defense of any such action, suit or proceedings if and to the extent that, in the opinion of the Indemnified Person and its counsel, such action, suit or proceeding involves the potential imposition of criminal liability on the Indemnified Person, or there exists a conflict or adversity of interest between the Indemnified Person and the Indemnifying Party, in such event the Indemnified Person, and (ii) shall not settle or consent to the entry of any judgment in any action, suit or proceeding without the consent of the Indemnified Person, which shall not be reasonably withheld, conditioned or delayed.

- 18.2 Consequential Damages. Other than the Liquidated Damages heretofore described, in no event shall either Party be liable under any provision of this LGIA for any losses, damages, costs or expenses for any special, indirect, incidental, consequential, or punitive damages, including but not limited to loss of profit or revenue, loss of the use of equipment, cost of capital, cost of temporary equipment or services, whether based in whole or in part in contract, in tort, including negligence, strict liability, or any other theory of liability; provided, however, that damages for which a Party may be liable to the other Party under another agreement will not be considered to be special, indirect, incidental, or consequential damages hereunder.
- **18.3 Insurance**. Each party shall, at its own expense, maintain in force throughout the period of this LGIA, and until released by the other Party, the following minimum insurance coverages, with insurers authorized to do business in the state where the Point of Interconnection is located:
 - **18.3.1** Employers' Liability and Workers' Compensation Insurance providing statutory benefits in accordance with the laws and regulations of the state in which the Point of Interconnection is located.
 - 18.3.2 Commercial General Liability Insurance including premises and operations, personal injury, broad form property damage, broad form blanket contractual liability coverage (including coverage for the contractual indemnification) products and completed operations coverage, coverage for explosion, collapse and underground hazards, independent contractors coverage, coverage for pollution to the extent normally available and punitive damages to the extent normally available and a cross liability endorsement, with minimum limits of One Million Dollars (\$1,000,000) per occurrence/One Million Dollars (\$1,000,000) aggregate combined single limit for personal injury, bodily injury, including death and property damage.
 - **18.3.3** Comprehensive Automobile Liability Insurance for coverage of owned and non-owned and hired vehicles, trailers or semi-trailers designed for travel on public roads, with a minimum, combined single limit of One Million Dollars (\$1,000,000) per occurrence for bodily injury, including death, and property damage.
 - 18.3.4 Excess Public Liability Insurance over and above the Employers' Liability Commercial General Liability and Comprehensive Automobile Liability Insurance coverage, with a minimum combined single limit of Twenty Million Dollars (\$20,000,000) per occurrence/Twenty Million Dollars (\$20,000,000) aggregate.

- 18.3.5 The Commercial General Liability Insurance, Comprehensive Automobile Insurance and Excess Public Liability Insurance policies shall name the other Party, its parent, associated and Affiliate companies and their respective directors, officers, agents, servants and employees ("Other Party Group") as additional insured. All policies shall contain provisions whereby the insurers waive all rights of subrogation in accordance with the provisions of this LGIA against the Other Party Group and provide thirty (30) Calendar Days advance written notice to the Other Party Group prior to anniversary date of cancellation or any material change in coverage or condition.
- 18.3.6 The Commercial General Liability Insurance, Comprehensive Automobile Liability Insurance and Excess Public Liability Insurance policies shall contain provisions that specify that the policies are primary and shall apply to such extent without consideration for other policies separately carried and shall state that each insured is provided coverage as though a separate policy had been issued to each, except the insurer's liability shall not be increased beyond the amount for which the insurer would have been liable had only one insured been covered. Each Party shall be responsible for its respective deductibles or retentions.
- 18.3.7 The Commercial General Liability Insurance, Comprehensive Automobile Liability Insurance and Excess Public Liability Insurance policies, if written on a Claims First Made Basis, shall be maintained in full force and effect for two (2) years after termination of this LGIA, which coverage may be in the form of tail coverage or extended reporting period coverage if agreed by the Parties.
- 18.3.8 The requirements contained herein as to the types and limits of all insurance to be maintained by the Parties are not intended to and shall not in any manner, limit or qualify the liabilities and obligations assumed by the Parties under this LGIA.
- 18.3.9 Within ten (10) Calendar Delays following execution of this LGIA, and as soon as practicable after the end of each fiscal year or at the renewal of the insurance policy and in any event within ninety (90) Calendar Delays thereafter, each Party shall provide certification of all insurance required in this LGIA, executed by each insurer or by an authorized representative of each insurer.
- **18.3.10** Notwithstanding In addition to the foregoing, each Party may self-

insure to meet the minimum insurance requirements of Articles 18.3.2 through 18.3.8 to the extent it maintains a self-insurance program; provided that, such Party's senior secured debt is rated at investment grade or better by Standard & Poor's and that its self-insurance program meets the minimum insurance requirements of Articles 18.3.2 through 18.3.8. For any period of time that a Party's senior secured debt is unrated by Standard & Poor's or is rated at less than investment grade by Standard & Poor's, such Party shall comply with the insurance requirements applicable to it under Articles 18.3.2 through 18.3.9. In the event that a Party is permitted to self-insure pursuant to this article, it shall certify tonotify the other Party with a letter of self-insurance that it meets the requirements to self-insure and that its self-insurance program meets the minimum insurance requirements in a manner consistent with that specified in Article 18.3.9.

18.3.11 The Parties agree to report to each other in writing as soon as practical all accidents or occurrences resulting in injuries to any person, including death, and any property damage arising out of this LGIA.

Article 19. Assignment

19.1 **Assignment**. This LGIA may be assigned by either Party only with the written consent of the other; provided that either Party may assign this LGIA without the consent of the other Party to any Affiliate of the assigning Party with an equal or greater credit rating and with the legal authority and operational ability to satisfy the obligations of the assigning Party under this LGIA; and provided further that Interconnection Customer shall have the right to assign this LGIA, without the consent of Transmission Provider, for collateral security purposes to aid in providing financing for the Large Generating Facility, provided that Interconnection Customer will promptly notify Transmission Provider of any such assignment. Any financing arrangement entered into by Interconnection Customer pursuant to this article will provide that prior to or upon the exercise of the secured party's, trustee's or mortgagee's assignment rights pursuant to said arrangement, the secured creditor, the trustee or mortgagee will notify Transmission Provider of the date and particulars of any such exercise of assignment right(s), including providing the Transmission Provider with proof that it meets the requirements of Articles 11.5 and 18.3. Any attempted assignment that violates this article is void and ineffective. Any assignment under this LGIA shall not relieve a Party of its obligations, nor shall a Party's obligations be enlarged, in whole or in part, by reason thereof. Where required, consent to assignment will not be unreasonably withheld, conditioned or delayed.

Article 20. Severability

20.1 Severability. If any provision in this LGIA is finally determined to be invalid, void or unenforceable by any court or other Governmental Authority having jurisdiction, such determination shall not invalidate, void or make unenforceable any other provision, agreement or covenant of this LGIA; provided that if Interconnection Customer (or any third party, but only if such third party is not acting at the direction of Transmission Provider) seeks and obtains such a final determination with respect to any provision of the Alternate Option (Article 5.1.2), or the Negotiated Option (Article 5.1.4), then none of these provisions shall thereafter have any force or effect and the Parties' rights and obligations shall be governed solely by the Standard Option (Article 5.1.1).

Article 21. Comparability

21.1 Comparability. The Parties will comply with all applicable comparability and code of conduct laws, rules and regulations, as amended from time to time.

Article 22. Confidentiality

22.1 Confidentiality. Confidential Information shall include, without limitation, all information relating to a Party's technology, research and development, business affairs, and pricing, and any information supplied by either of the Parties to the other prior to the execution of this LGIA.

Information is Confidential Information only if it is clearly designated or marked in writing as confidential on the face of the document, or, if the information is conveyed orally or by inspection, if the Party providing the information orally informs the Party receiving the information that the information is confidential.

If requested by either Party, the other Party shall provide in writing, the basis for asserting that the information referred to in this Article 22 warrants confidential treatment, and the requesting Party may disclose such writing to the appropriate Governmental Authority. Each Party shall be responsible for the costs associated with affording confidential treatment to its information.

The release of Confidential Information shall be subject to Applicable Laws and Regulations and Applicable Reliability Standards.

22.1.1 Term. During the term of this LGIA, and for a period of three (3) years after the expiration or termination of this LGIA, except as otherwise provided in this Article 22, each Party shall hold

in confidence and shall not disclose to any person Confidential Information.

- **22.1.2 Scope.** Confidential Information shall not include information that the receiving Party can demonstrate: (1) is generally available to the public other than as a result of a disclosure by the receiving Party; (2) was in the lawful possession of the receiving Party on a non-confidential basis before receiving it from the disclosing Party; (3) was supplied to the receiving Party without restriction by a third party, who, to the knowledge of the receiving Party after due inquiry, was under no obligation to the disclosing Party to keep such information confidential; (4) was independently developed by the receiving Party without reference to Confidential Information of the disclosing Party; (5) is, or becomes, publicly known, through no wrongful act or omission of the receiving Party or Breach of this LGIA; or (6) is required, in accordance with Article 22.1.7 of the LGIA, Order of Disclosure, to be disclosed by any Governmental Authority or is otherwise required to be disclosed by law or subpoena, or is necessary in any legal proceeding establishing rights and obligations under this LGIA. Information designated as Confidential Information will no longer be deemed confidential if the Party that designated the information as confidential notifies the other Party that it no longer is confidential.
- 22.1.3 Release of Confidential Information. Neither Party shall release or disclose Confidential Information to any other person, except to its Affiliates (limited by the Standards of Conduct requirements), subcontractors, employees, consultants, or to parties who may be or considering providing financing to or equity participation with Interconnection Customer, or to potential purchasers or assignees of Interconnection Customer, on a need-to-know basis in connection with this LGIA, unless such person has first been advised of the confidentiality provisions of this Article 22 and has agreed to comply with such provisions. Notwithstanding the foregoing, a Party providing Confidential Information to any person shall remain primarily responsible for any release of Confidential Information in contravention of this Article 22.
- **22.1.4 Rights**. Each Party retains all rights, title, and interest in the Confidential Information that each Party discloses to the other

Party. The disclosure by each Party to the other Party of Confidential Information shall not be deemed a waiver by either Party or any other person or entity of the right to protect the Confidential Information from public disclosure.

- 22.1.5 No Warranties. By providing Confidential Information, neither Party makes any warranties or representations as to its accuracy or completeness. In addition, by supplying Confidential Information, neither Party obligates itself to provide any particular information or Confidential Information to the other Party nor to enter into any further agreements or proceed with any other relationship or joint venture.
- **22.1.6 Standard of Care**. Each Party shall use at least the same standard of care to protect Confidential Information it receives as it uses to protect its own Confidential Information from unauthorized disclosure, publication or dissemination. Each Party may use Confidential Information solely to fulfill its obligations to the other Party under this LGIA or its regulatory requirements.
- 22.1.7 Order of Disclosure. If a court or a Government Authority or entity with the right, power, and apparent authority to do so requests or requires either Party, by subpoena, oral deposition, interrogatories, requests for production of documents, administrative order, or otherwise, to disclose Confidential Information, that Party shall provide the other Party with prompt notice of such request(s) or requirement(s) so that the other Party may seek an appropriate protective order or waive compliance with the terms of this LGIA. Notwithstanding the absence of a protective order or waiver, the Party may disclose such Confidential Information which, in the opinion of its counsel, the Party is legally compelled to disclose. Each Party will use Reasonable Efforts to obtain reliable assurance that confidential treatment will be accorded any Confidential Information so furnished.
- **22.1.8 Termination of Agreement**. Upon termination of this LGIA for any reason, each Party shall, within ten (10) Calendar Days of receipt of a written request from the other Party, use Reasonable Efforts to destroy, erase, or delete (with such destruction, erasure, and deletion certified in writing to the other Party) or return to the other Party, without retaining

copies thereof, any and all written or electronic Confidential Information received from the other Party.

22.1.9 Remedies. The Parties agree that monetary damages would be inadequate to compensate a Party for the other Party's Breach of its obligations under this Article 22. Each Party accordingly agrees that the other Party shall be entitled to equitable relief, by way of injunction or otherwise, if the first Party Breaches or threatens to Breach its obligations under this Article 22, which equitable relief shall be granted without bond or proof of damages, and the receiving Party shall not plead in defense that there would be an adequate remedy at law. Such remedy shall not be deemed an exclusive remedy for the Breach of this Article 22, but shall be in addition to all other remedies available at law or in equity. The Parties further acknowledge and agree that the covenants contained herein are necessary for the protection of legitimate business interests and are reasonable in scope. No Party, however, shall be liable for indirect, incidental, or consequential or punitive damages of any nature or kind resulting from or arising in connection with this Article 22.

22.1.10 Disclosure to FERC, its Staff, or a State. Notwithstanding anything in this Article 22 to the contrary, and pursuant to 18 CFR section 1b.20, if FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to this LGIA, the Party shall provide the requested information to FERC or its staff, within the time provided for in the request for information. In providing the information to FERC or its staff, the Party must, consistent with 18 CFR section 388.112, request that the information be treated as confidential and non-public by FERC and its staff and that the information be withheld from public disclosure. Parties are prohibited from notifying the other Party to this LGIA prior to the release of the Confidential Information to FERC or its staff. The Party shall notify the other Party to the LGIA when it is notified by FERC or its staff that a request to release Confidential Information has been received by FERC, at which time either of the Parties may respond before such information would be made public, pursuant to 18 CFR section 388.112. Requests from a state regulatory body conducting a confidential investigation shall be treated in a

22.1.11

similar manner if consistent with the applicable state rules and regulations.

Subject to the exception in Article 22.1.10, any information that a Party claims is competitively sensitive, commercial or financial information under this LGIA ("Confidential Information") shall not be disclosed by the other Party to any person not employed or retained by the other Party, except to the extent disclosure is (i) required by law; (ii) reasonably deemed by the disclosing Party to be required to be disclosed in connection with a dispute between or among the Parties, or the defense of litigation or dispute; (iii) otherwise permitted by consent of the other Party, such consent not to be unreasonably withheld; or (iv) necessary to fulfill its obligations under this LGIA or as a transmission service provider or a Control Area operator including disclosing the Confidential Information to an RTO or ISO or to a regional or national reliability organization. The Party asserting confidentiality shall notify the other Party in writing of the information it claims is confidential. Prior to any disclosures of the other Party's Confidential Information under this subparagraph, or if any third party or Governmental Authority makes any request or demand for any of the information described in this subparagraph, the disclosing Party agrees to promptly notify the other Party in writing and agrees to assert confidentiality and cooperate with the other Party in seeking to protect the Confidential Information from public disclosure by confidentiality agreement, protective order or other reasonable measures.

Article 23. Environmental Releases

23.1 Each Party shall notify the other Party, first orally and then in writing, of the release of any Hazardous Substances, any asbestos or lead abatement activities, or any type of remediation activities related to the Large Generating Facility or the Interconnection Facilities, each of which may reasonably be expected to affect the other Party. The notifying Party shall: (i) provide the notice as soon as practicable, provided such Party makes a good faith effort to provide the notice no later than twenty-four hours after such Party becomes aware of the occurrence; and (ii) promptly furnish to the other Party copies of any publicly available reports filed with any Governmental Authorities addressing such events.

Article 24. Information Requirements

- **24.1 Information Acquisition**. Transmission Provider and Interconnection Customer shall submit specific information regarding the electrical characteristics of their respective facilities to each other as described below and in accordance with Applicable Reliability Standards.
- 24.2 Information Submission by Transmission Provider. The initial information submission by Transmission Provider shall occur no later than one hundred eighty (180) Calendar Days prior to Trial Operation and shall include Transmission System information necessary to allow Interconnection Customer to select equipment and meet any system protection and stability requirements, unless otherwise agreed to by the Parties. On a monthly basis Transmission Provider shall provide Interconnection Customer a status report on the construction and installation of Transmission Provider's Interconnection Facilities and Network Upgrades, including, but not limited to, the following information: (1) progress to date; (2) a description of the activities since the last report (3) a description of the action items for the next period; and (4) the delivery status of equipment ordered.
- 24.3 Updated Information Submission by Interconnection Customer. The updated information submission by Interconnection Customer, including manufacturer information, shall occur no later than one hundred eighty (180) Calendar Days prior to the Trial Operation. Interconnection Customer shall submit a completed copy of the Large Generating Facility data requirements contained in Appendix 1 to the LGIP. It shall also include any additional information provided to Transmission Provider for the Definitive Interconnection System Impact StudyFeasibility and Interconnection Facilities Study. Information in this submission shall be the most current Large Generating Facility design or expected performance data. Information submitted for stability models shall be compatible with Transmission Provider standard models. If there is no compatible model, Interconnection Customer will work with a consultant mutually agreed to by the Parties to develop and supply a standard model and associated information.

If Interconnection Customer's data is materially different from what was originally provided to Transmission Provider pursuant to the Interconnection Study Agreement between Transmission Provider and Interconnection Customer, then Transmission Provider will conduct appropriate studies to determine the impact on Transmission Provider Transmission System based on the actual data submitted pursuant to this Article 24.3. The Interconnection Customer shall not begin Trial Operation until such studies are completed.

24.4 Information Supplementation. Prior to the Operation Date, the Parties shall supplement their information submissions described above in this Article 24 with any and all "as-built" Large Generating Facility information or "as-tested"

performance information that differs from the initial submissions or, alternatively, written confirmation that no such differences exist. The Interconnection Customer shall conduct tests on the Large Generating Facility as required by Good Utility Practice such as an open circuit "step voltage" test on the Large Generating Facility to verify proper operation of the Large Generating Facility's automatic voltage regulator.

Unless otherwise agreed, the test conditions shall include: (1) Large Generating Facility at synchronous speed; (2) automatic voltage regulator on and in voltage control mode; and (3) a five percent change in Large Generating Facility terminal voltage initiated by a change in the voltage regulators reference voltage. Interconnection Customer shall provide validated test recordings showing the responses of Large Generating Facility terminal and field voltages. In the event that direct recordings of these voltages is impractical, recordings of other voltages or currents that mirror the response of the Large Generating Facility's terminal or field voltage are acceptable if information necessary to translate these alternate quantities to actual Large Generating Facility terminal or field voltages is provided. Large Generating Facility testing shall be conducted and results provided to Transmission Provider for each individual generating unit in a station. Subsequent to the Operation Date, Interconnection Customer shall provide Transmission Provider any information changes due to equipment replacement, repair, or adjustment. Transmission Provider shall provide Interconnection Customer any information changes due to equipment replacement, repair or adjustment in the directly connected substation or any adjacent Transmission Provider-owned substation that may affect Interconnection Customer's Interconnection Facilities equipment ratings, protection or operating requirements. The Parties shall provide such information no later than thirty (30) Calendar Days after the date of the equipment replacement, repair or adjustment.

Article 25. Information Access and Audit Rights

- **25.1 Information Access**. Each Party (the "disclosing Party") shall make available to the other Party information that is in the possession of the disclosing Party and is necessary in order for the other Party to: (i) verify the costs incurred by the disclosing Party for which the other Party is responsible under this LGIA; and (ii) carry out its obligations and responsibilities under this LGIA. The Parties shall not use such information for purposes other than those set forth in this Article 25.1 and to enforce their rights under this LGIA.
- **Reporting of Non-Force Majeure Events**. Each Party (the "notifying Party") shall notify the other Party when the notifying Party becomes aware of its inability to comply with the provisions of this LGIA for a reason other than a Force Majeure event. The Parties agree to cooperate with each other and provide

necessary information regarding such inability to comply, including the date, duration, reason for the inability to comply, and corrective actions taken or planned to be taken with respect to such inability to comply. Notwithstanding the foregoing, notification, cooperation or information provided under this article shall not entitle the Party receiving such notification to allege a cause for anticipatory breach of this LGIA.

25.3 Audit Rights. Subject to the requirements of confidentiality under Article 22 of this LGIA, each Party shall have the right, during normal business hours, and upon prior reasonable notice to the other Party, to audit at its own expense the other Party's accounts and records pertaining to either Party's performance or either Party's satisfaction of obligations under this LGIA. Such audit rights shall include audits of the other Party's costs, calculation of invoiced amounts, Transmission Provider's efforts to allocate responsibility for the provision of reactive support to the Transmission System, Transmission Provider's efforts to allocate responsibility for interruption or reduction of generation on the Transmission System, and each Party's actions in an Emergency Condition. Any audit authorized by this article shall be performed at the offices where such accounts and records are maintained and shall be limited to those portions of such accounts and records that relate to each Party's performance and satisfaction of obligations under this LGIA. Each Party shall keep such accounts and records for a period equivalent to the audit rights periods described in Article 25.4.

25.4 Audit Rights Periods.

25.4.1 Audit Rights Period for Construction-Related Accounts and Records.

Accounts and records related to the design, engineering, procurement, and construction of Transmission Provider's Interconnection Facilities and Network Upgrades shall be subject to audit for a period of twenty-four months following Transmission Provider's issuance of a final invoice in accordance with Article 12.2.

25.4.2 Audit Rights Period for All Other Accounts and Records.

Accounts and records related to either Party's performance or satisfaction of all obligations under this LGIA other than those described in Article 25.4.1 shall be subject to audit as follows: (i) for an audit relating to cost obligations, the applicable audit rights period shall be twenty-four months after the auditing Party's receipt of an invoice giving rise to such cost obligations; and (ii) for an audit relating to all other obligations, the applicable audit rights period shall be twenty-four months after the event for which the audit is sought.

25.5 Audit Results. If an audit by a Party determines that an overpayment or an underpayment has occurred, a notice of such overpayment or underpayment shall be given to the other Party together with those records from the audit which support such determination.

Article 26. Subcontractors

- **26.1 General**. Nothing in this LGIA shall prevent a Party from utilizing the services of any subcontractor as it deems appropriate to perform its obligations under this LGIA; provided, however, that each Party shall require its subcontractors to comply with all applicable terms and conditions of this LGIA in providing such services and each Party shall remain primarily liable to the other Party for the performance of such subcontractor.
- **26.2 Responsibility of Principal**. The creation of any subcontract relationship shall not relieve the hiring Party of any of its obligations under this LGIA. The hiring Party shall be fully responsible to the other Party for the acts or omissions of any subcontractor the hiring Party hires as if no subcontract had been made; provided, however, that in no event shall Transmission Provider be liable for the actions or inactions of Interconnection Customer or its subcontractors with respect to obligations of Interconnection Customer under Article 5 of this LGIA. Any applicable obligation imposed by this LGIA upon the hiring Party shall be equally binding upon, and shall be construed as having application to, any subcontractor of such Party.
- **26.3 No Limitation by Insurance**. The obligations under this Article 26 will not be limited in any way by any limitation of subcontractor's insurance.

Article 27. Disputes

27.1 Submission. In the event either Party has a dispute, or asserts a claim, that arises out of or in connection with this LGIA or its performance, such Party (the "disputing Party") shall provide the other Party with written notice of the dispute or claim ("Notice of Dispute"). Such dispute or claim shall be referred to a designated senior representative of each Party for resolution on an informal basis as promptly as practicable after receipt of the Notice of Dispute by the other Party. In the event the designated representatives are unable to resolve the claim or dispute through unassisted or assisted negotiations within thirty (30) Calendar Days of the other Party's receipt of the Notice of Dispute, such claim or dispute may, upon mutual agreement of the Parties, be submitted to arbitration and resolved in accordance with the arbitration procedures set forth below. In the event the Parties do not agree to submit such claim or dispute to arbitration, each

Party may exercise whatever rights and remedies it may have in equity or at law consistent with the terms of this LGIA.

- 27.2 **External Arbitration Procedures.** Any arbitration initiated under this LGIA shall be conducted before a single neutral arbitrator appointed by the Parties. If the Parties fail to agree upon a single arbitrator within ten (10) Calendar Days of the submission of the dispute to arbitration, each Party shall choose one arbitrator who shall sit on a three-member arbitration panel. The two arbitrators so chosen shall within twenty (20) Calendar Days select a third arbitrator to chair the arbitration panel. In either case, the arbitrators shall be knowledgeable in electric utility matters, including electric transmission and bulk power issues, and shall not have any current or past substantial business or financial relationships with any party to the arbitration (except prior arbitration). The arbitrator(s) shall provide each of the Parties an opportunity to be heard and, except as otherwise provided herein, shall conduct the arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("Arbitration Rules") and any applicable FERC regulations or RTO rules; provided, however, in the event of a conflict between the Arbitration Rules and the terms of this Article 27, the terms of this Article 27 shall prevail.
- 27.3 Arbitration Decisions. Unless otherwise agreed by the Parties, the arbitrator(s) shall render a decision within ninety (90) Calendar Days of appointment and shall notify the Parties in writing of such decision and the reasons therefor. The arbitrator(s) shall be authorized only to interpret and apply the provisions of this LGIA and shall have no power to modify or change any provision of this Agreement in any manner. The decision of the arbitrator(s) shall be final and binding upon the Parties, and judgment on the award may be entered in any court having jurisdiction. The decision of the arbitrator(s) may be appealed solely on the grounds that the conduct of the arbitrator(s), or the decision itself, violated the standards set forth in the Federal Arbitration Act or the Administrative Dispute Resolution Act. The final decision of the arbitrator must also be filed with FERC if it affects jurisdictional rates, terms and conditions of service, Interconnection Facilities, or Network Upgrades.
- **27.4 Costs**. Each Party shall be responsible for its own costs incurred during the arbitration process and for the following costs, if applicable: (1) the cost of the arbitrator chosen by the Party to sit on the three member panel and one half of the cost of the third arbitrator chosen; or (2) one half the cost of the single arbitrator jointly chosen by the Parties.

Article 28. Representations, Warranties, and Covenants

28.1 General. Each Party makes the following representations, warranties and

covenants:

- 28.1.1 Good Standing. Such Party is duly organized, validly existing and in good standing under the laws of the state in which it is organized, formed, or incorporated, as applicable; that it is qualified to do business in the state or states in which the Large Generating Facility, Interconnection Facilities and Network Upgrades owned by such Party, as applicable, are located; and that it has the corporate power and authority to own its properties, to carry on its business as now being conducted and to enter into this LGIA and carry out the transactions contemplated hereby and perform and carry out all covenants and obligations on its part to be performed under and pursuant to this LGIA.
- **28.1.2 Authority**. Such Party has the right, power and authority to enter into this LGIA, to become a Party hereto and to perform its obligations hereunder. This LGIA is a legal, valid and binding obligation of such Party, enforceable against such Party in accordance with its terms, except as the enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting creditors' rights generally and by general equitable principles (regardless of whether enforceability is sought in a proceeding in equity or at law).
- **28.1.3 No Conflict**. The execution, delivery and performance of this LGIA does not violate or conflict with the organizational or formation documents, or bylaws or operating agreement, of such Party, or any judgment, license, permit, order, material agreement or instrument applicable to or binding upon such Party or any of its assets.
- **28.1.4 Consent and Approval**. Such Party has sought or obtained, or, in accordance with this LGIA will seek or obtain, each consent, approval, authorization, order, or acceptance by any Governmental Authority in connection with the execution, delivery and performance of this LGIA, and it will provide to any Governmental Authority notice of any actions under this LGIA that are required by Applicable Laws and Regulations.

Article 29. Joint Operating Committee

29.1 Joint Operating Committee. Except in the case of ISOs and RTOs, Transmission Provider shall constitute a Joint Operating Committee to coordinate operating and technical considerations of Interconnection Service. At least six (6) months prior

to the expected Initial Synchronization Date, Interconnection Customer and Transmission Provider shall each appoint one representative and one alternate to the Joint Operating Committee. Each Interconnection Customer shall notify Transmission Provider of its appointment in writing. Such appointments may be changed at any time by similar notice. The Joint Operating Committee shall meet as necessary, but not less than once each calendar year, to carry out the duties set forth herein. The Joint Operating Committee shall hold a meeting at the request of either Party, at a time and place agreed upon by the representatives. The Joint Operating Committee shall perform all of its duties consistent with the provisions of this LGIA. Each Party shall cooperate in providing to the Joint Operating Committee all information required in the performance of the Joint Operating Committee's duties. All decisions and agreements, if any, made by the Joint Operating Committee, shall be evidenced in writing. The duties of the Joint Operating Committee shall include the following:

- **29.1.1** Establish data requirements and operating record requirements.
- **29.1.2** Review the requirements, standards, and procedures for data acquisition equipment, protective equipment, and any other equipment or software.
- **29.1.3** Annually review the one (1) year forecast of maintenance and planned outage schedules of Transmission Provider's and Interconnection Customer's facilities at the Point of Interconnection.
- **29.1.4** Coordinate the scheduling of maintenance and planned outages on the Interconnection Facilities, the Large Generating Facility and other facilities that impact the normal operation of the interconnection of the Large Generating Facility to the Transmission System.
- **29.1.5** Ensure that information is being provided by each Party regarding equipment availability.
- **29.1.6** Perform such other duties as may be conferred upon it by mutual agreement of the Parties.

Article 30. Miscellaneous

- **30.1 Binding Effect**. This LGIA and the rights and obligations hereof, shall be binding upon and shall inure to the benefit of the successors and assigns of the Parties hereto.
- **30.2** Conflicts. In the event of a conflict between the body of this LGIA and any

attachment, appendices or exhibits hereto, the terms and provisions of the body of this LGIA shall prevail and be deemed the final intent of the Parties.

- 30.3 **Rules of Interpretation**. This LGIA, unless a clear contrary intention appears, shall be construed and interpreted as follows: (1) the singular number includes the plural number and vice versa; (2) reference to any person includes such person's successors and assigns but, in the case of a Party, only if such successors and assigns are permitted by this LGIA, and reference to a person in a particular capacity excludes such person in any other capacity or individually; (3) reference to any agreement (including this LGIA), document, instrument or tariff means such agreement, document, instrument, or tariff as amended or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof; (4) reference to any Applicable Laws and Regulations means such Applicable Laws and Regulations as amended, modified, codified, or reenacted, in whole or in part, and in effect from time to time, including, if applicable, rules and regulations promulgated thereunder; (5) unless expressly stated otherwise, reference to any Article, Section or Appendix means such Article of this LGIA or such Appendix to this LGIA, or such Section to the LGIP or such Appendix to the LGIP, as the case may be; (6) "hereunder", "hereof", "herein", "hereto" and words of similar import shall be deemed references to this LGIA as a whole and not to any particular Article or other provision hereof or thereof; (7) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding such term; and (8) relative to the determination of any period of time, "from" means "from and including", "to" means "to but excluding" and "through" means "through and including".
- 30.4 Entire Agreement. This LGIA, including all Appendices and Schedules attached hereto, constitutes the entire agreement between the Parties with reference to the subject matter hereof, and supersedes all prior and contemporaneous understandings or agreements, oral or written, between the Parties with respect to the subject matter of this LGIA. There are no other agreements, representations, warranties, or covenants which constitute any part of the consideration for, or any condition to, either Party's compliance with its obligations under this LGIA.
- **30.5 No Third Party Beneficiaries**. This LGIA is not intended to and does not create rights, remedies, or benefits of any character whatsoever in favor of any persons, corporations, associations, or entities other than the Parties, and the obligations herein assumed are solely for the use and benefit of the Parties, their successors in interest and, where permitted, their assigns.
- **30.6 Waiver**. The failure of a Party to this LGIA to insist, on any occasion, upon strict performance of any provision of this LGIA will not be considered a waiver of any obligation, right, or duty of, or imposed upon, such Party.

Any waiver at any time by either Party of its rights with respect to this LGIA shall not be deemed a continuing waiver or a waiver with respect to any other failure to comply with any other obligation, right, duty of this LGIA. Termination or Default of this LGIA for any reason by Interconnection Customer shall not constitute a waiver of Interconnection Customer's legal rights to obtain an interconnection from Transmission Provider. Any waiver of this LGIA shall, if requested, be provided in writing.

- **30.7 Headings**. The descriptive headings of the various Articles of this LGIA have been inserted for convenience of reference only and are of no significance in the interpretation or construction of this LGIA.
- **30.8 Multiple Counterparts**. This LGIA may be executed in two or more counterparts, each of which is deemed an original but all constitute one and the same instrument.
- **30.9 Amendment**. The Parties may by mutual agreement amend this LGIA by a written instrument duly executed by the Parties.
- **30.10 Modification by the Parties**. The Parties may by mutual agreement amend the Appendices to this LGIA by a written instrument duly executed by the Parties. Such amendment shall become effective and a part of this LGIA upon satisfaction of all Applicable Laws and Regulations.
- 30.11 Reservation of Rights. Transmission Provider shall have the right to make a unilateral filing with FERC to modify this LGIA with respect to any rates, terms and conditions, charges, classifications of service, rule or regulation under section 205 or any other applicable provision of the Federal Power Act and FERC's rules and regulations thereunder, and Interconnection Customer shall have the right to make a unilateral filing with FERC to modify this LGIA pursuant to section 206 or any other applicable provision of the Federal Power Act and FERC's rules and regulations thereunder; provided that each Party shall have the right to protest any such filing by the other Party and to participate fully in any proceeding before FERC in which such modifications may be considered. Nothing in this LGIA shall limit the rights of the Parties or of FERC under sections 205 or 206 of the Federal Power Act and FERC's rules and regulations thereunder, except to the extent that the Parties otherwise mutually agree as provided herein.
- **30.12 No Partnership**. This LGIA shall not be interpreted or construed to create an association, joint venture, agency relationship, or partnership between the Parties or to impose any partnership obligation or partnership liability upon either Party. Neither Party shall have any right, power or authority to enter into any agreement

or undertaking for, or act on behalf of, or to act as or be an agent or representative of, or to otherwise bind, the other Party.

IN WITNESS WHEREOF, the Parties have executed this LGIA in duplicate originals, each of which shall constitute and be an original effective Agreement between the Parties.

[Insert name of Transmission Provider or Transmission Owner, if applicable]		
By:		By:
Title:		Title:
Date:		Date:
[Insert name of Interconnection Customer]		
By:		
Title:		
Date:		

Appendices: Interconnection Facilities, Network Upgrades and Distribution Upgrade

Appendices to LGIA

Appendix A Interconnection Facilities, Network Upgrades, and Distribution

Upgrades

Appendix B Milestones

Appendix C Interconnection Details

Appendix D Security Arrangements Details

Appendix E Commercial Operation Date

Appendix F Addresses for Delivery of Notices and Billings

Appendix G Interconnection Requirements for a Wind Generating Plant

Appendix A: Interconnection Facilities, Network Upgrades and Distribution **Upgrades**

Appendix A to LGIA

Interconnection Facilities, Network Upgrades and Distribution Upgrades

- **Interconnection Facilities:** 1.
 - (a) [insert Interconnection Customer's Interconnection Facilities]:
 - (b) [insert Transmission Provider's Interconnection Facilities]:
- **Network Upgrades:** 2.
 - (a) [insert Stand Alone Network Upgrades]:
 - (b) [insert Other Network Upgrades]:
- **Distribution Upgrades: 3.**

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Appendix B: Milestones

Appendix B to LGIA
Milestones

Appendix C: Interconnection Details

Appendix C to LGIA

Interconnection Details¹

¹ This Appendix C identifies the facilities required for an Interconnection Customer's Interconnection Request. If the Interconnection Customer's Interconnection Request was studied as part of a Queue Cluster, the costs for Network Upgrades associated with the specific Queue Cluster will be allocated pursuant to Section 4.2.2 of the Large Generator Interconnection Procedures in Attachment M of EPE's OATT.

Appendix D: Security Arrangements Details

Appendix D to LGIA

Security Arrangements Details

Infrastructure security of Transmission System equipment and operations and control hardware and software is essential to ensure day-to-day Transmission System reliability and operational security. FERC will expect all entities, including Transmission Providers, market participants, and Interconnection Customers interconnected to the Transmission System to comply with the Reliability Standards promulgated by FERC and administered by the North American Electric Reliability Corporation (NERC) and associated Regional Reliability Organizations (RRO). In EPE's service territory, the RRO is the Western Electricity Coordinating Council (WECC).

Appendix E: Commercial Operation Date

Appendix E to LGIA

Commercial Operation Date

This Appendix E is a part of the LGIA between Transmission Provider and Interconnection Customer.

	[Date]	
	[Transmission Provider Address]	
	Re: Large Generating Facility	
	Dear:	
On [Date] [Interconnection Customer] has completed Trial Operation This letter confirms that [Interconnection Customer] commenced Operation of Unit No at the Large Generating Facility, effective as of [one day].		
	Thank you.	
	[Signature]	
	[Interconnection Customer Representative]	

Appendix F: Addresses for Delivery of Notices and Billings

Appendix F to LGIA

	Addresses for Delivery of Notices and Billings
Notices:	
Tra	ansmission Provider:
[To	be supplied.]
Inte	erconnection Customer:
[To	be supplied.]
Billings a	nd Payments:
Tra	nsmission Provider:
[To	be supplied.]
Inte	erconnection Customer:
[To	be supplied.]
Alternati	ve Forms of Delivery of Notices (telephone, facsimile or email):
Tra	nnsmission Provider:
[To	be supplied.]
Inte	erconnection Customer:
[To	be supplied.]

Appendix G: Interconnection Requirements for a Wind Generating Plant

Appendix G to LGIA

Interconnection Requirements for a

Wind Generating Plant

Appendix G sets forth requirements and provisions specific to a wind generating plant. All other requirements of this LGIA continue to apply to wind generating plant interconnections.

A. Technical Standards Applicable to a Wind Generating Plant

i. Low Voltage Ride-Through (LVRT) Capability

A wind generating plant shall be able to remain online during voltage disturbances up to the time periods and associated voltage levels set forth in the standard below. The LVRT standard provides for a transition period standard and a post-transition period standard.

Transition Period LVRT Standard

The transition period standard applies to wind generating plants subject to FERC Order 661 that have either: (i) interconnection agreements signed and filed with the Commission, filed with the Commission in unexecuted form, or filed with the Commission as non-conforming agreements between January 1, 2006 and December 31, 2006, with a scheduled in-service date no later than December 31, 2007, or (ii) wind generating turbines subject to a wind turbine procurement contract executed prior to December 31, 2005, for delivery through 2007.

1. Wind generating plants are required to remain in-service during three-phase faults

with normal clearing (which is a time period of approximately 4 - 9 cycles) and single line to ground faults with delayed clearing, and subsequent post-fault voltage recovery to prefault voltage unless clearing the fault effectively disconnects the generator from the system. The clearing time requirement for a three-phase fault will be specific to the wind generating plant substation location, as determined by and documented by the ***Transmission Provider**. The maximum clearing time the wind generating plant shall be required to withstand for a threephase fault shall be 9 cycles at a voltage as low as 0.15 p.u., as measured at the high side of the wind generating plant step-up transformer (i.e. the transformer that steps the voltage up to the transmission interconnection voltage or "GSU"), after which, if the fault remains following the location-specific normal clearing time for three-phase faults, the wind generating plant may disconnect from the transmission system.

- 2. This requirement does not apply to faults that would occur between the wind generator terminals and the high side of the GSU or to faults that would result in a voltage lower than 0.15 per unit on the high side of the GSU serving the facility.
- 3. Wind generating plants may be tripped after the fault period if this action is intended as part of a special protection system.
- Wind generating plants may meet the LVRT requirements of this standard by the 4. performance of the generators or by installing additional equipment (e.g., Static VAr Compensator, etc.) within the wind generating plant or by a combination of generator performance and additional equipment.

1.

5. Existing individual generator units that are, or have been, interconnected to the network at the same location at the effective date of the Appendix G LVRT Standard are exempt from meeting the Appendix G LVRT Standard for the remaining life of the existing generation equipment. Existing individual generator units that are replaced are required to meet the Appendix G LVRT Standard.

Post-transition Period LVRT Standard

All wind generating plants subject to FERC Order No. 661 and not covered by the transition period described above must meet the following requirements:

- Wind generating plants are required to remain in-service during three-phase faults with normal clearing (which is a time period of approximately 4 - 9 cycles) and single line to ground faults with delayed clearing, and subsequent post-fault voltage recovery to prefault voltage unless clearing the fault effectively disconnects the generator from the system. The clearing time requirement for a three-phase fault will be specific to the wind generating plant substation location, as determined by and documented by the ***Transmission Provider**. The maximum clearing time the wind generating plant shall be required to withstand for a threephase fault shall be 9 cycles after which, if the fault remains following the location-specific normal clearing time for three-phase faults, the wind generating plant may disconnect from the transmission system. A wind generating plant shall remain interconnected during such a fault on the transmission system for a voltage level as low as zero volts, as measured at the high voltage side of the wind GSU.
- 2. This requirement does not apply to faults that would occur between the wind

- generator terminals and the high side of the GSU.
- 3. Wind generating plants may be tripped after the fault period if this action is intended as part of a special protection system.
- 4. Wind generating plants may meet the LVRT requirements of this standard by the performance of the generators or by installing additional equipment (e.g., Static VAr Compensator) within the wind generating plant or by a combination of generator performance and additional equipment.
- 5. Existing individual generator units that are, or have been, interconnected to the network at the same location at the effective date of the Appendix G LVRT Standard are exempt from meeting the Appendix G LVRT Standard for the remaining life of the existing generation equipment. Existing individual generator units that are replaced are required to meet the Appendix G LVRT Standard.

ii. Power Factor Design Criteria (Reactive Power)

The following reactive power requirements apply only to a newly interconnecting wind generating plant that has executed a Facilities Study Agreement as of the effective date of the Final Rule establishing the reactive power requirements for non-synchronous generators in section 9.6.1 of this LGIA (Order No. 827). A wind generating plant to which this provision applies shall maintain a power factor within the range of 0.95 leading to 0.95 lagging, measured at the Point of Interconnection as defined in this LGIA, if the Transmission Provider's System Impact Study shows that such a requirement is necessary to ensure safety or reliability. The power factor range standard can be met by using, for example, power electronics designed to supply this level of reactive capability

(taking into account any limitations due to voltage level, real power output, etc.) or fixed and switched capacitors if agreed to by the Transmission Provider, or a combination of the two. The Interconnection Customer shall not disable power factor equipment while the wind plant is in operation. Wind plants shall also be able to provide sufficient dynamic voltage support in lieu of the power system stabilizer and automatic voltage regulation at the generator excitation system if the System Impact Study shows this to be required for system safety or reliability.

iii. Supervisory Control and Data Acquisition (SCADA) Capability

The wind plant shall provide SCADA capability to transmit data and receive instructions from the Transmission Provider to protect system reliability. The Transmission Provider and the wind plant Interconnection Customer shall determine what SCADA information is essential for the proposed wind plant, taking into account the size of the plant and its characteristics, location, and importance in maintaining generation resource adequacy and transmission system reliability in its area.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Appendix 7 INTERCONNECTION PROCEDURES FOR A WIND GENERATING PLANT

APPENDIX 7 TO LGIP

INTERCONNECTION PROCEDURES FOR A WIND GENERATING PLANT

Appendix G sets forth procedures specific to a wind generating plant. All other requirements of this LGIP continue to apply to wind generating plant interconnections.

A. Special Procedures Applicable to Wind Generators

The wind plant Interconnection Customer, in completing the Interconnection

Request required by section 3.3 of this LGIP, may provide to the Transmission Provider a set of preliminary electrical design specifications depicting the wind plant as a single equivalent generator. Upon satisfying these and other applicable Interconnection Request conditions, the wind plant may enter the queue and receive the base case data as provided for in this LGIP.

No later than six months after submitting an Interconnection Request completed in this manner, the wind plant Interconnection Customer must submit completed detailed electrical design specifications and other data (including collector system layout data) needed to allow the Transmission Provider to complete the System Impact study.

ATTACHMENT M

Large Generator Interconnection Procedures and Agreement

Standard Large Generator Interconnection Procedures (LGIP)

(Applicable to Generating Facilities that exceed 20 MW)

TABLE OF CONTENTS

Section 1.	Definitions
Section 2. 2.1 2.2 2.3 2.4	Scope and Application Application of Standard Large Generator Interconnection Procedures Comparability Base Case Data No Applicability to Transmission Service
Section 3.	Interconnection Requests
3.1	General
3.2	Identification of Types of Interconnection Services
	3.2.1 Energy Resource Interconnection Service
	3.2.1.1 The Product
	3.2.1.2 The Study
	3.2.2 Network Resource Interconnection Service
	3.2.2.1 The Product
	3.2.2.2 The Study
3.3	Utilization of Surplus Interconnection Service
	3.3.1 Surplus Interconnection Service Requests
3.4	Valid Interconnection Request
	3.4.1 Initiating an Interconnection Request
	3.4.2 Acknowledgment of Interconnection Request
	3.4.3 Deficiencies in Interconnection Request
3.5	3.4.4 Scoping Meeting
3.3	OASIS Posting 3.5.1 OASIS Posting
	3.5.2 Requirement to Post Interconnection Study Metrics
	3.5.2.1 Interconnection System Impact Studies Processing Time
	3.5.2.2 Interconnection Facilities Studies Processing Time
	3.5.2.3 Interconnection Service Requests Withdrawn from
	Interconnection Queue
3.6	Coordination with Affected Systems
3.7	Withdrawal
	3.7.1 Withdrawal Penalty
	3.7.1.1 Calculation of the Withdrawal Penalty
	3.7.1.2 Distribution of the Withdrawal Penalty
3.8	Identification of Contingent Facilities
Section 4.	Interconnection Request Evaluation Process
4.1	Queue Position

4.1.1 Assignment of Queue Position

4.2 4.3 4.4	4.1.2 Higher Queue Position Clustering and the Definitive Study Process 4.2.1 Initiation of a Resource Solicitation Cluster 4.2.2 Network Upgrades Cost Allocation Transferability of Queue Position Modifications
Section 5.	Procedures for Interconnection Requests Submitted Prior to Effective Date of Standard Large Generator Interconnection Procedures
5.1 5.2	Transition to the First-Ready, First-Served Process New Transmission Provider
Section 6.	Informational Interconnection Study
6.1	Informational Interconnection Study Agreement
6.2	Scope of Informational Interconnection Study
6.3	Informational Interconnection Study Procedures
Section 7.	Definitive Interconnection System Impact Study
7.1	Definitive Interconnection System Impact Study Agreement
7.2	Execution of Definitive Interconnection System Impact Study Agreement
7.3	Scope of Definitive Interconnection System Impact Study
7.4	DefinitiveInterconnection System Impact Study Procedures
7.5	Meeting with Transmission Provider
7.6	Re-Study
7.7	Readiness Milestones and Site Control
	7.7.1 Readiness Milestone 1 ("M1")
	7.7.2 Readiness Milestone 2 ("M2")
	7.7.3 Readiness Milestone 3 ("M3")
	7.7.4 Readiness Milestone 4 ("M3")
	7.7.5 Security Requirements
	7.7.6 [Reserved for Future Use]
	7.7.7 Site Control
Section 8.	Interconnection Facilities Study
8.1	Interconnection Facilities Study Agreement
8.2	Scope of Interconnection Facilities Study
8.3	Interconnection Facilities Study Procedures
8.4	Meeting with Transmission Provider
8.5	Re-Study

Section 9. Engineering & Procurement ('E&P') Agreement

Section 10. [Reserved for Future Use]

- Section 11. Standard Large Generator Interconnection Agreement (LGIA)
 - 11.1 Tender
 - 11.2 Negotiation
 - 11.3 Execution and Filing
 - 11.4 Commencement of Interconnection Activities
- Section 12. Construction of Transmission Provider's Interconnection Facilities and Network Upgrades
 - Schedule 12.1
 - 12.2 Construction Sequencing
 - 12.2.1 General
 - 12.2.2 Advance Construction of Network Upgrades that are an Obligation of an Entity other than Interconnection Customer
 - 12.2.3 Advancing Construction of Network Upgrades that are Part of an Expansion Plan of the Transmission Provider
 - 12.2.4 Amended Interconnection System Impact Study
- Section 13. Miscellaneous
 - 13.1 Confidentiality
 - 13.1.1 Scope
 - 13.1.2 Release of Confidential Information
 - 13.1.3 Rights
 - 13.1.4 No Warranties
 - 13.1.5 Standard of Care
 - 13.1.6 Order of Disclosure
 - 13.1.7 Remedies
 - 13.1.8 Disclosure to FERC, its Staff, or a State
 - 13.2 Delegation of Responsibility
 - Obligation for Study Costs and Withdrawal Penalty 13.3
 - Third Parties Conducting Studies 13.4
 - 13.5 Disputes
 - 13.5.1 Submission
 - 13.5.2 External Arbitration Procedures
 - 13.5.3 Arbitration Decisions
 - 13.5.4 Costs
 - 13.5.5 Non-binding dispute resolution procedures
 - 13.6 Local Furnishing Bonds
 - 13.6.1 Transmission Providers That Own Facilities Financed by Local Furnishing Bonds
 - 13.6.2 Alternative Procedures for Requesting Interconnection Service

- Appendix 1 Interconnection Request for a Large Generating Facility
- Appendix 2 Definitive Interconnection System Impact Study Agreement
- Appendix 3 Interconnection Facilities Study Agreement
- Appendix 4 Transitional Study Agreements
 - Transitional Facilities Interconnection Study Agreement 4.1
 - Transitional Cluster Study Agreement 4.2
- Appendix 5 Informational Study Forms
 - Informational Interconnection Study Request 5.1
 - Informational Interconnection Study Agreement 5.2

Appendix 6 - Standard Large Generator Interconnection Agreement

Section 1. Definitions

Adverse System Impact shall mean the negative effects due to technical or operational limits on conductors or equipment being exceeded that may compromise the safety and reliability of the electric system.

Affected System shall mean an electric system other than the Transmission Provider's Transmission System that may be affected by the proposed interconnection.

Affected System Operator shall mean the entity that operates an Affected System.

Affiliate shall mean, with respect to a corporation, partnership or other entity, each such other corporation, partnership or other entity that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such corporation, partnership or other entity.

Ancillary Services shall mean those services that are necessary to support the transmission of capacity and energy from resources to loads while maintaining reliable operation of the Transmission Provider's Transmission System in accordance with Good Utility Practice.

Applicable Laws and Regulations shall mean all duly promulgated applicable federal, state and local laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders, permits and other duly authorized actions of any Governmental Authority.

Applicable Reliability Council shall mean the reliability council applicable to the Transmission System to which the Generating Facility is directly interconnected.

Applicable Reliability Standards shall mean the requirements and guidelines of NERC, the Applicable Reliability Council, and the Control Area of the Transmission System to which the Generating Facility is directly interconnected.

Base Case shall mean the base case power flow, short circuit, and stability data bases used for the Interconnection Studies by the Transmission Provider or Interconnection Customer.

Breach shall mean the failure of a Party to perform or observe any material term or condition of the Standard Large Generator Interconnection Agreement.

Breaching Party shall mean a Party that is in Breach of the Standard Large Generator Interconnection Agreement.

Business Day shall mean Monday through Friday, excluding Federal Holidays.

Calendar Day shall mean any day including Saturday, Sunday or a Federal Holiday.

Cluster shall mean a group of Interconnection Requests (one or more) that are studied together for the purpose of conducting Interconnection Studies.

Cluster Study shall mean an Interconnection Study evaluating one or more Interconnection Requests.

Clustering shall mean the process whereby a group of Interconnection Requests is studied together, instead of serially, for the purpose of conducting Interconnection Studies.

Commercial Operation shall mean the status of a Generating Facility that has commenced generating electricity for sale, excluding electricity generated during Trial Operation.

Commercial Operation Date of a unit shall mean the date on which the Generating Facility commences Commercial Operation as agreed to by the Parties pursuant to Appendix E to the Standard Large Generator Interconnection Agreement.

Confidential Information shall mean any confidential, proprietary or trade secret information of a plan, specification, pattern, procedure, design, device, list, concept, policy or compilation relating to the present or planned business of a Party, which is designated as confidential by the Party supplying the information, whether conveyed orally, electronically, in writing, through inspection, or otherwise.

Contingent Facilities shall mean those unbuilt Interconnection Facilities and Network Upgrades upon which the Interconnection Request's costs, timing, and study findings are dependent, and if delayed or not built, could cause a need for Re-Studies of the Interconnection Request or a reassessment of the Interconnection Facilities and/or Network Upgrades and/or costs and timing.

Control Area shall mean an electrical system or systems bounded by interconnection metering and telemetry, capable of controlling generation to maintain its interchange schedule with other Control Areas and contributing to frequency regulation of the interconnection. A Control Area must be certified by an Applicable NERC Regional Reliability Entity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC.

Customer Engagement Window shall have the meaning set forth in Section 4.2 of

the LGIP.

Default shall mean the failure of a Breaching Party to cure its Breach in accordance with Article 17 of the Standard Large Generator Interconnection Agreement.

Definitive Interconnection Study Process ("Definitive Interconnection Study") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study.

Definitive Interconnection System Impact Study ("DISIS") shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System.

Definitive Interconnection System Impact Study Agreement ("DISIS Agreement") shall mean the form of agreement contained in the Appendices of the LGIP for conducting the Definitive Interconnection System Impact Study.

Definitive Interconnection System Impact Study Cluster ("DISIS Cluster") shall mean an engineering study that evaluates the impact of the proposed interconnection(s) on the safety and reliability of the Transmission System and, if applicable, an Affected System.

DISIS Request Window shall have the meaning set forth in Section 4.2 of the LGIP.

Dispute Resolution shall mean the procedure for resolution of a dispute between the Parties in which they will first attempt to resolve the dispute on an informal basis.

Distribution System shall mean the Transmission Provider's facilities and equipment used to transmit electricity to ultimate usage points such as homes and industries directly from nearby generators or from interchanges with higher voltage transmission networks which transport bulk power over longer distances. The voltage levels at which distribution systems operate differ among areas.

Distribution Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Distribution System at or beyond the Point of Interconnection to facilitate interconnection of the Generating Facility and render the transmission service necessary to effect Interconnection Customer's wholesale sale of electricity in interstate commerce. Distribution Upgrades do not include Interconnection Facilities.

Effective Date shall mean the date on which the Standard Large Generator Interconnection Agreement becomes effective upon execution by the Parties subject to acceptance by FERC, or if filed unexecuted, upon the date specified by FERC.

Emergency Condition shall mean a condition or situation: (1) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (2) that, in the case of a Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to Transmission Provider's Transmission System, Transmission Provider's Interconnection Facilities or the electric systems of others to which the Transmission Provider's Transmission System is directly connected; or (3) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Generating Facility or Interconnection Customer's Interconnection Facilities. System restoration and black start shall be considered Emergency Conditions; provided that Interconnection Customer is not obligated by the Standard Large Generator Interconnection Agreement to possess black start capability.

Energy Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to connect its Generating Facility to the Transmission Provider's Transmission System to be eligible to deliver the Generating Facility's electric output using the existing firm or nonfirm capacity of the Transmission Provider's Transmission System on an as available basis. Energy Resource Interconnection Service in and of itself does not convey transmission service.

Engineering & Procurement (E&P) Agreement shall mean an agreement that authorizes the Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection in order to advance the implementation of the Interconnection Request.

Environmental Law shall mean Applicable Laws or Regulations relating to pollution or protection of the environment or natural resources.

Federal Power Act shall mean the Federal Power Act, as amended, 16 U.S.C. §§ 791a et seq.

FERC shall mean the Federal Energy Regulatory Commission (Commission) or its successor.

Force Majeure shall mean any act of God, labor disturbance, act of the public enemy, war, insurrection, riot, fire, storm or flood, explosion, breakage or accident to machinery or equipment, any order, regulation or restriction imposed by governmental,

military or lawfully established civilian authorities, or any other cause beyond a Party's control. A Force Majeure event does not include acts of negligence or intentional wrongdoing by the Party claiming Force Majeure.

Generating Facility shall mean Interconnection Customer's device for the production and/or storage for later injection of electricity identified in the Interconnection Request, but shall not include the Interconnection Customer's Interconnection Facilities.

Generating Facility Capacity shall mean the net capacity of the Generating Facility and the aggregate net capacity of the Generating Facility where it includes multiple energy production devices.

Good Utility Practice shall mean any of the practices, methods and acts engaged in or approved by a significant portion of the electric industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act to the exclusion of all others, but rather to be acceptable practices, methods, or acts generally accepted in the region.

Governmental Authority shall mean any federal, state, local or other governmental regulatory or administrative agency, court, commission, department, board, or other governmental subdivision, legislature, rulemaking board, tribunal, or other governmental authority having jurisdiction over the Parties, their respective facilities, or the respective services they provide, and exercising or entitled to exercise any administrative, executive, police, or taxing authority or power; provided, however, that such term does not include Interconnection Customer, Transmission Provider, or any Affiliate thereof.

Hazardous Substances shall mean any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "radioactive substances," "contaminants," "pollutants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, or any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

Informational Interconnection Study shall mean an analysis based on assumptions specified by Interconnection Customer in the Informational Interconnection Study Agreement.

Informational Interconnection Study Agreement shall mean the form of

agreement contained in Appendix 5 of the LGIP for conducting the Informational Interconnection Study.

Initial Deposit shall mean a deposit of \$160,000 for an Interconnection Request of up to 75 MW, or \$250,000 for an Interconnection Request of 75 MW or greater, that Interconnection Customer submits to Transmission Provider upon initiating an Interconnection Request.

Initial Synchronization Date shall mean the date upon which the Generating Facility is initially synchronized and upon which Trial Operation begins.

In-Service Date shall mean the date upon which the Interconnection Customer reasonably expects it will be ready to begin use of the Transmission Provider's Interconnection Facilities to obtain back feed power.

Interconnection Customer shall mean any entity, including the Transmission Provider, Transmission Owner or any of the Affiliates or subsidiaries of either, that proposes to interconnect its Generating Facility with the Transmission Provider's Transmission System.

Interconnection Customer's Interconnection Facilities shall mean all facilities and equipment, as identified in Appendix A of the Standard Large Generator Interconnection Agreement, that are located between the Generating Facility and the Point of Change of Ownership, including any modification, addition, or upgrades to such facilities and equipment necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Customer's Interconnection Facilities are sole use facilities.

Interconnection Facilities shall mean the Transmission Provider's Interconnection Facilities and the Interconnection Customer's Interconnection Facilities. Collectively, Interconnection Facilities include all facilities and equipment between the Generating Facility and the Point of Interconnection, including any modification, additions or upgrades that are necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Interconnection Facilities Study shall mean a study conducted by the Transmission Provider or a third party consultant for the Interconnection Customer to determine a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades as identified in the Definitive Interconnection System Impact Study), the cost of those facilities, and the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System. The scope of the study is

defined in Section 8 of the Standard Large Generator Interconnection Procedures.

Interconnection Facilities Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Facilities Study.

Interconnection Request shall mean an Interconnection Customer's request, in the form of Appendix 1 to the Standard Large Generator Interconnection Procedures, in accordance with the Tariff, to interconnect a new Generating Facility, or to increase the capacity of, or make a Material Modification to the operating characteristics of, an existing Generating Facility that is interconnected with the Transmission Provider's Transmission System.

Interconnection Service shall mean the service provided by the Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to the Transmission Provider's Transmission System and enabling it to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Standard Large Generator Interconnection Agreement and, if applicable, the Transmission Provider's Tariff.

Interconnection Study shall mean any of the following studies: the Informational Interconnection Study, the Definitive Interconnection System Impact Study, and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection Study Agreement shall mean any of the following studies: the Informational Interconnection Study Agreement, the Definitive Interconnection System Impact Study and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

IRS shall mean the Internal Revenue Service.

Joint Operating Committee shall be a group made up of representatives from Interconnection Customers and the Transmission Provider to coordinate operating and technical considerations of Interconnection Service.

Large Generating Facility shall mean a Generating Facility having a Generating Facility Capacity of more than 20 MW.

Loss shall mean any and all losses relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney

fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's performance, or non performance of its obligations under the Standard Large Generator Interconnection Agreement on behalf of the Indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the Indemnifying Party.

Material Modification shall mean those modifications that have a material impact on the cost or timing of any Interconnection Request with an equal or later Queue Position.

Metering Equipment shall mean all metering equipment installed or to be installed at the Generating Facility pursuant to the Standard Large Generator Interconnection Agreement at the metering points, including but not limited to instrument transformers, MWh-meters, data acquisition equipment, transducers, remote terminal unit, communications equipment, phone lines, and fiber optics.

NERC shall mean the North American Electric Reliability Council or its successor organization.

Network Resource shall mean any designated generating resource owned, purchased, or leased by a Network Customer under the Network Integration Transmission Service Tariff. Network Resources do not include any resource, or any portion thereof, that is committed for sale to third parties or otherwise cannot be called upon to meet the Network Customer's Network Load on a non-interruptible basis.

Network Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to integrate its Large Generating Facility with the Transmission Provider's Transmission System (1) in a manner comparable to that in which the Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an RTO or ISO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service in and of itself does not convey transmission service.

Network Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Transmission System required at or beyond the point at which the Interconnection Facilities connect to the Transmission Provider's Transmission System to accommodate the interconnection of the Large Generating Facility to the Transmission Provider's Transmission System.

Notice of Dispute shall mean a written notice of a dispute or claim that arises out of or in connection with the Standard Large Generator Interconnection Agreement or its performance.

OASIS shall mean the Transmission Provider's Open Access Same-Time Information System.

Party or Parties shall mean Transmission Provider, Transmission Owner, Interconnection Customer or any combination of the above.

Permissible Technological Advancement shall mean a modification to a specific technology type submitted in an Interconnection Customer's Interconnection Request that: (1) is not a Material Modification; (2) is not a change in generation technology or fuel type, but may include advancements to other technology such as, e.g., advancements to turbines, inverters, plant supervisory controls, or other technological advancements that may affect a generating facility's ability to produce ancillary services; (3) results in electrical performance that is equal to or better than the electrical performance expected prior to the technology change; (4) does not degrade the electrical characteristics of the generating equipment (e.g., the ratings, impedances, efficiencies, capabilities, and performance of the equipment under steady state and dynamic conditions); (5) does not increase the Interconnection Customer's requested Interconnection Service and (6) does not cause any reliability concerns (i.e., materially impact the transmission system with regard to short circuit capability limits, steady-state thermal or voltage limits or dynamic system stability and response).

Phase ("Phase 1, Phase 2, Phase 3 or Phase 4") shall mean a distinct part of the Definitive Study Process as described in Section 7.

Point of Change of Ownership shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Customer's Interconnection Facilities connect to the Transmission Provider's Interconnection Facilities.

Point of Interconnection shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Facilities connect to the Transmission Provider's Transmission System.

Provisional Interconnection Service shall mean Interconnection Service provided by Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to Transmission Provider's Transmission System and enabling that Transmission System to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Provisional Large Generator Interconnection Agreement and, if applicable, the Tariff.

Provisional Large Generator Interconnection Agreement shall mean the interconnection agreement for Provisional Interconnection Service established

between Transmission Provider and/or the Transmission Owner and the Interconnection Customer. This agreement shall take the form of the Large Generator Interconnection Agreement, modified for provisional purposes.

Queue shall mean a queue for valid Interconnection Requests for the Definitive Interconnection Study Process.

Queue Position shall mean the order of a valid Interconnection Request, relative to all other pending valid Interconnection Requests, in the Definitive Interconnection Study Process. The Queue Position is established based upon the date and time the Interconnection Customer satisfied all of the requirements of this Attachment M to enter the Definitive Study Process.

Reasonable Efforts shall mean, with respect to an action required to be attempted or taken by a Party under the Standard Large Generator Interconnection Agreement, efforts that are timely and consistent with Good Utility Practice and are otherwise substantially equivalent to those a Party would use to protect its own interests.

Resource Plan shall mean any process authorized or required by Applicable Laws and Regulations for, *inter alia*, the selection of Generating Facilities.

Resource Planning Entity shall mean any entity required to develop a Resource Plan or conduct a Resource Solicitation Process.

Resource Solicitation Cluster shall mean a Cluster Study associated with a Resource Plan or related process.

Resource Solicitation Process shall mean any process authorized or required by Applicable Laws and Regulations for the acquisition of Network Resources.

Scoping Meeting shall mean the meeting between representatives of the Interconnection Customer and Transmission Provider conducted for the purpose of discussing the proposed interconnection request, alternative interconnection options, to exchange information including any transmission data and earlier study evaluations that would be reasonably expected to affect such interconnection options, to analyze such information, and to determine the potential feasible Points of Interconnection.

Site Control shall mean the exclusive land right to develop, construct, operate, ad maintain the Generating Facility over the term of expected operation of the Generating Facility. Site Control shall include the right to develop, construct, operate and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by documentation establishing: (1) ownership of, a leasehold interest in, or a right to develop a site of sufficient sized to construct and operate the Generating Facility and

associated Interconnection Facilities; (2) an option to purchase or acquire a leasehold interest in a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities; or (3) any other documentation that clearly demonstrates the right of the Interconnection Customer to exclusively occupy a site of sufficient size to construct and operate the Generating Facility. Site Control f or any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all co-located projects that meet the aforementioned provisions of this Site Control definition.

Small Generating Facility shall mean a Generating Facility that has a Generating Facility Capacity of no more than 20 MW.

Stand Alone Network Upgrades shall mean Network Upgrades that are not part of an Affected System that an Interconnection Customer may construct without affecting day-to-day operations of the Transmission System during their construction. Both the Transmission Provider and the Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify them in Appendix A to the Standard Large Generator Interconnection Agreement. If the Transmission Provider and Interconnection Customer disagree about whether a particular Network Upgrade is a Stand Alone Network Upgrade, the Transmission Provider must provide the Interconnection Customer a written technical explanation outlining why the Transmission Provider does not consider the Network Upgrade to be a Stand Alone Network Upgrade within 15 days of its determination.

Standard Engineering and Procurement Agreement shall mean the form of engineering and procurement agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Agreement (LGIA) shall mean the form of interconnection agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Procedures (LGIP) shall mean the interconnection procedures applicable to an Interconnection Request pertaining to a Large Generating Facility that are included in the Transmission Provider's Tariff.

Start Date shall mean that actual date of the start of the System Impact Study as evidenced by the initiation of the development of the required Base Case(s), or the initiation of the technical study work, i.e., start of the powerflow/stability computer runs, whichever is applicable.

Surplus Interconnection Service shall mean any unneeded portion of Interconnection Service established in a Large Generator Interconnection Agreement, such that if Surplus Interconnection Service is utilized, the total amount of Interconnection Service at the Point of Interconnection would remain the same.

System Protection Facilities shall mean the equipment, including necessary protection signal communications equipment, required to protect (1) the Transmission Provider's Transmission System from faults or other electrical disturbances occurring at the Generating Facility and (2) the Generating Facility from faults or other electrical system disturbances occurring on the Transmission Provider's Transmission System or on other delivery systems or other generating systems to which the Transmission Provider's Transmission System is directly connected.

Tariff shall mean the Transmission Provider's Tariff through which open access transmission service and Interconnection Service are offered, as filed with FERC, and as amended or supplemented from time to time, or any successor tariff.

Transmission Owner shall mean an entity that owns, leases or otherwise possesses an interest in the portion of the Transmission System at the Point of Interconnection and may be a Party to the Standard Large Generator Interconnection Agreement to the extent necessary.

Transmission Provider shall mean the public utility (or its designated agent) that owns, controls, or operates transmission or distribution facilities used for the transmission of electricity in interstate commerce and provides transmission service under the Tariff. The term Transmission Provider should be read to include the Transmission Owner when the Transmission Owner is separate from the Transmission Provider.

Transmission Provider's Interconnection Facilities shall mean all facilities and equipment owned, controlled, or operated by the Transmission Provider from the Point of Change of Ownership to the Point of Interconnection as identified in Appendix A to the Standard Large Generator Interconnection Agreement, including any modifications, additions or upgrades to such facilities and equipment. Transmission Provider's Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Transmission System shall mean the facilities owned, controlled or operated by the Transmission Provider or Transmission Owner that are used to provide transmission service under the Tariff.

Trial Operation shall mean the period during which Interconnection Customer is engaged in on-site test operations and commissioning of the Generating Facility prior to

Commercial Operation.

Withdrawal Penalty shall have the meaning set forth in Section 3.7.1 of the LGIP.

Section 2. Scope and Application

2.1 Application of Standard Large Generator Interconnection Procedures Sections 2 through 13 apply to processing an Interconnection Request pertaining to a Large Generating Facility. Small Generating Facilities requesting NRIS shall be processed under the LGIP.

2.2 Comparability

Transmission Provider shall receive, process and analyze all Interconnection Requests in a timely manner as set forth in this LGIP. Transmission Provider will use the same Reasonable Efforts in processing and analyzing Interconnection Requests from all Interconnection Customers, whether the Generating Facilities are owned by Transmission Provider, its subsidiaries or Affiliates or others.

2.3 Base Case Data

Transmission Provider shall maintain, consistent with Applicable Laws and Regulations and Applicable Reliability Standards, base power flow, short circuit and stability databases, including all underlying assumptions, and contingency list on either its OASIS site or a password-protected website, subject to confidentiality provisions in LGIP Section 13.1. In addition, Transmission Provider shall maintain network models and underlying assumptions on either its OASIS site or a password-protected website. Such network models and underlying assumptions should reasonably represent those used during the most recent Interconnection Study and be representative of current system conditions. If Transmission Provider posts this information on a password-protected website, a link to the information must be provided on Transmission Provider's OASIS site. Transmission Provider is permitted to require that Interconnection Customers, OASIS site users and password-protected website users sign a confidentiality agreement before the release of commercially sensitive information or Critical Energy Infrastructure Information in the Base Case data. Such databases and lists, hereinafter referred to as Base Cases, shall include all (1) generation projects and (2) transmission projects, including merchant transmission projects that are proposed for the Transmission System for which a transmission expansion plan has been submitted and approved by the applicable authority.

2.4 No Applicability to Transmission Service

Nothing in this LGIP shall constitute a request for transmission service or

confer upon an Interconnection Customer any right to receive transmission service.

Section 3. Interconnection Requests

3.1 General

An Interconnection Customer shall submit to Transmission Provider an Interconnection Request in the form of Appendix 1 to this LGIP and the Initial Deposit of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater. Transmission Provider shall apply the Initial Deposit toward the cost of the Definitive Interconnection Study Process. Interconnection Customer shall submit a separate Interconnection Request for each site and may submit multiple Interconnection Requests for a single site. Interconnection Customer must submit an Initial Deposit with each Interconnection Request even when more than one request is submitted for a single site. An Interconnection Request to evaluate one site at two different voltage levels shall be treated as two Interconnection Requests. Similarly, an Interconnection Request to evaluate one interconnection at two or more different sites (two or more Points of Interconnection) shall be treated as separate Interconnection Requests.

At Interconnection Customer's option, Transmission Provider and Interconnection Customer will identify alternative Point(s) of Interconnection and configurations at the Scoping Meeting to evaluate in this process and attempt to eliminate alternatives in a reasonable fashion given resources and information available. Interconnection Customer will select the definitive Point of Interconnection to be studied no later than the execution of the Definitive Interconnection System Impact Study Agreement. For purposes of clustering Interconnection Service requests, Transmission Provider may make reasonable changes to the requested Point(s) of Interconnection to facilitate efficient interconnection of Interconnection Customers at common points of interconnection. Transmission Provider shall notify Interconnection Customers in writing of any intended changes to the requested Point(s) of Interconnection and the Point(s) of Interconnection shall only change upon mutual agreement.

Transmission Provider shall use the process below to consider requests for Interconnection Service below the Generating Facility Capacity. These requests for Interconnection Service shall be studied at the level of Interconnection Service requested for purposes of Interconnection Facilities, Network Upgrades, and associated costs but may be subject to other studies

at the full Generating Facility Capacity to ensure safety and reliability of the system, with the study costs borne by the Interconnection Customer. If after the additional studies are complete, Transmission Provider determines that additional Network Upgrades are necessary, then Transmission Provider must: (1) specify which additional Network Upgrade costs are based on which studies; and (2) provide a detailed explanation of why the additional Network Upgrades are necessary. Any Interconnection Facility and/or Network Upgrade costs required for safety and reliability also will be borne by the Interconnection Customer. Interconnection Customers may be subject to additional control technologies as well as testing and validation of those technologies consistent with Article 6 of the LGIA. The necessary control technologies and protection systems shall be established in Appendix C of the executed, or requested to be filed unexecuted, LGIA.

3.2 Identification of Types of Interconnection Services

At the time the Interconnection Request is submitted, Interconnection Customer must request either Energy Resource Interconnection Service or Network Resource Interconnection Service, as described below. Interconnection Customer may designate only one type of Interconnection Service for each separate Interconnection Request in the Queue. The type of Interconnection Service must be finalized on submission of the executed Definitive System Impact Study Agreement and may only be changed after the start of the Definitive Study Process between Phase 2 and Phase 3 of the Definitive Interconnection Study Process and only if a Cluster must be re-studied in Phase 3 and otherwise may not be changed.

3.2.1 Energy Resource Interconnection Service

3.2.1.1 The Product.

Energy Resource Interconnection Service allows
Interconnection Customer to connect the Large
Generating Facility to the Transmission System and be
eligible to deliver the Large Generating Facility's output
using the existing firm or non-firm capacity of the
Transmission System on an "as available" basis. Energy
Resource Interconnection Service does not in and of
itself convey any right to deliver electricity to any
specific customer or Point of Delivery.

3.2.1.2 The Study.

The study consists of short circuit/fault duty, steady state (thermal and voltage) and stability analyses. The short circuit/fault duty analysis would identify direct

Interconnection Facilities required and the Network Upgrades necessary to address short circuit issues associated with the Interconnection Facilities. The stability and steady state studies would identify necessary upgrades to allow full output of the proposed Large Generating Facility and would also identify the maximum allowed output, at the time the study is performed, of the interconnecting Large Generating Facility without requiring additional Network Upgrades.

3.2.2 Network Resource Interconnection Service

3.2.2.1 The Product.

Transmission Provider must conduct the necessary studies and construct the Network Upgrades needed to integrate the Large Generating Facility: (1) in a manner comparable to that in which Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an ISO or RTO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service allows Interconnection Customer's Large Generating Facility to be designated as a Network Resource, up to the Large Generating Facility's full output, on the same basis as existing Network Resources interconnected to Transmission Provider's Transmission System, and to be studied as a Network Resource on the assumption that such a designation will occur. If the Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resource within Transmission Provider's Control Area, the Interconnection Customer must provide the point of delivery or the geographic location on EPE's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area.

3.2.2.2 The Study.

The Interconnection Study for Network Resource Interconnection Service shall assure that Interconnection Customer's Large Generating Facility meets the requirements for Network Resource Interconnection Service and as a general matter, that such Large Generating Facility's interconnection is also studied with Transmission Provider's Transmission System at peak load, under a variety of severely stressed conditions, to determine whether, with the Large Generating Facility at full output, the aggregate of generation in the local area can be delivered to the aggregate of load on Transmission Provider's Transmission System, consistent with Transmission Provider's reliability criteria and procedures. This approach assumes that some portion of existing Network Resources' output is displaced by the output of Interconnection Customer's Large Generating Facility. Network Resource Interconnection Service in and of itself does not convey any right to deliver electricity to any specific customer or Point of Delivery. The Transmission Provider may also study the Transmission System under non-peak load conditions. However, upon request by the Interconnection Customer, the Transmission Provider must explain in writing to the Interconnection Customer why the study of non-peak load conditions is required for reliability purposes.

3.3 Utilization of Surplus Interconnection Service.

Transmission Provider will use the process below to allow an Interconnection Customer to utilize or transfer Surplus Interconnection Service at an existing Point of Interconnection. The original Interconnection Customer or one of its affiliates shall have priority to utilize Surplus Interconnection Service. If the existing Interconnection Customer or one of its affiliates does not exercise its priority, then that service may be made available to other potential Interconnection Customers.

3.3.1 Surplus Interconnection Service Requests.

Surplus Interconnection Service requests may be made by the existing Interconnection Customer whose Generating Facility is already interconnected or one of its affiliates. Surplus Interconnection Service requests also may be made by another Interconnection Customer. Transmission Provider shall use the process in Section 3.3.2 in evaluating Interconnection Requests for Surplus Interconnection Service. Studies for Surplus Interconnection Service shall consist of reactive power, short circuit/fault duty, stability analyses, and any other appropriate studies.

Steady-state (thermal/voltage) analyses may be performed as necessary to ensure that all required reliability conditions are studied. If the Surplus Interconnection Service was not studied under off-peak conditions, off-peak steady state analyses shall be performed to the required level necessary to demonstrate reliable operation of the Surplus Interconnection Service. If the original System Impact Study is not available for the Surplus Interconnection Service, both off-peak and peak analysis may need to be performed for the existing Generating Facility associated with the request for Surplus Interconnection Service. The reactive power, short circuit/fault duty, stability, and steady-state analyses for Surplus Interconnection Service will identify any additional Interconnection Facilities and/or Network Upgrades necessary.

3.3.2 Process for Securing Surplus Interconnection Service.

An existing (original) Interconnection Customer whose Large Generating Facility is already interconnected, may make available its Surplus Interconnection Service at the existing Point of Interconnection for the Large Generating Facility using the process outlined in this Section 3.3.2. The original Interconnection Customer may retain any surplus for itself, or may transfer it to an entity of its own choosing, whether an affiliate or a non-affiliate. The amount of Surplus Interconnection Service made available cannot exceed the total amount of Interconnection Service at the Point of Interconnection established in the original Interconnection Customer's Large Generator Interconnection Agreement.

A. Either the original Interconnection Customer, its affiliate or a third-party Interconnection Customer (transferee) shall notify the Transmission Provider in writing of a request to utilize or transfer any Surplus Interconnection Service made available by the original Interconnection Customer at the existing Point of Interconnection. The written Surplus Interconnection Service request should include, at a minimum, the following information; (1) the amount of Surplus Interconnection Service (in MW) proposed to be made available; (2) the start date and end date that the Surplus Interconnection Service is to be made available; (3) the type of service (Energy Resource Interconnection Service or Network Resource Interconnection Service) to be made available as Surplus Interconnection Service, provided however, that if the original Interconnection Customer's Large Generator Interconnection Agreement provides for Energy Resource Interconnection Service, any Surplus Interconnection Service must also be Energy Resource Interconnection Service; (4) a one-line diagram illustrating how the Generating Facility that is to use the Surplus Interconnection Service will connect to the original

Interconnection Customer's Interconnection Facilities; (5) details on the Generating Facility that is to use the Surplus Interconnection Service (type of resource, characteristics, modeling information); (6) the signature of the original Interconnection Customer if the original Interconnection Customer is not the entity submitting the Surplus Interconnection Service request; and (7) any other relevant conditions of the original Interconnection Customer on its proposed utilization or transfer. To aid the original Interconnection Customer in initiating this process, Transmission Provider may post on OASIS a blank request form for Surplus Interconnection Service, identifying the required fields of data set forth above.

- B. Transmission Provider shall process such surplus interconnection service requests separately from other requests pending in its non-surplus interconnection queue. All requests for Surplus Interconnection Service will be subject to technical studies, as necessary, to ensure the reliable use of Surplus Interconnection Service. The Transmission Provider shall evaluate if the original System Impact Studies are still applicable and sufficient to accommodate the new request, and may require new or additional studies to evaluate system impacts. Examples of circumstances in which the Transmission Provider would require new studies include, but are not limited to, circumstances in which the original System Impact Study(ies) are more than 3 years old and/or circumstances in which new generation facilities have been added to the Transmission Provider's system since the original Interconnection Customer's Large Generating Facility was studied. Examples of studies that may be necessary and appropriate to examine the potential for system impacts include, but are not limited to:
 - i. Powerflow studies (steady-state, thermal/voltage),
 - ii. Stability studies (voltage/angular),
 - iii. Short circuit/fault duty studies,
 - iv. Reactive power studies
 - v. Transient, harmonic and/or sub-transient studies, particularly if the generation technology of the generation facility using the Surplus Interconnection Service is different from the generation technology in the original Interconnection Customer's Large Generation Interconnection Agreement.
- C. Should Transmission Provider determine that studies are necessary to evaluate a request for Surplus Interconnection Service, Transmission Provider will collect from the requestor a study deposit of \$25,000, subject to true-up to reflect the actual cost of the studies.

- D. If the technical studies show that the addition of Network Upgrades will be required for the use of Surplus Interconnection Service due to different technology or short circuit requirements, or for other reasons, Surplus Interconnection Service shall be available up to the amount that can be accommodated without Network Upgrades. The Generating Facility receiving Surplus Interconnection Service is to interconnect to the original Interconnection Customer's Interconnection Facilities. Changes or additions to the original Interconnection Customer's Interconnection Facilities may be necessary and permissible. The need for changes to other facilities identified in the original Interconnection Customer's Interconnection Agreement, including Network Upgrades, would make Surplus Interconnection Service unavailable.
- E. An agreement for Surplus Interconnection Service shall be developed between the Transmission Provider, the original Interconnection Customer and the Surplus Interconnection Service Customer. The Transmission Provider will file the agreement for Surplus Interconnection Service with the Commission. The agreement will contain the parameters of the Surplus Interconnection Service, its level of service, term of service, and any conditions arising from the results of the Transmission Provider's evaluation of system impacts. Upon Commission approval of the Surplus Interconnection Agreement and concurrent with the effective date of the Surplus Interconnection Service, the original Interconnection Customer's Interconnection Agreement shall be deemed to be adjusted to lower the level of Interconnection Service by an amount equal to the Surplus Interconnection Service, without the need for any filings or approvals.
- F. The use of Surplus Interconnection Service does not convey any promise or grant of transmission service.
- G. Surplus Interconnection Service cannot be offered until all facilities required for the original Interconnection Customer's interconnection service (including all Contingent Facilities) are constructed and In Service.
- H. Surplus Interconnection Service cannot be offered if the original Interconnection Customer's Generating Facility is scheduled to retire and permanently cease Commercial Operation before the Surplus Interconnection Service Customer's Generating Facility begins Commercial Operation.

3.4 Valid Interconnection Request

3.4.1 Initiating an Interconnection Request

An Interconnection Customer wishing to join the Definitive Interconnection Study Process shall submit a completed Interconnection Request to Transmission Provider within, and no later than the close of, the DISIS Request Window. To initiate an Interconnection Request, Interconnection Customer must submit all of the following:

- (i) Initial Deposit of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater;
- (ii) A completed application in the form of Appendix 1 (including all technical information);
- A demonstration of Site Control as defined in Section 1 and (iii) addressed further in Section 7.7.7 of the LGIP. Specifications for acceptable site size for the purposes of demonstrating Site Control are posted on Transmission Provider's OASIS website. Interconnection Customer may propose alternative specifications for site size to those posted on OASIS for Transmission Provider approval. In the event Transmission Provider and Interconnection Customer cannot reach agreement related to adequacy of site size, Transmission Provider will accept a Professional Engineer (licensed in the state of Texas or New Mexico, as applicable depending on the state in which the Generating Facility is to be located) stamped site plan drawing that depicts the proposed generation arrangement and specifies the maximum facility output for that arrangement;
- (iv) A Point of Interconnection;
- (v) If the request is for NRIS and if Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resourced within Transmission Provider's Control Area, the point of delivery or the geographic location on Transmission Provider's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area;

- (vi) A Generating Facility size (MW) (and requested Interconnection Service amount if the requested Interconnection Service is less than the Generating Facility Capacity);
- (vii) One of the following Readiness Milestone ("M1") options totaling the entire capacity of the Generating Facility (or requested Interconnection Service amount if the requested Interconnection Service amount is less than the Generating Facility Capacity or security equal to one times the study deposit described above (see Initial Deposit) in the form of an irrevocable letter of credit or cash *in lieu of* the Readiness Milestone. The security is refunded to the Interconnection Customer according to Section 7.7.5.
 - a. Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale (1) of the constructed Generating Facility, or (2) of the Generating Facility's energy, or (3) of the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
 - b. Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or
 - c. Provisional Large Generator Interconnection
 Agreement filed with FERC that contains a
 commitment to move forward with constructing the
 Generating Facility and is not suspended; and
- (viii) Security equal to one times the study Initial Deposit in the form of an irrevocable letter of credit or cash. The security is refunded to the Interconnection Customer according to Section 7.7.

The expected In-Service Date of the new Large Generating Facility or increase in capacity of the existing Generating Facility shall be no more than the process window for the regional expansion planning period (or in the absence of a regional planning process, the process window for Transmission Provider's expansion planning period) not to exceed seven years from the date the Interconnection Request is received by Transmission Provider, unless Interconnection Customer

demonstrates that engineering, permitting and construction of the new Large Generating Facility or increase in capacity of the existing Generating Facility will take longer than the regional expansion planning period. The In-Service Date may succeed the date the Interconnection Request is received by Transmission Provider by a period up to ten years, or longer where Interconnection Customer and Transmission Provider agree, such agreement not to be unreasonably withheld.

3.4.2 Acknowledgment of Interconnection Request

Transmission Provider shall acknowledge receipt of the Interconnection Request within five (5) Business Days of the close of the DISIS Request Window and attach a copy of the received Interconnection Request to the acknowledgement.

3.4.3 Deficiencies in Interconnection Request

An Interconnection Request will not be considered to be a valid request until all items in Section 3.4.1 have been received by Transmission Provider. If an Interconnection Request fails to meet the requirements set forth in Section 3.4.1, Transmission Provider shall notify Interconnection Customer within five (5) Business Days of the close of the DISIS Request Window of the reasons for such failure and that the Interconnection Request does not constitute a valid request. Interconnection Customer shall provide Transmission Provider the additional requested information needed to constitute a valid request within ten (10) Business Days after receipt of such notice. At any time, if Transmission Provider identifies issues with technical data provided by Interconnection Customer, Interconnection Customer and Transmission Provider shall work expeditiously and in good faith to remedy any data issues. If the data issues are remedied by the Interconnection Customer without adversely affecting the Transmission Provider's study timeline, the remedy will be accommodated. Failure by Interconnection Customer to comply with this Section 3.4.3 shall be treated in accordance with Section 3.7.

Transmission Provider shall determine if the information contained in the Interconnection Request is adequately sufficient to start the Definitive System Impact Study by the close of the Customer Engagement Window.

3.4.4 Scoping Meeting

Within ten (10) Business Days after the close of the DISIS Request

Window, Transmission Provider shall host a Scoping Meeting for all Interconnection Requests received in that DISIS Request Window. If requested by Interconnection Customer, Transmission Provider shall also hold individual customer specific Scoping Meetings, which must be requested no later than fifteen (15) business days after the close of the DISIS Request Window.

The purpose of the Scoping Meeting shall be to discuss alternative interconnection options, to exchange information including any transmission data that would reasonably be expected to impact such interconnection options, to analyze such information and to determine the potential feasible Points of Interconnection. Transmission Provider and Interconnection Customer will bring to the meeting such technical data, including, but not limited to: (i) general facility loadings, (ii) general instability issues, (iii) general short circuit issues, (iv) general voltage issues, and (v) general reliability issues as may be reasonably required to accomplish the purpose of the meeting. Transmission Provider and Interconnection Customer will also bring to the meeting personnel and other resources as may be reasonably required to accomplish the purpose of the meeting in the time allocated for the meeting. On the basis of the meeting, Interconnection Customer shall designate its Point of Interconnection. The duration of the meeting shall be sufficient to accomplish its purpose.

3.5 OASIS Posting

3.5.1 OASIS Posting

Transmission Provider will maintain on its OASIS a list of all Interconnection Requests. The list will identify, for each Interconnection Request: (i) the maximum summer and winter megawatt electrical output; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the status of the Interconnection Request, including Queue Position; (vi) the type of Interconnection Service being requested; (vii) the availability of any studies related to the Interconnection Request; (viii) the date of the Interconnection Request; (ix) the type of Generating Facility to be constructed (combined cycle, base load or combustion turbine and fuel type); and (x) for Interconnection Requests that have not resulted in a completed interconnection, an explanation as to why it was not completed. Except in the case of an Affiliate, the list will not disclose the identity of Interconnection Customer until Interconnection Customer executes an LGIA or requests that Transmission Provider file an unexecuted LGIA with FERC.

Before holding a Scoping Meeting with its Affiliate, Transmission Provider shall post on OASIS an advance notice of its intent to do so. Transmission Provider shall post to its OASIS site any deviations from the study timelines set forth herein. Interconnection Study reports shall be posted to Transmission Provider's OASIS site subsequent to the meeting between Interconnection Customer and Transmission Provider to discuss the applicable study results. Transmission Provider shall also post any known deviations in the Large Generating Facility's In-Service Date.

3.5.2 Requirement to Post Interconnection Study Metrics

Transmission Provider will maintain on its OASIS or its website summary statistics related to processing Interconnection Studies pursuant to Interconnection Requests, updated quarterly. If Transmission Provider posts this information on its website, a link to the information must be provided on Transmission Provider's OASIS site. For each calendar quarter, Transmission Providers must calculate and post the information detailed in sections 3.5.2.1 through 3.5.2.3. Where an Interconnection Customer is a member of a Queue Cluster and such customer tenders to Transmission Provider an executed study agreement before the close of the Queue Cluster Window, Transmission Provider will deem its date of receipt of the study agreement to be the day immediately after the close of the Queue Cluster Window for purposes of calculating and posting processing time under Sections 3.5.2.1 and 3.5.2.2.

3.5.2.1 Interconnection System Impact Studies processing time.

- (A) Number of Interconnection Requests that had Phase 1 Studies completed within Transmission Provider's coordinated region during the reporting quarter, and the number of Interconnection Requests that had Phase 2 and Phase 3 Studies completed within Transmission Provider's coordinated region during the reporting quarter,
- (B) Number of Interconnection Requests that had Phase 1 Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than ninety (90) Calendar Days after the close of the Queue Cluster Window, or more than ninety (90) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact Study Agreements, whichever is later, and the number

of Interconnection Requests that had Phase 2 and Phase 3 Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than one hundred fifty (150) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact Study Agreements,

- (C) At the end of the reporting quarter, the number of active valid Interconnection Requests with ongoing incomplete System Impact Studies where such Interconnection Requests had executed Definitive Interconnection System Impact Study Agreements received by Transmission Provider more than one hundred fifty (150)Calendar Days after the close of the Queue Cluster Window, or more than one hundred fifty (150) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection Impact Study Agreements, whichever is later, before the reporting quarter end,
- (D) Mean time (in days), Definitive Interconnection System Impact Studies completed within Transmission Provider's coordinated region during the reporting quarter, from the date when Transmission Provider received the executed Interconnection System Impact Study Agreement(s) to the date when Transmission Provider provided the completed Interconnection System Impact Study to the Interconnection Customer(s),
- (E) Percentage of Interconnection System Impact Studies exceeding the timelines set forth above to complete this reporting quarter, calculated as the sum of 3.5.2.1(B) plus 3.5.2.1(C) divided by the sum of 3.5.2.1(A) plus 3.5.2.1(C)).

3.5.2.2 Interconnection Facilities Studies processing time.

(A) Number of Interconnection Requests that had Interconnection Facilities Studies that are completed within Transmission Provider's coordinated region during the reporting quarter and tendered to the Interconnection Customer in draft form,

- (B) Number of Interconnection Requests that had Interconnection Facilities Studies that are completed within Transmission Provider's coordinated region during the reporting quarter that were completed and tendered to the Interconnection Customer in draft form more than ninety (90) Calendar Days after receipt by Transmission Provider of the Interconnection Customer's executed Interconnection Facilities Study Agreement,
- (C) At the end of the reporting quarter, the number of active valid Interconnection Service requests with ongoing incomplete Interconnection Facilities Studies where such Interconnection Requests had executed Interconnection Facilities Studies Agreement received by Transmission Provider more than ninety (90) Calendar Days before the reporting quarter end,
- (D) Mean time (in days), for Interconnection Facilities Studies completed within Transmission Provider's coordinated region during the reporting quarter, calculated from the date when Transmission Provider received the executed Interconnection Facilities Study Agreement to the date when Transmission Provider provided the completed draft Interconnection Facilities Study to the Interconnection Customer,
- (E) Percentage of delayed Interconnection Facilities Studies this reporting quarter, calculated as the sum of 3.5.2.2(B) plus 3.5.2.2(C) divided by the sum of 3.5.2.2(A) plus 3.5.2.2(C)).

3.5.2.3 Interconnection Service requests withdrawn from Interconnection queue.

- (A) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter,
- (B) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of any interconnection studies or execution of any interconnection study agreements,
- (C) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the

reporting quarter before completion of an Interconnection System Impact Study,

- (D) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of an Interconnection Facilities Study,
- (E) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue after execution of a generator interconnection agreement or Interconnection Customer requests the filing of an unexecuted, new interconnection agreement,
- (F) Mean time (in days), for all withdrawn Interconnection Requests, from the date when the request was determined to be valid to when Transmission Provider received the request to withdraw from the queue.

3.5.3

Transmission Provider is required to post on OASIS or its website the measures in paragraph 3.5.2.1(A) through paragraph 3.5.2.3(F) for each calendar quarter within 30 days of the end of the calendar quarter. Transmission Provider will keep the quarterly measures posted on OASIS or its website for three calendar years with the first required report to be in the first quarter of 2020. If Transmission Provider retains this information on its website, a link to the information must be provided on Transmission Provider's OASIS site.

3.5.4

In the event that any of the values calculated in paragraphs 3.5.2.1(E), or 3.5.2.2(E) exceeds 25 percent for two consecutive calendar quarters, Transmission Provider will have to comply with the measures below for the next four consecutive calendar quarters and must continue reporting this information until Transmission Provider reports four consecutive calendar quarters without the values calculated in 3.5.2.1(E), or 3.5.2.2(E) exceeding 25 percent for two consecutive calendar quarters:

(i) Transmission Provider must submit a report to the Commission describing the reason for each study or group of clustered studies pursuant to an Interconnection Request that exceeded its deadline (i.e., 90, 150 or 90 days) for completion (excluding any allowance for Reasonable Efforts). Transmission Provider must describe the reasons for each study delay and any steps taken to remedy these specific issues and, if applicable, prevent such delays in the future. The report must be filed at the Commission within 45 days of the end of the calendar quarter.

(ii) Transmission Provider shall aggregate the total number of employee-hours and third party consultant hours expended towards interconnection studies within its coordinated region that quarter and post on OASIS or its website. If Transmission Provider posts this information on its website, a link to the information must be provided on Transmission Provider's OASIS site. This information is to be posted within 30 days of the end of the calendar quarter.

3.6 Coordination with Affected Systems

Transmission Provider will coordinate the conduct of any studies required to determine the impact of the Interconnection Request on Affected Systems with Affected System Operators and, if possible, include those results (if available) in its applicable Interconnection Study within the time frame specified in this LGIP. Transmission Provider will include such Affected System Operators in all meetings held with Interconnection Customer as required by this LGIP. Interconnection Customer will cooperate with Transmission Provider in all matters related to the conduct of studies and the determination of modifications to Affected Systems. A Transmission Provider which may be an Affected System shall cooperate with Transmission Provider with whom interconnection has been requested in all matters related to the conduct of studies and the determination of modifications to Affected Systems. It is the responsibility of the Affected System owner to provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to (i) complete any interconnection studies and (ii) construct any necessary interconnection facilities and network upgrades needed to reliably interconnect at the requested service level.

3.7 Withdrawal

Interconnection Customer may withdraw its Interconnection Request at any time by written notice of such withdrawal to Transmission Provider. In addition, if Interconnection Customer fails to adhere to all requirements of this LGIP, except as provided in Section 13.5 (Disputes), Transmission

Provider shall deem the Interconnection Request to be withdrawn and shall provide written notice to Interconnection Customer of the deemed withdrawal and an explanation of the reasons for such deemed withdrawal. Upon receipt of such written notice, Interconnection Customer shall have fifteen (15) Business Days in which to either respond with information or actions that cures the deficiency or to notify Transmission Provider of its intent to pursue Dispute Resolution.

Withdrawal shall result in the loss of Interconnection Customer's Queue Position. If Interconnection Customer disputes the withdrawal and loss of its Queue Position, then during Dispute Resolution, Interconnection Customer's Interconnection Request is eliminated from the Queue until such time that the outcome of Dispute Resolution would restore its Queue Position. An Interconnection Customer that withdraws or is deemed to have withdrawn its Interconnection Request shall pay to Transmission Provider all costs that Transmission Provider prudently incurs with respect to that Interconnection Request prior to Transmission Provider's receipt of notice described above. Interconnection Customer must pay all monies due to Transmission Provider before it is allowed to obtain any Interconnection Study data or results.

In the case of a withdrawal, Transmission Provider shall (i) update the OASIS Queue Position posting; (ii) impose the Withdrawal Penalty described in Section 3.7.1; (iii) refund any security after settling the final invoice (see Section 7.7.5); and (iv) refund to Interconnection Customer any of the refundable portion of Interconnection Customer's study deposit that exceeds the share of the costs that Transmission Provider has incurred (less any non-refundable amounts described in this LGIP). The application of interest on amounts subject to refund shall be made in a manner consistent with Commission precedent and shall be calculated in accordance with section 35.19a(a)(2) of FERC's regulations. In the event of such withdrawal, Transmission Provider, subject to the confidentiality provisions of Section 13.1, shall provide, at Interconnection Customer's request, all information that Transmission Provider developed for any completed study conducted up to the date of withdrawal of the Interconnection Request.

3.7.1 Withdrawal Penalty

Interconnection Customers shall be subject to a Withdrawal Penalty if they withdraw their request from the Queue or the Generating Facility does not otherwise reach Commercial Operation unless the withdrawal does not negatively affect the timing or cost of equal or lower queued projects.

3.7.1.1 Calculation of the Withdrawal Penalty.

If the Interconnection Customer provided a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be equal to the higher of the study deposit or one (1) times of its actual allocated cost of the Definitive Interconnection Study Process. If the Interconnection Customer did not provide a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be dependent on the Phase in which the Interconnection Customer withdraws and shall be calculated as follows: 1. If the Interconnection Customer withdraws in Phase 1 (after M1, but before M2), the Withdrawal Penalty shall be the higher of the study deposit or two (2) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one (1) million dollars. 2. If the Interconnection Customer withdraws in Phase 2 (after M2, but before M3), the Withdrawal Penalty shall be the higher of the study deposit or three (3) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one and one half (1.5) million dollars. 3. If the Interconnection Customer withdraws in Phase 3 (after M3, but before M4), the study cost obligation shall be the higher of the study deposit or five (5) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at two (2) million dollars. 4. If the Interconnection Customer withdraws in Phase 4 (after M4, but before M5), the Withdrawal Penalty shall be the higher of the study deposit or seven (7) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at two and a half (2.5) million dollars.

The Withdrawal Penalty for any customer that has executed an LGIA is the higher of the study deposit or nine (9) times its actual allocated cost of the Definitive Interconnection Study Process.

3.7.1.2 Distribution of the Withdrawal Penalty.

Any Withdrawal Penalty revenues shall be used to fund generation interconnection studies. Withdrawal Penalty revenues shall first be applied, in the form of a bill credit, to not-yet-invoiced study costs for other Interconnection Customers in the same cluster, and to the extent that such studies are fully credited, shall be applied to study costs of future clusters in Queue order. Withdrawn Interconnection Customers shall not receive a bill credit associated with Withdrawal Penalties. Distribution of Withdrawal Penalty revenues to a specific study shall

not exceed the total actual study costs. Allocation of Withdrawal Penalty revenues within a cluster to a specific customer shall be comparable to the allocation of study costs described in Section 4.2. Specifically, the Withdrawal Penalty revenue distribution to each customer in a specific cluster, shall be (1) fifty percent (50%) on a per capita basis based on number of Interconnection Requests in the applicable Cluster; and (2) fifty percent (50%) to Interconnection Customers on a pro-rata basis based on requested megawatts included in the applicable Cluster. Distribution of Withdrawal Penalty revenue associated with Readiness Milestone 5 shall not be distributed to the remaining customers in that cluster until all customers in that cluster have reached Commercial Operation and thereafter shall be distributed as described above. Transmission Provider shall not change the distribution of Withdrawal Penalty revenue without authorization by the Commission. Transmission Provider shall post the Withdrawal Penalty balance on its OASIS site.

3.8 Identification of Contingent Facilities.

The Transmission Provider shall identify Contingent Facilities in the System Impact Study Report (including any restudy reports) using the steps outlined below. The method permits the parties to determine why a specific Contingent Facility was identified and how it relates to the Interconnection Request.

3.8.1 Baseline assumptions.

Transmission Provider uses a technical screening process to identify Contingent Facilities that starts with the baseline assumption that the following are in service: (i) Generating Facilities that are directly interconnected to the Transmission System; (ii) Generating Facilities that are interconnected to Affected Systems where the Affected System(s) have communicated to Transmission Provider, or Transmission Provider otherwise has determined, that such facilities may have an impact on the Interconnection Request; (iii) Generating Facilities that have a pending higher-queued Interconnection Request to interconnect to the Transmission System and their associated Interconnection Facilities and Network Upgrade requirements, to the extent those higher-queued Interconnection Requests have been the subject of a System Impact Study and/or a Facilities Study; (iv) Generating Facilities that executed an interconnection agreement, or requested that an unexecuted interconnection agreement be filed with FERC, and their associated Interconnection Facilities and Network Upgrades; (v) higher-queued requests for transmission service and

their associated facilities or upgrade requirements to the extent they have an impact on the Interconnection Request and to the extent that those higher-queued requests for transmission service have been the subject of a System Impact Study and/or a Facilities Study; (vi) Transmission Provider's transmission expansion plan components; and (vii) the transmission expansion plan components of third-party transmission providers, to the extent that Transmission Provider has determined that they have an impact on the Interconnection Request.

With respect to the treatment of higher-queued requests for interconnection service and/or transmission service, in situations in which (a) the higher-queued requests have not yet proceeded through the System Impact Study process to reach the stage where facilities necessary to accommodate the requests have been identified, or (b) facilities associated with higher-queued requests for service change as a result of queue withdrawal or otherwise, Transmission Provider shall adjust its baseline assumptions with the most current information available and produce a re-study System Impact Study Report on Interconnection Customer's Interconnection Request.

The re-study report will use the method described in this Section to permit the parties to determine why a specific Contingent Facility was identified and how it relates to the Interconnection Request.

3.8.2. Technical Screening Process.

The technical screening process for identifying Contingent Facilities is comprised of the following steps:

Step 1: Identify Potential Contingent Facilities.

Transmission Provider will review all applicable Interconnection and Transmission Service study results for higher-queued Interconnection or Transmission Service Requests to identify any unbuilt Interconnection Facilities and/or Network Upgrades as potential Contingent Facilities to be evaluated pursuant to Steps 2-5 below.

Step 2: Remove a Potential Contingent Facility and Perform Applicable Contingency Analyses. The Transmission Provider will take a potential Contingent Facility (and its associated unbuilt Generating Facility) out of service in its study model and: (a) perform steady state, short circuit, voltage stability, and/or transient stability analyses to determine if the Transmission System demonstrates acceptable pre- and post-contingency system performance, in accordance with current Transmission Provider, WECC, NERC, or Reliability Coordinator criteria or standards; and (b) document the

resulting Transmission System performance deficiencies following the analysis in Step 2(a). In implementing this step in situations in which higher-queued clustered requests involve a potential Contingent Facility, Transmission Provider will remove all unbuilt Interconnection Facilities and/or Network Upgrades in the cluster, as well as their associated Generating Facilities in the cluster.

Step 3: Add the proposed Generating Facility into Model and Rerun Contingency Analyses. Transmission Provider will add the proposed Generating Facility into the model after taking the potential Contingent Facility and its associated Generating Facility out of service as provided in Step 2 above, and: (a) perform the same analysis for the added proposed Generating Facility as the analysis outlined in Step 2(a) for the removed potential Contingent Facility; and (b) document the resulting Transmission System performance deficiencies following the analysis in Step 3(a).

Step 4: Apply Threshold and Categorize. If the Transmission System performance deficiencies observed in Step 3(b) are: (a) exacerbated by one percent (1%) or greater than the Transmission System performance deficiencies initially observed in Step 2(b), then the potential Contingent Facility that is individually evaluated in Step 2 will be deemed a Contingent Facility; or (b) exacerbated by less than one percent (1%) than the Transmission System performance deficiencies initially observed in Step 2(b), then the potential Contingent Facility that is individually evaluated in Step 2 will not be deemed a Contingent Facility. The only variation from this analysis will apply to short circuit criteria. For the performance of the short circuit analysis of the Transmission Provider's Transmission System in its Balancing Authority Area, all generation directly connected to the Transmission Provider's Transmission System is assumed to be connected and synchronized to the grid. The fault current at any substation that contains one or more circuit breakers exceeding 95% of the interruption rating of any circuit breaker in the substation where the fault is taken will be considered a Contingent Facility.

Step 5: Repeat for Each Identified Potential Contingent Facility. Transmission Provider will repeat Steps 2-4 for each potential Contingent Facility identified in Step 1.

Step 6: Per Se Contingent Facilities. Notwithstanding Steps 1-5, an Interconnection Facility or Network Upgrade of a higher-queued request for service shall automatically be deemed a Contingent

Facility if such Interconnection Facility or Network Upgrade would be necessary for the proper functioning of the proposed Generating Facility's System Protection Facilities (as defined in Appendix 6 to Attachment M of Transmission Provider's OATT).

- 3.8.3. The Interconnection System Impact Study report will list Contingent Facilities in an appendix, which will include: (a) a description of each Contingent Facility; and (b) the Interconnection Request, transmission service request or planned project for which the Contingent Facility was initially required. This list of Contingent Facilities is subject to updates if a System Impact Study is re-studied pursuant to the Tariff under the LGIP or the provisions governing transmission service requests. In addition, where the Transmission Provider has identified an Affected System pursuant to Section 3.6 and facilities have been identified to mitigate adverse impacts on an Affected System, such facilities shall be included on the list of Contingent Facilities to the extent they have an impact on the Interconnection Request.
- **3.8.4.** If requested by the Interconnection Customer, and if readily available and not commercially sensitive, Transmission Provider will also provide an estimate of the costs of and the in-service date for each Contingent Facility, which may be subject to later updates if a Contingent Facility's estimated costs and in-service dates change.

Section 4. Interconnection Request Evaluation Process

4.1 Queue Position

4.1.1 Assignment of Queue Postion

Transmission Provider shall assign a Queue Position as follows: the Queue Position within the Queue shall be assigned based upon the date and time of receipt of all items required pursuant to the provisions of Section 3.4. There is no queue for informational Interconnection Studies.

4.1.2 Higher Queue Position

A higher Queue Position assigned to an Interconnection Request is one that has been placed "earlier" in the Queue in relation to another Interconnection Request that is assigned a lower Queue Position. All requests studied in a Cluster shall be considered equally queued but Clusters initiated earlier in time shall be considered to have a higher Queue Position than clusters initiated later. The Queue Position of an Interconnection Request shall have

no bearing on the allocation of the cost of the upgrades identified in the applicable Cluster Study (such costs will be allocated among Interconnection Requests in accordance with Section 4.2.2. Moving a Point of Interconnection shall result in a loss of Queue Position if it is deemed a Material Modification under Section 4.4.3.

4.2 Clustering and the Definitive Study Process

The diagram attached in the At-A-Glance Reference Sheet provides an overview of the Definitive Interconnection Study Process.

Transmission Provider shall accept Interconnection Requests during an approximate sixty (60) Calendar Day period referred to as the "DISIS Request Window." A DISIS Request Window shall open annually on or about February 1st and close on March 31th or the following Business Day if March 31th falls on a weekend or NERC recognized holiday. A second DISIS Request Window shall open annually on August 1st and close on September 30th or the following Business Day if September 30th falls on a Saturday or Sunday.

If one or more valid requests are received, for sixty (60) Calendar Days following the close of the DISIS Request Window (the "Customer Engagement Window"), Transmission Provider shall work with applicable Interconnection Customers to test models, verify data, hold stakeholder meetings (including Scoping Meetings, as appropriate), work with requestors to address any modeling irregularities identified by Transmission Provider in its evaluation of the Interconnection Request, and generally prepare for the start of the Definitive Interconnection System Impact Study. Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar.

At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7.

Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study.

4.2.1 Initiation of a Resource Solicitation Cluster

Upon request of a Resource Planning Entity, Transmission Provider may initiate the study of a Resource Solicitation Cluster. The Resource Solicitation Cluster shall respect Queue Position and shall be studied as its own Cluster. Within ten (10) Business Days of receipt of a request to perform a Resource Solicitation Cluster that includes valid Interconnection Requests as described in Section 3.4, Transmission Provider and Resource Planning Entity shall meet to determine a mutually agreeable scope of study and timeframe to initiate the Resource Solicitation Cluster. The timeline shall indicate the close of the Customer Engagement Window for that Resource Solicitation Cluster. Thereafter the Definitive Interconnection System Impact Study shall proceed as described in Section 7.

In order to initiate Transmission Provider's study of Interconnection Requests made in connection with a Resource Solicitation Process Resource Planning Entity must: (a) act as the authorized representative for all Interconnection Requests submitted to the Resource Solicitation Cluster; (b) submit all Interconnection Requests arising from the Resource Solicitation Process at the same time to ensure an equal Queue Position for all Generating Facilities included in the Resource Solicitation Cluster; (c) cooperate with Transmission Provider in conducting the studies; and (d) request a reasonable

number of different combinations of such Interconnection Requests to meet Resource Planning Entity's identified need and assumptions in the Resource Solicitation Process. Such studies in connection with a Resource Solicitation Process shall be implemented based upon Queue Position (relative to higher or lower queued clusters) and shall consider Resource Planning Entity's needs and assumptions identified in the Resource Solicitation Process. Transmission Provider shall not be obligated to implement a Resource Solicitation Cluster more than once a year.

The Resource Planning Entity may submit for inclusion in the Resource Solicitation Process an Interconnection Request for a Generating Facility that already has a higher Queue Position pursuant to Section 4.1.1. A Generating Facility that initially is associated with a Queue Position through the Resource Solicitation Process may also reserve a lower Queue Position separate from the Resource Solicitation Process pursuant to Section 4.1.1. In either case, Interconnection Customer must meet all requirements associated with maintaining each Queue Position for the Generating Facility. In the event a Generating Facility has multiple Queue Positions, it shall not be double counted in the study models.

A Generating Facility in the Resource Solicitation Process is subject to study according to the Queue Position of the Resource Solicitation Cluster. A Generating Facility that is not a part of the Resource Solicitation Process is also subject to study according to its Queue Position. All studies must be performed in accordance with the provisions of the LGIP, and may not be delayed as a result of the Resource Solicitation Process.

After Transmission Provider completes the Definitive Interconnection System Impact Studies for the requested combinations, the results will be provided (Phase 1 Reports, Phase 2 Reports, Phase 3 Reports, etc.; as applicable under Section 7.4) to the Resource Planning Entity for use in the Resource Solicitation Process. The results will be posted on Transmission Provider's OASIS consistent with the posting of other study results.

After receipt of the Phase 1 Report, Resource Planning Entity must select one of the studied combinations to advance to Phase 2 of the study process. After receipt of the Phase 2 Report, Resource Planning Entity must select one of the studied combinations prior to the commencement of any Interconnection Facilities Study associated

with the Resource Solicitation Process. Prior to the completion of the Interconnection Facilities Study of all of the components of the selected combination, Resource Planning Entity may replace components, subject to any necessary Re-Study pursuant to Sections 7.6 or 8.5. While conducting the Definitive Interconnection Study Process, Transmission Provider may suspend further action on the Interconnection Requests in the Resource Solicitation Process that are not included in the selected combination. Once a Generating Facility is rejected in the Resource Solicitation Process, the Generating Facility shall lose the Queue Position it held as part of the Resource Solicitation Process. If a Generating Facility is selected by Resource Planning Entity at the conclusion of the Resource Solicitation Process, the Generating Facility may no longer maintain more than one Queue Position.

4.2.2 Network Upgrades Cost Allocation

For Network Upgrades identified in a Queue Cluster study, the Transmission Provider shall calculate each Interconnection Customer's share of Network Upgrade costs in the following manner:

- (a) The costs for station equipment, including all switching stations, will be allocated on a *pro rata* basis based on the number of Generating Facilities interconnecting at an individual station.
- (b) The costs for all transmission lines, transformers and voltage-support related Network Upgrades will be allocated on a *pro rata* basis based on the proportional capacity of each individual Generating Facility in the Queue Cluster requiring such Network Upgrades.

4.3 Transferability of Queue Position

An Interconnection Customer may transfer its Queue Position to another entity only if such entity acquires the specific Generating Facility identified in the Interconnection Request and the Point of Interconnection does not change.

4.4 Modifications

Interconnection Customer shall submit to Transmission Provider, in writing, modifications to any information provided in the Interconnection Request. Interconnection Customer shall retain its Queue Position if the modifications are in accordance with Sections 4.4.1, 4.4.2 or 4.4.5, or are determined not to

be Material Modifications pursuant to Section 4.4.3.

Notwithstanding the above, during the course of the Interconnection Studies, either Interconnection Customer or Transmission Provider may identify changes to the planned interconnection that may improve the costs and benefits (including reliability) of the interconnection, and the ability of the proposed change to accommodate the Interconnection Request. Subject to the foregoing sentence, and provided, however, they do not result in a material modification, to the extent the identified changes are acceptable to Transmission Provider, Interconnection Customer and potentially impacted Interconnection Customers in the same Cluster, such acceptance not to be unreasonably withheld, Transmission Provider shall modify the Point of Interconnection and/or configuration in accordance with such changes and proceed with any re-studies necessary to do so in accordance with Section 7.6 and Section 8.5 as applicable and Interconnection Customer shall retain its Queue Position.

- 4.4.1 No later than thirty (30) Calendar Days after the close of the DISIS Request Window and prior to the return of the executed Definitive Interconnection System Impact Study Agreement to Transmission Provider, the modifications permitted under this Section shall include specifically: (a) a decrease of up to 60 percent of electrical output (MW) of the proposed project, through either (1) a decrease in plant size or (2) a decrease in Interconnection Service level (consistent with the process described in Section 3.1) accomplished by applying Transmission Provider-approved injection-limiting equipment; (b) modifying the technical parameters associated with the Large Generating Facility step-up transformer impedance characteristics; and (c) modifying the interconnection configuration. For plant increases, the incremental increase in plant output will go to the end of the Queue for the purposes of cost allocation and study analysis.
- 4.4.2 Prior to the return of the executed Interconnection Facilities Study Agreement to Transmission Provider, the modifications permitted under this Section shall include specifically: (a) additional 15 percent decrease of electrical output of the proposed project through either (1) a decrease in plant size (MW) or (2) a decrease in Interconnection Service level (consistent with the process described in Section 3.1) accomplished by applying Transmission Provider-approved injection-limiting equipment; (b) Large Generating Facility technical parameters associated with modifications to Large Generating Facility technology and transformer impedances; provided, however,

the incremental costs associated with those modifications are the responsibility of the requesting Interconnection Customer; and (c) a Permissible Technological Advancement for the Large Generating Facility after the submission of the Interconnection Request. Section 4.4.6 specifies a separate technological change procedure including the requisite information and process that will be followed to assess whether the Interconnection Customer's proposed technological advancement under Section 4.4.2(c) is a Material Modification. Section 1 contains a definition of Permissible Technological Advancement.

- 4.4.3 Prior to making any modification other than those specifically permitted by Sections 4.4.1, 4.4.2, and 4.4.6, Interconnection Customer may first request that Transmission Provider evaluate whether such modification is a Material Modification. In response to Interconnection Customer's request, Transmission Provider shall evaluate the proposed modifications prior to making them and inform Interconnection Customer in writing of whether the modifications would constitute a Material Modification. Any change to the Point of Interconnection, except those deemed acceptable under Sections 4.4.1, or so allowed elsewhere, shall constitute a Material Modification. Interconnection Customer may then withdraw the proposed modification or proceed with a new Interconnection Request for such modification.
- **4.4.4** Upon receipt of Interconnection Customer's request for modification permitted under this Section 4.4, Transmission Provider shall commence and perform any necessary additional studies as soon as practicable, but in no event shall Transmission Provider commence such studies later than thirty (30) Calendar Days after receiving notice of the modification of Interconnection Customer's request. Any additional studies resulting from such modification shall be done at Interconnection Customer's cost.
- 4.4.5 Extensions of less than three (3) cumulative years in the Commercial Operation Date of the Generating Facility to which the Interconnection Request relates are not material and should be handled through construction sequencing; provided, however, that extensions may necessitate a determination of whether additional studies are required pursuant to Applicable Laws and Regulations and Applicable Reliability Standards. The initial requested Commercial Operation Date used for this calculation is determined from the date

proposed in the initial Interconnection Request. Such cumulative extensions are inclusive of extensions requested after execution by Interconnection Customer of the LGIA.

4.4.6 Technological Change Procedure.

4.4.6.1 Interconnection Customer Technological Advancement Request.

- (a) At any time after the submission of an Interconnection Request, but before the execution of an Interconnection Facility Study Agreement by Interconnection Customer, an Interconnection Customer may submit a written request pursuant to this Section to include additional or substituted technological components for its Large Generating Facility that differ from the description of the Large Generating Facility in its Interconnection Request.
- (b) To timely perfect its Technological Advancement Request, Interconnection Customer shall submit the following to Transmission Provider: (i) completed Technological Advancement Request form submitted on the request template provided by Transmission Provider on its OASIS site; (ii) a \$25,000 deposit; (iii) an updated version of the Interconnection Request for a Large Generating Facility, found at Appendix 1 of this LGIP, that reflects the data associated with the change in technology that Interconnection Customer seeks to incorporate; (iv) to the extent applicable, updated modeling data.

Updated modeling data is applicable if the technology change results in a change in the time of use of the facility, i.e., from peak to off peak, or capacity factor of the facility; a change in the steady-state thermal and/or voltage limits of the facility; a change in harmonics; a possible increase in the short circuit capability of the facility; or a possible change in the transient or dynamic response of the facility.

Where a technology change does not fall into one of the categories above, but is a technology change that may reduce the Network Upgrades previously identified in the interconnection study process associated with the Interconnection Customer's Large Generating Facility, the Transmission Provider may require a Technological

Advancement Study using updated modeling data for purposes of implementing the identification of Network Upgrades necessary to accommodate the Interconnection Request under this LGIP and for the implementation of the crediting provisions set forth in Section 11.4 of the LGIA. Among the changes that Transmission Provider may evaluate under this Section 4.4.6 include, but are not limited to, advancements that improve, rather than degrade the electrical characteristics of the generating equipment (e.g., the ratings, impedances, efficiencies, capabilities, and performance of the equipment under steady state and dynamic conditions) without changing the real power (MW) output or reactive power (MVAR) output of the Large Generating Facility.

- (c) Interconnection Customer's Technological Advancement Request must demonstrate how the proposed technological advancement (i) results in equal to or better electrical performance, (ii) does not increase the Interconnection Customer's requested interconnection service, and (iii) does not cause any reliability concerns (i.e., material impacts to the transmission system, including impacts to short circuit capability limits, steady-state thermal and voltage limits, or dynamic system stability and response).
- (d) An Interconnection Customer may have no more than one Technological Advancement Request pending at any one time. If the Technological Advancement Request is submitted during the time allocated under the LGIP for Interconnection Customer to execute and return a System Impact or Facilities Study Agreement to Transmission Provider, the deadline for execution and return of the System Impact or Facilities Study Agreement will be suspended while Transmission Provider analyzes the Technological Advancement Request in accordance with Section 4.4.6.
- (e) If Transmission Provider is performing an Interconnection System Impact Study, or other study for the Interconnection Request at the time that Interconnection Customer submits a Technological Advancement Request, Transmission Provider shall suspend work on any such pending studies until it has completed its analysis of the Technological Advancement Request and any Technological Advancement Study.

(f) Interconnection Customer shall fill in, sign and submit, together with its Technological Advancement Request, a Technological Advancement Study Agreement. A form of Technological Advancement Study Agreement is posted on Transmission Provider's OASIS.

4.4.6.2 Initial Analysis of Technological Advancement Request.

- (a) After the Interconnection Customer's Technological Advancement Request is received pursuant to Section 4.4.6.1, the Transmission Provider will perform an initial analysis to determine whether the proposed technological advancement is a Permissible Technological Advancement without the need of additional study.
- (b) If the Transmission Provider determines on the basis of its initial analysis that Interconnection Customer has demonstrated that the proposed technological advancement is a Permissible Technological Advancement without the need for additional study, the Transmission Provider will incorporate the technological advancement into Interconnection Customer's Interconnection Request.
- (c) If the Transmission Provider determines on the basis of its initial analysis that Interconnection Customer has not demonstrated that the proposed technological advancement is a Permissible Technological Advancement, then the Technological Advancement Request will be treated as a request for modification of the Interconnection Request under Section 4.4.3.
- (d) If the Transmission Provider determines on the basis of its initial analysis that further study is required to conclude whether the Technological Advancement Request is a Permissible Technological Advancement, Transmission Provider will require that a Technological Advancement Study be performed at the sole expense of the Interconnection Customer consistent with Sections 4.4.6.3, 4.4.6.4, 4.4.6.5.
- (e) Any difference between the deposit provided under Section 4.4.6.1(b) and the actual cost of providing the initial analysis of the request shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate, along with an invoice describing any charges.

4.4.6.3 Technological Advancement Study Notification:

If after its initial analysis of a Technological Advancement Request, Transmission Provider determines that a Technological Advancement Study is necessary to determine whether the requested technological advancement constitutes a Permissible Technological Advancement, Transmission Provider shall notify Interconnection Customer in writing that such a study is necessary, and shall perform such study pursuant to Section 4.4.6.4.

4.4.6.4 Technological Advancement Study Procedures:

- (a) The Technological Advancement Study shall seek to determine (i) whether the proposed technological advancement is a Permissible Technological Advancement, by focusing on whether the proposed technological advancement will result in equal or better electrical performance than the Large Generating Facility described in the Interconnection Request, and whether the proposed technological advancement will cause any reliability concerns (i.e., material impacts to the transmission system, including impacts to short circuit capability limits, steady-state thermal and voltage limits, or dynamic system stability and response); and (ii) if the proposed technological advancement is determined not to be a Permissible Technological Advancement, whether the proposed technological advancement is a Material Modification. The Technological Advancement Study may include steady-state, reactive power, short circuit/fault duty, stability analyses, and any other appropriate studies that Transmission Provider deems necessary to determine whether the technological advancement results in electrical performance that is equal to or better than the electrical performance expected prior to the technology change, and whether such technological advancement causes any reliability concerns.
- (b) Interconnection Customer shall cooperate with Transmission Provider to provide any additional information that Transmission Provider may require to complete the Technological Advancement Study. If the Transmission Provider determines that it requires additional technical information to complete the Technological Advancement

Study, Transmission Provider shall notify the Interconnection Customer of the additional technical information required, and Interconnection Customer shall work in good faith with Transmission Provider to promptly provide such information.

- (c) Upon completion of the Technological Advancement Study, Transmission Provider shall provide Interconnection Customer notice of its study conclusions. Upon request, Transmission Provider shall also provide Interconnection Customer supporting documentation, subject to confidentiality arrangements consistent with Section 13.1.
- (d) If the Technological Advancement Study determines that the proposed technological advancement is either (i) a Permissible Technological Advancement, or (ii) is not a Permissible Technological Advancement but does not constitute a Material Modification, then the Interconnection Request shall be amended to reflect the technological advancement.
- (e) If the Technological Advancement Study determines that the proposed technological advancement is not a Permissible Technological Advancement and also constitutes a Material Modification, Transmission Provider shall provide an explanation for this conclusion. Interconnection Customer may then withdraw the proposed modification or proceed with a new Interconnection Request for such modification.
- (f) Any difference between the deposit provided under Section 4.4.6.1(b) and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate, along with an invoice describing any charges.

4.4.6.5 Time for Completing Initial Analysis and Technological Advancement Study.

Within thirty (30) Calendar Days of receipt of the Interconnection Customer's Technological Advancement Request submitted pursuant to Section 4.4.6.1, Transmission Provider shall complete all analysis and study obligations under this Section 4.4.6 and determine whether the

Technological Advancement Request is a Permissible Technological Advancement or Material Modification, subject to Interconnection Customer's performance of its obligations under Section 4.4.6.1(f) and Section 4.4.6.4(b).

4.4.6.6 Treatment of Other Interconnection Studies During and After Technological Advancement Study.

Upon completion of the Transmission Provider's initial analysis of a Technological Advancement Request and any Technological Advancement Study, Transmission Provider and Interconnection Customer shall amend any existing Interconnection System Impact Study Agreement, or other Interconnection Study Agreements as necessary to incorporate elements of the requested technological advancement or the results of the Technological Advancement Study. Transmission Provider may require additional time or information to complete or re-run studies that were suspended during the pendency of the Technological Advancement Request. A single study may be suspended no more than once as a result of a Technological Advancement Request. If a subsequent Technological Advancement Request is received by Transmission Provider in such circumstances. Transmission Provider will process the subsequent Technological Advancement Request as soon as it completes the study that had been previously suspended.

Section 5. Procedures for Interconnection Requests Submitted Prior to Effective Date of Standard Large Generator Interconnection Procedures

5.1 Transition to the First-Ready, First-Served Process

5.1.1 An Interconnection Customer assigned a Queue Position prior to the effective date of this LGIP shall retain that Queue Position subject to the requirements set forth in Sections 5.1.1.1 and 5.1.1.2. An Interconnection Customer that fails to meet these requirements shall have its Interconnection Request deemed withdrawn pursuant to Section 3.7, without prejudice to the Interconnection Customer's ability to request an Informational Interconnection Study for which there is no Queue Position. Any unused deposit amounts of withdrawn Interconnection Requests shall be returned pursuant to Section 3.7. If an Interconnection Customer elects to continue with an

Informational Interconnection Study or alternatively with a Transitional Interconnection Facilities Study or a Transitional Cluster Study as described below, Transmission Provider shall retain the current study deposits, and Interconnection Customer shall be responsible for the entire cost of all studies pursuant to Sections 6 and 4.

5.1.1.1 Transitional Projects.

An Interconnection Customer that has a) a final System Impact Study Report that identifies facilities required to feasibly interconnect and b) an Interconnection Facilities Study Agreement that was executed prior to the effective date of this Section 5 may opt to continue with the Interconnection Facilities Study process if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Interconnection Facilities Study Agreement in the form of Appendix 4.1 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All of the following are required:

- a) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades identified in the System Impact Study Report. The deposit shall be equal to one hundred percent (100%) of the costs identified for Transmission Provider's Interconnection Facilities and Network Upgrades in the System Impact Study Report and will be reconciled to actual costs after the associated facilities are in service. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied towards future construction costs described in the LGIA. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, the deposit is fully refundable once the final invoice for study costs and Withdrawal Penalty is settled. The deposit shall be in the form of an irrevocable letter of credit upon which the Transmission Provider may draw or cash.
- b) Exclusive Site Control for the entire Generating Facility and any Interconnection Customer's Interconnection Facilities.
- c) Interconnection Customer shall provide one following:
 - i. A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or

- ii. Reasonable evidence that the Generating Facility is included in an approved Resource Plan or Resource Solicitation Process, or
- iii. An executed Provisional Large Generator Interconnection Agreement filed with FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the publication of the final Interconnection Facilities Study Report or the Interconnection Request shall be deemed withdrawn pursuant to Section 3.7 unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of facilities if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's total study cost is imposed.

5.1.1.2 Combined System Impact and Interconnection Facilities Transitional Cluster Study.

An Interconnection Customer with an assigned Queue Position prior to the effective date of this Section 5 may opt to enter the combined system impact and interconnection facilities transitional cluster study ("Transitional Cluster Study") if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Cluster Study Agreement in the form of Appendix 4.2 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All Interconnection Requests that enter the Transitional Cluster Study shall be considered to have an equal Queue Position, and identified upgrade costs shall be allocated according to Section 4.2.2 of the LGIP. The Transitional Cluster Study costs shall be allocated

according to the method described in Section 4.2. Interconnection Customer may make a one-time extension to its requested Commercial Operation date upon entry into the Transitional Cluster Study and such an extension shall not be past May 1, 2027.

All of the following are required:

- a) Choice of requesting either ERIS or NRIS
- b) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades expected to be identified in the Transitional Cluster Study. The deposit shall be equal to five million dollars (\$5,000,000) and be in the form of an irrevocable letter of credit upon which the Transmission Provider may draw or cash. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied towards future construction costs described in the LGIA. Any amounts in excess of the actual construction costs shall be returned to the customer. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, the deposit is fully refundable once the final invoice for study costs and Withdrawal Penalty is settled.
- c) Exclusive Site Control for the entire Generating Facility.
- d) Interconnection Customer shall provide one of the following:
 - i. A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or
 - ii. Reasonable evidence that the Generating Facility is included in an approved Resource Plan or Resource Solicitation Process, or

iii. An executed Provisional Large Generator Interconnection Agreement filed with FERC that is not in suspension with 1) a commitment to construct the facility, 2) a Commercial Operation Date no later than May 1, 2027 and 3) a security deposit in addition to the five million dollars identified in 5.1.1.2.b where the total security deposit represents a reasonable estimation of the potential costs that could be ultimately allocated to the project in the transitional cluster study.

After the Transitional Cluster Study report is published, the remaining process shall proceed according to Section 11 of this LGIP. All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the tender of the draft LGIA or the Interconnection Request is deemed withdrawn unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of Transmission Provider's Interconnection Facilities or Network Upgrades if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit (see Section 3.4) is imposed.

An Interconnection Customer under an LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 (and not in suspension) that has not built its Large Generating Facility shall have 60 Calendar Days from the effective date of this Section 5.1.1.3 to submit to Transmission Provider a written commitment to (a) start construction of the Generating Facility no later than two (2) years of the effective date of this Section 5.1.1.3 and reach

Commercial Operation no later than three (3) years of the effective date of this Section 5.1.1.3 and (b) pay a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit if it withdraws, terminates its LGIA, or otherwise does not reach Commercial Operation. If it does not timely make this submission, the LGIA is deemed terminated, and no Withdrawal Penalty shall be assessed. A deemed termination shall be without prejudice to the Interconnection Customer's ability to submit a new Interconnection Request when it is willing and able to satisfy the tariff criteria under the Definitive Interconnection Process.

For any LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 and currently in suspension, the Interconnection Customer shall have 60 Calendar Days from lifting its suspension to make the submission to the Transmission Provider addressed above.

5.2 New Transmission Provider

If Transmission Provider transfers control of its Transmission System to a successor Transmission Provider during the period when an Interconnection Request is pending, the original Transmission Provider shall transfer to the successor Transmission Provider any amount of the deposit or payment with interest thereon that exceeds the cost that it incurred to evaluate the request for interconnection (less any non-refundable amounts). Any difference between such net amount and the deposit or payment required by this LGIP shall be paid by or refunded to the Interconnection Customer, as appropriate. The original Transmission Provider shall coordinate with the successor Transmission Provider to complete any Interconnection Study, as appropriate, that the original Transmission Provider has begun but has not completed. If Transmission Provider has tendered a draft LGIA to Interconnection Customer but Interconnection Customer has not either executed the LGIA or requested the filing of an unexecuted LGIA with FERC, unless otherwise provided, Interconnection Customer must complete negotiations with the successor Transmission Provider.

Section 6. Informational Interconnection Study

6.1 Informational Interconnection Study Agreement.

At any time, a customer may request, and Transmission Provider (either itself or through a consultant) shall perform a reasonable number of Informational Interconnection Studies from May 1 to November 1 each year. An Informational Interconnection Study shall consist of analyses of the type performed in Phase I of the DISIS (power flow and voltage analyses). Interconnection Customer shall submit a separate Informational Interconnection Request for each site and may submit multiple Informational Interconnection Requests for a single site. Interconnection Customer must submit a deposit with each Informational Interconnection Request even when more than one request is submitted for a single site. An Informational Interconnection Request to evaluate one site at two different voltage levels shall be treated as two Informational Interconnection Requests. The request shall use the form in Appendix 5.1 of the LGIP and shall describe the assumptions that Interconnection Customer wishes Transmission Provider to study within the scope described in Section 6.2 of the LGIP below. Within five (5) Business Days after receipt of a request for an Informational Interconnection Study, Transmission Provider shall provide to Interconnection Customer an Informational Interconnection Study Agreement in the form of Appendix 5.2.

The Informational Interconnection Study Agreement shall: (i) include the scope of work for the Informational Interconnection Study (ii) specify the technical data that Interconnection Customer must provide, (iii) specify the Informational Interconnection Study case and assumptions, and (iv) identify the Transmission Provider's estimate of the cost of the Informational Interconnection Study. To the extent known by Transmission Provider, such estimate shall include any costs expected to be incurred by any Affected System whose participation is necessary to complete the Informational Interconnection Study. Notwithstanding the above, Transmission Provider shall not be required as a result of an Informational Interconnection Study request to conduct any additional Interconnection Studies with respect to any other Interconnection Request.

Interconnection Customer shall execute the Informational Interconnection Study Agreement within ten (10) Business Days of receipt of an agreed upon scope of work and deliver the Informational Interconnection Study Agreement, the technical data, and a study deposit to Transmission Provider in the amount of \$100,000.00.

6.2 Scope of Informational Interconnection Study.

The intent of the Informational Interconnection Study is to aid Interconnection Customer in its business decisions related to interconnection of generation facilities prior to entering the Definitive Interconnection Process. The Informational Interconnection Study shall consist of analysis based on the assumptions and scope of work specified by Interconnection Customer in the Informational Interconnection Study Agreement. The Informational Interconnection Study shall identify the potential Transmission Provider's Interconnection Facilities and the Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the results and assumptions of the Informational Interconnection Study. The Informational Interconnection Study shall be performed solely for informational purposes. Transmission Provider shall use Reasonable Efforts to coordinate the study with any Affected Systems that may be affected by the types of Interconnection Services that are being studied. Transmission Provider shall utilize existing studies to the extent practicable in conducting the Informational Interconnection Study.

6.3 Informational Interconnection Study Procedures.

The executed Informational Interconnection Study Agreement, the deposit, and technical and other data called for therein must be provided to Transmission Provider within ten (10) Business Days of Interconnection Customer receipt of the Informational Interconnection Request. Transmission Provider shall use Reasonable Efforts to complete the Informational Interconnection Study within 70 Calendar Days from the date of the Informational Interconnection Study Agreement. If Transmission Provider is unable to complete the Informational Interconnection Study within such time period, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required. Any difference between the study payment and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation and work papers and databases or data developed in the preparation of the Informational Interconnection Study, subject to confidentiality arrangements consistent with Section 13.1.

Section 7. Definitive Interconnection System Impact Study

7.1 Definitive Interconnection System Impact Study Agreement

Unless otherwise agreed, pursuant to the Scoping Meeting provided for in Section 3.4.4, within thirty (30) Calendar Days acknowledgement of a valid Interconnection Request indicating that a Definitive Interconnection System Impact Study is to be performed, Transmission Provider shall provide to Interconnection Customer a DISIS Agreement in the form of Appendix 2 to

this LGIP. The DISIS Agreement shall provide that Interconnection Customer shall compensate Transmission Provider for the actual cost of the DISIS. No later than Fifteen (15) Business Days after the close of the DISIS Request Window, Transmission Provider shall provide to Interconnection Customer a non-binding updated good faith estimate of the cost and timeframe for completing the Definitive Interconnection System Impact Study.

7.2 Execution of Definitive Interconnection System Impact Study
Agreement Interconnection Customer shall execute the DISIS Agreement
and deliver the executed DISIS Agreement to Transmission Provider no later
than thirty (30) Calendar Days after its receipt.

7.3 Scope of Definitive Interconnection System Impact Study

The Definitive Interconnection System Impact Study shall evaluate the impact of the proposed interconnection on the reliability of the Transmission System. The Definitive Interconnection System Impact Study will consider the Base Case as well as all generating facilities (and with respect to (iii) below, any identified Network Upgrades associated with such higher queued requests) that, on the date the DISIS Request Window closes: (i) are existing and directly interconnected to the Transmission System; (ii) are existing and interconnected to Affected Systems and may have an impact on the Interconnection Request; (iii) have a pending higher queued Interconnection Request to interconnect to the Transmission System; and (iv) have no Queue Position but have executed an LGIA or requested that an unexecuted LGIA be filed with FERC. Generating facilities with pending higher or equal queued NRIS requests, or requests associated with Firm Transmission Service shall generally be modeled at full output while existing generation may be re-dispatched to accommodate new requests in the model. Higher queued ERIS requests or in-service ERIS generators without associated Firm Transmission Service may be dispatched at zero in some study models.

As discussed in more detail in Section 7.4 below, the Definitive Interconnection System Impact Study is a phased study where the first phase (Phase 1) consists of a power flow and voltage analysis that is followed by a phase (Phase 2) that consists of a short circuit analysis and a stability analysis, and may also include another appropriate study (for example, an electromagnetic transient analysis, if appropriate). Any DISIS re-studies (Phase 3) shall consist of a power flow/voltage analysis, a short circuit analysis, and/or a stability analysis as needed. The Definitive Interconnection System Impact Study report will state the assumptions upon

which it is based; state the results of the analyses; and provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to correct any problems identified in those analyses and implement the interconnection. For purposes of determining necessary Interconnection Facilities and Network Upgrades, the Definitive Interconnection System Impact Study shall consider the level of Interconnection Service requested by the Interconnection Customer, unless otherwise required to study the full Generating Facility Capacity due to safety or reliability concerns. The Definitive Interconnection System Impact Study will provide a list of facilities that are required as a result of the Interconnection Request and a non-binding good faith estimate of cost responsibility and a non-binding good faith estimated time to construct.

7.4 Definitive Interconnection System Impact Study Procedures

Transmission Provider shall coordinate the Definitive Interconnection System Impact Study with any Affected System that is affected by the Interconnection Request pursuant to Section 3.6 above. Transmission Provider shall utilize existing studies to the extent practicable when it performs the DISIS. Interconnection Requests for DISIS may be submitted only within the DISIS Request Window and Transmission Provider shall initiate the Definitive Interconnection Study Process pursuant to Section 4.2.

The At-A-Glance Reference Sheet attached provides an overview and timeline of the Definitive Interconnection Study Process, including the phases and milestones associated with the Definitive Interconnection System Impact Study.

- a. The DISIS Cluster shall consist of all eligible Interconnection Requests that have executed a DISIS Agreement and have provided all required information before the close of the DISIS Request Window. Transmission Provider shall use Reasonable Efforts to complete the first phase (Phase 1) consisting of a power flow and voltage analysis within ninety (90) Calendar Days. The Phase 1 Report shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades that are expected to be required as a result of the Interconnection Request(s) and a non-binding good-faith indicative level estimate of cost responsibility and a non-binding good-faith estimated time to construct. Transmission Provider shall meet with each cluster participant ("Phase 1 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 1 results on OASIS.
- b. Within thirty (30) Calendar Days of the Phase 1 Report, all Interconnection

Customers are required to provide Readiness Milestone 2 ("M2") and continued evidence of Site Control as described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

- Interconnection Customers whose M2 and Site Control are accepted by c. Transmission Provider shall continue in to the second phase ("Phase 2") of the Definitive Interconnection System Impact Study. Phase 2 consists of an updated power flow/voltage analysis (if necessary), stability analysis and short circuit analysis, and any other appropriate study (for example, an electromagnetic transient analysis, if appropriate) for the Interconnection Customers remaining in the DISIS Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 2 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that DISIS Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 2 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 2 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 2 results on OASIS.
- d. Within thirty (30) Calendar Days of the Phase 2 Report, each Interconnection Customer is required to provide Readiness Milestone 3 ("M3") and additional evidence of Site Control described in Section 7.7.7. Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.
 - i. If all Interconnection Customers in the Cluster provide M3 and no Interconnection Customers withdraw from the Queue at this stage, the Definitive Interconnection Study Process advances to the Interconnection Facilities Study (Section 8). Transmission Provider shall electronically notify Interconnection Customers in the Cluster that Phase 3 is not required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.

- ii. If one or more Interconnection Customer withdraws from the Cluster, Transmission Provider shall determine if a full system impact re-study is necessary. If Transmission Provider determines a re-study is not necessary and Phase 3 is not required, Transmission Provider shall provide an updated Phase 2 Report within thirty (30) Calendar Days of such determination and the Definitive Interconnection Study Process advances to the Interconnection Facilities Study (Section 8). When the updated Phase 2 report is issued, Transmission Provider shall electronically notify Interconnection Customers in the Cluster that Phase 3 is not required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.
- If one or more Interconnection Customers withdraws from the Cluster iii. and Transmission Provider determines a full system impact re-study is necessary, Transmission Provider will continue with System Impact re-studies ("Phase 3") as described in Section 7.4.e below, until Transmission Provider determines that no further re-studies are required. If a customer withdraws after Section 7.4.d.i or Section 7.4.d.ii or during the Interconnection Facilities Study and Transmission Provider determines system impact level studies are necessary, the Cluster shall be restudied under the terms of Phase 3. Transmission Provider shall electronically notify Interconnection Customers in the Cluster and post on OASIS that a re-study is required. Interconnection Customers that have elected NRIS may make a onetime change between Phase 2 and Phase 3 (before the re-study starts) to ERIS if they notify Transmission Provider of such change in election within five (5) Business Days of the Transmission Provider's notification the first re-study is required.
- e. Interconnection Customers whose M3 and additional evidence of Site Control is accepted by Transmission Provider shall continue with the third phase ("Phase 3") of the Definitive Interconnection System Impact Study. Phase 3 may consist of updated power flow/voltage analysis, stability analysis, and/or short circuit and other analyses if necessary for the Interconnection Customers remaining in the Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 3 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 3 Report shall identify each Interconnection

Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 3 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 3 results on OASIS. If additional re-studies are required before moving to Phase 4 below, within thirty (30) Calendar Days of the Phase 3 Report, all Interconnection Customers are required to provide an updated Readiness Milestone 3 ("M3"). Readiness Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7. by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7. Transmission Provider shall electronically notify Interconnection Customers in the Cluster when no further re-studies are required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.

- f. Within twenty (20) Calendar Days of the notice that no System Impact re-studies are needed, each Interconnection Customer is required to provide Readiness Milestone 4 ("M4"), Site Control requirements described in Section 7.7.7, and an executed Interconnection Facilities Agreement in the form of Appendix 3 (completed and including all required data identified therein). Readiness Milestones for the Definitive Interconnection System Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.
- g. Twenty (20) Calendar Days after the notice that no further Re-Studies are needed, Transmission Provider shall proceed with the Interconnection Facilities Study phase ("Phase 4") of the Definitive Study Process, described in detail in Section 8 below. An additional study deposit is not required for Phase 4.

At the request of Interconnection Customer or at any time Transmission Provider determines that it will not meet the indicated time frame for completing the DISIS, Transmission Provider shall notify Interconnection Customer(s) as to the schedule status of the DISIS Cluster. If Transmission Provider is unable to complete the DISIS within the time period, it shall notify Interconnection Customer(s) and provide an estimated completion date with an explanation of the reasons why additional time is required.

Upon request, Transmission Provider shall provide Interconnection Customer all supporting documentation, workpapers and relevant pre-Interconnection Request and post-Interconnection Request power flow, short circuit and stability databases for the DISIS, subject to confidentiality arrangements consistent with Section 13.1.

7.5 Meeting with Transmission Provider

Within ten (10) Business Days of providing a DISIS report to Interconnection Customer and posting the report on OASIS, Transmission Provider and Interconnection Customer shall meet to discuss the study results of the Interconnection System Impact Study, unless otherwise mutually agreed upon by the Parties.

7.6 Re-Study

If Re-Study Definitive of the Interconnection System Impact Study other than the re-study described above in 7.4.e is required due to a higher or equal priority queued project dropping out of the Queue, or a modification of a higher queued project subject to 4.4, Transmission Provider shall notify Interconnection Customer(s) in writing. The Transmission Provider shall use Reasonable Efforts to ensure such Re-Study take no longer than one hundred fifty (150) Calendar Days from the date of notice. Any cost of Re-Study shall be borne by Interconnection Customer(s) being re-studied.

7.7 Readiness Milestones and Site Control

Readiness Milestones are required throughout the Definitive Interconnection Study Process to demonstrate readiness. A customer that does not sufficiently demonstrate readiness by providing a Readiness Milestones is subject to withdrawal as described in Section 3.7 which may include additional penalties.

There are three Readiness Milestone options that demonstrate readiness through the study process (i.e., for Readiness Milestones 1 (M1) through Readiness Milestones 4 (M4).

7.7.1 Readiness Milestone 1 ("M1").

M1 is satisfied by any one of the three options below (also described in 3.4.1) at Interconnection Customer's option. M1 may also be satisfied by providing additional security described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness.

a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i)

the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.

- b) Reasonable evidence the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c) Provisional Large Generator Interconnection Agreement accepted for filing at FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

7.7.2 Readiness Milestone 2 ("M2").

M2 is satisfied by any one of the three options below at Interconnection Customer's option. M2 may also be satisfied by providing additional security as described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

- a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.
- b) Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c) Provisional Large Generator Interconnection Agreement accepted for filing at FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

7.7.3 Readiness Milestone 3 ("M3").

M3 is satisfied by any one of the three options below at Interconnection Customer's option. M3 may also be satisfied by providing additional security described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

a) Executed contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's

ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.

- b) Reasonable evidence that the project has been selected in an approved Resource Plan or Resource Solicitation Process; or
- c) An unsuspended Provisional Large Generator Interconnection Agreement accepted for filing by FERC with reasonable evidence that the Generating Facility and Interconnection Facilities have commenced design and engineering.

7.7.4 Readiness Milestone 4 ("M4").

M4 is satisfied by any one of the three options below at Interconnection Customer's option. M4 may also be satisfied by providing additional security as described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness.

- a) Executed contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services and capacity if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
- b) Reasonable evidence that the project has been selected in an approved Resource Plan or Resource Solicitation Process; or
- c) An unsuspended Provisional Large Generator Interconnection Agreement accepted for filing by FERC with reasonable evidence that the Generating Facility and Interconnection Facilities have commenced construction.

7.7.5 Security Requirements.

A table showing the security required in each milestone is provided in the At-A-Glance Reference Sheet attachment. The security amount is dependent on if the customer provided a Readiness Milestone and the study phase the customer is entering. All security described below shall be in the form of an irrevocable letter of credit upon which Transmission Provider may draw or cash. The security is refunded to the Interconnection Customer upon withdrawal, LGIA termination, or Commercial Operation after any final invoice is settled. If cash is

provided as security, it shall be refunded plus interest, where the interest is calculated in accordance with the methodology set forth in FERC's regulations at 18 CFR § 35.19a(a)(2)(iii) from the date the security is received to the date that it is refunded. Security may be drawn upon if costs under this LGIP including the LGIA remain unpaid as per this LGIP and the attached LGIA. As part of a valid interconnection request all Interconnection Customers must provide security equal to the study deposit amount as described in Section 3.4. The security provided in Section 3.4.1 will be applied towards the amount of security required for M5. An Interconnection Customer may opt to provide security in lieu of providing Readiness Milestones 1 through 4, as described above in Sections 7.7.1, 7.7.2, 7.7.3, and 7.7.4. The security provided is applied towards the security amount required for each successive milestone if the Interconnection Customer does not withdraw from the queue. For example, the security provided for M2 is applied to the amount of security required for M3.

In lieu of providing a demonstration of readiness for Milestones 1 through 4, the amount of security required is a multiple of the study deposit described in Section 3.4 and is in addition to the security required for all Interconnection Customers under Sections 3.4. The additional amount of security required for each milestone for Interconnection Customers that do not provide a demonstration of readiness is:

M1 = 1 times the study deposit amount

M2 = 2 times the study deposit amount

M3 = 4 times the study deposit amount

M4 = 6 times the study deposit amount

For clarity, the total (i.e., inclusive of the security required under Section 3.4) amount of security required for each milestone for Interconnection Customers that do not provide a demonstration of readiness is:

M1 = 2 times the study deposit amount

M2 = 3 times the study deposit amount

M3 = 5 times the study deposit amount

M4 = 7 times the study deposit amount

All Interconnection Customers are required to provide security in order to satisfy Readiness Milestone 5 (M5) when the LGIA is executed as described in Section 11.3. The amount of security required for M5 is equal to nine (9) times the Interconnection Customer's share of the Definitive Interconnection Study Process study costs. If this amount is not known, the study deposit amount shall be used as an estimate of study cost until such amounts are known. If initially estimated, M5 shall be updated when the final invoice for actual study costs is issued. As this M5 amount is the total security required to satisfy Readiness Milestone 5, any security provided pursuant to Sections 3.4 and 7.7 shall be applied towards the Readiness Milestone 5 amount when the LGIA is executed. The Interconnection Customer shall only be responsible to provide the incremental amount of security to the Transmission Provider and any excess security provided shall be refunded to the Interconnection Customer. Transmission Provider shall refund all security provided under this section to the Interconnection Customer upon achieving Commercial Operation.

7.7.6 [Reserved for Future Use]

7.7.7 Site Control.

In addition to the above Readiness Milestones, Site Control is required to determine increased readiness through the Definitive Interconnection Study Process. Additional information on Site Control is posted on Transmission Provider's OASIS.

- a) Before entering Phase 1 (concurrent with M1) demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- b) Before entering Phase 2 (concurrent with M2): continued demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- c) Before entering Phase 3 (concurrent with M3): demonstration of 60% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.

- d) Before entering Phase 4 (concurrent with M4): demonstration of 75% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- e) Before executing an LGIA (concurrent with M5): demonstration of 90% Site Control and 50% Site Control of Interconnection Customer's Interconnection Facilities is required.

Section 8. Interconnection Facilities Study

8.1 Interconnection Facilities Study Agreement

Simultaneously with the notice to Interconnection Customer(s) that Phase 3 is complete or not required, Transmission Provider shall provide to Interconnection Customer an Interconnection Facilities Study Agreement in the form of Appendix 3 to this LGIP. Within five (5) Business Days following the DISIS results (Phase 2 or Phase 3) meeting, Transmission Provider shall provide to Interconnection Customer a non-binding good faith estimate of the cost and timeframe for completing the Interconnection Facilities Study. The Interconnection Facilities Study Agreement shall provide that Interconnection Customer shall compensate Transmission Provider for the actual cost of the Interconnection Facilities Study. Interconnection Customer shall execute the Interconnection Facilities Study Agreement and deliver the executed Interconnection Facilities Study Agreement to Transmission Provider within twenty (20) Calendar Days after its receipt, together with the required technical data, Readiness Milestone 4 and the Site Control requirements described in Section 7.7.7. Interconnection Customers that do not provide the Readiness Milestone (or additional security in lieu of the Readiness Milestone) and provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

8.2 Scope of Interconnection Facilities Study

The Interconnection Facilities Study shall specify and provide a non-binding estimate of the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the Phase 1 and Phase 2 (and/or Phase 3) Reports (as appropriate) in accordance with Good Utility Practice to physically and electrically connect the Interconnection Facilities to the Transmission System. The Interconnection Facilities Study shall also identify the electrical switching configuration of the connection equipment, including, without limitation: the transformer, switchgear,

meters, and other station equipment; the nature and estimated cost of any Transmission Provider's Interconnection Facilities and Network Upgrades necessary to accomplish the interconnection; and an estimate of the time required to complete the construction and installation of such facilities. The Interconnection Facilities Study will also identify any potential control equipment for requests for Interconnection Service that are lower than the Generating Facility Capacity.

8.3 Interconnection Facilities Study Procedures

Transmission Provider shall coordinate the Interconnection Facilities Study with any Affected System pursuant to Section 3.6 above.

Transmission Provider shall utilize existing studies to the extent practicable in performing the Interconnection Facilities Study. Transmission Provider shall use Reasonable Efforts to complete the study and issue a draft Interconnection Facilities Study report to Interconnection Customer within ninety (90) Calendar Days after acceptance of the Interconnection Facilities Agreement and Readiness Milestone 4.

At the request of Interconnection Customer, Transmission Provider shall notify Interconnection Customer as to the schedule status of the Interconnection Facilities Study. If Transmission Provider is unable to complete the Interconnection Facilities Study and issue a draft Interconnection Facilities Study report within the time identified, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required.

Interconnection Customer may, within thirty (30) Calendar Days after receipt of the draft Interconnection Facilities Study report, provide written comments to Transmission Provider, which Transmission Provider shall consider in completing the final Interconnection Facilities Study report. Transmission Provider shall issue the final Interconnection Facilities Study report within fifteen (15) Business Days of receiving Interconnection Customer's comments or promptly upon receiving Interconnection Customer's statement that it will not provide comments. Transmission Provider may reasonably extend such fifteen-day period upon notice to Interconnection Customer if Interconnection Customer's comments require Transmission Provider to perform additional analyses or make other significant modifications prior to the issuance of the final Interconnection Facilities Study report. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation, workpapers, and databases or data developed in the preparation of the Interconnection Facilities Study, subject to confidentiality arrangements consistent with

Section 13.1.

8.4 Meeting with Transmission Provider

Within ten (10) Business Days of providing a draft Interconnection Facilities Study report to Interconnection Customer, Transmission Provider and Interconnection Customer shall meet to discuss the results of the Interconnection Facilities Study, unless otherwise mutually agreed upon by the Parties.

8.5 Re-Study

If Re-Study of the Interconnection Facilities Study is required due to a higher or equal priority queued project dropping out of the Queue or a modification of a higher queued project pursuant to Section 4.4, Transmission Provider shall so notify Interconnection Customer(s) in writing. The Transmission Provider shall use Reasonable Efforts to complete the Re-Study in no longer than sixty (60) Calendar Days from the date of notice. Re-studies that require re-running the system impact study analysis may take longer than sixty days. Any cost of Re-Study shall be borne by the Interconnection Customer(s) being re-studied.

Section 9. Engineering & Procurement ('E&P') Agreement

Prior to executing an LGIA, an Interconnection Customer may, in order to advance the implementation of its interconnection, request and Transmission Provider shall offer the Interconnection Customer, an E&P Agreement that authorizes Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection. However, Transmission Provider shall not be obligated to offer an E&P Agreement if Interconnection Customer is in Dispute Resolution as a result of an allegation that Interconnection Customer has failed to meet any Readiness Milestones or comply with any prerequisites specified in other parts of the LGIP. The E&P Agreement is an optional procedure and it will not alter the Interconnection Customer's Queue Position or In-Service Date. The E&P Agreement shall provide for Interconnection Customer to pay the cost of all activities authorized by Interconnection Customer and to make advance payments or provide other satisfactory security for such costs.

Interconnection Customer shall pay the cost of such authorized activities and any cancellation costs for equipment that is already ordered for its interconnection, which cannot be mitigated as hereafter described, whether or not such items or equipment later become unnecessary. If Interconnection Customer withdraws its application for interconnection or either Party terminates the E&P Agreement, to the extent the equipment ordered can be canceled under reasonable terms, Interconnection Customer

shall be obligated to pay the associated cancellation costs. To the extent that the equipment cannot be reasonably canceled, Transmission Provider may elect: (i) to take title to the equipment, in which event Transmission Provider shall refund Interconnection Customer any amounts paid by Interconnection Customer for such equipment and shall pay the cost of delivery of such equipment, or (ii) to transfer title to and deliver such equipment to Interconnection Customer, in which event Interconnection Customer shall pay any unpaid balance and cost of delivery of such equipment.

Section 10. [Reserved for Future Use]

Section 11. Standard Large Generator Interconnection Agreement (LGIA)

11.1 Tender

Interconnection Customer shall tender comments on the draft Interconnection Facilities Study Report within thirty (30) Calendar Days of receipt of the report. Within thirty (30) Calendar Days after the comments are submitted or after the Interconnection Customer notifies Transmission Provider that it will provide no comments, Transmission Provider shall tender a draft LGIA, together with draft appendices. The draft LGIA shall be in the form of Transmission Provider's FERC-approved standard form LGIA, which is in Appendix 6. Interconnection Customer shall return the completed draft appendices within thirty (30) Calendar Days so that Transmission Provider may prepare a final LGIA. Transmission Provider shall tender a final LGIA, together with final appendices, within thirty (30) Calendar Days after the expiration of the thirty (30) Calendar Day window in which the Interconnection Customer is to return completed draft appendices.

11.2 Negotiation

Notwithstanding Section 11.1, at the request of Interconnection Customer Transmission Provider shall begin negotiations with Interconnection Customer concerning the appendices to the LGIA at any time after Interconnection Customer executes the Interconnection Facilities Study Agreement. Transmission Provider and Interconnection Customer shall negotiate concerning any disputed provisions of the appendices to the draft LGIA for not more than sixty (60) Calendar Days after tender of the final Interconnection Facilities Study Report. If Interconnection Customer determines that negotiations are at an impasse, Interconnection Customer may request termination of the negotiations at any time after tender of the

draft LGIA pursuant to Section 11.1 and request submission of the unexecuted LGIA with FERC or initiate Dispute Resolution procedures pursuant to Section 13.5. If Interconnection Customer requests termination of the negotiations, but within sixty (60) Calendar Days thereafter fails to request either the filing of the unexecuted LGIA or initiate Dispute Resolution, it shall be deemed to have withdrawn its Interconnection Request. Unless otherwise agreed by the Parties, if Interconnection Customer has not executed the LGIA, requested filing of an unexecuted LGIA, or initiated Dispute Resolution procedures pursuant to Section 13.5 within sixty (60) Calendar Days of tender of draft LGIA, it shall be deemed to have withdrawn its Interconnection Request. Transmission Provider shall provide to Interconnection Customer a final LGIA within fifteen (15) Business Days after the completion of the negotiation process.

11.3 Execution and Filing

Within fifteen (15) Business Days after receipt of the final LGIA, Interconnection Customer shall provide Transmission Provider (A) reasonable evidence that continued Site Control as defined in Section 7.7.7 and (B) post Readiness Milestone 5 security equal to nine (9) times that Interconnection Customer's share of the Definitive Interconnection Study Process study costs. If the actual study costs are not known at the time, study costs shall be estimated as the study deposit described in Section 3.4, and the M5 amount shall be updated when the study costs are known. If the Interconnection Customer does not reach Commercial Operation, upon payment of any final invoice, including any Withdrawal Penalty, Readiness Milestone 5 shall be refunded to the Interconnection Customer, including any accumulated interest, if applicable. If the Interconnection Customer reaches Commercial Operation, Readiness Milestone 5 is refunded to the Interconnection Customer including any accumulated interest, if applicable. Within fifteen (15) Business Days after receipt of the final LGIA, Interconnection Customer also shall provide reasonable evidence that one or more of the following milestones in the development of the Large Generating Facility, at Interconnection Customer election, has been achieved: (i) the execution of a contract for the supply or transportation of fuel to the Large Generating Facility (applicable to Generating Facilities that require fuel transportation; not available for wind or solar resources); (ii) the execution of a contract for the supply of cooling water to the Large Generating Facility (applicable to Generating Facilities that require cooling water for the production process; not available for wind or solar resources); (iii) execution of a contract for the engineering for, procurement of major equipment for, or construction of, the Large Generating Facility; (iv) execution of a contract for the sale of electric energy or capacity from the Large Generating Facility; or (v) application for an air, water, or land use

permit.

At the same time that Interconnection Customer makes its submission of the items identified above (due within fifteen (15) Business Days after receipt of the final LGIA), Interconnection Customer shall either: (i) execute two originals of the tendered LGIA and return them to Transmission Provider; or (ii) request in writing that Transmission Provider file with FERC an LGIA in unexecuted form. As soon as practicable, but not later than ten (10) Business Days after receiving either the two executed originals of the tendered LGIA (if it does not conform with a FERC-approved standard form of interconnection agreement) or the request to file an unexecuted LGIA, Transmission Provider shall file the LGIA with FERC, together with its explanation of any matters as to which Interconnection Customer and Transmission Provider disagree and support for the costs that Transmission Provider proposes to charge to Interconnection Customer under the LGIA. An unexecuted LGIA should contain terms and conditions deemed appropriate by Transmission Provider for the Interconnection Request. If the Parties agree to proceed with design, procurement, and construction of facilities and upgrades under the agreed-upon terms of the unexecuted LGIA, they may proceed pending FERC action.

11.4 Commencement of Interconnection Activities

If Interconnection Customer executes the final LGIA, Transmission Provider and Interconnection Customer shall perform their respective obligations in accordance with the terms of the LGIA, subject to modification by FERC. Upon submission of an unexecuted LGIA, Interconnection Customer and Transmission Provider shall promptly comply with the unexecuted LGIA, subject to modification by FERC.

Section 12. Construction of Transmission Provider's Interconnection Facilities and Network Upgrades

12.1 Schedule

Transmission Provider and Interconnection Customer shall negotiate in good faith concerning a schedule for the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades.

12.2 Construction Sequencing

12.2.1 General

In general, the In-Service Date of an Interconnection Customers seeking interconnection to the Transmission System will determine the sequence of construction of Network Upgrades.

12.2.2 Advance Construction of Network Upgrades that are an Obligation of an Entity other than Interconnection Customer

An Interconnection Customer with an LGIA, in order to maintain its In-Service Date, may request that Transmission Provider advance to the extent necessary the completion of Network Upgrades that: (i) were assumed in the Interconnection Studies for such Interconnection Customer, (ii) are necessary to support such In-Service Date, and (iii) would otherwise not be completed, pursuant to a contractual obligation of an entity other than Interconnection Customer that is seeking interconnection to the Transmission System, in time to support such In-Service Date. Upon such request, Transmission Provider will use Reasonable Efforts to advance the construction of such Network Upgrades to accommodate such request; provided that Interconnection Customer commits to pay Transmission Provider: (i) any associated expediting costs and (ii) the cost of such Network Upgrades.

Transmission Provider will refund to Interconnection Customer both the expediting costs and the cost of Network Upgrades, in accordance with Article 11.4 of the LGIA. Consequently, the entity with a contractual obligation to construct such Network Upgrades shall be obligated to pay only that portion of the costs of the Network Upgrades that Transmission Provider has not refunded to Interconnection Customer. Payment by that entity shall be due on the date that it would have been due had there been no request for advance construction. Transmission Provider shall forward to Interconnection Customer the amount paid by the entity with a contractual obligation to construct the Network Upgrades as payment in full for the outstanding balance owed to Interconnection Customer. Transmission Provider then shall refund to that entity the amount that it paid for the Network Upgrades, in accordance with Article 11.4 of the LGIA.

12.2.3 Advancing Construction of Network Upgrades that are Part of an Expansion Plan of the Transmission Provider An Interconnection Customer with an LGIA, in order to maintain its In-Service Date, may request that Transmission Provider advance to the extent necessary the

completion of Network Upgrades that: (i) are necessary to support such In-Service Date and (ii) would otherwise not be completed, pursuant to an expansion plan of Transmission Provider, in time to support such In-Service Date. Upon such request, Transmission Provider will use Reasonable Efforts to advance the construction of such Network Upgrades to accommodate such request; provided that Interconnection Customer commits to pay Transmission Provider any associated expediting costs. Interconnection Customer shall be entitled to transmission credits, if any, for any expediting costs paid.

12.2.4 Amended Definitive Interconnection System Impact Study

A Definitive Interconnection System Impact Study may be amended to determine the facilities necessary to support the requested In-Service Date. This amended study will include those transmission and Large Generating Facilities that are expected to be in service on or before the requested In-Service Date.

Section 13. Miscellaneous

13.1 Confidentiality

Confidential Information shall include, without limitation, all information relating to a Party's technology, research and development, business affairs, and pricing, and any information supplied by either of the Parties to the other prior to the execution of an LGIA.

Information is Confidential Information only if it is clearly designated or marked in writing as confidential on the face of the document, or, if the information is conveyed orally or by inspection, if the Party providing the information orally informs the Party receiving the information that the information is confidential.

If requested by either Party, the other Party shall provide in writing, the basis for asserting that the information referred to in this Section warrants confidential treatment, and the requesting Party may disclose such writing to the appropriate Governmental Authority. Each Party shall be responsible for the costs associated with affording confidential treatment to its information.

The release of Confidential Information shall be subject to Applicable Laws and Regulations and Applicable Reliability Standards.

13.1.1 Scope

Confidential Information shall not include information that the receiving Party can demonstrate: (1) is generally available to the public other than as a result of a disclosure by the receiving Party; (2) was in the lawful possession of the receiving Party on a non-confidential basis before receiving it from the disclosing Party; (3) was supplied to the receiving Party without restriction by a third party, who, to the knowledge of the receiving Party after due inquiry, was under no obligation to the disclosing Party to keep such information confidential; (4) was independently developed by the receiving Party without reference to Confidential Information of the disclosing Party; (5) is, or becomes, publicly known, through no wrongful act or omission of the receiving Party or Breach of the LGIA; or (6) is required, in accordance with Section 13.1.6, Order of Disclosure, to be disclosed by any Governmental Authority or is otherwise required to be disclosed by law or subpoena, or is necessary in any legal proceeding establishing rights and obligations under the LGIA. Information designated as Confidential Information will no longer be deemed confidential if the Party that designated the information as confidential notifies the other Party that it no longer is confidential.

13.1.2 Release of Confidential Information

Neither Party shall release or disclose Confidential Information to any other person, except to its Affiliates (limited by the Standards of Conduct requirements), employees, consultants, or to parties who may be or considering providing financing to or equity participation with Interconnection Customer, or to potential purchasers or assignees of Interconnection Customer, on a need-to-know basis in connection with these procedures, unless such person has first been advised of the confidentiality provisions of this Section 13.1 and has agreed to comply with such provisions. Notwithstanding the foregoing, a Party providing Confidential Information to any person shall remain primarily responsible for any release of Confidential Information in contravention of this Section 13.1.

13.1.3 Rights

Each Party retains all rights, title, and interest in the

Confidential Information that each Party discloses to the other Party. The disclosure by each Party to the other Party of Confidential Information shall not be deemed a waiver by either Party or any other person or entity of the right to protect the Confidential Information from public disclosure.

13.1.4 No Warranties

By providing Confidential Information, neither Party makes any warranties or representations as to its accuracy or completeness. In addition, by supplying Confidential Information, neither Party obligates itself to provide any particular information or Confidential Information to the other Party nor to enter into any further agreements or proceed with any other relationship or joint venture.

13.1.5 Standard of Care

Each Party shall use at least the same standard of care to protect Confidential Information it receives as it uses to protect its own Confidential Information from unauthorized disclosure, publication or dissemination. Each Party may use Confidential Information solely to fulfill its obligations to the other Party under these procedures or its regulatory requirements.

13.1.6 Order of Disclosure

If a court or a Government Authority or entity with the right, power, and apparent authority to do so requests or requires either Party, by subpoena, oral deposition, interrogatories, requests for production of documents, administrative order, or otherwise, to disclose Confidential Information, that Party shall provide the other Party with prompt notice of such request(s) or requirement(s) so that the other Party may seek an appropriate protective order or waive compliance with the terms of the LGIA. Notwithstanding the absence of a protective order or waiver, the Party may disclose such Confidential Information which, in the opinion of its counsel, the Party is legally compelled to disclose. Each Party will use Reasonable Efforts to obtain reliable assurance that confidential treatment will be accorded any Confidential Information so furnished.

13.1.7 Remedies

The Parties agree that monetary damages would be inadequate

to compensate a Party for the other Party's Breach of its obligations under this Section 13.1. Each Party accordingly agrees that the other Party shall be entitled to equitable relief, by way of injunction or otherwise, if the first Party Breaches or threatens to Breach its obligations under this Section 13.1, which equitable relief shall be granted without bond or proof of damages, and the receiving Party shall not plead in defense that there would be an adequate remedy at law. Such remedy shall not be deemed an exclusive remedy for the Breach of this Section 13.1, but shall be in addition to all other remedies available at law or in equity. The Parties further acknowledge and agree that the covenants contained herein are necessary for the protection of legitimate business interests and are reasonable in scope. No Party, however, shall be liable for indirect, incidental, or consequential or punitive damages of any nature or kind resulting from or arising in connection with this Section 13.1.

13.1.8 Disclosure to FERC, its Staff, or a State

Notwithstanding anything in this Section 13.1 to the contrary, and pursuant to 18 CFR section 1b.20, if FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to the LGIP, the Party shall provide the requested information to FERC or its staff, within the time provided for in the request for information. In providing the information to FERC or its staff, the Party must, consistent with 18 CFR section 388.112, request that the information be treated as confidential and non public by FERC and its staff and that the information be withheld from public disclosure. Parties are prohibited from notifying the other Party prior to the release of the Confidential Information to FERC or its staff. The Party shall notify the other Party to the LGIA when its is notified by FERC or its staff that a request to release Confidential Information has been received by FERC, at which time either of the Parties may respond before such information would be made public, pursuant to 18 CFR section 388.112. Requests from a state regulatory body conducting a confidential investigation shall be treated in a similar manner, consistent with applicable state rules and regulations.

Subject to the exception in Section 13.1.8, any information that a Party claims is competitively sensitive, commercial or

financial information ("Confidential Information") shall not be disclosed by the other Party to any person not employed or retained by the other Party, except to the extent disclosure is (i) required by law; (ii) reasonably deemed by the disclosing Party to be required to be disclosed in connection with a dispute between or among the Parties, or the defense of litigation or dispute; (iii) otherwise permitted by consent of the other Party, such consent not to be unreasonably withheld; or (iv) necessary to fulfill its obligations under this LGIP or as a transmission service provider or a Control Area operator including disclosing the Confidential Information to an RTO or ISO or to a subregional, regional or national reliability organization or planning group. The Party asserting confidentiality shall notify the other Party in writing of the information it claims is confidential. Prior to any disclosures of the other Party's Confidential Information under this subparagraph, or if any third party or Governmental Authority makes any request or demand for any of the information described in this subparagraph, the disclosing Party agrees to promptly notify the other Party in writing and agrees to assert confidentiality and cooperate with the other Party in seeking to protect the Confidential Information from public disclosure by confidentiality agreement, protective order or other reasonable measures.

- 13.1.10 This provision shall not apply to any information that was or is hereafter in the public domain (except as a result of a Breach of this provision).
- 13.1.11 Transmission Provider shall, at Interconnection Customer's election, destroy, in a confidential manner, or return the Confidential Information provided at the time of Confidential Information is no longer needed.

13.2 Delegation of Responsibility

Transmission Provider may use the services of subcontractors as it deems appropriate to perform its obligations under this LGIP. Transmission Provider shall remain primarily liable to Interconnection Customer for the performance of such subcontractors and compliance with its obligations of this LGIP. The subcontractor shall keep all information provided confidential and shall use such information solely for the performance of such obligation for which it was provided and no other purpose.

13.3 Obligation for Study Costs and Withdrawal Penalty

Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Interconnection Studies and the Withdrawal Penalty, as applicable. Any difference between the Initial Deposit and the actual cost of the Interconnection Study(ies) shall be paid by or refunded, except as otherwise provided herein, to Interconnection Customer. Any invoices for Interconnection Studies shall include a detailed and itemized accounting of the cost of each Interconnection Study as well as the Withdrawal Penalty, if applicable. Interconnection Customer shall pay any such undisputed costs within thirty (30) Calendar Days of receipt of an invoice therefor. Transmission Provider shall not be obligated to perform or continue to perform any studies unless Interconnection Customer has paid all undisputed amounts in compliance herewith. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided under this LGIP to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

13.4 Third Parties Conducting Studies

If (i) at the time of the signing of an Interconnection Study Agreement there is disagreement as to the estimated time to complete an Interconnection Study, (ii) Interconnection Customer receives notice pursuant to Sections 7.4 or 8.3 that Transmission Provider will not complete an Interconnection Study within the applicable timeframe for such Interconnection Study, or (iii) Interconnection Customer receives neither the Interconnection Study nor a notice under Sections 7.4 or 8.3 within the applicable timeframe for such Interconnection Study, then Interconnection Customer may require Transmission Provider to utilize a third party consultant reasonably acceptable to Interconnection Customer and Transmission Provider to perform such Interconnection Study under the direction of Transmission Provider. At other times, Transmission Provider may also utilize a third party consultant to perform such Interconnection Study, either in response to a general request of Interconnection Customer, or on its own volition.

In all cases, use of a third party consultant shall be in accord with Article 26 of the LGIA (Subcontractors) and limited to situations

where Transmission Provider determines that doing so will help maintain or accelerate the study process for Interconnection Customer's pending Interconnection Request and not interfere with Transmission Provider's progress on Interconnection Studies for other pending Interconnection Requests. In cases where Interconnection Customer requests use of a third party consultant to perform such Interconnection Study, Interconnection Customer and Transmission Provider shall negotiate all of the pertinent terms and conditions, including reimbursement arrangements and the estimated study completion date and study review deadline. Transmission Provider shall convey all workpapers, data bases, study results and all other supporting documentation prepared to date with respect to the Interconnection Request as soon as practicable upon Interconnection Customer's request subject to the confidentiality provision in Section 13.1. In any case, such third party contract may be entered into with either Interconnection Customer or Transmission Provider at Transmission Provider's discretion. In the case of (iii) Interconnection Customer maintains its right to submit a claim to Dispute Resolution to recover the costs of such third party study. Such third party consultant shall be required to comply with this LGIP, Article 26 of the LGIA (Subcontractors), and the relevant Tariff procedures and protocols as would apply if Transmission Provider were to conduct the Interconnection Study and shall use the information provided to it solely for purposes of performing such services and for no other purposes. Transmission Provider shall cooperate with such third party consultant and Interconnection Customer to complete and issue the Interconnection Study in the shortest reasonable time.

13.5 Disputes

13.5.1 Submission

In the event either Party has a dispute, or asserts a claim, that arises out of or in connection with the LGIA, the LGIP, or their performance, such Party (the "disputing Party") shall provide the other Party with written notice of the dispute or claim ("Notice of Dispute"). Such dispute or claim shall be referred to a designated senior representative of each Party for resolution on an informal basis as promptly as practicable after receipt of the Notice of Dispute by the other Party. In the event the designated representatives are unable to resolve the claim or dispute through unassisted or assisted negotiations within thirty (30) Calendar Days of the other Party's receipt of the

Notice of Dispute, such claim or dispute may, upon mutual agreement of the Parties, be submitted to arbitration and resolved in accordance with the arbitration procedures set forth below. In the event the Parties do not agree to submit such claim or dispute to arbitration, each Party may exercise whatever rights and remedies it may have in equity or at law consistent with the terms of this LGIA.

13.5.2 External Arbitration Procedures

Any arbitration initiated under these procedures shall be conducted before a single neutral arbitrator appointed by the Parties. If the Parties fail to agree upon a single arbitrator within ten (10) Calendar Days of the submission of the dispute to arbitration, each Party shall choose one arbitrator who shall sit on a three-member arbitration panel. The two arbitrators so chosen shall within twenty (20) Calendar Days select a third arbitrator to chair the arbitration panel. In either case, the arbitrators shall be knowledgeable in electric utility matters, including electric transmission and bulk power issues, and shall not have any current or past substantial business or financial relationships with any party to the arbitration (except prior arbitration). The arbitrator(s) shall provide each of the Parties an opportunity to be heard and, except as otherwise provided herein, shall conduct the arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("Arbitration Rules") and any applicable FERC regulations or RTO rules; provided, however, in the event of a conflict between the Arbitration Rules and the terms of this Section 13, the terms of this Section 13 shall prevail.

13.5.3 Arbitration Decisions

Unless otherwise agreed by the Parties, the arbitrator(s) shall render a decision within ninety (90) Calendar Days of appointment and shall notify the Parties in writing of such decision and the reasons therefor. The arbitrator(s) shall be authorized only to interpret and apply the provisions of the LGIA and LGIP and shall have no power to modify or change any provision of the LGIA and LGIP in any manner. The decision of the arbitrator(s) shall be final and binding upon the Parties, and judgment on the award may be entered in any court having jurisdiction. The decision of the arbitrator(s) may be appealed solely on the grounds that the conduct of the

arbitrator(s), or the decision itself, violated the standards set forth in the Federal Arbitration Act or the Administrative Dispute Resolution Act. The final decision of the arbitrator must also be filed with FERC if it affects jurisdictional rates, terms and conditions of service, Interconnection Facilities, or Network Upgrades.

13.5.4 Costs

Each Party shall be responsible for its own costs incurred during the arbitration process and for the following costs, if applicable: (1) the cost of the arbitrator chosen by the Party to sit on the three member panel and one half of the cost of the third arbitrator chosen; or (2) one half the cost of the single arbitrator jointly chosen by the Parties.

13.5.5 Non-Binding Dispute Resolution Procedures.

If a Party has submitted a Notice of Dispute pursuant to section 13.5.1, and the Parties are unable to resolve the claim or dispute through unassisted or assisted negotiations within the thirty (30) Calendar Days provided in that section, and the Parties cannot reach mutual agreement to pursue the section 13.5 arbitration process, a Party may request that Transmission Provider engage in Non-binding Dispute Resolution pursuant to this section by providing written notice to Transmission Provider ("Request for Non-binding Dispute Resolution"). Conversely, either Party may file a Request for Non-binding Dispute Resolution pursuant to this section without first seeking mutual agreement to pursue the section 13.5 arbitration process. The process in section 13.5.5 shall serve as an alternative to, and not a replacement of, the section 13.5 arbitration process. Pursuant to this process, a Transmission Provider must within 30 Calendar Days of receipt of the Request for Non-binding Dispute Resolution appoint a neutral decision-maker that is an independent subcontractor that shall not have any current or past substantial business or financial relationships with either Party. Unless otherwise agreed by the Parties, the decision-maker shall render a decision within sixty (60) Calendar Days of appointment and shall notify the Parties in writing of such decision and reasons therefore. This decision-maker shall be authorized

only to interpret and apply the provisions of the LGIP and LGIA and shall have no power to modify or change any provision of the LGIP and LGIA in any manner. The result reached in this process is not binding, but, unless otherwise agreed, the Parties may cite the record and decision in the non-binding dispute resolution process in future dispute resolution processes, including in a section 13.5 arbitration, or in a Federal Power Act section 206 complaint. Each Party shall be responsible for its own costs incurred during the process and the cost of the decision-maker shall be divided equally among each Party to the dispute.

13.6 Local Furnishing Bonds

13.6.1 Transmission Providers That Own Facilities Financed by Local Furnishing Bonds

This provision is applicable only to a Transmission Provider that has financed facilities for the local furnishing of electric energy with tax-exempt bonds, as described in Section 142(f) of the Internal Revenue Code ("local furnishing bonds"). Notwithstanding any other provision of this LGIA and LGIP, Transmission Provider shall not be required to provide Interconnection Service to Interconnection Customer pursuant to this LGIA and LGIP if the provision of such Transmission Service would jeopardize the tax-exempt status of any local furnishing bond(s) used to finance Transmission Provider's facilities that would be used in providing such Interconnection Service.

13.6.2 Alternative Procedures for Requesting Interconnection Service

If Transmission Provider determines that the provision of Interconnection Service requested by Interconnection Customer would jeopardize the tax exempt status of any local furnishing bond(s) used to finance its facilities that would be used in providing such Interconnection Service, it shall advise the Interconnection Customer within thirty (30) Calendar Days of receipt of the Interconnection Request.

Interconnection Customer thereafter may renew its request for interconnection using the process specified in Section 5.2(ii) of the Transmission Provider's Tariff.

Filed Date: 05/31/2023

At-A-Glance Reference Sheet

Milestone	Total Security Required	Total Security Required	Demonstration of	Site Control
	(Multiple of Section 3.4	(Multiple of Section 3.4	Site Control for All	of ICIF's
	Study Deposit) If	Study Deposit)	Fuel Types	
	Demonstration of	If Demonstration of		
	Readiness	Readiness IS NOT		
	<u>IS</u> Provided	Provided		
M1	1x	2x	50%	0%
M2	1x	3x	50%	0%
M3	1x	5x	60%	0%
M4	1x	7x	75%	0%
M5	9x	9x	90%	50%

Phase 1: Power Flow/Voltage: Within 90 calendar days

- Transmission Provider to perform Power Flow and Voltage Analyses.
- Transmission Provider to complete Phase 1 report within 90 calendar days after the close of the Queue Cluster Window.
- Transmission Provider to hold results meeting within 10 business days of posting of DISIS Phase 1 results.
- Interconnection Customer to demonstrate M2 Readiness within 30 calendar days following the Phase 1 report.

Phase 2: Stability/Short Circuit: Within 150 calendar days

- Transmission Provider to complete Phase 2 analyses within 150 calendar days after receipt of all Interconnection Customers' executed Interconnection System Impact Study Agreements.
- Transmission Provider to hold Phase 2 results meeting within 10 business days of posting of Phase 2 report.
- Interconnection Customer to demonstrate M3 (if Re-Study is necessary) within 30 calendar days following the Phase 2 report, or demonstrate M4 Readiness (if no Re-Study is necessary) within 20 calendar days following Transmission Provider's notice that no Re-Study is needed.

**Phase 3: Iterative Cluster System Impact Re-Study: Within 150 calendar days

May not be necessary

- If a Re-Study is needed, Transmission Provider perform Phase 3 Re-Study within 150 calendar days after acceptance of all Interconnection Customers' executed Interconnection System Impact Study Agreements.
- Transmission Provider to hold results meeting within 10 business days of posting results.
- Interconnection Customer to demonstrate M4 Readiness within 20 calendar days following Transmission Provider's notice that no further System Impact Re-Study is needed.

Phase 4: Facilities Study: Within 90 calendar days

- Transmission Provider to complete Facilities Study, complete and submit draft Facilities Study Report to Interconnection Customer within 90 calendar days after acceptance of all Interconnection Facilities Study Agreements.
- Transmission Provider to hold results meeting within 10 business days of posting of Phase 4 report.
- Interconnection Customer has opportunity to provide written comments of Facilities Study Report to Transmission Provider within 30 calendar days of receipt of draft Facilities Study Report.

Filed Date: 05/31/2023

• Transmission Provider to issue Final Facilities Study Report to Customer within 15 business days of receiving Interconnection Customer's comments or promptly upon receiving Interconnection Customer's statement that it will not provide comments.

Phase 5: LGIA

- Transmission Provider to provide Interconnection Customer with draft LGIA within 30 calendar days of receipt of Interconnection Customer's Facilities Study comments.
- Interconnection Customer to return completed draft appendices within 30 calendar days of receipt of draft LGIA.
- Deadline for LGIA negotiations to be completed is within 60 calendar days after tender by Transmission Provider of the final Interconnection Facilities Study Report.
- Interconnection Customer to satisfy Readiness Milestone 5 within 15 business days of receiving final LGIA.
- Deadline for Interconnection Customer to execute the LGIA (or instruct that it be filed unexecuted) is 15 business days after receipt of the final LGIA.

Appendices

APPENDIX 1 to LGIP INTERCONNECTION REQUEST FOR A LARGE GENERATING FACILITY

1.	The undersigned Interconnection Customer submits this request to interconnect its Large Generating Facility with Transmission Provider's Transmission System pursuant to a Tariff.					
2.	This Interconnection Request is for (check one): A proposed new Large Generating Facility. An increase in the generating capacity or a Material Modification of an existing Generating Facility. A Generating Facility proposed for inclusion in a resource solicitation process.					
3.	The type of interconnection service requested (check one): Energy Resource Interconnection Service Network Resource Interconnection Service					
4.	Interd	Interconnection Customer provides the following information:				
	a.	Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility;				
	b.	Maximum summer at degrees C and winter at degrees C megawatt electrical output of the proposed new Large Generating Facility or the amount of megawatt increase in the generating capacity of an existing Generating Facility;				
	C.	General description of the equipment configuration;				
	d.	Commercial Operation Date (Day, Month, and Year);				
	e.	Name, address, telephone number, and e-mail address of Interconnection Customer's contact person;				
	f.	Approximate location of the proposed Point of Interconnection (optional);				
	g.	Interconnection Customer Data (set forth in Attachment A)				
	h. i.	Primary frequency response operating range for electric storage resources. Requested capacity (in MW) of Interconnection Service (if lower than the Generating Facility Capacity).				
5.	Interd LGIP	connection Customer provides applicable study deposit amount as specified in the				
		\$160,000 for requests of less than 75 MW \$250,000 for requests of 75 MW and greater				
6.	Interd	connection Customer provides Readiness Milestone 1 (M1) as specified in the LGIP.				

M1 is satisfied by any one of the three options below (also described in the LGIP) at Interconnection Customer's option. M1 may also be satisfied by providing additional security described in Section 7.7.5 in lieu of providing one of the three options to

demonstrate readiness.

- i. Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
- ii. Reasonable evidence the project has been selected in a Resource Plan or Resource Solicitation Process; or
- iii. Provisional Large Generator Interconnection Agreement accepted for filing with FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.
- Interconnection Customer provides security equal to one times the study deposit 7. described in Section 3 of the LGIP in the form of an irrevocable letter of credit or cash.
- 8. If requesting NRIS: Interconnection Customer provides the expected point of delivery to deliver within the Transmission Provider's Control Area.
- 9. Interconnection Customer provides Evidence of Site Control as specified in the LGIP and Transmission Provider's business practices posted on OASIS, as applicable.
- 10. This Interconnection Request shall be submitted to the representative indicated below:

[To be completed by Transmission Provider]

11. Representative of Interconnection Customer to contact:

[To be completed by Interconnection Customer]

12. This Interconnection Request is submitted by:

Name of Interconnection Customer:
By (signature):
Name (type or print):
Title:
Date:

Attachment A to Appendix 1 Interconnection Request

LARGE GENERATING FACILITY DATA

	UNIT RATINGS	
kVA	°F	Voltage
Power Factor		
Speed (RPM)		
Short Circuit Ratio	Frequency, Hertz	
Stator Amperes at Rated kVA		Field Volts
Max Turbine MW	°F	
Primary frequency resp	onse operating range for	electric storage resources
Minimum State of Cha Maximum State of Cha		
Maximum State of Cha	<u></u>	
COMBINED TUR	RBINE-GENERATOR-EXCI	TER INERTIA DATA
Inertia Constant, H = Moment-of-Inertia, WR ² =	kW sec/kVA	
wionient-or-mentia, wix =	ID. II.	
REACT	ANCE DATA (PER UNIT-R	ATED KVA)
DII	RECT AXIS QUADRATURE	AXIS
Synchronous – saturated	X _{dv}	X _{qv}
Synchronous – unsaturated	X_{di}	X _{qi}
Transient – saturated	X' _{dv}	X' _{qv}
Transient – unsaturated	X' _{di}	X' _{qi}
Subtransient – saturated	X" _{dv}	X" _{qv}
Subtransient – unsaturated	X" _{di}	X" _{qi}
Negative Sequence – saturated	X2 _v	
Negative Sequence – unsaturate	ed X2 _i	
Zero Sequence – saturated	X0 _v	
Zero Sequence – unsaturated	X0 _i	
Leakage Reactance	XI_m	

FIELD TIME CONSTANT DATA (SEC)

Open Circuit	I do	I 'qo
Three-Phase Short Circuit Transic	ent T' _{d3}	T' _q
Line to Line Short Circuit Transier	nt T' _{d1}	
Short Circuit Subtransient	T" _d	T" _q
Open Circuit Subtransient	T' _{d2}	
Line to Neutral Short Circuit Trans	sient T" _{do}	T" _{qo}
ARMAT	URE TIME CONSTANT	DATA (SEC)
Line to Line Short Circuit	Га3 Га2 Га1	
NOTE: If requested information is	not applicable, indicate	by marking "N/A."
	BILITY AND PLANT CO	
ARMATURE V	VINDING RESISTANCE	E DATA (PER UNIT)
Negative I	₹ ₁ ₹ ₂ ₹ ₀	
Rotor Short Time Thermal Capac Field Current at Rated kVA, Arma Field Current at Rated kVA and A Three Phase Armature Winding C Field Winding Resistance = Armature Winding Resistance (Pe	uture Voltage and PF =_ rmature Voltage, 0 PF = Capacitance =m ohms°C	=amps nicrofarad
	CURVES	
Provide Saturation, Vee, Read Designate normal and emergency		city Temperature Correction curves erating range for multiple curves.
GENERATOR S	STEP-UP TRANSFORM	IER DATA RATINGS
Capacity Self-cooled/Ma	ximum Nameplate	
/	kVA	

Note: A completed General Electric Company Power Systems Load Flow (PSLF) data sheet or

List of adjustable setpoints for the protective equipment or software:

other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate to the proposed device, then they shall be provided with the Interconnection Request and discussed at Scoping Meeting.

Project Information: Site Control and Adequacy

Total acres required to construct the Generating Facility:
Total acres under site control for the Generating Facility at the time of application:
Is Site Control required for Interconnection Facilities, i.e. transmission gen-tie or substation, to interconnect the Generating Facility? $__$ Y $__$ N
If yes, how many miles of gen-tie right-of-way are required? What is the total number of acres required to build the gen-tie? How many miles of gen-tie right-of-way are under Site Control at the time of this application?
List any local, state, or federal government permits required to construct the Generating Facility and any applicable Interconnection Facilities, i.e. transmission gen-tie:

INDUCTION GENERATORS

(*)	Field Volts:
(*)	Field Amperes:
	Motoring Power (kW):
(*)	Neutral Grounding Resistor (If Applicable:
(*)	1 ₂ ² t or K (Heating Time Constant):
(*)	Rotor Resistance:
(*)	Stator Resistance:
(*)	Stator Reactance:
(*)	Rotor Reactance:
(*)	Magnetizing Reactance:
(*)	Short Circuit Reactance:
(*)	Exciting Current:
(*)	Temperature Rise:
(*)	Frame Size:
	Design Letter:
	Reactive Power Required In Vars (No Load):
	Reactive Power Required In Vars (Full Load):
(*)	Total Rotating Inertia, H:Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Interconnection Request to determine if the information designated by (*) is required.

extended.

APPENDIX 2 to LGIP DEFINITIVE INTERCONNECTION SYSTEM IMPACT STUDY AGREEMENT

and be	etween the	State	ofa a _ , ("Transmi	, a ission Provide	, , er"). Int	day of organized a of of organized a off organized a off organized a contraction of the off organized as the off or	and existing on Custon the laws oustomer and	, 20 by under the laws omer,") and of the State of d Transmission
				REC	ITALS	;		
WHEREAS , Interconnection Customer is proposing to develop a Large Generating Facility or generating capacity addition to an existing Generating Facility consistent with the Interconnection Request submitted by Interconnection Customer dated; and								
	WHEREAS , Interconnection Customer desires to interconnect the Large Generating Facility with the Transmission System; and					ing Facility with		
WHEREAS , Interconnection Customer has requested Transmission Provider to perform a Definitive Interconnection System Impact Study to assess the impact of interconnecting the Large Generating Facility to the Transmission System, and of any Affected Systems; and								
WHEREAS , Interconnection Customer commits to provide certain Readiness Milestones through the Definitive Interconnection Study process as described in the LGIP.								
NOW, THEREFORE , in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:								
1.0	When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.					d shall have the		
2.0	a Defi					ssion Provider sh udy consistent wi		
3.0		•		e Interconnec ttachment A to		ystem Impact Sti Agreement.	udy shall be	subject to the
4.0	inform to any reserv Custo during Interc	nation proversity modification with modification with the contraction model in the contraction m	vided by Intactions in acciright to renay reason ourse of Customer	terconnection cordance with quest additionably become the Definitive makes a pern	Custon Sectional te necession nissible	t Study shall be mer in the Interco on 4.4 of the LG chnical informat sary consistent erconnection Sy e modification in	onnection R GIP. Transm tion from I with Good ystem Imp accordance	equest, subject ission Provider Interconnection Utility Practice pact Study. If with the LGIP,

5.0 The Definitive Interconnection System Impact Study reports (though Phase 2, and including Phase 3 if applicable) shall provide the following information as appropriate:

- identification of any circuit breaker short circuit capability limits exceeded as a result of the interconnection;
- identification of any thermal overload or voltage limit violations resulting from the interconnection;
- identification of any instability or inadequately damped response to system disturbances resulting from the interconnection; and
- description and non-binding, good faith estimated cost of facilities required to interconnect the Large Generating Facility to the Transmission System and to address the identified short circuit, instability, and power flow issues.
- 6.0 Interconnection Customer shall provide the deposit as specified in Section 3.4 of the LGIP for the performance of the Definitive Interconnection System Impact Study and the Interconnection Facilities Study. Transmission Provider's good faith estimate for the time of completion of the Definitive Interconnection System Impact Study (Phase 2) is [insert date].

Upon receipt of the Interconnection Facilities Study results (Phase 4 Results), or withdrawal of the Interconnection Request, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Definitive Interconnection System Impact Study and the actual costs of the Interconnection Facilities Study, and the Withdrawal Penalty, as applicable, allocated according to the LGIP.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, except as otherwise provided herein. As provided in the LGIP, Interconnection Customer has thirty (30) Calendar Days of receipt of an invoice from Transmission Provider to pay any undisputed costs. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

7.0 Miscellaneous. The Definitive Interconnection System Impact Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, that are consistent with regional practices, Applicable Laws and Regulations and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS THEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

Ву:	_ By:
Title:	Title:
Date:	_ Date:
[Insert name of Interconnection Custor	ner]
Ву:	_
Title:	
Date:	

[Insert name of Transmission Provider or Transmission Owner, if applicable]

Attachment A to Appendix 2 **Definitive Interconnection System Impact Study Agreement**

ASSUMPTIONS USED IN CONDUCTING THE **DEFINITIVE INTERCONNECTION SYSTEM IMPACT STUDY**

The Definitive Interconnection System Impact Study shall be based upon the information set forth in the Interconnection Request(s) and results of applicable prior studies, subject to any modifications in accordance with Section 4.4 of the LGIP, and the following assumptions:

Designation of Point of Interconnection and configuration to be studied.

[Above assumptions to be completed by Interconnection Customer and other assumptions to be provided by Interconnection Customer and Transmission Provider]

APPENDIX 3 to LGIP INTERCONNECTION FACILITIES STUDY AGREEMENT

THIS	AGREE	MENT i	s made a	nd entered int	o this_	day of_		, 20	by
and b	etween_			, a		organized ar	nd existing u	nder the	laws
of	the		of			("Interconnection			
			a			existing under	the laws of	the Sta	te of
			_, ("Trans	mission Provid	ler "). Ir	nterconnection Cu	stomer and [*]	Transmi	ssion
Provi	der each	n may be	referred	to as a "Party,"	or coll	ectively as the "Pa	rties."		
				RE	CITALS	6			
gene	rating ca	pacity a	ddition to a	an existing Gei	nerating	to develop a Larg g Facility consister ed; a	it with the Int	•	-
	-			stomer desire (check one o		erconnect the Larg	e Generatino	g Facility	with
		Networ	k Resour	ce Interconnec	tion Se	rvice			
		Energy	Resource	e Interconnecti	on Ser	vice:			

WHEREAS, Transmission Provider has completed a Definitive Interconnection System Impact Study (the "System Impact Study") and provided the results of said study to Interconnection Customer; and

WHEREAS, Interconnection Customer has requested Transmission Provider to perform an Interconnection Facilities Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the Definitive Interconnection System Impact Study in accordance with Good Utility Practice to physically and electrically connect the Large Generating Facility to the Transmission System.

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- 1.0 When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- 2.0 Interconnection Customer elects and Transmission Provider shall cause an Interconnection Facilities Study consistent with this LGIP to be performed in accordance with the Tariff.
- 3.0 The scope of the Interconnection Facilities Study shall be subject to the assumptions set forth in Attachment A to this Agreement.
- 4.0 The Interconnection Facilities Study report (i) shall provide a description of, estimated cost of, schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the Definitive Interconnection System Impact Study.

- 5.0 Interconnection Customer shall meet the requirements specified under the LGIP prior to the performance of the Interconnection Facilities Study. The time for completion of the Interconnection Facilities Study is specified in Attachment A.
- 6.0 Interconnection Customer shall have provided the deposit as specified in Section 3.4.1 of the LGIP for the performance of the Definitive Interconnection System Impact Study and the Interconnection Facilities Study.

Upon receipt of the Interconnection Facilities Study results (Phase 4 Results), Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Definitive Interconnection System Impact Study and the actual costs of the Interconnection Facilities Study, which includes costs allocated according to the LGIP, and the Withdrawal Penalty calculated pursuant to Section 3.7.1, if applicable.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, except as otherwise provided herein. As provided in the LGIP, Interconnection Customer has thirty (30) Calendar Days of receipt of an invoice from Transmission Provider to pay any undisputed costs. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]					
By:	_ By:				
Title:	_Title:				
Date:	_ Date:				
[Insert name of Interconnection Custome	er]				
By:					
Title:					
Date:	_				

Attachment A to Appendix 3 **Interconnection Facilities Study Agreement**

DATA FORM TO BE PROVIDED BY INTERCONNECTION CUSTOMER WITH THE INTERCONNECTION FACILITIES STUDY AGREEMENT

Provide location plan and simplified one-line diagram of the plant and station facilities. For staged projects, please indicate future generation, transmission circuits, etc.

One set of metering is required for each generation connection to the new ring bus or existing Transmission Provider station. Number of generation connections:

On the one line diagram indicate the generation capacity attached at each metering location. (Maximum load on CT/PT)

On the one line diagram indicate the location of auxiliary power. (Minimum load on CT/PT) Amps
Will an alternate source of auxiliary power be available during CT/PT maintenance?Yes No
Will a transfer bus on the generation side of the metering require that each meter set be designed for the total plant generation? Yes No (Please indicate on one line diagram).
What type of control system or PLC will be located at Interconnection Customer's Large Generating Facility?
What protocol does the control system or PLC use?
Please provide a 7.5-minute quadrangle of the site. Sketch the plant, station, transmission line, and property line.
Physical dimensions of the proposed interconnection station:
Bus length from generation to interconnection station:
Line length from interconnection station to Transmission Provider's transmission line.
Tower number observed in the field. (Painted on tower leg)*
Number of third party easements required for transmission lines*:

^{*} To be completed in coordination with Transmission Provider.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Is the Large Generating Facility in Transmis	ssion Provider's service area?	
Yes No	Local	provider:
Please provide proposed schedule dates:		
Begin Construction	Date:	
Generator step-up transformer receives back feed power	Date:	
Generation Testing	Date:	
Commercial Operation	Date:	

4.0

APPENDIX 4.1 to LGIP

Transitional Interconnection Facilities Study Agreement

THIS A	AGREE	MENT i	s made an	nd entered into	o this_	day	of	, 2	0	_ by
of	the	State	ofa	, a 	, ,	organize ("Interconnectionexisting underconnection	tion der the la	Customer,") aws of the	State	aws and e of
Provide	er each	may be	_, ("Transm referred to	nission Provid o as a "Party,"	er"). In or coll	terconnection ectively as the	Custome "Parties.	er and Trans	smis	sion
				REC	CITALS	3				
genera	iting ca	pacity ac	ddition to a	n existing Ger	erating	to develop a Facility consi	stent with	•		-
			ection Cus tem; and	tomer desires	to inte	rconnect the L	₋arge Ge	nerating Fac	ility	with
proces engine Systen	sing its ering, _l n Impa	Intercor procurer ct Study	nnection Fa nent and c in accord	acilities Study construction w dance with G	to spectork new ood Ut	sted Transmi cify and estimated eded to imple ility Practice is ssion System;	ate the co ement the to physic	ost of the equesion	uipm s of	ent, the
an Inte	erconne	ection F	acilities St		ent pri	I and Transmi or to the effe and				
						certain require onnection Fac				
		EFORE, reed as		ration of and	subjec	to the mutua	l covenai	nts containe	d he	rein
1.0			_	•	•	alization, the t s FERC-appro	•		nave	the
2.0						ssion Provide with this LGIP				
3.0	forth in	n Attach	ment A to		ent whi	udy shall be so ch shall be th ement.				

The Interconnection Facilities Study report (i) shall provide a description, estimated cost of (consistent with Attachment A), schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the most recently published System Impact Study.

- Interconnection Customer has met certain requirements described in the LGIP. The time 5.0 for completion of the Interconnection Facilities Study is specified in Attachment A.
- 6.0 Interconnection Customer shall have previously provided the deposit for the performance of the Interconnection Facilities Study.
 - Upon receipt of the Interconnection Facilities Study results, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Interconnection Facilities Study, as allocated pursuant to the LGIP.
 - Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.
- 7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]						
Ву:	_ By:					
Title:	Title:					
Date:	_ Date:					
[Insert name of Interconnection Custon	ner]					
By:	_					
Title:	-					
Date:						

Attachment A to Appendix 4.1 Transitional Interconnection Facilities Study Agreement

ASSUMPTIONS USED IN CONDUCTING THE TRANSITIONAL INTERCONNECTION FACILITIES STUDY

[Assumptions to be completed by Interconnection Customer and Transmission Provider]

APPENDIX 4.2 to LGIP Transitional Cluster Study Agreement

THIS	AGREE	MENT	is made an	d entered into this	s day of_	, 20 l	bу
and b	etween_			, a	organized ar	nd existing under the law	٧S
of	the	State	of	,	("Interconnection	Customer,") ar	nd
			a .		existing under	the laws of the State	of
			<u> </u>	,		stomer and Transmission	วท
Provid	ler each	may be	referred to	as a "Party," or c	ollectively as the "Pa	rties."	
				D=0:T4			
				RECITA	LS		
gener	ating ca _l	oacity a	ddition to ar	n existing Generat	•	ge Generating Facility of the state of the second section in the second	
				tomer desires to ir (check one only):	nterconnect the Larg	e Generating Facility wi	th
		Networ	k Resource	e Interconnection S	Service		
		Energy	Resource	Interconnection S	ervice; and		

WHEREAS, Interconnection Customer has requested Transmission Provider to perform a "Transitional Cluster Study," which is a combined system impact and interconnection facility Cluster Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to physically and electrically connect the Large Generating Facility to the Transmission System; and

WHEREAS, Interconnection Customer has provided certain requirements described in the LGIP including a deposit of five million dollars (\$5,000,000) on expected Transmission Provider's Interconnection Facilities and Network Upgrades; and

WHEREAS, Interconnection Customer has a valid Queue Position as of the effective date of Transmission Provider's Definitive Interconnection Study Process; and

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- 1.0 When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- 2.0 Interconnection Customer elects and Transmission Provider shall cause to be performed a combined system impact and interconnection facility Cluster Study.
- 3.0 The Transitional Cluster Study shall be based upon the technical information provided by Interconnection Customer in the Interconnection Request. Transmission Provider reserves the right to request additional technical information from Interconnection Customer as may reasonably become necessary consistent with Good Utility Practice during the course of the Transitional Cluster Study and Interconnection Customer shall

provide such data as quickly as reasonable.

- 4.0 The Transitional Cluster Study report shall provide the following information:
 - identification of any circuit breaker short circuit capability limits exceeded as a result of the interconnection;
 - identification of any thermal overload or voltage limit violations resulting from the interconnection:
 - identification of any instability or inadequately damped response to system disturbances resulting from the interconnection and
 - shall provide a description, estimated cost of, schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the most recently published System Impact Study.
- 5.0 Interconnection Customer has met certain requirements described in the LGIP.
- 6.0 Interconnection Customer shall have previously provided a deposit for the performance of Interconnection Studies.
 - Upon receipt of the Transitional Cluster Study results, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Transitional Cluster Study.
 - Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.
- 7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]					
Ву:	By:				
Title:	_ Title:				
Date:	_ Date:				
[Insert name of Interconnection Custon	mer]				
Ву:					
Title:					
Date:					

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Attachment A to Appendix 4.2 **Transitional Cluster Study Agreement**

ASSUMPTIONS USED IN CONDUCTING THE TRANSITIONAL CLUSTER STUDY (A COMBINED SYSTEM IMPACT AND INTERCONNECTION FACILITIES STUDY)

[Assumptions to be completed by Interconnection Customer and Transmission Provider]

APPENDIX 5.1 to LGIP INFORMATIONAL INTERCONNECTION STUDY REQUEST

1.	The undersigned Interconnection Customer submits this request to evaluate the interconnection of its Generating Facility with Transmission Provider's Transmission System pursuant to the Tariff.										
2.	The type of interconnection service to be evaluated (check one):Energy Resource Interconnection ServiceNetwork Resource Interconnection Service										
3.	Interco	Interconnection Customer provides the following information:									
	a.	Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility;									
	b.	Maximum summer at degrees C and winter at degrees C megawatt electrical output of the proposed new Large Generating Facility or the amount of megawatt increase in the generating capacity of an existing Generating Facility;									
	C.	General description of the equipment configuration;									
	d.	Commercial Operation Date to be studied (Day, Month, and Year);									
	e.	Name, address, telephone number, and e-mail address of Interconnection Customer's contact person;									
	f.	Approximate location of the proposed Point of Interconnection;									
	g.	Interconnection Customer Data (set forth in Attachment A)									
	h.	Primary frequency response operating range for electric storage resources.									
	i.	Requested capacity (in MW) of Interconnection Service (if lower than the Generating Facility Capacity); and									
	j.	A Scope of Work including any additional information that may be reasonably required.									
4.	Study	deposit amount as specified in the LGIP.									
5.	For study purposes, the point of delivery to deliver within the Control Area.										
6.	This Informational Interconnection Study Request shall be submitted to the representative indicated below:										
	[To be	completed by Transmission Provider]									
7.	Repres	sentative of Interconnection Customer to contact:									
	[To be	[To be completed by Interconnection Customer]									

This Interconnection Request is submitted by: 8.

> Name of Interconnection Customer: ______ By (signature): _____ Name (type or print): _____

Attachment A to Appendix 5.1 Informational Interconnection Study Request

LARGE GENERATING FACILITY DATA

	UNII RATINGS	
kVA	°F	Voltage
Power Factor		
Speed (RPM)		
Short Circuit Ratio	Frequency, Hertz	
Stator Amperes at Rated kVA		Field Volts
Max Turbine MW	°F	
Primary frequency resp	oonse operating range for	electric storage resources.
Minimum State of Cha	arge:	
Maximum State of Ch		
COMBINED TUI	RBINE-GENERATOR-EXCI	TER INERTIA DATA
Inertia Constant, H =	kW sec/kVA	
Inertia Constant, $H = $ Moment-of-Inertia, $WR^2 = $	lb. ft. ²	
REACT	ANCE DATA (PER UNIT-R	ATED KVA)
DI	RECT AXIS QUADRATURE	EAXIS
Synchronous – saturated	X _{dv}	X _{qv}
Synchronous – unsaturated	X_{di}	X _{qi}
Transient – saturated	X dv	X' _{qv}
Transient – unsaturated	X' _{di}	X' _{qi}
Subtransient – saturated	X" _{dv}	X" _{qv}
Subtransient – unsaturated	X" _{di}	X" _{qi}
Negative Sequence – saturated	X2 _v	
Negative Sequence – unsaturat	ed X2 _i	
Zero Sequence – saturated	X0 _v	
Zero Sequence – unsaturated	X0 _i	
Leakage Reactance	XI _m	

FIELD TIME CONSTANT DATA (SEC)

Open Circuit		T' _{do}	T' _{qo}	
Three-Phase S	Short Circuit Transien	t T' _{d3}	T'q	
Line to Line Sh	nort Circuit Transient	T' _{d1}		
Short Circuit S	ubtransient	T" _d	T" _q	
Open Circuit S	ubtransient	T' _{d2}		
Line to Neutral	Short Circuit Transie	ent T" _{do}	T" _{qo}	
	ARMATUI	RE TIME CONSTA	NT DATA (SEC)	
Three Phase S Line to Line Sh Line to Neutral	nort Circuit T _{a2}	3 5		
NOTE: If reque	ested information is n	ot applicable, indic	ate by marking "N/A."	
		LITY AND PLANT GENERATING FA	CONFIGURATION ACILITY DATA	
	ARMATURE WI	NDING RESISTAN	ICE DATA (PER UNIT)	
Positive Negative Zero	R_2			
Field Current a Field Current a Three Phase A Field Winding	me Thermal Capacity at Rated kVA, Armatu at Rated kVA and Arn armature Winding Ca Resistance = ding Resistance (Per	re Voltage and PF nature Voltage, 0 F pacitance =° _ ohms°	PF =amps _microfarad C	
		CURVES		
			pacity Temperature Correction operating range for multiple cu	
	GENERATOR ST	EP-UP TRANSFO	RMER DATA RATINGS	
Capacity	Self-cooled/Maxi	mum Nameplate		
	/	kVA		

Voltage Ratio (Generator Side/System	n side/Tertiary)			
/	<i>_</i>	kV		
Winding Connections (Low V/High V/T	Tertiary V (Delta	or Wye))		
Fixed Taps Available				
Present Tap Setting				
If more than one transformer stage is the Transmission System, please p transformer type.				
	IMPEDANC	E		
Positive Z ₁ (on self-cooled kVA rating)	%		_X/R	
Zero Z ₀ (on self-cooled kVA rating)	%		_X/R	
EXCI	ITATION SYST	EM DATA		
Identify appropriate IEEE model block (PSS) for computer representation in excitation system and PSS constants	power system	stability simulati		
GOV	ERNOR SYST	EM DATA		
Identify appropriate IEEE model block power system stability simulations and the model.				
v	VIND GENERA	TORS		
Number of generators to be inter	rconnected pu	rsuant to this	Interconnection	Request:
Elevation: Sing	gle Phase	Three Phas	se	
Inverter manufacturer, model name, n				
List of adjustable setpoints for the prot				

Note: A completed General Electric Company Power Systems Load Flow (PSLF) data sheet or

other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate to the proposed device, then they shall be provided with the study request and discussed with the Transmission Provider in advance of the study.

INDUCTION GENERATORS

(*)	Field Volts:
	Field Amperes:
	Motoring Power (kW):
(*)	Neutral Grounding Resistor (If Applicable:
(*)	1 ₂ ² t or K (Heating Time Constant):
(*)	Rotor Resistance:
(*)	Stator Resistance:
(*)	Stator Reactance:
	Rotor Reactance:
(*)	Magnetizing Reactance:
(*)	Short Circuit Reactance:
(*)	Exciting Current:
(*)	Temperature Rise:
(*)	Frame Size:
	Design Letter:
	Reactive Power Required In Vars (No Load):
	Reactive Power Required In Vars (Full Load):
(*)	Total Rotating Inertia, H:Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Informational Interconnection Study Request to determine if the information designated by (*) is required.

6.0

APPENDIX 5.2 to LGIP INFORMATIONAL INTERCONNECTION STUDY AGREEMENT

THIS .	AGREEMENT is made and entered into thisday of,
20	by and between
organi	zed and existing under the laws of the State of
(IIIICH	connection customer,) and a
existin	g under the laws of the State of, ("Transmission Provider""). Interconnection
	mer and Transmission Provider each may be referred to as a "Party," or collectively as the
"Partie	S. The state of th
	RECITALS
	REAS , Interconnection Customer is evaluating developing a Large Generating Facility or ating capacity addition to an existing Generating Facility and
	REAS , Interconnection Customer is proposing to evaluate an interconnection with the mission System; and
	REAS, Interconnection Customer has submitted to Transmission Provider an Informational onnection Study Interconnection Request; and
	THEREFORE , in consideration of and subject to the mutual covenants contained herein rties agree as follows:
1.0	When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
2.0	Interconnection Customer elects and Transmission Provider shall cause an Informational Interconnection Study consistent with this LGIP to be performed in accordance with the Tariff.
3.0	The scope of the Informational Interconnection Study shall be subject to the assumptions set forth in Attachment A to this Agreement.
4.0	The Informational Interconnection Study shall be performed solely for informational purposes.
5.0	The Informational Interconnection Study report shall provide a sensitivity analysis based on the assumptions specified by Interconnection Customer in Attachment A to this Agreement. The Informational Interconnection Study shall identify Transmission Provider's Interconnection Facilities and the Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the assumptions specified by Interconnection Customer in Attachment A.

Interconnection Customer shall provide a deposit as identified in the LGIP for the

performance of the Informational Interconnection Study. Transmission Provider's good faith estimate for the time of completion of the Informational Interconnection Study is [insert date].

Upon receipt of the Informational Interconnection Study, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Informational Interconnection Study.

Any difference between the initial deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.

7.0 Miscellaneous. The Informational Interconnection Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

By:	_ By:		
Title:	_Title:		
Date:	_ Date:		
[Insert name of Interconnection Customer]			
Ву:	_		
Title:	_		
Date:			

Ilnsert name of Transmission Provider or Transmission Owner, if applicable

Appendix 6. STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT (LGIA)

> **Appendix 6 to the Standard Large Generator Interconnection Procedures**

STANDARD LARGE GENERATOR **INTERCONNECTION AGREEMENT (LGIA)**

(Applicable to Generating Facilities that exceed 20 MW)

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Note: This form of agreement is also used for a Small Generating Facility seeking Network Resource Interconnection Service. Small Generating Facilities are studied under the Large Generator Interconnection Procedures.

Appendix 6. STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT (LGIA)

TABLE OF CONTENTS

ъ			
к	ec1	ta.	S

recruis	
Article 1.	Definitions
Article 2.	Effective Date, Term, and Termination
2.1	Effective Date
2.2	\mathcal{C}
2.3	2.3.1 Written Notice
2.4	2.3.2 Default
2.4	
2.5	
2.6	Survival
Article 3.	Regulatory Filings
3.1	Filing
Article 4.	Scope of Service
4.1	1
	4.1.1 Energy Resource Interconnection Service
	4.1.1.1 The Product
	4.1.1.2 Transmission Delivery Service Implications
	4.1.2 Network Resource Interconnection Service
	4.1.2.1 The Product
	4.1.2.2 Transmission Delivery Service Implications
4.2	
4.3	
4.4	No Transmission Delivery Service
4.5	Interconnection Customer Provided Services
Article 5.	Interconnection Facilities Engineering, Procurement, and Construction
5.1	Options
	5.1.1 Standard Option
	5.1.2 Alternate Option
	5.1.3 Option to Build
	5.1.4 Negotiated Option
5.2	General Conditions Applicable to Option to Build

- 5.4 Power System Stabilizers
- 5.5 **Equipment Procurement**
- 5.6 **Construction Commencement**
- 5.7 **Work Progress**
- 5.8 Information Exchange
- 5.9 Other Interconnection Options
 - 5.9.1 Limited Operation
 - 5.9.2 Provisional Interconnection Service
- Interconnection Customer's Interconnection Facilities ('ICIF')
 - 5.10.1 Interconnection Customer's Interconnection Facility Specifications
 - 5.10.2 Transmission Provider's Review
 - 5.10.3 ICIF Construction
- 5.11 Transmission Provider's Interconnection Facilities Construction
- 5.12 Access Rights
- 5.13 Lands of Other Property Owners
- 5.14 Permits
- 5.15 Early Construction of Base Case Facilities
- 5.16 Suspension
 - 5.16.1 Effect of Missed Interconnection Customer Milestones
 - 5.16.2 Effect of Suspension; Parties' Obligations 5.17 Taxes
 - 5.17.1 Interconnection Customer Payments Not Taxable
 - 5.17.2 Representations and Covenants
 - 5.17.3 Indemnification for the Cost Consequences of Current Tax Liability Imposed Upon the Transmission Provider
 - 5.17.4 Tax Gross-Up Amount
 - 5.17.5 Private Letter Ruling or Change or Clarification of Law
 - 5.17.6 Subsequent Taxable Events
 - 5.17.7 Contests
 - 5.17.8 Refund
 - 5.17.9 Taxes Other Than Income Taxes
 - 5.17.10Transmission Owners Who Are Not Transmission Providers
- 5.18 Tax Status
- 5.19 Modification
 - 5.19.1 General.
 - 5.19.2 Standards
 - 5.19.3 Modification Costs
- Article 6. Testing and Inspection
 - Pre-Commercial Operation Date Testing and Modifications 6.1
 - Post-Commercial Operation Date Testing and Modifications 6.2
 - 6.3 Right to Observe Testing
 - Right to Inspect 6.4

Article 7.	Metering
7.1	General
7.2	Check Meters
7.3	Standards
7.4	Testing of Metering Equipment
7.5	
Article 8.	Communications
8.1	Interconnection Customer Obligations
8.2	Remote Terminal Unit
8.3	No Annexation
8.4	Provision of Data from a Variable Energy Resource
Article 9.	Operations
9.1	General
9.2	Control Area Notification
9.3	Transmission Provider Obligations
9.4	Interconnection Customer Obligations
9.5	Start-Up and Synchronization
9.6	Reactive Power and Primary Frequency Response
	9.6.1 Power Factor Design Criteria
	9.6.2 Voltage Schedules
	9.6.2.1 Voltage Regulators
	9.6.3 Payment for Reactive Power
	9.6.4 Primary Frequency Response
	9.6.4.1 Governor or Equivalent Controls
	9.6.4.2 Timely and Sustained Response
	9.6.4.3 Exemptions
	9.6.4.4 Electric Storage Resources
9.7	Outages and Interruptions
	9.7.1 Outages.
	9.7.1.1 Outage Authority and Coordination
	9.7.1.2 Outage Schedules
	9.7.1.3 Outage Restoration
	9.7.2 Interruption of Service
	9.7.3 Under-Frequency and Over Frequency Conditions
	9.7.4 System Protection and Other Control Requirements
	9.7.4.1 System Protection Facilities
	9.7.5 Requirements for Protection
	9.7.6 Power Quality
9.8	Switching and Tagging Rules
9.9	Use of Interconnection Facilities by Third Parties

	9.9.1 Purpose of Interconnection Facilities
	9.9.2 Third Party Users
9.10	Disturbance Analysis Data Exchange

- Article 10. Maintenance
 - 10.1 Transmission Provider Obligations
 - 10.2 Interconnection Customer Obligations
 - 10.3 Coordination
 - 10.4 Secondary Systems
 - 10.5 Operating and Maintenance Expenses
- Article 11. Performance Obligation
 - 11.1 Interconnection Customer Interconnection Facilities
 - Transmission Provider's Interconnection Facilities 11.2
 - 11.3 Network Upgrades and Distribution Upgrades
 - 11.4 Transmission Credits
 - 11.4.1 Repayment of Amounts Advanced for Network Upgrades
 - 11.4.2 Special Provisions for Affected Systems
 - 11.5 Provision of Security
 - 11.6 Interconnection Customer Compensation
 - 11.6.1 Interconnection Customer Compensation for Actions During Emergency Condition
- Article 12. Invoice
 - 12.1 General
 - 12.2 Final Invoice
 - 12.3 Payment
 - 12.4 Disputes
- Article 13. **Emergencies**
 - 13.1 Definition
 - 13.2 Obligations
 - 13.3 Notice
 - 13.4 Immediate Action
 - 13.5 Transmission Provider Authority
 - 13.5.1 General
 - 13.5.2 Reduction and Disconnection
 - 13.6 Interconnection Customer Authority
 - 13.7 Limited Liability
- Article 14. Regulatory Requirements and Governing Law
 - Regulatory Requirements 14.1
 - 14.2 Governing Law

15.2 15.3	Notices General Billings and Payments Alternative Forms of Notice Operations and Maintenance Notice
	Force Majeure Force Majeure
Article 17. 17.1	Default Default 17.1.1 General 17.1.2 Right to Terminate
	Indemnity, Consequential Damages and Insurance Indemnity 18.1.1 Indemnified Person 18.1.2 Indemnifying Party 18.1.3 Indemnity Procedures
	Consequential Damages Insurance
	Assignment Assignment
	Severability Severability
	Comparability Comparability
	Confidentiality Confidentiality 22.1.1 Term 22.1.2 Scope 22.1.3 Release of Confidential Information 22.1.4 Rights 22.1.5 No Warranties 22.1.6 Standard of Care 22.1.7 Order of Disclosure 22.1.8 Termination of Agreement

22.1.9 Remedies

22.1.10 Disclosure to FERC, its Staff, or a State

Article 23	Environmental Releases
Alticle 23.	Environmental Releases
Article 24.	Information Requirements
24.1	Information Acquisition
24.2	Information Submission by Transmission Provider
24.3	Updated Information Submission by Interconnection Customer
24.4	Information Supplementation
Article 25.	Information Access and Audit Rights
25.1	Information Access
25.2	Reporting of Non-Force Majeure Events
25.3	Audit Rights
25.4	Audit Rights Periods
23.4	25.4.1 Audit Rights Period for Construction-Related Accounts and Records
	25.4.2 Audit Rights Period for All Other Accounts and Records
25.5	Audit Results
23.3	Tradit Testins
Article 26.	Subcontractors
26.1	General
26.2	Responsibility of Principal
26.3	No Limitation by Insurance
Article 27.	Disputes
27.1	•
	External Arbitration Procedures
	Arbitration Decisions
27.4	
27,1	Costs
Article 28.	Representations, Warranties, and Covenants
28.1	General
	28.1.1 Good Standing
	28.1.2 Authority
	28.1.3 No Conflict
	28.1.4 Consent and Approval
Article 29.	Joint Operating Committee
29.1	Joint Operating Committee
Article 30.	Miscellaneous
30.1	Binding Effect
30.2	Conflicts

- 30.3 Rules of Interpretation
- 30.4 Entire Agreement
- 30.5 No Third Party Beneficiaries
- 30.6 Waiver
- 30.7 Headings
- 30.8 Multiple Counterparts
- 30.9 Amendment
- 30.10 Modification by the Parties
- 30.11 Reservation of Rights
- 30.12 No Partnership
- Appendix A Interconnection Facilities, Network Upgrades, and Distribution Upgrades
- Appendix B Milestones
- Appendix C Interconnection Details
- Appendix D Security Arrangements Details
- Appendix E Commercial Operation Date
- Appendix F Addresses for Delivery of Notices and Billings

Standard Large Generator Interconnection Agreement

STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT

THIS STANDADD I ADOE CENEDATOD INTEDCONNECTION

THIS STANDARD LANGE GENERATOR INTERCONNECTIVE	OI1
AGREEMENT ("Agreement") is made and entered into this day of _	
20, by and between, a	
organized and existing under the laws of the State/Commonwealth of	
("Interconnection Customer" with a Large Generating Facility), and	
, a	organized
and existing under the laws of the State/Commonwealth of	
("Transmission Provider and/or Transmission Owner"). Interconnection Co	ustomer and
Transmission Provider each may be referred to as a "Party" or collectively as	the "Parties."

Recitals

WHEREAS, Transmission Provider operates the Transmission System; and

WHEREAS, Interconnection Customer intends to own, lease and/or control and operate the Generating Facility identified as a Large Generating Facility in Appendix C to this Agreement; and,

WHEREAS, Interconnection Customer and Transmission Provider have agreed to enter into this Agreement for the purpose of interconnecting the Large Generating Facility with the Transmission System;

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein, it is agreed:

When used in this Standard Large Generator Interconnection Agreement, terms with initial capitalization that are not defined in Article 1 shall have the meanings specified in the Article in which they are used or the Open Access Transmission Tariff (Tariff).

Article 1. Definitions

Adverse System Impact shall mean the negative effects due to technical or operational limits on conductors or equipment being exceeded that may compromise the safety and reliability of the electric system.

Affected System shall mean an electric system other than the Transmission Provider's Transmission System that may be affected by the proposed interconnection.

Affected System Operator shall mean the entity that operates an Affected System.

Affiliate shall mean, with respect to a corporation, partnership or other entity, each such other corporation, partnership or other entity that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such corporation, partnership or other entity.

Ancillary Services shall mean those services that are necessary to support the transmission of capacity and energy from resources to loads while maintaining reliable operation of the Transmission Provider's Transmission System in accordance with Good Utility Practice.

Applicable Laws and Regulations shall mean all duly promulgated applicable federal, state and local laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders, permits and other duly authorized actions of any Governmental Authority.

Applicable Reliability Council shall mean the reliability council applicable to the Transmission System to which the Generating Facility is directly interconnected.

Applicable Reliability Standards shall mean the requirements and guidelines of NERC, the Applicable Reliability Council, and the Control Area of the Transmission System to which the Generating Facility is directly interconnected.

Base Case shall mean the base case power flow, short circuit, and stability data bases used for the Interconnection Studies by the Transmission Provider or Interconnection Customer.

Breach shall mean the failure of a Party to perform or observe any material term or condition of the Standard Large Generator Interconnection Agreement.

Breaching Party shall mean a Party that is in Breach of the Standard Large Generator Interconnection Agreement.

Business Day shall mean Monday through Friday, excluding Federal Holidays.

Calendar Day shall mean any day including Saturday, Sunday or a Federal Holiday.

Cluster shall mean a group of Interconnection Requests that are studied together for the purpose of conducting the Interconnection Studies. A Cluster is sometimes referred to as a Queue Cluster.

Clustering shall mean the process whereby a group of Interconnection Requests is

studied together, instead of serially, for the purpose of conducting the Interconnection Studies.

Commercial Operation shall mean the status of a Generating Facility that has commenced generating electricity for sale, excluding electricity generated during Trial Operation.

Commercial Operation Date of a unit shall mean the date on which the Generating Facility commences Commercial Operation as agreed to by the Parties pursuant to Appendix E to the Standard Large Generator Interconnection Agreement.

Confidential Information shall mean any confidential, proprietary or trade secret information of a plan, specification, pattern, procedure, design, device, list, concept, policy or compilation relating to the present or planned business of a Party, which is designated as confidential by the Party supplying the information, whether conveyed orally, electronically, in writing, through inspection, or otherwise.

Control Area shall mean an electrical system or systems bounded by interconnection metering and telemetry, capable of controlling generation to maintain its interchange schedule with other Control Areas and contributing to frequency regulation of the interconnection. A Control Area must be certified by the Applicable NERC Regional Reliability Entity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC.

Default shall mean the failure of a Breaching Party to cure its Breach in accordance with Article 17 of the Standard Large Generator Interconnection Agreement.

Dispute Resolution shall mean the procedure for resolution of a dispute between the Parties in which they will first attempt to resolve the dispute on an informal basis.

Definitive Interconnection Study Process ("Definitive Interconnection Study") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study Process.

Definitive Interconnection System Impact Study ("DISIS") shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System.

Definitive Interconnection System Impact Study Agreement ("DISIS Agreement") shall mean the form of agreement contained in the Appendix of the LGIP for conducting the Definitive Interconnection System Impact Study.

DISIS Request Window shall have the meaning set forth in Section 4.2 of the LGIP.

Distribution System shall mean the Transmission Provider's facilities and equipment used to transmit electricity to ultimate usage points such as homes and industries directly from nearby generators or from interchanges with higher voltage transmission networks which transport bulk power over longer distances. The voltage levels at which distribution systems operate differ among areas.

Distribution Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Distribution System at or beyond the Point of Interconnection to facilitate interconnection of the Generating Facility and render the transmission service necessary to effect Interconnection Customer's wholesale sale of electricity in interstate commerce. Distribution Upgrades do not include Interconnection Facilities.

Effective Date shall mean the date on which the Standard Large Generator Interconnection Agreement becomes effective upon execution by the Parties subject to acceptance by FERC, or if filed unexecuted, upon the date specified by FERC.

Emergency Condition shall mean a condition or situation: (1) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (2) that, in the case of a Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to Transmission Provider's Transmission System, Transmission Provider's Interconnection Facilities or the electric systems of others to which the Transmission Provider's Transmission System is directly connected; or (3) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Generating Facility or Interconnection Customer's Interconnection Facilities. System restoration and black start shall be considered Emergency Conditions; provided, that Interconnection Customer is not obligated by the Standard Large Generator Interconnection Agreement to possess black start capability.

Energy Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to connect its Generating Facility to the Transmission Provider's Transmission System to be eligible to deliver the Generating Facility's electric output using the existing firm or nonfirm capacity of the Transmission Provider's Transmission System on an as available basis. Energy Resource Interconnection Service in and of itself does not convey transmission service.

Engineering & Procurement (E&P) Agreement shall mean an agreement that authorizes the Transmission Provider to begin engineering and procurement of long

lead-time items necessary for the establishment of the interconnection in order to advance the implementation of the Interconnection Request.

Environmental Law shall mean Applicable Laws or Regulations relating to pollution or protection of the environment or natural resources.

Federal Power Act shall mean the Federal Power Act, as amended, 16 U.S.C. §§ 791a et seq.

FERC shall mean the Federal Energy Regulatory Commission (Commission) or its successor.

Force Majeure shall mean any act of God, labor disturbance, act of the public enemy, war, insurrection, riot, fire, storm or flood, explosion, breakage or accident to machinery or equipment, any order, regulation or restriction imposed by governmental, military or lawfully established civilian authorities, or any other cause beyond a Party's control. A Force Majeure event does not include acts of negligence or intentional wrongdoing by the Party claiming Force Majeure.

Generating Facility shall mean Interconnection Customer's device for the production and/or storage for later injection of electricity identified in the Interconnection Request, but shall not include the Interconnection Customer's Interconnection Facilities.

Generating Facility Capacity shall mean the net capacity of the Generating Facility and the aggregate net capacity of the Generating Facility where it includes multiple energy production devices.

Good Utility Practice shall mean any of the practices, methods and acts engaged in or approved by a significant portion of the electric industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act to the exclusion of all others, but rather to be acceptable practices, methods, or acts generally accepted in the region.

Governmental Authority shall mean any federal, state, local or other governmental regulatory or administrative agency, court, commission, department, board, or other governmental subdivision, legislature, rulemaking board, tribunal, or other governmental authority having jurisdiction over the Parties, their respective facilities, or the respective services they provide, and exercising or entitled to exercise any administrative, executive, police, or taxing authority or power; provided, however, that such term does not include Interconnection Customer, Transmission Provider, or any

Affiliate thereof.

Hazardous Substances shall mean any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "radioactive substances," "contaminants," "pollutants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, or any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

Initial Synchronization Date shall mean the date upon which the Generating Facility is initially synchronized and upon which Trial Operation begins.

In-Service Date shall mean the date upon which the Interconnection Customer reasonably expects it will be ready to begin use of the Transmission Provider's Interconnection Facilities to obtain back feed power.

Interconnection Customer shall mean any entity, including the Transmission Provider, Transmission Owner or any of the Affiliates or subsidiaries of either, that proposes to interconnect its Generating Facility with the Transmission Provider's Transmission System. Where the Interconnection Customer and the Transmission Provider are the same entity, application of the following provisions set forth herein is not required: provisions governing the posting of security (in Articles 5 and 11), provisions governing taxes and reimbursements for tax liability (in Article 5), provisions governing indemnity, consequential damages and insurance (in Article 18), and provisions governing billing and payment (in Articles 12 and 15) (invoices are not required, but may be generated and tendered for administrative use to aid in the identification and accounting of costs).

Interconnection Customer's Interconnection Facilities shall mean all facilities and equipment, as identified in Appendix A of the Standard Large Generator Interconnection Agreement, that are located between the Generating Facility and the Point of Change of Ownership, including any modification, addition, or upgrades to such facilities and equipment necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System.

Interconnection Customer's Interconnection Facilities are sole use facilities.

Interconnection Facilities shall mean the Transmission Provider's Interconnection Facilities and the Interconnection Customer's Interconnection Facilities. Collectively, Interconnection Facilities include all facilities and equipment between the Generating Facility and the Point of Interconnection, including any modification, additions or upgrades that are necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network

Upgrades or Network Upgrades.

Interconnection Facilities Study shall mean a study conducted by the Transmission Provider or a third party consultant for the Interconnection Customer to determine a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades as identified in the Definitive Interconnection System Impact Study), the cost of those facilities, and the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System. The scope of the study is defined in Section 8 of the Standard Large Generator Interconnection Procedures.

Interconnection Facilities Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Facilities Study.

Interconnection Request shall mean an Interconnection Customer's request, in the form of Appendix 1 to the Standard Large Generator Interconnection Procedures, in accordance with the Tariff, to interconnect a new Generating Facility, or to increase the capacity of, or make a Material Modification to the operating characteristics of, an existing Generating Facility that is interconnected with the Transmission Provider's Transmission System.

Interconnection Service shall mean the service provided by the Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to the Transmission Provider's Transmission System and enabling it to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Standard Large Generator Interconnection Agreement and, if applicable, the Transmission Provider's Tariff.

Interconnection Study shall mean any of the following studies: the Informational Interconnection Study, the Definitive Interconnection System Impact Study, and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection Study Agreement shall mean any of the following agreements: the Informational Interconnection Study Agreement, the Definitive Interconnection System Impact Study Agreement, or the Interconnection Facilities Study Agreement described in the Standard Large Generator Interconnection Procedures.

IRS shall mean the Internal Revenue Service.

Joint Operating Committee shall be a group made up of representatives from Interconnection Customers and the Transmission Provider to coordinate operating and technical considerations of Interconnection Service.

Large Generating Facility shall mean a Generating Facility having a Generating Facility Capacity of more than 20 MW.

Loss shall mean any and all losses relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's performance, or non performance of its obligations under the Standard Large Generator Interconnection Agreement on behalf of the indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the indemnifying Party.

Material Modification shall mean those modifications that have a material impact on the cost or timing of any Interconnection Request with a later or equal queue priority date.

Metering Equipment shall mean all metering equipment installed or to be installed at the Generating Facility pursuant to the Standard Large Generator Interconnection Agreement at the metering points, including but not limited to instrument transformers, MWh-meters, data acquisition equipment, transducers, remote terminal unit, communications equipment, phone lines, and fiber optics.

NERC shall mean the North American Electric Reliability Council or its successor organization.

Network Resource shall mean any designated generating resource owned, purchased, or leased by a Network Customer under the Network Integration Transmission Service Tariff. Network Resources do not include any resource, or any portion thereof, that is committed for sale to third parties or otherwise cannot be called upon to meet the Network Customer's Network Load on a non-interruptible basis.

Network Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to integrate its Large Generating Facility with the Transmission Provider's Transmission System (1) in a manner comparable to that in which the Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an RTO or ISO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service in and of itself does not convey transmission service.

Network Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Transmission System required at or beyond the point at which the Interconnection Facilities connect to the Transmission Provider's Transmission System to accommodate the interconnection of the Large Generating Facility to the Transmission Provider's Transmission System.

Notice of Dispute shall mean a written notice of a dispute or claim that arises out of or in connection with the Standard Large Generator Interconnection Agreement or its performance.

OASIS shall mean the Transmission Provider's Open Access Same-Time Informational System.

Party or Parties shall mean Transmission Provider, Transmission Owner, Interconnection Customer or any combination of the above.

Phase ("Phase 1, Phase 2, Phase 3 or "Phase 4") shall mean a distinct part of the Definitive Study Process as described in Sections 7 and 8 of the LGIP.

Point of Change of Ownership shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Customer's Interconnection Facilities connect to the Transmission Provider's Interconnection Facilities.

Point of Interconnection shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Facilities connect to the Transmission Provider's Transmission System.

Provisional Interconnection Service shall mean Interconnection Service provided by Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to Transmission Provider's Transmission System and enabling that Transmission System to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Provisional Large Generator Interconnection Agreement and, if applicable, the Tariff.

Provisional Large Generator Interconnection Agreement shall mean the interconnection agreement for Provisional Interconnection Service established between Transmission Provider and/or the Transmission Owner and the Interconnection Customer. This agreement shall take the form of the Large Generator Interconnection Agreement, modified for provisional purposes. Provisional Large Generator Interconnection Agreements are not eligible for suspension.

Queue shall mean a queue for valid Interconnection Requests for the Definitive Interconnection Study Process.

Queue Position shall mean the order of a valid Interconnection Request, relative to all other pending valid Interconnection Requests, in the Definitive Interconnection Study Process. The Queue Position is established based upon the date and time

Interconnection Customer satisfies all of the requirements of the LGIP to enter the Definitive Interconnection Study Process. Valid Interconnection Requests in a single Cluster are considered equal-queued.

Readiness Milestone(s) shall have the meaning set forth in the LGIP.

Reasonable Efforts shall mean, with respect to an action required to be attempted or taken by a Party under the Standard Large Generator Interconnection Agreement, efforts that are timely and consistent with Good Utility Practice and are otherwise substantially equivalent to those a Party would use to protect its own interests.

Resource Plan shall mean any process authorized or required by Applicable Laws and Regulations for, *inter alia*, the selection of Generating Facilities interconnected to the Transmission System of Transmission Provider.

Resource Planning Entity shall mean any entity subject to or conducting a Resource Solicitation Process.

Resource Solicitation Cluster shall mean a Cluster Study associated with a Resource Planning Process.

Resource Solicitation Process shall mean any process authorized or required by Applicable Laws and Regulations for the acquisition of Network Resources by an entity interconnected to the Transmission System of Transmission Provider.

Scoping Meeting shall mean the meeting between representatives of the Interconnection Customer and Transmission Provider conducted for the purpose of discussing the proposed Interconnection Request, alternative interconnection options, to exchange information including any transmission data and earlier study evaluations that would be reasonably expected to affect such interconnection options, to analyze such information, and to determine the potential feasible Points of Interconnection.

Site Control shall mean the exclusive land right to develop, construct, operate, and maintain the Generating Facility over the term of expected operation of the Generating Facility. Site Control shall include the right to develop, construct, operate, and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by documentation reasonably demonstrating: (1) ownership of, a leasehold interest in, or a right to develop a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Customer's Interconnection Facilities; (2) an option to purchase or acquire a leasehold interest in a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities; or (3) any other documentation that clearly demonstrates the right of the Interconnection Customer to exclusively or occupy a site of sufficient size to construct and operate the

Generating Facility. Site Control for any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all co-located projects that meet the aforementioned provisions of this Site Control definition.

Small Generating Facility shall mean a Generating Facility that has a Generating Facility Capacity of no more than 20 MW.

Stand Alone Network Upgrades shall mean Network Upgrades that are not part of an Affected System that an Interconnection Customer may construct without affecting day-to-day operations of the Transmission System during their construction. Both the Transmission Provider and the Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify them in Appendix A to the Standard Large Generator Interconnection Agreement. If the Transmission Provider and Interconnection Customer disagree about whether a particular Network Upgrade is a Stand Alone Network Upgrade, the Transmission Provider must provide the Interconnection Customer a written technical explanation outlining why the Transmission Provider does not consider the Network Upgrade to be a Stand Alone Network Upgrade within 15 days of its determination.

Standard Large Generator Interconnection Agreement (LGIA) shall mean the form of interconnection agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Procedures (LGIP) shall mean the interconnection procedures applicable to an Interconnection Request pertaining to a Large Generating Facility that are included in the Transmission Provider's Tariff.

Surplus Interconnection Service shall mean any unneeded portion of Interconnection Service established in a Large Generator Interconnection Agreement, such that if Surplus Interconnection Service is utilized the total amount of Interconnection Service at the Point of Interconnection would remain the same.

System Protection Facilities shall mean the equipment, including necessary protection signal communications equipment, required to protect (1) the Transmission Provider's Transmission System from faults or other electrical disturbances occurring at the Generating Facility and (2) the Generating Facility from faults or other electrical system disturbances occurring on the Transmission Provider's Transmission System or on other delivery systems or other generating systems to which the Transmission Provider's Transmission System is directly connected.

Tariff shall mean the Transmission Provider's Tariff through which open access transmission service and Interconnection Service are offered, as filed with FERC, and as amended or supplemented from time to time, or any successor tariff.

Transmission Owner shall mean an entity that owns, leases or otherwise possesses an interest in the portion of the Transmission System at the Point of Interconnection and may be a Party to the Standard Large Generator Interconnection Agreement to the extent necessary.

Transmission Provider shall mean the public utility (or its designated agent) that owns, controls, or operates transmission or distribution facilities used for the transmission of electricity in interstate commerce and provides transmission service under the Tariff. The term Transmission Provider should be read to include the Transmission Owner when the Transmission Owner is separate from the Transmission Provider.

Transmission Provider's Interconnection Facilities shall mean all facilities and equipment owned, controlled or operated by the Transmission Provider from the Point of Change of Ownership to the Point of Interconnection as identified in Appendix A to the Standard Large Generator Interconnection Agreement, including any modifications, additions or upgrades to such facilities and equipment. Transmission Provider's Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Transmission System shall mean the facilities owned, controlled or operated by the Transmission Provider or Transmission Owner that are used to provide transmission service under the Tariff.

Trial Operation shall mean the period during which Interconnection Customer is engaged in on-site test operations and commissioning of the Generating Facility prior to Commercial Operation.

Variable Energy Resource shall mean a device for the production of electricity that is characterized by an energy source that: (1) is renewable; (2) cannot be stored by the facility owner or operator; and (3) has variability that is beyond the control of the facility owner or operator.

Withdrawal Penalty shall have the meaning set forth in Section 3.7 of the LGIP. Article 2. Effective Date, Term, and Termination

- **2.1 Effective Date**. This LGIA shall become effective upon execution by the Parties subject to acceptance by FERC (if applicable), or if filed unexecuted, upon the date specified by FERC. Transmission Provider shall promptly file this LGIA with FERC upon execution in accordance with Article 3.1, if required.
- **2.2 Term of Agreement**. Subject to the provisions of Article 2.3, this LGIA shall remain in effect for a period of ten (10) years from the Effective Date or such other

longer period as Interconnection Customer may request (Term to be specified in individual agreements) and shall be automatically renewed for each successive one-year period thereafter.

2.3 Termination Procedures.

- **2.3.1 Written Notice**. This LGIA may be terminated by Interconnection Customer after giving Transmission Provider ninety (90) Calendar Days advance written notice, or by Transmission Provider notifying FERC after the Generating Facility permanently ceases Commercial Operation. This LGIA shall be terminated by Transmission Provider if the Generating Facility or a portion of the Generating Facility fails to achieve Commercial Operation by the Commercial Operation Date established in accordance with Section 4.4.5 of the LGIP, including any extension provided thereunder.
- **2.3.2 Default**. Either Party may terminate this LGIA in accordance with Article 17.
- **2.3.3.** Notwithstanding Articles 2.3.1 and 2.3.2, no termination shall become effective until the Parties have complied with all Applicable Laws and Regulations applicable to such termination, including the filing with FERC of a notice of termination of this LGIA, which notice has been accepted for filing by FERC.
- Article 2.3 above, each Party shall pay all costs incurred (including any cancellation costs relating to orders or contracts for Interconnection Facilities and equipment) or charges assessed by the other Party, as of the date of the other Party's receipt of such notice of termination, that are the responsibility of the Terminating Party under this LGIA. In the event of termination by a Party, the Parties shall use commercially Reasonable Efforts to mitigate the costs, damages and charges arising as a consequence of termination. Upon termination of this LGIA, unless otherwise ordered or approved by FERC:
 - 2.4.1 With respect to any portion of Transmission Provider's Interconnection Facilities that have not yet been constructed or installed, Transmission Provider shall to the extent possible and with Interconnection Customer's authorization cancel any pending orders of, or return, any materials or equipment for, or contracts for construction of, such facilities; provided that in the event Interconnection Customer elects not to authorize such cancellation, Interconnection Customer shall assume all payment obligations with respect to such materials, equipment, and contracts, and Transmission Provider shall deliver such material and equipment, and, if necessary, assign

such contracts, to Interconnection Customer as soon as practicable, at Interconnection Customer's expense. To the extent that Interconnection Customer has already paid Transmission Provider for any or all such costs of materials or equipment not taken by Interconnection Customer, Transmission Provider shall promptly refund such amounts to Interconnection Customer, less any costs, including penalties incurred by Transmission Provider to cancel any pending orders of or return such materials, equipment, or contracts.

If an Interconnection Customer terminates this LGIA, it shall be responsible for all costs incurred in association with that Interconnection Customer's interconnection, including any cancellation costs relating to orders or contracts for Interconnection Facilities and equipment, and other expenses including any Network Upgrades for which Transmission Provider has incurred expenses and has not been reimbursed by Interconnection Customer.

- **2.4.2** Transmission Provider may, at its option, retain any portion of such materials, equipment, or facilities that Interconnection Customer chooses not to accept delivery of, in which case Transmission Provider shall be responsible for all costs associated with procuring such materials, equipment, or facilities.
- **2.4.3** With respect to any portion of the Interconnection Facilities, and any other facilities already installed or constructed pursuant to the terms of this LGIA, Interconnection Customer shall be responsible for all costs associated with the removal, relocation or other disposition or retirement of such materials, equipment, or facilities.
- 2.4.4 Transmission Provider shall refund the security provided under Section 10.3 of the LGIP including any accumulated interest, if applicable. Notwithstanding the foregoing, prior to remitting such security, plus accumulated interest if applicable, Transmission Provider shall offset against such security, and accumulated interest, any unpaid costs or penalties arising out of this Agreement or the LGIP. Monies due the Interconnection Customer shall be remitted within 90 Calendar Days of termination.
- **2.4.5** In the event a Withdrawal Penalty applies to a termination under this section, the Withdrawal Penalty shall have the meaning of and be calculated pursuant to Section 3.7.1 of the LGIP.
- **2.5 Disconnection**. Upon termination of this LGIA, the Parties will take all appropriate steps to disconnect the Large Generating Facility from the Transmission System.

All costs required to effectuate such disconnection shall be borne by the terminating Party, unless such termination resulted from the nonterminating Party's Default of this LGIA or such non-terminating Party otherwise is responsible for these costs under this LGIA.

2.6 Survival. This LGIA shall continue in effect after termination to the extent necessary to provide for final billings and payments and for costs incurred hereunder, including billings and payments pursuant to this LGIA; to permit the determination and enforcement of liability and indemnification obligations arising from acts or events that occurred while this LGIA was in effect; and to permit each Party to have access to the lands of the other Party pursuant to this LGIA or other applicable agreements, to disconnect, remove or salvage its own facilities and equipment.

Article 3. Regulatory Filings

3.1 Filing. Transmission Provider shall file this LGIA (and any amendment hereto) with the appropriate Governmental Authority, if required. Interconnection Customer may request that any information so provided be subject to the confidentiality provisions of Article 22. If Interconnection Customer has executed this LGIA, or any amendment thereto, Interconnection Customer shall reasonably cooperate with Transmission Provider with respect to such filing and to provide any information reasonably requested by Transmission Provider needed to comply with applicable regulatory requirements.

Article 4. Scope of Service

4.1 Interconnection Product Options. Interconnection Customer has selected the following (checked) type of Interconnection Service:

4.1.1 Energy Resource Interconnection Service

- 4.1.1.1 The Product. Energy Resource Interconnection Service allows Interconnection Customer to connect the Large Generating Facility to the Transmission System and be eligible to deliver the Large Generating Facility's output using the existing firm or non-firm capacity of the Transmission System on an "as available" basis. To the extent Interconnection Customer wants to receive Energy Resource Interconnection Service, Transmission Provider shall construct facilities identified in Attachment A.
- **4.1.1.2 Transmission Delivery Service Implications**. Under Energy

Resource Interconnection Service, Interconnection Customer will be eligible to inject power from the Large Generating Facility into and deliver power across the interconnecting Transmission Provider's Transmission System on an "as available" basis up to the amount of MWs identified in the applicable stability and steady state studies to the extent the upgrades initially required to qualify for Energy Resource Interconnection Service have been constructed. Where eligible to do so (e.g., PJM, ISONE, NYISO), Interconnection Customer may place a bid to sell into the market up to the maximum identified Large Generating Facility output, subject to any conditions specified in the interconnection service approval, and the Large Generating Facility will be dispatched to the extent Interconnection Customer's bid clears. In all other instances, no transmission delivery service from the Large Generating Facility is assured, but Interconnection Customer may obtain Point-to-Point Transmission Service, Network Integration Transmission Service, or be used for secondary network transmission service, pursuant to Transmission Provider's Tariff, up to the maximum output identified in the stability and steady state studies. In those instances, in order for Interconnection Customer to obtain the right to deliver or inject energy beyond the Large Generating Facility Point of Interconnection or to improve its ability to do so, transmission delivery service must be obtained pursuant to the provisions of Transmission Provider's Tariff. The Interconnection Customer's ability to inject its Large Generating Facility output beyond the Point of Interconnection, therefore, will depend on the existing capacity of Transmission Provider's Transmission System at such time as a transmission service request is made that would accommodate such delivery. The provision of firm Point-to-Point Transmission Service or Network Integration Transmission Service may require the construction of additional Network Upgrades.

4.1.2 Network Resource Interconnection Service.

4.1.2.1 The Product. Transmission Provider must conduct the necessary studies and construct the Network Upgrades needed to integrate the Large Generating Facility (1) in a manner comparable to that in which Transmission Provider integrates its generating facilities to serve native load customers; or (2) in

an ISO or RTO with market based congestion management, in the same manner as all Network Resources. To the extent Interconnection Customer wants to receive Network Resource Interconnection Service, Transmission Provider shall construct the facilities identified in Attachment A to this LGIA.

4.1.2.2 Transmission Delivery Service Implications. Network

Resource Interconnection Service allows Interconnection Customer's Large Generating Facility to be designated by any Network Customer under the Tariff on Transmission Provider's Transmission System as a Network Resource, up to the Large Generating Facility's full output, on the same basis as existing Network Resources interconnected to Transmission Provider's Transmission System, and to be studied as a Network Resource on the assumption that such a designation will occur. Although Network Resource Interconnection Service does not convey a reservation of transmission service, any Network Customer under the Tariff can utilize its network service under the Tariff to obtain delivery of energy from the interconnected Interconnection Customer's Large Generating Facility in the same manner as it accesses Network Resources. A Large Generating Facility receiving Network Resource Interconnection Service may also be used to provide Ancillary Services after technical studies and/or periodic analyses are performed with respect to the Large Generating Facility's ability to provide any applicable Ancillary Services, provided that such studies and analyses have been or would be required in connection with the provision of such Ancillary Services by any existing Network Resource. However, if an Interconnection Customer's Large Generating Facility has not been designated as a Network Resource by any load, it cannot be required to provide Ancillary Services except to the extent such requirements extend to all generating facilities that are similarly situated. The provision of Network Integration Transmission Service or firm Point-to-Point Transmission Service may require additional studies and the construction of additional upgrades. Because such studies and upgrades would be associated with a request for delivery service under the Tariff, cost responsibility for the studies and upgrades would be in accordance with FERC's policy for pricing transmission delivery services. Network Resource Interconnection Service does not necessarily provide Interconnection Customer with the capability to physically deliver the output of its Large

Generating Facility to any particular load on Transmission Provider's Transmission System without incurring congestion costs. In the event of transmission constraints on Transmission Provider's Transmission System, Interconnection Customer's Large Generating Facility shall be subject to the applicable congestion management procedures in Transmission Provider's Transmission System in the same manner as Network Resources.

There is no requirement either at the time of study or interconnection, or at any point in the future, that Interconnection Customer's Large Generating Facility be designated as a Network Resource by a Network Service Customer under the Tariff or that Interconnection Customer identify a specific buyer (or sink). To the extent a Network Customer does designate the Large Generating Facility as a Network Resource, it must do so pursuant to Transmission Provider's Tariff.

Once an Interconnection Customer satisfies the requirements for obtaining Network Resource Interconnection Service, any future transmission service request for delivery from the Large Generating Facility within Transmission Provider's Transmission System of any amount of capacity and/or energy, up to the amount initially studied, will not require that any additional studies be performed or that any further upgrades associated with such Large Generating Facility be undertaken, regardless of whether or not such Large Generating Facility is ever designated by a Network Customer as a Network Resource and regardless of changes in ownership of the Large Generating Facility. However, the reduction or elimination of congestion or redispatch costs may require additional studies and the construction of additional upgrades.

To the extent Interconnection Customer enters into an arrangement for long term transmission service for deliveries from the Large Generating Facility outside Transmission Provider's Transmission System, such request may require additional studies and upgrades in order for Transmission Provider to grant such request.

4.2 Provision of Service. Transmission Provider shall provide Interconnection Service for the Large Generating Facility at the Point of Interconnection.

- 4.3 Performance Standards. Each Party shall perform all of its obligations under this LGIA in accordance with Applicable Laws and Regulations, Applicable Reliability Standards, and Good Utility Practice, and to the extent a Party is required or prevented or limited in taking any action by such regulations and standards, such Party shall not be deemed to be in Breach of this LGIA for its compliance therewith. If such Party is a Transmission Provider or Transmission Owner, then that Party shall amend the LGIA and submit the amendment to FERC for approval.
- **4.4 No Transmission Delivery Service**. The execution of this LGIA does not constitute a request for, nor the provision of, any transmission delivery service under Transmission Provider's Tariff, and does not convey any right to deliver electricity to any specific customer or Point of Delivery.
- **4.5 Interconnection Customer Provided Services**. The services provided by Interconnection Customer under this LGIA are set forth in Article 9.6 and Article 13.5.1. Interconnection Customer shall be paid for such services in accordance with Article 11.6.

Article 5. Interconnection Facilities Engineering, Procurement, and Construction

- Customer shall select the In-Service Date, Initial Synchronization Date, and Commercial Operation Date; and either Standard Option or Alternate Option set forth below for completion of Transmission Provider's Interconnection Facilities and Network Upgrades as set forth in Appendix A, and such dates and selected option shall be set forth in Appendix B, Milestones. At the same time, Interconnection Customer shall indicate whether it elects to exercise the Option to Build set forth in Article 5.1.3 below. If the dates designated by Interconnection Customer are not acceptable to Transmission Provider, Transmission Provider shall so notify Interconnection Customer within thirty (30) Calendar Days. Upon receipt of the notification that Interconnection Customer's designated dates are not acceptable to Transmission Provider, the Interconnection Customer shall notify Transmission Provider within thirty (30) Calendar Days whether it elects to exercise the Option to Build if it has not already elected to exercise the Option to Build.
 - 5.1.1 Standard Option. Transmission Provider shall design, procure, and construct Transmission Provider's Interconnection Facilities and Network Upgrades, using Reasonable Efforts to complete Transmission Provider's Interconnection Facilities and Network Upgrades by the dates set forth in Appendix B, Milestones. Transmission Provider shall not be required to undertake any action which is inconsistent with its standard safety practices, its material and equipment specifications, its design criteria and construction

procedures, its labor agreements, and Applicable Laws and Regulations. In the event Transmission Provider reasonably expects that it will not be able to complete Transmission Provider's Interconnection Facilities and Network Upgrades by the specified dates, Transmission Provider shall promptly provide written notice to Interconnection Customer and shall undertake Reasonable Efforts to meet the earliest dates thereafter.

5.1.2 Alternate Option. If the dates designated by Interconnection Customer are acceptable to Transmission Provider, Transmission Provider shall so notify Interconnection Customer within thirty (30) Calendar Days, and shall assume responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities by the designated dates.

If Transmission Provider subsequently fails to complete Transmission Provider's Interconnection Facilities by the In-Service Date, to the extent necessary to provide back feed power; or fails to complete Network Upgrades by the Initial Synchronization Date to the extent necessary to allow for Trial Operation at full power output, unless other arrangements are made by the Parties for such Trial Operation; or fails to complete the Network Upgrades by the Commercial Operation Date, as such dates are reflected in Appendix B, Milestones; Transmission Provider shall pay Interconnection Customer liquidated damages in accordance with Article 5.3, Liquidated Damages, provided, however, the dates designated by Interconnection Customer shall be extended day for day for each day that the applicable RTO or ISO refuses to grant clearances to install equipment.

- 5.1.3 Option to Build. Interconnection Customer shall have the option to assume responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades on the dates specified in Article 5.1.2. Transmission Provider and Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify such Stand Alone Network Upgrades in Appendix A. Except for Stand Alone Network Upgrades, Interconnection Customer shall have no right to construct Network Upgrades under this option.
- 5.1.4 Negotiated Option. If the dates designated by Interconnection Customer are not acceptable to Transmission Provider, the Parties shall in good faith attempt to negotiate terms and conditions (including revision of the specified dates and liquidated damages, the provision of incentives, or the procurement and construction of all facilities other than Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades if the Interconnection Customer elects to exercise the Option to Build under

- Article 5.1.3). If the Parties are unable to reach agreement on such terms and conditions, then, pursuant to article 5.1.1 (Standard Option), Transmission Provider shall assume responsibility for the design, procurement and construction of all facilities other than Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades if the Interconnection Customer elects to exercise the Option to Build.
- **5.2 General Conditions Applicable to Option to Build**. If Interconnection Customer assumes responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades,
 - (1) Interconnection Customer shall engineer, procure equipment, and construct Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades (or portions thereof) using Good Utility Practice and using standards and specifications provided in advance by Transmission Provider;
 - (2) Interconnection Customer's engineering, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades shall comply with all requirements of law to which Transmission Provider would be subject in the engineering, procurement or construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
 - (3) Transmission Provider shall review and approve the engineering design, equipment acceptance tests, and the construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
 - (4) prior to commencement of construction, Interconnection Customer shall provide to Transmission Provider a schedule for construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades, and shall promptly respond to requests for information from Transmission Provider;
 - (5) at any time during construction, Transmission Provider shall have the right to gain unrestricted access to Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades and to conduct inspections of the same;
 - (6) at any time during construction, should any phase of the engineering, equipment procurement, or construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades not meet the

standards and specifications provided by Transmission Provider, Interconnection Customer shall be obligated to remedy deficiencies in that portion of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;

- (7) Interconnection Customer shall indemnify Transmission Provider for claims arising from Interconnection Customer's construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades under the terms and procedures applicable to Article 18.1 Indemnity;
- (8) Interconnection Customer shall transfer control of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades to Transmission Provider;
- (9) Unless Parties otherwise agree, Interconnection Customer shall transfer ownership of Transmission Provider's Interconnection Facilities and Stand-Alone Network Upgrades to Transmission Provider;
- (10) Transmission Provider shall approve and accept for operation and maintenance Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades to the extent engineered, procured, and constructed in accordance with this Article 5.2; and
- (11) Interconnection Customer shall deliver to Transmission Provider "asbuilt" drawings, information, and any other documents that are reasonably required by Transmission Provider to assure that the Interconnection Facilities and Stand Alone Network Upgrades are built to the standards and specifications required by Transmission Provider.
- (12) If Interconnection Customer exercises the Option to Build pursuant to Article 5.1.3, Interconnection Customer shall pay Transmission Provider an agreed upon amount of [\$ PLACEHOLDER] for Transmission Provider to execute the responsibilities enumerated to Transmission Provider under Article 5.2. Transmission Provider shall invoice Interconnection Customer for this total amount to be divided on a monthly basis pursuant to Article 12.
- **5.3 Liquidated Damages**. The actual damages to Interconnection Customer, in the event Transmission Provider's Interconnection Facilities or Network Upgrades are not completed by the dates designated by Interconnection Customer and accepted by Transmission Provider pursuant to subparagraphs 5.1.2 or 5.1.4, above, may include Interconnection Customer's fixed operation and maintenance costs and lost opportunity costs. Such actual damages are uncertain and impossible to determine at

this time. Because of such uncertainty, any liquidated damages paid by Transmission Provider to Interconnection Customer in the event that Transmission Provider does not complete any portion of Transmission Provider's Interconnection Facilities or Network Upgrades by the applicable dates, shall be an amount equal to ½ of 1 percent per day of the actual cost of Transmission Provider's Interconnection Facilities and Network Upgrades, in the aggregate, for which Transmission Provider has assumed responsibility to design, procure and construct.

However, in no event shall the total liquidated damages exceed 20 percent of the actual cost of Transmission Provider's Interconnection Facilities and Network Upgrades for which Transmission Provider has assumed responsibility to design, procure, and construct. The foregoing payments will be made by Transmission Provider to Interconnection Customer as just compensation for the damages caused to Interconnection Customer, which actual damages are uncertain and impossible to determine at this time, and as reasonable liquidated damages, but not as a penalty or a method to secure performance of this LGIA. Liquidated damages, when the Parties agree to them, are the exclusive remedy for the Transmission Provider's failure to meet its schedule.

No liquidated damages shall be paid to Interconnection Customer if: (1) Interconnection Customer is not ready to commence use of Transmission Provider's Interconnection Facilities or Network Upgrades to take the delivery of power for the Large Generating Facility's Trial Operation or to export power from the Large Generating Facility on the specified dates, unless Interconnection Customer would have been able to commence use of Transmission Provider's Interconnection Facilities or Network Upgrades to take the delivery of power for Large Generating Facility's Trial Operation or to export power from the Large Generating Facility, but for Transmission Provider's delay; (2) Transmission Provider's failure to meet the specified dates is the result of the action or inaction of Interconnection Customer or any other Interconnection Customer who has entered into an LGIA with Transmission Provider or any cause beyond Transmission Provider's reasonable control or reasonable ability to cure; (3) the Interconnection Customer has assumed responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades; or (4) the Parties have otherwise agreed.

Power System Stabilizers. The Interconnection Customer shall procure, install, maintain and operate Power System Stabilizers in accordance with the guidelines and procedures established by the Applicable Reliability Council. Transmission Provider reserves the right to reasonably establish minimum acceptable settings for any installed Power System Stabilizers, subject to the design and operating limitations of the Large Generating Facility. If the Large Generating Facility's Power System Stabilizers are removed from service or not capable of automatic

- operation, Interconnection Customer shall immediately notify Transmission Provider's system operator, or its designated representative. The requirements of this paragraph shall not apply to wind generators.
- 5.5 **Equipment Procurement**. If responsibility for construction of Transmission Provider's Interconnection Facilities or Network Upgrades is to be borne by Transmission Provider, then Transmission Provider shall commence design of Transmission Provider's Interconnection Facilities or Network Upgrades and procure necessary equipment as soon as practicable after all of the following conditions are satisfied, unless the Parties otherwise agree in writing:
 - **5.5.1** Transmission Provider has completed the Interconnection Facilities Study pursuant to the Interconnection Facilities Study Agreement;
 - **5.5.2** Transmission Provider has received written authorization to proceed with design and procurement from Interconnection Customer by the date specified in Appendix B, Milestones; and
 - **5.5.3** Interconnection Customer has provided security to Transmission Provider in accordance with Article 11.5 by the dates specified in Appendix B, Milestones.
- 5.6 **Construction Commencement.** Transmission Provider shall commence construction of Transmission Provider's Interconnection Facilities and Network Upgrades for which it is responsible as soon as practicable after the following additional conditions are satisfied:
 - **5.6.1** Approval of the appropriate Governmental Authority has been obtained for any facilities requiring regulatory approval;
 - **5.6.2** Necessary real property rights and rights-of-way have been obtained, to the extent required for the construction of a discrete aspect of Transmission Provider's Interconnection Facilities and Network Upgrades;
 - **5.6.3** Transmission Provider has received written authorization to proceed with construction from Interconnection Customer by the date specified in Appendix B, Milestones; and
 - **5.6.4** Interconnection Customer has provided security to Transmission Provider in accordance with Article 11.5 by the dates specified in Appendix B, Milestones.
- 5.7 Work Progress. The Parties will keep each other advised periodically as to the

progress of their respective design, procurement and construction efforts. Either Party may, at any time, request a progress report from the other Party. If, at any time, Interconnection Customer determines that the completion of Transmission Provider's Interconnection Facilities will not be required until after the specified In-Service Date, Interconnection Customer will provide written notice to Transmission Provider of such later date upon which the completion of Transmission Provider's Interconnection Facilities will be required.

5.8 Information Exchange. As soon as reasonably practicable after the Effective Date, the Parties shall exchange information regarding the design and compatibility of the Parties' Interconnection Facilities and compatibility of the Interconnection Facilities with Transmission Provider's Transmission System, and shall work diligently and in good faith to make any necessary design changes.

5.9 Other Interconnection Options

- 5.9.1. Limited Operation. If any of Transmission Provider's Interconnection Facilities or Network Upgrades are not reasonably expected to be completed prior to the Commercial Operation Date of the Large Generating Facility, Transmission Provider shall, upon the request and at the expense of Interconnection Customer, perform operating studies on a timely basis to determine the extent to which the Large Generating Facility and Interconnection Customer's Interconnection Facilities may operate prior to the completion of Transmission Provider's Interconnection Facilities or Network Upgrades consistent with Applicable Laws and Regulations, Applicable Reliability Standards, Good Utility Practice, and this LGIA. Transmission Provider shall permit Interconnection Customer to operate the Large Generating Facility and Interconnection Customer's Interconnection Facilities in accordance with the results of such studies.
- Interconnection Customer, and prior to completion of requisite
 Interconnection Facilities, Network Upgrades, Distribution Upgrades,
 or System Protection Facilities Transmission Provider may execute a
 Provisional Large Generator Interconnection Agreement or
 Interconnection Customer may request the filing of an unexecuted
 Provisional Large Generator Interconnection Agreement with the
 Interconnection Customer for limited Interconnection Service at the
 discretion of Transmission Provider based upon an evaluation that
 will consider the results of available studies. Transmission Provider
 shall determine, through available studies or additional studies as
 necessary, whether stability, short circuit, thermal, and/or voltage
 issues would arise if Interconnection Customer interconnects without

modifications to the Generating Facility or Transmission System. Transmission Provider shall determine whether any Interconnection Facilities, Network Upgrades, Distribution Upgrades, or System Protection Facilities that are necessary to meet the requirements of NERC, or any applicable Regional Entity for the interconnection of a new, modified and/or expanded Generating Facility are in place prior to the commencement of Interconnection Service from the Generating Facility. Where available studies indicate that such, Interconnection Facilities, Network Upgrades, Distribution Upgrades, and/or System Protection Facilities that are required for the interconnection of a new, modified and/or expanded Generating Facility are not currently in place, Transmission Provider will perform a study, at the Interconnection Customer's expense, to confirm the facilities that are required for Provisional Interconnection Service. The maximum permissible output of the Generating Facility in the Provisional Large Generator Interconnection Agreement shall be studied and updated on an annual frequency at the Interconnection Customer's expense. Interconnection Customer assumes all risk and liabilities with respect to changes between the Provisional Large Generator Interconnection Agreement and the Large Generator Interconnection Agreement, including changes in output limits and Interconnection Facilities, Network Upgrades, Distribution Upgrades, and/or System Protection Facilities cost responsibilities.

5.10 Interconnection Customer's Interconnection Facilities ('ICIF').

Interconnection Customer shall, at its expense, design, procure, construct, own and install the ICIF, as set forth in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades.

5.10.1 Interconnection Customer's Interconnection Facility Specifications.

Interconnection Customer shall submit initial specifications for the ICIF, including System Protection Facilities, to Transmission Provider at least one hundred eighty (180) Calendar Days prior to the Initial Synchronization Date; and final specifications for review and comment at least ninety (90) Calendar Days prior to the Initial Synchronization Date. Transmission Provider shall review such specifications to ensure that the ICIF are compatible with the technical specifications, operational control, and safety requirements of Transmission Provider and comment on such specifications within thirty (30) Calendar Days of Interconnection Customer's submission. All specifications provided hereunder shall be deemed confidential.

5.10.2 Transmission Provider's Review. Transmission Provider's review of Interconnection Customer's final specifications shall not be construed as

confirming, endorsing, or providing a warranty as to the design, fitness, safety, durability or reliability of the Large Generating Facility, or the ICIF. Interconnection Customer shall make such changes to the ICIF as may reasonably be required by Transmission Provider, in accordance with Good Utility Practice, to ensure that the ICIF are compatible with the technical specifications, operational control, and safety requirements of Transmission Provider.

- **5.10.3 ICIF Construction**. The ICIF shall be designed and constructed in accordance with Good Utility Practice. Within one hundred twenty (120) Calendar Days after the Commercial Operation Date, unless the Parties agree on another mutually acceptable deadline, Interconnection Customer shall deliver to Transmission Provider "as-built" drawings, information and documents for the ICIF, such as: a one-line diagram, a site plan showing the Large Generating Facility and the ICIF, plan and elevation drawings showing the layout of the ICIF, a relay functional diagram, relaying AC and DC schematic wiring diagrams and relay settings for all facilities associated with Interconnection Customer's step-up transformers, the facilities connecting the Large Generating Facility to the step-up transformers and the ICIF, and the impedances (determined by factory tests) for the associated step-up transformers and the Large Generating Facility. The Interconnection Customer shall provide Transmission Provider specifications for the excitation system, automatic voltage regulator, Large Generating Facility control and protection settings, transformer tap settings, and communications, if applicable.
- 5.11 Transmission Provider's Interconnection Facilities Construction. Transmission Provider's Interconnection Facilities shall be designed and constructed in accordance with Good Utility Practice. Upon request, within one hundred twenty (120) Calendar Days after the Commercial Operation Date, unless the Parties agree on another mutually acceptable deadline, Transmission Provider shall deliver to Interconnection Customer the following "as-built" drawings, information and documents for Transmission Provider's Interconnection Facilities (include appropriate drawings and relay diagrams).

Transmission Provider will obtain control of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades upon completion of such facilities.

5.12 Access Rights. Upon reasonable notice and supervision by a Party, and subject to any required or necessary regulatory approvals, a Party ("Granting Party") shall furnish at no cost to the other Party ("Access Party") any rights of use, licenses, rights of way and easements with respect to lands owned or controlled by the

Granting Party, its agents (if allowed under the applicable agency agreement), or any Affiliate, that are necessary to enable the Access Party to obtain ingress and egress to construct, operate, maintain, repair, test (or witness testing), inspect, replace or remove facilities and equipment to: (i) interconnect the Large Generating Facility with the Transmission System; (ii) operate and maintain the Large Generating Facility, the Interconnection Facilities and the Transmission System; and (iii) disconnect or remove the Access Party's facilities and equipment upon termination of this LGIA. In exercising such licenses, rights of way and easements, the Access Party shall not unreasonably disrupt or interfere with normal operation of the Granting Party's business and shall adhere to the safety rules and procedures established in advance, as may be changed from time to time, by the Granting Party and provided to the Access Party.

- 5.13 Lands of Other Property Owners. If any part of Transmission Provider or Transmission Owner's Interconnection Facilities and/or Network Upgrades is to be installed on property owned by persons other than Interconnection Customer or Transmission Provider or Transmission Owner shall at Interconnection Customer's expense use efforts, similar in nature and extent to those that it typically undertakes on its own behalf or on behalf of its Affiliates, including use of its eminent domain authority, and to the extent consistent with state law, to procure from such persons any rights of use, licenses, rights of way and easements that are necessary to construct, operate, maintain, test, inspect, replace or remove Transmission Provider or Transmission Owner's Interconnection Facilities and/or Network Upgrades upon such property.
- 5.14 Permits. Transmission Provider or Transmission Owner and Interconnection Customer shall cooperate with each other in good faith in obtaining all permits, licenses and authorizations that are necessary to accomplish the interconnection in compliance with Applicable Laws and Regulations. With respect to this paragraph, Transmission Provider or Transmission Owner shall provide permitting assistance to Interconnection Customer comparable to that provided to Transmission Provider's own, or an Affiliate's generation.
- request Transmission Provider to construct, and Transmission Provider shall construct, using Reasonable Efforts to accommodate Interconnection Customer's In-Service Date, all or any portion of any Network Upgrades required for Interconnection Customer to be interconnected to the Transmission System which are included in the Base Case of the Facilities Study for Interconnection Customer, and which also are required to be constructed for another Interconnection Customer, but where such construction is not scheduled to be completed in time to achieve Interconnection Customer's In-Service Date.

5.16 **Suspension**. Interconnection Customer reserves the right, upon written notice to Transmission Provider, to suspend at any time all work by Transmission Provider associated with the construction and installation of Transmission Provider's Interconnection Facilities and/or Network Upgrades required under this LGIA with the condition that Transmission System shall be left in a safe and reliable condition in accordance with Good Utility Practice and Transmission Provider's safety and reliability criteria. In such event, Interconnection Customer shall be responsible for all reasonable and necessary costs which Transmission Provider (i) has incurred pursuant to this LGIA prior to the suspension and (ii) incurs in suspending such work, including any costs incurred to perform such work as may be necessary to ensure the safety of persons and property and the integrity of the Transmission System during such suspension and, if applicable, any costs incurred in connection with the cancellation or suspension of material, equipment and labor contracts which Transmission Provider cannot reasonably avoid; provided, however, that prior to canceling or suspending any such material, equipment or labor contract, Transmission Provider shall obtain Interconnection Customer's authorization to do SO.

Transmission Provider shall invoice Interconnection Customer for such costs pursuant to Article 12 and shall use due diligence to minimize its costs. In the event Interconnection Customer suspends work by Transmission Provider required under this LGIA pursuant to this Article 5.16, and has not requested Transmission Provider to recommence the work required under this LGIA on or before the expiration of three (3) years following commencement of such suspension, this LGIA shall be deemed terminated. The three-year period shall begin on the date the suspension is requested, or the date of the written notice to Transmission Provider, if no effective date is specified.

5.16.1 Effect of Missed Interconnection Customer Milestones.

If Interconnection Customer fails to provide notice of suspension pursuant to Article 5.15, and Interconnection Customer fails to fulfill or complete any Interconnection Customer milestone provided in Appendix B, this constitutes a Breach under this LGIA. Depending upon the consequences of the Breach and effectiveness of the cure pursuant to Article 17, Transmission Provider's Appendix B milestones may be revised following consultation with Interconnection Customer, consistent with Reasonable Efforts, and in consideration of all relevant circumstances. Parties shall employ Reasonable Efforts to maintain their remaining respective Appendix B milestones.

5.16.2 Effect of Suspension; Parties' Obligations.

In the event that Interconnection Customer suspends work pursuant to

this Article 5.16, the applicable construction duration, timelines and schedules set forth in Appendix B shall be suspended during the period of suspension. Should Interconnection Customer thereafter request that work be recommenced, Appendix A and Appendix B may be revised to account for construction sequencing and modified milestones. If the Commercial Operation Date is extended beyond three (3) cumulative years described in Section 4.4.5 of the LGIP and Article 2.3.1 of this LGIA, such an extension may be considered a Material Modification and result in the termination of the LGIA under Article 2.3.1. Interconnection Customer is required to maintain Site Control while this LGIA is in effect, including during suspension.

5.17 Taxes.

- 5.17.1 Interconnection Customer Payments Not Taxable. The Parties intend that all payments or property transfers made by Interconnection Customer to Transmission Provider for the installation of Transmission Provider's Interconnection Facilities and the Network Upgrades shall be non-taxable, either as contributions to capital, or as an advance, in accordance with the Internal Revenue Code and any applicable state income tax laws and shall not be taxable as contributions in aid of construction or otherwise under the Internal Revenue Code and any applicable state income tax laws.
- 5.17.2 Representations and Covenants. In accordance with IRS Notice 2001-82 and IRS Notice 88-129, Interconnection Customer represents and covenants that (i) ownership of the electricity generated at the Large Generating Facility will pass to another party prior to the transmission of the electricity on the Transmission System, (ii) for income tax purposes, the amount of any payments and the cost of any property transferred to Transmission Provider for Transmission Provider's Interconnection Facilities will be capitalized by Interconnection Customer as an intangible asset and recovered using the straight-line method over a useful life of twenty (20) years, and (iii) any portion of Transmission Provider's Interconnection Facilities that is a "dual-use intertie," within the meaning of IRS Notice 88-129, is reasonably expected to carry only a de minimis amount of electricity in the direction of the Large Generating Facility. For this purpose, "de minimis amount" means no more than 5 percent of the total power flows in both directions, calculated in accordance with the "5 percent test" set forth in IRS Notice 88-129. This is not intended to be an exclusive list of the relevant conditions that must be met to conform to IRS requirements for non-taxable treatment.

At Transmission Provider's request, Interconnection Customer shall provide Transmission Provider with a report from an independent engineer confirming its representation in clause (iii), above.

Transmission Provider represents and covenants that the cost of Transmission Provider's Interconnection Facilities paid for by Interconnection Customer will have no net effect on the base upon which rates are determined.

5.17.3 Indemnification for the Cost Consequences of Current Tax Liability Imposed Upon the Transmission Provider.

Notwithstanding Article 5.17.1, Interconnection Customer shall protect, indemnify and hold harmless Transmission Provider from the cost consequences of any current tax liability imposed against Transmission Provider as the result of payments or property transfers made by Interconnection Customer to Transmission Provider under this LGIA for Interconnection Facilities, as well as any interest and penalties, other than interest and penalties attributable to any delay caused by Transmission Provider.

Transmission Provider shall not include a gross-up for the cost consequences of any current tax liability in the amounts it charges Interconnection Customer under this LGIA unless (i) Transmission Provider has determined, in good faith, that the payments or property transfers made by Interconnection Customer to Transmission Provider should be reported as income subject to taxation or (ii) any Governmental Authority directs Transmission Provider to report payments or property as income subject to taxation; provided, however, that Transmission Provider may require Interconnection Customer to provide security for Interconnection Facilities, in a form reasonably acceptable to Transmission Provider (such as a parental guarantee or a letter of credit), in an amount equal to the cost consequences of any current tax liability under this Article 5.17. Interconnection Customer shall reimburse Transmission Provider for such costs on a fully grossed-up basis, in accordance with Article 5.17.4, within thirty (30) Calendar Days of receiving written notification from Transmission Provider of the amount due, including detail about how the amount was calculated.

The indemnification obligation shall terminate at the earlier of (1) the expiration of the ten year testing period and the applicable statute of limitation, as it may be extended by Transmission Provider upon request of the IRS, to keep these years open for audit or adjustment, or

(2) the occurrence of a subsequent taxable event and the payment of any related indemnification obligations as contemplated by this Article 5.17.

5.17.4 Tax Gross-Up Amount. Interconnection Customer's liability for the cost consequences of any current tax liability under this Article 5.17 shall be calculated on a fully grossed-up basis. Except as may otherwise be agreed to by the parties, this means that Interconnection Customer will pay Transmission Provider, in addition to the amount paid for the Interconnection Facilities and Network Upgrades, an amount equal to (1) the current taxes imposed on Transmission Provider ("Current Taxes") on the excess of (a) the gross income realized by Transmission Provider as a result of payments or property transfers made by Interconnection Customer to Transmission Provider under this LGIA (without regard to any payments under this Article 5.17) (the "Gross Income Amount") over (b) the present value of future tax deductions for depreciation that will be available as a result of such payments or property transfers (the "Present Value Depreciation Amount"), plus (2) an additional amount sufficient to permit Transmission Provider to receive and retain, after the payment of all Current Taxes, an amount equal to the net amount described in clause (1).

> For this purpose, (i) Current Taxes shall be computed based on Transmission Provider's composite federal and state tax rates at the time the payments or property transfers are received and Transmission Provider will be treated as being subject to tax at the highest marginal rates in effect at that time (the "Current Tax Rate"), and (ii) the Present Value Depreciation Amount shall be computed by discounting Transmission Provider's anticipated tax depreciation deductions as a result of such payments or property transfers by Transmission Provider's current weighted average cost of capital. Thus, the formula for calculating Interconnection Customer's liability to Transmission Owner pursuant to this Article 5.17.4 can be expressed as follows: (Current Tax Rate x (Gross Income Amount -Present Value of Tax Depreciation))/(1Current Tax Rate). Interconnection Customer's estimated tax liability in the event taxes are imposed shall be stated in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades.

5.17.5 Private Letter Ruling or Change or Clarification of Law. At Interconnection Customer's request and expense, Transmission Provider shall file with the IRS a request for a private letter ruling as

to whether any property transferred or sums paid, or to be paid, by Interconnection Customer to Transmission Provider under this LGIA are subject to federal income taxation. Interconnection Customer will prepare the initial draft of the request for a private letter ruling, and will certify under penalties of perjury that all facts represented in such request are true and accurate to the best of Interconnection Customer's knowledge. Transmission Provider and Interconnection Customer shall cooperate in good faith with respect to the submission of such request.

Transmission Provider shall keep Interconnection Customer fully informed of the status of such request for a private letter ruling and shall execute either a privacy act waiver or a limited power of attorney, in a form acceptable to the IRS, that authorizes Interconnection Customer to participate in all discussions with the IRS regarding such request for a private letter ruling. Transmission Provider shall allow Interconnection Customer to attend all meetings with IRS officials about the request and shall permit Interconnection Customer to prepare the initial drafts of any follow-up letters in connection with the request.

- Subsequent Taxable Events. If, within 10 years from the date on which the relevant Transmission Provider's Interconnection Facilities are placed in service, (i) Interconnection Customer Breaches the covenants contained in Article 5.17.2, (ii) a "disqualification event" occurs within the meaning of IRS Notice 88-129, or (iii) this LGIA terminates and Transmission Provider retains ownership of the Interconnection Facilities and Network Upgrades, Interconnection Customer shall pay a tax gross-up for the cost consequences of any current tax liability imposed on Transmission Provider, calculated using the methodology described in Article 5.17.4 and in accordance with IRS Notice 90-60.
- 5.17.7 Contests. In the event any Governmental Authority determines that Transmission Provider's receipt of payments or property constitutes income that is subject to taxation, Transmission Provider shall notify Interconnection Customer, in writing, within thirty (30) Calendar Days of receiving notification of such determination by a Governmental Authority. Upon the timely written request by Interconnection Customer and at Interconnection Customer's sole expense, Transmission Provider may appeal, protest, seek abatement of, or otherwise oppose such determination. Upon Interconnection Customer's written request and sole expense, Transmission Provider

may file a claim for refund with respect to any taxes paid under this Article 5.17, whether or not it has received such a determination. Transmission Provider reserves the right to make all decisions with regard to the prosecution of such appeal, protest, abatement or other contest, including the selection of counsel and compromise or settlement of the claim, but Transmission Provider shall keep Interconnection Customer informed, shall consider in good faith suggestions from Interconnection Customer about the conduct of the contest, and shall reasonably permit Interconnection Customer or an Interconnection Customer representative to attend contest proceedings.

Interconnection Customer shall pay to Transmission Provider on a periodic basis, as invoiced by Transmission Provider, Transmission Provider's documented reasonable costs of prosecuting such appeal, protest, abatement or other contest. At any time during the contest, Transmission Provider may agree to a settlement either with Interconnection Customer's consent or after obtaining written advice from nationally-recognized tax counsel, selected by Transmission Provider, but reasonably acceptable to Interconnection Customer, that the proposed settlement represents a reasonable settlement given the hazards of litigation.

Interconnection Customer's obligation shall be based on the amount of the settlement agreed to by Interconnection Customer, or if a higher amount, so much of the settlement that is supported by the written advice from nationally-recognized tax counsel selected under the terms of the preceding sentence. The settlement amount shall be calculated on a fully grossed-up bases to cover any related cost consequences of the current tax liability. Any settlement without Interconnection Customer's consent or such written advice will relieve Interconnection Customer from any obligation to indemnify Transmission Provider for the tax at issue in the contest.

5.17.8 Refund. In the event that (a) a private letter ruling is issued to Transmission Provider which holds that any amount paid or the value of any property transferred by Interconnection Customer to Transmission Provider under the terms of this LGIA is not subject to federal income taxation, (b) any legislative change or administrative announcement, notice, ruling or other determination makes it reasonably clear to Transmission Provider in good faith that any amount paid or the value of any property transferred by Interconnection Customer to Transmission Provider under the terms

of this LGIA is not taxable to Transmission Provider, (c) any abatement, appeal, protest, or other contest results in a determination that any payments or transfers made by Interconnection Customer to Transmission Provider are not subject to federal income tax, or (d) if Transmission Provider receives a refund from any taxing authority for any overpayment of tax attributable to any payment or property transfer made by Interconnection Customer to Transmission Provider pursuant to this LGIA, Transmission Provider shall promptly refund to Interconnection Customer the following:

- (i) any payment made by Interconnection Customer under this Article 5.17 for taxes that is attributable to the amount determined to be non-taxable, together with interest thereon,
- (ii) interest on any amounts paid by Interconnection Customer to Transmission Provider for such taxes which Transmission Provider did not submit to the taxing authority, calculated in accordance with the methodology set forth in FERC's regulations at 18 CFR §35.19a(a)(2)(iii) from the date payment was made by Interconnection Customer to the date Transmission Provider refunds such payment to Interconnection Customer, and
- (iii) with respect to any such taxes paid by Transmission Provider, any refund or credit Transmission Provider receives or to which it may be entitled from any Governmental Authority, interest (or that portion thereof attributable to the payment described in clause (i), above) owed to Transmission Provider for such overpayment of taxes (including any reduction in interest otherwise payable by Transmission Provider to any Governmental Authority resulting from an offset or credit); provided, however, that Transmission Provider will remit such amount promptly to Interconnection Customer only after and to the extent that Transmission Provider has received a tax refund, credit or offset from any Governmental Authority for any applicable overpayment of income tax related to Transmission Provider's Interconnection Facilities.

The intent of this provision is to leave the Parties, to the extent practicable, in the event that no taxes are due with respect to any payment for Interconnection Facilities and Network Upgrades hereunder, in the same position they would have been in had no such

tax payments been made.

- 5.17.9 **Taxes Other Than Income Taxes.** Upon the timely request by Interconnection Customer, and at Interconnection Customer's sole expense, Transmission Provider may appeal, protest, seek abatement of, or otherwise contest any tax (other than federal or state income tax) asserted or assessed against Transmission Provider for which Interconnection Customer may be required to reimburse Transmission Provider under the terms of this LGIA. Interconnection Customer shall pay to Transmission Provider on a periodic basis, as invoiced by Transmission Provider, Transmission Provider's documented reasonable costs of prosecuting such appeal, protest, abatement, or other contest. Interconnection Customer and Transmission Provider shall cooperate in good faith with respect to any such contest. Unless the payment of such taxes is a prerequisite to an appeal or abatement or cannot be deferred, no amount shall be payable by Interconnection Customer to Transmission Provider for such taxes until they are assessed by a final, non-appealable order by any court or agency of competent jurisdiction. In the event that a tax payment is withheld and ultimately due and payable after appeal, Interconnection Customer will be responsible for all taxes, interest and penalties, other than penalties attributable to any delay caused by Transmission Provider.
- Transmission Owners Who Are Not Transmission Providers. If Transmission Provider is not the same entity as the Transmission Owner, then (i) all references in this Article 5.17 to Transmission Provider shall be deemed also to refer to and to include the Transmission Owner, as appropriate, and (ii) this LGIA shall not become effective until such Transmission Owner shall have agreed in writing to assume all of the duties and obligations of Transmission Provider under this Article 5.17 of this LGIA.
- **5.18 Tax Status**. Each Party shall cooperate with the other to maintain the other Party's tax status. Nothing in this LGIA is intended to adversely affect any Transmission Provider's tax exempt status with respect to the issuance of bonds including, but not limited to, Local Furnishing Bonds.

5.19 Modification.

5.19.1 General. Either Party may undertake modifications to its facilities. If a Party plans to undertake a modification that reasonably may be expected to affect the other Party's facilities, that Party shall provide

to the other Party sufficient information regarding such modification so that the other Party may evaluate the potential impact of such modification prior to commencement of the work. Such information shall be deemed to be confidential hereunder and shall include information concerning the timing of such modifications and whether such modifications are expected to interrupt the flow of electricity from the Large Generating Facility. The Party desiring to perform such work shall provide the relevant drawings, plans, and specifications to the other Party at least ninety (90) Calendar Days in advance of the commencement of the work or such shorter period upon which the Parties may agree, which agreement shall not unreasonably be withheld, conditioned or delayed.

In the case of Large Generating Facility modifications that do not require Interconnection Customer to submit an Interconnection Request, Transmission Provider shall provide, within thirty (30) Calendar Days (or such other time as the Parties may agree), an estimate of any additional modifications to the Transmission System, Transmission Provider's Interconnection Facilities or Network Upgrades necessitated by such Interconnection Customer modification and a good faith estimate of the costs thereof.

- **Standards**. Any additions, modifications, or replacements made to a Party's facilities shall be designed, constructed and operated in accordance with this LGIA and Good Utility Practice.
- 5.19.3 Modification Costs. Interconnection Customer shall not be directly assigned for the costs of any additions, modifications, or replacements that Transmission Provider makes to Transmission Provider's Interconnection Facilities or the Transmission System to facilitate the interconnection of a third party to Transmission Provider's Interconnection Facilities or the Transmission System, or to provide transmission service to a third party under Transmission Provider's Tariff. Interconnection Customer shall be responsible for the costs of any additions, modifications, or replacements to Interconnection Customer's Interconnection Facilities that may be necessary to maintain or upgrade such Interconnection Customer's Interconnection Facilities consistent with Applicable Laws and Regulations, Applicable Reliability Standards or Good Utility Practice.

Article 6. Testing and Inspection

- 6.1 Pre-Commercial Operation Date Testing and Modifications. Prior to the Commercial Operation Date, Transmission Provider shall test Transmission Provider's Interconnection Facilities and Network Upgrades and Interconnection Customer shall test the Large Generating Facility and Interconnection Customer's Interconnection Facilities to ensure their safe and reliable operation. Similar testing may be required after initial operation. Each Party shall make any modifications to its facilities that are found to be necessary as a result of such testing. Interconnection Customer shall bear the cost of all such testing and modifications. Interconnection Customer shall generate test energy at the Large Generating Facility only if it has arranged for the delivery of such test energy.
- 6.2 Post-Commercial Operation Date Testing and Modifications. Each Party shall at its own expense perform routine inspection and testing of its facilities and equipment in accordance with Good Utility Practice as may be necessary to ensure the continued interconnection of the Large Generating Facility with the Transmission System in a safe and reliable manner. Each Party shall have the right, upon advance written notice, to require reasonable additional testing of the other Party's facilities, at the requesting Party's expense, as may be in accordance with Good Utility Practice.
- **Right to Observe Testing**. Each Party shall notify the other Party in advance of its performance of tests of its Interconnection Facilities. The other Party has the right, at its own expense, to observe such testing.
- 6.4 **Right to Inspect**. Each Party shall have the right, but shall have no obligation to: (i) observe the other Party's tests and/or inspection of any of its System Protection Facilities and other protective equipment, including Power System Stabilizers; (ii) review the settings of the other Party's System Protection Facilities and other protective equipment; and (iii) review the other Party's maintenance records relative to the Interconnection Facilities, the System Protection Facilities and other protective equipment. A Party may exercise these rights from time to time as it deems necessary upon reasonable notice to the other Party. The exercise or non-exercise by a Party of any such rights shall not be construed as an endorsement or confirmation of any element or condition of the Interconnection Facilities or the System Protection Facilities or other protective equipment or the operation thereof, or as a warranty as to the fitness, safety, desirability, or reliability of same. Any information that a Party obtains through the exercise of any of its rights under this Article 6.4 shall be deemed to be Confidential Information and treated pursuant to Article 22 of this LGIA.

Article 7. Metering

7.1 General. Each Party shall comply with the Applicable Reliability Council

requirements. Unless otherwise agreed by the Parties, Transmission Provider shall install Metering Equipment at the Point of Interconnection prior to any operation of the Large Generating Facility and shall own, operate, test and maintain such Metering Equipment. Power flows to and from the Large Generating Facility shall be measured at or, at Transmission Provider's option, compensated to, the Point of Interconnection. Transmission Provider shall provide metering quantities, in analog and/or digital form, to Interconnection Customer upon request. Interconnection Customer shall bear all reasonable documented costs associated with the purchase, installation, operation, testing and maintenance of the Metering Equipment.

- 7.2 Check Meters. Interconnection Customer, at its option and expense, may install and operate, on its premises and on its side of the Point of Interconnection, one or more check meters to check Transmission Provider's meters. Such check meters shall be for check purposes only and shall not be used for the measurement of power flows for purposes of this LGIA, except as provided in Article 7.4 below. The check meters shall be subject at all reasonable times to inspection and examination by Transmission Provider or its designee. The installation, operation and maintenance thereof shall be performed entirely by Interconnection Customer in accordance with Good Utility Practice.
- **7.3 Standards**. Transmission Provider shall install, calibrate, and test revenue quality Metering Equipment in accordance with applicable ANSI standards.
- Testing of Metering Equipment. Transmission Provider shall inspect and test all 7.4 Transmission Provider-owned Metering Equipment upon installation and at least once every two (2) years thereafter. If requested to do so by Interconnection Customer, Transmission Provider shall, at Interconnection Customer's expense, inspect or test Metering Equipment more frequently than every two (2) years. Transmission Provider shall give reasonable notice of the time when any inspection or test shall take place, and Interconnection Customer may have representatives present at the test or inspection. If at any time Metering Equipment is found to be inaccurate or defective, it shall be adjusted, repaired or replaced at Interconnection Customer's expense, in order to provide accurate metering, unless the inaccuracy or defect is due to Transmission Provider's failure to maintain, then Transmission Provider shall pay. If Metering Equipment fails to register, or if the measurement made by Metering Equipment during a test varies by more than two percent from the measurement made by the standard meter used in the test, Transmission Provider shall adjust the measurements by correcting all measurements for the period during which Metering Equipment was in error by using Interconnection Customer's check meters, if installed. If no such check meters are installed or if the period cannot be reasonably ascertained, the adjustment shall be for the period immediately preceding the test of the Metering Equipment equal to one-half the

time from the date of the last previous test of the Metering Equipment.

7.5 Metering Data. At Interconnection Customer's expense, the metered data shall be telemetered to one or more locations designated by Transmission Provider and one or more locations designated by Interconnection Customer. Such telemetered data shall be used, under normal operating conditions, as the official measurement of the amount of energy delivered from the Large Generating Facility to the Point of Interconnection.

Article 8. Communications

- 8.1 **Interconnection Customer Obligations**. Interconnection Customer shall maintain satisfactory operating communications with Transmission Provider's Transmission System dispatcher or representative designated by Transmission Provider. Interconnection Customer shall provide standard voice line, dedicated voice line and facsimile communications at its Large Generating Facility control room or central dispatch facility through use of either the public telephone system, or a voice communications system that does not rely on the public telephone system. Interconnection Customer shall also provide the dedicated data circuit(s) necessary to provide Interconnection Customer data to Transmission Provider as set forth in Appendix D, Security Arrangements Details. The data circuit(s) shall extend from the Large Generating Facility to the location(s) specified by Transmission Provider. Any required maintenance of such communications equipment shall be performed by Interconnection Customer. Operational communications shall be activated and maintained under, but not be limited to, the following events: system paralleling or separation, scheduled and unscheduled shutdowns, equipment clearances, and hourly and daily load data.
- 8.2 Remote Terminal Unit. Prior to the Initial Synchronization Date of the Large Generating Facility, a Remote Terminal Unit, or equivalent data collection and transfer equipment acceptable to the Parties, shall be installed by Interconnection Customer, or by Transmission Provider at Interconnection Customer's expense, to gather accumulated and instantaneous data to be telemetered to the location(s) designated by Transmission Provider through use of a dedicated point-to-point data circuit(s) as indicated in Article 8.1. The communication protocol for the data circuit(s) shall be specified by Transmission Provider. Instantaneous bidirectional analog real power and reactive power flow information must be telemetered directly to the location(s) specified by Transmission Provider.

Each Party will promptly advise the other Party if it detects or otherwise learns of any metering, telemetry or communications equipment errors or malfunctions that require the attention and/or correction by the other Party. The Party owning such equipment shall correct such error or malfunction as soon as reasonably feasible.

- **8.3 No Annexation**. Any and all equipment placed on the premises of a Party shall be and remain the property of the Party providing such equipment regardless of the mode and manner of annexation or attachment to real property, unless otherwise mutually agreed by the Parties.
- 8.4 Provision of Data from a Variable Energy Resource. The Interconnection Customer whose Generating Facility is a Variable Energy Resource shall provide meteorological and forced outage data to the Transmission Provider to the extent necessary for the Transmission Provider's development and deployment of power production forecasts for that class of Variable Energy Resources. The Interconnection Customer with a Variable Energy Resource having wind as the energy source, at a minimum, will be required to provide the Transmission Provider with site-specific meteorological data including: temperature, wind speed, wind direction, and atmospheric pressure. The Interconnection Customer with a Variable Energy Resource having solar as the energy source, at a minimum, will be required to provide the Transmission Provider with site-specific meteorological data including: temperature, atmospheric pressure, and irradiance. The Transmission Provider and Interconnection Customer whose Generating Facility is a Variable Energy Resource shall mutually agree to any additional meteorological data that are required for the development and deployment of a power production forecast. The Interconnection Customer whose Generating Facility is a Variable Energy Resource also shall submit data to the Transmission Provider regarding all forced outages to the extent necessary for the Transmission Provider's development and deployment of power production forecasts for that class of Variable Energy Resources. The exact specifications of the meteorological and forced outage data to be provided by the Interconnection Customer to the Transmission Provider, including the frequency and timing of data submittals, shall be made taking into account the size and configuration of the Variable Energy Resource, its characteristics, location, and its importance in maintaining generation resource adequacy and transmission system reliability in its area. All requirements for meteorological and forced outage data must be commensurate with the power production forecasting employed by the Transmission Provider. Such requirements for meteorological and forced outage data are set forth in Appendix C, Interconnection Details, of this LGIA, as they may change from time to time.

Article 9. Operations

9.1 General. Each Party shall comply with the Applicable Reliability Council requirements. Each Party shall provide to the other Party all information that may reasonably be required by the other Party to comply with Applicable Laws and Regulations and Applicable Reliability Standards.

- Date, Interconnection Customer shall notify Transmission Provider in writing of the Control Area in which the Large Generating Facility will be located. If Interconnection Customer elects to locate the Large Generating Facility in a Control Area other than the Control Area in which the Large Generating Facility is physically located, and if permitted to do so by the relevant transmission tariffs, all necessary arrangements, including but not limited to those set forth in Article 7 and Article 8 of this LGIA, and remote Control Area generator interchange agreements, if applicable, and the appropriate measures under such agreements, shall be executed and implemented prior to the placement of the Large Generating Facility in the other Control Area.
- 9.3 Transmission Provider Obligations. Transmission Provider shall cause the Transmission System and Transmission Provider's Interconnection Facilities to be operated, maintained and controlled in a safe and reliable manner and in accordance with this LGIA. Transmission Provider may provide operating instructions to Interconnection Customer consistent with this LGIA and Transmission Provider's operating protocols and procedures as they may change from time to time. Transmission Provider will consider changes to its operating protocols and procedures proposed by Interconnection Customer.
- 9.4 Interconnection Customer Obligations. Interconnection Customer shall at its own expense operate, maintain and control the Large Generating Facility and Interconnection Customer's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA. Interconnection Customer shall operate the Large Generating Facility and Interconnection Customer's Interconnection Facilities in accordance with all applicable requirements of the Control Area of which it is part, as such requirements are set forth in Appendix C, Interconnection Details, of this LGIA. Appendix C, Interconnection Details, will be modified to reflect changes to the requirements as they may change from time to time. Either Party may request that the other Party provide copies of the requirements set forth in Appendix C, Interconnection Details, of this LGIA.
- **9.5 Start-Up and Synchronization**. Consistent with the Parties' mutually acceptable procedures, Interconnection Customer is responsible for the proper synchronization of the Large Generating Facility to Transmission Provider's Transmission System.
- 9.6 Reactive Power and Primary Frequency Response.
 - 9.6.1 Power Factor Design Criteria.
 - **9.6.1.1 Synchronous Generation.** Interconnection Customer shall design the Large Generating Facility to maintain a composite power delivery at continuous rated power output at the Point of

Interconnection at a power factor within the range of 0.95 leading to 0.95 lagging, unless the Transmission Provider has established different requirements that apply to all generators in the Control Area on a comparable basis.

- 9.6.1.2 Non-Synchronous Generation. Interconnection Customer shall design the Large Generating Facility to maintain a composite power delivery at continuous rated power output at the high-side of the generator substation at a power factor within the range of 0.95 leading to 0.95 lagging, unless the Transmission Provider has established a different power factor range that applies to all non-synchronous generators in the Control Area on a comparable basis. This power factor range standard shall be dynamic and can be met using, for example, power electronics designed to supply this level of reactive capability (taking into account any limitations due to voltage level, real power output, etc.) or fixed and switched capacitors, or a combination of the two. This requirement shall only apply to newly interconnecting non-synchronous generators that have not yet executed a Facilities Study Agreement as of the effective date of the Final Rule establishing this requirement (Order No. 827).
- **9.6.2** Voltage Schedules. Once Interconnection Customer has synchronized the Large Generating Facility with the Transmission System, Transmission Provider shall require Interconnection Customer to operate the Large Generating Facility to produce or absorb reactive power within the design limitations of the Large Generating Facility set forth in Article 9.6.1 (Power Factor Design Criteria). Transmission Provider's voltage schedules shall treat all sources of reactive power in the Control Area in an equitable and not unduly discriminatory manner. Transmission Provider shall exercise Reasonable Efforts to provide Interconnection Customer with such schedules at least one (1) day in advance, and may make changes to such schedules as necessary to maintain the reliability of the Transmission System. Interconnection Customer shall operate the Large Generating Facility to maintain the specified output voltage or power factor at the Point of Interconnection within the design limitations of the Large Generating Facility set forth in Article 9.6.1 (Power Factor Design Criteria). If Interconnection Customer is unable to maintain the specified voltage or power factor, it shall promptly notify the System Operator.
 - **9.6.2.1 Voltage Regulators**. Whenever the Large Generating Facility

is operated in parallel with the Transmission System and voltage regulators are capable of operation, Interconnection Customer shall operate the Large Generating Facility with its voltage regulators in automatic operation. If the Large Generating Facility's voltage regulators are not capable of such automatic operation, Interconnection Customer shall immediately notify Transmission Provider's system operator, or its designated representative, and ensure that such Large Generating Facility's reactive power production or absorption (measured in MVARs) are within the design capability of the Large Generating Facility's generating unit(s) and steady state stability limits. Interconnection Customer shall not cause its Large Generating Facility to disconnect automatically or instantaneously from the Transmission System or trip any generating unit comprising the Large Generating Facility for an under or over frequency condition unless the abnormal frequency condition persists for a time period beyond the limits set forth in ANSI/IEEE Standard C37.106, or such other standard as applied to other generators in the Control Area on a comparable basis.

- 9.6.3 Payment for Reactive Power. Transmission Provider is required to pay Interconnection Customer for reactive power that Interconnection Customer provides or absorbs from the Large Generating Facility when Transmission Provider requests Interconnection Customer to operate its Large Generating Facility outside the range specified in Article 9.6.1, provided that if Transmission Provider pays its own or affiliated generators for reactive power service within the specified range, it must also pay Interconnection Customer. Payments shall be pursuant to Article 11.6 or such other agreement to which the Parties have otherwise agreed.
- 9.6.4 Primary Frequency Response. Interconnection Customer shall ensure the primary frequency response capability of its Large Generating Facility by installing, maintaining, and operating a functioning governor or equivalent controls. The term "functioning governor or equivalent controls" as used herein shall mean the required hardware and/or software that provides frequency responsive real power control with the ability to sense changes in system frequency and autonomously adjust the Large Generating Facility's real power output in accordance with the droop and deadband parameters and in the direction needed to correct frequency deviations. Interconnection Customer is required to install a governor or equivalent controls with the capability of operating: (1) with a maximum 5 percent droop and ±0.036 Hz deadband; or (2) in accordance with the relevant droop, deadband, and

timely and sustained response settings from an approved NERC Reliability Standard providing for equivalent or more stringent parameters. The droop characteristic shall be: (1) based on the nameplate capacity of the Large Generating Facility, and shall be linear in the range of frequencies between 59 to 61 Hz that are outside of the deadband parameter; or (2) based on an approved NERC Reliability Standard providing for an equivalent or more stringent parameter. The deadband parameter shall be: the range of frequencies above and below nominal (60 Hz) in which the governor or equivalent controls is not expected to adjust the Large Generating Facility's real power output in response to frequency deviations. The deadband shall be implemented: (1) without a step to the droop curve, that is, once the frequency deviation exceeds the deadband parameter, the expected change in the Large Generating Facility's real power output in response to frequency deviations shall start from zero and then increase (for under-frequency deviations) or decrease (for over-frequency deviations) linearly in proportion to the magnitude of the frequency deviation; or (2) in accordance with an approved NERC Reliability Standard providing for an equivalent or more stringent parameter. Interconnection Customer shall notify Transmission Provider that the primary frequency response capability of the Large Generating Facility has been tested and confirmed during commissioning. Once Interconnection Customer has synchronized the Large Generating Facility with the Transmission System, Interconnection Customer shall operate the Large Generating Facility consistent with the provisions specified in Sections 9.6.4.1 and 9.6.4.2 of this Agreement. The primary frequency response requirements contained herein shall apply to both synchronous and non-synchronous Large Generating Facilities.

9.6.4.1 Governor or Equivalent Controls. Whenever the Large Generating Facility is operated in parallel with the Transmission System, Interconnection Customer shall operate the Large Generating Facility with its governor or equivalent controls in service and responsive to frequency. Interconnection Customer shall: (1) in coordination with Transmission Provider and/or the relevant balancing authority, set the deadband parameter to: (1) a maximum of ± 0.036 Hz and set the droop parameter to a maximum of 5 percent; or (2) implement the relevant droop and deadband settings from an approved NERC Reliability Standard that provides for equivalent or more stringent parameters. Interconnection Customer shall be required to provide the status and settings of the governor or equivalent controls to Transmission Provider and/or the relevant balancing authority upon request. If Interconnection Customer needs to operate the Large

Generating Facility with its governor or equivalent controls not in service, Interconnection Customer shall immediately notify Transmission Provider and the relevant balancing authority, and provide both with the following information: (1) the operating status of the governor or equivalent controls (i.e., whether it is currently out of service or when it will be taken out of service); (2) the reasons for removing the governor or equivalent controls from service; and (3) a reasonable estimate of when the governor or equivalent controls will be returned to service. Interconnection Customer shall make Reasonable Efforts to return its governor or equivalent controls into service as soon as practicable. Interconnection Customer shall make Reasonable Efforts to keep outages of the Large Generating Facility's governor or equivalent controls to a minimum whenever the Large Generating Facility is operated in parallel with the Transmission System.

- 9.6.4.2 Timely and Sustained Response. Interconnection Customer shall ensure that the Large Generating Facility's real power response to sustained frequency deviations outside of the deadband setting is automatically provided and shall begin immediately after frequency deviates outside of the deadband, and to the extent the Large Generating Facility has operating capability in the direction needed to correct the frequency deviation. Interconnection Customer shall not block or otherwise inhibit the ability of the governor or equivalent controls to respond and shall ensure that the response is not inhibited, except under certain operational constraints including, but not limited to, ambient temperature limitations, physical energy limitations, outages of mechanical equipment, or regulatory requirements. The Large Generating Facility shall sustain the real power response at least until system frequency returns to a value within the deadband setting of the governor or equivalent controls. A Commission-approved Reliability Standard with equivalent or more stringent requirements shall supersede the above requirements.
- **9.6.4.3 Exemptions**. Large Generating Facilities that are regulated by the United States Nuclear Regulatory Commission shall be exempt from Sections 9.6.4, 9.6.4.1, and 9.6.4.2 of this Agreement. Large Generating Facilities that are behind the meter generation that is sized-to-load (i.e., the thermal load and the generation are near-balanced in real-time operation

and the generation is primarily controlled to maintain the unique thermal, chemical, or mechanical output necessary for the operating requirements of its host facility) shall be required to install primary frequency response capability in accordance with the droop and deadband capability requirements specified in Section 9.6.4, but shall be otherwise exempt from the operating requirements in Sections 9.6.4, 9.6.4.1, 9.6.4.2, and 9.6.4.4 of this Agreement.

9.6.4.4

Electric Storage Resources. Interconnection Customer interconnecting an electric storage resource shall establish an operating range in Appendix C of its LGIA that specifies a minimum state of charge and a maximum state of charge between which the electric storage resource will be required to provide primary frequency response consistent with the conditions set forth in Sections 9.6.4, 9.6.4.1, 9.6.4.2, and 9.6.4.3 of this Agreement. Appendix C shall specify whether the operating range is static or dynamic, and shall consider (1) the expected magnitude of frequency deviations in the interconnection; (2) the expected duration that system frequency will remain outside of the deadband parameter in the interconnection; (3) the expected incidence of frequency deviations outside of the deadband parameter in the interconnection; (4) the physical capabilities of the electric storage resource; (5) operational limitations of the electric storage resource due to manufacturer specifications; and (6) any other relevant factors agreed to by Transmission Provider and Interconnection Customer, and in consultation with the relevant transmission owner or balancing authority as appropriate. If the operating range is dynamic, then Appendix C must establish how frequently the operating range will be reevaluated and the factors that may be considered during its reevaluation.

Interconnection Customer's electric storage resource is required to provide timely and sustained primary frequency response consistent with Section 9.6.4.2 of this Agreement when it is online and dispatched to inject electricity to the Transmission System and/or receive electricity from the Transmission System. This excludes circumstances when the electric storage resource is not dispatched to inject electricity to the Transmission System and/or dispatched to receive electricity from the Transmission System. If Interconnection

Customer's electric storage resource is charging at the time of a frequency deviation outside of its deadband parameter, it is to increase (for over-frequency deviations) or decrease (for under-frequency deviations) the rate at which it is charging in accordance with its droop parameter. Interconnection Customer's electric storage resource is not required to change from charging to discharging, or vice versa, unless the response necessitated by the droop and deadband settings requires it to do so and it is technically capable of making such a transition.

9.7 Outages and Interruptions.

9.7.1 Outages.

- 9.7.1.1 Outage Authority and Coordination. Each Party may in accordance with Good Utility Practice in coordination with the other Party remove from service any of its respective Interconnection Facilities or Network Upgrades that may impact the other Party's facilities as necessary to perform maintenance or testing or to install or replace equipment. Absent an Emergency Condition, the Party scheduling a removal of such facility(ies) from service will use Reasonable Efforts to schedule such removal on a date and time mutually acceptable to the Parties. In all circumstances, any Party planning to remove such facility(ies) from service shall use Reasonable Efforts to minimize the effect on the other Party of such removal.
- 9.7.1.2 Outage Schedules. Transmission Provider shall post scheduled outages of its transmission facilities on the OASIS. Interconnection Customer shall submit its planned maintenance schedules for the Large Generating Facility to Transmission Provider for a minimum of a rolling twenty-four month period. Interconnection Customer shall update its planned maintenance schedules as necessary. Transmission Provider may request Interconnection Customer to reschedule its maintenance as necessary to maintain the reliability of the Transmission System; provided, however, adequacy of generation supply shall not be a criterion in determining Transmission System reliability. Transmission Provider shall compensate Interconnection Customer for any additional direct costs that Interconnection Customer incurs as a result of

having to reschedule maintenance, including any additional overtime, breaking of maintenance contracts or other costs above and beyond the cost Interconnection Customer would have incurred absent Transmission Provider's request to reschedule maintenance. Interconnection Customer will not be eligible to receive compensation, if during the twelve (12) months prior to the date of the scheduled maintenance, Interconnection Customer had modified its schedule of maintenance activities.

- 9.7.1.3 Outage Restoration. If an outage on a Party's Interconnection Facilities or Network Upgrades adversely affects the other Party's operations or facilities, the Party that owns or controls the facility that is out of service shall use Reasonable Efforts to promptly restore such facility(ies) to a normal operating condition consistent with the nature of the outage. The Party that owns or controls the facility that is out of service shall provide the other Party, to the extent such information is known, information on the nature of the Emergency Condition, an estimated time of restoration, and any corrective actions required. Initial verbal notice shall be followed up as soon as practicable with written notice explaining the nature of the outage.
- **9.7.2 Interruption of Service**. If required by Good Utility Practice to do so, Transmission Provider may require Interconnection Customer to interrupt or reduce deliveries of electricity if such delivery of electricity could adversely affect Transmission Provider's ability to perform such activities as are necessary to safely and reliably operate and maintain the Transmission System. The following provisions shall apply to any interruption or reduction permitted under this Article 9.7.2:
 - **9.7.2.1** The interruption or reduction shall continue only for so long as reasonably necessary under Good Utility Practice;
 - 9.7.2.2 Any such interruption or reduction shall be made on an equitable, non-discriminatory basis with respect to all generating facilities directly connected to the Transmission System;
 - 9.7.2.3 When the interruption or reduction must be made under circumstances which do not allow for advance notice,
 Transmission Provider shall notify Interconnection Customer

by telephone as soon as practicable of the reasons for the curtailment, interruption, or reduction, and, if known, its expected duration. Telephone notification shall be followed by written notification as soon as practicable;

- 9.7.2.4 Except during the existence of an Emergency Condition, when the interruption or reduction can be scheduled without advance notice, Transmission Provider shall notify Interconnection Customer in advance regarding the timing of such scheduling and further notify Interconnection Customer of the expected duration. Transmission Provider shall coordinate with Interconnection Customer using Good Utility Practice to schedule the interruption or reduction during periods of least impact to Interconnection Customer and Transmission Provider;
- 9.7.2.5 The Parties shall cooperate and coordinate with each other to the extent necessary in order to restore the Large Generating Facility, Interconnection Facilities, and the Transmission System to their normal operating state, consistent with system conditions and Good Utility Practice.
- 9.7.3 Under-Frequency and Over Frequency Conditions. The Transmission System is designed to automatically activate a load-shed program as required by the Applicable Reliability Council in the event of an under-frequency system disturbance. Interconnection Customer shall implement under-frequency and over-frequency relay set points for the Large Generating Facility as required by the Applicable Reliability Council to ensure "ride through" capability of the Transmission System. Large Generating Facility response to frequency deviations of pre-determined magnitudes, both under-frequency and over-frequency deviations, shall be studied and coordinated with Transmission Provider in accordance with Good Utility Practice. The term "ride through" as used herein shall mean the ability of a Generating Facility to stay connected to and synchronized with the Transmission System during system disturbances within a range of under-frequency and over-frequency conditions, in accordance with Good Utility Practice.
- 9.7.4 System Protection and Other Control Requirements.
 - **9.7.4.1 System Protection Facilities**. Interconnection Customer shall, at its expense, install, operate and maintain System Protection Facilities as a part of the Large Generating Facility or

Interconnection Customer's Interconnection Facilities.
Transmission Provider shall install at Interconnection
Customer's expense any System Protection Facilities that may
be required on Transmission Provider's Interconnection
Facilities or the Transmission System as a result of the
interconnection of the Large Generating Facility and
Interconnection Customer's Interconnection Facilities.

- **9.7.4.2** Each Party's protection facilities shall be designed and coordinated with other systems in accordance with Good Utility Practice.
- **9.7.4.3** Each Party shall be responsible for protection of its facilities consistent with Good Utility Practice.
- 9.7.4.4 Each Party's protective relay design shall incorporate the necessary test switches to perform the tests required in Article 6. The required test switches will be placed such that they allow operation of lockout relays while preventing breaker failure schemes from operating and causing unnecessary breaker operations and/or the tripping of Interconnection Customer's units.
- **9.7.4.5** Each Party will test, operate and maintain System Protection Facilities in accordance with Good Utility Practice.
- 9.7.4.6 Prior to the In-Service Date, and again prior to the Commercial Operation Date, each Party or its agent shall perform a complete calibration test and functional trip test of the System Protection Facilities. At intervals suggested by Good Utility Practice and following any apparent malfunction of the System Protection Facilities, each Party shall perform both calibration and functional trip tests of its System Protection Facilities. These tests do not require the tripping of any in-service generation unit. These tests do, however, require that all protective relays and lockout contacts be activated.
- 9.7.5 Requirements for Protection. In compliance with Good Utility Practice, Interconnection Customer shall provide, install, own, and maintain relays, circuit breakers and all other devices necessary to remove any fault contribution of the Large Generating Facility to any short circuit occurring on the Transmission System not otherwise isolated by Transmission Provider's equipment, such that the removal of the fault contribution shall be

coordinated with the protective requirements of the Transmission System. Such protective equipment shall include, without limitation, a disconnecting device or switch with load-interrupting capability located between the Large Generating Facility and the Transmission System at a site selected upon mutual agreement (not to be unreasonably withheld, conditioned or delayed) of the Parties. Interconnection Customer shall be responsible for protection of the Large Generating Facility and Interconnection Customer's other equipment from such conditions as negative sequence currents, over- or under-frequency, sudden load rejection, over- or under-voltage, and generator loss-of-field. Interconnection Customer shall be solely responsible to disconnect the Large Generating Facility and Interconnection Customer's other equipment if conditions on the Transmission System could adversely affect the Large Generating Facility.

- 9.7.6 Power Quality. Neither Party's facilities shall cause excessive voltage flicker nor introduce excessive distortion to the sinusoidal voltage or current waves as defined by ANSI Standard C84.1-1989, in accordance with IEEE Standard 519, or any applicable superseding electric industry standard. In the event of a conflict between ANSI Standard C84.1-1989, or any applicable superseding electric industry standard, ANSI Standard C84.1-1989, or the applicable superseding electric industry standard, shall control.
- 9.8 Switching and Tagging Rules. Each Party shall provide the other Party a copy of its switching and tagging rules that are applicable to the other Party's activities. Such switching and tagging rules shall be developed on a non-discriminatory basis. The Parties shall comply with applicable switching and tagging rules, as amended from time to time, in obtaining clearances for work or for switching operations on equipment.
- 9.9 Use of Interconnection Facilities by Third Parties.
 - **9.9.1 Purpose of Interconnection Facilities**. Except as may be required by Applicable Laws and Regulations, or as otherwise agreed to among the Parties, the Interconnection Facilities shall be constructed for the sole purpose of interconnecting the Large Generating Facility to the Transmission System and shall be used for no other purpose.
 - 9.9.2 Third Party Users. If required by Applicable Laws and Regulations or if the Parties mutually agree, such agreement not to be unreasonably withheld, to allow one or more third parties to use Transmission Provider's Interconnection Facilities, or any part thereof, Interconnection Customer will be entitled to compensation for the capital expenses it incurred in connection with the Interconnection Facilities based upon the pro rata use of the

Interconnection Facilities by Transmission Provider, all third party users, and Interconnection Customer, in accordance with Applicable Laws and Regulations or upon some other mutually-agreed upon methodology. In addition, cost responsibility for ongoing costs, including operation and maintenance costs associated with the Interconnection Facilities, will be allocated between Interconnection Customer and any third party users based upon the pro rata use of the Interconnection Facilities by Transmission Provider, all third party users, and Interconnection Customer, in accordance with Applicable Laws and Regulations or upon some other mutually agreed upon methodology. If the issue of such compensation or allocation cannot be resolved through such negotiations, it shall be submitted to FERC for resolution.

9.10 Disturbance Analysis Data Exchange. The Parties will cooperate with one another in the analysis of disturbances to either the Large Generating Facility or Transmission Provider's Transmission System by gathering and providing access to any information relating to any disturbance, including information from oscillography, protective relay targets, breaker operations and sequence of events records, and any disturbance information required by Good Utility Practice.

Article 10. Maintenance

- **10.1 Transmission Provider Obligations**. Transmission Provider shall maintain the Transmission System and Transmission Provider's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA.
- **10.2 Interconnection Customer Obligations**. Interconnection Customer shall maintain the Large Generating Facility and Interconnection Customer's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA.
- **10.3 Coordination**. The Parties shall confer regularly to coordinate the planning, scheduling and performance of preventive and corrective maintenance on the Large Generating Facility and the Interconnection Facilities.
- 10.4 Secondary Systems. Each Party shall cooperate with the other in the inspection, maintenance, and testing of control or power circuits that operate below 600 volts, AC or DC, including, but not limited to, any hardware, control or protective devices, cables, conductors, electric raceways, secondary equipment panels, transducers, batteries, chargers, and voltage and current transformers that directly affect the operation of a Party's facilities and equipment which may reasonably be expected to impact the other Party. Each Party shall provide advance notice to the other Party before undertaking any work on such circuits, especially on electrical circuits involving circuit breaker trip and close contacts, current transformers, or

potential transformers.

10.5 Operating and Maintenance Expenses. Subject to the provisions herein addressing the use of facilities by others, and except for operations and maintenance expenses associated with modifications made for providing interconnection or transmission service to a third party and such third party pays for such expenses, Interconnection Customer shall be responsible for all reasonable expenses including overheads, associated with: (1) owning, operating, maintaining, repairing, and replacing Interconnection Customer's Interconnection Facilities; and (2) operation, maintenance, repair and replacement of Transmission Provider's Interconnection Facilities.

Article 11. Performance Obligation

- 11.1 Interconnection Customer Interconnection Facilities. Interconnection Customer shall design, procure, construct, install, own and/or control Interconnection Customer Interconnection Facilities described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades, at its sole expense.
- 11.2 Transmission Provider's Interconnection Facilities. Transmission Provider or Transmission Owner shall design, procure, construct, install, own and/or control the Transmission Provider's Interconnection Facilities described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades, at the sole expense of the Interconnection Customer.
- 11.3 Network Upgrades and Distribution Upgrades. Transmission Provider or Transmission Owner shall design, procure, construct, install, and own the Network Upgrades and Distribution Upgrades described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades. The Interconnection Customer shall be responsible for all costs related to Distribution Upgrades. Unless Transmission Provider or Transmission Owner elects to fund the capital for the Network Upgrades, they shall be solely funded by Interconnection Customer.

11.4 Transmission Credits.

11.4.1 Repayment of Amounts Advanced for Network Upgrades.

Interconnection Customer shall be entitled to a cash repayment, equal to the total amount paid to Transmission Provider and Affected System Operator, if any, for the Network Upgrades, including any tax gross-up or other tax-related payments associated with Network Upgrades, and not refunded to Interconnection Customer pursuant to

Article 5.17.8 or otherwise, to be paid to Interconnection Customer on a dollar-for-dollar basis for the non-usage sensitive portion of transmission charges, as payments are made under Transmission Provider's Tariff and Affected System's Tariff for transmission services with respect to the Large Generating Facility. Any repayment shall include interest calculated in accordance with the methodology set forth in FERC's regulations at 18 C.F.R. § 35.19a(a)(2)(iii) from the date of any payment for Network Upgrades through the date on which the Interconnection Customer receives a repayment of such payment pursuant to this subparagraph. Interconnection Customer may assign such repayment rights to any person.

Notwithstanding the foregoing, Interconnection Customer, Transmission Provider, and Affected System Operator may adopt any alternative payment schedule that is mutually agreeable so long as Transmission Provider and Affected System Operator take one of the following actions no later than five years from the Commercial Operation Date: (1) return to Interconnection Customer any amounts advanced for Network Upgrades not previously repaid, or (2) declare in writing that Transmission Provider or Affected System Operator will continue to provide payments to Interconnection Customer on a dollar-for-dollar basis for the non-usage sensitive portion of transmission charges, or develop an alternative schedule that is mutually agreeable and provides for the return of all amounts advanced for Network Upgrades not previously repaid; however, full reimbursement shall not extend beyond twenty (20) years from the Commercial Operation Date.

If the Large Generating Facility fails to achieve Commercial Operation, but it or another Generating Facility is later constructed and makes use of the Network Upgrades, Transmission Provider and Affected System Operator shall at that time reimburse Interconnection Customer for the amounts advanced for the Network Upgrades. Before any such reimbursement can occur, the Interconnection Customer, or the entity that ultimately constructs the Generating Facility, if different, is responsible for identifying the entity to which reimbursement must be made.

11.4.2 Special Provisions for Affected Systems. Unless Transmission Provider provides, under the LGIA, for the repayment of amounts advanced to Affected System Operator for Network Upgrades, Interconnection Customer and Affected System Operator shall enter

into an agreement that provides for such repayment. The agreement shall specify the terms governing payments to be made by Interconnection Customer to the Affected System Operator as well as the repayment by the Affected System Operator.

- 11.4.3 Notwithstanding any other provision of this LGIA, nothing herein shall be construed as relinquishing or foreclosing any rights, including but not limited to firm transmission rights, capacity rights, transmission congestion rights, or transmission credits, that Interconnection Customer, shall be entitled to, now or in the future under any other agreement or tariff as a result of, or otherwise associated with, the transmission capacity, if any, created by the Network Upgrades, including the right to obtain cash reimbursements or transmission credits for transmission service that is not associated with the Large Generating Facility.
- 11.5 Provision of Security. At least thirty (30) Calendar Days prior to the commencement of the procurement, installation, or construction of a discrete portion of a Transmission Provider's Interconnection Facilities, Network Upgrades, or Distribution Upgrades, Interconnection Customer shall provide Transmission Provider, at Interconnection Customer's option, a guarantee, a surety bond, letter of credit or other form of security that is reasonably acceptable to Transmission Provider and is consistent with the Uniform Commercial Code of the jurisdiction identified in Article 14.2.1. Such security for payment shall be in an amount sufficient to cover the costs for constructing, procuring and installing the applicable portion of Transmission Provider's Interconnection Facilities, Network Upgrades, or Distribution Upgrades and shall be reduced on a dollar-for-dollar basis for payments made to Transmission Provider for these purposes.

In addition:

- The guarantee must be made by an entity that meets the creditworthiness requirements of Transmission Provider, and contain terms and conditions that guarantee payment of any amount that may be due from Interconnection Customer, up to an agreed-to maximum amount.
- 11.5.2 The letter of credit must be issued by a financial institution reasonably acceptable to Transmission Provider and must specify a reasonable expiration date.
- The surety bond must be issued by an insurer reasonably acceptable to Transmission Provider and must specify a reasonable expiration date.

- 11.6 **Interconnection Customer Compensation**. If Transmission Provider requests or directs Interconnection Customer to provide a service pursuant to Articles 9.6.3 (Payment for Reactive Power), or 13.5.1 of this LGIA, Transmission Provider shall compensate Interconnection Customer in accordance with Interconnection Customer's applicable rate schedule then in effect unless the provision of such service(s) is subject to an RTO or ISO FERC-approved rate schedule. Interconnection Customer shall serve Transmission Provider or RTO or ISO with any filing of a proposed rate schedule at the time of such filing with FERC. To the extent that no rate schedule is in effect at the time the Interconnection Customer is required to provide or absorb any Reactive Power under this LGIA, Transmission Provider agrees to compensate Interconnection Customer in such amount as would have been due Interconnection Customer had the rate schedule been in effect at the time service commenced; provided, however, that such rate schedule must be filed at FERC or other appropriate Governmental Authority within sixty (60) Calendar Days of the commencement of service.
 - 11.6.1 Interconnection Customer Compensation for Actions During Emergency Condition. Transmission Provider or RTO or ISO shall compensate Interconnection Customer for its provision of real and reactive power and other Emergency Condition services that Interconnection Customer provides to support the Transmission System during an Emergency Condition in accordance with Article 11.6.

Article 12. Invoice

- **12.1 General**. Each Party shall submit to the other Party, on a monthly basis, invoices of amounts due for the preceding month. Each invoice shall state the month to which the invoice applies and fully describe the services and equipment provided. The Parties may discharge mutual debts and payment obligations due and owing to each other on the same date through netting, in which case all amounts a Party owes to the other Party under this LGIA, including interest payments or credits, shall be netted so that only the net amount remaining due shall be paid by the owing Party.
- 12.2 Final Invoice. Within six months after completion of the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades, Transmission Provider shall provide an invoice of the final cost of the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades and shall set forth such costs in sufficient detail to enable Interconnection Customer to compare the actual costs with the estimates and to ascertain deviations, if any, from the cost estimates. Transmission Provider shall refund to Interconnection Customer any amount by which the actual payment by Interconnection Customer

for estimated costs exceeds the actual costs of construction within thirty (30) Calendar Days of the issuance of such final construction invoice.

- 12.3 Payment. Invoices shall be rendered to the paying Party at the address specified in Appendix F. The Party receiving the invoice shall pay the invoice within thirty (30) Calendar Days of receipt. All payments shall be made in immediately available funds payable to the other Party, or by wire transfer to a bank named and account designated by the invoicing Party. Payment of invoices by either Party will not constitute a waiver of any rights or claims either Party may have under this LGIA. If Interconnection Customer has not paid the final invoice following a withdrawal within thirty (30) Calendar Days, Transmission Provider shall draw upon the security provided under this LGIA to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.
- 12.4 Disputes. In the event of a billing dispute between Transmission Provider and Interconnection Customer, Transmission Provider shall continue to provide Interconnection Service under this LGIA as long as Interconnection Customer: (i) continues to make all payments not in dispute; and (ii) pays to Transmission Provider or into an independent escrow account the portion of the invoice in dispute, pending resolution of such dispute. If Interconnection Customer fails to meet these two requirements for continuation of service, then Transmission Provider may provide notice to Interconnection Customer of a Default pursuant to Article 17. Within thirty (30) Calendar Days after the resolution of the dispute, the Party that owes money to the other Party shall pay the amount due with interest calculated in accord with the methodology set forth in FERC's regulations at 18 CFR § 35.19a(a)(2)(iii).

Article 13. Emergencies

13.1 Definition. "Emergency Condition" shall mean a condition or situation: (i) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (ii) that, in the case of Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to the Transmission System, Transmission Provider's Interconnection Facilities or the Transmission Systems of others to which the Transmission System is directly connected; or (iii) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Large Generating Facility or Interconnection Customer's Interconnection Facilities' System restoration and black start shall be considered Emergency Conditions; provided, that Interconnection Customer is not obligated by

this LGIA to possess black start capability.

- **13.2 Obligations**. Each Party shall comply with the Emergency Condition procedures of the applicable ISO/RTO, NERC, the Applicable Reliability Council, Applicable Laws and Regulations, and any emergency procedures agreed to by the Joint Operating Committee.
- 13.3 **Notice**. Transmission Provider shall notify Interconnection Customer promptly when it becomes aware of an Emergency Condition that affects Transmission Provider's Interconnection Facilities or the Transmission System that may reasonably be expected to affect Interconnection Customer's operation of the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Interconnection Customer shall notify Transmission Provider promptly when it becomes aware of an Emergency Condition that affects the Large Generating Facility or Interconnection Customer's Interconnection Facilities that may reasonably be expected to affect the Transmission System or Transmission Provider's Interconnection Facilities. To the extent information is known, the notification shall describe the Emergency Condition, the extent of the damage or deficiency, the expected effect on the operation of Interconnection Customer's or Transmission Provider's facilities and operations, its anticipated duration and the corrective action taken and/or to be taken. The initial notice shall be followed as soon as practicable with written notice.
- 13.4 Immediate Action. Unless, in Interconnection Customer's reasonable judgment, immediate action is required, Interconnection Customer shall obtain the consent of Transmission Provider, such consent to not be unreasonably withheld, prior to performing any manual switching operations at the Large Generating Facility or Interconnection Customer's Interconnection Facilities in response to an Emergency Condition either declared by Transmission Provider or otherwise regarding the Transmission System.

13.5 Transmission Provider Authority.

13.5.1 General. Transmission Provider may take whatever actions or inactions with regard to the Transmission System or Transmission Provider's Interconnection Facilities it deems necessary during an Emergency Condition in order to (i) preserve public health and safety, (ii) preserve the reliability of the Transmission System or Transmission Provider's Interconnection Facilities, (iii) limit or prevent damage, and (iv) expedite restoration of service.

Transmission Provider shall use Reasonable Efforts to minimize the effect of such actions or inactions on the Large Generating Facility or

Interconnection Customer's Interconnection Facilities. Transmission Provider may, on the basis of technical considerations. require the Large Generating Facility to mitigate an Emergency Condition by taking actions necessary and limited in scope to remedy the Emergency Condition, including, but not limited to, directing Interconnection Customer to shut-down, startup, increase or decrease the real or reactive power output of the Large Generating Facility; implementing a reduction or disconnection pursuant to Article 13.5.2; directing Interconnection Customer to assist with blackstart (if available) or restoration efforts; or altering the outage schedules of the Large Generating Facility and Interconnection Customer's Interconnection Facilities. Interconnection Customer shall comply with all of Transmission Provider's operating instructions concerning Large Generating Facility real power and reactive power output within the manufacturer's design limitations of the Large Generating Facility's equipment that is in service and physically available for operation at the time, in compliance with Applicable Laws and Regulations.

- 13.5.2 Reduction and Disconnection. Transmission Provider may reduce Interconnection Service or disconnect the Large Generating Facility or Interconnection Customer's Interconnection Facilities, when such, reduction or disconnection is necessary under Good Utility Practice due to Emergency Conditions. These rights are separate and distinct from any right of curtailment of Transmission Provider pursuant to Transmission Provider's Tariff. When Transmission Provider can schedule the reduction or disconnection in advance. Transmission Provider shall notify Interconnection Customer of the reasons, timing and expected duration of the reduction or disconnection. Transmission Provider shall coordinate with Interconnection Customer using Good Utility Practice to schedule the reduction or disconnection during periods of least impact to Interconnection Customer and Transmission Provider. Any reduction or disconnection shall continue only for so long as reasonably necessary under Good Utility Practice. The Parties shall cooperate with each other to restore the Large Generating Facility, the Interconnection Facilities, and the Transmission System to their normal operating state as soon as practicable consistent with Good Utility Practice.
- **13.6 Interconnection Customer Authority**. Consistent with Good Utility Practice and the LGIA and the LGIP, Interconnection Customer may take actions or inactions with regard to the Large Generating Facility or Interconnection Customer's Interconnection Facilities during an Emergency Condition in order to (i) preserve

public health and safety, (ii) preserve the reliability of the Large Generating Facility or Interconnection Customer's Interconnection Facilities, (iii) limit or prevent damage, and (iv) expedite restoration of service. Interconnection Customer shall use Reasonable Efforts to minimize the effect of such actions or inactions on the Transmission System and Transmission Provider's Interconnection Facilities. Transmission Provider shall use Reasonable Efforts to assist Interconnection Customer in such actions.

13.7 Limited Liability. Except as otherwise provided in Article 11.6.1 of this LGIA, neither Party shall be liable to the other for any action it takes in responding to an Emergency Condition so long as such action is made in good faith and is consistent with Good Utility Practice.

Article 14. Regulatory Requirements and Governing Law

14.1 Regulatory Requirements. Each Party's obligations under this LGIA shall be subject to its receipt of any required approval or certificate from one or more Governmental Authorities in the form and substance satisfactory to the applying Party, or the Party making any required filings with, or providing notice to, such Governmental Authorities, and the expiration of any time period associated therewith. Each Party shall in good faith seek and use its Reasonable Efforts to obtain such other approvals. Nothing in this LGIA shall require Interconnection Customer to take any action that could result in its inability to obtain, or its loss of, status or exemption under the Federal Power Act, the Public Utility Holding Company Act of 1935, as amended, or the Public Utility Regulatory Policies Act of 1978.

14.2 Governing Law.

- 14.2.1 The validity, interpretation and performance of this LGIA and each of its provisions shall be governed by the laws of the state where the Point of Interconnection is located, without regard to its conflicts of law principles.
- 14.2.2 This LGIA is subject to all Applicable Laws and Regulations.
- 14.3.3 Each Party expressly reserves the right to seek changes in, appeal, or otherwise contest any laws, orders, rules, or regulations of a Governmental Authority.

Article 15. Notices.

15.1 General. Unless otherwise provided in this LGIA, any notice, demand or request

required or permitted to be given by either Party to the other and any instrument required or permitted to be tendered or delivered by either Party in writing to the other shall be effective when delivered and may be so given, tendered or delivered, by recognized national courier, or by depositing the same with the United States Postal Service with postage prepaid, for delivery by certified or registered mail, addressed to the Party, or personally delivered to the Party, at the address set out in Appendix F, Addresses for Delivery of Notices and Billings.

Either Party may change the notice information in this LGIA by giving five (5) Business Days written notice prior to the effective date of the change.

- **15.2 Billings and Payments**. Billings and payments shall be sent to the addresses set out in Appendix F.
- **15.3 Alternative Forms of Notice**. Any notice or request required or permitted to be given by a Party to the other and not required by this Agreement to be given in writing may be so given by telephone, facsimile or email to the telephone numbers and email addresses set out in Appendix F.
- **15.4** Operations and Maintenance Notice. Each Party shall notify the other Party in writing of the identity of the person(s) that it designates as the point(s) of contact with respect to the implementation of Articles 9 and 10.

Article 16. Force Majeure

- 16.1 Force Majeure.
 - **16.1.1** Economic hardship is not considered a Force Majeure event.
 - Neither Party shall be considered to be in Default with respect to any obligation hereunder, (including obligations under Article 4), other than the obligation to pay money when due, if prevented from fulfilling such obligation by Force Majeure. A Party unable to fulfill any obligation hereunder (other than an obligation to pay money when due) by reason of Force Majeure shall give notice and the full particulars of such Force Majeure to the other Party in writing or by telephone as soon as reasonably possible after the occurrence of the cause relied upon. Telephone notices given pursuant to this article shall be confirmed in writing as soon as reasonably possible and shall specifically state full particulars of the Force Majeure, the time and date when the Force Majeure occurred and when the Force Majeure is reasonably expected to cease. The Party

affected shall exercise due diligence to remove such disability with reasonable dispatch, but shall not be required to accede or agree to any provision not satisfactory to it in order to settle and terminate a strike or other labor disturbance.

Article 17. Default

17.1 Default

- Oeneral. No Default shall exist where such failure to discharge an obligation (other than the payment of money) is the result of Force Majeure as defined in this LGIA or the result of an act of omission of the other Party. Upon a Breach, the non-breaching Party shall give written notice of such Breach to the breaching Party. Except as provided in Article 17.1.2, the breaching Party shall have thirty (30) Calendar Days from receipt of the Default notice within which to cure such Breach; provided however, if such Breach is not capable of cure within thirty (30) Calendar Days, the breaching Party shall commence such cure within thirty (30) Calendar Days after notice and continuously and diligently complete such cure within ninety (90) Calendar Days from receipt of the Default notice; and, if cured within such time, the Breach specified in such notice shall cease to exist.
- **Right to Terminate.** If a Breach is not cured as provided in this article, or if a Breach is not capable of being cured within the period provided for herein, the non-breaching Party shall have the right to declare a Default and terminate this LGIA by written notice at any time until cure occurs, and be relieved of any further obligation hereunder and, whether or not that Party terminates this LGIA, to recover from the breaching Party all amounts due hereunder, plus all other damages and remedies to which it is entitled at law or in equity. The provisions of this article will survive termination of this LGIA.

Article 18. Indemnity, Consequential Damages and Insurance

18.1 Indemnity. The Parties shall at all times indemnify, defend, and hold the other Party harmless from, any and all damages, losses, claims, including claims and actions relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's action or inactions of its obligations under this LGIA on behalf of the Indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the indemnified Party.

- **18.1.1 Indemnified Person**. If an Indemnified Person is entitled to indemnification under this Article 18 as a result of a claim by a third party, and the Indemnifying Party fails, after notice and reasonable opportunity to proceed under Article 18.1, to assume the defense of such claim, such Indemnified Person may at the expense of the Indemnifying Party contest, settle or consent to the entry of any judgment with respect to, or pay in full, such claim.
- **18.1.2** Indemnifying Party. If an Indemnifying Party is obligated to indemnify and hold any Indemnified Person harmless under this Article 18, the amount owing to the Indemnified Person shall be the amount of such Indemnified Person's actual Loss, net of any insurance or other recovery.
- 18.1.3 Indemnity Procedures. Promptly after receipt by an Indemnified Person of any claim or notice of the commencement of any action or administrative or legal proceeding or investigation as to which the indemnity provided for in Article 18.1 may apply, the Indemnified Person shall notify the Indemnifying Party of such fact. Any failure of or delay in such notification shall not affect a Party's indemnification obligation unless such failure or delay is materially prejudicial to the Indemnifying Party.

The Indemnifying Party shall have the right to assume the defense thereof with counsel designated by such Indemnifying Party and reasonably satisfactory to the Indemnified Person. If the defendants in any such action include one or more Indemnified Persons and the Indemnifying Party and if the Indemnified Person reasonably concludes that there may be legal defenses available to it and/or other Indemnified Persons which are different from or additional to those available to the Indemnifying Party, the Indemnified Person shall have the right to select separate counsel to assert such legal defenses and to otherwise participate in the defense of such action on its own behalf. In such instances, the Indemnifying Party shall only be required to pay the fees and expenses of one additional attorney to represent an Indemnified Person or Indemnified Persons having such differing or additional legal defenses.

The Indemnified Person shall be entitled, at its expense, to participate in any such action, suit or proceeding, the defense of which has been assumed by the Indemnifying Party. Notwithstanding the foregoing, the Indemnifying Party (i) shall not be entitled to assume and control the defense of any such action, suit or proceedings if and to the extent

that, in the opinion of the Indemnified Person and its counsel, such action, suit or proceeding involves the potential imposition of criminal liability on the Indemnified Person, or there exists a conflict or adversity of interest between the Indemnified Person and the Indemnifying Party, in such event the Indemnifying Party shall pay the reasonable expenses of the Indemnified Person, and (ii) shall not settle or consent to the entry of any judgment in any action, suit or proceeding without the consent of the Indemnified Person, which shall not be reasonably withheld, conditioned or delayed.

- 18.2 Consequential Damages. Other than the Liquidated Damages heretofore described, in no event shall either Party be liable under any provision of this LGIA for any losses, damages, costs or expenses for any special, indirect, incidental, consequential, or punitive damages, including but not limited to loss of profit or revenue, loss of the use of equipment, cost of capital, cost of temporary equipment or services, whether based in whole or in part in contract, in tort, including negligence, strict liability, or any other theory of liability; provided, however, that damages for which a Party may be liable to the other Party under another agreement will not be considered to be special, indirect, incidental, or consequential damages hereunder.
- **18.3 Insurance**. Each party shall, at its own expense, maintain in force throughout the period of this LGIA, and until released by the other Party, the following minimum insurance coverages, with insurers authorized to do business in the state where the Point of Interconnection is located:
 - 18.3.1 Employers' Liability and Workers' Compensation Insurance providing statutory benefits in accordance with the laws and regulations of the state in which the Point of Interconnection is located.
 - 18.3.2 Commercial General Liability Insurance including premises and operations, personal injury, broad form property damage, broad form blanket contractual liability coverage (including coverage for the contractual indemnification) products and completed operations coverage, coverage for explosion, collapse and underground hazards, independent contractors coverage, coverage for pollution to the extent normally available and punitive damages to the extent normally available and a cross liability endorsement, with minimum limits of One Million Dollars (\$1,000,000) per occurrence/One Million Dollars (\$1,000,000) aggregate combined single limit for personal injury, bodily injury, including death and property damage.

- 18.3.3 Comprehensive Automobile Liability Insurance for coverage of owned and non-owned and hired vehicles, trailers or semi-trailers designed for travel on public roads, with a minimum, combined single limit of One Million Dollars (\$1,000,000) per occurrence for bodily injury, including death, and property damage.
- 18.3.4 Excess Public Liability Insurance over and above the Employers' Liability Commercial General Liability and Comprehensive Automobile Liability Insurance coverage, with a minimum combined single limit of Twenty Million Dollars (\$20,000,000) per occurrence/Twenty Million Dollars (\$20,000,000) aggregate.
- 18.3.5 The Commercial General Liability Insurance, Comprehensive Automobile Insurance and Excess Public Liability Insurance policies shall name the other Party, its parent, associated and Affiliate companies and their respective directors, officers, agents, servants and employees ("Other Party Group") as additional insured. All policies shall contain provisions whereby the insurers waive all rights of subrogation in accordance with the provisions of this LGIA against the Other Party Group and provide thirty (30) Calendar Days advance written notice to the Other Party Group prior to anniversary date of cancellation or any material change in coverage or condition.
- 18.3.6 The Commercial General Liability Insurance, Comprehensive Automobile Liability Insurance and Excess Public Liability Insurance policies shall contain provisions that specify that the policies are primary and shall apply to such extent without consideration for other policies separately carried and shall state that each insured is provided coverage as though a separate policy had been issued to each, except the insurer's liability shall not be increased beyond the amount for which the insurer would have been liable had only one insured been covered. Each Party shall be responsible for its respective deductibles or retentions.
- 18.3.7 The Commercial General Liability Insurance, Comprehensive Automobile Liability Insurance and Excess Public Liability Insurance policies, if written on a Claims First Made Basis, shall be maintained in full force and effect for two (2) years after termination of this LGIA, which coverage may be in the form of tail coverage or extended reporting period coverage if agreed by the Parties.
- 18.3.8 The requirements contained herein as to the types and limits of all insurance to be maintained by the Parties are not intended to and shall

not in any manner, limit or qualify the liabilities and obligations assumed by the Parties under this LGIA.

- 18.3.9 Within ten (10) Calendar Days following execution of this LGIA, and as soon as practicable after the end of each fiscal year or at the renewal of the insurance policy and in any event within ninety (90) Calendar Days thereafter, each Party shall provide certification of all insurance required in this LGIA, executed by each insurer or by an authorized representative of each insurer.
- In addition to the foregoing, each Party may self-insure to meet the minimum insurance requirements of Articles 18.3.2 through 18.3.8 to the extent it maintains a self-insurance program; provided that, such Party's self-insurance program meets the minimum insurance requirements of Articles 18.3.2 through 18.3.8. In the event that a Party is permitted to self-insure pursuant to this article, it shall certify to the other Party with a letter of self-insurance that it meets the requirements to self-insure and that its self-insurance program meets the minimum insurance requirements in a manner consistent with that specified in Article 18.3.9.
- 18.3.11 The Parties agree to report to each other in writing as soon as practical all accidents or occurrences resulting in injuries to any person, including death, and any property damage arising out of this LGIA.

Article 19. Assignment

Assignment. This LGIA may be assigned by either Party only with the written 19.1 consent of the other; provided that either Party may assign this LGIA without the consent of the other Party to any Affiliate of the assigning Party with an equal or greater credit rating and with the legal authority and operational ability to satisfy the obligations of the assigning Party under this LGIA; and provided further that Interconnection Customer shall have the right to assign this LGIA, without the consent of Transmission Provider, for collateral security purposes to aid in providing financing for the Large Generating Facility, provided that Interconnection Customer will promptly notify Transmission Provider of any such assignment. Any financing arrangement entered into by Interconnection Customer pursuant to this article will provide that prior to or upon the exercise of the secured party's, trustee's or mortgagee's assignment rights pursuant to said arrangement, the secured creditor, the trustee or mortgagee will notify Transmission Provider of the date and particulars of any such exercise of assignment right(s), including providing the Transmission Provider with proof that it meets the requirements of Articles 11.5 and 18.3. Any attempted assignment that violates this article is void

and ineffective. Any assignment under this LGIA shall not relieve a Party of its obligations, nor shall a Party's obligations be enlarged, in whole or in part, by reason thereof. Where required, consent to assignment will not be unreasonably withheld, conditioned or delayed.

Article 20. Severability

20.1 Severability. If any provision in this LGIA is finally determined to be invalid, void or unenforceable by any court or other Governmental Authority having jurisdiction, such determination shall not invalidate, void or make unenforceable any other provision, agreement or covenant of this LGIA; provided that if Interconnection Customer (or any third party, but only if such third party is not acting at the direction of Transmission Provider) seeks and obtains such a final determination with respect to any provision of the Alternate Option (Article 5.1.2), or the Negotiated Option (Article 5.1.4), then none of these provisions shall thereafter have any force or effect and the Parties' rights and obligations shall be governed solely by the Standard Option (Article 5.1.1).

Article 21. Comparability

21.1 Comparability. The Parties will comply with all applicable comparability and code of conduct laws, rules and regulations, as amended from time to time.

Article 22. Confidentiality

22.1 Confidentiality. Confidential Information shall include, without limitation, all information relating to a Party's technology, research and development, business affairs, and pricing, and any information supplied by either of the Parties to the other prior to the execution of this LGIA.

Information is Confidential Information only if it is clearly designated or marked in writing as confidential on the face of the document, or, if the information is conveyed orally or by inspection, if the Party providing the information orally informs the Party receiving the information that the information is confidential.

If requested by either Party, the other Party shall provide in writing, the basis for asserting that the information referred to in this Article 22 warrants confidential treatment, and the requesting Party may disclose such writing to the appropriate Governmental Authority. Each Party shall be responsible for the costs associated with affording confidential treatment to its information.

The release of Confidential Information shall be subject to Applicable Laws and Regulations and Applicable Reliability Standards.

- **Term**. During the term of this LGIA, and for a period of three (3) years after the expiration or termination of this LGIA, except as otherwise provided in this Article 22, each Party shall hold in confidence and shall not disclose to any person Confidential Information.
- 22.1.2 **Scope**. Confidential Information shall not include information that the receiving Party can demonstrate: (1) is generally available to the public other than as a result of a disclosure by the receiving Party; (2) was in the lawful possession of the receiving Party on a non-confidential basis before receiving it from the disclosing Party; (3) was supplied to the receiving Party without restriction by a third party, who, to the knowledge of the receiving Party after due inquiry, was under no obligation to the disclosing Party to keep such information confidential; (4) was independently developed by the receiving Party without reference to Confidential Information of the disclosing Party; (5) is, or becomes, publicly known, through no wrongful act or omission of the receiving Party or Breach of this LGIA; or (6) is required, in accordance with Article 22.1.7 of the LGIA, Order of Disclosure, to be disclosed by any Governmental Authority or is otherwise required to be disclosed by law or subpoena, or is necessary in any legal proceeding establishing rights and obligations under this LGIA. Information designated as Confidential Information will no longer be deemed confidential if the Party that designated the information as confidential notifies the other Party that it no longer is confidential.
- 22.1.3 Release of Confidential Information. Neither Party shall release or disclose Confidential Information to any other person, except to its Affiliates (limited by the Standards of Conduct requirements), subcontractors, employees, consultants, or to parties who may be or considering providing financing to or equity participation with Interconnection Customer, or to potential purchasers or assignees of Interconnection Customer, on a need-to-know basis in connection with this LGIA, unless such person has first been advised of the confidentiality provisions of this Article 22 and has agreed to comply with such provisions. Notwithstanding the foregoing, a Party providing Confidential Information to any person shall remain primarily responsible for any release

of Confidential Information in contravention of this Article 22.

- **Rights**. Each Party retains all rights, title, and interest in the Confidential Information that each Party discloses to the other Party. The disclosure by each Party to the other Party of Confidential Information shall not be deemed a waiver by either Party or any other person or entity of the right to protect the Confidential Information from public disclosure.
- No Warranties. By providing Confidential Information, neither Party makes any warranties or representations as to its accuracy or completeness. In addition, by supplying Confidential Information, neither Party obligates itself to provide any particular information or Confidential Information to the other Party nor to enter into any further agreements or proceed with any other relationship or joint venture.
- 22.1.6 Standard of Care. Each Party shall use at least the same standard of care to protect Confidential Information it receives as it uses to protect its own Confidential Information from unauthorized disclosure, publication or dissemination. Each Party may use Confidential Information solely to fulfill its obligations to the other Party under this LGIA or its regulatory requirements.
- 22.1.7 **Order of Disclosure**. If a court or a Government Authority or entity with the right, power, and apparent authority to do so requests or requires either Party, by subpoena, oral deposition, interrogatories, requests for production of documents, administrative order, or otherwise, to disclose Confidential Information, that Party shall provide the other Party with prompt notice of such request(s) or requirement(s) so that the other Party may seek an appropriate protective order or waive compliance with the terms of this LGIA. Notwithstanding the absence of a protective order or waiver, the Party may disclose such Confidential Information which, in the opinion of its counsel, the Party is legally compelled to disclose. Each Party will use Reasonable Efforts to obtain reliable assurance that confidential treatment will be accorded any Confidential Information so furnished.
- **Termination of Agreement**. Upon termination of this LGIA for any reason, each Party shall, within ten (10) Calendar Days

of receipt of a written request from the other Party, use Reasonable Efforts to destroy, erase, or delete (with such destruction, erasure, and deletion certified in writing to the other Party) or return to the other Party, without retaining copies thereof, any and all written or electronic Confidential Information received from the other Party.

- 22.1.9 **Remedies**. The Parties agree that monetary damages would be inadequate to compensate a Party for the other Party's Breach of its obligations under this Article 22. Each Party accordingly agrees that the other Party shall be entitled to equitable relief, by way of injunction or otherwise, if the first Party Breaches or threatens to Breach its obligations under this Article 22, which equitable relief shall be granted without bond or proof of damages, and the receiving Party shall not plead in defense that there would be an adequate remedy at law. Such remedy shall not be deemed an exclusive remedy for the Breach of this Article 22, but shall be in addition to all other remedies available at law or in equity. The Parties further acknowledge and agree that the covenants contained herein are necessary for the protection of legitimate business interests and are reasonable in scope. No Party, however, shall be liable for indirect, incidental, or consequential or punitive damages of any nature or kind resulting from or arising in connection with this Article 22.
- 22,1,10 Disclosure to FERC, its Staff, or a State. Notwithstanding anything in this Article 22 to the contrary, and pursuant to 18 CFR section 1b.20, if FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to this LGIA, the Party shall provide the requested information to FERC or its staff, within the time provided for in the request for information. In providing the information to FERC or its staff, the Party must, consistent with 18 CFR section 388.112, request that the information be treated as confidential and non-public by FERC and its staff and that the information be withheld from public disclosure. Parties are prohibited from notifying the other Party to this LGIA prior to the release of the Confidential Information to FERC or its staff. The Party shall notify the other Party to the LGIA when it is notified by FERC or its staff that a request to release Confidential Information has been received by FERC,

at which time either of the Parties may respond before such information would be made public, pursuant to 18 CFR section 388.112. Requests from a state regulatory body conducting a confidential investigation shall be treated in a similar manner if consistent with the applicable state rules and regulations.

22.1.11

Subject to the exception in Article 22.1.10, any information that a Party claims is competitively sensitive, commercial or financial information under this LGIA ("Confidential Information") shall not be disclosed by the other Party to any person not employed or retained by the other Party, except to the extent disclosure is (i) required by law; (ii) reasonably deemed by the disclosing Party to be required to be disclosed in connection with a dispute between or among the Parties, or the defense of litigation or dispute; (iii) otherwise permitted by consent of the other Party, such consent not to be unreasonably withheld; or (iv) necessary to fulfill its obligations under this LGIA or as a transmission service provider or a Control Area operator including disclosing the Confidential Information to an RTO or ISO or to a regional or national reliability organization. The Party asserting confidentiality shall notify the other Party in writing of the information it claims is confidential. Prior to any disclosures of the other Party's Confidential Information under this subparagraph, or if any third party or Governmental Authority makes any request or demand for any of the information described in this subparagraph, the disclosing Party agrees to promptly notify the other Party in writing and agrees to assert confidentiality and cooperate with the other Party in seeking to protect the Confidential Information from public disclosure by confidentiality agreement, protective order or other reasonable measures.

Article 23. Environmental Releases

23.1 Each Party shall notify the other Party, first orally and then in writing, of the release of any Hazardous Substances, any asbestos or lead abatement activities, or any type of remediation activities related to the Large Generating Facility or the Interconnection Facilities, each of which may reasonably be expected to affect the other Party. The notifying Party shall: (i) provide the notice as soon as practicable, provided such Party makes a good faith effort to provide the notice no later than twenty-four hours after such Party becomes aware of the occurrence; and (ii) promptly furnish to the other Party copies of any publicly available reports filed

with any Governmental Authorities addressing such events.

Article 24. Information Requirements

- **24.1 Information Acquisition**. Transmission Provider and Interconnection Customer shall submit specific information regarding the electrical characteristics of their respective facilities to each other as described below and in accordance with Applicable Reliability Standards.
- 24.2 Information Submission by Transmission Provider. The initial information submission by Transmission Provider shall occur no later than one hundred eighty (180) Calendar Days prior to Trial Operation and shall include Transmission System information necessary to allow Interconnection Customer to select equipment and meet any system protection and stability requirements, unless otherwise agreed to by the Parties. On a monthly basis Transmission Provider shall provide Interconnection Customer a status report on the construction and installation of Transmission Provider's Interconnection Facilities and Network Upgrades, including, but not limited to, the following information: (1) progress to date; (2) a description of the activities since the last report (3) a description of the action items for the next period; and (4) the delivery status of equipment ordered.
- 24.3 Updated Information Submission by Interconnection Customer. The updated information submission by Interconnection Customer, including manufacturer information, shall occur no later than one hundred eighty (180) Calendar Days prior to the Trial Operation. Interconnection Customer shall submit a completed copy of the Large Generating Facility data requirements contained in Appendix 1 to the LGIP. It shall also include any additional information provided to Transmission Provider for the Definitive Interconnection System Impact Study and Interconnection Facilities Study. Information in this submission shall be the most current Large Generating Facility design or expected performance data. Information submitted for stability models shall be compatible with Transmission Provider standard models. If there is no compatible model, Interconnection Customer will work with a consultant mutually agreed to by the Parties to develop and supply a standard model and associated information.

If Interconnection Customer's data is materially different from what was originally provided to Transmission Provider pursuant to the Interconnection Study Agreement between Transmission Provider and Interconnection Customer, then Transmission Provider will conduct appropriate studies to determine the impact on Transmission Provider Transmission System based on the actual data submitted pursuant to this Article 24.3. The Interconnection Customer shall not begin Trial Operation until such studies are completed.

24.4 Information Supplementation. Prior to the Operation Date, the Parties shall supplement their information submissions described above in this Article 24 with any and all "as-built" Large Generating Facility information or "as-tested" performance information that differs from the initial submissions or, alternatively, written confirmation that no such differences exist. The Interconnection Customer shall conduct tests on the Large Generating Facility as required by Good Utility Practice such as an open circuit "step voltage" test on the Large Generating Facility to verify proper operation of the Large Generating Facility's automatic voltage regulator.

Unless otherwise agreed, the test conditions shall include: (1) Large Generating Facility at synchronous speed; (2) automatic voltage regulator on and in voltage control mode; and (3) a five percent change in Large Generating Facility terminal voltage initiated by a change in the voltage regulators reference voltage. Interconnection Customer shall provide validated test recordings showing the responses of Large Generating Facility terminal and field voltages. In the event that direct recordings of these voltages is impractical, recordings of other voltages or currents that mirror the response of the Large Generating Facility's terminal or field voltage are acceptable if information necessary to translate these alternate quantities to actual Large Generating Facility terminal or field voltages is provided. Large Generating Facility testing shall be conducted and results provided to Transmission Provider for each individual generating unit in a station. Subsequent to the Operation Date, Interconnection Customer shall provide Transmission Provider any information changes due to equipment replacement, repair, or adjustment. Transmission Provider shall provide Interconnection Customer any information changes due to equipment replacement, repair or adjustment in the directly connected substation or any adjacent Transmission Provider-owned substation that may affect Interconnection Customer's Interconnection Facilities equipment ratings, protection or operating requirements. The Parties shall provide such information no later than thirty (30) Calendar Days after the date of the equipment replacement, repair or adjustment.

Article 25. Information Access and Audit Rights

- 25.1 Information Access. Each Party (the "disclosing Party") shall make available to the other Party information that is in the possession of the disclosing Party and is necessary in order for the other Party to: (i) verify the costs incurred by the disclosing Party for which the other Party is responsible under this LGIA; and (ii) carry out its obligations and responsibilities under this LGIA. The Parties shall not use such information for purposes other than those set forth in this Article 25.1 and to enforce their rights under this LGIA.
- 25.2 Reporting of Non-Force Majeure Events. Each Party (the "notifying Party") shall

notify the other Party when the notifying Party becomes aware of its inability to comply with the provisions of this LGIA for a reason other than a Force Majeure event. The Parties agree to cooperate with each other and provide necessary information regarding such inability to comply, including the date, duration, reason for the inability to comply, and corrective actions taken or planned to be taken with respect to such inability to comply. Notwithstanding the foregoing, notification, cooperation or information provided under this article shall not entitle the Party receiving such notification to allege a cause for anticipatory breach of this LGIA.

25.3 Audit Rights. Subject to the requirements of confidentiality under Article 22 of this LGIA, each Party shall have the right, during normal business hours, and upon prior reasonable notice to the other Party, to audit at its own expense the other Party's accounts and records pertaining to either Party's performance or either Party's satisfaction of obligations under this LGIA. Such audit rights shall include audits of the other Party's costs, calculation of invoiced amounts, Transmission Provider's efforts to allocate responsibility for the provision of reactive support to the Transmission System, Transmission Provider's efforts to allocate responsibility for interruption or reduction of generation on the Transmission System, and each Party's actions in an Emergency Condition. Any audit authorized by this article shall be performed at the offices where such accounts and records are maintained and shall be limited to those portions of such accounts and records that relate to each Party's performance and satisfaction of obligations under this LGIA. Each Party shall keep such accounts and records for a period equivalent to the audit rights periods described in Article 25.4.

25.4 Audit Rights Periods.

- Audit Rights Period for Construction-Related Accounts and Records. Accounts and records related to the design, engineering, procurement, and construction of Transmission Provider's Interconnection Facilities and Network Upgrades shall be subject to audit for a period of twenty-four months following Transmission Provider's issuance of a final invoice in accordance with Article 12.2.
- Accounts and records related to either Party's performance or satisfaction of all obligations under this LGIA other than those described in Article 25.4.1 shall be subject to audit as follows: (i) for an audit relating to cost obligations, the applicable audit rights period shall be twenty-four months after the auditing Party's receipt of an invoice giving rise to such cost obligations; and (ii) for an audit relating to all other obligations, the applicable audit rights period shall be twenty-four months after the event for which the audit is sought.

25.5 Audit Results. If an audit by a Party determines that an overpayment or an underpayment has occurred, a notice of such overpayment or underpayment shall be given to the other Party together with those records from the audit which support such determination.

Article 26. Subcontractors

- **26.1 General**. Nothing in this LGIA shall prevent a Party from utilizing the services of any subcontractor as it deems appropriate to perform its obligations under this LGIA; provided, however, that each Party shall require its subcontractors to comply with all applicable terms and conditions of this LGIA in providing such services and each Party shall remain primarily liable to the other Party for the performance of such subcontractor.
- **26.2 Responsibility of Principal**. The creation of any subcontract relationship shall not relieve the hiring Party of any of its obligations under this LGIA. The hiring Party shall be fully responsible to the other Party for the acts or omissions of any subcontractor the hiring Party hires as if no subcontract had been made; provided, however, that in no event shall Transmission Provider be liable for the actions or inactions of Interconnection Customer or its subcontractors with respect to obligations of Interconnection Customer under Article 5 of this LGIA. Any applicable obligation imposed by this LGIA upon the hiring Party shall be equally binding upon, and shall be construed as having application to, any subcontractor of such Party.
- **26.3 No Limitation by Insurance**. The obligations under this Article 26 will not be limited in any way by any limitation of subcontractor's insurance.

Article 27. Disputes

27.1 Submission. In the event either Party has a dispute, or asserts a claim, that arises out of or in connection with this LGIA or its performance, such Party (the "disputing Party") shall provide the other Party with written notice of the dispute or claim ("Notice of Dispute"). Such dispute or claim shall be referred to a designated senior representative of each Party for resolution on an informal basis as promptly as practicable after receipt of the Notice of Dispute by the other Party. In the event the designated representatives are unable to resolve the claim or dispute through unassisted or assisted negotiations within thirty (30) Calendar Days of the other Party's receipt of the Notice of Dispute, such claim or dispute may, upon mutual agreement of the Parties, be submitted to arbitration and resolved in accordance with the arbitration procedures set forth below. In the event the Parties do not agree to submit such claim or dispute to arbitration, each Party may exercise

whatever rights and remedies it may have in equity or at law consistent with the terms of this LGIA.

- 27.2 **External Arbitration Procedures.** Any arbitration initiated under this LGIA shall be conducted before a single neutral arbitrator appointed by the Parties. If the Parties fail to agree upon a single arbitrator within ten (10) Calendar Days of the submission of the dispute to arbitration, each Party shall choose one arbitrator who shall sit on a three-member arbitration panel. The two arbitrators so chosen shall within twenty (20) Calendar Days select a third arbitrator to chair the arbitration panel. In either case, the arbitrators shall be knowledgeable in electric utility matters, including electric transmission and bulk power issues, and shall not have any current or past substantial business or financial relationships with any party to the arbitration (except prior arbitration). The arbitrator(s) shall provide each of the Parties an opportunity to be heard and, except as otherwise provided herein, shall conduct the arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("Arbitration Rules") and any applicable FERC regulations or RTO rules; provided, however, in the event of a conflict between the Arbitration Rules and the terms of this Article 27, the terms of this Article 27 shall prevail.
- 27.3 Arbitration Decisions. Unless otherwise agreed by the Parties, the arbitrator(s) shall render a decision within ninety (90) Calendar Days of appointment and shall notify the Parties in writing of such decision and the reasons therefor. The arbitrator(s) shall be authorized only to interpret and apply the provisions of this LGIA and shall have no power to modify or change any provision of this Agreement in any manner. The decision of the arbitrator(s) shall be final and binding upon the Parties, and judgment on the award may be entered in any court having jurisdiction. The decision of the arbitrator(s) may be appealed solely on the grounds that the conduct of the arbitrator(s), or the decision itself, violated the standards set forth in the Federal Arbitration Act or the Administrative Dispute Resolution Act. The final decision of the arbitrator must also be filed with FERC if it affects jurisdictional rates, terms and conditions of service, Interconnection Facilities, or Network Upgrades.
- **27.4 Costs**. Each Party shall be responsible for its own costs incurred during the arbitration process and for the following costs, if applicable: (1) the cost of the arbitrator chosen by the Party to sit on the three member panel and one half of the cost of the third arbitrator chosen; or (2) one half the cost of the single arbitrator jointly chosen by the Parties.

Article 28. Representations, Warranties, and Covenants

28.1 General. Each Party makes the following representations, warranties and

covenants:

- **28.1.1 Good Standing**. Such Party is duly organized, validly existing and in good standing under the laws of the state in which it is organized, formed, or incorporated, as applicable; that it is qualified to do business in the state or states in which the Large Generating Facility, Interconnection Facilities and Network Upgrades owned by such Party, as applicable, are located; and that it has the corporate power and authority to own its properties, to carry on its business as now being conducted and to enter into this LGIA and carry out the transactions contemplated hereby and perform and carry out all covenants and obligations on its part to be performed under and pursuant to this LGIA.
- **Authority**. Such Party has the right, power and authority to enter into this LGIA, to become a Party hereto and to perform its obligations hereunder. This LGIA is a legal, valid and binding obligation of such Party, enforceable against such Party in accordance with its terms, except as the enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting creditors' rights generally and by general equitable principles (regardless of whether enforceability is sought in a proceeding in equity or at law).
- **No Conflict**. The execution, delivery and performance of this LGIA does not violate or conflict with the organizational or formation documents, or bylaws or operating agreement, of such Party, or any judgment, license, permit, order, material agreement or instrument applicable to or binding upon such Party or any of its assets.
- **28.1.4 Consent and Approval**. Such Party has sought or obtained, or, in accordance with this LGIA will seek or obtain, each consent, approval, authorization, order, or acceptance by any Governmental Authority in connection with the execution, delivery and performance of this LGIA, and it will provide to any Governmental Authority notice of any actions under this LGIA that are required by Applicable Laws and Regulations.

Article 29. Joint Operating Committee

29.1 Joint Operating Committee. Except in the case of ISOs and RTOs, Transmission Provider shall constitute a Joint Operating Committee to coordinate operating and technical considerations of Interconnection Service. At least six (6) months prior

to the expected Initial Synchronization Date, Interconnection Customer and Transmission Provider shall each appoint one representative and one alternate to the Joint Operating Committee. Each Interconnection Customer shall notify Transmission Provider of its appointment in writing. Such appointments may be changed at any time by similar notice. The Joint Operating Committee shall meet as necessary, but not less than once each calendar year, to carry out the duties set forth herein. The Joint Operating Committee shall hold a meeting at the request of either Party, at a time and place agreed upon by the representatives. The Joint Operating Committee shall perform all of its duties consistent with the provisions of this LGIA. Each Party shall cooperate in providing to the Joint Operating Committee all information required in the performance of the Joint Operating Committee's duties. All decisions and agreements, if any, made by the Joint Operating Committee, shall be evidenced in writing. The duties of the Joint Operating Committee shall include the following:

- **29.1.1** Establish data requirements and operating record requirements.
- 29.1.2 Review the requirements, standards, and procedures for data acquisition equipment, protective equipment, and any other equipment or software.
- 29.1.3 Annually review the one (1) year forecast of maintenance and planned outage schedules of Transmission Provider's and Interconnection Customer's facilities at the Point of Interconnection.
- 29.1.4 Coordinate the scheduling of maintenance and planned outages on the Interconnection Facilities, the Large Generating Facility and other facilities that impact the normal operation of the interconnection of the Large Generating Facility to the Transmission System.
- 29.1.5 Ensure that information is being provided by each Party regarding equipment availability.
- 29.1.6 Perform such other duties as may be conferred upon it by mutual agreement of the Parties.

Article 30. Miscellaneous

- **30.1 Binding Effect**. This LGIA and the rights and obligations hereof, shall be binding upon and shall inure to the benefit of the successors and assigns of the Parties hereto.
- **30.2** Conflicts. In the event of a conflict between the body of this LGIA and any

attachment, appendices or exhibits hereto, the terms and provisions of the body of this LGIA shall prevail and be deemed the final intent of the Parties.

- 30.3 Rules of Interpretation. This LGIA, unless a clear contrary intention appears, shall be construed and interpreted as follows: (1) the singular number includes the plural number and vice versa; (2) reference to any person includes such person's successors and assigns but, in the case of a Party, only if such successors and assigns are permitted by this LGIA, and reference to a person in a particular capacity excludes such person in any other capacity or individually; (3) reference to any agreement (including this LGIA), document, instrument or tariff means such agreement, document, instrument, or tariff as amended or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof; (4) reference to any Applicable Laws and Regulations means such Applicable Laws and Regulations as amended, modified, codified, or reenacted, in whole or in part, and in effect from time to time, including, if applicable, rules and regulations promulgated thereunder; (5) unless expressly stated otherwise, reference to any Article, Section or Appendix means such Article of this LGIA or such Appendix to this LGIA, or such Section to the LGIP or such Appendix to the LGIP, as the case may be; (6) "hereunder", "hereof", "herein", "hereto" and words of similar import shall be deemed references to this LGIA as a whole and not to any particular Article or other provision hereof or thereof; (7) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding such term; and (8) relative to the determination of any period of time, "from" means "from and including", "to" means "to but excluding" and "through" means "through and including".
- 30.4 Entire Agreement. This LGIA, including all Appendices and Schedules attached hereto, constitutes the entire agreement between the Parties with reference to the subject matter hereof, and supersedes all prior and contemporaneous understandings or agreements, oral or written, between the Parties with respect to the subject matter of this LGIA. There are no other agreements, representations, warranties, or covenants which constitute any part of the consideration for, or any condition to, either Party's compliance with its obligations under this LGIA.
- **30.5 No Third Party Beneficiaries**. This LGIA is not intended to and does not create rights, remedies, or benefits of any character whatsoever in favor of any persons, corporations, associations, or entities other than the Parties, and the obligations herein assumed are solely for the use and benefit of the Parties, their successors in interest and, where permitted, their assigns.
- **30.6 Waiver**. The failure of a Party to this LGIA to insist, on any occasion, upon strict performance of any provision of this LGIA will not be considered a waiver of any obligation, right, or duty of, or imposed upon, such Party.

Any waiver at any time by either Party of its rights with respect to this LGIA shall not be deemed a continuing waiver or a waiver with respect to any other failure to comply with any other obligation, right, duty of this LGIA. Termination or Default of this LGIA for any reason by Interconnection Customer shall not constitute a waiver of Interconnection Customer's legal rights to obtain an interconnection from Transmission Provider. Any waiver of this LGIA shall, if requested, be provided in writing.

- **30.7 Headings**. The descriptive headings of the various Articles of this LGIA have been inserted for convenience of reference only and are of no significance in the interpretation or construction of this LGIA.
- **30.8 Multiple Counterparts**. This LGIA may be executed in two or more counterparts, each of which is deemed an original but all constitute one and the same instrument.
- **30.9 Amendment**. The Parties may by mutual agreement amend this LGIA by a written instrument duly executed by the Parties.
- **30.10 Modification by the Parties**. The Parties may by mutual agreement amend the Appendices to this LGIA by a written instrument duly executed by the Parties. Such amendment shall become effective and a part of this LGIA upon satisfaction of all Applicable Laws and Regulations.
- 30.11 Reservation of Rights. Transmission Provider shall have the right to make a unilateral filing with FERC to modify this LGIA with respect to any rates, terms and conditions, charges, classifications of service, rule or regulation under section 205 or any other applicable provision of the Federal Power Act and FERC's rules and regulations thereunder, and Interconnection Customer shall have the right to make a unilateral filing with FERC to modify this LGIA pursuant to section 206 or any other applicable provision of the Federal Power Act and FERC's rules and regulations thereunder; provided that each Party shall have the right to protest any such filing by the other Party and to participate fully in any proceeding before FERC in which such modifications may be considered. Nothing in this LGIA shall limit the rights of the Parties or of FERC under sections 205 or 206 of the Federal Power Act and FERC's rules and regulations thereunder, except to the extent that the Parties otherwise mutually agree as provided herein.
- **30.12 No Partnership**. This LGIA shall not be interpreted or construed to create an association, joint venture, agency relationship, or partnership between the Parties or to impose any partnership obligation or partnership liability upon either Party. Neither Party shall have any right, power or authority to enter into any agreement or undertaking for, or act on behalf of, or to act as or be an agent or representative of,

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

or to otherwise bind, the other Party.

IN WITNESS WHEREOF, the Parties have executed this LGIA in duplicate originals, each of which shall constitute and be an original effective Agreement between the Parties.

[Insert name of Transmission Provider or Transmission Owner, if applicable]		
By:	By:	
Title:	Title:	
Date:	Date:	
[Insert name of Interconnection Customer]		
By:		
Title:		
Date:		

Appendices: Interconnection Facilities, Network Upgrades and Distribution Upgrade

Appendices to LGIA

Appendix A Interconnection Facilities, Network Upgrades, and Distribution

Upgrades

Appendix B Milestones

Appendix C Interconnection Details

Appendix D Security Arrangements Details

Appendix E Commercial Operation Date

Appendix F Addresses for Delivery of Notices and Billings

Appendix G Interconnection Requirements for a Wind Generating Plant

Appendix A: Interconnection Facilities, Network Upgrades and Distribution **Upgrades**

Appendix A to LGIA

Interconnection Facilities, Network Upgrades and Distribution Upgrades

- **Interconnection Facilities:** 1.
 - (a) [insert Interconnection Customer's Interconnection Facilities]:
 - (b) [insert Transmission Provider's Interconnection Facilities]:
- **Network Upgrades:** 2.
 - (a) [insert Stand Alone Network Upgrades]:
 - (b) [insert Other Network Upgrades]:
- **Distribution Upgrades: 3.**

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Appendix B: Milestones

Appendix B to LGIA
Milestones

Appendix C: Interconnection Details

Appendix C to LGIA

Interconnection Details¹

¹ This Appendix C identifies the facilities required for an Interconnection Customer's Interconnection Request. If the Interconnection Customer's Interconnection Request was studied as part of a Queue Cluster, the costs for Network Upgrades associated with the specific Queue Cluster will be allocated pursuant to Section 4.2.2 of the Large Generator Interconnection Procedures in Attachment M of EPE's OATT.

Appendix D: Security Arrangements Details

Appendix D to LGIA

Security Arrangements Details

Infrastructure security of Transmission System equipment and operations and control hardware and software is essential to ensure day-to-day Transmission System reliability and operational security. FERC will expect all entities, including Transmission Providers, market participants, and Interconnection Customers interconnected to the Transmission System to comply with the Reliability Standards promulgated by FERC and administered by the North American Electric Reliability Corporation (NERC) and associated Regional Reliability Organizations (RRO). In EPE's service territory, the RRO is the Western Electricity Coordinating Council (WECC).

Appendix E: Commercial Operation Date

Appendix E to LGIA

Commercial Operation Date

This Appendix E is a part of the LGIA between Transmission Provider and Interconnection Customer.

[Da	te]
[Tra	ansmission Provider Address]
Re:	Large Generating Facility
Dea	r:
This 1	[Date] [Interconnection Customer] has completed Trial Operation of Unit No etter confirms that [Interconnection Customer] commenced Commercial of Unit No at the Large Generating Facility, effective as of [Date plus one)
	Thank you.
	[Signature]
	[Interconnection Customer Representative]

Appendix F: Addresses for Delivery of Notices and Billings

Appendix F to LGIA

	Addresses for Delivery of Notices and Billings
Notice	es:
	<u>Transmission Provider</u> :
	[To be supplied.]
	Interconnection Customer:
	[To be supplied.]
Billin	gs and Payments:
	<u>Transmission Provider</u> :
	[To be supplied.]
	<u>Interconnection Customer</u> :
	[To be supplied.]
Alteri	native Forms of Delivery of Notices (telephone, facsimile or email):
	<u>Transmission Provider</u> :
	[To be supplied.]
	Interconnection Customer:
	[To be supplied.]

Appendix G: Interconnection Requirements for a Wind Generating Plant

Appendix G to LGIA

Interconnection Requirements for a

Wind Generating Plant

Appendix G sets forth requirements and provisions specific to a wind generating plant. All other requirements of this LGIA continue to apply to wind generating plant interconnections.

A. Technical Standards Applicable to a Wind Generating Plant

i. Low Voltage Ride-Through (LVRT) Capability

A wind generating plant shall be able to remain online during voltage disturbances up to the time periods and associated voltage levels set forth in the standard below. The LVRT standard provides for a transition period standard and a post-transition period standard.

Transition Period LVRT Standard

The transition period standard applies to wind generating plants subject to FERC Order 661 that have either: (i) interconnection agreements signed and filed with the Commission, filed with the Commission in unexecuted form, or filed with the Commission as non-conforming agreements between January 1, 2006 and December 31, 2006, with a scheduled in-service date no later than December 31, 2007, or (ii) wind generating turbines subject to a wind turbine procurement contract executed prior to December 31, 2005, for delivery through 2007.

1. Wind generating plants are required to remain in-service during three-phase faults with normal clearing (which is a time period of approximately 4 - 9 cycles) and single line to ground faults with delayed clearing, and subsequent post-fault voltage recovery to prefault voltage unless clearing the fault effectively disconnects the generator from the system. The clearing time requirement for a three-phase fault will be specific to the wind generating plant substation location, as determined by and documented by the Transmission Provider. The maximum clearing time the wind generating plant shall be required to withstand for a three-phase fault shall be 9 cycles at a voltage as low as 0.15 p.u., as measured at the high side of the wind generating plant step-up transformer (i.e. the transformer that steps the voltage up to the transmission interconnection voltage or "GSU"), after which, if the fault remains following the location-specific normal clearing time for three-phase faults, the wind generating plant may disconnect from the transmission system.

- 2. This requirement does not apply to faults that would occur between the wind generator terminals and the high side of the GSU or to faults that would result in a voltage lower than 0.15 per unit on the high side of the GSU serving the facility.
- 3. Wind generating plants may be tripped after the fault period if this action is intended as part of a special protection system.
- 4. Wind generating plants may meet the LVRT requirements of this standard by the performance of the generators or by installing additional equipment (e.g., Static VAr Compensator, etc.) within the wind generating plant or by a combination of generator performance and additional equipment.
- 5. Existing individual generator units that are, or have been, interconnected to the

network at the same location at the effective date of the Appendix G LVRT Standard are exempt from meeting the Appendix G LVRT Standard for the remaining life of the existing generation equipment. Existing individual generator units that are replaced are required to meet the Appendix G LVRT Standard.

Post-transition Period LVRT Standard

All wind generating plants subject to FERC Order No. 661 and not covered by the transition period described above must meet the following requirements:

- 1. Wind generating plants are required to remain in-service during three-phase faults with normal clearing (which is a time period of approximately 4 - 9 cycles) and single line to ground faults with delayed clearing, and subsequent post-fault voltage recovery to prefault voltage unless clearing the fault effectively disconnects the generator from the system. The clearing time requirement for a three-phase fault will be specific to the wind generating plant substation location, as determined by and documented by the Transmission Provider. The maximum clearing time the wind generating plant shall be required to withstand for a three-phase fault shall be 9 cycles after which, if the fault remains following the location-specific normal clearing time for three-phase faults, the wind generating plant may disconnect from the transmission system. A wind generating plant shall remain interconnected during such a fault on the transmission system for a voltage level as low as zero volts, as measured at the high voltage side of the wind GSU.
- 2. This requirement does not apply to faults that would occur between the wind generator terminals and the high side of the GSU.

- 3. Wind generating plants may be tripped after the fault period if this action is intended as part of a special protection system.
- 4. Wind generating plants may meet the LVRT requirements of this standard by the performance of the generators or by installing additional equipment (e.g., Static VAr Compensator) within the wind generating plant or by a combination of generator performance and additional equipment.
- 5. Existing individual generator units that are, or have been, interconnected to the network at the same location at the effective date of the Appendix G LVRT Standard are exempt from meeting the Appendix G LVRT Standard for the remaining life of the existing generation equipment. Existing individual generator units that are replaced are required to meet the Appendix G LVRT Standard.

ii. Power Factor Design Criteria (Reactive Power)

The following reactive power requirements apply only to a newly interconnecting wind generating plant that has executed a Facilities Study Agreement as of the effective date of the Final Rule establishing the reactive power requirements for non-synchronous generators in section 9.6.1 of this LGIA (Order No. 827). A wind generating plant to which this provision applies shall maintain a power factor within the range of 0.95 leading to 0.95 lagging, measured at the Point of Interconnection as defined in this LGIA, if the Transmission Provider's System Impact Study shows that such a requirement is necessary to ensure safety or reliability. The power factor range standard can be met by using, for example, power electronics designed to supply this level of reactive capability (taking into account any limitations due to voltage level, real power output, etc.) or fixed and switched capacitors if agreed to by the Transmission Provider, or a combination of the two. The Interconnection Customer shall not disable power factor equipment while the wind plant is in operation. Wind plants shall also be able to provide sufficient dynamic voltage support in lieu of the power system stabilizer and automatic voltage regulation at the generator excitation system if the System Impact Study shows this to be required for system safety or reliability.

iii. Supervisory Control and Data Acquisition (SCADA) Capability

The wind plant shall provide SCADA capability to transmit data and receive instructions from the Transmission Provider to protect system reliability. The Transmission Provider and the wind plant Interconnection Customer shall determine what SCADA information is essential for the proposed wind plant, taking into account the size of the plant and its characteristics, location, and importance in maintaining generation resource adequacy and transmission system reliability in its area.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

FERC rendition of the electronically filed tariff records in Docket No. ER23-02031-000

Filing Data: CID: C000465

Filing Title: OATT Revisions Filing of Attachment M Related to Interconnection Queue

Company Filing Identifier: 183 Type of Filing Code: 10 Associated Filing Identifier:

Tariff Title: Open Access Transmission Tariff

Tariff ID: 5

Payment Confirmation: Suspension Motion:

Tariff Record Data:

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Attachment M, Large Generation Interconnection Procedures and Agreement, 1.2.0, A

Record Narative Name: Tariff Record ID: 65

Tariff Record Collation Value: 692615323 Tariff Record Parent Identifier: 0

Proposed Date: 2023-08-01

Priority Order: 770

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

ATTACHMENT M

Large Generator Interconnection Procedures and Agreement

Standard Large Generator Interconnection Procedures (LGIP)

(Applicable to Generating Facilities that exceed 20 MW)

TABLE OF CONTENTS

Section 1.	Definitions
Section 2. 2.1 2.2	
2.3	Base Case Data
2.4	No Applicability to Transmission Service
Section 3.	Interconnection Requests
3.1	General
3.2	Identification of Types of Interconnection Services
	3.2.1 Energy Resource Interconnection Service
	3.2.1.1 The Product
	3.2.1.2 The Study
	3.2.2 Network Resource Interconnection Service
	3.2.2.1 The Product
	3.2.2.2 The Study
3.3	Utilization of Surplus Interconnection Service
2.4	3.3.1 Surplus Interconnection Service Requests
3.4	Valid Interconnection Request
	3.4.1 Initiating an Interconnection Request
	3.4.2 Acknowledgment of Interconnection Request
	3.4.3 Deficiencies in Interconnection Request
2.5	3.4.4 Scoping Meeting
3.5	ϵ
	3.5.1 OASIS Posting
	3.5.2 Requirement to Post Interconnection Study Metrics
	3.5.2.1 Interconnection System Impact Studies Processing Time
	3.5.2.2 Interconnection Facilities Studies Processing Time
	3.5.2.3 Interconnection Service Requests Withdrawn from
3.6	Interconnection Queue
3.7	Coordination with Affected Systems Withdrawal
3./	3.7.1 Withdrawal Penalty
	3.7.1.1 Calculation of the Withdrawal Penalty
	3.7.1.2 Distribution of the Withdrawal Penalty
3.8	Identification of Contingent Facilities
Section 4.	Interconnection Request Evaluation Process
4.1	Queue Position
	Zurur z somon

4.1.1 Assignment of Queue Position

4.2	4.1.2 Higher Queue PositionClustering and the Definitive Study Process4.2.1 Initiation of a Resource Solicitation Cluster4.2.2 Network Upgrades Cost Allocation
4.3 4.4	Transferability of Queue Position Modifications
Section 5.	Procedures for Interconnection Requests Submitted Prior to Effective Date of Standard Large Generator Interconnection Procedures
5.1 5.2	Transition to the First-Ready, First-Served Process New Transmission Provider
Section 6.	Informational Interconnection Study
6.1	Informational Interconnection Study Agreement
6.2	Scope of Informational Interconnection Study
6.3	Informational Interconnection Study Procedures
Section 7.	Definitive Interconnection System Impact Study
7.1	Definitive Interconnection System Impact Study Agreement
7.2	Execution of Definitive Interconnection System Impact Study Agreement
7.3	Scope of Definitive Interconnection System Impact Study
7.4	DefinitiveInterconnection System Impact Study Procedures
7.5	Meeting with Transmission Provider
7.6	Re-Study
7.7	Readiness Milestones and Site Control
	7.7.1 Readiness Milestone 1 ("M1")
	7.7.2 Readiness Milestone 2 ("M2")
	7.7.3 Readiness Milestone 3 ("M3")
	7.7.4 Readiness Milestone 4 ("M3")
	7.7.5 Security Requirements
	7.7.6 [Reserved for Future Use]
	7.7.7 Site Control
Section 8.	Interconnection Facilities Study
8.1	Interconnection Facilities Study Agreement
8.2	Scope of Interconnection Facilities Study
8.3	Interconnection Facilities Study Procedures
8.4	Meeting with Transmission Provider
8.5	Re-Study
Section 9.	Engineering & Procurement ('E&P') Agreement

Section 10. [Reserved for Future Use]

- Section 11. Standard Large Generator Interconnection Agreement (LGIA)
 - 11.1 Tender
 - 11.2 Negotiation
 - 11.3 Execution and Filing
 - 11.4 Commencement of Interconnection Activities
- Section 12. Construction of Transmission Provider's Interconnection Facilities and Network Upgrades
 - Schedule 12.1
 - 12.2 Construction Sequencing
 - 12.2.1 General
 - 12.2.2 Advance Construction of Network Upgrades that are an Obligation of an Entity other than Interconnection Customer
 - 12.2.3 Advancing Construction of Network Upgrades that are Part of an Expansion Plan of the Transmission Provider
 - 12.2.4 Amended Interconnection System Impact Study
- Section 13. Miscellaneous
 - 13.1 Confidentiality
 - 13.1.1 Scope
 - 13.1.2 Release of Confidential Information
 - 13.1.3 Rights
 - 13.1.4 No Warranties
 - 13.1.5 Standard of Care
 - 13.1.6 Order of Disclosure
 - 13.1.7 Remedies
 - 13.1.8 Disclosure to FERC, its Staff, or a State
 - 13.2 Delegation of Responsibility
 - Obligation for Study Costs and Withdrawal Penalty 13.3
 - Third Parties Conducting Studies 13.4
 - 13.5 Disputes
 - 13.5.1 Submission
 - 13.5.2 External Arbitration Procedures
 - 13.5.3 Arbitration Decisions
 - 13.5.4 Costs
 - 13.5.5 Non-binding dispute resolution procedures
 - 13.6 Local Furnishing Bonds
 - 13.6.1 Transmission Providers That Own Facilities Financed by Local **Furnishing Bonds**
 - 13.6.2 Alternative Procedures for Requesting Interconnection Service

- Appendix 1 Interconnection Request for a Large Generating Facility
- Appendix 2 Definitive Interconnection System Impact Study Agreement
- Appendix 3 Interconnection Facilities Study Agreement
- Appendix 4 Transitional Study Agreements
 - Transitional Facilities Interconnection Study Agreement 4.1
 - Transitional Cluster Study Agreement 4.2
- Appendix 5 Informational Study Forms
 - Informational Interconnection Study Request 5.1
 - 5.2 Informational Interconnection Study Agreement

Appendix 6 - Standard Large Generator Interconnection Agreement

Section 1. Definitions

Adverse System Impact shall mean the negative effects due to technical or operational limits on conductors or equipment being exceeded that may compromise the safety and reliability of the electric system.

Affected System shall mean an electric system other than the Transmission Provider's Transmission System that may be affected by the proposed interconnection.

Affected System Operator shall mean the entity that operates an Affected System.

Affiliate shall mean, with respect to a corporation, partnership or other entity, each such other corporation, partnership or other entity that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such corporation, partnership or other entity.

Ancillary Services shall mean those services that are necessary to support the transmission of capacity and energy from resources to loads while maintaining reliable operation of the Transmission Provider's Transmission System in accordance with Good Utility Practice.

Applicable Laws and Regulations shall mean all duly promulgated applicable federal, state and local laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders, permits and other duly authorized actions of any Governmental Authority.

Applicable Reliability Council shall mean the reliability council applicable to the Transmission System to which the Generating Facility is directly interconnected.

Applicable Reliability Standards shall mean the requirements and guidelines of NERC, the Applicable Reliability Council, and the Control Area of the Transmission System to which the Generating Facility is directly interconnected.

Base Case shall mean the base case power flow, short circuit, and stability data bases used for the Interconnection Studies by the Transmission Provider or Interconnection Customer.

Breach shall mean the failure of a Party to perform or observe any material term or condition of the Standard Large Generator Interconnection Agreement.

Breaching Party shall mean a Party that is in Breach of the Standard Large Generator Interconnection Agreement.

Business Day shall mean Monday through Friday, excluding Federal Holidays.

Calendar Day shall mean any day including Saturday, Sunday or a Federal Holiday.

Cluster shall mean a group of Interconnection Requests (one or more) that are studied together for the purpose of conducting Interconnection Studies.

Cluster Study shall mean an Interconnection Study evaluating one or more Interconnection Requests.

Clustering shall mean the process whereby a group of Interconnection Requests is studied together, instead of serially, for the purpose of conducting Interconnection Studies.

Commercial Operation shall mean the status of a Generating Facility that has commenced generating electricity for sale, excluding electricity generated during Trial Operation.

Commercial Operation Date of a unit shall mean the date on which the Generating Facility commences Commercial Operation as agreed to by the Parties pursuant to Appendix E to the Standard Large Generator Interconnection Agreement.

Confidential Information shall mean any confidential, proprietary or trade secret information of a plan, specification, pattern, procedure, design, device, list, concept, policy or compilation relating to the present or planned business of a Party, which is designated as confidential by the Party supplying the information, whether conveyed orally, electronically, in writing, through inspection, or otherwise.

Contingent Facilities shall mean those unbuilt Interconnection Facilities and Network Upgrades upon which the Interconnection Request's costs, timing, and study findings are dependent, and if delayed or not built, could cause a need for Re-Studies of the Interconnection Request or a reassessment of the Interconnection Facilities and/or Network Upgrades and/or costs and timing.

Control Area shall mean an electrical system or systems bounded by interconnection metering and telemetry, capable of controlling generation to maintain its interchange schedule with other Control Areas and contributing to frequency regulation of the interconnection. A Control Area must be certified by an Applicable NERC Regional Reliability Entity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC.

Customer Engagement Window shall have the meaning set forth in Section 4.2 of the LGIP.

Default shall mean the failure of a Breaching Party to cure its Breach in accordance with Article 17 of the Standard Large Generator Interconnection Agreement.

Definitive Interconnection Study Process ("Definitive Interconnection Study") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study.

Definitive Interconnection System Impact Study ("DISIS") shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System.

Definitive Interconnection System Impact Study Agreement ("DISIS Agreement") shall mean the form of agreement contained in the Appendices of the LGIP for conducting the Definitive Interconnection System Impact Study.

Definitive Interconnection System Impact Study Cluster ("DISIS Cluster") shall mean an engineering study that evaluates the impact of the proposed interconnection(s) on the safety and reliability of the Transmission System and, if applicable, an Affected System.

DISIS Request Window shall have the meaning set forth in Section 4.2 of the LGIP.

Dispute Resolution shall mean the procedure for resolution of a dispute between the Parties in which they will first attempt to resolve the dispute on an informal basis.

Distribution System shall mean the Transmission Provider's facilities and equipment used to transmit electricity to ultimate usage points such as homes and industries directly from nearby generators or from interchanges with higher voltage transmission networks which transport bulk power over longer distances. The voltage levels at which distribution systems operate differ among areas.

Distribution Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Distribution System at or beyond the Point of Interconnection to facilitate interconnection of the Generating Facility and render the

transmission service necessary to effect Interconnection Customer's wholesale sale of electricity in interstate commerce. Distribution Upgrades do not include Interconnection Facilities.

Effective Date shall mean the date on which the Standard Large Generator Interconnection Agreement becomes effective upon execution by the Parties subject to acceptance by FERC, or if filed unexecuted, upon the date specified by FERC.

Emergency Condition shall mean a condition or situation: (1) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (2) that, in the case of a Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to Transmission Provider's Transmission System, Transmission Provider's Interconnection Facilities or the electric systems of others to which the Transmission Provider's Transmission System is directly connected; or (3) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Generating Facility or Interconnection Customer's Interconnection Facilities. System restoration and black start shall be considered Emergency Conditions; provided that Interconnection Customer is not obligated by the Standard Large Generator Interconnection Agreement to possess black start capability.

Energy Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to connect its Generating Facility to the Transmission Provider's Transmission System to be eligible to deliver the Generating Facility's electric output using the existing firm or nonfirm capacity of the Transmission Provider's Transmission System on an as available basis. Energy Resource Interconnection Service in and of itself does not convey transmission service.

Engineering & Procurement (E&P) Agreement shall mean an agreement that authorizes the Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection in order to advance the implementation of the Interconnection Request.

Environmental Law shall mean Applicable Laws or Regulations relating to pollution or protection of the environment or natural resources.

Federal Power Act shall mean the Federal Power Act, as amended, 16 U.S.C. §§ 791a et seq.

FERC shall mean the Federal Energy Regulatory Commission (Commission) or its successor.

Force Majeure shall mean any act of God, labor disturbance, act of the public enemy, war, insurrection, riot, fire, storm or flood, explosion, breakage or accident to machinery or equipment, any order, regulation or restriction imposed by governmental, military or lawfully established civilian authorities, or any other cause beyond a Party's control. A Force Majeure event does not include acts of negligence or intentional wrongdoing by the Party claiming Force Majeure.

Generating Facility shall mean Interconnection Customer's device for the production and/or storage for later injection of electricity identified in the Interconnection Request, but shall not include the Interconnection Customer's Interconnection Facilities.

Generating Facility Capacity shall mean the net capacity of the Generating Facility and the aggregate net capacity of the Generating Facility where it includes multiple energy production devices.

Good Utility Practice shall mean any of the practices, methods and acts engaged in or approved by a significant portion of the electric industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act to the exclusion of all others, but rather to be acceptable practices, methods, or acts generally accepted in the region.

Governmental Authority shall mean any federal, state, local or other governmental regulatory or administrative agency, court, commission, department, board, or other governmental subdivision, legislature, rulemaking board, tribunal, or other governmental authority having jurisdiction over the Parties, their respective facilities, or the respective services they provide, and exercising or entitled to exercise any administrative, executive, police, or taxing authority or power; provided, however, that such term does not include Interconnection Customer, Transmission Provider, or any Affiliate thereof.

Hazardous Substances shall mean any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "radioactive substances," "contaminants," "pollutants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, or any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

Informational Interconnection Study shall mean an analysis based on assumptions specified by Interconnection Customer in the Informational Interconnection Study Agreement.

Informational Interconnection Study Agreement shall mean the form of agreement contained in Appendix 5 of the LGIP for conducting the Informational Interconnection Study.

Initial Deposit shall mean a deposit of \$160,000 for an Interconnection Request of up to 75 MW, or \$250,000 for an Interconnection Request of 75 MW or greater, that Interconnection Customer submits to Transmission Provider upon initiating an Interconnection Request.

Initial Synchronization Date shall mean the date upon which the Generating Facility is initially synchronized and upon which Trial Operation begins.

In-Service Date shall mean the date upon which the Interconnection Customer reasonably expects it will be ready to begin use of the Transmission Provider's Interconnection Facilities to obtain back feed power.

Interconnection Customer shall mean any entity, including the Transmission Provider, Transmission Owner or any of the Affiliates or subsidiaries of either, that proposes to interconnect its Generating Facility with the Transmission Provider's Transmission System.

Interconnection Customer's Interconnection Facilities shall mean all facilities and equipment, as identified in Appendix A of the Standard Large Generator Interconnection Agreement, that are located between the Generating Facility and the Point of Change of Ownership, including any modification, addition, or upgrades to such facilities and equipment necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Customer's Interconnection Facilities are sole use facilities.

Interconnection Facilities shall mean the Transmission Provider's Interconnection Facilities and the Interconnection Customer's Interconnection Facilities. Collectively, Interconnection Facilities include all facilities and equipment between the Generating Facility and the Point of Interconnection, including any modification, additions or upgrades that are necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Interconnection Facilities Study shall mean a study conducted by the

Transmission Provider or a third party consultant for the Interconnection Customer to determine a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades as identified in the Definitive Interconnection System Impact Study), the cost of those facilities, and the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System. The scope of the study is defined in Section 8 of the Standard Large Generator Interconnection Procedures.

Interconnection Facilities Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Facilities Study.

Interconnection Request shall mean an Interconnection Customer's request, in the form of Appendix 1 to the Standard Large Generator Interconnection Procedures, in accordance with the Tariff, to interconnect a new Generating Facility, or to increase the capacity of, or make a Material Modification to the operating characteristics of, an existing Generating Facility that is interconnected with the Transmission Provider's Transmission System.

Interconnection Service shall mean the service provided by the Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to the Transmission Provider's Transmission System and enabling it to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Standard Large Generator Interconnection Agreement and, if applicable, the Transmission Provider's Tariff.

Interconnection Study shall mean any of the following studies: the Informational Interconnection Study, the Definitive Interconnection System Impact Study, and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection Study Agreement shall mean any of the following studies: the Informational Interconnection Study Agreement, the Definitive Interconnection System Impact Study and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

IRS shall mean the Internal Revenue Service.

Joint Operating Committee shall be a group made up of representatives from Interconnection Customers and the Transmission Provider to coordinate operating and technical considerations of Interconnection Service.

Large Generating Facility shall mean a Generating Facility having a Generating Facility Capacity of more than 20 MW.

Loss shall mean any and all losses relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's performance, or non performance of its obligations under the Standard Large Generator Interconnection Agreement on behalf of the Indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the Indemnifying Party.

Material Modification shall mean those modifications that have a material impact on the cost or timing of any Interconnection Request with an equal or later Queue Position.

Metering Equipment shall mean all metering equipment installed or to be installed at the Generating Facility pursuant to the Standard Large Generator Interconnection Agreement at the metering points, including but not limited to instrument transformers, MWh-meters, data acquisition equipment, transducers, remote terminal unit, communications equipment, phone lines, and fiber optics.

NERC shall mean the North American Electric Reliability Council or its successor organization.

Network Resource shall mean any designated generating resource owned, purchased, or leased by a Network Customer under the Network Integration Transmission Service Tariff. Network Resources do not include any resource, or any portion thereof, that is committed for sale to third parties or otherwise cannot be called upon to meet the Network Customer's Network Load on a non-interruptible basis.

Network Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to integrate its Large Generating Facility with the Transmission Provider's Transmission System (1) in a manner comparable to that in which the Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an RTO or ISO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service in and of itself does not convey transmission service.

Network Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Transmission System required at or beyond the point at which the Interconnection Facilities connect to the Transmission Provider's Transmission System to accommodate the interconnection of the Large Generating Facility to the Transmission Provider's Transmission System.

Notice of Dispute shall mean a written notice of a dispute or claim that arises out of or in connection with the Standard Large Generator Interconnection Agreement or its performance.

OASIS shall mean the Transmission Provider's Open Access Same-Time Information System.

Party or Parties shall mean Transmission Provider, Transmission Owner, Interconnection Customer or any combination of the above.

Permissible Technological Advancement shall mean a modification to a specific technology type submitted in an Interconnection Customer's Interconnection Request that: (1) is not a Material Modification; (2) is not a change in generation technology or fuel type, but may include advancements to other technology such as, e.g., advancements to turbines, inverters, plant supervisory controls, or other technological advancements that may affect a generating facility's ability to produce ancillary services; (3) results in electrical performance that is equal to or better than the electrical performance expected prior to the technology change; (4) does not degrade the electrical characteristics of the generating equipment (e.g., the ratings, impedances, efficiencies, capabilities, and performance of the equipment under steady state and dynamic conditions); (5) does not increase the Interconnection Customer's requested Interconnection Service and (6) does not cause any reliability concerns (i.e., materially impact the transmission system with regard to short circuit capability limits, steady-state thermal or voltage limits or dynamic system stability and response).

Phase ("Phase 1, Phase 2, Phase 3 or Phase 4") shall mean a distinct part of the Definitive Study Process as described in Section 7.

Point of Change of Ownership shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Customer's Interconnection Facilities connect to the Transmission Provider's Interconnection Facilities.

Point of Interconnection shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Facilities connect to the Transmission Provider's Transmission System.

Provisional Interconnection Service shall mean Interconnection Service provided by Transmission Provider associated with interconnecting the

Interconnection Customer's Generating Facility to Transmission Provider's Transmission System and enabling that Transmission System to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Provisional Large Generator Interconnection Agreement and, if applicable, the Tariff.

Provisional Large Generator Interconnection Agreement shall mean the interconnection agreement for Provisional Interconnection Service established between Transmission Provider and/or the Transmission Owner and the Interconnection Customer. This agreement shall take the form of the Large Generator Interconnection Agreement, modified for provisional purposes.

Queue shall mean a queue for valid Interconnection Requests for the Definitive Interconnection Study Process.

Queue Position shall mean the order of a valid Interconnection Request, relative to all other pending valid Interconnection Requests, in the Definitive Interconnection Study Process. The Queue Position is established based upon the date and time the Interconnection Customer satisfied all of the requirements of this Attachment M to enter the Definitive Study Process.

Reasonable Efforts shall mean, with respect to an action required to be attempted or taken by a Party under the Standard Large Generator Interconnection Agreement, efforts that are timely and consistent with Good Utility Practice and are otherwise substantially equivalent to those a Party would use to protect its own interests.

Resource Plan shall mean any process authorized or required by Applicable Laws and Regulations for, *inter alia*, the selection of Generating Facilities.

Resource Planning Entity shall mean any entity required to develop a Resource Plan or conduct a Resource Solicitation Process.

Resource Solicitation Cluster shall mean a Cluster Study associated with a Resource Plan or related process.

Resource Solicitation Process shall mean any process authorized or required by Applicable Laws and Regulations for the acquisition of Network Resources.

Scoping Meeting shall mean the meeting between representatives of the Interconnection Customer and Transmission Provider conducted for the purpose of discussing the proposed interconnection request, alternative interconnection options, to exchange information including any transmission data and earlier study evaluations that

would be reasonably expected to affect such interconnection options, to analyze such information, and to determine the potential feasible Points of Interconnection.

Site Control shall mean the exclusive land right to develop, construct, operate, ad maintain the Generating Facility over the term of expected operation of the Generating Facility. Site Control shall include the right to develop, construct, operate and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by documentation establishing: (1) ownership of, a leasehold interest in, or a right to develop a site of sufficient sized to construct and operate the Generating Facility and associated Interconnection Facilities; (2) an option to purchase or acquire a leasehold interest in a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities; or (3) any other documentation that clearly demonstrates the right of the Interconnection Customer to exclusively occupy a site of sufficient size to construct and operate the Generating Facility. Site Control f or any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all co-located projects that meet the aforementioned provisions of this Site Control definition.

Small Generating Facility shall mean a Generating Facility that has a Generating Facility Capacity of no more than 20 MW.

Stand Alone Network Upgrades shall mean Network Upgrades that are not part of an Affected System that an Interconnection Customer may construct without affecting day-to-day operations of the Transmission System during their construction. Both the Transmission Provider and the Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify them in Appendix A to the Standard Large Generator Interconnection Agreement. If the Transmission Provider and Interconnection Customer disagree about whether a particular Network Upgrade is a Stand Alone Network Upgrade, the Transmission Provider must provide the Interconnection Customer a written technical explanation outlining why the Transmission Provider does not consider the Network Upgrade to be a Stand Alone Network Upgrade within 15 days of its determination.

Standard Engineering and Procurement Agreement shall mean the form of engineering and procurement agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Agreement (LGIA) shall mean the form of interconnection agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Procedures (LGIP) shall mean the

interconnection procedures applicable to an Interconnection Request pertaining to a Large Generating Facility that are included in the Transmission Provider's Tariff.

Start Date shall mean that actual date of the start of the System Impact Study as evidenced by the initiation of the development of the required Base Case(s), or the initiation of the technical study work, i.e., start of the powerflow/stability computer runs, whichever is applicable.

Surplus Interconnection Service shall mean any unneeded portion of Interconnection Service established in a Large Generator Interconnection Agreement, such that if Surplus Interconnection Service is utilized, the total amount of Interconnection Service at the Point of Interconnection would remain the same.

System Protection Facilities shall mean the equipment, including necessary protection signal communications equipment, required to protect (1) the Transmission Provider's Transmission System from faults or other electrical disturbances occurring at the Generating Facility and (2) the Generating Facility from faults or other electrical system disturbances occurring on the Transmission Provider's Transmission System or on other delivery systems or other generating systems to which the Transmission Provider's Transmission System is directly connected.

Tariff shall mean the Transmission Provider's Tariff through which open access transmission service and Interconnection Service are offered, as filed with FERC, and as amended or supplemented from time to time, or any successor tariff.

Transmission Owner shall mean an entity that owns, leases or otherwise possesses an interest in the portion of the Transmission System at the Point of Interconnection and may be a Party to the Standard Large Generator Interconnection Agreement to the extent necessary.

Transmission Provider shall mean the public utility (or its designated agent) that owns, controls, or operates transmission or distribution facilities used for the transmission of electricity in interstate commerce and provides transmission service under the Tariff. The term Transmission Provider should be read to include the Transmission Owner when the Transmission Owner is separate from the Transmission Provider.

Transmission Provider's Interconnection Facilities shall mean all facilities and equipment owned, controlled, or operated by the Transmission Provider from the Point of Change of Ownership to the Point of Interconnection as identified in Appendix A to the Standard Large Generator Interconnection Agreement, including any modifications, additions or upgrades to such facilities and equipment. Transmission Provider's

Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Transmission System shall mean the facilities owned, controlled or operated by the Transmission Provider or Transmission Owner that are used to provide transmission service under the Tariff.

Trial Operation shall mean the period during which Interconnection Customer is engaged in on-site test operations and commissioning of the Generating Facility prior to Commercial Operation.

Withdrawal Penalty shall have the meaning set forth in Section 3.7.1 of the LGIP.

Section 2. Scope and Application

2.1 Application of Standard Large Generator Interconnection Procedures
Sections 2 through 13 apply to processing an Interconnection Request
pertaining to a Large Generating Facility. Small Generating Facilities
requesting NRIS shall be processed under the LGIP.

2.2 Comparability

Transmission Provider shall receive, process and analyze all Interconnection Requests in a timely manner as set forth in this LGIP. Transmission Provider will use the same Reasonable Efforts in processing and analyzing Interconnection Requests from all Interconnection Customers, whether the Generating Facilities are owned by Transmission Provider, its subsidiaries or Affiliates or others.

2.3 Base Case Data

Transmission Provider shall maintain, consistent with Applicable Laws and Regulations and Applicable Reliability Standards, base power flow, short circuit and stability databases, including all underlying assumptions, and contingency list on either its OASIS site or a password-protected website, subject to confidentiality provisions in LGIP Section 13.1. In addition, Transmission Provider shall maintain network models and underlying assumptions on either its OASIS site or a password-protected website. Such network models and underlying assumptions should reasonably represent those used during the most recent Interconnection Study and be representative of current system conditions. If Transmission Provider posts this information on a password-protected website, a link to the information must be provided on Transmission Provider's OASIS site. Transmission Provider is permitted to require that Interconnection

Customers, OASIS site users and password-protected website users sign a confidentiality agreement before the release of commercially sensitive information or Critical Energy Infrastructure Information in the Base Case data. Such databases and lists, hereinafter referred to as Base Cases, shall include all (1) generation projects and (2) transmission projects, including merchant transmission projects that are proposed for the Transmission System for which a transmission expansion plan has been submitted and approved by the applicable authority.

2.4 No Applicability to Transmission Service

Nothing in this LGIP shall constitute a request for transmission service or confer upon an Interconnection Customer any right to receive transmission service.

Section 3. Interconnection Requests

3.1 General

An Interconnection Customer shall submit to Transmission Provider an Interconnection Request in the form of Appendix 1 to this LGIP and the Initial Deposit of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater. Transmission Provider shall apply the Initial Deposit toward the cost of the Definitive Interconnection Study Process. Interconnection Customer shall submit a separate Interconnection Request for each site and may submit multiple Interconnection Requests for a single site. Interconnection Customer must submit an Initial Deposit with each Interconnection Request even when more than one request is submitted for a single site. An Interconnection Request to evaluate one site at two different voltage levels shall be treated as two Interconnection Requests. Similarly, an Interconnection Request to evaluate one interconnection at two or more different sites (two or more Points of Interconnection) shall be treated as separate Interconnection Requests.

At Interconnection Customer's option, Transmission Provider and Interconnection Customer will identify alternative Point(s) of Interconnection and configurations at the Scoping Meeting to evaluate in this process and attempt to eliminate alternatives in a reasonable fashion given resources and information available. Interconnection Customer will select the definitive Point of Interconnection to be studied no later than the execution of the Definitive Interconnection System Impact Study Agreement. For purposes of clustering Interconnection Service requests, Transmission Provider may make reasonable changes to the

requested Point(s) of Interconnection to facilitate efficient interconnection of Interconnection Customers at common points of interconnection. Transmission Provider shall notify Interconnection Customers in writing of any intended changes to the requested Point(s) of Interconnection and the Point(s) of Interconnection shall only change upon mutual agreement.

Transmission Provider shall use the process below to consider requests for Interconnection Service below the Generating Facility Capacity. These requests for Interconnection Service shall be studied at the level of Interconnection Service requested for purposes of Interconnection Facilities, Network Upgrades, and associated costs but may be subject to other studies at the full Generating Facility Capacity to ensure safety and reliability of the system, with the study costs borne by the Interconnection Customer. If after the additional studies are complete, Transmission Provider determines that additional Network Upgrades are necessary, then Transmission Provider must: (1) specify which additional Network Upgrade costs are based on which studies; and (2) provide a detailed explanation of why the additional Network Upgrades are necessary. Any Interconnection Facility and/or Network Upgrade costs required for safety and reliability also will be borne by the Interconnection Customer. Interconnection Customers may be subject to additional control technologies as well as testing and validation of those technologies consistent with Article 6 of the LGIA. The necessary control technologies and protection systems shall be established in Appendix C of the executed, or requested to be filed unexecuted, LGIA.

3.2 Identification of Types of Interconnection Services

At the time the Interconnection Request is submitted, Interconnection Customer must request either Energy Resource Interconnection Service or Network Resource Interconnection Service, as described below. Interconnection Customer may designate only one type of Interconnection Service for each separate Interconnection Request in the Queue. The type of Interconnection Service must be finalized on submission of the executed Definitive System Impact Study Agreement and may only be changed after the start of the Definitive Study Process between Phase 2 and Phase 3 of the Definitive Interconnection Study Process and only if a Cluster must be re-studied in Phase 3 and otherwise may not be changed.

3.2.1 Energy Resource Interconnection Service

3.2.1.1 The Product.

Energy Resource Interconnection Service allows Interconnection Customer to connect the Large Generating Facility to the Transmission System and be eligible to deliver the Large Generating Facility's output using the existing firm or non-firm capacity of the Transmission System on an "as available" basis. Energy Resource Interconnection Service does not in and of itself convey any right to deliver electricity to any specific customer or Point of Delivery.

3.2.1.2 The Study.

The study consists of short circuit/fault duty, steady state (thermal and voltage) and stability analyses. The short circuit/fault duty analysis would identify direct Interconnection Facilities required and the Network Upgrades necessary to address short circuit issues associated with the Interconnection Facilities. The stability and steady state studies would identify necessary upgrades to allow full output of the proposed Large Generating Facility and would also identify the maximum allowed output, at the time the study is performed, of the interconnecting Large Generating Facility without requiring additional Network Upgrades.

3.2.2 Network Resource Interconnection Service

3.2.2.1 The Product.

Transmission Provider must conduct the necessary studies and construct the Network Upgrades needed to integrate the Large Generating Facility: (1) in a manner comparable to that in which Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an ISO or RTO with market based congestion management, in the same manner as Network Resources. Network Resource Interconnection Service allows Interconnection Customer's Large Generating Facility to be designated as a Network Resource, up to the Large Generating Facility's full output, on the same basis as existing Network Resources interconnected to Transmission Provider's Transmission System, and to be studied as a Network Resource on the assumption that such a designation will occur. If the Transmission Provider has not been notified pursuant to Section 29.2 of Part

III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resource within Transmission Provider's Control Area, the Interconnection Customer must provide the point of delivery or the geographic location on EPE's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area.

3.2.2.2 The Study.

The Interconnection Study for Network Resource Interconnection Service shall assure that Interconnection Customer's Large Generating Facility meets the requirements for Network Resource Interconnection Service and as a general matter, that such Large Generating Facility's interconnection is also studied with Transmission Provider's Transmission System at peak load, under a variety of severely stressed conditions, to determine whether, with the Large Generating Facility at full output, the aggregate of generation in the local area can be delivered to the aggregate of load on Transmission Provider's Transmission System, consistent with Transmission Provider's reliability criteria and procedures. This approach assumes that some portion of existing Network Resources' output is displaced by the output of Interconnection Customer's Large Generating Facility. Network Resource Interconnection Service in and of itself does not convey any right to deliver electricity to any specific customer or Point of Delivery. The Transmission Provider may also study the Transmission System under non-peak load conditions. However, upon request by the Interconnection Customer, the Transmission Provider must explain in writing to the Interconnection Customer why the study of non-peak load conditions is required for reliability purposes.

3.3 Utilization of Surplus Interconnection Service.

Transmission Provider will use the process below to allow an Interconnection Customer to utilize or transfer Surplus Interconnection Service at an existing Point of Interconnection. The original Interconnection Customer or one of its affiliates shall have priority to

utilize Surplus Interconnection Service. If the existing Interconnection Customer or one of its affiliates does not exercise its priority, then that service may be made available to other potential Interconnection Customers.

3.3.1 Surplus Interconnection Service Requests.

Surplus Interconnection Service requests may be made by the existing Interconnection Customer whose Generating Facility is already interconnected or one of its affiliates. Surplus Interconnection Service requests also may be made by another Interconnection Customer. Transmission Provider shall use the process in Section 3.3.2 in evaluating Interconnection Requests for Surplus Interconnection Service. Studies for Surplus Interconnection Service shall consist of reactive power, short circuit/fault duty, stability analyses, and any other appropriate studies. Steady-state (thermal/voltage) analyses may be performed as necessary to ensure that all required reliability conditions are studied. If the Surplus Interconnection Service was not studied under off-peak conditions, off-peak steady state analyses shall be performed to the required level necessary to demonstrate reliable operation of the Surplus Interconnection Service. If the original System Impact Study is not available for the Surplus Interconnection Service, both off-peak and peak analysis may need to be performed for the existing Generating Facility associated with the request for Surplus Interconnection Service. The reactive power, short circuit/fault duty, stability, and steady-state analyses for Surplus Interconnection Service will identify any additional Interconnection Facilities and/or Network Upgrades necessary.

3.3.2 Process for Securing Surplus Interconnection Service.

An existing (original) Interconnection Customer whose Large Generating Facility is already interconnected, may make available its Surplus Interconnection Service at the existing Point of Interconnection for the Large Generating Facility using the process outlined in this Section 3.3.2. The original Interconnection Customer may retain any surplus for itself, or may transfer it to an entity of its own choosing, whether an affiliate or a non-affiliate. The amount of Surplus Interconnection Service made available cannot exceed the total amount of Interconnection Service at the Point of Interconnection established in the original Interconnection Customer's Large Generator Interconnection Agreement.

A. Either the original Interconnection Customer, its affiliate or a third-party Interconnection Customer (transferee) shall notify the Transmission Provider in writing of a request to utilize or transfer any Surplus Interconnection Service made available by the original

Interconnection Customer at the existing Point of Interconnection. The written Surplus Interconnection Service request should include, at a minimum, the following information; (1) the amount of Surplus Interconnection Service (in MW) proposed to be made available; (2) the start date and end date that the Surplus Interconnection Service is to be made available; (3) the type of service (Energy Resource Interconnection Service or Network Resource Interconnection Service) to be made available as Surplus Interconnection Service, provided however, that if the original Interconnection Customer's Large Generator Interconnection Agreement provides for Energy Resource Interconnection Service, any Surplus Interconnection Service must also be Energy Resource Interconnection Service; (4) a one-line diagram illustrating how the Generating Facility that is to use the Surplus Interconnection Service will connect to the original Interconnection Customer's Interconnection Facilities; (5) details on the Generating Facility that is to use the Surplus Interconnection Service (type of resource, characteristics, modeling information); (6) the signature of the original Interconnection Customer if the original Interconnection Customer is not the entity submitting the Surplus Interconnection Service request; and (7) any other relevant conditions of the original Interconnection Customer on its proposed utilization or transfer. To aid the original Interconnection Customer in initiating this process, Transmission Provider may post on OASIS a blank request form for Surplus Interconnection Service, identifying the required fields of data set forth above.

B. Transmission Provider shall process such surplus interconnection service requests separately from other requests pending in its non-surplus interconnection queue. All requests for Surplus Interconnection Service will be subject to technical studies, as necessary, to ensure the reliable use of Surplus Interconnection Service. The Transmission Provider shall evaluate if the original System Impact Studies are still applicable and sufficient to accommodate the new request, and may require new or additional studies to evaluate system impacts. Examples of circumstances in which the Transmission Provider would require new studies include, but are not limited to, circumstances in which the original System Impact Study(ies) are more than 3 years old and/or circumstances in which new generation facilities have been added to the Transmission Provider's system since the original Interconnection Customer's Large Generating Facility was studied. Examples of studies that may be necessary and appropriate to examine the potential for system impacts include, but are not limited to:

- i. Powerflow studies (steady-state, thermal/voltage),
- ii. Stability studies (voltage/angular),
- iii. Short circuit/fault duty studies,
- iv. Reactive power studies
- v. Transient, harmonic and/or sub-transient studies, particularly if the generation technology of the generation facility using the Surplus Interconnection Service is different from the generation technology in the original Interconnection Customer's Large Generation Interconnection Agreement.
- C. Should Transmission Provider determine that studies are necessary to evaluate a request for Surplus Interconnection Service, Transmission Provider will collect from the requestor a study deposit of \$25,000, subject to true-up to reflect the actual cost of the studies.
- D. If the technical studies show that the addition of Network Upgrades will be required for the use of Surplus Interconnection Service due to different technology or short circuit requirements, or for other reasons, Surplus Interconnection Service shall be available up to the amount that can be accommodated without Network Upgrades. The Generating Facility receiving Surplus Interconnection Service is to interconnect to the original Interconnection Customer's Interconnection Facilities. Changes or additions to the original Interconnection Customer's Interconnection Facilities may be necessary and permissible. The need for changes to other facilities identified in the original Interconnection Customer's Interconnection Agreement, including Network Upgrades, would make Surplus Interconnection Service unavailable.
- E. An agreement for Surplus Interconnection Service shall be developed between the Transmission Provider, the original Interconnection Customer and the Surplus Interconnection Service Customer. The Transmission Provider will file the agreement for Surplus Interconnection Service with the Commission. The agreement will contain the parameters of the Surplus Interconnection Service, its level of service, term of service, and any conditions arising from the results of the Transmission Provider's evaluation of system impacts. Upon Commission approval of the Surplus Interconnection Agreement and concurrent with the effective date of the Surplus Interconnection Service, the original Interconnection Customer's Interconnection Agreement shall be deemed to be adjusted to lower the level of Interconnection Service by an amount equal to the Surplus Interconnection Service, without the need for

any filings or approvals.

- F. The use of Surplus Interconnection Service does not convey any promise or grant of transmission service.
- G. Surplus Interconnection Service cannot be offered until all facilities required for the original Interconnection Customer's interconnection service (including all Contingent Facilities) are constructed and In Service.
- H. Surplus Interconnection Service cannot be offered if the original Interconnection Customer's Generating Facility is scheduled to retire and permanently cease Commercial Operation before the Surplus Interconnection Service Customer's Generating Facility begins Commercial Operation.

3.4 Valid Interconnection Request

3.4.1 Initiating an Interconnection Request

An Interconnection Customer wishing to join the Definitive Interconnection Study Process shall submit a completed Interconnection Request to Transmission Provider within, and no later than the close of, the DISIS Request Window. To initiate an Interconnection Request, Interconnection Customer must submit all of the following:

- (i) Initial Deposit of \$160,000 for an Interconnection Request of up to 75 MW or \$250,000 for an Interconnection Request of 75 MW or greater;
- (ii) A completed application in the form of Appendix 1 (including all technical information);
- (iii) A demonstration of Site Control as defined in Section 1 and addressed further in Section 7.7.7 of the LGIP.

 Specifications for acceptable site size for the purposes of demonstrating Site Control are posted on Transmission Provider's OASIS website. Interconnection Customer may propose alternative specifications for site size to those posted on OASIS for Transmission Provider approval. In the event Transmission Provider and Interconnection Customer cannot reach agreement related to adequacy of site size,

 Transmission Provider will accept a Professional Engineer (licensed in the state of Texas or New Mexico, as applicable depending on the state in which the Generating Facility is to

be located) stamped site plan drawing that depicts the proposed generation arrangement and specifies the maximum facility output for that arrangement;

- (iv) A Point of Interconnection;
- (v) If the request is for NRIS and if Transmission Provider has not been notified pursuant to Section 29.2 of Part III of the Tariff that Interconnection Customer's proposed Generating Facility is to be designated as a Network Resourced within Transmission Provider's Control Area, the point of delivery or the geographic location on Transmission Provider's system at which Interconnection Customer intends to deliver output out of Transmission Provider's Control Area;
- (vi) A Generating Facility size (MW) (and requested Interconnection Service amount if the requested Interconnection Service is less than the Generating Facility Capacity);
- (vii) One of the following Readiness Milestone ("M1") options totaling the entire capacity of the Generating Facility (or requested Interconnection Service amount if the requested Interconnection Service amount is less than the Generating Facility Capacity or security equal to one times the study deposit described above (see Initial Deposit) in the form of an irrevocable letter of credit or cash *in lieu of* the Readiness Milestone. The security is refunded to the Interconnection Customer according to Section 7.7.5.
 - a. Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale (1) of the constructed Generating Facility, or (2) of the Generating Facility's energy, or (3) of the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
 - b. Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or
 - c. Provisional Large Generator Interconnection
 Agreement filed with FERC that contains a
 commitment to move forward with constructing the
 Generating Facility and is not suspended; and

(viii) Security equal to one times the study Initial Deposit in the form of an irrevocable letter of credit or cash. The security is refunded to the Interconnection Customer according to Section 7.7.

The expected In-Service Date of the new Large Generating Facility or increase in capacity of the existing Generating Facility shall be no more than the process window for the regional expansion planning period (or in the absence of a regional planning process, the process window for Transmission Provider's expansion planning period) not to exceed seven years from the date the Interconnection Request is received by Transmission Provider, unless Interconnection Customer demonstrates that engineering, permitting and construction of the new Large Generating Facility or increase in capacity of the existing Generating Facility will take longer than the regional expansion planning period. The In-Service Date may succeed the date the Interconnection Request is received by Transmission Provider by a period up to ten years, or longer where Interconnection Customer and Transmission Provider agree, such agreement not to be unreasonably withheld.

3.4.2 Acknowledgment of Interconnection Request

Transmission Provider shall acknowledge receipt of the Interconnection Request within five (5) Business Days of the close of the DISIS Request Window and attach a copy of the received Interconnection Request to the acknowledgement.

3.4.3 Deficiencies in Interconnection Request

An Interconnection Request will not be considered to be a valid request until all items in Section 3.4.1 have been received by Transmission Provider. If an Interconnection Request fails to meet the requirements set forth in Section 3.4.1, Transmission Provider shall notify Interconnection Customer within five (5) Business Days of the close of the DISIS Request Window of the reasons for such failure and that the Interconnection Request does not constitute a valid request. Interconnection Customer shall provide Transmission Provider the additional requested information needed to constitute a valid request within ten (10) Business Days after receipt of such notice. At any time, if Transmission Provider identifies issues with technical data provided by Interconnection Customer,

Interconnection Customer and Transmission Provider shall work expeditiously and in good faith to remedy any data issues. If the data issues are remedied by the Interconnection Customer without adversely affecting the Transmission Provider's study timeline, the remedy will be accommodated. Failure by Interconnection Customer to comply with this Section 3.4.3 shall be treated in accordance with Section 3.7.

Transmission Provider shall determine if the information contained in the Interconnection Request is adequately sufficient to start the Definitive System Impact Study by the close of the Customer Engagement Window.

3.4.4 Scoping Meeting

Within ten (10) Business Days after the close of the DISIS Request Window, Transmission Provider shall host a Scoping Meeting for all Interconnection Requests received in that DISIS Request Window. If requested by Interconnection Customer, Transmission Provider shall also hold individual customer specific Scoping Meetings, which must be requested no later than fifteen (15) business days after the close of the DISIS Request Window.

The purpose of the Scoping Meeting shall be to discuss alternative interconnection options, to exchange information including any transmission data that would reasonably be expected to impact such interconnection options, to analyze such information and to determine the potential feasible Points of Interconnection. Transmission Provider and Interconnection Customer will bring to the meeting such technical data, including, but not limited to: (i) general facility loadings, (ii) general instability issues, (iii) general short circuit issues, (iv) general voltage issues, and (v) general reliability issues as may be reasonably required to accomplish the purpose of the meeting. Transmission Provider and Interconnection Customer will also bring to the meeting personnel and other resources as may be reasonably required to accomplish the purpose of the meeting in the time allocated for the meeting. On the basis of the meeting, Interconnection Customer shall designate its Point of Interconnection. The duration of the meeting shall be sufficient to accomplish its purpose.

3.5 OASIS Posting

3.5.1 OASIS Posting

Transmission Provider will maintain on its OASIS a list of all Interconnection Requests. The list will identify, for each Interconnection Request: (i) the maximum summer and winter megawatt electrical output; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the status of the Interconnection Request, including Queue Position; (vi) the type of Interconnection Service being requested; (vii) the availability of any studies related to the Interconnection Request; (viii) the date of the Interconnection Request; (ix) the type of Generating Facility to be constructed (combined cycle, base load or combustion turbine and fuel type); and (x) for Interconnection Requests that have not resulted in a completed interconnection, an explanation as to why it was not completed. Except in the case of an Affiliate, the list will not disclose the identity of Interconnection Customer until Interconnection Customer executes an LGIA or requests that Transmission Provider file an unexecuted LGIA with FERC. Before holding a Scoping Meeting with its Affiliate, Transmission Provider shall post on OASIS an advance notice of its intent to do so. Transmission Provider shall post to its OASIS site any deviations from the study timelines set forth herein. Interconnection Study reports shall be posted to Transmission Provider's OASIS site subsequent to the meeting between Interconnection Customer and Transmission Provider to discuss the applicable study results. Transmission Provider shall also post any known deviations in the Large Generating Facility's In-Service Date.

3.5.2 Requirement to Post Interconnection Study Metrics

Transmission Provider will maintain on its OASIS or its website summary statistics related to processing Interconnection Studies pursuant to Interconnection Requests, updated quarterly. If Transmission Provider posts this information on its website, a link to the information must be provided on Transmission Provider's OASIS site. For each calendar quarter, Transmission Providers must calculate and post the information detailed in sections 3.5.2.1 through 3.5.2.3. Where an Interconnection Customer is a member of a Queue Cluster and such customer tenders to Transmission Provider an executed study agreement before the close of the Queue Cluster Window, Transmission Provider will deem its date of receipt of the study agreement to be the day immediately after the close of the Queue Cluster Window for purposes of calculating and posting processing time under Sections 3.5.2.1 and 3.5.2.2.

3.5.2.1 Interconnection System Impact Studies processing time.

- (A) Number of Interconnection Requests that had Phase 1 Studies completed within Transmission Provider's coordinated region during the reporting quarter, and the number of Interconnection Requests that had Phase 2 and Phase 3 Studies completed within Transmission Provider's coordinated region during the reporting quarter,
- (B) Number of Interconnection Requests that had Phase 1 Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than ninety (90) Calendar Days after the close of the Queue Cluster Window, or more than ninety (90) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact Study Agreements, whichever is later, and the number of Interconnection Requests that had Phase 2 and Phase 3 Studies completed within Transmission Provider's coordinated region during the reporting quarter that were completed more than one hundred fifty (150) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection System Impact Study Agreements,
- (C) At the end of the reporting quarter, the number of active valid Interconnection Requests with ongoing incomplete System Impact Studies where such Interconnection Requests had executed Definitive Interconnection System Impact Study Agreements received by Transmission Provider more than one hundred fifty (150)Calendar Days after the close of the Queue Cluster Window, or more than one hundred fifty (150) Calendar Days after receipt by Transmission Provider of all Interconnection Customers' executed Interconnection Impact Study Agreements, whichever is later, before the reporting quarter end,
- (D) Mean time (in days), Definitive Interconnection System Impact Studies completed within Transmission Provider's

coordinated region during the reporting quarter, from the date when Transmission Provider received the executed Interconnection System Impact Study Agreement(s) to the date when Transmission Provider provided the completed Interconnection System Impact Study to the Interconnection Customer(s),

(E) Percentage of Interconnection System Impact Studies exceeding the timelines set forth above to complete this reporting quarter, calculated as the sum of 3.5.2.1(B) plus 3.5.2.1(C) divided by the sum of 3.5.2.1(A) plus 3.5.2.1(C)).

3.5.2.2 Interconnection Facilities Studies processing time.

- (A) Number of Interconnection Requests that had Interconnection Facilities Studies that are completed within Transmission Provider's coordinated region during the reporting quarter and tendered to the Interconnection Customer in draft form,
- (B) Number of Interconnection Requests that had Interconnection Facilities Studies that are completed within Transmission Provider's coordinated region during the reporting quarter that were completed and tendered to the Interconnection Customer in draft form more than ninety (90) Calendar Days after receipt by Transmission Provider of the Interconnection Customer's executed Interconnection Facilities Study Agreement,
- (C) At the end of the reporting quarter, the number of active valid Interconnection Service requests with ongoing incomplete Interconnection Facilities Studies where such Interconnection Requests had executed Interconnection Facilities Studies Agreement received by Transmission Provider more than ninety (90) Calendar Days before the reporting quarter end,
- (D) Mean time (in days), for Interconnection Facilities Studies completed within Transmission Provider's coordinated region during the reporting quarter, calculated from the date when Transmission Provider received the executed Interconnection Facilities Study Agreement to the date when Transmission Provider provided the completed

draft Interconnection Facilities Study to the Interconnection Customer,

(E) Percentage of delayed Interconnection Facilities Studies this reporting quarter, calculated as the sum of 3.5.2.2(B) plus 3.5.2.2(C) divided by the sum of 3.5.2.2(A) plus 3.5.2.2(C)).

3.5.2.3 Interconnection Service requests withdrawn from Interconnection queue.

- (A) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter,
- (B) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of any interconnection studies or execution of any interconnection study agreements,
- (C) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of an Interconnection System Impact Study,
- (D) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue during the reporting quarter before completion of an Interconnection Facilities Study,
- (E) Number of Interconnection Requests withdrawn from Transmission Provider's interconnection queue after execution of a generator interconnection agreement or Interconnection Customer requests the filing of an unexecuted, new interconnection agreement,
- (F) Mean time (in days), for all withdrawn Interconnection Requests, from the date when the request was determined to be valid to when Transmission Provider received the request to withdraw from the queue.

Transmission Provider is required to post on OASIS or its website the measures in paragraph 3.5.2.1(A) through paragraph 3.5.2.3(F) for each calendar quarter within 30 days of the end of the calendar quarter. Transmission Provider will keep the quarterly measures posted on OASIS or its website for three calendar years with the first required report to be in the first quarter of 2020. If Transmission Provider retains this information on its website, a link to the information must be provided on Transmission Provider's OASIS site.

3.5.4

In the event that any of the values calculated in paragraphs 3.5.2.1(E), or 3.5.2.2(E) exceeds 25 percent for two consecutive calendar quarters, Transmission Provider will have to comply with the measures below for the next four consecutive calendar quarters and must continue reporting this information until Transmission Provider reports four consecutive calendar quarters without the values calculated in 3.5.2.1(E), or 3.5.2.2(E) exceeding 25 percent for two consecutive calendar quarters:

- (i) Transmission Provider must submit a report to the Commission describing the reason for each study or group of clustered studies pursuant to an Interconnection Request that exceeded its deadline (i.e., 90, 150 or 90 days) for completion (excluding any allowance for Reasonable Efforts). Transmission Provider must describe the reasons for each study delay and any steps taken to remedy these specific issues and, if applicable, prevent such delays in the future. The report must be filed at the Commission within 45 days of the end of the calendar quarter.
- (ii) Transmission Provider shall aggregate the total number of employee-hours and third party consultant hours expended towards interconnection studies within its coordinated region that quarter and post on OASIS or its website. If Transmission Provider posts this information on its website, a link to the information must be provided on Transmission Provider's OASIS site. This information is to be posted within 30 days of the end of the calendar quarter.

3.6 Coordination with Affected Systems

Transmission Provider will coordinate the conduct of any studies required to determine the impact of the Interconnection Request on Affected Systems with Affected System Operators and, if possible, include those

results (if available) in its applicable Interconnection Study within the time frame specified in this LGIP. Transmission Provider will include such Affected System Operators in all meetings held with Interconnection Customer as required by this LGIP. Interconnection Customer will cooperate with Transmission Provider in all matters related to the conduct of studies and the determination of modifications to Affected Systems. A Transmission Provider which may be an Affected System shall cooperate with Transmission Provider with whom interconnection has been requested in all matters related to the conduct of studies and the determination of modifications to Affected Systems. It is the responsibility of the Affected System owner to provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to (i) complete any interconnection studies and (ii) construct any necessary interconnection facilities and network upgrades needed to reliably interconnect at the requested service level.

3.7 Withdrawal

Interconnection Customer may withdraw its Interconnection Request at any time by written notice of such withdrawal to Transmission Provider. In addition, if Interconnection Customer fails to adhere to all requirements of this LGIP, except as provided in Section 13.5 (Disputes), Transmission Provider shall deem the Interconnection Request to be withdrawn and shall provide written notice to Interconnection Customer of the deemed withdrawal and an explanation of the reasons for such deemed withdrawal. Upon receipt of such written notice, Interconnection Customer shall have fifteen (15) Business Days in which to either respond with information or actions that cures the deficiency or to notify Transmission Provider of its intent to pursue Dispute Resolution.

Withdrawal shall result in the loss of Interconnection Customer's Queue Position. If Interconnection Customer disputes the withdrawal and loss of its Queue Position, then during Dispute Resolution, Interconnection Customer's Interconnection Request is eliminated from the Queue until such time that the outcome of Dispute Resolution would restore its Queue Position. An Interconnection Customer that withdraws or is deemed to have withdrawn its Interconnection Request shall pay to Transmission Provider all costs that Transmission Provider prudently incurs with respect to that Interconnection Request prior to Transmission Provider's receipt of notice described above. Interconnection Customer must pay all monies due to Transmission Provider before it is allowed to obtain any Interconnection Study data or results.

In the case of a withdrawal, Transmission Provider shall (i) update the OASIS Queue Position posting; (ii) impose the Withdrawal Penalty described in Section 3.7.1; (iii) refund any security after settling the final invoice (see Section 7.7.5); and (iv) refund to Interconnection Customer any of the refundable portion of Interconnection Customer's study deposit that exceeds the share of the costs that Transmission Provider has incurred (less any non-refundable amounts described in this LGIP). The application of interest on amounts subject to refund shall be made in a manner consistent with Commission precedent and shall be calculated in accordance with section 35.19a(a)(2) of FERC's regulations. In the event of such withdrawal, Transmission Provider, subject to the confidentiality provisions of Section 13.1, shall provide, at Interconnection Customer's request, all information that Transmission Provider developed for any completed study conducted up to the date of withdrawal of the Interconnection Request.

3.7.1 Withdrawal Penalty

Interconnection Customers shall be subject to a Withdrawal Penalty if they withdraw their request from the Queue or the Generating Facility does not otherwise reach Commercial Operation unless the withdrawal does not negatively affect the timing or cost of equal or lower queued projects.

3.7.1.1 Calculation of the Withdrawal Penalty.

If the Interconnection Customer provided a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be equal to the higher of the study deposit or one (1) times of its actual allocated cost of the Definitive Interconnection Study Process. If the Interconnection Customer did not provide a demonstration of readiness, that Interconnection Customer's Withdrawal Penalty shall be dependent on the Phase in which the Interconnection Customer withdraws and shall be calculated as follows: 1. If the Interconnection Customer withdraws in Phase 1 (after M1, but before M2), the Withdrawal Penalty shall be the higher of the study deposit or two (2) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one (1) million dollars. 2. If the Interconnection Customer withdraws in Phase 2 (after M2, but before M3), the Withdrawal Penalty shall be the higher of the study deposit or three (3) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at one and one half

(1.5) million dollars. 3. If the Interconnection Customer withdraws in Phase 3 (after M3, but before M4), the study cost obligation shall be the higher of the study deposit or five (5) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at two (2) million dollars. 4. If the Interconnection Customer withdraws in Phase 4 (after M4, but before M5), the Withdrawal Penalty shall be the higher of the study deposit or seven (7) times its actual allocated cost of the Definitive Interconnection Study Process. This amount shall be capped at two and a half (2.5) million dollars.

The Withdrawal Penalty for any customer that has executed an LGIA is the higher of the study deposit or nine (9) times its actual allocated cost of the Definitive Interconnection Study Process.

3.7.1.2 Distribution of the Withdrawal Penalty.

Any Withdrawal Penalty revenues shall be used to fund generation interconnection studies. Withdrawal Penalty revenues shall first be applied, in the form of a bill credit, to not-yet-invoiced study costs for other Interconnection Customers in the same cluster, and to the extent that such studies are fully credited, shall be applied to study costs of future clusters in Queue order. Withdrawn Interconnection Customers shall not receive a bill credit associated with Withdrawal Penalties. Distribution of Withdrawal Penalty revenues to a specific study shall not exceed the total actual study costs. Allocation of Withdrawal Penalty revenues within a cluster to a specific customer shall be comparable to the allocation of study costs described in Section 4.2. Specifically, the Withdrawal Penalty revenue distribution to each customer in a specific cluster, shall be (1) fifty percent (50%) on a per capita basis based on number of Interconnection Requests in the applicable Cluster; and (2) fifty percent (50%) to Interconnection Customers on a pro-rata basis based on requested megawatts included in the applicable Cluster. Distribution of Withdrawal Penalty revenue associated with Readiness Milestone 5 shall not be distributed to the remaining customers in that cluster until all customers in that cluster have reached Commercial Operation and thereafter shall be distributed as described above. Transmission Provider shall not change the distribution of Withdrawal Penalty revenue without authorization by the Commission. Transmission Provider shall post the Withdrawal Penalty balance on its OASIS site.

3.8 Identification of Contingent Facilities.

The Transmission Provider shall identify Contingent Facilities in the System Impact Study Report (including any restudy reports) using the steps outlined below. The method permits the parties to determine why a specific Contingent Facility was identified and how it relates to the Interconnection Request.

3.8.1 Baseline assumptions.

Transmission Provider uses a technical screening process to identify Contingent Facilities that starts with the baseline assumption that the following are in service: (i) Generating Facilities that are directly interconnected to the Transmission System; (ii) Generating Facilities that are interconnected to Affected Systems where the Affected System(s) have communicated to Transmission Provider, or Transmission Provider otherwise has determined, that such facilities may have an impact on the Interconnection Request; (iii) Generating Facilities that have a pending higher-queued Interconnection Request to interconnect to the Transmission System and their associated Interconnection Facilities and Network Upgrade requirements, to the extent those higher-queued Interconnection Requests have been the subject of a System Impact Study and/or a Facilities Study; (iv) Generating Facilities that executed an interconnection agreement, or requested that an unexecuted interconnection agreement be filed with FERC, and their associated Interconnection Facilities and Network Upgrades; (v) higher-queued requests for transmission service and their associated facilities or upgrade requirements to the extent they have an impact on the Interconnection Request and to the extent that those higher-queued requests for transmission service have been the subject of a System Impact Study and/or a Facilities Study; (vi) Transmission Provider's transmission expansion plan components; and (vii) the transmission expansion plan components of third-party transmission providers, to the extent that Transmission Provider has determined that they have an impact on the Interconnection Request.

With respect to the treatment of higher-queued requests for interconnection service and/or transmission service, in situations in which (a) the higher-queued requests have not yet proceeded through the System Impact Study process to reach the stage where facilities necessary to accommodate the requests have been identified, or (b) facilities associated with higher-queued requests for service change as a result of queue withdrawal or otherwise, Transmission Provider

shall adjust its baseline assumptions with the most current information available and produce a re-study System Impact Study Report on Interconnection Customer's Interconnection Request. The re-study report will use the method described in this Section to permit the parties to determine why a specific Contingent Facility was identified and how it relates to the Interconnection Request.

3.8.2. Technical Screening Process.

The technical screening process for identifying Contingent Facilities is comprised of the following steps:

Step 1: Identify Potential Contingent Facilities.

Transmission Provider will review all applicable Interconnection and Transmission Service study results for higher-queued Interconnection or Transmission Service Requests to identify any unbuilt Interconnection Facilities and/or Network Upgrades as potential Contingent Facilities to be evaluated pursuant to Steps 2-5 below.

Step 2: Remove a Potential Contingent Facility and Perform **Applicable Contingency Analyses.** The Transmission Provider will take a potential Contingent Facility (and its associated unbuilt Generating Facility) out of service in its study model and: (a) perform steady state, short circuit, voltage stability, and/or transient stability analyses to determine if the Transmission System demonstrates acceptable pre- and post-contingency system performance, in accordance with current Transmission Provider, WECC, NERC, or Reliability Coordinator criteria or standards; and (b) document the resulting Transmission System performance deficiencies following the analysis in Step 2(a). In implementing this step in situations in which higher-queued clustered requests involve a potential Contingent Facility, Transmission Provider will remove all unbuilt Interconnection Facilities and/or Network Upgrades in the cluster, as well as their associated Generating Facilities in the cluster.

Step 3: Add the proposed Generating Facility into Model and Rerun Contingency Analyses. Transmission Provider will add the proposed Generating Facility into the model after taking the potential Contingent Facility and its associated Generating Facility out of service as provided in Step 2 above, and: (a) perform the same analysis for the added proposed Generating Facility as the analysis outlined in Step 2(a) for the removed potential Contingent Facility;

and (b) document the resulting Transmission System performance deficiencies following the analysis in Step 3(a).

Step 4: Apply Threshold and Categorize. If the Transmission System performance deficiencies observed in Step 3(b) are: (a) exacerbated by one percent (1%) or greater than the Transmission System performance deficiencies initially observed in Step 2(b), then the potential Contingent Facility that is individually evaluated in Step 2 will be deemed a Contingent Facility; or (b) exacerbated by less than one percent (1%) than the Transmission System performance deficiencies initially observed in Step 2(b), then the potential Contingent Facility that is individually evaluated in Step 2 will not be deemed a Contingent Facility. The only variation from this analysis will apply to short circuit criteria. For the performance of the short circuit analysis of the Transmission Provider's Transmission System in its Balancing Authority Area, all generation directly connected to the Transmission Provider's Transmission System is assumed to be connected and synchronized to the grid. The fault current at any substation that contains one or more circuit breakers exceeding 95% of the interruption rating of any circuit breaker in the substation where the fault is taken will be considered a Contingent Facility.

- Step 5: Repeat for Each Identified Potential Contingent Facility. Transmission Provider will repeat Steps 2-4 for each potential Contingent Facility identified in Step 1.
- **Step 6: Per Se Contingent Facilities.** Notwithstanding Steps 1-5, an Interconnection Facility or Network Upgrade of a higher-queued request for service shall automatically be deemed a Contingent Facility if such Interconnection Facility or Network Upgrade would be necessary for the proper functioning of the proposed Generating Facility's System Protection Facilities (as defined in Appendix 6 to Attachment M of Transmission Provider's OATT).
- **3.8.3.** The Interconnection System Impact Study report will list Contingent Facilities in an appendix, which will include: (a) a description of each Contingent Facility; and (b) the Interconnection Request, transmission service request or planned project for which the Contingent Facility was initially required. This list of Contingent Facilities is subject to updates if a System Impact Study is re-studied pursuant to the Tariff under the LGIP or the provisions governing transmission service requests. In addition, where the Transmission

Provider has identified an Affected System pursuant to Section 3.6 and facilities have been identified to mitigate adverse impacts on an Affected System, such facilities shall be included on the list of Contingent Facilities to the extent they have an impact on the Interconnection Request.

3.8.4. If requested by the Interconnection Customer, and if readily available and not commercially sensitive, Transmission Provider will also provide an estimate of the costs of and the in-service date for each Contingent Facility, which may be subject to later updates if a Contingent Facility's estimated costs and in-service dates change.

Section 4. Interconnection Request Evaluation Process

4.1 Queue Position

4.1.1 Assignment of Queue Postion

Transmission Provider shall assign a Queue Position as follows: the Queue Position within the Queue shall be assigned based upon the date and time of receipt of all items required pursuant to the provisions of Section 3.4. There is no queue for informational Interconnection Studies.

4.1.2 Higher Queue Position

A higher Queue Position assigned to an Interconnection Request is one that has been placed "earlier" in the Queue in relation to another Interconnection Request that is assigned a lower Queue Position. All requests studied in a Cluster shall be considered equally queued but Clusters initiated earlier in time shall be considered to have a higher Queue Position than clusters initiated later. The Queue Position of an Interconnection Request shall have no bearing on the allocation of the cost of the upgrades identified in the applicable Cluster Study (such costs will be allocated among Interconnection Requests in accordance with Section 4.2.2. Moving a Point of Interconnection shall result in a loss of Queue Position if it is deemed a Material Modification under Section 4.4.3.

4.2 Clustering and the Definitive Study Process

The diagram attached in the At-A-Glance Reference Sheet provides an overview of the Definitive Interconnection Study Process.

Transmission Provider shall accept Interconnection Requests during an approximate sixty (60) Calendar Day period referred to as the "DISIS Request Window." A DISIS Request Window shall open annually on or about February 1st and close on March 31th or the following Business Day if March 31th falls on a weekend or NERC recognized holiday. A second DISIS Request Window shall open annually on August 1st and close on September 30th or the following Business Day if September 30th falls on a Saturday or Sunday.

If one or more valid requests are received, for sixty (60) Calendar Days following the close of the DISIS Request Window (the "Customer Engagement Window"), Transmission Provider shall work with applicable Interconnection Customers to test models, verify data, hold stakeholder meetings (including Scoping Meetings, as appropriate), work with requestors to address any modeling irregularities identified by Transmission Provider in its evaluation of the Interconnection Request, and generally prepare for the start of the Definitive Interconnection System Impact Study. Notwithstanding the preceding sentence and upon written consent of all Interconnection Requests for a specific Cluster, Transmission Provider may shorten the "Customer Engagement Window" in order to start the Definitive Interconnection System Impact Study earlier. Within the first ten (10) Business Days following the close of the DISIS Request Window, Transmission Provider shall post on its OASIS site a list of Interconnection Requests for that Cluster. The list shall identify, for each Interconnection Request: (i) the requested amount of Interconnection Service; (ii) the location by county and state; (iii) the station or transmission line or lines where the interconnection will be made; (iv) the projected In-Service Date; (v) the type of Interconnection Service (vi) cluster being requested; and (vi) the type of Generating Facility to be constructed including fuel type such as wind, natural gas, coal, or solar.

At the end of the Customer Engagement Window, all Interconnection Requests deemed sufficient that have an executed DISIS Agreement shall be included in that DISIS Cluster. Any Interconnection Requests not deemed sufficient or undergoing Dispute Resolution at the close of the Customer Engagement Window shall not be included in that DISIS Cluster. Immediately following the Customer Engagement Window, Transmission Provider shall initiate the Definitive Interconnection System Impact Study described in more detail in Section 7.

Transmission Provider shall determine each Interconnection Customer's share of the DISIS Cluster Study by allocating the applicable study costs to Interconnection Customers on a per capita basis based on number of

Interconnection Requests included in the applicable Cluster. The Interconnection Facilities Study portion of the Definitive Interconnection Study Process is handled in the same manner. The Interconnection Facilities Study is a cluster study, and study costs are allocated to the participating Interconnection Customers in the same way as in the DISIS Cluster study.

4.2.1 Initiation of a Resource Solicitation Cluster

Upon request of a Resource Planning Entity, Transmission Provider may initiate the study of a Resource Solicitation Cluster. The Resource Solicitation Cluster shall respect Queue Position and shall be studied as its own Cluster. Within ten (10) Business Days of receipt of a request to perform a Resource Solicitation Cluster that includes valid Interconnection Requests as described in Section 3.4, Transmission Provider and Resource Planning Entity shall meet to determine a mutually agreeable scope of study and timeframe to initiate the Resource Solicitation Cluster. The timeline shall indicate the close of the Customer Engagement Window for that Resource Solicitation Cluster. Thereafter the Definitive Interconnection System Impact Study shall proceed as described in Section 7.

In order to initiate Transmission Provider's study of Interconnection Requests made in connection with a Resource Solicitation Process Resource Planning Entity must: (a) act as the authorized representative for all Interconnection Requests submitted to the Resource Solicitation Cluster; (b) submit all Interconnection Requests arising from the Resource Solicitation Process at the same time to ensure an equal Queue Position for all Generating Facilities included in the Resource Solicitation Cluster; (c) cooperate with Transmission Provider in conducting the studies; and (d) request a reasonable number of different combinations of such Interconnection Requests to meet Resource Planning Entity's identified need and assumptions in the Resource Solicitation Process. Such studies in connection with a Resource Solicitation Process shall be implemented based upon Queue Position (relative to higher or lower queued clusters) and shall consider Resource Planning Entity's needs and assumptions identified in the Resource Solicitation Process. Transmission Provider shall not be obligated to implement a Resource Solicitation Cluster more than once a year.

The Resource Planning Entity may submit for inclusion in the Resource Solicitation Process an Interconnection Request for a Generating Facility that already has a higher Queue Position pursuant to Section 4.1.1. A Generating Facility that initially is associated with a Queue Position through the Resource Solicitation Process may also reserve a lower Queue Position separate from the Resource Solicitation Process pursuant to Section 4.1.1. In either case, Interconnection Customer must meet all requirements associated with maintaining each Queue Position for the Generating Facility. In the event a Generating Facility has multiple Queue Positions, it shall not be double counted in the study models.

A Generating Facility in the Resource Solicitation Process is subject to study according to the Queue Position of the Resource Solicitation Cluster. A Generating Facility that is not a part of the Resource Solicitation Process is also subject to study according to its Queue Position. All studies must be performed in accordance with the provisions of the LGIP, and may not be delayed as a result of the Resource Solicitation Process.

After Transmission Provider completes the Definitive Interconnection System Impact Studies for the requested combinations, the results will be provided (Phase 1 Reports, Phase 2 Reports, Phase 3 Reports, etc.; as applicable under Section 7.4) to the Resource Planning Entity for use in the Resource Solicitation Process. The results will be posted on Transmission Provider's OASIS consistent with the posting of other study results.

After receipt of the Phase 1 Report, Resource Planning Entity must select one of the studied combinations to advance to Phase 2 of the study process. After receipt of the Phase 2 Report, Resource Planning Entity must select one of the studied combinations prior to the commencement of any Interconnection Facilities Study associated with the Resource Solicitation Process. Prior to the completion of the Interconnection Facilities Study of all of the components of the selected combination, Resource Planning Entity may replace components, subject to any necessary Re-Study pursuant to Sections 7.6 or 8.5. While conducting the Definitive Interconnection Study Process, Transmission Provider may suspend further action on the Interconnection Requests in the Resource Solicitation Process that are not included in the selected combination. Once a Generating Facility is rejected in the Resource Solicitation Process, the Generating Facility shall lose the Queue Position it held as part of the Resource Solicitation Process. If a Generating Facility is selected by Resource Planning Entity at the conclusion of the Resource Solicitation Process, the Generating

Facility may no longer maintain more than one Queue Position.

4.2.2 Network Upgrades Cost Allocation

For Network Upgrades identified in a Queue Cluster study, the Transmission Provider shall calculate each Interconnection Customer's share of Network Upgrade costs in the following manner:

- (a) The costs for station equipment, including all switching stations, will be allocated on a *pro rata* basis based on the number of Generating Facilities interconnecting at an individual station.
- (b) The costs for all transmission lines, transformers and voltage-support related Network Upgrades will be allocated on a *pro rata* basis based on the proportional capacity of each individual Generating Facility in the Queue Cluster requiring such Network Upgrades.

4.3 Transferability of Queue Position

An Interconnection Customer may transfer its Queue Position to another entity only if such entity acquires the specific Generating Facility identified in the Interconnection Request and the Point of Interconnection does not change.

4.4 Modifications

Interconnection Customer shall submit to Transmission Provider, in writing, modifications to any information provided in the Interconnection Request. Interconnection Customer shall retain its Queue Position if the modifications are in accordance with Sections 4.4.1, 4.4.2 or 4.4.5, or are determined not to be Material Modifications pursuant to Section 4.4.3.

Notwithstanding the above, during the course of the Interconnection Studies, either Interconnection Customer or Transmission Provider may identify changes to the planned interconnection that may improve the costs and benefits (including reliability) of the interconnection, and the ability of the proposed change to accommodate the Interconnection Request. Subject to the foregoing sentence, and provided, however, they do not result in a material modification, to the extent the identified changes are acceptable to Transmission Provider, Interconnection Customer and potentially impacted Interconnection Customers in the same Cluster, such acceptance not to be unreasonably withheld, Transmission Provider shall modify the Point of

Interconnection and/or configuration in accordance with such changes and proceed with any re-studies necessary to do so in accordance with Section 7.6 and Section 8.5 as applicable and Interconnection Customer shall retain its Queue Position.

- A.4.1 No later than thirty (30) Calendar Days after the close of the DISIS Request Window and prior to the return of the executed Definitive Interconnection System Impact Study Agreement to Transmission Provider, the modifications permitted under this Section shall include specifically: (a) a decrease of up to 60 percent of electrical output (MW) of the proposed project, through either (1) a decrease in plant size or (2) a decrease in Interconnection Service level (consistent with the process described in Section 3.1) accomplished by applying Transmission Provider-approved injection-limiting equipment; (b) modifying the technical parameters associated with the Large Generating Facility technology or the Large Generating Facility step-up transformer impedance characteristics; and (c) modifying the interconnection configuration. For plant increases, the incremental increase in plant output will go to the end of the Queue for the purposes of cost allocation and study analysis.
- **4.4.2** Prior to the return of the executed Interconnection Facilities Study Agreement to Transmission Provider, the modifications permitted under this Section shall include specifically: (a) additional 15 percent decrease of electrical output of the proposed project through either (1) a decrease in plant size (MW) or (2) a decrease in Interconnection Service level (consistent with the process described in Section 3.1) accomplished by applying Transmission Provider-approved injection-limiting equipment; (b) Large Generating Facility technical parameters associated with modifications to Large Generating Facility technology and transformer impedances; provided, however, the incremental costs associated with those modifications are the responsibility of the requesting Interconnection Customer; and (c) a Permissible Technological Advancement for the Large Generating Facility after the submission of the Interconnection Request. Section 4.4.6 specifies a separate technological change procedure including the requisite information and process that will be followed to assess whether the Interconnection Customer's proposed technological advancement under Section 4.4.2(c) is a Material Modification. Section 1 contains a definition of Permissible Technological Advancement.

- 4.4.3 Prior to making any modification other than those specifically permitted by Sections 4.4.1, 4.4.2, and 4.4.6, Interconnection Customer may first request that Transmission Provider evaluate whether such modification is a Material Modification. In response to Interconnection Customer's request, Transmission Provider shall evaluate the proposed modifications prior to making them and inform Interconnection Customer in writing of whether the modifications would constitute a Material Modification. Any change to the Point of Interconnection, except those deemed acceptable under Sections 4.4.1, or so allowed elsewhere, shall constitute a Material Modification. Interconnection Customer may then withdraw the proposed modification or proceed with a new Interconnection Request for such modification.
- **4.4.4** Upon receipt of Interconnection Customer's request for modification permitted under this Section 4.4, Transmission Provider shall commence and perform any necessary additional studies as soon as practicable, but in no event shall Transmission Provider commence such studies later than thirty (30) Calendar Days after receiving notice of the modification of Interconnection Customer's request. Any additional studies resulting from such modification shall be done at Interconnection Customer's cost.
- 4.4.5 Extensions of less than three (3) cumulative years in the Commercial Operation Date of the Generating Facility to which the Interconnection Request relates are not material and should be handled through construction sequencing; provided, however, that extensions may necessitate a determination of whether additional studies are required pursuant to Applicable Laws and Regulations and Applicable Reliability Standards. The initial requested Commercial Operation Date used for this calculation is determined from the date proposed in the initial Interconnection Request. Such cumulative extensions are inclusive of extensions requested after execution by Interconnection Customer of the LGIA.

4.4.6 Technological Change Procedure.

4.4.6.1 Interconnection Customer Technological Advancement Request.

(a) At any time after the submission of an Interconnection Request, but before the execution of an Interconnection Facility Study Agreement by Interconnection Customer, an Interconnection Customer may submit a written request pursuant to this Section to include additional or substituted technological components for its Large Generating Facility that differ from the description of the Large Generating Facility in its Interconnection Request.

(b) To timely perfect its Technological Advancement Request, Interconnection Customer shall submit the following to Transmission Provider: (i) completed Technological Advancement Request form submitted on the request template provided by Transmission Provider on its OASIS site; (ii) a \$25,000 deposit; (iii) an updated version of the Interconnection Request for a Large Generating Facility, found at Appendix 1 of this LGIP, that reflects the data associated with the change in technology that Interconnection Customer seeks to incorporate; (iv) to the extent applicable, updated modeling data.

Updated modeling data is applicable if the technology change results in a change in the time of use of the facility, i.e., from peak to off peak, or capacity factor of the facility; a change in the steady-state thermal and/or voltage limits of the facility; a change in harmonics; a possible increase in the short circuit capability of the facility; or a possible change in the transient or dynamic response of the facility.

Where a technology change does not fall into one of the categories above, but is a technology change that may reduce the Network Upgrades previously identified in the interconnection study process associated with the Interconnection Customer's Large Generating Facility, the Transmission Provider may require a Technological Advancement Study using updated modeling data for purposes of implementing the identification of Network Upgrades necessary to accommodate the Interconnection Request under this LGIP and for the implementation of the crediting provisions set forth in Section 11.4 of the LGIA. Among the changes that Transmission Provider may evaluate under this Section 4.4.6 include, but are not limited to, advancements that improve, rather than degrade the electrical characteristics of the generating equipment (e.g., the ratings, impedances, efficiencies, capabilities, and performance of the equipment under steady state and dynamic conditions)

without changing the real power (MW) output or reactive power (MVAR) output of the Large Generating Facility.

- (c) Interconnection Customer's Technological Advancement Request must demonstrate how the proposed technological advancement (i) results in equal to or better electrical performance, (ii) does not increase the Interconnection Customer's requested interconnection service, and (iii) does not cause any reliability concerns (i.e., material impacts to the transmission system, including impacts to short circuit capability limits, steady-state thermal and voltage limits, or dynamic system stability and response).
- (d) An Interconnection Customer may have no more than one Technological Advancement Request pending at any one time. If the Technological Advancement Request is submitted during the time allocated under the LGIP for Interconnection Customer to execute and return a System Impact or Facilities Study Agreement to Transmission Provider, the deadline for execution and return of the System Impact or Facilities Study Agreement will be suspended while Transmission Provider analyzes the Technological Advancement Request in accordance with Section 4.4.6.
- (e) If Transmission Provider is performing an Interconnection System Impact Study, or other study for the Interconnection Request at the time that Interconnection Customer submits a Technological Advancement Request, Transmission Provider shall suspend work on any such pending studies until it has completed its analysis of the Technological Advancement Request and any Technological Advancement Study.
- (f) Interconnection Customer shall fill in, sign and submit, together with its Technological Advancement Request, a Technological Advancement Study Agreement. A form of Technological Advancement Study Agreement is posted on Transmission Provider's OASIS.

4.4.6.2 Initial Analysis of Technological Advancement Request.

(a) After the Interconnection Customer's Technological Advancement Request is received pursuant to Section 4.4.6.1, the Transmission Provider will perform an initial analysis to determine whether the proposed technological advancement

is a Permissible Technological Advancement without the need of additional study.

- (b) If the Transmission Provider determines on the basis of its initial analysis that Interconnection Customer has demonstrated that the proposed technological advancement is a Permissible Technological Advancement without the need for additional study, the Transmission Provider will incorporate the technological advancement into Interconnection Customer's Interconnection Request.
- (c) If the Transmission Provider determines on the basis of its initial analysis that Interconnection Customer has not demonstrated that the proposed technological advancement is a Permissible Technological Advancement, then the Technological Advancement Request will be treated as a request for modification of the Interconnection Request under Section 4.4.3.
- (d) If the Transmission Provider determines on the basis of its initial analysis that further study is required to conclude whether the Technological Advancement Request is a Permissible Technological Advancement, Transmission Provider will require that a Technological Advancement Study be performed at the sole expense of the Interconnection Customer consistent with Sections 4.4.6.3, 4.4.6.4, 4.4.6.5.
- (e) Any difference between the deposit provided under Section 4.4.6.1(b) and the actual cost of providing the initial analysis of the request shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate, along with an invoice describing any charges.

4.4.6.3 Technological Advancement Study Notification:

If after its initial analysis of a Technological Advancement Request, Transmission Provider determines that a Technological Advancement Study is necessary to determine whether the requested technological advancement constitutes a Permissible Technological Advancement, Transmission Provider shall notify Interconnection Customer in writing that such a study is necessary, and shall perform such study pursuant to Section 4.4.6.4.

4.4.6.4 Technological Advancement Study Procedures:

- (a) The Technological Advancement Study shall seek to determine (i) whether the proposed technological advancement is a Permissible Technological Advancement, by focusing on whether the proposed technological advancement will result in equal or better electrical performance than the Large Generating Facility described in the Interconnection Request, and whether the proposed technological advancement will cause any reliability concerns (i.e., material impacts to the transmission system, including impacts to short circuit capability limits, steady-state thermal and voltage limits, or dynamic system stability and response); and (ii) if the proposed technological advancement is determined not to be a Permissible Technological Advancement, whether the proposed technological advancement is a Material Modification. The Technological Advancement Study may include steady-state, reactive power, short circuit/fault duty, stability analyses, and any other appropriate studies that Transmission Provider deems necessary to determine whether the technological advancement results in electrical performance that is equal to or better than the electrical performance expected prior to the technology change, and whether such technological advancement causes any reliability concerns.
- (b) Interconnection Customer shall cooperate with Transmission Provider to provide any additional information that Transmission Provider may require to complete the Technological Advancement Study. If the Transmission Provider determines that it requires additional technical information to complete the Technological Advancement Study, Transmission Provider shall notify the Interconnection Customer of the additional technical information required, and Interconnection Customer shall work in good faith with Transmission Provider to promptly provide such information.
- (c) Upon completion of the Technological Advancement Study, Transmission Provider shall provide Interconnection Customer notice of its study conclusions. Upon request, Transmission Provider shall also provide Interconnection Customer supporting documentation, subject to confidentiality arrangements consistent with Section 13.1.

- (d) If the Technological Advancement Study determines that the proposed technological advancement is either (i) a Permissible Technological Advancement, or (ii) is not a Permissible Technological Advancement but does not constitute a Material Modification, then the Interconnection Request shall be amended to reflect the technological advancement.
- (e) If the Technological Advancement Study determines that the proposed technological advancement is not a Permissible Technological Advancement and also constitutes a Material Modification, Transmission Provider shall provide an explanation for this conclusion. Interconnection Customer may then withdraw the proposed modification or proceed with a new Interconnection Request for such modification.
- (f) Any difference between the deposit provided under Section 4.4.6.1(b) and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate, along with an invoice describing any charges.

4.4.6.5 Time for Completing Initial Analysis and Technological Advancement Study.

Within thirty (30) Calendar Days of receipt of the Interconnection Customer's Technological Advancement Request submitted pursuant to Section 4.4.6.1, Transmission Provider shall complete all analysis and study obligations under this Section 4.4.6 and determine whether the Technological Advancement Request is a Permissible Technological Advancement or Material Modification, subject to Interconnection Customer's performance of its obligations under Section 4.4.6.1(f) and Section 4.4.6.4(b).

4.4.6.6 Treatment of Other Interconnection Studies During and After Technological Advancement Study.

Upon completion of the Transmission Provider's initial analysis of a Technological Advancement Request and any Technological Advancement Study, Transmission Provider

and Interconnection Customer shall amend any existing
Interconnection System Impact Study Agreement, or other
Interconnection Study Agreements as necessary to
incorporate elements of the requested technological
advancement or the results of the Technological
Advancement Study. Transmission Provider may require
additional time or information to complete or re-run studies
that were suspended during the pendency of the
Technological Advancement Request. A single study may be
suspended no more than once as a result of a Technological
Advancement Request. If a subsequent Technological
Advancement Request is received by Transmission Provider
in such circumstances, Transmission Provider will process the
subsequent Technological Advancement Request as soon as it
completes the study that had been previously suspended.

Section 5. Procedures for Interconnection Requests Submitted Prior to Effective Date of Standard Large Generator Interconnection Procedures

5.1 Transition to the First-Ready, First-Served Process

5.1.1 An Interconnection Customer assigned a Queue Position prior to the effective date of this LGIP shall retain that Queue Position subject to the requirements set forth in Sections 5.1.1.1 and 5.1.1.2. An Interconnection Customer that fails to meet these requirements shall have its Interconnection Request deemed withdrawn pursuant to Section 3.7, without prejudice to the Interconnection Customer's ability to request an Informational Interconnection Study for which there is no Queue Position. Any unused deposit amounts of withdrawn Interconnection Requests shall be returned pursuant to Section 3.7. If an Interconnection Customer elects to continue with an Informational Interconnection Study or alternatively with a Transitional Interconnection Facilities Study or a Transitional Cluster Study as described below, Transmission Provider shall retain the current study deposits, and Interconnection Customer shall be responsible for the entire cost of all studies pursuant to Sections 6 and 4.

5.1.1.1 Transitional Projects.

An Interconnection Customer that has a) a final System Impact Study Report that identifies facilities required to feasibly interconnect and b) an Interconnection Facilities Study Agreement that was executed prior to the effective date of this Section 5 may opt to continue with the Interconnection Facilities Study process if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Interconnection Facilities Study Agreement in the form of Appendix 4.1 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All of the following are required:

- a) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades identified in the System Impact Study Report. The deposit shall be equal to one hundred percent (100%) of the costs identified for Transmission Provider's Interconnection Facilities and Network Upgrades in the System Impact Study Report and will be reconciled to actual costs after the associated facilities are in service. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied towards future construction costs described in the LGIA. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, the deposit is fully refundable once the final invoice for study costs and Withdrawal Penalty is settled. The deposit shall be in the form of an irrevocable letter of credit upon which the Transmission Provider may draw or cash.
- b) Exclusive Site Control for the entire Generating Facility and any Interconnection Customer's Interconnection Facilities.
- c) Interconnection Customer shall provide one following:
 - i. A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or
 - ii. Reasonable evidence that the Generating Facility is included in an approved Resource Plan or Resource Solicitation Process, or
 - iii. An executed Provisional Large Generator Interconnection Agreement filed with FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the publication of the final Interconnection Facilities Study Report or the Interconnection Request shall be deemed withdrawn pursuant to Section 3.7 unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of facilities if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's total study cost is imposed.

5.1.1.2 Combined System Impact and Interconnection Facilities Transitional Cluster Study.

An Interconnection Customer with an assigned Queue Position prior to the effective date of this Section 5 may opt to enter the combined system impact and interconnection facilities transitional cluster study ("Transitional Cluster Study") if the Interconnection Customer: (1) meets each of the following requirements that demonstrate readiness; and (2) executes a Transitional Cluster Study Agreement in the form of Appendix 4.2 to the LGIP within thirty (30) Calendar Days of the Effective Date of this LGIP. All Interconnection Requests that enter the Transitional Cluster Study shall be considered to have an equal Queue Position, and identified upgrade costs shall be allocated according to Section 4.2.2 of the LGIP. The Transitional Cluster Study costs shall be allocated according to the method described in Section 4.2. Interconnection Customer may make a one-time extension to its requested Commercial Operation date upon entry into the Transitional Cluster Study and such an extension shall not be past May 1, 2027.

All of the following are required:

a) Choice of requesting either ERIS or NRIS

- b) A deposit on the Transmission Provider's Interconnection Facilities and Network Upgrades expected to be identified in the Transitional Cluster Study. The deposit shall be equal to five million dollars (\$5,000,000) and be in the form of an irrevocable letter of credit upon which the Transmission Provider may draw or cash. If the Interconnection Customer does not withdraw, the deposit shall be reconciled with and applied towards future construction costs described in the LGIA. Any amounts in excess of the actual construction costs shall be returned to the customer. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, the deposit is fully refundable once the final invoice for study costs and Withdrawal Penalty is settled.
- c) Exclusive Site Control for the entire Generating Facility.
- d) Interconnection Customer shall provide one of the following:
 - i. A contract, binding upon the parties to the contract, for sale of the Generating Facility's energy, or the entire constructed Generating Facility; where the term of sale is not less than five (5) years, or
 - ii. Reasonable evidence that the Generating Facility is included in an approved Resource Plan or Resource Solicitation Process, or
 - iii. An executed Provisional Large
 Generator Interconnection Agreement
 filed with FERC that is not in suspension
 with 1) a commitment to construct the
 facility, 2) a Commercial Operation Date
 no later than May 1, 2027 and 3) a
 security deposit in addition to the five

million dollars identified in 5.1.1.2.b where the total security deposit represents a reasonable estimation of the potential costs that could be ultimately allocated to the project in the transitional cluster study.

After the Transitional Cluster Study report is published, the remaining process shall proceed according to Section 11 of this LGIP. All LGIA negotiations shall be completed and the LGIA executed (or filed unexecuted) within sixty (60) Calendar Days of the tender of the draft LGIA or the Interconnection Request is deemed withdrawn unless extended by mutual agreement of Transmission Provider and Interconnection Customer. A change in the Commercial Operation Date shall not delay the construction of Transmission Provider's Interconnection Facilities or Network Upgrades if such delay negatively affects lower or equal queued projects. If the Interconnection Customer withdraws or otherwise does not reach Commercial Operation, a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit (see Section 3.4) is imposed.

5.1.1.3

An Interconnection Customer under an LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 (and not in suspension) that has not built its Large Generating Facility shall have 60 Calendar Days from the effective date of this Section 5.1.1.3 to submit to Transmission Provider a written commitment to (a) start construction of the Generating Facility no later than two (2) years of the effective date of this Section 5.1.1.3 and reach Commercial Operation no later than three (3) years of the effective date of this Section 5.1.1.3 and (b) pay a Withdrawal Penalty equal to nine (9) times the Interconnection Customer's Initial Deposit if it withdraws, terminates its LGIA, or otherwise does not reach Commercial Operation. If it does not timely

make this submission, the LGIA is deemed terminated, and no Withdrawal Penalty shall be assessed. A deemed termination shall be without prejudice to the Interconnection Customer's ability to submit a new Interconnection Request when it is willing and able to satisfy the tariff criteria under the Definitive Interconnection Process.

For any LGIA executed and in effect or made effective unexecuted prior to the effective date of this Section 5 and currently in suspension, the Interconnection Customer shall have 60 Calendar Days from lifting its suspension to make the submission to the Transmission Provider addressed above.

5.2 New Transmission Provider

If Transmission Provider transfers control of its Transmission System to a successor Transmission Provider during the period when an Interconnection Request is pending, the original Transmission Provider shall transfer to the successor Transmission Provider any amount of the deposit or payment with interest thereon that exceeds the cost that it incurred to evaluate the request for interconnection (less any non-refundable amounts). Any difference between such net amount and the deposit or payment required by this LGIP shall be paid by or refunded to the Interconnection Customer, as appropriate. The original Transmission Provider shall coordinate with the successor Transmission Provider to complete any Interconnection Study, as appropriate, that the original Transmission Provider has begun but has not completed. If Transmission Provider has tendered a draft LGIA to Interconnection Customer but Interconnection Customer has not either executed the LGIA or requested the filing of an unexecuted LGIA with FERC, unless otherwise provided, Interconnection Customer must complete negotiations with the successor Transmission Provider.

Section 6. Informational Interconnection Study

6.1 Informational Interconnection Study Agreement.

At any time, a customer may request, and Transmission Provider (either itself or through a consultant) shall perform a reasonable number of Informational Interconnection Studies from May 1 to November 1 each year. An Informational Interconnection Study shall consist of analyses of the type performed in Phase I of the DISIS (power flow and voltage

analyses). Interconnection Customer shall submit a separate Informational Interconnection Request for each site and may submit multiple Informational Interconnection Requests for a single site. Interconnection Customer must submit a deposit with each Informational Interconnection Request even when more than one request is submitted for a single site. An Informational Interconnection Request to evaluate one site at two different voltage levels shall be treated as two Informational Interconnection Requests. The request shall use the form in Appendix 5.1 of the LGIP and shall describe the assumptions that Interconnection Customer wishes Transmission Provider to study within the scope described in Section 6.2 of the LGIP below. Within five (5) Business Days after receipt of a request for an Informational Interconnection Study, Transmission Provider shall provide to Interconnection Customer an Informational Interconnection Study Agreement in the form of Appendix 5.2.

The Informational Interconnection Study Agreement shall: (i) include the scope of work for the Informational Interconnection Study (ii) specify the technical data that Interconnection Customer must provide, (iii) specify the Informational Interconnection Study case and assumptions, and (iv) identify the Transmission Provider's estimate of the cost of the Informational Interconnection Study. To the extent known by Transmission Provider, such estimate shall include any costs expected to be incurred by any Affected System whose participation is necessary to complete the Informational Interconnection Study. Notwithstanding the above, Transmission Provider shall not be required as a result of an Informational Interconnection Study request to conduct any additional Interconnection Studies with respect to any other Interconnection Request.

Interconnection Customer shall execute the Informational Interconnection Study Agreement within ten (10) Business Days of receipt of an agreed upon scope of work and deliver the Informational Interconnection Study Agreement, the technical data, and a study deposit to Transmission Provider in the amount of \$100,000.00.

6.2 Scope of Informational Interconnection Study.

The intent of the Informational Interconnection Study is to aid Interconnection Customer in its business decisions related to interconnection of generation facilities prior to entering the Definitive Interconnection Process. The Informational Interconnection Study shall consist of analysis based on the assumptions and scope of work specified by Interconnection Customer in the Informational Interconnection Study Agreement. The Informational Interconnection Study shall identify the potential Transmission Provider's Interconnection Facilities and the

Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the results and assumptions of the Informational Interconnection Study. The Informational Interconnection Study shall be performed solely for informational purposes. Transmission Provider shall use Reasonable Efforts to coordinate the study with any Affected Systems that may be affected by the types of Interconnection Services that are being studied. Transmission Provider shall utilize existing studies to the extent practicable in conducting the Informational Interconnection Study.

6.3 Informational Interconnection Study Procedures.

The executed Informational Interconnection Study Agreement, the deposit, and technical and other data called for therein must be provided to Transmission Provider within ten (10) Business Days of Interconnection Customer receipt of the Informational Interconnection Request. Transmission Provider shall use Reasonable Efforts to complete the Informational Interconnection Study within 70 Calendar Days from the date of the Informational Interconnection Study Agreement. If Transmission Provider is unable to complete the Informational Interconnection Study within such time period, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required. Any difference between the study payment and the actual cost of the study shall be paid to Transmission Provider or refunded to Interconnection Customer, as appropriate. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation and work papers and databases or data developed in the preparation of the Informational Interconnection Study, subject to confidentiality arrangements consistent with Section 13.1.

Section 7. Definitive Interconnection System Impact Study

7.1 Definitive Interconnection System Impact Study Agreement

Unless otherwise agreed, pursuant to the Scoping Meeting provided for in Section 3.4.4, within thirty (30) Calendar Days acknowledgement of a valid Interconnection Request indicating that a Definitive Interconnection System Impact Study is to be performed, Transmission Provider shall provide to Interconnection Customer a DISIS Agreement in the form of Appendix 2 to this LGIP. The DISIS Agreement shall provide that Interconnection Customer shall compensate Transmission Provider for the actual cost of the DISIS. No later than Fifteen (15) Business Days after the close of the DISIS Request Window, Transmission Provider shall provide to Interconnection Customer a non-binding updated good faith estimate of the cost and timeframe for completing the Definitive

Interconnection System Impact Study.

7.2 Execution of Definitive Interconnection System Impact Study Agreement Interconnection Customer shall execute the DISIS Agreement and deliver the executed DISIS Agreement to Transmission Provider no later than thirty (30) Calendar Days after its receipt.

7.3 Scope of Definitive Interconnection System Impact Study

The Definitive Interconnection System Impact Study shall evaluate the impact of the proposed interconnection on the reliability of the Transmission System. The Definitive Interconnection System Impact Study will consider the Base Case as well as all generating facilities (and with respect to (iii) below, any identified Network Upgrades associated with such higher queued requests) that, on the date the DISIS Request Window closes: (i) are existing and directly interconnected to the Transmission System; (ii) are existing and interconnected to Affected Systems and may have an impact on the Interconnection Request; (iii) have a pending higher queued Interconnection Request to interconnect to the Transmission System; and (iv) have no Queue Position but have executed an LGIA or requested that an unexecuted LGIA be filed with FERC. Generating facilities with pending higher or equal queued NRIS requests, or requests associated with Firm Transmission Service shall generally be modeled at full output while existing generation may be re-dispatched to accommodate new requests in the model. Higher queued ERIS requests or in-service ERIS generators without associated Firm Transmission Service may be dispatched at zero in some study models.

As discussed in more detail in Section 7.4 below, the Definitive Interconnection System Impact Study is a phased study where the first phase (Phase 1) consists of a power flow and voltage analysis that is followed by a phase (Phase 2) that consists of a short circuit analysis and a stability analysis, and may also include another appropriate study (for example, an electromagnetic transient analysis, if appropriate). Any DISIS re-studies (Phase 3) shall consist of a power flow/voltage analysis, a short circuit analysis, and/or a stability analysis as needed. The Definitive Interconnection System Impact Study report will state the assumptions upon which it is based; state the results of the analyses; and provide the requirements or potential impediments to providing the requested interconnection service, including a preliminary indication of the cost and length of time that would be necessary to correct any problems identified in those analyses and implement the interconnection. For purposes of

determining necessary Interconnection Facilities and Network Upgrades, the Definitive Interconnection System Impact Study shall consider the level of Interconnection Service requested by the Interconnection Customer, unless otherwise required to study the full Generating Facility Capacity due to safety or reliability concerns. The Definitive Interconnection System Impact Study will provide a list of facilities that are required as a result of the Interconnection Request and a non-binding good faith estimate of cost responsibility and a non-binding good faith estimated time to construct.

7.4 Definitive Interconnection System Impact Study Procedures

Transmission Provider shall coordinate the Definitive Interconnection System Impact Study with any Affected System that is affected by the Interconnection Request pursuant to Section 3.6 above. Transmission Provider shall utilize existing studies to the extent practicable when it performs the DISIS. Interconnection Requests for DISIS may be submitted only within the DISIS Request Window and Transmission Provider shall initiate the Definitive Interconnection Study Process pursuant to Section 4.2.

The At-A-Glance Reference Sheet attached provides an overview and timeline of the Definitive Interconnection Study Process, including the phases and milestones associated with the Definitive Interconnection System Impact Study.

- a. The DISIS Cluster shall consist of all eligible Interconnection Requests that have executed a DISIS Agreement and have provided all required information before the close of the DISIS Request Window. Transmission Provider shall use Reasonable Efforts to complete the first phase (Phase 1) consisting of a power flow and voltage analysis within ninety (90) Calendar Days. The Phase 1 Report shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades that are expected to be required as a result of the Interconnection Request(s) and a non-binding good-faith indicative level estimate of cost responsibility and a non-binding good-faith estimated time to construct. Transmission Provider shall meet with each cluster participant ("Phase 1 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 1 results on OASIS.
- b. Within thirty (30) Calendar Days of the Phase 1 Report, all Interconnection Customers are required to provide Readiness Milestone 2 ("M2") and continued evidence of Site Control as described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site

- Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.
- Interconnection Customers whose M2 and Site Control are accepted by c. Transmission Provider shall continue in to the second phase ("Phase 2") of the Definitive Interconnection System Impact Study. Phase 2 consists of an updated power flow/voltage analysis (if necessary), stability analysis and short circuit analysis, and any other appropriate study (for example, an electromagnetic transient analysis, if appropriate) for the Interconnection Customers remaining in the DISIS Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 2 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that DISIS Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 2 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 2 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 2 results on OASIS.
- d. Within thirty (30) Calendar Days of the Phase 2 Report, each Interconnection Customer is required to provide Readiness Milestone 3 ("M3") and additional evidence of Site Control described in Section 7.7.7. Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.
 - i. If all Interconnection Customers in the Cluster provide M3 and no Interconnection Customers withdraw from the Queue at this stage, the Definitive Interconnection Study Process advances to the Interconnection Facilities Study (Section 8). Transmission Provider shall electronically notify Interconnection Customers in the Cluster that Phase 3 is not required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.
 - ii. If one or more Interconnection Customer withdraws from the Cluster, Transmission Provider shall determine if a full system impact re-study is necessary. If Transmission Provider determines a

re-study is not necessary and Phase 3 is not required, Transmission Provider shall provide an updated Phase 2 Report within thirty (30) Calendar Days of such determination and the Definitive Interconnection Study Process advances to the Interconnection Facilities Study (Section 8). When the updated Phase 2 report is issued, Transmission Provider shall electronically notify Interconnection Customers in the Cluster that Phase 3 is not required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.

- iii. If one or more Interconnection Customers withdraws from the Cluster and Transmission Provider determines a full system impact re-study is necessary, Transmission Provider will continue with System Impact re-studies ("Phase 3") as described in Section 7.4.e below, until Transmission Provider determines that no further re-studies are required. If a customer withdraws after Section 7.4.d.i or Section 7.4.d.ii or during the Interconnection Facilities Study and Transmission Provider determines system impact level studies are necessary, the Cluster shall be restudied under the terms of Phase 3. Transmission Provider shall electronically notify Interconnection Customers in the Cluster and post on OASIS that a re-study is required. Interconnection Customers that have elected NRIS may make a onetime change between Phase 2 and Phase 3 (before the re-study starts) to ERIS if they notify Transmission Provider of such change in election within five (5) Business Days of the Transmission Provider's notification the first re-study is required.
- Interconnection Customers whose M3 and additional evidence of Site e. Control is accepted by Transmission Provider shall continue with the third phase ("Phase 3") of the Definitive Interconnection System Impact Study. Phase 3 may consist of updated power flow/voltage analysis, stability analysis, and/or short circuit and other analyses if necessary for the Interconnection Customers remaining in the Cluster. Transmission Provider shall use Reasonable Efforts to complete the Phase 3 analysis within one hundred fifty (150) Calendar Days. The results of this analysis shall identify Transmission Provider's Interconnection Facilities and Transmission Provider's Network Upgrades expected to be required to reliably interconnect the Generating Facilities in that Cluster at the requested Interconnection Service level and shall provide non-binding estimates for required upgrades. The Phase 3 Report shall identify each Interconnection Customer's estimated allocated costs for Transmission Provider's Interconnection Facilities and Transmission Provider's Network

Upgrades. Transmission Provider shall meet with each cluster participant ("Phase 3 Report Meeting") within ten (10) Business Days of publishing the DISIS Phase 3 results on OASIS. If additional re-studies are required before moving to Phase 4 below, within thirty (30) Calendar Days of the Phase 3 Report, all Interconnection Customers are required to provide an updated Readiness Milestone 3 ("M3"). Readiness Milestones for the Definitive Interconnection Study Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7. by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7. Transmission Provider shall electronically notify Interconnection Customers in the Cluster when no further re-studies are required and simultaneously provide the Interconnection Facilities Agreement in the form of Appendix 3.

- f. Within twenty (20) Calendar Days of the notice that no System Impact re-studies are needed, each Interconnection Customer is required to provide Readiness Milestone 4 ("M4"), Site Control requirements described in Section 7.7.7, and an executed Interconnection Facilities Agreement in the form of Appendix 3 (completed and including all required data identified therein). Readiness Milestones for the Definitive Interconnection System Process are described in Section 7.7. Interconnection Customers that do not provide the Readiness Milestone (or provide security in lieu of the Readiness Milestone) or do not provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.
- g. Twenty (20) Calendar Days after the notice that no further Re-Studies are needed, Transmission Provider shall proceed with the Interconnection Facilities Study phase ("Phase 4") of the Definitive Study Process, described in detail in Section 8 below. An additional study deposit is not required for Phase 4.

At the request of Interconnection Customer or at any time Transmission Provider determines that it will not meet the indicated time frame for completing the DISIS, Transmission Provider shall notify Interconnection Customer(s) as to the schedule status of the DISIS Cluster. If Transmission Provider is unable to complete the DISIS within the time period, it shall notify Interconnection Customer(s) and provide an estimated completion date with an explanation of the reasons why additional time is required. Upon request, Transmission Provider shall provide Interconnection Customer all supporting documentation, workpapers and

relevant pre-Interconnection Request and post-Interconnection Request power flow, short circuit and stability databases for the DISIS, subject to confidentiality arrangements consistent with Section 13.1.

7.5 Meeting with Transmission Provider

Within ten (10) Business Days of providing a DISIS report to Interconnection Customer and posting the report on OASIS, Transmission Provider and Interconnection Customer shall meet to discuss the study results of the Interconnection System Impact Study, unless otherwise mutually agreed upon by the Parties.

7.6 Re-Study

If Re-Study Definitive of the Interconnection System Impact Study other than the re-study described above in 7.4.e is required due to a higher or equal priority queued project dropping out of the Queue, or a modification of a higher queued project subject to 4.4, Transmission Provider shall notify Interconnection Customer(s) in writing. The Transmission Provider shall use Reasonable Efforts to ensure such Re-Study take no longer than one hundred fifty (150) Calendar Days from the date of notice. Any cost of Re-Study shall be borne by Interconnection Customer(s) being re-studied.

7.7 Readiness Milestones and Site Control

Readiness Milestones are required throughout the Definitive Interconnection Study Process to demonstrate readiness. A customer that does not sufficiently demonstrate readiness by providing a Readiness Milestones is subject to withdrawal as described in Section 3.7 which may include additional penalties.

There are three Readiness Milestone options that demonstrate readiness through the study process (i.e., for Readiness Milestones 1 (M1) through Readiness Milestones 4 (M4).

7.7.1 Readiness Milestone 1 ("M1").

M1 is satisfied by any one of the three options below (also described in 3.4.1) at Interconnection Customer's option. M1 may also be satisfied by providing additional security described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness.

a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating

Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.

- b) Reasonable evidence the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c) Provisional Large Generator Interconnection Agreement accepted for filing at FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

7.7.2 Readiness Milestone 2 ("M2").

M2 is satisfied by any one of the three options below at Interconnection Customer's option. M2 may also be satisfied by providing additional security as described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

- a) Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years.
- b) Reasonable evidence that the project has been selected in a Resource Plan or Resource Solicitation Process; or
- c) Provisional Large Generator Interconnection Agreement accepted for filing at FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.

7.7.3 Readiness Milestone 3 ("M3").

M3 is satisfied by any one of the three options below at Interconnection Customer's option. M3 may also be satisfied by providing additional security described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

a) Executed contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric

- storage resource; where the term of sale is not less than five (5) years.
- b) Reasonable evidence that the project has been selected in an approved Resource Plan or Resource Solicitation Process; or
- c) An unsuspended Provisional Large Generator Interconnection Agreement accepted for filing by FERC with reasonable evidence that the Generating Facility and Interconnection Facilities have commenced design and engineering.

7.7.4 Readiness Milestone 4 ("M4").

M4 is satisfied by any one of the three options below at Interconnection Customer's option. M4 may also be satisfied by providing additional security as described in Section 7.7.5 below in lieu of providing one of the three options to demonstrate readiness.

- a) Executed contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services and capacity if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
- b) Reasonable evidence that the project has been selected in an approved Resource Plan or Resource Solicitation Process; or
- c) An unsuspended Provisional Large Generator Interconnection Agreement accepted for filing by FERC with reasonable evidence that the Generating Facility and Interconnection Facilities have commenced construction.

7.7.5 Security Requirements.

A table showing the security required in each milestone is provided in the At-A-Glance Reference Sheet attachment. The security amount is dependent on if the customer provided a Readiness Milestone and the study phase the customer is entering. All security described below shall be in the form of an irrevocable letter of credit upon which Transmission Provider may draw or cash. The security is refunded to the Interconnection Customer upon withdrawal, LGIA termination, or Commercial Operation after any final invoice is settled. If cash is provided as security, it shall be refunded plus

interest, where the interest is calculated in accordance with the methodology set forth in FERC's regulations at 18 CFR § 35.19a(a)(2)(iii) from the date the security is received to the date that it is refunded. Security may be drawn upon if costs under this LGIP including the LGIA remain unpaid as per this LGIP and the attached LGIA. As part of a valid interconnection request all Interconnection Customers must provide security equal to the study deposit amount as described in Section 3.4. The security provided in Section 3.4.1 will be applied towards the amount of security required for M5. An Interconnection Customer may opt to provide security in lieu of providing Readiness Milestones 1 through 4, as described above in Sections 7.7.1, 7.7.2, 7.7.3, and 7.7.4. The security provided is applied towards the security amount required for each successive milestone if the Interconnection Customer does not withdraw from the queue. For example, the security provided for M2 is applied to the amount of security required for M3.

In lieu of providing a demonstration of readiness for Milestones 1 through 4, the amount of security required is a multiple of the study deposit described in Section 3.4 and is in addition to the security required for all Interconnection Customers under Sections 3.4. The additional amount of security required for each milestone for Interconnection Customers that do not provide a demonstration of readiness is:

M1 = 1 times the study deposit amount

M2 = 2 times the study deposit amount

M3 = 4 times the study deposit amount

M4 = 6 times the study deposit amount

For clarity, the total (i.e., inclusive of the security required under Section 3.4) amount of security required for each milestone for Interconnection Customers that do not provide a demonstration of readiness is:

M1 = 2 times the study deposit amount

M2 = 3 times the study deposit amount

M3 = 5 times the study deposit amount

M4 = 7 times the study deposit amount

All Interconnection Customers are required to provide security in order to satisfy Readiness Milestone 5 (M5) when the LGIA is executed as described in Section 11.3. The amount of security required for M5 is equal to nine (9) times the Interconnection Customer's share of the Definitive Interconnection Study Process study costs. If this amount is not known, the study deposit amount shall be used as an estimate of study cost until such amounts are known. If initially estimated, M5 shall be updated when the final invoice for actual study costs is issued. As this M5 amount is the total security required to satisfy Readiness Milestone 5, any security provided pursuant to Sections 3.4 and 7.7 shall be applied towards the Readiness Milestone 5 amount when the LGIA is executed. The Interconnection Customer shall only be responsible to provide the incremental amount of security to the Transmission Provider and any excess security provided shall be refunded to the Interconnection Customer. Transmission Provider shall refund all security provided under this section to the Interconnection Customer upon achieving Commercial Operation.

7.7.6 [Reserved for Future Use]

7.7.7 Site Control.

In addition to the above Readiness Milestones, Site Control is required to determine increased readiness through the Definitive Interconnection Study Process. Additional information on Site Control is posted on Transmission Provider's OASIS.

- a) Before entering Phase 1 (concurrent with M1) demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- b) Before entering Phase 2 (concurrent with M2): continued demonstration of 50% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- c) Before entering Phase 3 (concurrent with M3): demonstration of 60% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.

- d) Before entering Phase 4 (concurrent with M4): demonstration of 75% Site Control and 0% Site Control of Interconnection Customer's Interconnection Facilities is required.
- e) Before executing an LGIA (concurrent with M5): demonstration of 90% Site Control and 50% Site Control of Interconnection Customer's Interconnection Facilities is required.

Section 8. Interconnection Facilities Study

8.1 Interconnection Facilities Study Agreement

Simultaneously with the notice to Interconnection Customer(s) that Phase 3 is complete or not required, Transmission Provider shall provide to Interconnection Customer an Interconnection Facilities Study Agreement in the form of Appendix 3 to this LGIP. Within five (5) Business Days following the DISIS results (Phase 2 or Phase 3) meeting, Transmission Provider shall provide to Interconnection Customer a non-binding good faith estimate of the cost and timeframe for completing the Interconnection Facilities Study. The Interconnection Facilities Study Agreement shall provide that Interconnection Customer shall compensate Transmission Provider for the actual cost of the Interconnection Facilities Study. Interconnection Customer shall execute the Interconnection Facilities Study Agreement and deliver the executed Interconnection Facilities Study Agreement to Transmission Provider within twenty (20) Calendar Days after its receipt, together with the required technical data, Readiness Milestone 4 and the Site Control requirements described in Section 7.7.7. Interconnection Customers that do not provide the Readiness Milestone (or additional security in lieu of the Readiness Milestone) and provide Site Control described in Section 7.7.7 by the required date shall be deemed withdrawn from the Queue pursuant to Section 3.7.

.

8.2 Scope of Interconnection Facilities Study

The Interconnection Facilities Study shall specify and provide a non-binding estimate of the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the Phase 1 and Phase 2 (and/or Phase 3) Reports (as appropriate) in accordance with Good Utility Practice to physically and electrically connect the Interconnection Facilities to the Transmission System. The Interconnection Facilities Study shall also identify the electrical switching configuration of the connection equipment, including, without limitation: the transformer, switchgear, meters, and other station equipment; the nature

and estimated cost of any Transmission Provider's Interconnection Facilities and Network Upgrades necessary to accomplish the interconnection; and an estimate of the time required to complete the construction and installation of such facilities. The Interconnection Facilities Study will also identify any potential control equipment for requests for Interconnection Service that are lower than the Generating Facility Capacity.

8.3 Interconnection Facilities Study Procedures

Transmission Provider shall coordinate the Interconnection Facilities Study with any Affected System pursuant to Section 3.6 above.

Transmission Provider shall utilize existing studies to the extent practicable in performing the Interconnection Facilities Study. Transmission Provider shall use Reasonable Efforts to complete the study and issue a draft Interconnection Facilities Study report to Interconnection Customer within ninety (90) Calendar Days after acceptance of the Interconnection Facilities Agreement and Readiness Milestone 4.

At the request of Interconnection Customer, Transmission Provider shall notify Interconnection Customer as to the schedule status of the Interconnection Facilities Study. If Transmission Provider is unable to complete the Interconnection Facilities Study and issue a draft Interconnection Facilities Study report within the time identified, it shall notify Interconnection Customer and provide an estimated completion date and an explanation of the reasons why additional time is required.

Interconnection Customer may, within thirty (30) Calendar Days after receipt of the draft Interconnection Facilities Study report, provide written comments to Transmission Provider, which Transmission Provider shall consider in completing the final Interconnection Facilities Study report. Transmission Provider shall issue the final Interconnection Facilities Study report within fifteen (15) Business Days of receiving Interconnection Customer's comments or promptly upon receiving Interconnection Customer's statement that it will not provide comments. Transmission Provider may reasonably extend such fifteen-day period upon notice to Interconnection Customer if Interconnection Customer's comments require Transmission Provider to perform additional analyses or make other significant modifications prior to the issuance of the final Interconnection Facilities Study report. Upon request, Transmission Provider shall provide Interconnection Customer supporting documentation, workpapers, and databases or data developed in the preparation of the Interconnection Facilities Study, subject to confidentiality arrangements consistent with

Section 13.1.

8.4 Meeting with Transmission Provider

Within ten (10) Business Days of providing a draft Interconnection Facilities Study report to Interconnection Customer, Transmission Provider and Interconnection Customer shall meet to discuss the results of the Interconnection Facilities Study, unless otherwise mutually agreed upon by the Parties.

8.5 Re-Study

If Re-Study of the Interconnection Facilities Study is required due to a higher or equal priority queued project dropping out of the Queue or a modification of a higher queued project pursuant to Section 4.4, Transmission Provider shall so notify Interconnection Customer(s) in writing. The Transmission Provider shall use Reasonable Efforts to complete the Re-Study in no longer than sixty (60) Calendar Days from the date of notice. Re-studies that require re-running the system impact study analysis may take longer than sixty days. Any cost of Re-Study shall be borne by the Interconnection Customer(s) being re-studied.

Section 9. Engineering & Procurement ('E&P') Agreement

Prior to executing an LGIA, an Interconnection Customer may, in order to advance the implementation of its interconnection, request and Transmission Provider shall offer the Interconnection Customer, an E&P Agreement that authorizes Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection. However, Transmission Provider shall not be obligated to offer an E&P Agreement if Interconnection Customer is in Dispute Resolution as a result of an allegation that Interconnection Customer has failed to meet any Readiness Milestones or comply with any prerequisites specified in other parts of the LGIP. The E&P Agreement is an optional procedure and it will not alter the Interconnection Customer's Queue Position or In-Service Date. The E&P Agreement shall provide for Interconnection Customer to pay the cost of all activities authorized by Interconnection Customer and to make advance payments or provide other satisfactory security for such costs.

Interconnection Customer shall pay the cost of such authorized activities and any cancellation costs for equipment that is already ordered for its interconnection, which cannot be mitigated as hereafter described, whether or not such items or equipment later become unnecessary. If Interconnection Customer withdraws its application for interconnection or either Party terminates the E&P Agreement, to the extent the equipment

ordered can be canceled under reasonable terms, Interconnection Customer shall be obligated to pay the associated cancellation costs. To the extent that the equipment cannot be reasonably canceled, Transmission Provider may elect: (i) to take title to the equipment, in which event Transmission Provider shall refund Interconnection Customer any amounts paid by Interconnection Customer for such equipment and shall pay the cost of delivery of such equipment, or (ii) to transfer title to and deliver such equipment to Interconnection Customer, in which event Interconnection Customer shall pay any unpaid balance and cost of delivery of such equipment.

Section 10. [Reserved for Future Use]

Section 11. Standard Large Generator Interconnection Agreement (LGIA)

11.1 Tender

Interconnection Customer shall tender comments on the draft Interconnection Facilities Study Report within thirty (30) Calendar Days of receipt of the report. Within thirty (30) Calendar Days after the comments are submitted or after the Interconnection Customer notifies Transmission Provider that it will provide no comments, Transmission Provider shall tender a draft LGIA, together with draft appendices. The draft LGIA shall be in the form of Transmission Provider's FERC-approved standard form LGIA, which is in Appendix 6. Interconnection Customer shall return the completed draft appendices within thirty (30) Calendar Days so that Transmission Provider may prepare a final LGIA. Transmission Provider shall tender a final LGIA, together with final appendices, within thirty (30) Calendar Days after the expiration of the thirty (30) Calendar Day window in which the Interconnection Customer is to return completed draft appendices.

11.2 Negotiation

Notwithstanding Section 11.1, at the request of Interconnection Customer Transmission Provider shall begin negotiations with Interconnection Customer concerning the appendices to the LGIA at any time after Interconnection Customer executes the Interconnection Facilities Study Agreement. Transmission Provider and Interconnection Customer shall negotiate concerning any disputed provisions of the appendices to the draft LGIA for not more than sixty (60) Calendar Days after tender of the final Interconnection Facilities Study Report. If Interconnection Customer

determines that negotiations are at an impasse, Interconnection Customer may request termination of the negotiations at any time after tender of the draft LGIA pursuant to Section 11.1 and request submission of the unexecuted LGIA with FERC or initiate Dispute Resolution procedures pursuant to Section 13.5. If Interconnection Customer requests termination of the negotiations, but within sixty (60) Calendar Days thereafter fails to request either the filing of the unexecuted LGIA or initiate Dispute Resolution, it shall be deemed to have withdrawn its Interconnection Request. Unless otherwise agreed by the Parties, if Interconnection Customer has not executed the LGIA, requested filing of an unexecuted LGIA, or initiated Dispute Resolution procedures pursuant to Section 13.5 within sixty (60) Calendar Days of tender of draft LGIA, it shall be deemed to have withdrawn its Interconnection Request. Transmission Provider shall provide to Interconnection Customer a final LGIA within fifteen (15) Business Days after the completion of the negotiation process.

11.3 Execution and Filing

Within fifteen (15) Business Days after receipt of the final LGIA, Interconnection Customer shall provide Transmission Provider (A) reasonable evidence that continued Site Control as defined in Section 7.7.7 and (B) post Readiness Milestone 5 security equal to nine (9) times that Interconnection Customer's share of the Definitive Interconnection Study Process study costs. If the actual study costs are not known at the time, study costs shall be estimated as the study deposit described in Section 3.4, and the M5 amount shall be updated when the study costs are known. If the Interconnection Customer does not reach Commercial Operation, upon payment of any final invoice, including any Withdrawal Penalty, Readiness Milestone 5 shall be refunded to the Interconnection Customer, including any accumulated interest, if applicable. If the Interconnection Customer reaches Commercial Operation, Readiness Milestone 5 is refunded to the Interconnection Customer including any accumulated interest, if applicable. Within fifteen (15) Business Days after receipt of the final LGIA, Interconnection Customer also shall provide reasonable evidence that one or more of the following milestones in the development of the Large Generating Facility, at Interconnection Customer election, has been achieved: (i) the execution of a contract for the supply or transportation of fuel to the Large Generating Facility (applicable to Generating Facilities that require fuel transportation; not available for wind or solar resources); (ii) the execution of a contract for the supply of cooling water to the Large Generating Facility (applicable to Generating Facilities that require cooling water for the production process; not available for wind or solar resources); (iii) execution of a contract for the engineering for, procurement of major

equipment for, or construction of, the Large Generating Facility; (iv) execution of a contract for the sale of electric energy or capacity from the Large Generating Facility; or (v) application for an air, water, or land use permit.

At the same time that Interconnection Customer makes its submission of the items identified above (due within fifteen (15) Business Days after receipt of the final LGIA), Interconnection Customer shall either: (i) execute two originals of the tendered LGIA and return them to Transmission Provider; or (ii) request in writing that Transmission Provider file with FERC an LGIA in unexecuted form. As soon as practicable, but not later than ten (10) Business Days after receiving either the two executed originals of the tendered LGIA (if it does not conform with a FERC-approved standard form of interconnection agreement) or the request to file an unexecuted LGIA, Transmission Provider shall file the LGIA with FERC, together with its explanation of any matters as to which Interconnection Customer and Transmission Provider disagree and support for the costs that Transmission Provider proposes to charge to Interconnection Customer under the LGIA. An unexecuted LGIA should contain terms and conditions deemed appropriate by Transmission Provider for the Interconnection Request. If the Parties agree to proceed with design, procurement, and construction of facilities and upgrades under the agreed-upon terms of the unexecuted LGIA, they may proceed pending FERC action.

11.4 Commencement of Interconnection Activities

If Interconnection Customer executes the final LGIA, Transmission Provider and Interconnection Customer shall perform their respective obligations in accordance with the terms of the LGIA, subject to modification by FERC. Upon submission of an unexecuted LGIA, Interconnection Customer and Transmission Provider shall promptly comply with the unexecuted LGIA, subject to modification by FERC.

Section 12. Construction of Transmission Provider's Interconnection Facilities and Network Upgrades

12.1 Schedule

Transmission Provider and Interconnection Customer shall negotiate in good faith concerning a schedule for the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades.

12.2 Construction Sequencing

12.2.1 General

In general, the In-Service Date of an Interconnection Customers seeking interconnection to the Transmission System will determine the sequence of construction of Network Upgrades.

12.2.2 Advance Construction of Network Upgrades that are an Obligation of an Entity other than Interconnection Customer

An Interconnection Customer with an LGIA, in order to maintain its In-Service Date, may request that Transmission Provider advance to the extent necessary the completion of Network Upgrades that: (i) were assumed in the Interconnection Studies for such Interconnection Customer, (ii) are necessary to support such In-Service Date, and (iii) would otherwise not be completed, pursuant to a contractual obligation of an entity other than Interconnection Customer that is seeking interconnection to the Transmission System, in time to support such In-Service Date. Upon such request, Transmission Provider will use Reasonable Efforts to advance the construction of such Network Upgrades to accommodate such request; provided that Interconnection Customer commits to pay Transmission Provider: (i) any associated expediting costs and (ii) the cost of such Network Upgrades.

Transmission Provider will refund to Interconnection Customer both the expediting costs and the cost of Network Upgrades, in accordance with Article 11.4 of the LGIA. Consequently, the entity with a contractual obligation to construct such Network Upgrades shall be obligated to pay only that portion of the costs of the Network Upgrades that Transmission Provider has not refunded to Interconnection Customer. Payment by that entity shall be due on the date that it would have been due had there been no request for advance construction. Transmission Provider shall forward to Interconnection Customer the amount paid by the entity with a contractual obligation to construct the Network Upgrades as payment in full for the outstanding balance owed to Interconnection Customer. Transmission Provider then shall refund to that entity the amount that it paid for the Network Upgrades, in accordance with Article 11.4 of the LGIA.

12.2.3 **Advancing Construction of Network Upgrades that**

are Part of an Expansion Plan of the Transmission Provider An Interconnection Customer with an LGIA, in order to maintain its In-Service Date, may request that Transmission Provider advance to the extent necessary the completion of Network Upgrades that: (i) are necessary to support such In-Service Date and (ii) would otherwise not be completed, pursuant to an expansion plan of Transmission Provider, in time to support such In-Service Date. Upon such request, Transmission Provider will use Reasonable Efforts to advance the construction of such Network Upgrades to accommodate such request; provided that Interconnection Customer commits to pay Transmission Provider any associated expediting costs. Interconnection Customer shall be entitled to transmission credits, if any, for any expediting costs paid.

12.2.4 Amended Definitive Interconnection System Impact Study

A Definitive Interconnection System Impact Study may be amended to determine the facilities necessary to support the requested In-Service Date. This amended study will include those transmission and Large Generating Facilities that are expected to be in service on or before the requested In-Service Date.

Section 13. Miscellaneous

13.1 Confidentiality

Confidential Information shall include, without limitation, all information relating to a Party's technology, research and development, business affairs, and pricing, and any information supplied by either of the Parties to the other prior to the execution of an LGIA.

Information is Confidential Information only if it is clearly designated or marked in writing as confidential on the face of the document, or, if the information is conveyed orally or by inspection, if the Party providing the information orally informs the Party receiving the information that the information is confidential.

If requested by either Party, the other Party shall provide in writing, the basis for asserting that the information referred to in this Section warrants confidential treatment, and the requesting Party may disclose such writing to the appropriate Governmental Authority. Each Party shall be responsible for the costs associated with affording confidential treatment to

its information.

The release of Confidential Information shall be subject to Applicable Laws and Regulations and Applicable Reliability Standards.

13.1.1 Scope

Confidential Information shall not include information that the receiving Party can demonstrate: (1) is generally available to the public other than as a result of a disclosure by the receiving Party; (2) was in the lawful possession of the receiving Party on a non-confidential basis before receiving it from the disclosing Party; (3) was supplied to the receiving Party without restriction by a third party, who, to the knowledge of the receiving Party after due inquiry, was under no obligation to the disclosing Party to keep such information confidential; (4) was independently developed by the receiving Party without reference to Confidential Information of the disclosing Party; (5) is, or becomes, publicly known, through no wrongful act or omission of the receiving Party or Breach of the LGIA; or (6) is required, in accordance with Section 13.1.6, Order of Disclosure, to be disclosed by any Governmental Authority or is otherwise required to be disclosed by law or subpoena, or is necessary in any legal proceeding establishing rights and obligations under the LGIA. Information designated as Confidential Information will no longer be deemed confidential if the Party that designated the information as confidential notifies the other Party that it no longer is confidential.

13.1.2 Release of Confidential Information

Neither Party shall release or disclose Confidential Information to any other person, except to its Affiliates (limited by the Standards of Conduct requirements), employees, consultants, or to parties who may be or considering providing financing to or equity participation with Interconnection Customer, or to potential purchasers or assignees of Interconnection Customer, on a need-to-know basis in connection with these procedures, unless such person has first been advised of the confidentiality provisions of this Section 13.1 and has agreed to comply with such provisions. Notwithstanding the foregoing, a Party providing Confidential Information to any person shall remain primarily responsible for any release of Confidential Information in

contravention of this Section 13.1.

13.1.3 Rights

Each Party retains all rights, title, and interest in the Confidential Information that each Party discloses to the other Party. The disclosure by each Party to the other Party of Confidential Information shall not be deemed a waiver by either Party or any other person or entity of the right to protect the Confidential Information from public disclosure.

13.1.4 No Warranties

By providing Confidential Information, neither Party makes any warranties or representations as to its accuracy or completeness. In addition, by supplying Confidential Information, neither Party obligates itself to provide any particular information or Confidential Information to the other Party nor to enter into any further agreements or proceed with any other relationship or joint venture.

13.1.5 Standard of Care

Each Party shall use at least the same standard of care to protect Confidential Information it receives as it uses to protect its own Confidential Information from unauthorized disclosure, publication or dissemination. Each Party may use Confidential Information solely to fulfill its obligations to the other Party under these procedures or its regulatory requirements.

13.1.6 Order of Disclosure

If a court or a Government Authority or entity with the right, power, and apparent authority to do so requests or requires either Party, by subpoena, oral deposition, interrogatories, requests for production of documents, administrative order, or otherwise, to disclose Confidential Information, that Party shall provide the other Party with prompt notice of such request(s) or requirement(s) so that the other Party may seek an appropriate protective order or waive compliance with the terms of the LGIA. Notwithstanding the absence of a protective order or waiver, the Party may disclose such Confidential Information which, in the opinion of its counsel, the Party is legally compelled to disclose. Each Party will use Reasonable Efforts to obtain reliable assurance that confidential treatment will be accorded any Confidential

Information so furnished.

13.1.7 Remedies

The Parties agree that monetary damages would be inadequate to compensate a Party for the other Party's Breach of its obligations under this Section 13.1. Each Party accordingly agrees that the other Party shall be entitled to equitable relief, by way of injunction or otherwise, if the first Party Breaches or threatens to Breach its obligations under this Section 13.1, which equitable relief shall be granted without bond or proof of damages, and the receiving Party shall not plead in defense that there would be an adequate remedy at law. Such remedy shall not be deemed an exclusive remedy for the Breach of this Section 13.1, but shall be in addition to all other remedies available at law or in equity. The Parties further acknowledge and agree that the covenants contained herein are necessary for the protection of legitimate business interests and are reasonable in scope. No Party, however, shall be liable for indirect, incidental, or consequential or punitive damages of any nature or kind resulting from or arising in connection with this Section 13.1.

13.1.8 Disclosure to FERC, its Staff, or a State

Notwithstanding anything in this Section 13.1 to the contrary, and pursuant to 18 CFR section 1b.20, if FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to the LGIP, the Party shall provide the requested information to FERC or its staff, within the time provided for in the request for information. In providing the information to FERC or its staff, the Party must, consistent with 18 CFR section 388.112, request that the information be treated as confidential and non public by FERC and its staff and that the information be withheld from public disclosure. Parties are prohibited from notifying the other Party prior to the release of the Confidential Information to FERC or its staff. The Party shall notify the other Party to the LGIA when its is notified by FERC or its staff that a request to release Confidential Information has been received by FERC, at which time either of the Parties may respond before such information would be made public, pursuant to 18 CFR section 388.112. Requests from a state regulatory body conducting a confidential

investigation shall be treated in a similar manner, consistent with applicable state rules and regulations.

- 13.1.9 Subject to the exception in Section 13.1.8, any information that a Party claims is competitively sensitive, commercial or financial information ("Confidential Information") shall not be disclosed by the other Party to any person not employed or retained by the other Party, except to the extent disclosure is (i) required by law; (ii) reasonably deemed by the disclosing Party to be required to be disclosed in connection with a dispute between or among the Parties, or the defense of litigation or dispute; (iii) otherwise permitted by consent of the other Party, such consent not to be unreasonably withheld; or (iv) necessary to fulfill its obligations under this LGIP or as a transmission service provider or a Control Area operator including disclosing the Confidential Information to an RTO or ISO or to a subregional, regional or national reliability organization or planning group. The Party asserting confidentiality shall notify the other Party in writing of the information it claims is confidential. Prior to any disclosures of the other Party's Confidential Information under this subparagraph, or if any third party or Governmental Authority makes any request or demand for any of the information described in this subparagraph, the disclosing Party agrees to promptly notify the other Party in writing and agrees to assert confidentiality and cooperate with the other Party in seeking to protect the Confidential Information from public disclosure by confidentiality agreement, protective order or other reasonable measures.
- This provision shall not apply to any information that was or is hereafter in the public domain (except as a result of a Breach of this provision).
- 13.1.11 Transmission Provider shall, at Interconnection Customer's election, destroy, in a confidential manner, or return the Confidential Information provided at the time of Confidential Information is no longer needed.

13.2 Delegation of Responsibility

Transmission Provider may use the services of subcontractors as it deems appropriate to perform its obligations under this LGIP.

Transmission Provider shall remain primarily liable to

Interconnection Customer for the performance of such subcontractors and compliance with its obligations of this LGIP. The subcontractor shall keep all information provided confidential and shall use such information solely for the performance of such obligation for which it was provided and no other purpose.

13.3 Obligation for Study Costs and Withdrawal Penalty

Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Interconnection Studies and the Withdrawal Penalty, as applicable. Any difference between the Initial Deposit and the actual cost of the Interconnection Study(ies) shall be paid by or refunded, except as otherwise provided herein, to Interconnection Customer. Any invoices for Interconnection Studies shall include a detailed and itemized accounting of the cost of each Interconnection Study as well as the Withdrawal Penalty, if applicable. Interconnection Customer shall pay any such undisputed costs within thirty (30) Calendar Days of receipt of an invoice therefor. Transmission Provider shall not be obligated to perform or continue to perform any studies unless Interconnection Customer has paid all undisputed amounts in compliance herewith. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided under this LGIP to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

13.4 Third Parties Conducting Studies

If (i) at the time of the signing of an Interconnection Study Agreement there is disagreement as to the estimated time to complete an Interconnection Study, (ii) Interconnection Customer receives notice pursuant to Sections 7.4 or 8.3 that Transmission Provider will not complete an Interconnection Study within the applicable timeframe for such Interconnection Study, or (iii) Interconnection Customer receives neither the Interconnection Study nor a notice under Sections 7.4 or 8.3 within the applicable timeframe for such Interconnection Study, then Interconnection Customer may require Transmission Provider to utilize a third party consultant reasonably acceptable to Interconnection Customer and Transmission Provider to perform such Interconnection Study under the direction of Transmission Provider. At other times, Transmission Provider may also utilize a third party consultant to

perform such Interconnection Study, either in response to a general request of Interconnection Customer, or on its own volition.

In all cases, use of a third party consultant shall be in accord with Article 26 of the LGIA (Subcontractors) and limited to situations where Transmission Provider determines that doing so will help maintain or accelerate the study process for Interconnection Customer's pending Interconnection Request and not interfere with Transmission Provider's progress on Interconnection Studies for other pending Interconnection Requests. In cases where Interconnection Customer requests use of a third party consultant to perform such Interconnection Study, Interconnection Customer and Transmission Provider shall negotiate all of the pertinent terms and conditions, including reimbursement arrangements and the estimated study completion date and study review deadline. Transmission Provider shall convey all workpapers, data bases, study results and all other supporting documentation prepared to date with respect to the Interconnection Request as soon as practicable upon Interconnection Customer's request subject to the confidentiality provision in Section 13.1. In any case, such third party contract may be entered into with either Interconnection Customer or Transmission Provider at Transmission Provider's discretion. In the case of (iii) Interconnection Customer maintains its right to submit a claim to Dispute Resolution to recover the costs of such third party study. Such third party consultant shall be required to comply with this LGIP, Article 26 of the LGIA (Subcontractors), and the relevant Tariff procedures and protocols as would apply if Transmission Provider were to conduct the Interconnection Study and shall use the information provided to it solely for purposes of performing such services and for no other purposes. Transmission Provider shall cooperate with such third party consultant and Interconnection Customer to complete and issue the Interconnection Study in the shortest reasonable time.

13.5 Disputes

13.5.1 Submission

In the event either Party has a dispute, or asserts a claim, that arises out of or in connection with the LGIA, the LGIP, or their performance, such Party (the "disputing Party") shall provide the other Party with written notice of the dispute or claim ("Notice of Dispute"). Such dispute or claim shall be referred to a designated senior representative of each Party for

resolution on an informal basis as promptly as practicable after receipt of the Notice of Dispute by the other Party. In the event the designated representatives are unable to resolve the claim or dispute through unassisted or assisted negotiations within thirty (30) Calendar Days of the other Party's receipt of the Notice of Dispute, such claim or dispute may, upon mutual agreement of the Parties, be submitted to arbitration and resolved in accordance with the arbitration procedures set forth below. In the event the Parties do not agree to submit such claim or dispute to arbitration, each Party may exercise whatever rights and remedies it may have in equity or at law consistent with the terms of this LGIA.

13.5.2 External Arbitration Procedures

Any arbitration initiated under these procedures shall be conducted before a single neutral arbitrator appointed by the Parties. If the Parties fail to agree upon a single arbitrator within ten (10) Calendar Days of the submission of the dispute to arbitration, each Party shall choose one arbitrator who shall sit on a three-member arbitration panel. The two arbitrators so chosen shall within twenty (20) Calendar Days select a third arbitrator to chair the arbitration panel. In either case, the arbitrators shall be knowledgeable in electric utility matters, including electric transmission and bulk power issues, and shall not have any current or past substantial business or financial relationships with any party to the arbitration (except prior arbitration). The arbitrator(s) shall provide each of the Parties an opportunity to be heard and, except as otherwise provided herein, shall conduct the arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("Arbitration Rules") and any applicable FERC regulations or RTO rules; provided, however, in the event of a conflict between the Arbitration Rules and the terms of this Section 13, the terms of this Section 13 shall prevail.

13.5.3 Arbitration Decisions

Unless otherwise agreed by the Parties, the arbitrator(s) shall render a decision within ninety (90) Calendar Days of appointment and shall notify the Parties in writing of such decision and the reasons therefor. The arbitrator(s) shall be authorized only to interpret and apply the provisions of the LGIA and LGIP and shall have no power to modify or change

any provision of the LGIA and LGIP in any manner. The decision of the arbitrator(s) shall be final and binding upon the Parties, and judgment on the award may be entered in any court having jurisdiction. The decision of the arbitrator(s) may be appealed solely on the grounds that the conduct of the arbitrator(s), or the decision itself, violated the standards set forth in the Federal Arbitration Act or the Administrative Dispute Resolution Act. The final decision of the arbitrator must also be filed with FERC if it affects jurisdictional rates, terms and conditions of service, Interconnection Facilities, or Network Upgrades.

13.5.4 Costs

Each Party shall be responsible for its own costs incurred during the arbitration process and for the following costs, if applicable: (1) the cost of the arbitrator chosen by the Party to sit on the three member panel and one half of the cost of the third arbitrator chosen; or (2) one half the cost of the single arbitrator jointly chosen by the Parties.

13.5.5 Non-Binding Dispute Resolution Procedures.

If a Party has submitted a Notice of Dispute pursuant to section 13.5.1, and the Parties are unable to resolve the claim or dispute through unassisted or assisted negotiations within the thirty (30) Calendar Days provided in that section, and the Parties cannot reach mutual agreement to pursue the section 13.5 arbitration process, a Party may request that Transmission Provider engage in Non-binding Dispute Resolution pursuant to this section by providing written notice to Transmission Provider ("Request for Non-binding Dispute Resolution"). Conversely, either Party may file a Request for Non-binding Dispute Resolution pursuant to this section without first seeking mutual agreement to pursue the section 13.5 arbitration process. The process in section 13.5.5 shall serve as an alternative to, and not a replacement of, the section 13.5 arbitration process. Pursuant to this process, a Transmission Provider must within 30 Calendar Days of receipt of the Request for Non-binding Dispute Resolution appoint a

neutral decision-maker that is an independent subcontractor that shall not have any current or past substantial business or financial relationships with either Party. Unless otherwise agreed by the Parties, the decision-maker shall render a decision within sixty (60) Calendar Days of appointment and shall notify the Parties in writing of such decision and reasons therefore. This decision-maker shall be authorized only to interpret and apply the provisions of the LGIP and LGIA and shall have no power to modify or change any provision of the LGIP and LGIA in any manner. The result reached in this process is not binding, but, unless otherwise agreed, the Parties may cite the record and decision in the non-binding dispute resolution process in future dispute resolution processes, including in a section 13.5 arbitration, or in a Federal Power Act section 206 complaint. Each Party shall be responsible for its own costs incurred during the process and the cost of the decision-maker shall be divided equally among each Party to the dispute.

13.6 Local Furnishing Bonds

13.6.1 Transmission Providers That Own Facilities Financed by Local Furnishing Bonds

This provision is applicable only to a Transmission Provider that has financed facilities for the local furnishing of electric energy with tax-exempt bonds, as described in Section 142(f) of the Internal Revenue Code ("local furnishing bonds"). Notwithstanding any other provision of this LGIA and LGIP, Transmission Provider shall not be required to provide Interconnection Service to Interconnection Customer pursuant to this LGIA and LGIP if the provision of such Transmission Service would jeopardize the tax-exempt status of any local furnishing bond(s) used to finance Transmission Provider's facilities that would be used in providing such Interconnection Service.

13.6.2 Alternative Procedures for Requesting Interconnection Service

If Transmission Provider determines that the provision of

Interconnection Service requested by Interconnection Customer would jeopardize the tax exempt status of any local furnishing bond(s) used to finance its facilities that would be used in providing such Interconnection Service, it shall advise the Interconnection Customer within thirty (30) Calendar Days of receipt of the Interconnection Request.

Interconnection Customer thereafter may renew its request for interconnection using the process specified in Section 5.2(ii) of the Transmission Provider's Tariff.

At-A-Glance Reference Sheet

Milestone	Total Security Required (Multiple of Section 3.4 Study Deposit) If Demonstration of Readiness IS Provided	Total Security Required (Multiple of Section 3.4 Study Deposit) If Demonstration of Readiness IS NOT Provided	Demonstration of Site Control for All Fuel Types	Site Control of ICIF's
M1	1x	2x	50%	0%
M2	1x	3x	50%	0%
M3	1x	5x	60%	0%
M4	1x	7x	75%	0%
M5	9x	9x	90%	50%

Phase 1: Power Flow/Voltage: Within 90 calendar days

- Transmission Provider to perform Power Flow and Voltage Analyses.
- Transmission Provider to complete Phase 1 report within 90 calendar days after the close of the Queue Cluster Window.
- Transmission Provider to hold results meeting within 10 business days of posting of DISIS Phase 1 results.
- Interconnection Customer to demonstrate M2 Readiness within 30 calendar days following the Phase 1 report.

Phase 2: Stability/Short Circuit: Within 150 calendar days

- Transmission Provider to complete Phase 2 analyses within 150 calendar days after receipt of all Interconnection Customers' executed Interconnection System Impact Study Agreements.
- Transmission Provider to hold Phase 2 results meeting within 10 business days of posting of Phase 2 report.
- Interconnection Customer to demonstrate M3 (if Re-Study is necessary) within 30 calendar days following the Phase 2 report, or demonstrate M4 Readiness (if no Re-Study is necessary) within 20 calendar days following Transmission Provider's notice that no Re-Study is needed.

**Phase 3: Iterative Cluster System Impact Re-Study: Within 150 calendar days

Filed Date: 05/31/2023

May not be necessary

- If a Re-Study is needed, Transmission Provider perform Phase 3 Re-Study within 150 calendar days after acceptance of all Interconnection Customers' executed Interconnection System Impact Study Agreements.
- Transmission Provider to hold results meeting within 10 business days of posting results.
- Interconnection Customer to demonstrate M4 Readiness within 20 calendar days following Transmission Provider's notice that no further System Impact Re-Study is needed.

Phase 4: Facilities Study: Within 90 calendar days

- Transmission Provider to complete Facilities Study, complete and submit draft Facilities Study Report to Interconnection Customer within 90 calendar days after acceptance of all Interconnection Facilities Study Agreements.
- Transmission Provider to hold results meeting within 10 business days of posting of Phase 4 report.
- Interconnection Customer has opportunity to provide written comments of Facilities Study Report to Transmission Provider within 30 calendar days of receipt of draft Facilities Study Report.
- Transmission Provider to issue Final Facilities Study Report to Customer within 15 business days of receiving Interconnection Customer's comments or promptly upon receiving Interconnection Customer's statement that it will not provide comments.

Phase 5: LGIA

- Transmission Provider to provide Interconnection Customer with draft LGIA within 30 calendar days of receipt of Interconnection Customer's Facilities Study comments.
- Interconnection Customer to return completed draft appendices within 30 calendar days of receipt of draft LGIA.
- Deadline for LGIA negotiations to be completed is within 60 calendar days after tender by Transmission Provider of the final Interconnection Facilities Study Report.
- Interconnection Customer to satisfy Readiness Milestone 5 within 15 business days of receiving final LGIA.
- Deadline for Interconnection Customer to execute the LGIA (or instruct that it be filed unexecuted) is 15 business days after receipt of the final LGIA.

Appendices

LGIP.

APPENDIX 1 to LGIP INTERCONNECTION REQUEST FOR A LARGE GENERATING FACILITY

1.	The undersigned Interconnection Customer submits this request to interconnect its Large Generating Facility with Transmission Provider's Transmission System pursuant to a Tariff.
2.	This Interconnection Request is for (check one): A proposed new Large Generating Facility. An increase in the generating capacity or a Material Modification of an existing Generating Facility. A Generating Facility proposed for inclusion in a resource solicitation process.
3.	The type of interconnection service requested (check one): Energy Resource Interconnection Service Network Resource Interconnection Service
4.	Interconnection Customer provides the following information:
	a. Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility;
	b. Maximum summer at degrees C and winter at degrees C megawat electrical output of the proposed new Large Generating Facility or the amount of megawatt increase in the generating capacity of an existing Generating Facility;
	c. General description of the equipment configuration;
	d. Commercial Operation Date (Day, Month, and Year);
	e. Name, address, telephone number, and e-mail address of Interconnection Customer's contact person;
	f. Approximate location of the proposed Point of Interconnection (optional);
	g. Interconnection Customer Data (set forth in Attachment A)
	 h. Primary frequency response operating range for electric storage resources. i. Requested capacity (in MW) of Interconnection Service (if lower than the Generating Facility Capacity).
5.	Interconnection Customer provides applicable study deposit amount as specified in the LGIP.
	\$160,000 for requests of less than 75 MW \$250,000 for requests of 75 MW and greater
6.	Interconnection Customer provides Readiness Milestone 1 (M1) as specified in the

M1 is satisfied by any one of the three options below (also described in the LGIP) at Interconnection Customer's option. M1 may also be satisfied by providing additional

security described in Section 7.7.5 in lieu of providing one of the three options to demonstrate readiness.

- i. Executed term sheet (or comparable evidence) related to a contract, binding upon the parties to the contract, for sale of (i) the constructed Generating Facility, (ii) the Generating Facility's energy, or (iii) the Generating Facility's ancillary services if the Generating Facility is an electric storage resource; where the term of sale is not less than five (5) years;
- ii. Reasonable evidence the project has been selected in a Resource Plan or Resource Solicitation Process; or
- Provisional Large Generator Interconnection Agreement accepted for filing with iii. FERC. Such an agreement shall not be suspended and shall include a commitment to construct the Generating Facility.
- 7. Interconnection Customer provides security equal to one times the study deposit described in Section 3 of the LGIP in the form of an irrevocable letter of credit or cash.
- If requesting NRIS: Interconnection Customer provides the expected point of delivery to 8. deliver within the Transmission Provider's Control Area.
- 9. Interconnection Customer provides Evidence of Site Control as specified in the LGIP and Transmission Provider's business practices posted on OASIS, as applicable.
- 10. This Interconnection Request shall be submitted to the representative indicated below:

[To be completed by Transmission Provider]

11. Representative of Interconnection Customer to contact:

[To be completed by Interconnection Customer]

12. This Interconnection Request is submitted by:

Name of Interconnection Customer:
By (signature):
Name (type or print):
Title:
Date:

Attachment A to Appendix 1 Interconnection Request

LARGE GENERATING FACILITY DATA

UNIT RATINGS					
kVA	°F	Voltage			
Power Factor					
Speed (RPM)	Connection (e.g. Wye)				
Short Circuit Ratio	Frequency, Hertz				
Stator Amperes at Rated kVA	°F	Field Volts			
Max Turbine MW	г				
Primary frequency resp	onse operating range for	electric storage resources.			
Minimum State of Cha Maximum State of Cha		<u></u>			
COMBINED TUR	RBINE-GENERATOR-EXCI	TER INERTIA DATA			
Inertia Constant, H = Moment-of-Inertia, WR ² =	kW sec/kVA lb. ft.²				
	ANCE DATA (PER UNIT-RA	ATED KVA)			
DIRECT AXIS QUADRATURE AXIS					
Synchronous – saturated	X _{dv}	X _{qv}			
Synchronous – unsaturated	X_{di}	X_{qi}			
Transient – saturated	X' _{dv}	X' _{qv}			
Transient – unsaturated	X' _{di}	X' _{qi}			
Subtransient – saturated	X" _{dv}	X" _{qv}			
Subtransient – unsaturated	X" _{di}	X" _{qi}			
Negative Sequence – saturated	X2 _v				
Negative Sequence – unsaturate	ed X2 _i				
Zero Sequence – saturated	X0 _v				
Zero Sequence – unsaturated	X0 _i				
Leakage Reactance	XI _m				

FIELD TIME CONSTANT DATA (SEC)

Open Circuit			T' _{do}	T' _{qo}	
Three-Phase	Short Circuit Trans	sient	T' _{d3}		
Line to Line SI	hort Circuit Transi	ent	T' _{d1}		
Short Circuit S	Subtransient		T" _d	T" _q	<u> </u>
Open Circuit S	Subtransient		T' _{d2}		
Line to Neutra	l Short Circuit Tra	nsient	T" _{do}	T" _{qo}	<u></u>
	ARMA	TURE TIM	E CONSTA	NT DATA (SEC)	
Three Phase S Line to Line SI Line to Neutra		T _{a3} T _{a2} T _{a1}			
NOTE: If requ	ested information	is not appli	icable, indic	ate by marking "N	//A."
				CONFIGURATION	N
	ARMATURE	WINDING	RESISTAN	ICE DATA (PER	UNIT)
Positive Negative Zero		R ₁ R ₂ R ₀			
Field Current a Field Current a Three Phase A Field Winding	me Thermal Capa at Rated kVA, Arm at Rated kVA and Armature Winding Resistance = ding Resistance (F	nature Volta Armature \ Capacitano ohms	/oltage, 0 F ce =°(s°(PF =amps _microfarad C	;
			CURVES		
					ture Correction curves for multiple curves.
	GENERATOR	STEP-UP	TRANSFO	RMER DATA RA	TINGS
Capacity	Self-cooled/M	laximum Na	ameplate		
		kVA			

Voltage Ratio (Generator Side/System s	ide/Tertiary)			
		kV		
Winding Connections (Low V/High V/Te	rtiary V (Delta o	r Wye))		
		_		
Fixed Taps Available				
Present Tap Setting				
If more than one transformer stage is u Facility to the Transmission System, ple or transformer type.				
	IMPEDANCE			
Positive Z ₁ (on self-cooled kVA rating)	%		_X/R	
Zero Z ₀ (on self-cooled kVA rating)	<u>%</u>		_X/R	
EXCITA	ATION SYSTEM	I DATA		
Identify appropriate IEEE model block d (PSS) for computer representation in poexcitation system and PSS constants for	ower system sta	ability simulati		
GOVE	RNOR SYSTEM	I DATA		
Identify appropriate IEEE model block of in power system stability simulations and in the model.				
WIND AND OTHER N	ON-SYNCHRO	NOUS GENE	RATORS	
Number of generators to be interco	onnected pursu	uant to this	Interconnection	Request:
Elevation: Single	Phase	Three Phas	se	
Inverter manufacturer, model name, nur	nber, and versio	n:		
List of adjustable setpoints for the protect	ctive equipment	or software:		

Note: A completed General Electric Company Power Systems Load Flow (PSLF) data sheet or other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate to the proposed device, then they shall be provided with the Interconnection Request and discussed at Scoping Meeting.

Project Information: Site Control and Adequacy

Total acres required to construct the Generating Facility:
Total acres under site control for the Generating Facility at the time of application:
Is Site Control required for Interconnection Facilities, i.e. transmission gen-tie or substation, to interconnect the Generating Facility? $___$ Y $___$ N
If yes, how many miles of gen-tie right-of-way are required? What is the total number of acres required to build the gen-tie? How many miles of gen-tie right-of-way are under Site Control at the time of this application?
List any local, state, or federal government permits required to construct the Generating Facility and any applicable Interconnection Facilities, i.e. transmission gen-tie:

INDUCTION GENERATORS

(*)	Field Volts:
	Field Amperes:
(*)	Motoring Power (kW):
(*)	Neutral Grounding Resistor (If Applicable:
(*)	l ₂ ² t or K (Heating Time Constant):
(*)	Rotor Resistance:
(*)	Stator Resistance:
(*)	Stator Reactance:
(*)	Rotor Reactance:
(*)	Magnetizing Reactance:
(*)	Short Circuit Reactance:
(*)	Exciting Current:
(*)	Temperature Rise:
(*)	Frame Size:
٠,	Design Letter:
(*)	Reactive Power Required In Vars (No Load):
(*)	Reactive Power Required In Vars (Full Load):
(*)	Total Rotating Inertia, H:Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Interconnection Request to determine if the information designated by (*) is required.

extended.

5.0

APPENDIX 2 to LGIP DEFINITIVE INTERCONNECTION SYSTEM IMPACT STUDY AGREEMENT

THIS	S AGREEMENT is made and entered into this day	of, 20 I	by
of	S AGREEMENT is made and entered into this day between, a organized the State of, ("Interconnection a existing unconnection as continuous existinuous e	ier the laws of the State	OT
Provid	, ("Transmission Provider"). Interconnection vider each may be referred to as a "Party," or collectively as the	Customer and Transmission	on
1 10114	RECITALS	r dido.	
genera	EREAS, Interconnection Customer is proposing to develop a erating capacity addition to an existing Generating Farconnection Request submitted by Interconnection Customer dates.	acility consistent with the	
	EREAS , Interconnection Customer desires to interconnect the Transmission System; and	e Large Generating Facili	ity
Definit	EREAS, Interconnection Customer has requested Transmis nitive Interconnection System Impact Study to assess the irge Generating Facility to the Transmission System, and of any A	npact of interconnecting the	
	EREAS , Interconnection Customer commits to provide ce ugh the Definitive Interconnection Study process as described in		es
	N, THEREFORE , in consideration of and subject to the mutual Parties agreed as follows:	covenants contained here	in
1.0	When used in this Agreement, with initial capitalization, the meanings indicated in Transmission Provider's FERC-ag		ve
2.0	Interconnection Customer elects and Transmission Provider a Definitive Interconnection System Impact Study con accordance with the Tariff.		
3.0	The scope of the Definitive Interconnection System Impact assumptions set forth in Attachment A to this Agreement.	Study shall be subject to the	he
4.0	The Definitive Interconnection System Impact Study shall information provided by Interconnection Customer in the subject to any modifications in accordance with Section 4. Provider reserves the right to request additional Interconnection Customer as may reasonably become necutility Practice during the course of the Definitive Interconnection If Interconnection Customer makes a permissible modificated LGIP, the time to complete the Definitive Interconnection States.	e Interconnection Request 4 of the LGIP. Transmission technical information frout tessary consistent with Good tection System Impact Studiation in accordance with the	st, on om od dy. he

The Definitive Interconnection System Impact Study reports (though Phase 2, and

including Phase 3 if applicable) shall provide the following information as appropriate:

- identification of any circuit breaker short circuit capability limits exceeded as a result of the interconnection;
- identification of any thermal overload or voltage limit violations resulting from the interconnection;
- identification of any instability or inadequately damped response to system disturbances resulting from the interconnection; and
- description and non-binding, good faith estimated cost of facilities required to interconnect the Large Generating Facility to the Transmission System and to address the identified short circuit, instability, and power flow issues.
- 6.0 Interconnection Customer shall provide the deposit as specified in Section 3.4 of the LGIP for the performance of the Definitive Interconnection System Impact Study and the Interconnection Facilities Study. Transmission Provider's good faith estimate for the time of completion of the Definitive Interconnection System Impact Study (Phase 2) is [insert date].

Upon receipt of the Interconnection Facilities Study results (Phase 4 Results), or withdrawal of the Interconnection Request, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Definitive Interconnection System Impact Study and the actual costs of the Interconnection Facilities Study, and the Withdrawal Penalty, as applicable, allocated according to the LGIP.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, except as otherwise provided herein. As provided in the LGIP, Interconnection Customer has thirty (30) Calendar Days of receipt of an invoice from Transmission Provider to pay any undisputed costs. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

7.0 Miscellaneous. The Definitive Interconnection System Impact Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, that are consistent with regional practices, Applicable Laws and Regulations and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS THEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

Ву:	By:			
	Title:			
Date:	_ Date:			
[Insert name of Interconnection Customer]				
Ву:	_			
Title:	_			
Date:				

[Insert name of Transmission Provider or Transmission Owner, if applicable]

Attachment A to Appendix 2 **Definitive Interconnection System Impact Study Agreement**

ASSUMPTIONS USED IN CONDUCTING THE **DEFINITIVE INTERCONNECTION SYSTEM IMPACT STUDY**

The Definitive Interconnection System Impact Study shall be based upon the information set forth in the Interconnection Request(s) and results of applicable prior studies, subject to any modifications in accordance with Section 4.4 of the LGIP, and the following assumptions:

Designation of Point of Interconnection and configuration to be studied.

[Above assumptions to be completed by Interconnection Customer and other assumptions to be provided by Interconnection Customer and Transmission Provider]

APPENDIX 3 to LGIP INTERCONNECTION FACILITIES STUDY AGREEMENT

THIS AGREEMENT is made and entered and between, a of the State of a, ("Transmission Provider each may be referred to as a "Party	/ider "). Ir	organized and ("Interconnection existing under the organized connection Customates")	, 20 by I existing under the laws Customer,") and ne laws of the State of tomer and Transmission ties."			
RECITALS						
WHEREAS, Interconnection Customer is p generating capacity addition to an extinterconnection Request submitted by Interconnection	xisting (Senerating Facility	consistent with the			
WHEREAS , Interconnection Customer desires to interconnect the Large Generating Facility with the Transmission System under (check one only):						
Network Resource Interconne	ection Se	rvice				
Energy Resource Interconne	ction Ser	vice;				

WHEREAS, Transmission Provider has completed a Definitive Interconnection System Impact Study (the "System Impact Study") and provided the results of said study to Interconnection Customer: and

WHEREAS. Interconnection Customer has requested Transmission Provider to perform an Interconnection Facilities Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to implement the conclusions of the Definitive Interconnection System Impact Study in accordance with Good Utility Practice to physically and electrically connect the Large Generating Facility to the Transmission System.

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- 1.0 When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- 2.0 Interconnection Customer elects and Transmission Provider shall cause an Interconnection Facilities Study consistent with this LGIP to be performed in accordance with the Tariff.
- 3.0 The scope of the Interconnection Facilities Study shall be subject to the assumptions set forth in Attachment A to this Agreement.
- 4.0 The Interconnection Facilities Study report (i) shall provide a description of, estimated cost of, schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the Definitive Interconnection System Impact Study.

- 5.0 Interconnection Customer shall meet the requirements specified under the LGIP prior to the performance of the Interconnection Facilities Study. The time for completion of the Interconnection Facilities Study is specified in Attachment A.
- 6.0 Interconnection Customer shall have provided the deposit as specified in Section 3.4.1 of the LGIP for the performance of the Definitive Interconnection System Impact Study and the Interconnection Facilities Study.

Upon receipt of the Interconnection Facilities Study results (Phase 4 Results), Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Definitive Interconnection System Impact Study and the actual costs of the Interconnection Facilities Study, which includes costs allocated according to the LGIP, and the Withdrawal Penalty calculated pursuant to Section 3.7.1, if applicable.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate, except as otherwise provided herein. As provided in the LGIP, Interconnection Customer has thirty (30) Calendar Days of receipt of an invoice from Transmission Provider to pay any undisputed costs. If invoices are not paid within thirty (30) Calendar Days of receipt of an invoice, Transmission Provider shall draw upon the security provided to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]				
Ву:	By:	_		
Title:	Title:	_		
Date:	Date:			
[Insert name of Interconnection	on Customer]			
Ву:				
Title:				
Date:				

DATA FORM TO BE PROVIDED BY INTERCONNECTION CUSTOMER WITH THE INTERCONNECTION FACILITIES STUDY AGREEMENT

Provide location plan and simplified one-line diagram of the plant and station facilities. For staged projects, please indicate future generation, transmission circuits, etc.

One set of metering is required for each generation connection to the new ring bus or existing Transmission Provider station. Number of generation connections:

On the one line diagram indicate the generation capacity attached at each metering location. (Maximum load on CT/PT)

On the one line diagram indicate the location of auxiliary power. (Minimum load on CT/PT) Amps

Will an alternate source of auxiliary power be available during CT/PT maintenance?Yes No
Will a transfer bus on the generation side of the metering require that each meter set be designed for the total plant generation? Yes No (Please indicate on one line diagram).
What type of control system or PLC will be located at Interconnection Customer's Large Generating Facility?
What protocol does the control system or PLC use?
Please provide a 7.5-minute quadrangle of the site. Sketch the plant, station, transmission line, and property line.
Physical dimensions of the proposed interconnection station:
Bus length from generation to interconnection station:
Line length from interconnection station to Transmission Provider's transmission line.
Tower number observed in the field. (Painted on tower leg)*
Number of third party easements required for transmission lines*:

^{*} To be completed in coordination with Transmission Provider.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Is the Large Generating Facility in Transmission Provider's service area?				
Yes No	Local	provider:		
Please provide proposed schedule dates:				
Begin Construction	Date:	-		
Generator step-up transformer receives back feed power	Date:	-		
Generation Testing	Date:	-		
Commercial Operation	Date:	_		

4.0

APPENDIX 4.1 to LGIP

Transitional Interconnection Facilities Study Agreement

and be	tween_	State	of	l entered into th	organ	ized and ex	xisting unde	r the laws
			a a	ssion Provider" as a "Party," or	existing	under the	laws of the	State of
Provid	er each	may be i	referred to	as a "Party," or	collectively as	the "Parties	3."	113111133101
				RECIT	ALS			
genera	ating c	apacity	addition 1	tomer is propos to an existing d by Interconne	Generating	Facility	consistent	with the
			ection Cus System; an	tomer desires	to interconnec	t the Larg	e Generatir	g Facility
proces engine Systen	sing its ering, _l n Impa	Interconi procurem ct Study	nection Fac ent and co in accorda	stomer has re cilities Study to so construction work ance with Good cility to the Trans	specify and est needed to im Utility Praction	imate the complement the complement of the compl	cost of the e	quipment ons of the
an Int	erconne	ection Fa	acilities Stu	tomer has execudy Agreement on Study Proces	prior to the			
	ng a			omer has provid mission Provide				
		EFORE, i reed as f		ation of and sub	pject to the mu	tual covena	ants contain	ed hereir
1.0				nent, with initia Transmission P				shall have
2.0				elects and Tran ies Study cons				
3.0	forth in	n Attachn	nent A to t	nection Facilities this Agreement facilities Study <i>A</i>	which shall be	•		•

The Interconnection Facilities Study report (i) shall provide a description, estimated cost of (consistent with Attachment A), schedule for required facilities to interconnect the

Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the most recently published System Impact Study.

- 5.0 Interconnection Customer has met certain requirements described in the LGIP. The time for completion of the Interconnection Facilities Study is specified in Attachment A.
- 6.0 Interconnection Customer shall have previously provided the deposit for the performance of the Interconnection Facilities Study.

Upon receipt of the Interconnection Facilities Study results, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Interconnection Facilities Study, as allocated pursuant to the LGIP.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]				
Ву:	By:			
Title:	Title:			
Date:	Date:			
[Insert name of Interconnection Customer]				
Ву:	<u> </u>			
Title:	<u></u>			
Date:				

Attachment A to Appendix 4.1 Transitional Interconnection Facilities Study Agreement

ASSUMPTIONS USED IN CONDUCTING THE TRANSITIONAL INTERCONNECTION FACILITIES STUDY

[Assumptions to be completed by Interconnection Customer and Transmission Provider]

APPENDIX 4.2 to LGIP Transitional Cluster Study Agreement

organized and e ("Interconnection	, 20 by xisting under the laws Customer,") and			
nterconnection Custon	ner and Transmission			
s				
Generating Facility	consistent with the			
WHEREAS , Interconnection Customer desires to interconnect the Large Generating Facility with the Transmission System under (check one only):				
ervice				
rvice; and				
1	organized and e ("Interconnection existing under the nterconnection Custon llectively as the "Partie. g to develop a Large of Generating Facility on Customer dated interconnect the Large			

WHEREAS, Interconnection Customer has requested Transmission Provider to perform a "Transitional Cluster Study," which is a combined system impact and interconnection facility Cluster Study to specify and estimate the cost of the equipment, engineering, procurement and construction work needed to physically and electrically connect the Large Generating Facility to the Transmission System; and

WHEREAS, Interconnection Customer has provided certain requirements described in the LGIP including a deposit of five million dollars (\$5,000,000) on expected Transmission Provider's Interconnection Facilities and Network Upgrades; and

WHEREAS, Interconnection Customer has a valid Queue Position as of the effective date of Transmission Provider's Definitive Interconnection Study Process; and

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein the Parties agreed as follows:

- 1.0 When used in this Agreement, with initial capitalization, the terms specified shall have the meanings indicated in Transmission Provider's FERC-approved LGIP.
- 2.0 Interconnection Customer elects and Transmission Provider shall cause to be performed a combined system impact and interconnection facility Cluster Study.
- 3.0 The Transitional Cluster Study shall be based upon the technical information provided by Interconnection Customer in the Interconnection Request. Transmission Provider reserves the right to request additional technical information from Interconnection Customer as may reasonably become necessary consistent with Good Utility Practice during the course of the Transitional Cluster Study and Interconnection Customer shall

provide such data as quickly as reasonable.

- 4.0 The Transitional Cluster Study report shall provide the following information:
 - identification of any circuit breaker short circuit capability limits exceeded as a result of the interconnection;
 - identification of any thermal overload or voltage limit violations resulting from the interconnection:
 - identification of any instability or inadequately damped response to system disturbances resulting from the interconnection and
 - shall provide a description, estimated cost of, schedule for required facilities to interconnect the Large Generating Facility to the Transmission System and (ii) shall address the short circuit, instability, and power flow issues identified in the most recently published System Impact Study.
- 5.0 Interconnection Customer has met certain requirements described in the LGIP.
- 6.0 Interconnection Customer shall have previously provided a deposit for the performance of Interconnection Studies.

Upon receipt of the Transitional Cluster Study results, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Transitional Cluster Study.

Any difference between the study deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.

7.0 Miscellaneous. The Interconnection Facilities Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

[Insert name of Transmission Provider or Transmission Owner, if applicable]				
Ву:	By:			
Title:	Title:			
Date:	Date:			
[Insert name of Interconnection Customer]				
Ву:				
Title:				
Date:				

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Attachment A to Appendix 4.2 Transitional Cluster Study Agreement

ASSUMPTIONS USED IN CONDUCTING THE TRANSITIONAL CLUSTER STUDY (A COMBINED SYSTEM IMPACT AND INTERCONNECTION FACILITIES STUDY)

[Assumptions to be completed by Interconnection Customer and Transmission Provider]

APPENDIX 5.1 to LGIP INFORMATIONAL INTERCONNECTION STUDY REQUEST

1.	interconnection of its Generating Facility with Transmission Provider's Transmission System pursuant to the Tariff.					
2.	The type of interconnection service to be evaluated (check one): Energy Resource Interconnection Service Network Resource Interconnection Service					
3.	Interconnection Customer provides the following information:					
	a. Address or location or the proposed new Large Generating Facility site (to the extent known) or, in the case of an existing Generating Facility, the name and specific location of the existing Generating Facility;					
	b. Maximum summer at degrees C and winter at degrees C megawat electrical output of the proposed new Large Generating Facility or the amount o megawatt increase in the generating capacity of an existing Generating Facility;					
	c. General description of the equipment configuration;					
	d. Commercial Operation Date to be studied (Day, Month, and Year);					
	 e. Name, address, telephone number, and e-mail address of Interconnection Customer's contact person; 					
	Approximate location of the proposed Point of Interconnection;					
	g. Interconnection Customer Data (set forth in Attachment A)					
	h. Primary frequency response operating range for electric storage resources.					
	i. Requested capacity (in MW) of Interconnection Service (if lower than the					
	Generating Facility Capacity); and j. A Scope of Work including any additional information that may be reasonably required.					
4.	Study deposit amount as specified in the LGIP.					
5.	For study purposes, the point of delivery to deliver within the Control Area.					
6.	This Informational Interconnection Study Request shall be submitted to the representative indicated below:					
	[To be completed by Transmission Provider]					
7.	Representative of Interconnection Customer to contact:					
	[To be completed by Interconnection Customer]					

8. This Interconnection Request is submitted by:

Date: _____

Name of Interconnection Customer: By (signature): Name (type or print): _____

Attachment A to Appendix 5.1 Informational Interconnection Study Request

LARGE GENERATING FACILITY DATA

	UNIT RATINGS				
kVA	°F	Voltage			
Power Factor					
Speed (RPM)	Connection (e.g. Wye)				
	Frequency, Hertz				
Stator Amperes at Rated kVA		Field Volts			
Max Turbine MW	°F				
Primary frequency resp	onse operating range for e	lectric storage resources.			
Minimum State of Cha	rge:				
Maximum State of Cha					
COMBINED TUR	BINE-GENERATOR-EXCITI	ER INERTIA DATA			
Inertia Constant, H =	kW sec/kVA				
Moment-of-Inertia, WR ² =	lb. ft. ²				
REACTA	NCE DATA (PER UNIT-RA	TED KVA)			
DIF	DIRECT AXIS QUADRATURE AXIS				
Synchronous – saturated	X _{dv}	X _{qv}			
Synchronous – unsaturated	X_{di}	X_{qi}			
Transient – saturated	X'dv	X' _{qv}			
Transient – unsaturated	X'di	X'qi			
Subtransient – saturated	X"dv	X" _{qv}			
Subtransient – unsaturated	X"di	X"qi			
Negative Sequence – saturated	X2 _v				
Negative Sequence – unsaturate	d X2 _i				
Zero Sequence – saturated	X0 _v				
Zero Sequence – unsaturated	X0 _i				
Leakage Reactance	XI _m				

FIELD TIME CONSTANT DATA (SEC)

Open Circuit	T' _{do}	T'qo			
Three-Phase Short Circuit Transient	T' _{d3}				
Line to Line Short Circuit Transient	T' _{d1}	<u> </u>			
Short Circuit Subtransient	T" _d	T" _q			
Open Circuit Subtransient	T' _{d2}	<u> </u>			
Line to Neutral Short Circuit Transient	T" _{do}	T" _{qo}			
ARMATURE 1	TIME CONST	ANT DATA (SEC)			
Three Phase Short Circuit T_{a3} Line to Line Short Circuit T_{a2} Line to Neutral Short Circuit T_{a1}					
NOTE: If requested information is not a	pplicable, ind	cate by marking "N/A."			
MW CAPABILITY AND PLANT CONFIGURATION LARGE GENERATING FACILITY DATA					
ARMATURE WINDI	NG RESISTA	NCE DATA (PER UNIT)			
$\begin{array}{ccc} \text{Positive} & & R_1 _ \\ \text{Negative} & & R_2 _ \\ \text{Zero} & & R_0 _ \\ \end{array}$					
Rotor Short Time Thermal Capacity l_2^2t Field Current at Rated kVA, Armature V Field Current at Rated kVA and Armatu Three Phase Armature Winding Capaci Field Winding Resistance = of Armature Winding Resistance (Per Pha	/oltage and P re Voltage, 0 tance = nms	PF =amps microfarad °C			
	CURVES				
Provide Saturation, Vee, Reactive Designate normal and emergency Hydronical Provide Saturation, Vee, Reactive Saturation,		capacity Temperature Correction curves e operating range for multiple curves.			
GENERATOR STEP-	UP TRANSF	ORMER DATA RATINGS			
Capacity Self-cooled/Maximum	n Nameplate				
/k	:VA				

Voltage Ratio (Generator Side/System s	side/Tertiary)		
	k\	/	
Winding Connections (Low V/High V/Te	rtiary V (Delta or Wy	/e))	
Fixed Taps Available			
Present Tap Setting			
If more than one transformer stage is uthe Transmission System, please protransformer type.			
	IMPEDANCE		
Positive Z ₁ (on self-cooled kVA rating)	%	X/R	
Zero Z ₀ (on self-cooled kVA rating)	%	X/R	
EXCIT	ATION SYSTEM DA	NTA	
Identify appropriate IEEE model block of (PSS) for computer representation in prexcitation system and PSS constants for	ower system stability		
GOVE	RNOR SYSTEM DA	ATA .	
Identify appropriate IEEE model block of in power system stability simulations are in the model.			
WI	ND GENERATORS		
Number of generators to be interce	onnected pursuant	to this Interconnection	n Request:
Elevation: Single	PhaseTI	hree Phase	
Inverter manufacturer, model name, nur	nber, and version:		
List of adjustable setpoints for the prote	ctive equipment or s	oftware:	

Note: A completed General Electric Company Power Systems Load Flow (PSLF) data sheet or other compatible formats, such as IEEE and PTI power flow models, must be supplied with the Interconnection Request. If other data sheets are more appropriate to the proposed device, then they shall be provided with the study request and discussed with the Transmission Provider in advance of the study.

INDUCTION GENERATORS

(*)) Field Volts:	
(*)	Field Amperes:	
(*)	Motoring Power (kW):	
(*)	Neutral Grounding Resistor (If Applicable:	
(*)) I ₂ ² t or K (Heating Time Constant):	
(*)	Rotor Resistance:	
(*)	Stator Resistance:	
(*)	Stator Reactance:	
(*)) Rotor Reactance:	
) Magnetizing Reactance:	<u>-</u>
(*)) Short Circuit Reactance:	_
(*)	Exciting Current:	
) Temperature Rise:	
(*)	Frame Size:	
٠,) Design Letter:	
٠,) Reactive Power Required In Vars (No Loa	,
(*)) Reactive Power Required In Vars (Full Loa	,
(*)) Total Rotating Inertia, H:	Per Unit on KVA Base

Note: Please consult Transmission Provider prior to submitting the Informational Interconnection Study Request to determine if the information designated by (*) is required.

4.0

purposes.

APPENDIX 5.2 to LGIP INFORMATIONAL INTERCONNECTION STUDY AGREEMENT

INIS AGREEMENT IS made and entered into thisday of	
20 by and between,	а
20 by and between, organized and existing under the laws of the State of,	
("Interconnection Customer,") and	a
existing under the laws of the State of, ("Transmission Provider""). Interconnecti Customer and Transmission Provider each may be referred to as a "Party," or collectively as t "Parties."	
RECITALS	
WHEREAS , Interconnection Customer is evaluating developing a Large Generating Facility generating capacity addition to an existing Generating Facility and	or
WHEREAS , Interconnection Customer is proposing to evaluate an interconnection with t Transmission System; and	:he
WHEREAS , Interconnection Customer has submitted to Transmission Provider an Information Interconnection Study Interconnection Request; and	nal
NOW, THEREFORE , in consideration of and subject to the mutual covenants contained here the Parties agree as follows:	ein
1.0 When used in this Agreement, with initial capitalization, the terms specified shall ha the meanings indicated in Transmission Provider's FERC-approved LGIP.	ıve
2.0 Interconnection Customer elects and Transmission Provider shall cause an Information Interconnection Study consistent with this LGIP to be performed in accordance with tariff.	
3.0 The scope of the Informational Interconnection Study shall be subject to t assumptions set forth in Attachment A to this Agreement.	:he

The Informational Interconnection Study report shall provide a sensitivity analysis based 5.0 on the assumptions specified by Interconnection Customer in Attachment A to this Agreement. The Informational Interconnection Study shall identify Transmission Provider's Interconnection Facilities and the Network Upgrades that may be required to provide transmission service or Interconnection Service based upon the assumptions specified by Interconnection Customer in Attachment A.

The Informational Interconnection Study shall be performed solely for informational

- 6.0 Interconnection Customer shall provide a deposit as identified in the LGIP for the performance of the Informational Interconnection Study. Transmission Provider's good faith estimate for the time of completion of the Informational Interconnection Study is [insert date].
 - Upon receipt of the Informational Interconnection Study, Transmission Provider shall charge and Interconnection Customer shall pay the actual costs of the Informational Interconnection Study.
 - Any difference between the initial deposit and the actual cost of the study shall be paid by or refunded to Interconnection Customer, as appropriate.
- 7.0 Miscellaneous. The Informational Interconnection Study Agreement shall include standard miscellaneous terms including, but not limited to, indemnities, representations, disclaimers, warranties, governing law, amendment, execution, waiver, enforceability and assignment, that reflect best practices in the electric industry, and that are consistent with regional practices, Applicable Laws and Regulations, and the organizational nature of each Party. All of these provisions, to the extent practicable, shall be consistent with the provisions of the LGIP and the LGIA.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their duly authorized officers or agents on the day and year first above written.

•	
Ву:	By:
Title:	
Date:	_ Date:
[Insert name of Interconnection Custor	mer]
Ву:	_
Title:	_
Date:	

[Insert name of Transmission Provider or Transmission Owner, if applicable]

Appendix 6. STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT (LGIA)

> **Appendix 6 to the Standard Large Generator Interconnection Procedures**

STANDARD LARGE GENERATOR **INTERCONNECTION AGREEMENT (LGIA)**

(Applicable to Generating Facilities that exceed 20 MW)

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Note: This form of agreement is also used for a Small Generating Facility seeking Network Resource Interconnection Service. Small Generating Facilities are studied under the Large Generator Interconnection Procedures.

Appendix 6. STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT (LGIA)

TABLE OF CONTENTS

\mathbf{r}			
К	eci	ta	ls

Article 1.	Definitions
Article 2.	Effective Date, Term, and Termination
2.1	
2.2	
2.3	
	2.3.1 Written Notice
	2.3.2 Default
2.4	
	Disconnection
2.6	Survival
Article 3.	Regulatory Filings
3.1	Filing
Article 4.	Scope of Service
4.1	Interconnection Product Options
	4.1.1 Energy Resource Interconnection Service
	4.1.1.1 The Product
	4.1.1.2 Transmission Delivery Service Implications
	4.1.2 Network Resource Interconnection Service
	4.1.2.1 The Product
	4.1.2.2 Transmission Delivery Service Implications
4.2	
4.3	
4.4	5
4.5	Interconnection Customer Provided Services
Article 5.	Interconnection Facilities Engineering, Procurement, and Construction
5.1	Options
	5.1.1 Standard Option
	5.1.2 Alternate Option
	5.1.3 Option to Build
	5.1.4 Negotiated Option
5.2	General Conditions Applicable to Option to Build

- 5.4 Power System Stabilizers
- 5.5 **Equipment Procurement**
- 5.6 **Construction Commencement**
- 5.7 Work Progress
- 5.8 Information Exchange
- 5.9 Other Interconnection Options
 - 5.9.1 Limited Operation
 - 5.9.2 Provisional Interconnection Service
- Interconnection Customer's Interconnection Facilities ('ICIF')
 - 5.10.1 Interconnection Customer's Interconnection Facility Specifications
 - 5.10.2 Transmission Provider's Review
 - 5.10.3 ICIF Construction
- 5.11 Transmission Provider's Interconnection Facilities Construction
- 5.12 Access Rights
- 5.13 Lands of Other Property Owners
- 5.14 Permits
- 5.15 Early Construction of Base Case Facilities
- 5.16 Suspension
 - 5.16.1 Effect of Missed Interconnection Customer Milestones
 - 5.16.2 Effect of Suspension; Parties' Obligations 5.17 Taxes
 - 5.17.1 Interconnection Customer Payments Not Taxable
 - 5.17.2 Representations and Covenants
 - 5.17.3 Indemnification for the Cost Consequences of Current Tax Liability Imposed Upon the Transmission Provider
 - 5.17.4 Tax Gross-Up Amount
 - 5.17.5 Private Letter Ruling or Change or Clarification of Law
 - 5.17.6 Subsequent Taxable Events
 - 5.17.7 Contests
 - 5.17.8 Refund
 - 5.17.9 Taxes Other Than Income Taxes
 - 5.17.10Transmission Owners Who Are Not Transmission Providers
- 5.18 Tax Status
- 5.19 Modification
 - 5.19.1 General.
 - 5.19.2 Standards
 - 5.19.3 Modification Costs
- Article 6. Testing and Inspection
 - Pre-Commercial Operation Date Testing and Modifications 6.1
 - Post-Commercial Operation Date Testing and Modifications 6.2
 - 6.3 Right to Observe Testing
 - Right to Inspect 6.4

Article 7.	Metering
7.1	General
7.2	Check Meters
7.3	Standards
7.4	Testing of Metering Equipment
7.5	Metering Data
Article 8.	Communications
8.1	Interconnection Customer Obligations
8.2	
8.3	
8.4	Provision of Data from a Variable Energy Resource
Article 9.	Operations
9.1	
9.2	
9.3	
9.4	ϵ
9.5	Start-Up and Synchronization
9.6	Reactive Power and Primary Frequency Response
	9.6.1 Power Factor Design Criteria
	9.6.2 Voltage Schedules
	9.6.2.1 Voltage Regulators
	9.6.3 Payment for Reactive Power
	9.6.4 Primary Frequency Response
	9.6.4.1 Governor or Equivalent Controls
	9.6.4.2 Timely and Sustained Response
	9.6.4.3 Exemptions
	9.6.4.4 Electric Storage Resources
9.7	Outages and Interruptions
	9.7.1 Outages.
	9.7.1.1 Outage Authority and Coordination
	9.7.1.2 Outage Schedules
	9.7.1.3 Outage Restoration
	9.7.2 Interruption of Service
	9.7.3 Under-Frequency and Over Frequency Conditions
	9.7.4 System Protection and Other Control Requirements
	9.7.4.1 System Protection Facilities
	9.7.5 Requirements for Protection
	9.7.6 Power Quality
9.8	Switching and Tagging Rules
9.9	Use of Interconnection Facilities by Third Parties

9.10

9.9.1 Purpose of Interconnection Facilities 9.9.2 Third Party Users Disturbance Analysis Data Exchange
Maintenance

Article 10.

- 10.1 Transmission Provider Obligations
- 10.2 Interconnection Customer Obligations
- 10.3 Coordination
- 10.4 Secondary Systems
- 10.5 Operating and Maintenance Expenses

Article 11. Performance Obligation

- 11.1 Interconnection Customer Interconnection Facilities
- Transmission Provider's Interconnection Facilities 11.2
- 11.3 Network Upgrades and Distribution Upgrades
- 11.4 Transmission Credits
 - 11.4.1 Repayment of Amounts Advanced for Network Upgrades 11.4.2 Special Provisions for Affected Systems
- 11.5 Provision of Security
- 11.6 Interconnection Customer Compensation
- 11.6.1 Interconnection Customer Compensation for Actions During Emergency Condition

Article 12. Invoice

- 12.1 General
- 12.2 Final Invoice
- 12.3 Payment
- 12.4 Disputes

Article 13. **Emergencies**

- 13.1 Definition
- 13.2 Obligations
- 13.3 Notice
- 13.4 Immediate Action
- 13.5 Transmission Provider Authority
 - 13.5.1 General
 - 13.5.2 Reduction and Disconnection
- 13.6 Interconnection Customer Authority
- 13.7 Limited Liability

Article 14. Regulatory Requirements and Governing Law

- Regulatory Requirements 14.1
- 14.2 Governing Law

15.2 15.3	Notices General Billings and Payments Alternative Forms of Notice Operations and Maintenance Notice
	Force Majeure Force Majeure
Article 17. 17.1	Default Default 17.1.1 General 17.1.2 Right to Terminate
Article 18. 18.1	Indemnity, Consequential Damages and Insurance Indemnity 18.1.1 Indemnified Person 18.1.2 Indemnifying Party 18.1.3 Indemnity Procedures
	Consequential Damages Insurance
	Assignment Assignment
	Severability Severability
	Comparability Comparability
Article 22. 22.1	Confidentiality Confidentiality 22.1.1 Term 22.1.2 Scope 22.1.3 Release of Confidential Information 22.1.4 Rights 22.1.5 No Warranties 22.1.6 Standard of Care 22.1.7 Order of Disclosure 22.1.8 Termination of Agreement

22.1.9 Remedies

22.1.10 Disclosure to FERC, its Staff, or a State

Article 23.	Environmental Releases
Article 24.	Information Requirements
24.1	<u>.</u>
24.2	Information Submission by Transmission Provider
	Updated Information Submission by Interconnection Customer
24.4	Information Supplementation
Article 25.	Information Access and Audit Rights
25.1	Information Access
	Reporting of Non-Force Majeure Events
25.3	e e e e e e e e e e e e e e e e e e e
25.4	Č
	25.4.1 Audit Rights Period for Construction-Related Accounts and Records
	25.4.2 Audit Rights Period for All Other Accounts and Records
25.5	Audit Results
Article 26.	Subcontractors
	General
	Responsibility of Principal
26.3	No Limitation by Insurance
20.5	Two Difficulties of Historice
Article 27.	Disputes
	Submission
27.2	External Arbitration Procedures
27.3	Arbitration Decisions
	Costs
= 7	
Article 28.	Representations, Warranties, and Covenants
28.1	General
	28.1.1 Good Standing
	28.1.2 Authority
	28.1.3 No Conflict
	28.1.4 Consent and Approval
Article 29.	Joint Operating Committee
29.1	Joint Operating Committee Joint Operating Committee
29.1	Joint Operating Committee
Article 30.	Miscellaneous
30.1	Binding Effect
30.2	Conflicts

- 30.3 Rules of Interpretation
- 30.4 Entire Agreement
- 30.5 No Third Party Beneficiaries
- 30.6 Waiver
- 30.7 Headings
- 30.8 Multiple Counterparts
- 30.9 Amendment
- 30.10 Modification by the Parties
- 30.11 Reservation of Rights
- 30.12 No Partnership
- Appendix A Interconnection Facilities, Network Upgrades, and Distribution Upgrades
- Appendix B Milestones
- Appendix C Interconnection Details
- Appendix D Security Arrangements Details
- Appendix E Commercial Operation Date
- Appendix F Addresses for Delivery of Notices and Billings

Standard Large Generator Interconnection Agreement

STANDARD LARGE GENERATOR INTERCONNECTION AGREEMENT

THIS STANDARD I ARCE CENERATOR INTERCONNECTION

THIS STANDARD LARGE GENERATOR INTERCONNECTION	
AGREEMENT ("Agreement") is made and entered into this day of	
20, by and between, a	
organized and existing under the laws of the State/Commonwealth of	
("Interconnection Customer" with a Large Generating Facility), and	
, a org	ganized
and existing under the laws of the State/Commonwealth of	
("Transmission Provider and/or Transmission Owner"). Interconnection Custom	ner and
Transmission Provider each may be referred to as a "Party" or collectively as the	
"Parties."	

Recitals

WHEREAS, Transmission Provider operates the Transmission System; and

WHEREAS, Interconnection Customer intends to own, lease and/or control and operate the Generating Facility identified as a Large Generating Facility in Appendix C to this Agreement; and,

WHEREAS, Interconnection Customer and Transmission Provider have agreed to enter into this Agreement for the purpose of interconnecting the Large Generating Facility with the Transmission System;

NOW, THEREFORE, in consideration of and subject to the mutual covenants contained herein, it is agreed:

When used in this Standard Large Generator Interconnection Agreement, terms with initial capitalization that are not defined in Article 1 shall have the meanings specified in the Article in which they are used or the Open Access Transmission Tariff (Tariff).

Article 1. Definitions

Adverse System Impact shall mean the negative effects due to technical or operational limits on conductors or equipment being exceeded that may compromise the safety and reliability of the electric system.

Affected System shall mean an electric system other than the Transmission Provider's Transmission System that may be affected by the proposed interconnection. **Affected System Operator** shall mean the entity that operates an Affected System.

Affiliate shall mean, with respect to a corporation, partnership or other entity, each such other corporation, partnership or other entity that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such corporation, partnership or other entity.

Ancillary Services shall mean those services that are necessary to support the transmission of capacity and energy from resources to loads while maintaining reliable operation of the Transmission Provider's Transmission System in accordance with Good Utility Practice.

Applicable Laws and Regulations shall mean all duly promulgated applicable federal, state and local laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders, permits and other duly authorized actions of any Governmental Authority.

Applicable Reliability Council shall mean the reliability council applicable to the Transmission System to which the Generating Facility is directly interconnected.

Applicable Reliability Standards shall mean the requirements and guidelines of NERC, the Applicable Reliability Council, and the Control Area of the Transmission System to which the Generating Facility is directly interconnected.

Base Case shall mean the base case power flow, short circuit, and stability data bases used for the Interconnection Studies by the Transmission Provider or Interconnection Customer.

Breach shall mean the failure of a Party to perform or observe any material term or condition of the Standard Large Generator Interconnection Agreement.

Breaching Party shall mean a Party that is in Breach of the Standard Large Generator Interconnection Agreement.

Business Day shall mean Monday through Friday, excluding Federal Holidays.

Calendar Day shall mean any day including Saturday, Sunday or a Federal Holiday.

Cluster shall mean a group of Interconnection Requests that are studied together for the purpose of conducting the Interconnection Studies. A Cluster is sometimes

referred to as a Queue Cluster.

Clustering shall mean the process whereby a group of Interconnection Requests is studied together, instead of serially, for the purpose of conducting the Interconnection Studies.

Commercial Operation shall mean the status of a Generating Facility that has commenced generating electricity for sale, excluding electricity generated during Trial Operation.

Commercial Operation Date of a unit shall mean the date on which the Generating Facility commences Commercial Operation as agreed to by the Parties pursuant to Appendix E to the Standard Large Generator Interconnection Agreement.

Confidential Information shall mean any confidential, proprietary or trade secret information of a plan, specification, pattern, procedure, design, device, list, concept, policy or compilation relating to the present or planned business of a Party, which is designated as confidential by the Party supplying the information, whether conveyed orally, electronically, in writing, through inspection, or otherwise.

Control Area shall mean an electrical system or systems bounded by interconnection metering and telemetry, capable of controlling generation to maintain its interchange schedule with other Control Areas and contributing to frequency regulation of the interconnection. A Control Area must be certified by the Applicable NERC Regional Reliability Entity. Control Area shall have the same meaning as Balancing Authority Area as defined by NERC.

Default shall mean the failure of a Breaching Party to cure its Breach in accordance with Article 17 of the Standard Large Generator Interconnection Agreement.

Dispute Resolution shall mean the procedure for resolution of a dispute between the Parties in which they will first attempt to resolve the dispute on an informal basis.

Definitive Interconnection Study Process ("Definitive Interconnection Study") shall mean the complete definitive study process inclusive of the DISIS Request Window, Customer Engagement Window, Definitive Interconnection System Impact Study, and the Interconnection Facilities Study. Both the Resource Solicitation Cluster and the DISIS Cluster are processed under the Definitive Interconnection Study Process.

Definitive Interconnection System Impact Study ("DISIS") shall mean an engineering study that evaluates the impact of a Cluster of Interconnection Requests on the safety and reliability of the Transmission System and, if applicable, an Affected System.

Definitive Interconnection System Impact Study Agreement ("DISIS Agreement") shall mean the form of agreement contained in the Appendix of the LGIP for conducting the Definitive Interconnection System Impact Study.

DISIS Request Window shall have the meaning set forth in Section 4.2 of the LGIP.

Distribution System shall mean the Transmission Provider's facilities and equipment used to transmit electricity to ultimate usage points such as homes and industries directly from nearby generators or from interchanges with higher voltage transmission networks which transport bulk power over longer distances. The voltage levels at which distribution systems operate differ among areas.

Distribution Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Distribution System at or beyond the Point of Interconnection to facilitate interconnection of the Generating Facility and render the transmission service necessary to effect Interconnection Customer's wholesale sale of electricity in interstate commerce. Distribution Upgrades do not include Interconnection Facilities.

Effective Date shall mean the date on which the Standard Large Generator Interconnection Agreement becomes effective upon execution by the Parties subject to acceptance by FERC, or if filed unexecuted, upon the date specified by FERC.

Emergency Condition shall mean a condition or situation: (1) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (2) that, in the case of a Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to Transmission Provider's Transmission System, Transmission Provider's Interconnection Facilities or the electric systems of others to which the Transmission Provider's Transmission System is directly connected; or (3) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Generating Facility or Interconnection Customer's Interconnection Facilities. System restoration and black start shall be considered Emergency Conditions; provided, that Interconnection Customer is not obligated by the Standard Large Generator Interconnection Agreement to possess black start capability.

Energy Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to connect its Generating Facility to the Transmission Provider's Transmission System to be eligible to deliver the Generating Facility's electric output using the existing firm or nonfirm capacity of the Transmission

Provider's Transmission System on an as available basis. Energy Resource Interconnection Service in and of itself does not convey transmission service.

Engineering & Procurement (E&P) Agreement shall mean an agreement that authorizes the Transmission Provider to begin engineering and procurement of long lead-time items necessary for the establishment of the interconnection in order to advance the implementation of the Interconnection Request.

Environmental Law shall mean Applicable Laws or Regulations relating to pollution or protection of the environment or natural resources.

Federal Power Act shall mean the Federal Power Act, as amended, 16 U.S.C. §§ 791a et seq.

FERC shall mean the Federal Energy Regulatory Commission (Commission) or its successor.

Force Majeure shall mean any act of God, labor disturbance, act of the public enemy, war, insurrection, riot, fire, storm or flood, explosion, breakage or accident to machinery or equipment, any order, regulation or restriction imposed by governmental, military or lawfully established civilian authorities, or any other cause beyond a Party's control. A Force Majeure event does not include acts of negligence or intentional wrongdoing by the Party claiming Force Majeure.

Generating Facility shall mean Interconnection Customer's device for the production and/or storage for later injection of electricity identified in the Interconnection Request, but shall not include the Interconnection Customer's Interconnection Facilities.

Generating Facility Capacity shall mean the net capacity of the Generating Facility and the aggregate net capacity of the Generating Facility where it includes multiple energy production devices.

Good Utility Practice shall mean any of the practices, methods and acts engaged in or approved by a significant portion of the electric industry during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act to the exclusion of all others, but rather to be acceptable practices, methods, or acts generally accepted in the region.

Governmental Authority shall mean any federal, state, local or other

governmental regulatory or administrative agency, court, commission, department, board, or other governmental subdivision, legislature, rulemaking board, tribunal, or other governmental authority having jurisdiction over the Parties, their respective facilities, or the respective services they provide, and exercising or entitled to exercise any administrative, executive, police, or taxing authority or power; provided, however, that such term does not include Interconnection Customer, Transmission Provider, or any Affiliate thereof.

Hazardous Substances shall mean any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "radioactive substances," "contaminants," "pollutants," "toxic pollutants" or words of similar meaning and regulatory effect under any applicable Environmental Law, or any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any applicable Environmental Law.

Initial Synchronization Date shall mean the date upon which the Generating Facility is initially synchronized and upon which Trial Operation begins.

In-Service Date shall mean the date upon which the Interconnection Customer reasonably expects it will be ready to begin use of the Transmission Provider's Interconnection Facilities to obtain back feed power.

Interconnection Customer shall mean any entity, including the Transmission Provider, Transmission Owner or any of the Affiliates or subsidiaries of either, that proposes to interconnect its Generating Facility with the Transmission Provider's Transmission System. Where the Interconnection Customer and the Transmission Provider are the same entity, application of the following provisions set forth herein is not required: provisions governing the posting of security (in Articles 5 and 11), provisions governing taxes and reimbursements for tax liability (in Article 5), provisions governing indemnity, consequential damages and insurance (in Article 18), and provisions governing billing and payment (in Articles 12 and 15) (invoices are not required, but may be generated and tendered for administrative use to aid in the identification and accounting of costs).

Interconnection Customer's Interconnection Facilities shall mean all facilities and equipment, as identified in Appendix A of the Standard Large Generator Interconnection Agreement, that are located between the Generating Facility and the Point of Change of Ownership, including any modification, addition, or upgrades to such facilities and equipment necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Customer's Interconnection Facilities are sole use facilities.

Interconnection Facilities shall mean the Transmission Provider's Interconnection Facilities and the Interconnection Customer's Interconnection Facilities. Collectively, Interconnection Facilities include all facilities and equipment between the Generating Facility and the Point of Interconnection, including any modification, additions or upgrades that are necessary to physically and electrically interconnect the Generating Facility to the Transmission Provider's Transmission System. Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Interconnection Facilities Study shall mean a study conducted by the Transmission Provider or a third party consultant for the Interconnection Customer to determine a list of facilities (including Transmission Provider's Interconnection Facilities and Network Upgrades as identified in the Definitive Interconnection System Impact Study), the cost of those facilities, and the time required to interconnect the Generating Facility with the Transmission Provider's Transmission System. The scope of the study is defined in Section 8 of the Standard Large Generator Interconnection Procedures.

Interconnection Facilities Study Agreement shall mean the form of agreement contained in Appendix 3 of the Standard Large Generator Interconnection Procedures for conducting the Interconnection Facilities Study.

Interconnection Request shall mean an Interconnection Customer's request, in the form of Appendix 1 to the Standard Large Generator Interconnection Procedures, in accordance with the Tariff, to interconnect a new Generating Facility, or to increase the capacity of, or make a Material Modification to the operating characteristics of, an existing Generating Facility that is interconnected with the Transmission Provider's Transmission System.

Interconnection Service shall mean the service provided by the Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to the Transmission Provider's Transmission System and enabling it to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Standard Large Generator Interconnection Agreement and, if applicable, the Transmission Provider's Tariff.

Interconnection Study shall mean any of the following studies: the Informational Interconnection Study, the Definitive Interconnection System Impact Study, and the Interconnection Facilities Study described in the Standard Large Generator Interconnection Procedures.

Interconnection Study Agreement shall mean any of the following agreements: the Informational Interconnection Study Agreement, the Definitive Interconnection

System Impact Study Agreement, or the Interconnection Facilities Study Agreement described in the Standard Large Generator Interconnection Procedures.

IRS shall mean the Internal Revenue Service.

Joint Operating Committee shall be a group made up of representatives from Interconnection Customers and the Transmission Provider to coordinate operating and technical considerations of Interconnection Service.

Large Generating Facility shall mean a Generating Facility having a Generating Facility Capacity of more than 20 MW.

Loss shall mean any and all losses relating to injury to or death of any person or damage to property, demand, suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's performance, or non performance of its obligations under the Standard Large Generator Interconnection Agreement on behalf of the indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the indemnifying Party.

Material Modification shall mean those modifications that have a material impact on the cost or timing of any Interconnection Request with a later or equal queue priority date.

Metering Equipment shall mean all metering equipment installed or to be installed at the Generating Facility pursuant to the Standard Large Generator Interconnection Agreement at the metering points, including but not limited to instrument transformers, MWh-meters, data acquisition equipment, transducers, remote terminal unit, communications equipment, phone lines, and fiber optics.

NERC shall mean the North American Electric Reliability Council or its successor organization.

Network Resource shall mean any designated generating resource owned, purchased, or leased by a Network Customer under the Network Integration Transmission Service Tariff. Network Resources do not include any resource, or any portion thereof, that is committed for sale to third parties or otherwise cannot be called upon to meet the Network Customer's Network Load on a non-interruptible basis.

Network Resource Interconnection Service shall mean an Interconnection Service that allows the Interconnection Customer to integrate its Large Generating Facility with the Transmission Provider's Transmission System (1) in a manner comparable to that in which the Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an RTO or ISO with market based congestion

management, in the same manner as Network Resources. Network Resource Interconnection Service in and of itself does not convey transmission service.

Network Upgrades shall mean the additions, modifications, and upgrades to the Transmission Provider's Transmission System required at or beyond the point at which the Interconnection Facilities connect to the Transmission Provider's Transmission System to accommodate the interconnection of the Large Generating Facility to the Transmission Provider's Transmission System.

Notice of Dispute shall mean a written notice of a dispute or claim that arises out of or in connection with the Standard Large Generator Interconnection Agreement or its performance.

OASIS shall mean the Transmission Provider's Open Access Same-Time Informational System.

Party or **Parties** shall mean Transmission Provider, Transmission Owner, Interconnection Customer or any combination of the above.

Phase ("Phase 1, Phase 2, Phase 3 or "Phase 4") shall mean a distinct part of the Definitive Study Process as described in Sections 7 and 8 of the LGIP.

Point of Change of Ownership shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Customer's Interconnection Facilities connect to the Transmission Provider's Interconnection Facilities.

Point of Interconnection shall mean the point, as set forth in Appendix A to the Standard Large Generator Interconnection Agreement, where the Interconnection Facilities connect to the Transmission Provider's Transmission System.

Provisional Interconnection Service shall mean Interconnection Service provided by Transmission Provider associated with interconnecting the Interconnection Customer's Generating Facility to Transmission Provider's Transmission System and enabling that Transmission System to receive electric energy and capacity from the Generating Facility at the Point of Interconnection, pursuant to the terms of the Provisional Large Generator Interconnection Agreement and, if applicable, the Tariff.

Provisional Large Generator Interconnection Agreement shall mean the interconnection agreement for Provisional Interconnection Service established between Transmission Provider and/or the Transmission Owner and the Interconnection Customer. This agreement shall take the form of the Large

Generator Interconnection Agreement, modified for provisional purposes. Provisional Large Generator Interconnection Agreements are not eligible for suspension.

Queue shall mean a queue for valid Interconnection Requests for the Definitive Interconnection Study Process.

Queue Position shall mean the order of a valid Interconnection Request, relative to all other pending valid Interconnection Requests, in the Definitive Interconnection Study Process. The Queue Position is established based upon the date and time Interconnection Customer satisfies all of the requirements of the LGIP to enter the Definitive Interconnection Study Process. Valid Interconnection Requests in a single Cluster are considered equal-queued.

Readiness Milestone(s) shall have the meaning set forth in the LGIP.

Reasonable Efforts shall mean, with respect to an action required to be attempted or taken by a Party under the Standard Large Generator Interconnection Agreement, efforts that are timely and consistent with Good Utility Practice and are otherwise substantially equivalent to those a Party would use to protect its own interests.

Resource Plan shall mean any process authorized or required by Applicable Laws and Regulations for, *inter alia*, the selection of Generating Facilities interconnected to the Transmission System of Transmission Provider.

Resource Planning Entity shall mean any entity subject to or conducting a Resource Solicitation Process.

Resource Solicitation Cluster shall mean a Cluster Study associated with a Resource Planning Process.

Resource Solicitation Process shall mean any process authorized or required by Applicable Laws and Regulations for the acquisition of Network Resources by an entity interconnected to the Transmission System of Transmission Provider.

Scoping Meeting shall mean the meeting between representatives of the Interconnection Customer and Transmission Provider conducted for the purpose of discussing the proposed Interconnection Request, alternative interconnection options, to exchange information including any transmission data and earlier study evaluations that would be reasonably expected to affect such interconnection options, to analyze such information, and to determine the potential feasible Points of Interconnection.

Site Control shall mean the exclusive land right to develop, construct, operate, and maintain the Generating Facility over the term of expected operation of the

Generating Facility. Site Control shall include the right to develop, construct, operate, and maintain Interconnection Customer's Interconnection Facilities. Site Control may be demonstrated by documentation reasonably demonstrating: (1) ownership of, a leasehold interest in, or a right to develop a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Customer's Interconnection Facilities; (2) an option to purchase or acquire a leasehold interest in a site of sufficient size to construct and operate the Generating Facility and associated Interconnection Facilities; or (3) any other documentation that clearly demonstrates the right of the Interconnection Customer to exclusively or occupy a site of sufficient size to construct and operate the Generating Facility. Site Control for any co-located project is demonstrated by a contract or other agreement demonstrating shared land use for all co-located projects that meet the aforementioned provisions of this Site Control definition.

Small Generating Facility shall mean a Generating Facility that has a Generating Facility Capacity of no more than 20 MW.

Stand Alone Network Upgrades shall mean Network Upgrades that are not part of an Affected System that an Interconnection Customer may construct without affecting day-to-day operations of the Transmission System during their construction. Both the Transmission Provider and the Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify them in Appendix A to the Standard Large Generator Interconnection Agreement. If the Transmission Provider and Interconnection Customer disagree about whether a particular Network Upgrade is a Stand Alone Network Upgrade, the Transmission Provider must provide the Interconnection Customer a written technical explanation outlining why the Transmission Provider does not consider the Network Upgrade to be a Stand Alone Network Upgrade within 15 days of its determination.

Standard Large Generator Interconnection Agreement (LGIA) shall mean the form of interconnection agreement applicable to an Interconnection Request pertaining to a Large Generating Facility that is included in the Transmission Provider's Tariff.

Standard Large Generator Interconnection Procedures (LGIP) shall mean the interconnection procedures applicable to an Interconnection Request pertaining to a Large Generating Facility that are included in the Transmission Provider's Tariff.

Surplus Interconnection Service shall mean any unneeded portion of Interconnection Service established in a Large Generator Interconnection Agreement, such that if Surplus Interconnection Service is utilized the total amount of Interconnection Service at the Point of Interconnection would remain the same.

System Protection Facilities shall mean the equipment, including necessary protection signal communications equipment, required to protect (1) the Transmission Provider's Transmission System from faults or other electrical disturbances occurring at the Generating Facility and (2) the Generating Facility from faults or other electrical system disturbances occurring on the Transmission Provider's Transmission System or on other delivery systems or other generating systems to which the Transmission Provider's Transmission System is directly connected.

Tariff shall mean the Transmission Provider's Tariff through which open access transmission service and Interconnection Service are offered, as filed with FERC, and as amended or supplemented from time to time, or any successor tariff.

Transmission Owner shall mean an entity that owns, leases or otherwise possesses an interest in the portion of the Transmission System at the Point of Interconnection and may be a Party to the Standard Large Generator Interconnection Agreement to the extent necessary.

Transmission Provider shall mean the public utility (or its designated agent) that owns, controls, or operates transmission or distribution facilities used for the transmission of electricity in interstate commerce and provides transmission service under the Tariff. The term Transmission Provider should be read to include the Transmission Owner when the Transmission Owner is separate from the Transmission Provider.

Transmission Provider's Interconnection Facilities shall mean all facilities and equipment owned, controlled or operated by the Transmission Provider from the Point of Change of Ownership to the Point of Interconnection as identified in Appendix A to the Standard Large Generator Interconnection Agreement, including any modifications, additions or upgrades to such facilities and equipment. Transmission Provider's Interconnection Facilities are sole use facilities and shall not include Distribution Upgrades, Stand Alone Network Upgrades or Network Upgrades.

Transmission System shall mean the facilities owned, controlled or operated by the Transmission Provider or Transmission Owner that are used to provide transmission service under the Tariff.

Trial Operation shall mean the period during which Interconnection Customer is engaged in on-site test operations and commissioning of the Generating Facility prior to Commercial Operation.

Variable Energy Resource shall mean a device for the production of electricity that is characterized by an energy source that: (1) is renewable; (2) cannot be stored by the facility owner or operator; and (3) has variability that is beyond the control of the facility owner or operator.

Withdrawal Penalty shall have the meaning set forth in Section 3.7 of the LGIP.

Article 2. Effective Date, Term, and Termination

- **2.1 Effective Date**. This LGIA shall become effective upon execution by the Parties subject to acceptance by FERC (if applicable), or if filed unexecuted, upon the date specified by FERC. Transmission Provider shall promptly file this LGIA with FERC upon execution in accordance with Article 3.1, if required.
- **2.2 Term of Agreement**. Subject to the provisions of Article 2.3, this LGIA shall remain in effect for a period of ten (10) years from the Effective Date or such other longer period as Interconnection Customer may request (Term to be specified in individual agreements) and shall be automatically renewed for each successive one-year period thereafter.

2.3 Termination Procedures.

- 2.3.1 Written Notice. This LGIA may be terminated by Interconnection Customer after giving Transmission Provider ninety (90) Calendar Days advance written notice, or by Transmission Provider notifying FERC after the Generating Facility permanently ceases Commercial Operation. This LGIA shall be terminated by Transmission Provider if the Generating Facility or a portion of the Generating Facility fails to achieve Commercial Operation by the Commercial Operation Date established in accordance with Section 4.4.5 of the LGIP, including any extension provided thereunder.
- **2.3.2 Default**. Either Party may terminate this LGIA in accordance with Article 17.
- **2.3.3.** Notwithstanding Articles 2.3.1 and 2.3.2, no termination shall become effective until the Parties have complied with all Applicable Laws and Regulations applicable to such termination, including the filing with FERC of a notice of termination of this LGIA, which notice has been accepted for filing by FERC.
- 2.4 Termination Costs. If a Party elects to terminate this Agreement pursuant to Article 2.3 above, each Party shall pay all costs incurred (including any cancellation costs relating to orders or contracts for Interconnection Facilities and equipment) or charges assessed by the other Party, as of the date of the other Party's receipt of such notice of termination, that are the responsibility of the Terminating Party under this LGIA. In the event of termination by a Party, the Parties shall use commercially Reasonable Efforts to mitigate the costs, damages

and charges arising as a consequence of termination. Upon termination of this LGIA, unless otherwise ordered or approved by FERC:

2.4.1 With respect to any portion of Transmission Provider's Interconnection Facilities that have not yet been constructed or installed, Transmission Provider shall to the extent possible and with Interconnection Customer's authorization cancel any pending orders of, or return, any materials or equipment for, or contracts for construction of, such facilities; provided that in the event Interconnection Customer elects not to authorize such cancellation, Interconnection Customer shall assume all payment obligations with respect to such materials, equipment, and contracts, and Transmission Provider shall deliver such material and equipment, and, if necessary, assign such contracts, to Interconnection Customer as soon as practicable, at Interconnection Customer's expense. To the extent that Interconnection Customer has already paid Transmission Provider for any or all such costs of materials or equipment not taken by Interconnection Customer, Transmission Provider shall promptly refund such amounts to Interconnection Customer, less any costs, including penalties incurred by Transmission Provider to cancel any pending orders of or return such materials, equipment, or contracts.

If an Interconnection Customer terminates this LGIA, it shall be responsible for all costs incurred in association with that Interconnection Customer's interconnection, including any cancellation costs relating to orders or contracts for Interconnection Facilities and equipment, and other expenses including any Network Upgrades for which Transmission Provider has incurred expenses and has not been reimbursed by Interconnection Customer.

- 2.4.2 Transmission Provider may, at its option, retain any portion of such materials, equipment, or facilities that Interconnection Customer chooses not to accept delivery of, in which case Transmission Provider shall be responsible for all costs associated with procuring such materials, equipment, or facilities.
- 2.4.3 With respect to any portion of the Interconnection Facilities, and any other facilities already installed or constructed pursuant to the terms of this LGIA, Interconnection Customer shall be responsible for all costs associated with the removal, relocation or other disposition or retirement of such materials, equipment, or facilities.
- **2.4.4** Transmission Provider shall refund the security provided under Section 10.3 of the LGIP including any accumulated interest, if applicable.

Notwithstanding the foregoing, prior to remitting such security, plus accumulated interest if applicable, Transmission Provider shall offset against such security, and accumulated interest, any unpaid costs or penalties arising out of this Agreement or the LGIP. Monies due the Interconnection Customer shall be remitted within 90 Calendar Days of termination.

- **2.4.5** In the event a Withdrawal Penalty applies to a termination under this section, the Withdrawal Penalty shall have the meaning of and be calculated pursuant to Section 3.7.1 of the LGIP.
- 2.5 Disconnection. Upon termination of this LGIA, the Parties will take all appropriate steps to disconnect the Large Generating Facility from the Transmission System. All costs required to effectuate such disconnection shall be borne by the terminating Party, unless such termination resulted from the non-terminating Party's Default of this LGIA or such non-terminating Party otherwise is responsible for these costs under this LGIA.
- 2.6 Survival. This LGIA shall continue in effect after termination to the extent necessary to provide for final billings and payments and for costs incurred hereunder, including billings and payments pursuant to this LGIA; to permit the determination and enforcement of liability and indemnification obligations arising from acts or events that occurred while this LGIA was in effect; and to permit each Party to have access to the lands of the other Party pursuant to this LGIA or other applicable agreements, to disconnect, remove or salvage its own facilities and equipment.

Article 3. Regulatory Filings

3.1 Filing. Transmission Provider shall file this LGIA (and any amendment hereto) with the appropriate Governmental Authority, if required. Interconnection Customer may request that any information so provided be subject to the confidentiality provisions of Article 22. If Interconnection Customer has executed this LGIA, or any amendment thereto, Interconnection Customer shall reasonably cooperate with Transmission Provider with respect to such filing and to provide any information reasonably requested by Transmission Provider needed to comply with applicable regulatory requirements.

Article 4. Scope of Service

4.1 Interconnection Product Options. Interconnection Customer has selected the following (checked) type of Interconnection Service:

4.1.1 Energy Resource Interconnection Service

- 4.1.1.1 The Product. Energy Resource Interconnection Service allows Interconnection Customer to connect the Large Generating Facility to the Transmission System and be eligible to deliver the Large Generating Facility's output using the existing firm or non-firm capacity of the Transmission System on an "as available" basis. To the extent Interconnection Customer wants to receive Energy Resource Interconnection Service, Transmission Provider shall construct facilities identified in Attachment A.
- 4.1.1.2 **Transmission Delivery Service Implications.** Under Energy Resource Interconnection Service, Interconnection Customer will be eligible to inject power from the Large Generating Facility into and deliver power across the interconnecting Transmission Provider's Transmission System on an "as available" basis up to the amount of MWs identified in the applicable stability and steady state studies to the extent the upgrades initially required to qualify for Energy Resource Interconnection Service have been constructed. Where eligible to do so (e.g., PJM, ISONE, NYISO), Interconnection Customer may place a bid to sell into the market up to the maximum identified Large Generating Facility output, subject to any conditions specified in the interconnection service approval, and the Large Generating Facility will be dispatched to the extent Interconnection Customer's bid clears. In all other instances, no transmission delivery service from the Large Generating Facility is assured, but Interconnection Customer may obtain Point-to-Point Transmission Service, Network Integration Transmission Service, or be used for secondary network transmission service, pursuant to Transmission Provider's Tariff, up to the maximum output identified in the stability and steady state studies. In those instances, in order for Interconnection Customer to obtain the right to deliver or inject energy beyond the Large Generating Facility Point of Interconnection or to improve its ability to do so, transmission delivery service must be obtained pursuant to the provisions of Transmission Provider's Tariff. The Interconnection Customer's ability to inject its Large Generating Facility output beyond the Point of Interconnection, therefore, will depend on the existing

capacity of Transmission Provider's Transmission System at such time as a transmission service request is made that would accommodate such delivery. The provision of firm Point-to-Point Transmission Service or Network Integration Transmission Service may require the construction of additional Network Upgrades.

4.1.2 Network Resource Interconnection Service.

- 4.1.2.1 The Product. Transmission Provider must conduct the necessary studies and construct the Network Upgrades needed to integrate the Large Generating Facility (1) in a manner comparable to that in which Transmission Provider integrates its generating facilities to serve native load customers; or (2) in an ISO or RTO with market based congestion management, in the same manner as all Network Resources. To the extent Interconnection Customer wants to receive Network Resource Interconnection Service, Transmission Provider shall construct the facilities identified in Attachment A to this LGIA.
- 4.1.2.2 **Transmission Delivery Service Implications.** Network Resource Interconnection Service allows Interconnection Customer's Large Generating Facility to be designated by any Network Customer under the Tariff on Transmission Provider's Transmission System as a Network Resource, up to the Large Generating Facility's full output, on the same basis as existing Network Resources interconnected to Transmission Provider's Transmission System, and to be studied as a Network Resource on the assumption that such a designation will occur. Although Network Resource Interconnection Service does not convey a reservation of transmission service, any Network Customer under the Tariff can utilize its network service under the Tariff to obtain delivery of energy from the interconnected Interconnection Customer's Large Generating Facility in the same manner as it accesses Network Resources. A Large Generating Facility receiving Network Resource Interconnection Service may also be used to provide Ancillary Services after technical studies and/or periodic analyses are performed with respect to the Large Generating Facility's ability to provide any applicable Ancillary Services, provided that such studies and analyses have been or would be required in connection with

the provision of such Ancillary Services by any existing Network Resource. However, if an Interconnection Customer's Large Generating Facility has not been designated as a Network Resource by any load, it cannot be required to provide Ancillary Services except to the extent such requirements extend to all generating facilities that are similarly situated. The provision of Network Integration Transmission Service or firm Point-to-Point Transmission Service may require additional studies and the construction of additional upgrades. Because such studies and upgrades would be associated with a request for delivery service under the Tariff, cost responsibility for the studies and upgrades would be in accordance with FERC's policy for pricing transmission delivery services. Network Resource Interconnection Service does not necessarily provide Interconnection Customer with the capability to physically deliver the output of its Large Generating Facility to any particular load on Transmission Provider's Transmission System without incurring congestion costs. In the event of transmission constraints on Transmission Provider's Transmission System, Interconnection Customer's Large Generating Facility shall be subject to the applicable congestion management procedures in Transmission Provider's Transmission System in the same manner as Network Resources.

There is no requirement either at the time of study or interconnection, or at any point in the future, that Interconnection Customer's Large Generating Facility be designated as a Network Resource by a Network Service Customer under the Tariff or that Interconnection Customer identify a specific buyer (or sink). To the extent a Network Customer does designate the Large Generating Facility as a Network Resource, it must do so pursuant to Transmission Provider's Tariff.

Once an Interconnection Customer satisfies the requirements for obtaining Network Resource Interconnection Service, any future transmission service request for delivery from the Large Generating Facility within Transmission Provider's Transmission System of any amount of capacity and/or energy, up to the amount initially studied, will not require that any additional studies be performed or that any further

upgrades associated with such Large Generating Facility be undertaken, regardless of whether or not such Large Generating Facility is ever designated by a Network Customer as a Network Resource and regardless of changes in ownership of the Large Generating Facility. However, the reduction or elimination of congestion or redispatch costs may require additional studies and the construction of additional upgrades.

To the extent Interconnection Customer enters into an arrangement for long term transmission service for deliveries from the Large Generating Facility outside Transmission Provider's Transmission System, such request may require additional studies and upgrades in order for Transmission Provider to grant such request.

- **4.2 Provision of Service**. Transmission Provider shall provide Interconnection Service for the Large Generating Facility at the Point of Interconnection.
- 4.3 Performance Standards. Each Party shall perform all of its obligations under this LGIA in accordance with Applicable Laws and Regulations, Applicable Reliability Standards, and Good Utility Practice, and to the extent a Party is required or prevented or limited in taking any action by such regulations and standards, such Party shall not be deemed to be in Breach of this LGIA for its compliance therewith. If such Party is a Transmission Provider or Transmission Owner, then that Party shall amend the LGIA and submit the amendment to FERC for approval.
- **4.4 No Transmission Delivery Service**. The execution of this LGIA does not constitute a request for, nor the provision of, any transmission delivery service under Transmission Provider's Tariff, and does not convey any right to deliver electricity to any specific customer or Point of Delivery.
- **4.5 Interconnection Customer Provided Services**. The services provided by Interconnection Customer under this LGIA are set forth in Article 9.6 and Article 13.5.1. Interconnection Customer shall be paid for such services in accordance with Article 11.6.

Article 5. Interconnection Facilities Engineering, Procurement, and Construction

5.1 Options. Unless otherwise mutually agreed to between the Parties, Interconnection Customer shall select the In-Service Date, Initial Synchronization Date, and Commercial Operation Date; and either Standard Option or Alternate

Option set forth below for completion of Transmission Provider's Interconnection Facilities and Network Upgrades as set forth in Appendix A, and such dates and selected option shall be set forth in Appendix B, Milestones. At the same time, Interconnection Customer shall indicate whether it elects to exercise the Option to Build set forth in Article 5.1.3 below. If the dates designated by Interconnection Customer are not acceptable to Transmission Provider, Transmission Provider shall so notify Interconnection Customer within thirty (30) Calendar Days. Upon receipt of the notification that Interconnection Customer's designated dates are not acceptable to Transmission Provider, the Interconnection Customer shall notify Transmission Provider within thirty (30) Calendar Days whether it elects to exercise the Option to Build if it has not already elected to exercise the Option to Build.

- 5.1.1 Standard Option. Transmission Provider shall design, procure, and construct Transmission Provider's Interconnection Facilities and Network Upgrades, using Reasonable Efforts to complete Transmission Provider's Interconnection Facilities and Network Upgrades by the dates set forth in Appendix B, Milestones. Transmission Provider shall not be required to undertake any action which is inconsistent with its standard safety practices, its material and equipment specifications, its design criteria and construction procedures, its labor agreements, and Applicable Laws and Regulations. In the event Transmission Provider reasonably expects that it will not be able to complete Transmission Provider's Interconnection Facilities and Network Upgrades by the specified dates, Transmission Provider shall promptly provide written notice to Interconnection Customer and shall undertake Reasonable Efforts to meet the earliest dates thereafter.
- **5.1.2 Alternate Option**. If the dates designated by Interconnection Customer are acceptable to Transmission Provider, Transmission Provider shall so notify Interconnection Customer within thirty (30) Calendar Days, and shall assume responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities by the designated dates.

If Transmission Provider subsequently fails to complete Transmission Provider's Interconnection Facilities by the In-Service Date, to the extent necessary to provide back feed power; or fails to complete Network Upgrades by the Initial Synchronization Date to the extent necessary to allow for Trial Operation at full power output, unless other arrangements are made by the Parties for such Trial Operation; or fails to complete the Network Upgrades by the Commercial Operation Date, as such dates are reflected in Appendix B, Milestones; Transmission Provider shall pay Interconnection Customer liquidated damages in accordance with Article 5.3, Liquidated Damages, provided, however, the dates designated by

Interconnection Customer shall be extended day for day for each day that the applicable RTO or ISO refuses to grant clearances to install equipment.

- 5.1.3 Option to Build. Interconnection Customer shall have the option to assume responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades on the dates specified in Article 5.1.2. Transmission Provider and Interconnection Customer must agree as to what constitutes Stand Alone Network Upgrades and identify such Stand Alone Network Upgrades in Appendix A. Except for Stand Alone Network Upgrades, Interconnection Customer shall have no right to construct Network Upgrades under this option.
- 5.1.4 Negotiated Option. If the dates designated by Interconnection Customer are not acceptable to Transmission Provider, the Parties shall in good faith attempt to negotiate terms and conditions (including revision of the specified dates and liquidated damages, the provision of incentives, or the procurement and construction of all facilities other than Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades if the Interconnection Customer elects to exercise the Option to Build under Article 5.1.3). If the Parties are unable to reach agreement on such terms and conditions, then, pursuant to article 5.1.1 (Standard Option), Transmission Provider shall assume responsibility for the design, procurement and construction of all facilities other than Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades if the Interconnection Customer elects to exercise the Option to Build.
- **5.2 General Conditions Applicable to Option to Build**. If Interconnection Customer assumes responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades,
 - (1) Interconnection Customer shall engineer, procure equipment, and construct Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades (or portions thereof) using Good Utility Practice and using standards and specifications provided in advance by Transmission Provider;
 - (2) Interconnection Customer's engineering, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades shall comply with all requirements of law to which Transmission Provider would be subject in the engineering, procurement or construction of Transmission Provider's Interconnection Facilities and

Stand Alone Network Upgrades;

- (3) Transmission Provider shall review and approve the engineering design, equipment acceptance tests, and the construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
- (4) prior to commencement of construction, Interconnection Customer shall provide to Transmission Provider a schedule for construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades, and shall promptly respond to requests for information from Transmission Provider;
- (5) at any time during construction, Transmission Provider shall have the right to gain unrestricted access to Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades and to conduct inspections of the same;
- (6) at any time during construction, should any phase of the engineering, equipment procurement, or construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades not meet the standards and specifications provided by Transmission Provider, Interconnection Customer shall be obligated to remedy deficiencies in that portion of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades;
- (7) Interconnection Customer shall indemnify Transmission Provider for claims arising from Interconnection Customer's construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades under the terms and procedures applicable to Article 18.1 Indemnity;
- (8) Interconnection Customer shall transfer control of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades to Transmission Provider;
- (9) Unless Parties otherwise agree, Interconnection Customer shall transfer ownership of Transmission Provider's Interconnection Facilities and Stand-Alone Network Upgrades to Transmission Provider;
- (10) Transmission Provider shall approve and accept for operation and maintenance Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades to the extent engineered, procured, and constructed in accordance with this Article 5.2; and

- (11) Interconnection Customer shall deliver to Transmission Provider "asbuilt" drawings, information, and any other documents that are reasonably required by Transmission Provider to assure that the Interconnection Facilities and Stand Alone Network Upgrades are built to the standards and specifications required by Transmission Provider.
- (12) If Interconnection Customer exercises the Option to Build pursuant to Article 5.1.3, Interconnection Customer shall pay Transmission Provider an agreed upon amount of [\$ PLACEHOLDER] for Transmission Provider to execute the responsibilities enumerated to Transmission Provider under Article 5.2. Transmission Provider shall invoice Interconnection Customer for this total amount to be divided on a monthly basis pursuant to Article 12.
- 5.3 Liquidated Damages. The actual damages to Interconnection Customer, in the event Transmission Provider's Interconnection Facilities or Network Upgrades are not completed by the dates designated by Interconnection Customer and accepted by Transmission Provider pursuant to subparagraphs 5.1.2 or 5.1.4, above, may include Interconnection Customer's fixed operation and maintenance costs and lost opportunity costs. Such actual damages are uncertain and impossible to determine at this time. Because of such uncertainty, any liquidated damages paid by Transmission Provider to Interconnection Customer in the event that Transmission Provider does not complete any portion of Transmission Provider's Interconnection Facilities or Network Upgrades by the applicable dates, shall be an amount equal to ½ of 1 percent per day of the actual cost of Transmission Provider's Interconnection Facilities and Network Upgrades, in the aggregate, for which Transmission Provider has assumed responsibility to design, procure and construct.

However, in no event shall the total liquidated damages exceed 20 percent of the actual cost of Transmission Provider's Interconnection Facilities and Network Upgrades for which Transmission Provider has assumed responsibility to design, procure, and construct. The foregoing payments will be made by Transmission Provider to Interconnection Customer as just compensation for the damages caused to Interconnection Customer, which actual damages are uncertain and impossible to determine at this time, and as reasonable liquidated damages, but not as a penalty or a method to secure performance of this LGIA. Liquidated damages, when the Parties agree to them, are the exclusive remedy for the Transmission Provider's failure to meet its schedule.

No liquidated damages shall be paid to Interconnection Customer if: (1) Interconnection Customer is not ready to commence use of Transmission

Provider's Interconnection Facilities or Network Upgrades to take the delivery of power for the Large Generating Facility's Trial Operation or to export power from the Large Generating Facility on the specified dates, unless Interconnection Customer would have been able to commence use of Transmission Provider's Interconnection Facilities or Network Upgrades to take the delivery of power for Large Generating Facility's Trial Operation or to export power from the Large Generating Facility, but for Transmission Provider's delay; (2) Transmission Provider's failure to meet the specified dates is the result of the action or inaction of Interconnection Customer or any other Interconnection Customer who has entered into an LGIA with Transmission Provider or any cause beyond Transmission Provider's reasonable control or reasonable ability to cure; (3) the Interconnection Customer has assumed responsibility for the design, procurement and construction of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades; or (4) the Parties have otherwise agreed.

- Power System Stabilizers. The Interconnection Customer shall procure, install, maintain and operate Power System Stabilizers in accordance with the guidelines and procedures established by the Applicable Reliability Council. Transmission Provider reserves the right to reasonably establish minimum acceptable settings for any installed Power System Stabilizers, subject to the design and operating limitations of the Large Generating Facility. If the Large Generating Facility's Power System Stabilizers are removed from service or not capable of automatic operation, Interconnection Customer shall immediately notify Transmission Provider's system operator, or its designated representative. The requirements of this paragraph shall not apply to wind generators.
- 5.5 Equipment Procurement. If responsibility for construction of Transmission Provider's Interconnection Facilities or Network Upgrades is to be borne by Transmission Provider, then Transmission Provider shall commence design of Transmission Provider's Interconnection Facilities or Network Upgrades and procure necessary equipment as soon as practicable after all of the following conditions are satisfied, unless the Parties otherwise agree in writing:
 - **5.5.1** Transmission Provider has completed the Interconnection Facilities Study pursuant to the Interconnection Facilities Study Agreement;
 - **5.5.2** Transmission Provider has received written authorization to proceed with design and procurement from Interconnection Customer by the date specified in Appendix B, Milestones; and
 - **5.5.3** Interconnection Customer has provided security to Transmission Provider in accordance with Article 11.5 by the dates specified in Appendix B, Milestones.

- **5.6** Construction Commencement. Transmission Provider shall commence construction of Transmission Provider's Interconnection Facilities and Network Upgrades for which it is responsible as soon as practicable after the following additional conditions are satisfied:
 - **5.6.1** Approval of the appropriate Governmental Authority has been obtained for any facilities requiring regulatory approval;
 - **5.6.2** Necessary real property rights and rights-of-way have been obtained, to the extent required for the construction of a discrete aspect of Transmission Provider's Interconnection Facilities and Network Upgrades;
 - **5.6.3** Transmission Provider has received written authorization to proceed with construction from Interconnection Customer by the date specified in Appendix B, Milestones; and
 - **5.6.4** Interconnection Customer has provided security to Transmission Provider in accordance with Article 11.5 by the dates specified in Appendix B, Milestones.
- 5.7 Work Progress. The Parties will keep each other advised periodically as to the progress of their respective design, procurement and construction efforts. Either Party may, at any time, request a progress report from the other Party. If, at any time, Interconnection Customer determines that the completion of Transmission Provider's Interconnection Facilities will not be required until after the specified In-Service Date, Interconnection Customer will provide written notice to Transmission Provider of such later date upon which the completion of Transmission Provider's Interconnection Facilities will be required.
- 5.8 Information Exchange. As soon as reasonably practicable after the Effective Date, the Parties shall exchange information regarding the design and compatibility of the Parties' Interconnection Facilities and compatibility of the Interconnection Facilities with Transmission Provider's Transmission System, and shall work diligently and in good faith to make any necessary design changes.

5.9 Other Interconnection Options

5.9.1. Limited Operation. If any of Transmission Provider's Interconnection Facilities or Network Upgrades are not reasonably expected to be completed prior to the Commercial Operation Date of the Large Generating Facility, Transmission Provider shall, upon the request and at the expense of Interconnection Customer, perform operating studies on a timely basis to

determine the extent to which the Large Generating Facility and Interconnection Customer's Interconnection Facilities may operate prior to the completion of Transmission Provider's Interconnection Facilities or Network Upgrades consistent with Applicable Laws and Regulations, Applicable Reliability Standards, Good Utility Practice, and this LGIA. Transmission Provider shall permit Interconnection Customer to operate the Large Generating Facility and Interconnection Customer's Interconnection Facilities in accordance with the results of such studies.

5.9.2 Provisional Interconnection Service. Upon the request of Interconnection Customer, and prior to completion of requisite Interconnection Facilities, Network Upgrades, Distribution Upgrades, or System Protection Facilities Transmission Provider may execute a Provisional Large Generator Interconnection Agreement or Interconnection Customer may request the filing of an unexecuted Provisional Large Generator Interconnection Agreement with the Interconnection Customer for limited Interconnection Service at the discretion of Transmission Provider based upon an evaluation that will consider the results of available studies. Transmission Provider shall determine, through available studies or additional studies as necessary, whether stability, short circuit, thermal, and/or voltage issues would arise if Interconnection Customer interconnects without modifications to the Generating Facility or Transmission System. **Transmission** Provider shall determine whether any Interconnection Facilities, Network Upgrades, Distribution Upgrades, or System Protection Facilities that are necessary to meet the requirements of NERC, or any applicable Regional Entity for the interconnection of a new, modified and/or expanded Generating Facility are in place prior to the commencement of Interconnection Service from the Generating Facility. Where available studies indicate that such, Interconnection Facilities, Network Upgrades, Distribution Upgrades, and/or System Protection Facilities that are required for the interconnection of a new, modified and/or expanded Generating Facility are not currently in place, Transmission Provider will perform a study, at the Interconnection Customer's expense, to confirm the facilities that are required for Provisional Interconnection Service. The maximum permissible output of the Generating Facility in the Provisional Large Generator Interconnection Agreement shall be studied and updated on an annual frequency at the Interconnection Customer's expense.

Interconnection Customer assumes all risk and liabilities with respect to changes between the Provisional Large Generator Interconnection Agreement and the Large Generator Interconnection Agreement, including changes in output limits and Interconnection Facilities, Network Upgrades, Distribution Upgrades, and/or System Protection Facilities cost responsibilities.

5.10 Interconnection Customer's Interconnection Facilities ('ICIF').

Interconnection Customer shall, at its expense, design, procure, construct, own and install the ICIF, as set forth in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades.

- 5.10.1 Interconnection Customer's Interconnection Facility Specifications.
 - Interconnection Customer shall submit initial specifications for the ICIF, including System Protection Facilities, to Transmission Provider at least one hundred eighty (180) Calendar Days prior to the Initial Synchronization Date; and final specifications for review and comment at least ninety (90) Calendar Days prior to the Initial Synchronization Date. Transmission Provider shall review such specifications to ensure that the ICIF are compatible with the technical specifications, operational control, and safety requirements of Transmission Provider and comment on such specifications within thirty (30) Calendar Days of Interconnection Customer's submission. All specifications provided hereunder shall be deemed confidential.
- **5.10.2 Transmission Provider's Review**. Transmission Provider's review of Interconnection Customer's final specifications shall not be construed as confirming, endorsing, or providing a warranty as to the design, fitness, safety, durability or reliability of the Large Generating Facility, or the ICIF. Interconnection Customer shall make such changes to the ICIF as may reasonably be required by Transmission Provider, in accordance with Good Utility Practice, to ensure that the ICIF are compatible with the technical specifications, operational control, and safety requirements of Transmission Provider.
- **5.10.3 ICIF Construction**. The ICIF shall be designed and constructed in accordance with Good Utility Practice. Within one hundred twenty (120) Calendar Days after the Commercial Operation Date, unless the Parties agree on another mutually acceptable deadline, Interconnection Customer shall deliver to Transmission Provider "as-built" drawings, information and documents for the ICIF, such as: a one-line diagram, a site plan showing the Large Generating Facility and the ICIF, plan and elevation drawings

showing the layout of the ICIF, a relay functional diagram, relaying AC and DC schematic wiring diagrams and relay settings for all facilities associated with Interconnection Customer's step-up transformers, the facilities connecting the Large Generating Facility to the step-up transformers and the ICIF, and the impedances (determined by factory tests) for the associated step-up transformers and the Large Generating Facility. The Interconnection Customer shall provide Transmission Provider specifications for the excitation system, automatic voltage regulator, Large Generating Facility control and protection settings, transformer tap settings, and communications, if applicable.

5.11 Transmission Provider's Interconnection Facilities Construction.

Transmission Provider's Interconnection Facilities shall be designed and constructed in accordance with Good Utility Practice. Upon request, within one hundred twenty (120) Calendar Days after the Commercial Operation Date, unless the Parties agree on another mutually acceptable deadline, Transmission Provider shall deliver to Interconnection Customer the following "as-built" drawings, information and documents for Transmission Provider's Interconnection Facilities (include appropriate drawings and relay diagrams).

Transmission Provider will obtain control of Transmission Provider's Interconnection Facilities and Stand Alone Network Upgrades upon completion of such facilities.

- 5.12 Access Rights. Upon reasonable notice and supervision by a Party, and subject to any required or necessary regulatory approvals, a Party ("Granting Party") shall furnish at no cost to the other Party ("Access Party") any rights of use, licenses, rights of way and easements with respect to lands owned or controlled by the Granting Party, its agents (if allowed under the applicable agency agreement), or any Affiliate, that are necessary to enable the Access Party to obtain ingress and egress to construct, operate, maintain, repair, test (or witness testing), inspect, replace or remove facilities and equipment to: (i) interconnect the Large Generating Facility with the Transmission System; (ii) operate and maintain the Large Generating Facility, the Interconnection Facilities and the Transmission System; and (iii) disconnect or remove the Access Party's facilities and equipment upon termination of this LGIA. In exercising such licenses, rights of way and easements, the Access Party shall not unreasonably disrupt or interfere with normal operation of the Granting Party's business and shall adhere to the safety rules and procedures established in advance, as may be changed from time to time, by the Granting Party and provided to the Access Party.
- **5.13 Lands of Other Property Owners**. If any part of Transmission Provider or Transmission Owner's Interconnection Facilities and/or Network Upgrades is to

be installed on property owned by persons other than Interconnection Customer or Transmission Provider or Transmission Owner, Transmission Provider or Transmission Owner shall at Interconnection Customer's expense use efforts, similar in nature and extent to those that it typically undertakes on its own behalf or on behalf of its Affiliates, including use of its eminent domain authority, and to the extent consistent with state law, to procure from such persons any rights of use, licenses, rights of way and easements that are necessary to construct, operate, maintain, test, inspect, replace or remove Transmission Provider or Transmission Owner's Interconnection Facilities and/or Network Upgrades upon such property.

- 5.14 Permits. Transmission Provider or Transmission Owner and Interconnection Customer shall cooperate with each other in good faith in obtaining all permits, licenses and authorizations that are necessary to accomplish the interconnection in compliance with Applicable Laws and Regulations. With respect to this paragraph, Transmission Provider or Transmission Owner shall provide permitting assistance to Interconnection Customer comparable to that provided to Transmission Provider's own, or an Affiliate's generation.
- 5.15 Early Construction of Base Case Facilities. Interconnection Customer may request Transmission Provider to construct, and Transmission Provider shall construct, using Reasonable Efforts to accommodate Interconnection Customer's In-Service Date, all or any portion of any Network Upgrades required for Interconnection Customer to be interconnected to the Transmission System which are included in the Base Case of the Facilities Study for Interconnection Customer, and which also are required to be constructed for another Interconnection Customer, but where such construction is not scheduled to be completed in time to achieve Interconnection Customer's In-Service Date.
- Suspension. Interconnection Customer reserves the right, upon written notice to 5.16 Transmission Provider, to suspend at any time all work by Transmission Provider associated with the construction and installation of Transmission Provider's Interconnection Facilities and/or Network Upgrades required under this LGIA with the condition that Transmission System shall be left in a safe and reliable condition in accordance with Good Utility Practice and Transmission Provider's safety and reliability criteria. In such event, Interconnection Customer shall be responsible for all reasonable and necessary costs which Transmission Provider (i) has incurred pursuant to this LGIA prior to the suspension and (ii) incurs in suspending such work, including any costs incurred to perform such work as may be necessary to ensure the safety of persons and property and the integrity of the Transmission System during such suspension and, if applicable, any costs incurred in connection with the cancellation or suspension of material, equipment and labor contracts which Transmission Provider cannot reasonably avoid; provided, however, that prior to canceling or suspending any such material, equipment or

labor contract, Transmission Provider shall obtain Interconnection Customer's authorization to do so.

Transmission Provider shall invoice Interconnection Customer for such costs pursuant to Article 12 and shall use due diligence to minimize its costs. In the event Interconnection Customer suspends work by Transmission Provider required under this LGIA pursuant to this Article 5.16, and has not requested Transmission Provider to recommence the work required under this LGIA on or before the expiration of three (3) years following commencement of such suspension, this LGIA shall be deemed terminated. The three-year period shall begin on the date the suspension is requested, or the date of the written notice to Transmission Provider, if no effective date is specified.

5.16.1 Effect of Missed Interconnection Customer Milestones.

If Interconnection Customer fails to provide notice of suspension pursuant to Article 5.15, and Interconnection Customer fails to fulfill or complete any Interconnection Customer milestone provided in Appendix B, this constitutes a Breach under this LGIA. Depending upon the consequences of the Breach and effectiveness of the cure pursuant to Article 17, Transmission Provider's Appendix B milestones may be revised following consultation with Interconnection Customer, consistent with Reasonable Efforts, and in consideration of all relevant circumstances. Parties shall employ Reasonable Efforts to maintain their remaining respective Appendix B milestones.

5.16.2 Effect of Suspension; Parties' Obligations.

In the event that Interconnection Customer suspends work pursuant to this Article 5.16, the applicable construction duration, timelines and schedules set forth in Appendix B shall be suspended during the period of suspension. Should Interconnection Customer thereafter request that work be recommenced, Appendix A and Appendix B may be revised to account for construction sequencing and modified milestones. If the Commercial Operation Date is extended beyond three (3) cumulative years described in Section 4.4.5 of the LGIP and Article 2.3.1 of this LGIA, such an extension may be considered a Material Modification and result in the termination of the LGIA under Article 2.3.1. Interconnection Customer is required to maintain Site Control while this LGIA is in effect, including during suspension.

5.17 Taxes.

- 5.17.1 Interconnection Customer Payments Not Taxable. The Parties intend that all payments or property transfers made by Interconnection Customer to Transmission Provider for the installation of Transmission Provider's Interconnection Facilities and the Network Upgrades shall be non-taxable, either as contributions to capital, or as an advance, in accordance with the Internal Revenue Code and any applicable state income tax laws and shall not be taxable as contributions in aid of construction or otherwise under the Internal Revenue Code and any applicable state income tax laws.
- 5.17.2 Representations and Covenants. In accordance with IRS Notice 2001-82 and IRS Notice 88-129, Interconnection Customer represents and covenants that (i) ownership of the electricity generated at the Large Generating Facility will pass to another party prior to the transmission of the electricity on the Transmission System, (ii) for income tax purposes, the amount of any payments and the cost of any property transferred to Transmission Provider for Transmission Provider's Interconnection Facilities will be capitalized by Interconnection Customer as an intangible asset and recovered using the straight-line method over a useful life of twenty (20) years, and (iii) any portion of Transmission Provider's Interconnection Facilities that is a "dual-use intertie." within the meaning of IRS Notice 88-129, is reasonably expected to carry only a de minimis amount of electricity in the direction of the Large Generating Facility. For this purpose, "de minimis amount" means no more than 5 percent of the total power flows in both directions, calculated in accordance with the "5 percent test" set forth in IRS Notice 88-129. This is not intended to be an exclusive list of the relevant conditions that must be met to conform to IRS requirements for non-taxable treatment.

At Transmission Provider's request, Interconnection Customer shall provide Transmission Provider with a report from an independent engineer confirming its representation in clause (iii), above. Transmission Provider represents and covenants that the cost of Transmission Provider's Interconnection Facilities paid for by Interconnection Customer will have no net effect on the base upon which rates are determined.

5.17.3 Indemnification for the Cost Consequences of Current Tax Liability Imposed Upon the Transmission Provider.

Notwithstanding Article 5.17.1, Interconnection Customer shall

protect, indemnify and hold harmless Transmission Provider from the cost consequences of any current tax liability imposed against Transmission Provider as the result of payments or property transfers made by Interconnection Customer to Transmission Provider under this LGIA for Interconnection Facilities, as well as any interest and penalties, other than interest and penalties attributable to any delay caused by Transmission Provider.

Transmission Provider shall not include a gross-up for the cost consequences of any current tax liability in the amounts it charges Interconnection Customer under this LGIA unless (i) Transmission Provider has determined, in good faith, that the payments or property transfers made by Interconnection Customer to Transmission Provider should be reported as income subject to taxation or (ii) any Governmental Authority directs Transmission Provider to report payments or property as income subject to taxation; provided, however, that Transmission Provider may require Interconnection Customer to provide security for Interconnection Facilities, in a form reasonably acceptable to Transmission Provider (such as a parental guarantee or a letter of credit), in an amount equal to the cost consequences of any current tax liability under this Article 5.17. Interconnection Customer shall reimburse Transmission Provider for such costs on a fully grossed-up basis, in accordance with Article 5.17.4, within thirty (30) Calendar Days of receiving written notification from Transmission Provider of the amount due, including detail about how the amount was calculated.

The indemnification obligation shall terminate at the earlier of (1) the expiration of the ten year testing period and the applicable statute of limitation, as it may be extended by Transmission Provider upon request of the IRS, to keep these years open for audit or adjustment, or (2) the occurrence of a subsequent taxable event and the payment of any related indemnification obligations as contemplated by this Article 5.17.

5.17.4 Tax Gross-Up Amount. Interconnection Customer's liability for the cost consequences of any current tax liability under this Article 5.17 shall be calculated on a fully grossed-up basis. Except as may otherwise be agreed to by the parties, this means that Interconnection Customer will pay Transmission Provider, in addition to the amount paid for the Interconnection Facilities and Network Upgrades, an amount equal to (1) the current taxes imposed on Transmission Provider ("Current Taxes") on the excess of (a) the gross income

5.17.5

realized by Transmission Provider as a result of payments or property transfers made by Interconnection Customer to Transmission Provider under this LGIA (without regard to any payments under this Article 5.17) (the "Gross Income Amount") over (b) the present value of future tax deductions for depreciation that will be available as a result of such payments or property transfers (the "Present Value Depreciation Amount"), plus (2) an additional amount sufficient to permit Transmission Provider to receive and retain, after the payment of all Current Taxes, an amount equal to the net amount described in clause (1).

For this purpose, (i) Current Taxes shall be computed based on Transmission Provider's composite federal and state tax rates at the time the payments or property transfers are received and Transmission Provider will be treated as being subject to tax at the highest marginal rates in effect at that time (the "Current Tax Rate"), and (ii) the Present Value Depreciation Amount shall be computed by discounting Transmission Provider's anticipated tax depreciation deductions as a result of such payments or property transfers by Transmission Provider's current weighted average cost of capital. Thus, the formula for calculating Interconnection Customer's liability to Transmission Owner pursuant to this Article 5.17.4 can be expressed as follows: (Current Tax Rate x (Gross Income Amount - Present Value of Tax Depreciation))/(1Current Tax Rate). Interconnection Customer's estimated tax liability in the event taxes are imposed shall be stated in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades.

Interconnection Customer's request and expense, Transmission Provider shall file with the IRS a request for a private letter ruling as to whether any property transferred or sums paid, or to be paid, by Interconnection Customer to Transmission Provider under this LGIA are subject to federal income taxation. Interconnection Customer will prepare the initial draft of the request for a private letter ruling, and will certify under penalties of perjury that all facts represented in such request are true and accurate to the best of Interconnection

Customer shall cooperate in good faith with respect to the

submission of such request.

Private Letter Ruling or Change or Clarification of Law. At

Transmission Provider shall keep Interconnection Customer fully informed of the status of such request for a private letter ruling and

Customer's knowledge. Transmission Provider and Interconnection

shall execute either a privacy act waiver or a limited power of attorney, in a form acceptable to the IRS, that authorizes Interconnection Customer to participate in all discussions with the IRS regarding such request for a private letter ruling. Transmission Provider shall allow Interconnection Customer to attend all meetings with IRS officials about the request and shall permit Interconnection Customer to prepare the initial drafts of any follow-up letters in connection with the request.

- Subsequent Taxable Events. If, within 10 years from the date on which the relevant Transmission Provider's Interconnection Facilities are placed in service, (i) Interconnection Customer Breaches the covenants contained in Article 5.17.2, (ii) a "disqualification event" occurs within the meaning of IRS Notice 88-129, or (iii) this LGIA terminates and Transmission Provider retains ownership of the Interconnection Facilities and Network Upgrades, Interconnection Customer shall pay a tax gross-up for the cost consequences of any current tax liability imposed on Transmission Provider, calculated using the methodology described in Article 5.17.4 and in accordance with IRS Notice 90-60.
- 5.17.7 **Contests**. In the event any Governmental Authority determines that Transmission Provider's receipt of payments or property constitutes income that is subject to taxation, Transmission Provider shall notify Interconnection Customer, in writing, within thirty (30) Calendar Days of receiving notification of such determination by a Governmental Authority. Upon the timely written request by Interconnection Customer and at Interconnection Customer's sole expense, Transmission Provider may appeal, protest, seek abatement of, or otherwise oppose such determination. Upon Interconnection Customer's written request and sole expense, Transmission Provider may file a claim for refund with respect to any taxes paid under this Article 5.17, whether or not it has received such a determination. Transmission Provider reserves the right to make all decisions with regard to the prosecution of such appeal, protest, abatement or other contest, including the selection of counsel and compromise or settlement of the claim, but Transmission Provider shall keep Interconnection Customer informed, shall consider in good faith suggestions from Interconnection Customer about the conduct of the contest, and shall reasonably permit Interconnection Customer or an Interconnection Customer representative to attend contest proceedings.

Interconnection Customer shall pay to Transmission Provider on a periodic basis, as invoiced by Transmission Provider, Transmission Provider's documented reasonable costs of prosecuting such appeal, protest, abatement or other contest. At any time during the contest, Transmission Provider may agree to a settlement either with Interconnection Customer's consent or after obtaining written advice from nationally-recognized tax counsel, selected by Transmission Provider, but reasonably acceptable to Interconnection Customer, that the proposed settlement represents a reasonable settlement given the hazards of litigation.

Interconnection Customer's obligation shall be based on the amount of the settlement agreed to by Interconnection Customer, or if a higher amount, so much of the settlement that is supported by the written advice from nationally-recognized tax counsel selected under the terms of the preceding sentence. The settlement amount shall be calculated on a fully grossed-up bases to cover any related cost consequences of the current tax liability. Any settlement without Interconnection Customer's consent or such written advice will relieve Interconnection Customer from any obligation to indemnify Transmission Provider for the tax at issue in the contest.

5.17.8 **Refund.** In the event that (a) a private letter ruling is issued to Transmission Provider which holds that any amount paid or the value of any property transferred by Interconnection Customer to Transmission Provider under the terms of this LGIA is not subject to federal income taxation, (b) any legislative change or administrative announcement, notice, ruling or other determination makes it reasonably clear to Transmission Provider in good faith that any amount paid or the value of any property transferred by Interconnection Customer to Transmission Provider under the terms of this LGIA is not taxable to Transmission Provider, (c) any abatement, appeal, protest, or other contest results in a determination that any payments or transfers made by Interconnection Customer to Transmission Provider are not subject to federal income tax, or (d) if Transmission Provider receives a refund from any taxing authority for any overpayment of tax attributable to any payment or property transfer made by Interconnection Customer to Transmission Provider pursuant to this LGIA, Transmission Provider shall promptly refund to Interconnection Customer the following:

(i) any payment made by Interconnection Customer under this Article 5.17 for taxes that is attributable to the amount

determined to be non-taxable, together with interest thereon,

- (ii) interest on any amounts paid by Interconnection Customer to Transmission Provider for such taxes which Transmission Provider did not submit to the taxing authority, calculated in accordance with the methodology set forth in FERC's regulations at 18 CFR §35.19a(a)(2)(iii) from the date payment was made by Interconnection Customer to the date Transmission Provider refunds such payment to Interconnection Customer, and
- (iii) with respect to any such taxes paid by Transmission Provider, any refund or credit Transmission Provider receives or to which it may be entitled from any Governmental Authority, interest (or that portion thereof attributable to the payment described in clause (i), above) owed to Transmission Provider for such overpayment of taxes (including any reduction in interest otherwise payable by Transmission Provider to any Governmental Authority resulting from an offset or credit); provided, however, that Transmission Provider will remit such amount promptly to Interconnection Customer only after and to the extent that Transmission Provider has received a tax refund, credit or offset from any Governmental Authority for any applicable overpayment of income tax related to Transmission Provider's Interconnection Facilities.

The intent of this provision is to leave the Parties, to the extent practicable, in the event that no taxes are due with respect to any payment for Interconnection Facilities and Network Upgrades hereunder, in the same position they would have been in had no such tax payments been made.

Interconnection Customer, and at Interconnection Customer's sole expense, Transmission Provider may appeal, protest, seek abatement of, or otherwise contest any tax (other than federal or state income tax) asserted or assessed against Transmission Provider for which Interconnection Customer may be required to reimburse Transmission Provider under the terms of this LGIA. Interconnection Customer shall pay to Transmission Provider on a periodic basis, as invoiced by Transmission Provider, Transmission Provider's documented reasonable costs of prosecuting such appeal,

protest, abatement, or other contest. Interconnection Customer and Transmission Provider shall cooperate in good faith with respect to any such contest. Unless the payment of such taxes is a prerequisite to an appeal or abatement or cannot be deferred, no amount shall be payable by Interconnection Customer to Transmission Provider for such taxes until they are assessed by a final, non-appealable order by any court or agency of competent jurisdiction. In the event that a tax payment is withheld and ultimately due and payable after appeal, Interconnection Customer will be responsible for all taxes, interest and penalties, other than penalties attributable to any delay caused by Transmission Provider.

- Transmission Owners Who Are Not Transmission Providers. If Transmission Provider is not the same entity as the Transmission Owner, then (i) all references in this Article 5.17 to Transmission Provider shall be deemed also to refer to and to include the Transmission Owner, as appropriate, and (ii) this LGIA shall not become effective until such Transmission Owner shall have agreed in writing to assume all of the duties and obligations of Transmission Provider under this Article 5.17 of this LGIA.
- **5.18 Tax Status**. Each Party shall cooperate with the other to maintain the other Party's tax status. Nothing in this LGIA is intended to adversely affect any Transmission Provider's tax exempt status with respect to the issuance of bonds including, but not limited to, Local Furnishing Bonds.

5.19 Modification.

5.19.1 **General**. Either Party may undertake modifications to its facilities. If a Party plans to undertake a modification that reasonably may be expected to affect the other Party's facilities, that Party shall provide to the other Party sufficient information regarding such modification so that the other Party may evaluate the potential impact of such modification prior to commencement of the work. Such information shall be deemed to be confidential hereunder and shall include information concerning the timing of such modifications and whether such modifications are expected to interrupt the flow of electricity from the Large Generating Facility. The Party desiring to perform such work shall provide the relevant drawings, plans, and specifications to the other Party at least ninety (90) Calendar Days in advance of the commencement of the work or such shorter period upon which the Parties may agree, which agreement shall not unreasonably be withheld, conditioned or delayed.

In the case of Large Generating Facility modifications that do not require Interconnection Customer to submit an Interconnection Request, Transmission Provider shall provide, within thirty (30) Calendar Days (or such other time as the Parties may agree), an estimate of any additional modifications to the Transmission System, Transmission Provider's Interconnection Facilities or Network Upgrades necessitated by such Interconnection Customer modification and a good faith estimate of the costs thereof.

- **Standards**. Any additions, modifications, or replacements made to a Party's facilities shall be designed, constructed and operated in accordance with this LGIA and Good Utility Practice.
- 5.19.3 Modification Costs. Interconnection Customer shall not be directly assigned for the costs of any additions, modifications, or replacements that Transmission Provider makes to Transmission Provider's Interconnection Facilities or the Transmission System to facilitate the interconnection of a third party to Transmission Provider's Interconnection Facilities or the Transmission System, or to provide transmission service to a third party under Transmission Provider's Tariff. Interconnection Customer shall be responsible for the costs of any additions, modifications, or replacements to Interconnection Customer's Interconnection Facilities that may be necessary to maintain or upgrade such Interconnection Customer's Interconnection Facilities consistent with Applicable Laws and Regulations, Applicable Reliability Standards or Good Utility Practice.

Article 6. Testing and Inspection

- 6.1 Pre-Commercial Operation Date Testing and Modifications. Prior to the Commercial Operation Date, Transmission Provider shall test Transmission Provider's Interconnection Facilities and Network Upgrades and Interconnection Customer shall test the Large Generating Facility and Interconnection Customer's Interconnection Facilities to ensure their safe and reliable operation. Similar testing may be required after initial operation. Each Party shall make any modifications to its facilities that are found to be necessary as a result of such testing. Interconnection Customer shall bear the cost of all such testing and modifications. Interconnection Customer shall generate test energy at the Large Generating Facility only if it has arranged for the delivery of such test energy.
- **6.2 Post-Commercial Operation Date Testing and Modifications**. Each Party shall

at its own expense perform routine inspection and testing of its facilities and equipment in accordance with Good Utility Practice as may be necessary to ensure the continued interconnection of the Large Generating Facility with the Transmission System in a safe and reliable manner. Each Party shall have the right, upon advance written notice, to require reasonable additional testing of the other Party's facilities, at the requesting Party's expense, as may be in accordance with Good Utility Practice.

- **6.3 Right to Observe Testing**. Each Party shall notify the other Party in advance of its performance of tests of its Interconnection Facilities. The other Party has the right, at its own expense, to observe such testing.
- 6.4 **Right to Inspect**. Each Party shall have the right, but shall have no obligation to: (i) observe the other Party's tests and/or inspection of any of its System Protection Facilities and other protective equipment, including Power System Stabilizers; (ii) review the settings of the other Party's System Protection Facilities and other protective equipment; and (iii) review the other Party's maintenance records relative to the Interconnection Facilities, the System Protection Facilities and other protective equipment. A Party may exercise these rights from time to time as it deems necessary upon reasonable notice to the other Party. The exercise or non-exercise by a Party of any such rights shall not be construed as an endorsement or confirmation of any element or condition of the Interconnection Facilities or the System Protection Facilities or other protective equipment or the operation thereof, or as a warranty as to the fitness, safety, desirability, or reliability of same. Any information that a Party obtains through the exercise of any of its rights under this Article 6.4 shall be deemed to be Confidential Information and treated pursuant to Article 22 of this LGIA.

Article 7. Metering

- 7.1 General. Each Party shall comply with the Applicable Reliability Council requirements. Unless otherwise agreed by the Parties, Transmission Provider shall install Metering Equipment at the Point of Interconnection prior to any operation of the Large Generating Facility and shall own, operate, test and maintain such Metering Equipment. Power flows to and from the Large Generating Facility shall be measured at or, at Transmission Provider's option, compensated to, the Point of Interconnection. Transmission Provider shall provide metering quantities, in analog and/or digital form, to Interconnection Customer upon request. Interconnection Customer shall bear all reasonable documented costs associated with the purchase, installation, operation, testing and maintenance of the Metering Equipment.
- 7.2 Check Meters. Interconnection Customer, at its option and expense, may install

and operate, on its premises and on its side of the Point of Interconnection, one or more check meters to check Transmission Provider's meters. Such check meters shall be for check purposes only and shall not be used for the measurement of power flows for purposes of this LGIA, except as provided in Article 7.4 below. The check meters shall be subject at all reasonable times to inspection and examination by Transmission Provider or its designee. The installation, operation and maintenance thereof shall be performed entirely by Interconnection Customer in accordance with Good Utility Practice.

- **7.3 Standards**. Transmission Provider shall install, calibrate, and test revenue quality Metering Equipment in accordance with applicable ANSI standards.
- 7.4 **Testing of Metering Equipment**. Transmission Provider shall inspect and test all Transmission Provider-owned Metering Equipment upon installation and at least once every two (2) years thereafter. If requested to do so by Interconnection Customer, Transmission Provider shall, at Interconnection Customer's expense, inspect or test Metering Equipment more frequently than every two (2) years. Transmission Provider shall give reasonable notice of the time when any inspection or test shall take place, and Interconnection Customer may have representatives present at the test or inspection. If at any time Metering Equipment is found to be inaccurate or defective, it shall be adjusted, repaired or replaced at Interconnection Customer's expense, in order to provide accurate metering, unless the inaccuracy or defect is due to Transmission Provider's failure to maintain, then Transmission Provider shall pay. If Metering Equipment fails to register, or if the measurement made by Metering Equipment during a test varies by more than two percent from the measurement made by the standard meter used in the test, Transmission Provider shall adjust the measurements by correcting all measurements for the period during which Metering Equipment was in error by using Interconnection Customer's check meters, if installed. If no such check meters are installed or if the period cannot be reasonably ascertained, the adjustment shall be for the period immediately preceding the test of the Metering Equipment equal to one-half the time from the date of the last previous test of the Metering Equipment.
- 7.5 Metering Data. At Interconnection Customer's expense, the metered data shall be telemetered to one or more locations designated by Transmission Provider and one or more locations designated by Interconnection Customer. Such telemetered data shall be used, under normal operating conditions, as the official measurement of the amount of energy delivered from the Large Generating Facility to the Point of Interconnection.

Article 8. Communications

- 8.1 **Interconnection Customer Obligations**. Interconnection Customer shall maintain satisfactory operating communications with Transmission Provider's Transmission System dispatcher or representative designated by Transmission Provider. Interconnection Customer shall provide standard voice line, dedicated voice line and facsimile communications at its Large Generating Facility control room or central dispatch facility through use of either the public telephone system, or a voice communications system that does not rely on the public telephone system. Interconnection Customer shall also provide the dedicated data circuit(s) necessary to provide Interconnection Customer data to Transmission Provider as set forth in Appendix D, Security Arrangements Details. The data circuit(s) shall extend from the Large Generating Facility to the location(s) specified by Transmission Provider. Any required maintenance of such communications equipment shall be performed by Interconnection Customer. Operational communications shall be activated and maintained under, but not be limited to, the following events: system paralleling or separation, scheduled and unscheduled shutdowns, equipment clearances, and hourly and daily load data.
- 8.2 Remote Terminal Unit. Prior to the Initial Synchronization Date of the Large Generating Facility, a Remote Terminal Unit, or equivalent data collection and transfer equipment acceptable to the Parties, shall be installed by Interconnection Customer, or by Transmission Provider at Interconnection Customer's expense, to gather accumulated and instantaneous data to be telemetered to the location(s) designated by Transmission Provider through use of a dedicated point-to-point data circuit(s) as indicated in Article 8.1. The communication protocol for the data circuit(s) shall be specified by Transmission Provider. Instantaneous bidirectional analog real power and reactive power flow information must be telemetered directly to the location(s) specified by Transmission Provider.

Each Party will promptly advise the other Party if it detects or otherwise learns of any metering, telemetry or communications equipment errors or malfunctions that require the attention and/or correction by the other Party. The Party owning such equipment shall correct such error or malfunction as soon as reasonably feasible.

- **8.3 No Annexation**. Any and all equipment placed on the premises of a Party shall be and remain the property of the Party providing such equipment regardless of the mode and manner of annexation or attachment to real property, unless otherwise mutually agreed by the Parties.
- **8.4 Provision of Data from a Variable Energy Resource**. The Interconnection Customer whose Generating Facility is a Variable Energy Resource shall provide meteorological and forced outage data to the Transmission Provider to the extent necessary for the Transmission Provider's development and deployment of power production forecasts for that class of Variable Energy Resources. The

Interconnection Customer with a Variable Energy Resource having wind as the energy source, at a minimum, will be required to provide the Transmission Provider with site-specific meteorological data including: temperature, wind speed, wind direction, and atmospheric pressure. The Interconnection Customer with a Variable Energy Resource having solar as the energy source, at a minimum, will be required to provide the Transmission Provider with site-specific meteorological data including: temperature, atmospheric pressure, and irradiance. The Transmission Provider and Interconnection Customer whose Generating Facility is a Variable Energy Resource shall mutually agree to any additional meteorological data that are required for the development and deployment of a power production forecast. The Interconnection Customer whose Generating Facility is a Variable Energy Resource also shall submit data to the Transmission Provider regarding all forced outages to the extent necessary for the Transmission Provider's development and deployment of power production forecasts for that class of Variable Energy Resources. The exact specifications of the meteorological and forced outage data to be provided by the Interconnection Customer to the Transmission Provider, including the frequency and timing of data submittals, shall be made taking into account the size and configuration of the Variable Energy Resource, its characteristics, location, and its importance in maintaining generation resource adequacy and transmission system reliability in its area. All requirements for meteorological and forced outage data must be commensurate with the power production forecasting employed by the Transmission Provider. Such requirements for meteorological and forced outage data are set forth in Appendix C, Interconnection Details, of this LGIA, as they may change from time to time.

Article 9. Operations

- 9.1 General. Each Party shall comply with the Applicable Reliability Council requirements. Each Party shall provide to the other Party all information that may reasonably be required by the other Party to comply with Applicable Laws and Regulations and Applicable Reliability Standards.
- Ontrol Area Notification. At least three months before Initial Synchronization Date, Interconnection Customer shall notify Transmission Provider in writing of the Control Area in which the Large Generating Facility will be located. If Interconnection Customer elects to locate the Large Generating Facility in a Control Area other than the Control Area in which the Large Generating Facility is physically located, and if permitted to do so by the relevant transmission tariffs, all necessary arrangements, including but not limited to those set forth in Article 7 and Article 8 of this LGIA, and remote Control Area generator interchange agreements, if applicable, and the appropriate measures under such agreements, shall be executed and implemented prior to the placement of the Large Generating

Facility in the other Control Area.

- 9.3 Transmission Provider Obligations. Transmission Provider shall cause the Transmission System and Transmission Provider's Interconnection Facilities to be operated, maintained and controlled in a safe and reliable manner and in accordance with this LGIA. Transmission Provider may provide operating instructions to Interconnection Customer consistent with this LGIA and Transmission Provider's operating protocols and procedures as they may change from time to time. Transmission Provider will consider changes to its operating protocols and procedures proposed by Interconnection Customer.
- 9.4 Interconnection Customer Obligations. Interconnection Customer shall at its own expense operate, maintain and control the Large Generating Facility and Interconnection Customer's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA. Interconnection Customer shall operate the Large Generating Facility and Interconnection Customer's Interconnection Facilities in accordance with all applicable requirements of the Control Area of which it is part, as such requirements are set forth in Appendix C, Interconnection Details, of this LGIA. Appendix C, Interconnection Details, will be modified to reflect changes to the requirements as they may change from time to time. Either Party may request that the other Party provide copies of the requirements set forth in Appendix C, Interconnection Details, of this LGIA.
- 9.5 Start-Up and Synchronization. Consistent with the Parties' mutually acceptable procedures, Interconnection Customer is responsible for the proper synchronization of the Large Generating Facility to Transmission Provider's Transmission System.
- 9.6 Reactive Power and Primary Frequency Response.
 - 9.6.1 Power Factor Design Criteria.
 - 9.6.1.1 Synchronous Generation. Interconnection Customer shall design the Large Generating Facility to maintain a composite power delivery at continuous rated power output at the Point of Interconnection at a power factor within the range of 0.95 leading to 0.95 lagging, unless the Transmission Provider has established different requirements that apply to all generators in the Control Area on a comparable basis.
 - 9.6.1.2 Non-Synchronous Generation. Interconnection Customer shall design the Large Generating Facility to maintain a composite power delivery at continuous rated power output at the high-side of the generator substation at a power factor

within the range of 0.95 leading to 0.95 lagging, unless the Transmission Provider has established a different power factor range that applies to all non-synchronous generators in the Control Area on a comparable basis. This power factor range standard shall be dynamic and can be met using, for example, power electronics designed to supply this level of reactive capability (taking into account any limitations due to voltage level, real power output, etc.) or fixed and switched capacitors, or a combination of the two. This requirement shall only apply to newly interconnecting non-synchronous generators that have not yet executed a Facilities Study Agreement as of the effective date of the Final Rule establishing this requirement (Order No. 827).

- 9.6.2 Voltage Schedules. Once Interconnection Customer has synchronized the Large Generating Facility with the Transmission System, Transmission Provider shall require Interconnection Customer to operate the Large Generating Facility to produce or absorb reactive power within the design limitations of the Large Generating Facility set forth in Article 9.6.1 (Power Factor Design Criteria). Transmission Provider's voltage schedules shall treat all sources of reactive power in the Control Area in an equitable and not unduly discriminatory manner. Transmission Provider shall exercise Reasonable Efforts to provide Interconnection Customer with such schedules at least one (1) day in advance, and may make changes to such schedules as necessary to maintain the reliability of the Transmission System. Interconnection Customer shall operate the Large Generating Facility to maintain the specified output voltage or power factor at the Point of Interconnection within the design limitations of the Large Generating Facility set forth in Article 9.6.1 (Power Factor Design Criteria). If Interconnection Customer is unable to maintain the specified voltage or power factor, it shall promptly notify the System Operator.
 - 9.6.2.1 Voltage Regulators. Whenever the Large Generating Facility is operated in parallel with the Transmission System and voltage regulators are capable of operation, Interconnection Customer shall operate the Large Generating Facility with its voltage regulators in automatic operation. If the Large Generating Facility's voltage regulators are not capable of such automatic operation, Interconnection Customer shall immediately notify Transmission Provider's system operator, or its designated representative, and ensure that such Large Generating Facility's reactive power production or absorption

(measured in MVARs) are within the design capability of the Large Generating Facility's generating unit(s) and steady state stability limits. Interconnection Customer shall not cause its Large Generating Facility to disconnect automatically or instantaneously from the Transmission System or trip any generating unit comprising the Large Generating Facility for an under or over frequency condition unless the abnormal frequency condition persists for a time period beyond the limits set forth in ANSI/IEEE Standard C37.106, or such other standard as applied to other generators in the Control Area on a comparable basis.

- 9.6.3 Payment for Reactive Power. Transmission Provider is required to pay Interconnection Customer for reactive power that Interconnection Customer provides or absorbs from the Large Generating Facility when Transmission Provider requests Interconnection Customer to operate its Large Generating Facility outside the range specified in Article 9.6.1, provided that if Transmission Provider pays its own or affiliated generators for reactive power service within the specified range, it must also pay Interconnection Customer. Payments shall be pursuant to Article 11.6 or such other agreement to which the Parties have otherwise agreed.
- 9.6.4 Primary Frequency Response. Interconnection Customer shall ensure the primary frequency response capability of its Large Generating Facility by installing, maintaining, and operating a functioning governor or equivalent controls. The term "functioning governor or equivalent controls" as used herein shall mean the required hardware and/or software that provides frequency responsive real power control with the ability to sense changes in system frequency and autonomously adjust the Large Generating Facility's real power output in accordance with the droop and deadband parameters and in the direction needed to correct frequency deviations. Interconnection Customer is required to install a governor or equivalent controls with the capability of operating: (1) with a maximum 5 percent droop and ± 0.036 Hz deadband; or (2) in accordance with the relevant droop, deadband, and timely and sustained response settings from an approved NERC Reliability Standard providing for equivalent or more stringent parameters. The droop characteristic shall be: (1) based on the nameplate capacity of the Large Generating Facility, and shall be linear in the range of frequencies between 59 to 61 Hz that are outside of the deadband parameter; or (2) based on an approved NERC Reliability Standard providing for an equivalent or more stringent parameter. The deadband parameter shall be: the range of frequencies above and below nominal (60 Hz) in which the governor or equivalent controls is not expected to adjust the Large Generating Facility's

real power output in response to frequency deviations. The deadband shall be implemented: (1) without a step to the droop curve, that is, once the frequency deviation exceeds the deadband parameter, the expected change in the Large Generating Facility's real power output in response to frequency deviations shall start from zero and then increase (for under-frequency deviations) or decrease (for over-frequency deviations) linearly in proportion to the magnitude of the frequency deviation; or (2) in accordance with an approved NERC Reliability Standard providing for an equivalent or more stringent parameter. Interconnection Customer shall notify Transmission Provider that the primary frequency response capability of the Large Generating Facility has been tested and confirmed during commissioning. Once Interconnection Customer has synchronized the Large Generating Facility with the Transmission System, Interconnection Customer shall operate the Large Generating Facility consistent with the provisions specified in Sections 9.6.4.1 and 9.6.4.2 of this Agreement. The primary frequency response requirements contained herein shall apply to both synchronous and non-synchronous Large Generating Facilities.

9.6.4.1 Governor or Equivalent Controls. Whenever the Large Generating Facility is operated in parallel with the Transmission System, Interconnection Customer shall operate the Large Generating Facility with its governor or equivalent controls in service and responsive to frequency. Interconnection Customer shall: (1) in coordination with Transmission Provider and/or the relevant balancing authority, set the deadband parameter to: (1) a maximum of ± 0.036 Hz and set the droop parameter to a maximum of 5 percent; or (2) implement the relevant droop and deadband settings from an approved NERC Reliability Standard that provides for equivalent or more stringent parameters. Interconnection Customer shall be required to provide the status and settings of the governor or equivalent controls to Transmission Provider and/or the relevant balancing authority upon request. If Interconnection Customer needs to operate the Large Generating Facility with its governor or equivalent controls not in service, Interconnection Customer shall immediately notify Transmission Provider and the relevant balancing authority, and provide both with the following information: (1) the operating status of the governor or equivalent controls (i.e., whether it is currently out of service or when it will be taken out of service); (2) the reasons for removing the governor or equivalent controls from service;

and (3) a reasonable estimate of when the governor or equivalent controls will be returned to service. Interconnection Customer shall make Reasonable Efforts to return its governor or equivalent controls into service as soon as practicable. Interconnection Customer shall make Reasonable Efforts to keep outages of the Large Generating Facility's governor or equivalent controls to a minimum whenever the Large Generating Facility is operated in parallel with the Transmission System.

- 9.6.4.2 Timely and Sustained Response. Interconnection Customer shall ensure that the Large Generating Facility's real power response to sustained frequency deviations outside of the deadband setting is automatically provided and shall begin immediately after frequency deviates outside of the deadband, and to the extent the Large Generating Facility has operating capability in the direction needed to correct the frequency deviation. Interconnection Customer shall not block or otherwise inhibit the ability of the governor or equivalent controls to respond and shall ensure that the response is not inhibited, except under certain operational constraints including, but not limited to, ambient temperature limitations, physical energy limitations, outages of mechanical equipment, or regulatory requirements. The Large Generating Facility shall sustain the real power response at least until system frequency returns to a value within the deadband setting of the governor or equivalent controls. A Commission-approved Reliability Standard with equivalent or more stringent requirements shall supersede the above requirements.
- 9.6.4.3 Exemptions. Large Generating Facilities that are regulated by the United States Nuclear Regulatory Commission shall be exempt from Sections 9.6.4, 9.6.4.1, and 9.6.4.2 of this Agreement. Large Generating Facilities that are behind the meter generation that is sized-to-load (i.e., the thermal load and the generation are near-balanced in real-time operation and the generation is primarily controlled to maintain the unique thermal, chemical, or mechanical output necessary for the operating requirements of its host facility) shall be required to install primary frequency response capability in accordance with the droop and deadband capability requirements specified in Section 9.6.4, but shall be otherwise

9.6.4.4

exempt from the operating requirements in Sections 9.6.4, 9.6.4.1, 9.6.4.2, and 9.6.4.4 of this Agreement.

Electric Storage Resources. Interconnection Customer interconnecting an electric storage resource shall establish an operating range in Appendix C of its LGIA that specifies a minimum state of charge and a maximum state of charge between which the electric storage resource will be required to provide primary frequency response consistent with the conditions set forth in Sections 9.6.4, 9.6.4.1, 9.6.4.2, and 9.6.4.3 of this Agreement. Appendix C shall specify whether the operating range is static or dynamic, and shall consider (1) the expected magnitude of frequency deviations in the interconnection; (2) the expected duration that system frequency will remain outside of the deadband parameter in the interconnection; (3) the expected incidence of frequency deviations outside of the deadband parameter in the interconnection; (4) the physical capabilities of the electric storage resource; (5) operational limitations of the electric storage resource due to manufacturer specifications; and (6) any other relevant factors agreed to by Transmission Provider and Interconnection Customer, and in consultation with the relevant transmission owner or balancing authority as appropriate. If the operating range is dynamic, then Appendix C must establish how frequently the operating range will be reevaluated and the factors that may be considered during its

reevaluation.

Interconnection Customer's electric storage resource is required to provide timely and sustained primary frequency response consistent with Section 9.6.4.2 of this Agreement when it is online and dispatched to inject electricity to the Transmission System and/or receive electricity from the Transmission System. This excludes circumstances when the electric storage resource is not dispatched to inject electricity to the Transmission System and/or dispatched to receive electricity from the Transmission System. If Interconnection Customer's electric storage resource is charging at the time of a frequency deviation outside of its deadband parameter, it is to increase (for over-frequency deviations) or decrease (for under-frequency deviations) the rate at which it is charging in accordance with its droop parameter. Interconnection Customer's electric storage resource is not required to change

from charging to discharging, or vice versa, unless the response necessitated by the droop and deadband settings requires it to do so and it is technically capable of making such a transition.

9.7 Outages and Interruptions.

9.7.1 Outages.

- 9.7.1.1 Outage Authority and Coordination. Each Party may in accordance with Good Utility Practice in coordination with the other Party remove from service any of its respective Interconnection Facilities or Network Upgrades that may impact the other Party's facilities as necessary to perform maintenance or testing or to install or replace equipment. Absent an Emergency Condition, the Party scheduling a removal of such facility(ies) from service will use Reasonable Efforts to schedule such removal on a date and time mutually acceptable to the Parties. In all circumstances, any Party planning to remove such facility(ies) from service shall use Reasonable Efforts to minimize the effect on the other Party of such removal.
- 9.7.1.2 Outage Schedules. Transmission Provider shall post scheduled outages of its transmission facilities on the OASIS. Interconnection Customer shall submit its planned maintenance schedules for the Large Generating Facility to Transmission Provider for a minimum of a rolling twenty-four month period. Interconnection Customer shall update its planned maintenance schedules as necessary. Transmission Provider may request Interconnection Customer to reschedule its maintenance as necessary to maintain the reliability of the Transmission System; provided, however, adequacy of generation supply shall not be a criterion in determining Transmission System reliability. Transmission Provider shall compensate Interconnection Customer for any additional direct costs that Interconnection Customer incurs as a result of having to reschedule maintenance, including any additional overtime, breaking of maintenance contracts or other costs above and beyond the cost Interconnection Customer would have incurred absent Transmission Provider's request to reschedule maintenance. Interconnection Customer will not be eligible to receive

compensation, if during the twelve (12) months prior to the date of the scheduled maintenance, Interconnection Customer had modified its schedule of maintenance activities.

- 9.7.1.3 Outage Restoration. If an outage on a Party's Interconnection Facilities or Network Upgrades adversely affects the other Party's operations or facilities, the Party that owns or controls the facility that is out of service shall use Reasonable Efforts to promptly restore such facility(ies) to a normal operating condition consistent with the nature of the outage. The Party that owns or controls the facility that is out of service shall provide the other Party, to the extent such information is known, information on the nature of the Emergency Condition, an estimated time of restoration, and any corrective actions required. Initial verbal notice shall be followed up as soon as practicable with written notice explaining the nature of the outage.
- **9.7.2 Interruption of Service**. If required by Good Utility Practice to do so, Transmission Provider may require Interconnection Customer to interrupt or reduce deliveries of electricity if such delivery of electricity could adversely affect Transmission Provider's ability to perform such activities as are necessary to safely and reliably operate and maintain the Transmission System. The following provisions shall apply to any interruption or reduction permitted under this Article 9.7.2:
 - **9.7.2.1** The interruption or reduction shall continue only for so long as reasonably necessary under Good Utility Practice;
 - 9.7.2.2 Any such interruption or reduction shall be made on an equitable, non-discriminatory basis with respect to all generating facilities directly connected to the Transmission System;
 - 9.7.2.3 When the interruption or reduction must be made under circumstances which do not allow for advance notice, Transmission Provider shall notify Interconnection Customer by telephone as soon as practicable of the reasons for the curtailment, interruption, or reduction, and, if known, its expected duration. Telephone notification shall be followed by written notification as soon as practicable;
 - **9.7.2.4** Except during the existence of an Emergency Condition,

when the interruption or reduction can be scheduled without advance notice, Transmission Provider shall notify Interconnection Customer in advance regarding the timing of such scheduling and further notify Interconnection Customer of the expected duration. Transmission Provider shall coordinate with Interconnection Customer using Good Utility Practice to schedule the interruption or reduction during periods of least impact to Interconnection Customer and Transmission Provider;

- 9.7.2.5 The Parties shall cooperate and coordinate with each other to the extent necessary in order to restore the Large Generating Facility, Interconnection Facilities, and the Transmission System to their normal operating state, consistent with system conditions and Good Utility Practice.
- 9.7.3 Under-Frequency and Over Frequency Conditions. The Transmission System is designed to automatically activate a load-shed program as required by the Applicable Reliability Council in the event of an under-frequency system disturbance. Interconnection Customer shall implement under-frequency and over-frequency relay set points for the Large Generating Facility as required by the Applicable Reliability Council to ensure "ride through" capability of the Transmission System. Large Generating Facility response to frequency deviations of pre-determined magnitudes, both under-frequency and over-frequency deviations, shall be studied and coordinated with Transmission Provider in accordance with Good Utility Practice. The term "ride through" as used herein shall mean the ability of a Generating Facility to stay connected to and synchronized with the Transmission System during system disturbances within a range of under-frequency and over-frequency conditions, in accordance with Good Utility Practice.
- 9.7.4 System Protection and Other Control Requirements.
 - 9.7.4.1 System Protection Facilities. Interconnection Customer shall, at its expense, install, operate and maintain System Protection Facilities as a part of the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Transmission Provider shall install at Interconnection Customer's expense any System Protection Facilities that may be required on Transmission Provider's Interconnection Facilities or the Transmission System as a result of the interconnection of the Large Generating Facility and

Interconnection Customer's Interconnection Facilities.

- 9.7.4.2 Each Party's protection facilities shall be designed and coordinated with other systems in accordance with Good Utility Practice.
- **9.7.4.3** Each Party shall be responsible for protection of its facilities consistent with Good Utility Practice.
- 9.7.4.4 Each Party's protective relay design shall incorporate the necessary test switches to perform the tests required in Article 6. The required test switches will be placed such that they allow operation of lockout relays while preventing breaker failure schemes from operating and causing unnecessary breaker operations and/or the tripping of Interconnection Customer's units.
- **9.7.4.5** Each Party will test, operate and maintain System Protection Facilities in accordance with Good Utility Practice.
- Prior to the In-Service Date, and again prior to the Commercial Operation Date, each Party or its agent shall perform a complete calibration test and functional trip test of the System Protection Facilities. At intervals suggested by Good Utility Practice and following any apparent malfunction of the System Protection Facilities, each Party shall perform both calibration and functional trip tests of its System Protection Facilities. These tests do not require the tripping of any in-service generation unit. These tests do, however, require that all protective relays and lockout contacts be activated.
- 9.7.5 Requirements for Protection. In compliance with Good Utility Practice, Interconnection Customer shall provide, install, own, and maintain relays, circuit breakers and all other devices necessary to remove any fault contribution of the Large Generating Facility to any short circuit occurring on the Transmission System not otherwise isolated by Transmission Provider's equipment, such that the removal of the fault contribution shall be coordinated with the protective requirements of the Transmission System. Such protective equipment shall include, without limitation, a disconnecting device or switch with load-interrupting capability located between the Large Generating Facility and the Transmission System at a site selected upon mutual agreement (not to be unreasonably withheld,

conditioned or delayed) of the Parties. Interconnection Customer shall be responsible for protection of the Large Generating Facility and Interconnection Customer's other equipment from such conditions as negative sequence currents, over- or under-frequency, sudden load rejection, over- or under-voltage, and generator loss-of-field. Interconnection Customer shall be solely responsible to disconnect the Large Generating Facility and Interconnection Customer's other equipment if conditions on the Transmission System could adversely affect the Large Generating Facility.

- 9.7.6 Power Quality. Neither Party's facilities shall cause excessive voltage flicker nor introduce excessive distortion to the sinusoidal voltage or current waves as defined by ANSI Standard C84.1-1989, in accordance with IEEE Standard 519, or any applicable superseding electric industry standard. In the event of a conflict between ANSI Standard C84.1-1989, or any applicable superseding electric industry standard, ANSI Standard C84.1-1989, or the applicable superseding electric industry standard, shall control.
- 9.8 Switching and Tagging Rules. Each Party shall provide the other Party a copy of its switching and tagging rules that are applicable to the other Party's activities. Such switching and tagging rules shall be developed on a non-discriminatory basis. The Parties shall comply with applicable switching and tagging rules, as amended from time to time, in obtaining clearances for work or for switching operations on equipment.
- 9.9 Use of Interconnection Facilities by Third Parties.
 - **9.9.1 Purpose of Interconnection Facilities**. Except as may be required by Applicable Laws and Regulations, or as otherwise agreed to among the Parties, the Interconnection Facilities shall be constructed for the sole purpose of interconnecting the Large Generating Facility to the Transmission System and shall be used for no other purpose.
 - 9.9.2 Third Party Users. If required by Applicable Laws and Regulations or if the Parties mutually agree, such agreement not to be unreasonably withheld, to allow one or more third parties to use Transmission Provider's Interconnection Facilities, or any part thereof, Interconnection Customer will be entitled to compensation for the capital expenses it incurred in connection with the Interconnection Facilities based upon the pro rata use of the Interconnection Facilities by Transmission Provider, all third party users, and Interconnection Customer, in accordance with Applicable Laws and Regulations or upon some other mutually-agreed upon methodology.

In addition, cost responsibility for ongoing costs, including operation and maintenance costs associated with the Interconnection Facilities, will be allocated between Interconnection Customer and any third party users based upon the pro rata use of the Interconnection Facilities by Transmission Provider, all third party users, and Interconnection Customer, in accordance with Applicable Laws and Regulations or upon some other mutually agreed upon methodology. If the issue of such compensation or allocation cannot be resolved through such negotiations, it shall be submitted to FERC for resolution.

9.10 Disturbance Analysis Data Exchange. The Parties will cooperate with one another in the analysis of disturbances to either the Large Generating Facility or Transmission Provider's Transmission System by gathering and providing access to any information relating to any disturbance, including information from oscillography, protective relay targets, breaker operations and sequence of events records, and any disturbance information required by Good Utility Practice.

Article 10. Maintenance

- **10.1 Transmission Provider Obligations**. Transmission Provider shall maintain the Transmission System and Transmission Provider's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA.
- **10.2** Interconnection Customer Obligations. Interconnection Customer shall maintain the Large Generating Facility and Interconnection Customer's Interconnection Facilities in a safe and reliable manner and in accordance with this LGIA.
- **10.3** Coordination. The Parties shall confer regularly to coordinate the planning, scheduling and performance of preventive and corrective maintenance on the Large Generating Facility and the Interconnection Facilities.
- 10.4 Secondary Systems. Each Party shall cooperate with the other in the inspection, maintenance, and testing of control or power circuits that operate below 600 volts, AC or DC, including, but not limited to, any hardware, control or protective devices, cables, conductors, electric raceways, secondary equipment panels, transducers, batteries, chargers, and voltage and current transformers that directly affect the operation of a Party's facilities and equipment which may reasonably be expected to impact the other Party. Each Party shall provide advance notice to the other Party before undertaking any work on such circuits, especially on electrical circuits involving circuit breaker trip and close contacts, current transformers, or potential transformers.

10.5 Operating and Maintenance Expenses. Subject to the provisions herein addressing the use of facilities by others, and except for operations and maintenance expenses associated with modifications made for providing interconnection or transmission service to a third party and such third party pays for such expenses, Interconnection Customer shall be responsible for all reasonable expenses including overheads, associated with: (1) owning, operating, maintaining, repairing, and replacing Interconnection Customer's Interconnection Facilities; and (2) operation, maintenance, repair and replacement of Transmission Provider's Interconnection Facilities.

Article 11. Performance Obligation

- 11.1 Interconnection Customer Interconnection Facilities. Interconnection Customer shall design, procure, construct, install, own and/or control Interconnection Customer Interconnection Facilities described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades, at its sole expense.
- 11.2 Transmission Provider's Interconnection Facilities. Transmission Provider or Transmission Owner shall design, procure, construct, install, own and/or control the Transmission Provider's Interconnection Facilities described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades, at the sole expense of the Interconnection Customer.
- 11.3 Network Upgrades and Distribution Upgrades. Transmission Provider or Transmission Owner shall design, procure, construct, install, and own the Network Upgrades and Distribution Upgrades described in Appendix A, Interconnection Facilities, Network Upgrades and Distribution Upgrades. The Interconnection Customer shall be responsible for all costs related to Distribution Upgrades. Unless Transmission Provider or Transmission Owner elects to fund the capital for the Network Upgrades, they shall be solely funded by Interconnection Customer.

11.4 Transmission Credits.

11.4.1 Repayment of Amounts Advanced for Network Upgrades.

Interconnection Customer shall be entitled to a cash repayment, equal to the total amount paid to Transmission Provider and Affected System Operator, if any, for the Network Upgrades, including any tax gross-up or other tax-related payments associated with Network Upgrades, and not refunded to Interconnection Customer pursuant to Article 5.17.8 or otherwise, to be paid to

Interconnection Customer on a dollar-for-dollar basis for the non-usage sensitive portion of transmission charges, as payments are made under Transmission Provider's Tariff and Affected System's Tariff for transmission services with respect to the Large Generating Facility. Any repayment shall include interest calculated in accordance with the methodology set forth in FERC's regulations at 18 C.F.R. § 35.19a(a)(2)(iii) from the date of any payment for Network Upgrades through the date on which the Interconnection Customer receives a repayment of such payment pursuant to this subparagraph. Interconnection Customer may assign such repayment rights to any person.

Notwithstanding the foregoing, Interconnection Customer, Transmission Provider, and Affected System Operator may adopt any alternative payment schedule that is mutually agreeable so long as Transmission Provider and Affected System Operator take one of the following actions no later than five years from the Commercial Operation Date: (1) return to Interconnection Customer any amounts advanced for Network Upgrades not previously repaid, or (2) declare in writing that Transmission Provider or Affected System Operator will continue to provide payments to Interconnection Customer on a dollar-for-dollar basis for the non-usage sensitive portion of transmission charges, or develop an alternative schedule that is mutually agreeable and provides for the return of all amounts advanced for Network Upgrades not previously repaid; however, full reimbursement shall not extend beyond twenty (20) years from the Commercial Operation Date.

If the Large Generating Facility fails to achieve Commercial Operation, but it or another Generating Facility is later constructed and makes use of the Network Upgrades, Transmission Provider and Affected System Operator shall at that time reimburse Interconnection Customer for the amounts advanced for the Network Upgrades. Before any such reimbursement can occur, the Interconnection Customer, or the entity that ultimately constructs the Generating Facility, if different, is responsible for identifying the entity to which reimbursement must be made.

11.4.2 Special Provisions for Affected Systems. Unless Transmission Provider provides, under the LGIA, for the repayment of amounts advanced to Affected System Operator for Network Upgrades, Interconnection Customer and Affected System Operator shall enter into an agreement that provides for such repayment. The agreement

shall specify the terms governing payments to be made by Interconnection Customer to the Affected System Operator as well as the repayment by the Affected System Operator.

- 11.4.3 Notwithstanding any other provision of this LGIA, nothing herein shall be construed as relinquishing or foreclosing any rights, including but not limited to firm transmission rights, capacity rights, transmission congestion rights, or transmission credits, that Interconnection Customer, shall be entitled to, now or in the future under any other agreement or tariff as a result of, or otherwise associated with, the transmission capacity, if any, created by the Network Upgrades, including the right to obtain cash reimbursements or transmission credits for transmission service that is not associated with the Large Generating Facility.
- 11.5 Provision of Security. At least thirty (30) Calendar Days prior to the commencement of the procurement, installation, or construction of a discrete portion of a Transmission Provider's Interconnection Facilities, Network Upgrades, or Distribution Upgrades, Interconnection Customer shall provide Transmission Provider, at Interconnection Customer's option, a guarantee, a surety bond, letter of credit or other form of security that is reasonably acceptable to Transmission Provider and is consistent with the Uniform Commercial Code of the jurisdiction identified in Article 14.2.1. Such security for payment shall be in an amount sufficient to cover the costs for constructing, procuring and installing the applicable portion of Transmission Provider's Interconnection Facilities, Network Upgrades, or Distribution Upgrades and shall be reduced on a dollar-for-dollar basis for payments made to Transmission Provider for these purposes.

In addition:

- The guarantee must be made by an entity that meets the creditworthiness requirements of Transmission Provider, and contain terms and conditions that guarantee payment of any amount that may be due from Interconnection Customer, up to an agreed-to maximum amount.
- 11.5.2 The letter of credit must be issued by a financial institution reasonably acceptable to Transmission Provider and must specify a reasonable expiration date.
- 11.5.3 The surety bond must be issued by an insurer reasonably acceptable to Transmission Provider and must specify a reasonable expiration date.

- 11.6 **Interconnection Customer Compensation**. If Transmission Provider requests or directs Interconnection Customer to provide a service pursuant to Articles 9.6.3 (Payment for Reactive Power), or 13.5.1 of this LGIA, Transmission Provider shall compensate Interconnection Customer in accordance with Interconnection Customer's applicable rate schedule then in effect unless the provision of such service(s) is subject to an RTO or ISO FERC-approved rate schedule. Interconnection Customer shall serve Transmission Provider or RTO or ISO with any filing of a proposed rate schedule at the time of such filing with FERC. To the extent that no rate schedule is in effect at the time the Interconnection Customer is required to provide or absorb any Reactive Power under this LGIA, Transmission Provider agrees to compensate Interconnection Customer in such amount as would have been due Interconnection Customer had the rate schedule been in effect at the time service commenced; provided, however, that such rate schedule must be filed at FERC or other appropriate Governmental Authority within sixty (60) Calendar Days of the commencement of service.
 - 11.6.1 Interconnection Customer Compensation for Actions During Emergency Condition. Transmission Provider or RTO or ISO shall compensate Interconnection Customer for its provision of real and reactive power and other Emergency Condition services that Interconnection Customer provides to support the Transmission System during an Emergency Condition in accordance with Article 11.6.

Article 12. Invoice

- 12.1 General. Each Party shall submit to the other Party, on a monthly basis, invoices of amounts due for the preceding month. Each invoice shall state the month to which the invoice applies and fully describe the services and equipment provided. The Parties may discharge mutual debts and payment obligations due and owing to each other on the same date through netting, in which case all amounts a Party owes to the other Party under this LGIA, including interest payments or credits, shall be netted so that only the net amount remaining due shall be paid by the owing Party.
- 12.2 Final Invoice. Within six months after completion of the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades, Transmission Provider shall provide an invoice of the final cost of the construction of Transmission Provider's Interconnection Facilities and the Network Upgrades and shall set forth such costs in sufficient detail to enable Interconnection Customer to compare the actual costs with the estimates and to ascertain deviations, if any, from the cost estimates. Transmission Provider shall refund to

Interconnection Customer any amount by which the actual payment by Interconnection Customer for estimated costs exceeds the actual costs of construction within thirty (30) Calendar Days of the issuance of such final construction invoice.

- 12.3 Payment. Invoices shall be rendered to the paying Party at the address specified in Appendix F. The Party receiving the invoice shall pay the invoice within thirty (30) Calendar Days of receipt. All payments shall be made in immediately available funds payable to the other Party, or by wire transfer to a bank named and account designated by the invoicing Party. Payment of invoices by either Party will not constitute a waiver of any rights or claims either Party may have under this LGIA. If Interconnection Customer has not paid the final invoice following a withdrawal within thirty (30) Calendar Days, Transmission Provider shall draw upon the security provided under this LGIA to settle all accounts, which shall include any offsets of amounts due and owing by Transmission Provider. After the final invoice is paid and all accounts are settled, Transmission Provider shall refund all remaining security.
- 12.4 Disputes. In the event of a billing dispute between Transmission Provider and Interconnection Customer, Transmission Provider shall continue to provide Interconnection Service under this LGIA as long as Interconnection Customer: (i) continues to make all payments not in dispute; and (ii) pays to Transmission Provider or into an independent escrow account the portion of the invoice in dispute, pending resolution of such dispute. If Interconnection Customer fails to meet these two requirements for continuation of service, then Transmission Provider may provide notice to Interconnection Customer of a Default pursuant to Article 17. Within thirty (30) Calendar Days after the resolution of the dispute, the Party that owes money to the other Party shall pay the amount due with interest calculated in accord with the methodology set forth in FERC's regulations at 18 CFR § 35.19a(a)(2)(iii).

Article 13. Emergencies

13.1 Definition. "Emergency Condition" shall mean a condition or situation: (i) that in the judgment of the Party making the claim is imminently likely to endanger life or property; or (ii) that, in the case of Transmission Provider, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to the Transmission System, Transmission Provider's Interconnection Facilities or the Transmission Systems of others to which the Transmission System is directly connected; or (iii) that, in the case of Interconnection Customer, is imminently likely (as determined in a non-discriminatory manner) to cause a material adverse effect on the security of, or damage to, the Large Generating Facility or Interconnection Customer's

Interconnection Facilities' System restoration and black start shall be considered Emergency Conditions; provided, that Interconnection Customer is not obligated by this LGIA to possess black start capability.

- **13.2 Obligations**. Each Party shall comply with the Emergency Condition procedures of the applicable ISO/RTO, NERC, the Applicable Reliability Council, Applicable Laws and Regulations, and any emergency procedures agreed to by the Joint Operating Committee.
- 13.3 **Notice.** Transmission Provider shall notify Interconnection Customer promptly when it becomes aware of an Emergency Condition that affects Transmission Provider's Interconnection Facilities or the Transmission System that may reasonably be expected to affect Interconnection Customer's operation of the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Interconnection Customer shall notify Transmission Provider promptly when it becomes aware of an Emergency Condition that affects the Large Generating Facility or Interconnection Customer's Interconnection Facilities that may reasonably be expected to affect the Transmission System or Transmission Provider's Interconnection Facilities. To the extent information is known, the notification shall describe the Emergency Condition, the extent of the damage or deficiency, the expected effect on the operation of Interconnection Customer's or Transmission Provider's facilities and operations, its anticipated duration and the corrective action taken and/or to be taken. The initial notice shall be followed as soon as practicable with written notice.
- 13.4 Immediate Action. Unless, in Interconnection Customer's reasonable judgment, immediate action is required, Interconnection Customer shall obtain the consent of Transmission Provider, such consent to not be unreasonably withheld, prior to performing any manual switching operations at the Large Generating Facility or Interconnection Customer's Interconnection Facilities in response to an Emergency Condition either declared by Transmission Provider or otherwise regarding the Transmission System.

13.5 Transmission Provider Authority.

13.5.1 General. Transmission Provider may take whatever actions or inactions with regard to the Transmission System or Transmission Provider's Interconnection Facilities it deems necessary during an Emergency Condition in order to (i) preserve public health and safety, (ii) preserve the reliability of the Transmission System or Transmission Provider's Interconnection Facilities, (iii) limit or prevent damage, and (iv) expedite restoration of service.

Transmission Provider shall use Reasonable Efforts to minimize the effect of such actions or inactions on the Large Generating Facility or Interconnection Customer's Interconnection Facilities. Transmission Provider may, on the basis of technical considerations, require the Large Generating Facility to mitigate an Emergency Condition by taking actions necessary and limited in scope to remedy the Emergency Condition, including, but not limited to, directing Interconnection Customer to shut-down, startup, increase or decrease the real or reactive power output of the Large Generating Facility; implementing a reduction or disconnection pursuant to Article 13.5.2; directing Interconnection Customer to assist with blackstart (if available) or restoration efforts; or altering the outage schedules of the Large Generating Facility and Interconnection Customer's Interconnection Facilities. Interconnection Customer shall comply with all of Transmission Provider's operating instructions concerning Large Generating Facility real power and reactive power output within the manufacturer's design limitations of the Large Generating Facility's equipment that is in service and physically available for operation at the time, in compliance with Applicable Laws and Regulations.

13.5.2 Reduction and Disconnection. Transmission Provider may reduce Interconnection Service or disconnect the Large Generating Facility or Interconnection Customer's Interconnection Facilities, when such, reduction or disconnection is necessary under Good Utility Practice due to Emergency Conditions. These rights are separate and distinct from any right of curtailment of Transmission Provider pursuant to Transmission Provider's Tariff. When Transmission Provider can schedule the reduction or disconnection in advance, Transmission Provider shall notify Interconnection Customer of the reasons, timing and expected duration of the reduction or disconnection. Transmission Provider shall coordinate with Interconnection Customer using Good Utility Practice to schedule the reduction or disconnection during periods of least impact to Interconnection Customer and Transmission Provider. Any reduction or disconnection shall continue only for so long as reasonably necessary under Good Utility Practice. The Parties shall cooperate with each other to restore the Large Generating Facility, the Interconnection Facilities, and the Transmission System to their normal operating state as soon as practicable consistent with Good Utility Practice.

13.6 Interconnection Customer Authority. Consistent with Good Utility Practice and

the LGIA and the LGIP, Interconnection Customer may take actions or inactions with regard to the Large Generating Facility or Interconnection Customer's Interconnection Facilities during an Emergency Condition in order to (i) preserve public health and safety, (ii) preserve the reliability of the Large Generating Facility or Interconnection Customer's Interconnection Facilities, (iii) limit or prevent damage, and (iv) expedite restoration of service. Interconnection Customer shall use Reasonable Efforts to minimize the effect of such actions or inactions on the Transmission System and Transmission Provider's Interconnection Facilities. Transmission Provider shall use Reasonable Efforts to assist Interconnection Customer in such actions.

13.7 Limited Liability. Except as otherwise provided in Article 11.6.1 of this LGIA, neither Party shall be liable to the other for any action it takes in responding to an Emergency Condition so long as such action is made in good faith and is consistent with Good Utility Practice.

Article 14. Regulatory Requirements and Governing Law

14.1 Regulatory Requirements. Each Party's obligations under this LGIA shall be subject to its receipt of any required approval or certificate from one or more Governmental Authorities in the form and substance satisfactory to the applying Party, or the Party making any required filings with, or providing notice to, such Governmental Authorities, and the expiration of any time period associated therewith. Each Party shall in good faith seek and use its Reasonable Efforts to obtain such other approvals. Nothing in this LGIA shall require Interconnection Customer to take any action that could result in its inability to obtain, or its loss of, status or exemption under the Federal Power Act, the Public Utility Holding Company Act of 1935, as amended, or the Public Utility Regulatory Policies Act of 1978.

14.2 Governing Law.

- 14.2.1 The validity, interpretation and performance of this LGIA and each of its provisions shall be governed by the laws of the state where the Point of Interconnection is located, without regard to its conflicts of law principles.
- 14.2.2 This LGIA is subject to all Applicable Laws and Regulations.
- 14.3.3 Each Party expressly reserves the right to seek changes in, appeal, or otherwise contest any laws, orders, rules, or regulations of a Governmental Authority.

Article 15. Notices.

15.1 General. Unless otherwise provided in this LGIA, any notice, demand or request required or permitted to be given by either Party to the other and any instrument required or permitted to be tendered or delivered by either Party in writing to the other shall be effective when delivered and may be so given, tendered or delivered, by recognized national courier, or by depositing the same with the United States Postal Service with postage prepaid, for delivery by certified or registered mail, addressed to the Party, or personally delivered to the Party, at the address set out in Appendix F, Addresses for Delivery of Notices and Billings.

Either Party may change the notice information in this LGIA by giving five (5) Business Days written notice prior to the effective date of the change.

- **15.2 Billings and Payments**. Billings and payments shall be sent to the addresses set out in Appendix F.
- **15.3 Alternative Forms of Notice**. Any notice or request required or permitted to be given by a Party to the other and not required by this Agreement to be given in writing may be so given by telephone, facsimile or email to the telephone numbers and email addresses set out in Appendix F.
- **15.4** Operations and Maintenance Notice. Each Party shall notify the other Party in writing of the identity of the person(s) that it designates as the point(s) of contact with respect to the implementation of Articles 9 and 10.

Article 16. Force Majeure

16.1 Force Majeure.

- **16.1.1** Economic hardship is not considered a Force Majeure event.
- 16.1.2 Neither Party shall be considered to be in Default with respect to any obligation hereunder, (including obligations under Article 4), other than the obligation to pay money when due, if prevented from fulfilling such obligation by Force Majeure. A Party unable to fulfill any obligation hereunder (other than an obligation to pay money when due) by reason of Force Majeure shall give notice and the full particulars of such Force Majeure to the other Party in writing or by telephone as soon as reasonably possible after the occurrence of the cause relied upon. Telephone notices given pursuant to this article shall be confirmed in writing as soon as reasonably possible

and shall specifically state full particulars of the Force Majeure, the time and date when the Force Majeure occurred and when the Force Majeure is reasonably expected to cease. The Party affected shall exercise due diligence to remove such disability with reasonable dispatch, but shall not be required to accede or agree to any provision not satisfactory to it in order to settle and terminate a strike or other labor disturbance.

Article 17. Default

17.1 Default

- Obligation (other than the payment of money) is the result of Force Majeure as defined in this LGIA or the result of an act of omission of the other Party. Upon a Breach, the non-breaching Party shall give written notice of such Breach to the breaching Party. Except as provided in Article 17.1.2, the breaching Party shall have thirty (30) Calendar Days from receipt of the Default notice within which to cure such Breach; provided however, if such Breach is not capable of cure within thirty (30) Calendar Days, the breaching Party shall commence such cure within thirty (30) Calendar Days after notice and continuously and diligently complete such cure within ninety (90) Calendar Days from receipt of the Default notice; and, if cured within such time, the Breach specified in such notice shall cease to exist.
- Right to Terminate. If a Breach is not cured as provided in this article, or if a Breach is not capable of being cured within the period provided for herein, the non-breaching Party shall have the right to declare a Default and terminate this LGIA by written notice at any time until cure occurs, and be relieved of any further obligation hereunder and, whether or not that Party terminates this LGIA, to recover from the breaching Party all amounts due hereunder, plus all other damages and remedies to which it is entitled at law or in equity. The provisions of this article will survive termination of this LGIA.

Article 18. Indemnity, Consequential Damages and Insurance

18.1 Indemnity. The Parties shall at all times indemnify, defend, and hold the other Party harmless from, any and all damages, losses, claims, including claims and actions relating to injury to or death of any person or damage to property, demand,

suits, recoveries, costs and expenses, court costs, attorney fees, and all other obligations by or to third parties, arising out of or resulting from the other Party's action or inactions of its obligations under this LGIA on behalf of the Indemnifying Party, except in cases of gross negligence or intentional wrongdoing by the indemnified Party.

- **18.1.1 Indemnified Person**. If an Indemnified Person is entitled to indemnification under this Article 18 as a result of a claim by a third party, and the Indemnifying Party fails, after notice and reasonable opportunity to proceed under Article 18.1, to assume the defense of such claim, such Indemnified Person may at the expense of the Indemnifying Party contest, settle or consent to the entry of any judgment with respect to, or pay in full, such claim.
- **18.1.2** Indemnifying Party. If an Indemnifying Party is obligated to indemnify and hold any Indemnified Person harmless under this Article 18, the amount owing to the Indemnified Person shall be the amount of such Indemnified Person's actual Loss, net of any insurance or other recovery.
- 18.1.3 Indemnity Procedures. Promptly after receipt by an Indemnified Person of any claim or notice of the commencement of any action or administrative or legal proceeding or investigation as to which the indemnity provided for in Article 18.1 may apply, the Indemnified Person shall notify the Indemnifying Party of such fact. Any failure of or delay in such notification shall not affect a Party's indemnification obligation unless such failure or delay is materially prejudicial to the Indemnifying Party.

The Indemnifying Party shall have the right to assume the defense thereof with counsel designated by such Indemnifying Party and reasonably satisfactory to the Indemnified Person. If the defendants in any such action include one or more Indemnified Persons and the Indemnifying Party and if the Indemnified Person reasonably concludes that there may be legal defenses available to it and/or other Indemnified Persons which are different from or additional to those available to the Indemnifying Party, the Indemnified Person shall have the right to select separate counsel to assert such legal defenses and to otherwise participate in the defense of such action on its own behalf. In such instances, the Indemnifying Party shall only be required to pay the fees and expenses of one additional attorney to represent an Indemnified Person or Indemnified Persons having such differing or additional legal defenses.

The Indemnified Person shall be entitled, at its expense, to participate in any such action, suit or proceeding, the defense of which has been assumed by the Indemnifying Party.

Notwithstanding the foregoing, the Indemnifying Party (i) shall not be entitled to assume and control the defense of any such action, suit or proceedings if and to the extent that, in the opinion of the Indemnified Person and its counsel, such action, suit or proceeding involves the potential imposition of criminal liability on the Indemnified Person, or there exists a conflict or adversity of interest between the Indemnified Person and the Indemnifying Party, in such event the Indemnified Person, and (ii) shall not settle or consent to the entry of any judgment in any action, suit or proceeding without the consent of the Indemnified Person, which shall not be reasonably withheld, conditioned or delayed.

- 18.2 Consequential Damages. Other than the Liquidated Damages heretofore described, in no event shall either Party be liable under any provision of this LGIA for any losses, damages, costs or expenses for any special, indirect, incidental, consequential, or punitive damages, including but not limited to loss of profit or revenue, loss of the use of equipment, cost of capital, cost of temporary equipment or services, whether based in whole or in part in contract, in tort, including negligence, strict liability, or any other theory of liability; provided, however, that damages for which a Party may be liable to the other Party under another agreement will not be considered to be special, indirect, incidental, or consequential damages hereunder.
- **18.3 Insurance**. Each party shall, at its own expense, maintain in force throughout the period of this LGIA, and until released by the other Party, the following minimum insurance coverages, with insurers authorized to do business in the state where the Point of Interconnection is located:
 - 18.3.1 Employers' Liability and Workers' Compensation Insurance providing statutory benefits in accordance with the laws and regulations of the state in which the Point of Interconnection is located.
 - 18.3.2 Commercial General Liability Insurance including premises and operations, personal injury, broad form property damage, broad form blanket contractual liability coverage (including coverage for the contractual indemnification) products and completed operations coverage, coverage for explosion, collapse and underground hazards,

independent contractors coverage, coverage for pollution to the extent normally available and punitive damages to the extent normally available and a cross liability endorsement, with minimum limits of One Million Dollars (\$1,000,000) per occurrence/One Million Dollars (\$1,000,000) aggregate combined single limit for personal injury, bodily injury, including death and property damage.

- 18.3.3 Comprehensive Automobile Liability Insurance for coverage of owned and non-owned and hired vehicles, trailers or semi-trailers designed for travel on public roads, with a minimum, combined single limit of One Million Dollars (\$1,000,000) per occurrence for bodily injury, including death, and property damage.
- 18.3.4 Excess Public Liability Insurance over and above the Employers' Liability Commercial General Liability and Comprehensive Automobile Liability Insurance coverage, with a minimum combined single limit of Twenty Million Dollars (\$20,000,000) per occurrence/Twenty Million Dollars (\$20,000,000) aggregate.
- 18.3.5 The Commercial General Liability Insurance, Comprehensive Automobile Insurance and Excess Public Liability Insurance policies shall name the other Party, its parent, associated and Affiliate companies and their respective directors, officers, agents, servants and employees ("Other Party Group") as additional insured. All policies shall contain provisions whereby the insurers waive all rights of subrogation in accordance with the provisions of this LGIA against the Other Party Group and provide thirty (30) Calendar Days advance written notice to the Other Party Group prior to anniversary date of cancellation or any material change in coverage or condition.
- 18.3.6 The Commercial General Liability Insurance, Comprehensive Automobile Liability Insurance and Excess Public Liability Insurance policies shall contain provisions that specify that the policies are primary and shall apply to such extent without consideration for other policies separately carried and shall state that each insured is provided coverage as though a separate policy had been issued to each, except the insurer's liability shall not be increased beyond the amount for which the insurer would have been liable had only one insured been covered. Each Party shall be responsible for its respective deductibles or retentions.
- 18.3.7 The Commercial General Liability Insurance, Comprehensive Automobile Liability Insurance and Excess Public Liability

Insurance policies, if written on a Claims First Made Basis, shall be maintained in full force and effect for two (2) years after termination of this LGIA, which coverage may be in the form of tail coverage or extended reporting period coverage if agreed by the Parties.

- 18.3.8 The requirements contained herein as to the types and limits of all insurance to be maintained by the Parties are not intended to and shall not in any manner, limit or qualify the liabilities and obligations assumed by the Parties under this LGIA.
- 18.3.9 Within ten (10) Calendar Days following execution of this LGIA, and as soon as practicable after the end of each fiscal year or at the renewal of the insurance policy and in any event within ninety (90) Calendar Days thereafter, each Party shall provide certification of all insurance required in this LGIA, executed by each insurer or by an authorized representative of each insurer.
- In addition to the foregoing, each Party may self-insure to meet the minimum insurance requirements of Articles 18.3.2 through 18.3.8 to the extent it maintains a self-insurance program; provided that, such Party's self-insurance program meets the minimum insurance requirements of Articles 18.3.2 through 18.3.8. In the event that a Party is permitted to self-insure pursuant to this article, it shall certify to the other Party with a letter of self-insurance that it meets the requirements to self-insure and that its self-insurance program meets the minimum insurance requirements in a manner consistent with that specified in Article 18.3.9.
- 18.3.11 The Parties agree to report to each other in writing as soon as practical all accidents or occurrences resulting in injuries to any person, including death, and any property damage arising out of this LGIA.

Article 19. Assignment

19.1 Assignment. This LGIA may be assigned by either Party only with the written consent of the other; provided that either Party may assign this LGIA without the consent of the other Party to any Affiliate of the assigning Party with an equal or greater credit rating and with the legal authority and operational ability to satisfy the obligations of the assigning Party under this LGIA; and provided further that Interconnection Customer shall have the right to assign this LGIA, without the consent of Transmission Provider, for collateral security purposes to aid in providing financing for the Large Generating Facility, provided that

Interconnection Customer will promptly notify Transmission Provider of any such assignment. Any financing arrangement entered into by Interconnection Customer pursuant to this article will provide that prior to or upon the exercise of the secured party's, trustee's or mortgagee's assignment rights pursuant to said arrangement, the secured creditor, the trustee or mortgagee will notify Transmission Provider of the date and particulars of any such exercise of assignment right(s), including providing the Transmission Provider with proof that it meets the requirements of Articles 11.5 and 18.3. Any attempted assignment that violates this article is void and ineffective. Any assignment under this LGIA shall not relieve a Party of its obligations, nor shall a Party's obligations be enlarged, in whole or in part, by reason thereof. Where required, consent to assignment will not be unreasonably withheld, conditioned or delayed.

Article 20. Severability

20.1 Severability. If any provision in this LGIA is finally determined to be invalid, void or unenforceable by any court or other Governmental Authority having jurisdiction, such determination shall not invalidate, void or make unenforceable any other provision, agreement or covenant of this LGIA; provided that if Interconnection Customer (or any third party, but only if such third party is not acting at the direction of Transmission Provider) seeks and obtains such a final determination with respect to any provision of the Alternate Option (Article 5.1.2), or the Negotiated Option (Article 5.1.4), then none of these provisions shall thereafter have any force or effect and the Parties' rights and obligations shall be governed solely by the Standard Option (Article 5.1.1).

Article 21. Comparability

21.1 Comparability. The Parties will comply with all applicable comparability and code of conduct laws, rules and regulations, as amended from time to time.

Article 22. Confidentiality

22.1 Confidentiality. Confidential Information shall include, without limitation, all information relating to a Party's technology, research and development, business affairs, and pricing, and any information supplied by either of the Parties to the other prior to the execution of this LGIA.

Information is Confidential Information only if it is clearly designated or marked in writing as confidential on the face of the document, or, if the information is conveyed orally or by inspection, if the Party providing the information orally informs the Party receiving the information that the information is confidential.

If requested by either Party, the other Party shall provide in writing, the basis for asserting that the information referred to in this Article 22 warrants confidential treatment, and the requesting Party may disclose such writing to the appropriate Governmental Authority. Each Party shall be responsible for the costs associated with affording confidential treatment to its information.

The release of Confidential Information shall be subject to Applicable Laws and Regulations and Applicable Reliability Standards.

- **Term**. During the term of this LGIA, and for a period of three (3) years after the expiration or termination of this LGIA, except as otherwise provided in this Article 22, each Party shall hold in confidence and shall not disclose to any person Confidential Information.
- 22.1.2 **Scope**. Confidential Information shall not include information that the receiving Party can demonstrate: (1) is generally available to the public other than as a result of a disclosure by the receiving Party; (2) was in the lawful possession of the receiving Party on a non-confidential basis before receiving it from the disclosing Party; (3) was supplied to the receiving Party without restriction by a third party, who, to the knowledge of the receiving Party after due inquiry, was under no obligation to the disclosing Party to keep such information confidential; (4) was independently developed by the receiving Party without reference to Confidential Information of the disclosing Party; (5) is, or becomes, publicly known, through no wrongful act or omission of the receiving Party or Breach of this LGIA; or (6) is required, in accordance with Article 22.1.7 of the LGIA, Order of Disclosure, to be disclosed by any Governmental Authority or is otherwise required to be disclosed by law or subpoena, or is necessary in any legal proceeding establishing rights and obligations under this LGIA. Information designated as Confidential Information will no longer be deemed confidential if the Party that designated the information as confidential notifies the other Party that it no longer is confidential.
- **Release of Confidential Information**. Neither Party shall release or disclose Confidential Information to any other person, except to its Affiliates (limited by the Standards of Conduct requirements), subcontractors, employees, consultants, or to parties who may be or considering

providing financing to or equity participation with Interconnection Customer, or to potential purchasers or assignees of Interconnection Customer, on a need-to-know basis in connection with this LGIA, unless such person has first been advised of the confidentiality provisions of this Article 22 and has agreed to comply with such provisions. Notwithstanding the foregoing, a Party providing Confidential Information to any person shall remain primarily responsible for any release of Confidential Information in contravention of this Article 22.

- **Rights**. Each Party retains all rights, title, and interest in the Confidential Information that each Party discloses to the other Party. The disclosure by each Party to the other Party of Confidential Information shall not be deemed a waiver by either Party or any other person or entity of the right to protect the Confidential Information from public disclosure.
- 22.1.5 No Warranties. By providing Confidential Information, neither Party makes any warranties or representations as to its accuracy or completeness. In addition, by supplying Confidential Information, neither Party obligates itself to provide any particular information or Confidential Information to the other Party nor to enter into any further agreements or proceed with any other relationship or joint venture.
- 22.1.6 Standard of Care. Each Party shall use at least the same standard of care to protect Confidential Information it receives as it uses to protect its own Confidential Information from unauthorized disclosure, publication or dissemination. Each Party may use Confidential Information solely to fulfill its obligations to the other Party under this LGIA or its regulatory requirements.
- 22.1.7 Order of Disclosure. If a court or a Government Authority or entity with the right, power, and apparent authority to do so requests or requires either Party, by subpoena, oral deposition, interrogatories, requests for production of documents, administrative order, or otherwise, to disclose Confidential Information, that Party shall provide the other Party with prompt notice of such request(s) or requirement(s) so that the other Party may seek an appropriate protective

order or waive compliance with the terms of this LGIA. Notwithstanding the absence of a protective order or waiver, the Party may disclose such Confidential Information which, in the opinion of its counsel, the Party is legally compelled to disclose. Each Party will use Reasonable Efforts to obtain reliable assurance that confidential treatment will be accorded any Confidential Information so furnished.

- 22.1.8 Termination of Agreement. Upon termination of this LGIA for any reason, each Party shall, within ten (10) Calendar Days of receipt of a written request from the other Party, use Reasonable Efforts to destroy, erase, or delete (with such destruction, erasure, and deletion certified in writing to the other Party) or return to the other Party, without retaining copies thereof, any and all written or electronic Confidential Information received from the other Party.
- 22.1.9 **Remedies**. The Parties agree that monetary damages would be inadequate to compensate a Party for the other Party's Breach of its obligations under this Article 22. Each Party accordingly agrees that the other Party shall be entitled to equitable relief, by way of injunction or otherwise, if the first Party Breaches or threatens to Breach its obligations under this Article 22, which equitable relief shall be granted without bond or proof of damages, and the receiving Party shall not plead in defense that there would be an adequate remedy at law. Such remedy shall not be deemed an exclusive remedy for the Breach of this Article 22, but shall be in addition to all other remedies available at law or in equity. The Parties further acknowledge and agree that the covenants contained herein are necessary for the protection of legitimate business interests and are reasonable in scope. No Party, however, shall be liable for indirect, incidental, or consequential or punitive damages of any nature or kind resulting from or arising in connection with this Article 22.
- **22.1.10 Disclosure to FERC, its Staff, or a State**. Notwithstanding anything in this Article 22 to the contrary, and pursuant to 18 CFR section 1b.20, if FERC or its staff, during the course of an investigation or otherwise, requests information from one of the Parties that is otherwise required to be maintained in confidence pursuant to this LGIA, the Party shall provide the requested information to FERC or its staff, within the time

provided for in the request for information. In providing the information to FERC or its staff, the Party must, consistent with 18 CFR section 388.112, request that the information be treated as confidential and non-public by FERC and its staff and that the information be withheld from public disclosure. Parties are prohibited from notifying the other Party to this LGIA prior to the release of the Confidential Information to FERC or its staff. The Party shall notify the other Party to the LGIA when it is notified by FERC or its staff that a request to release Confidential Information has been received by FERC, at which time either of the Parties may respond before such information would be made public, pursuant to 18 CFR section 388.112. Requests from a state regulatory body conducting a confidential investigation shall be treated in a similar manner if consistent with the applicable state rules and regulations.

22.1.11

Subject to the exception in Article 22.1.10, any information that a Party claims is competitively sensitive, commercial or financial information under this LGIA ("Confidential Information") shall not be disclosed by the other Party to any person not employed or retained by the other Party, except to the extent disclosure is (i) required by law; (ii) reasonably deemed by the disclosing Party to be required to be disclosed in connection with a dispute between or among the Parties, or the defense of litigation or dispute; (iii) otherwise permitted by consent of the other Party, such consent not to be unreasonably withheld; or (iv) necessary to fulfill its obligations under this LGIA or as a transmission service provider or a Control Area operator including disclosing the Confidential Information to an RTO or ISO or to a regional or national reliability organization. The Party asserting confidentiality shall notify the other Party in writing of the information it claims is confidential. Prior to any disclosures of the other Party's Confidential Information under this subparagraph, or if any third party or Governmental Authority makes any request or demand for any of the information described in this subparagraph, the disclosing Party agrees to promptly notify the other Party in writing and agrees to assert confidentiality and cooperate with the other Party in seeking to protect the Confidential Information from public disclosure by confidentiality agreement, protective order or other reasonable measures.

Article 23. Environmental Releases

23.1 Each Party shall notify the other Party, first orally and then in writing, of the release of any Hazardous Substances, any asbestos or lead abatement activities, or any type of remediation activities related to the Large Generating Facility or the Interconnection Facilities, each of which may reasonably be expected to affect the other Party. The notifying Party shall: (i) provide the notice as soon as practicable, provided such Party makes a good faith effort to provide the notice no later than twenty-four hours after such Party becomes aware of the occurrence; and (ii) promptly furnish to the other Party copies of any publicly available reports filed with any Governmental Authorities addressing such events.

Article 24. Information Requirements

- **24.1 Information Acquisition**. Transmission Provider and Interconnection Customer shall submit specific information regarding the electrical characteristics of their respective facilities to each other as described below and in accordance with Applicable Reliability Standards.
- 24.2 Information Submission by Transmission Provider. The initial information submission by Transmission Provider shall occur no later than one hundred eighty (180) Calendar Days prior to Trial Operation and shall include Transmission System information necessary to allow Interconnection Customer to select equipment and meet any system protection and stability requirements, unless otherwise agreed to by the Parties. On a monthly basis Transmission Provider shall provide Interconnection Customer a status report on the construction and installation of Transmission Provider's Interconnection Facilities and Network Upgrades, including, but not limited to, the following information: (1) progress to date; (2) a description of the activities since the last report (3) a description of the action items for the next period; and (4) the delivery status of equipment ordered.
- 24.3 Updated Information Submission by Interconnection Customer. The updated information submission by Interconnection Customer, including manufacturer information, shall occur no later than one hundred eighty (180) Calendar Days prior to the Trial Operation. Interconnection Customer shall submit a completed copy of the Large Generating Facility data requirements contained in Appendix 1 to the LGIP. It shall also include any additional information provided to Transmission Provider for the Definitive Interconnection System Impact Study and Interconnection Facilities Study. Information in this submission shall be the most current Large Generating Facility design or expected performance data. Information submitted for stability models shall be compatible with Transmission Provider standard models. If there is no compatible model, Interconnection

Customer will work with a consultant mutually agreed to by the Parties to develop and supply a standard model and associated information.

If Interconnection Customer's data is materially different from what was originally provided to Transmission Provider pursuant to the Interconnection Study Agreement between Transmission Provider and Interconnection Customer, then Transmission Provider will conduct appropriate studies to determine the impact on Transmission Provider Transmission System based on the actual data submitted pursuant to this Article 24.3. The Interconnection Customer shall not begin Trial Operation until such studies are completed.

24.4 Information Supplementation. Prior to the Operation Date, the Parties shall supplement their information submissions described above in this Article 24 with any and all "as-built" Large Generating Facility information or "as-tested" performance information that differs from the initial submissions or, alternatively, written confirmation that no such differences exist. The Interconnection Customer shall conduct tests on the Large Generating Facility as required by Good Utility Practice such as an open circuit "step voltage" test on the Large Generating Facility to verify proper operation of the Large Generating Facility's automatic voltage regulator.

Unless otherwise agreed, the test conditions shall include: (1) Large Generating Facility at synchronous speed; (2) automatic voltage regulator on and in voltage control mode; and (3) a five percent change in Large Generating Facility terminal voltage initiated by a change in the voltage regulators reference voltage. Interconnection Customer shall provide validated test recordings showing the responses of Large Generating Facility terminal and field voltages. In the event that direct recordings of these voltages is impractical, recordings of other voltages or currents that mirror the response of the Large Generating Facility's terminal or field voltage are acceptable if information necessary to translate these alternate quantities to actual Large Generating Facility terminal or field voltages is provided. Large Generating Facility testing shall be conducted and results provided to Transmission Provider for each individual generating unit in a station. Subsequent to the Operation Date, Interconnection Customer shall provide Transmission Provider any information changes due to equipment replacement, repair, or adjustment. Transmission Provider shall provide Interconnection Customer any information changes due to equipment replacement, repair or adjustment in the directly connected substation or any adjacent Transmission Provider-owned substation that may affect Interconnection Customer's Interconnection Facilities equipment ratings, protection or operating requirements. The Parties shall provide such information no later than thirty (30) Calendar Days after the date of the equipment replacement, repair or adjustment.

Article 25. Information Access and Audit Rights

- 25.1 Information Access. Each Party (the "disclosing Party") shall make available to the other Party information that is in the possession of the disclosing Party and is necessary in order for the other Party to: (i) verify the costs incurred by the disclosing Party for which the other Party is responsible under this LGIA; and (ii) carry out its obligations and responsibilities under this LGIA. The Parties shall not use such information for purposes other than those set forth in this Article 25.1 and to enforce their rights under this LGIA.
- 25.2 Reporting of Non-Force Majeure Events. Each Party (the "notifying Party") shall notify the other Party when the notifying Party becomes aware of its inability to comply with the provisions of this LGIA for a reason other than a Force Majeure event. The Parties agree to cooperate with each other and provide necessary information regarding such inability to comply, including the date, duration, reason for the inability to comply, and corrective actions taken or planned to be taken with respect to such inability to comply. Notwithstanding the foregoing, notification, cooperation or information provided under this article shall not entitle the Party receiving such notification to allege a cause for anticipatory breach of this LGIA.
- 25.3 Audit Rights. Subject to the requirements of confidentiality under Article 22 of this LGIA, each Party shall have the right, during normal business hours, and upon prior reasonable notice to the other Party, to audit at its own expense the other Party's accounts and records pertaining to either Party's performance or either Party's satisfaction of obligations under this LGIA. Such audit rights shall include audits of the other Party's costs, calculation of invoiced amounts, Transmission Provider's efforts to allocate responsibility for the provision of reactive support to the Transmission System, Transmission Provider's efforts to allocate responsibility for interruption or reduction of generation on the Transmission System, and each Party's actions in an Emergency Condition. Any audit authorized by this article shall be performed at the offices where such accounts and records are maintained and shall be limited to those portions of such accounts and records that relate to each Party's performance and satisfaction of obligations under this LGIA. Each Party shall keep such accounts and records for a period equivalent to the audit rights periods described in Article 25.4.

25.4 Audit Rights Periods.

Audit Rights Period for Construction-Related Accounts and Records. Accounts and records related to the design, engineering, procurement, and construction of Transmission Provider's Interconnection Facilities and Network Upgrades shall be subject to

audit for a period of twenty-four months following Transmission Provider's issuance of a final invoice in accordance with Article 12.2.

- Audit Rights Period for All Other Accounts and Records.

 Accounts and records related to either Party's performance or satisfaction of all obligations under this LGIA other than those described in Article 25.4.1 shall be subject to audit as follows: (i) for an audit relating to cost obligations, the applicable audit rights period shall be twenty-four months after the auditing Party's receipt of an invoice giving rise to such cost obligations; and (ii) for an audit relating to all other obligations, the applicable audit rights period shall be twenty-four months after the event for which the audit is sought.
- **25.5 Audit Results**. If an audit by a Party determines that an overpayment or an underpayment has occurred, a notice of such overpayment or underpayment shall be given to the other Party together with those records from the audit which support such determination.

Article 26. Subcontractors

- **26.1 General**. Nothing in this LGIA shall prevent a Party from utilizing the services of any subcontractor as it deems appropriate to perform its obligations under this LGIA; provided, however, that each Party shall require its subcontractors to comply with all applicable terms and conditions of this LGIA in providing such services and each Party shall remain primarily liable to the other Party for the performance of such subcontractor.
- **26.2 Responsibility of Principal**. The creation of any subcontract relationship shall not relieve the hiring Party of any of its obligations under this LGIA. The hiring Party shall be fully responsible to the other Party for the acts or omissions of any subcontractor the hiring Party hires as if no subcontract had been made; provided, however, that in no event shall Transmission Provider be liable for the actions or inactions of Interconnection Customer or its subcontractors with respect to obligations of Interconnection Customer under Article 5 of this LGIA. Any applicable obligation imposed by this LGIA upon the hiring Party shall be equally binding upon, and shall be construed as having application to, any subcontractor of such Party.
- **26.3 No Limitation by Insurance**. The obligations under this Article 26 will not be limited in any way by any limitation of subcontractor's insurance.

Article 27. Disputes

- 27.1 Submission. In the event either Party has a dispute, or asserts a claim, that arises out of or in connection with this LGIA or its performance, such Party (the "disputing Party") shall provide the other Party with written notice of the dispute or claim ("Notice of Dispute"). Such dispute or claim shall be referred to a designated senior representative of each Party for resolution on an informal basis as promptly as practicable after receipt of the Notice of Dispute by the other Party. In the event the designated representatives are unable to resolve the claim or dispute through unassisted or assisted negotiations within thirty (30) Calendar Days of the other Party's receipt of the Notice of Dispute, such claim or dispute may, upon mutual agreement of the Parties, be submitted to arbitration and resolved in accordance with the arbitration procedures set forth below. In the event the Parties do not agree to submit such claim or dispute to arbitration, each Party may exercise whatever rights and remedies it may have in equity or at law consistent with the terms of this LGIA.
- 27.2 **External Arbitration Procedures.** Any arbitration initiated under this LGIA shall be conducted before a single neutral arbitrator appointed by the Parties. If the Parties fail to agree upon a single arbitrator within ten (10) Calendar Days of the submission of the dispute to arbitration, each Party shall choose one arbitrator who shall sit on a three-member arbitration panel. The two arbitrators so chosen shall within twenty (20) Calendar Days select a third arbitrator to chair the arbitration panel. In either case, the arbitrators shall be knowledgeable in electric utility matters, including electric transmission and bulk power issues, and shall not have any current or past substantial business or financial relationships with any party to the arbitration (except prior arbitration). The arbitrator(s) shall provide each of the Parties an opportunity to be heard and, except as otherwise provided herein, shall conduct the arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("Arbitration Rules") and any applicable FERC regulations or RTO rules; provided, however, in the event of a conflict between the Arbitration Rules and the terms of this Article 27, the terms of this Article 27 shall prevail.
- 27.3 Arbitration Decisions. Unless otherwise agreed by the Parties, the arbitrator(s) shall render a decision within ninety (90) Calendar Days of appointment and shall notify the Parties in writing of such decision and the reasons therefor. The arbitrator(s) shall be authorized only to interpret and apply the provisions of this LGIA and shall have no power to modify or change any provision of this Agreement in any manner. The decision of the arbitrator(s) shall be final and binding upon the Parties, and judgment on the award may be entered in any court having jurisdiction. The decision of the arbitrator(s) may be appealed solely on the grounds that the conduct of the arbitrator(s), or the decision itself, violated the

standards set forth in the Federal Arbitration Act or the Administrative Dispute Resolution Act. The final decision of the arbitrator must also be filed with FERC if it affects jurisdictional rates, terms and conditions of service, Interconnection Facilities, or Network Upgrades.

27.4 Costs. Each Party shall be responsible for its own costs incurred during the arbitration process and for the following costs, if applicable: (1) the cost of the arbitrator chosen by the Party to sit on the three member panel and one half of the cost of the third arbitrator chosen; or (2) one half the cost of the single arbitrator jointly chosen by the Parties.

Article 28. Representations, Warranties, and Covenants

- **28.1 General.** Each Party makes the following representations, warranties and covenants:
 - 28.1.1 Good Standing. Such Party is duly organized, validly existing and in good standing under the laws of the state in which it is organized, formed, or incorporated, as applicable; that it is qualified to do business in the state or states in which the Large Generating Facility, Interconnection Facilities and Network Upgrades owned by such Party, as applicable, are located; and that it has the corporate power and authority to own its properties, to carry on its business as now being conducted and to enter into this LGIA and carry out the transactions contemplated hereby and perform and carry out all covenants and obligations on its part to be performed under and pursuant to this LGIA.
 - **Authority**. Such Party has the right, power and authority to enter into this LGIA, to become a Party hereto and to perform its obligations hereunder. This LGIA is a legal, valid and binding obligation of such Party, enforceable against such Party in accordance with its terms, except as the enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting creditors' rights generally and by general equitable principles (regardless of whether enforceability is sought in a proceeding in equity or at law).
 - **No Conflict**. The execution, delivery and performance of this LGIA does not violate or conflict with the organizational or formation documents, or bylaws or operating agreement, of such Party, or any judgment, license, permit, order, material agreement or instrument applicable to or binding upon such Party or any of its assets.

28.1.4 Consent and Approval. Such Party has sought or obtained, or, in accordance with this LGIA will seek or obtain, each consent, approval, authorization, order, or acceptance by any Governmental Authority in connection with the execution, delivery and performance of this LGIA, and it will provide to any Governmental Authority notice of any actions under this LGIA that are required by Applicable Laws and Regulations.

Article 29. Joint Operating Committee

- 29.1 Joint Operating Committee. Except in the case of ISOs and RTOs, Transmission Provider shall constitute a Joint Operating Committee to coordinate operating and technical considerations of Interconnection Service. At least six (6) months prior to the expected Initial Synchronization Date, Interconnection Customer and Transmission Provider shall each appoint one representative and one alternate to the Joint Operating Committee. Each Interconnection Customer shall notify Transmission Provider of its appointment in writing. Such appointments may be changed at any time by similar notice. The Joint Operating Committee shall meet as necessary, but not less than once each calendar year, to carry out the duties set forth herein. The Joint Operating Committee shall hold a meeting at the request of either Party, at a time and place agreed upon by the representatives. The Joint Operating Committee shall perform all of its duties consistent with the provisions of this LGIA. Each Party shall cooperate in providing to the Joint Operating Committee all information required in the performance of the Joint Operating Committee's duties. All decisions and agreements, if any, made by the Joint Operating Committee, shall be evidenced in writing. The duties of the Joint Operating Committee shall include the following:
 - **29.1.1** Establish data requirements and operating record requirements.
 - 29.1.2 Review the requirements, standards, and procedures for data acquisition equipment, protective equipment, and any other equipment or software.
 - 29.1.3 Annually review the one (1) year forecast of maintenance and planned outage schedules of Transmission Provider's and Interconnection Customer's facilities at the Point of Interconnection.
 - 29.1.4 Coordinate the scheduling of maintenance and planned outages on the Interconnection Facilities, the Large Generating Facility and other facilities that impact the normal operation of the interconnection of the Large Generating Facility to the Transmission

System.

- 29.1.5 Ensure that information is being provided by each Party regarding equipment availability.
- 29.1.6 Perform such other duties as may be conferred upon it by mutual agreement of the Parties.

Article 30. Miscellaneous

- **30.1 Binding Effect**. This LGIA and the rights and obligations hereof, shall be binding upon and shall inure to the benefit of the successors and assigns of the Parties hereto.
- **30.2** Conflicts. In the event of a conflict between the body of this LGIA and any attachment, appendices or exhibits hereto, the terms and provisions of the body of this LGIA shall prevail and be deemed the final intent of the Parties.
- Rules of Interpretation. This LGIA, unless a clear contrary intention appears, 30.3 shall be construed and interpreted as follows: (1) the singular number includes the plural number and vice versa; (2) reference to any person includes such person's successors and assigns but, in the case of a Party, only if such successors and assigns are permitted by this LGIA, and reference to a person in a particular capacity excludes such person in any other capacity or individually; (3) reference to any agreement (including this LGIA), document, instrument or tariff means such agreement, document, instrument, or tariff as amended or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof; (4) reference to any Applicable Laws and Regulations means such Applicable Laws and Regulations as amended, modified, codified, or reenacted, in whole or in part, and in effect from time to time, including, if applicable, rules and regulations promulgated thereunder; (5) unless expressly stated otherwise, reference to any Article, Section or Appendix means such Article of this LGIA or such Appendix to this LGIA, or such Section to the LGIP or such Appendix to the LGIP, as the case may be; (6) "hereunder", "hereof", "herein", "hereto" and words of similar import shall be deemed references to this LGIA as a whole and not to any particular Article or other provision hereof or thereof; (7) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding such term; and (8) relative to the determination of any period of time, "from" means "from and including", "to" means "to but excluding" and "through" means "through and including".
- **30.4** Entire Agreement. This LGIA, including all Appendices and Schedules attached hereto, constitutes the entire agreement between the Parties with reference to the

subject matter hereof, and supersedes all prior and contemporaneous understandings or agreements, oral or written, between the Parties with respect to the subject matter of this LGIA. There are no other agreements, representations, warranties, or covenants which constitute any part of the consideration for, or any condition to, either Party's compliance with its obligations under this LGIA.

- **30.5 No Third Party Beneficiaries**. This LGIA is not intended to and does not create rights, remedies, or benefits of any character whatsoever in favor of any persons, corporations, associations, or entities other than the Parties, and the obligations herein assumed are solely for the use and benefit of the Parties, their successors in interest and, where permitted, their assigns.
- **30.6 Waiver**. The failure of a Party to this LGIA to insist, on any occasion, upon strict performance of any provision of this LGIA will not be considered a waiver of any obligation, right, or duty of, or imposed upon, such Party.
 - Any waiver at any time by either Party of its rights with respect to this LGIA shall not be deemed a continuing waiver or a waiver with respect to any other failure to comply with any other obligation, right, duty of this LGIA. Termination or Default of this LGIA for any reason by Interconnection Customer shall not constitute a waiver of Interconnection Customer's legal rights to obtain an interconnection from Transmission Provider. Any waiver of this LGIA shall, if requested, be provided in writing.
- **30.7 Headings**. The descriptive headings of the various Articles of this LGIA have been inserted for convenience of reference only and are of no significance in the interpretation or construction of this LGIA.
- **30.8 Multiple Counterparts**. This LGIA may be executed in two or more counterparts, each of which is deemed an original but all constitute one and the same instrument.
- **30.9 Amendment**. The Parties may by mutual agreement amend this LGIA by a written instrument duly executed by the Parties.
- **30.10 Modification by the Parties**. The Parties may by mutual agreement amend the Appendices to this LGIA by a written instrument duly executed by the Parties. Such amendment shall become effective and a part of this LGIA upon satisfaction of all Applicable Laws and Regulations.
- **30.11 Reservation of Rights**. Transmission Provider shall have the right to make a unilateral filing with FERC to modify this LGIA with respect to any rates, terms and conditions, charges, classifications of service, rule or regulation under section

205 or any other applicable provision of the Federal Power Act and FERC's rules and regulations thereunder, and Interconnection Customer shall have the right to make a unilateral filing with FERC to modify this LGIA pursuant to section 206 or any other applicable provision of the Federal Power Act and FERC's rules and regulations thereunder; provided that each Party shall have the right to protest any such filing by the other Party and to participate fully in any proceeding before FERC in which such modifications may be considered. Nothing in this LGIA shall limit the rights of the Parties or of FERC under sections 205 or 206 of the Federal Power Act and FERC's rules and regulations thereunder, except to the extent that the Parties otherwise mutually agree as provided herein.

30.12 No Partnership. This LGIA shall not be interpreted or construed to create an association, joint venture, agency relationship, or partnership between the Parties or to impose any partnership obligation or partnership liability upon either Party. Neither Party shall have any right, power or authority to enter into any agreement or undertaking for, or act on behalf of, or to act as or be an agent or representative of, or to otherwise bind, the other Party.

IN WITNESS WHEREOF, the Parties have executed this LGIA in duplicate originals, each of which shall constitute and be an original effective Agreement between the Parties.

[Insert name of Transmission Provider or Transmission Owner, if applicable]			
By:	By:		
Title:	Title:		
Date:	Date:		
[Insert name of Interconnection Customer]			
By:			
Title:			
Date:			

Appendices: Interconnection Facilities, Network Upgrades and Distribution Upgrade

Appendices to LGIA

Appendix A Interconnection Facilities, Network Upgrades, and Distribution

Upgrades

Appendix B Milestones

Appendix C Interconnection Details

Appendix D Security Arrangements Details

Appendix E Commercial Operation Date

Appendix F Addresses for Delivery of Notices and Billings

Appendix G Interconnection Requirements for a Wind Generating Plant

Appendix A: Interconnection Facilities, Network Upgrades and Distribution **Upgrades**

Appendix A to LGIA

Interconnection Facilities, Network Upgrades and Distribution Upgrades

- **Interconnection Facilities:** 1.
 - (a) [insert Interconnection Customer's Interconnection Facilities]:
 - (b) [insert Transmission Provider's Interconnection Facilities]:
- **Network Upgrades:** 2.
 - (a) [insert Stand Alone Network Upgrades]:
 - (b) [insert Other Network Upgrades]:
- **Distribution Upgrades:** 3.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Appendix B: Milestones

Appendix B to LGIA
Milestones

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Appendix C: Interconnection Details

Appendix C to LGIA

Interconnection Details

Appendix D: Security Arrangements Details

Appendix D to LGIA

Security Arrangements Details

Infrastructure security of Transmission System equipment and operations and control hardware and software is essential to ensure day-to-day Transmission System reliability and operational security. FERC will expect all entities, including Transmission Providers, market participants, and Interconnection Customers interconnected to the Transmission System to comply with the Reliability Standards promulgated by FERC and administered by the North American Electric Reliability Corporation (NERC) and associated Regional Reliability Organizations (RRO). In EPE's service territory, the RRO is the Western Electricity Coordinating Council (WECC).

Appendix E: Commercial Operation Date

Appendix E to LGIA

Commercial Operation Date

This Appendix E is a part of the LGIA between Transmission Provider and Interconnection Customer.

	[Date]		
I	[Transmission Provider Address]		
]	Re: Large Generating Facility		
]	Dear:		
On [Date] [Interconnection Customer] has completed Trial Operation of Unit No at the Large Generating Facility, effective as of [Date pone day].			
	Thank you.		
	[Signature]		
	[Interconnection Customer Representative]		

Appendix F: Addresses for Delivery of Notices and Billings

Appendix F to LGIA

	Addresses for Delivery of Notices and Billings
Notic	es:
	<u>Transmission Provider</u> :
	[To be supplied.]
	Interconnection Customer:
	[To be supplied.]
Billin	gs and Payments:
	<u>Transmission Provider:</u>
	[To be supplied.]
	Interconnection Customer:
	[To be supplied.]
Alter	native Forms of Delivery of Notices (telephone, facsimile or email):
	<u>Transmission Provider</u> :
	[To be supplied.]
	Interconnection Customer:
	[To be supplied.]

Appendix G: Interconnection Requirements for a Wind Generating Plant

Appendix G to LGIA

Interconnection Requirements for a

Wind Generating Plant

Appendix G sets forth requirements and provisions specific to a wind generating plant. All other requirements of this LGIA continue to apply to wind generating plant interconnections.

A. Technical Standards Applicable to a Wind Generating Plant

i. Low Voltage Ride-Through (LVRT) Capability

A wind generating plant shall be able to remain online during voltage disturbances up to the time periods and associated voltage levels set forth in the standard below. The LVRT standard provides for a transition period standard and a post-transition period standard.

Transition Period LVRT Standard

The transition period standard applies to wind generating plants subject to FERC Order 661 that have either: (i) interconnection agreements signed and filed with the Commission, filed with the Commission in unexecuted form, or filed with the Commission as non-conforming agreements between January 1, 2006 and December 31, 2006, with a scheduled in-service date no later than December 31, 2007, or (ii) wind generating turbines subject to a wind turbine procurement contract executed prior to December 31, 2005, for delivery through 2007.

1. Wind generating plants are required to remain in-service during three-phase faults with normal clearing (which is a time period of approximately 4 - 9 cycles) and single line to ground faults with delayed clearing, and subsequent post-fault voltage recovery to prefault voltage unless clearing the fault effectively disconnects the generator from the system. The clearing time requirement for a three-phase fault will be specific to the wind generating plant substation location, as determined by and documented by the Transmission Provider. The maximum clearing time the wind generating plant shall be required to withstand for a three-phase fault shall be 9 cycles at a voltage as low as 0.15 p.u., as measured at the high side of the wind generating plant step-up transformer (i.e. the transformer that steps the voltage up to the transmission interconnection voltage or "GSU"), after which, if the fault remains following the location-specific normal clearing time for three-phase faults, the wind generating plant may disconnect from the transmission system.

- This requirement does not apply to faults that would occur between the wind 2. generator terminals and the high side of the GSU or to faults that would result in a voltage lower than 0.15 per unit on the high side of the GSU serving the facility.
- 3. Wind generating plants may be tripped after the fault period if this action is intended as part of a special protection system.
- 4. Wind generating plants may meet the LVRT requirements of this standard by the performance of the generators or by installing additional equipment (e.g., Static VAr Compensator, etc.) within the wind generating plant or by a combination of generator performance and additional equipment.

5. Existing individual generator units that are, or have been, interconnected to the network at the same location at the effective date of the Appendix G LVRT Standard are exempt from meeting the Appendix G LVRT Standard for the remaining life of the existing generation equipment. Existing individual generator units that are replaced are required to meet the Appendix G LVRT Standard.

Post-transition Period LVRT Standard

All wind generating plants subject to FERC Order No. 661 and not covered by the transition period described above must meet the following requirements:

Wind generating plants are required to remain in-service during three-phase faults 1. with normal clearing (which is a time period of approximately 4 - 9 cycles) and single line to ground faults with delayed clearing, and subsequent post-fault voltage recovery to prefault voltage unless clearing the fault effectively disconnects the generator from the system. The clearing time requirement for a three-phase fault will be specific to the wind generating plant substation location, as determined by and documented by the Transmission Provider. The maximum clearing time the wind generating plant shall be required to withstand for a three-phase fault shall be 9 cycles after which, if the fault remains following the location-specific normal clearing time for three-phase faults, the wind generating plant may disconnect from the transmission system. A wind generating plant shall remain interconnected during such a fault on the transmission system for a voltage level as low as zero volts, as measured at the high voltage side of the wind GSU.

- 2. This requirement does not apply to faults that would occur between the wind generator terminals and the high side of the GSU.
- 3. Wind generating plants may be tripped after the fault period if this action is intended as part of a special protection system.
- 4. Wind generating plants may meet the LVRT requirements of this standard by the performance of the generators or by installing additional equipment (e.g., Static VAr Compensator) within the wind generating plant or by a combination of generator performance and additional equipment.
- 5. Existing individual generator units that are, or have been, interconnected to the network at the same location at the effective date of the Appendix G LVRT Standard are exempt from meeting the Appendix G LVRT Standard for the remaining life of the existing generation equipment. Existing individual generator units that are replaced are required to meet the Appendix G LVRT Standard.

ii. Power Factor Design Criteria (Reactive Power)

The following reactive power requirements apply only to a newly interconnecting wind generating plant that has executed a Facilities Study Agreement as of the effective date of the Final Rule establishing the reactive power requirements for non-synchronous generators in section 9.6.1 of this LGIA (Order No. 827). A wind generating plant to which this provision applies shall maintain a power factor within the range of 0.95 leading to 0.95 lagging, measured at the Point of Interconnection as defined in this LGIA, if the Transmission Provider's System Impact Study shows that such a requirement is necessary to ensure safety or reliability. The power factor range standard can be met by

using, for example, power electronics designed to supply this level of reactive capability (taking into account any limitations due to voltage level, real power output, etc.) or fixed and switched capacitors if agreed to by the Transmission Provider, or a combination of the two. The Interconnection Customer shall not disable power factor equipment while the wind plant is in operation. Wind plants shall also be able to provide sufficient dynamic voltage support in lieu of the power system stabilizer and automatic voltage regulation at the generator excitation system if the System Impact Study shows this to be required for system safety or reliability.

iii. Supervisory Control and Data Acquisition (SCADA) Capability

The wind plant shall provide SCADA capability to transmit data and receive instructions from the Transmission Provider to protect system reliability. The Transmission Provider and the wind plant Interconnection Customer shall determine what SCADA information is essential for the proposed wind plant, taking into account the size of the plant and its characteristics, location, and importance in maintaining generation resource adequacy and transmission system reliability in its area.

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Document Accession #: 20230531-5351 Filed Date: 05/31/2023

Document Content(s)	
Transmittal Letter_Attachment M 2023 05 31.pdf	.1
Appendix A to Transmittal Letter.docx	. 8
Marked183-27a47792-3a06-4ab3-b78a-c1c04fac421d.pdf	44
Clean183-82cbe1bb-6bf7-4a2d-adf2-e16c6b1f0d9c.pdf	01
FERC GENERATED TARIFF FILING.rtf5	28